

Identifying the psychosocial needs of Emirati and expatriate breast cancer survivors in the U.A.E.: A mixed method investigation in a hospital setting

Thesis

For the degree of Doctor of Philosophy

At the University of Salford

By

Maria Aamir

Submitted on 20th January 2022

School of Health and Society

University of Salford



CONTENTS

CONTENTS 2			
LIST OF FIGURES			
LIST OF TA	LIST OF TABLES		
ACKNOWI	ACKNOWLEDGEMENTS		
DEDICATIONS			
DECLARATION 1			
ABSTRACT 1		15	
Chapter 1:]	INTRODUCTION	17	
1.1 RE	SEARCH QUESTION	17	
1.2 RE	SEARCH PROBLEM	17	
1.3 AI	MS AND OBJECTIVES	19	
1.4 SIG	GNIFICANCE OF THE STUDY	20	
1.5 OF	GANISATION OF THESIS	21	
1.6 BA	CKGROUND	23	
1.6.1	UNITED ARAB EMIRATES POPULATION	23	
1.6.2	UAE HEALTHCARE SYSTEM	25	
1.6.3	ABU DHABI HEALTHCARE SYSTEM	25	
1.6.4	CANCER DEFINITION AND BIOLOGY	27	
1.6.5	CANCER DIAGNOSIS	27	
1.6.6	CANCER – A GLOBAL PROBLEM	28	
1.6.7	BREAST CANCER EPIDEMIOLOGY	29	
1.6.8	CANCER SURVIVOR AND CANCER SURVIVORSHIP	30	
1.6.9	CANCER AND QUALITY OF LIFE (QOL)	31	
1.6.10	UNDERSTANDING CANCER AND CULTURE	32	



	1.6.	11	CANCER SURVIVORSHIP CARE IN THE UAE	36
	1.7	PE	RSONAL EXPERIENCE	38
С	hapter	2:]	LITERATURE REVIEW	39
	2.1	Lľ	TERATURE SEARCH STRATEGY	39
	2.2	TH	IEORETICAL FRAMEWORK	40
	2.3	GA	APS IN THE CURRENT LITERATURE	59
	2.3.	1	EVIDENCE-BASED HIGH-QUALITY CANCER CARE	60
	2.3.	2	CANCER SURVIVORSHIP CARE	64
	2.3.	3	PSYCHOSOCIAL NEEDS OF CANCER SURVIVORS	67
	2.3.	4	CANCER NEED ASSESSMENT INSTRUMENTS	75
	2.3.	5	CANCER ASSOCIATION WITH CULTURE	81
	2.3.	6	PSYCHOSOCIAL NEEDS OF CANCER SURVIVORS IN THE	E UNITED
	ARA	AB	EMIRATES	84
	2.4	CA	NCER DIAGNOSIS & SURVIVORSHIP	85
	2.5	BR	EAST CANCER SURVIVORSHIP	87
	2.6	BR	EAST CANCER SURVIVORS' CONCERNS	90
	2.7	KE	EY DOMAINS OF BREAST CANCER SURVIVORS' CONCERNS	95
	2.7.	1	PHYSICAL CONCERNS OF BREAST CANCER SURVIVORS	96
	2.7. SUI	2 RVI	INFORMATION OR EDUCATIONAL CONCERNS OF BREAST VORS	CANCER 101
	2.7. SUI	3 RVI	SOCIAL AND FINANCIAL CONCERNS OF BREAST VORS	CANCER 109
	2.7.	4	EMOTIONAL CONCERNS OF BREAST CANCER SURVIVORS	115
	2.7.	5	SPIRITUAL CONCERNS OF BREAST CANCER SURVIVORS	119
	2.8	DI	SPARITIES IN BREAST CANCER SURVIVORSHIP	122
С	Chapter	3: 1	METHODOLOGY	131
	3.1	PH	ILOSOPHICAL ORIENTATION OF RESEARCH DESIGN	131



3	.2	OVERVIEW OF MIXED METHODS	133
3	.3	JUSTIFICATION OF SELECTED METHOD	138
3	.4	DATA INTEGRATION	142
3	.5	PHASE ONE: QUANTITATIVE METHODOLOGY	144
	3.5.	.1 ETHICAL APPROVAL	144
	3.5.	.2 PARTICIPANTS & SETTING	146
	3.5.	.3 RESPONSE RATE OF THE SURVEY	148
	3.5.	.4 INSTRUMENT	148
	3.5.	.5 DATA COLLECTION	149
	3.5.	.6 QUANTITATIVE DATA ANALYSIS	151
3	.6	PHASE TWO: QUALITATIVE METHODOLOGY	151
	3.6.	.1 ETHICAL APPROVAL	151
	3.6.	.2 PARTICIPANTS & SETTING	153
	3.6.	.3 INTERVIEW GUIDE	155
	3.6.	.4 DATA COLLECTION	156
	3.7	QUALITATIVE DATA ANALYSIS METHODS – AN OVERVIEW	160
Cha	pter	4: RESULTS	178
4	.1 DE	ESCRIPTIVE DATA ANALYSIS	179
	4.1.	1 DEMOGRAPHIC CHARACTERSTICS OF BREAST CANCER SUB 180	VIVORS
	4.1.	.2 BREAST CANCER SURVIVORS' (a) PREFERRED LEARNING Se	OURCES,
	(b) '	TOPICS OF INTEREST & (c) PRIMARY SOURCE OF STRENGTH (SE	CTION 9-
	11)	183	
	Prin	mary source of strength	196
	4.1.	.3 MOST PREVELANT PSYCHOSCOIAL CONCERNS OF BREAST	CANCER
	- 501	K V I V UKS	19/



4.1.4 MOST PREVALENT PSYCHOSOCIAL CONCERNS OF BREAST CANCER
SURVIVORS BY YEAR OF DIAGNOSIS200
4.1.5 SEVERITYOFBREASTCANCERSURVIVORS'PSYCHOSOCIALCONCERNS202
4.1.6ASSOCIATIONBETWEENBREASTCANCERSURVIVORS'(A)NATIONALITY & AGE (B)NATIONALITY & STAGE OF CANCER207
4.1.7INFLUENCE OF NATIONALITY ON PSYCHOSOCIAL NEEDS SCORESOF BREAST CANCER SURVIVORS214
4.1.8 CORRELATION BETWEEN QUALITY OF LIFE & PSYCHOSOCIAL
CONCERNS OF BREAST CANCER SURVIVORS 218
4.1.9 RELATIONSHIP BETWEEN (a) DEMOGRAPHIC FACTORS & PSYCHOSOCIAL CONCERNS, (b) PSYCHOSOCIAL CONCERNS & QUALITY OF
LIFE OF BREAST CANCER SURVIVORS 219
4.2 QUALITATIVE DATA ANALYSIS 231
4.2.1 INTRODUCTION 231
4.2.2 THEMATIC ANALYSIS 234
4.2.3 SUMMARY OF QUANTITATIVE & QUALITATIVE RESULTS 281
Chapter 5: DISCUSSION 285
5.1 DEMOGRAPHICS AND CLINICAL CHARACTERISTICS OF BREAST
CANCER SURVIVORS 286
5.2DIAGNOSIS AND CANCER EXPERIENCE287
5.3ACCESS TO CARE289
5.4HEALTH DECISION-MAKING291
5.5THE ROLE OF SUPPORT FORUMS293
5.6PSYCHOSOCIAL CONCERNS OF BREAST CANCER SURVIVORS294
5.6.1PHYSICAL CONCERNS OF BREAST CANCER SURVIVORS295
5.6.2 INFORMATION CONCERNS OF BREAST CANCER SURVIVORS 299



5.0.5 Sound find find the concentry of bitters	Γ CANCER
SURVIVORS	302
5.6.4 EMOTIONAL CONCERNS OF BREAST CANCER SURVIVOR	as 304
5.6.5 SPIRITUAL OR RELIGIOUS CONCERNS OF BREAST	Г CANCER
SURVIVORS	307
5.6.6 QUALITY OF LIFE OF BREAST CANCER SURVIVORS	310
5.7 REFLECTIONS	311
5.8 STRENGTHS AND LIMITATIONS	315
5.9 IMPLICATIONS	318
Chapter 6: CONCLUSIONS	324
Chapter 7: REFERENCES	326
Chapter 8: APPENDICES	490
APPENDIX 1a: GENERAL CONSENT FOR HEALTHCARE SERVICES -	ORIGINAL
	492
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI	ERVICES -
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED	ERVICES - 497
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO	ERVICES - 497 CONTACT
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO PATIENTS THROUGH THE REGISTRY BY TELEPHONE	ERVICES - 497 CONTACT 498
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO PATIENTS THROUGH THE REGISTRY BY TELEPHONE APPENDIX 2a: IRB APPROVAL – TAWAM HOSPITAL	ERVICES - 497 CONTACT 498 499
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO PATIENTS THROUGH THE REGISTRY BY TELEPHONE APPENDIX 2a: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2b: IRB APPROVAL – TAWAM HOSPITAL	ERVICES - 497 CONTACT 498 499 500
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO PATIENTS THROUGH THE REGISTRY BY TELEPHONE APPENDIX 2a: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2b: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2c: IRB APPROVAL – UNIVERSITY OF SALFORD	ERVICES - 497 CONTACT 498 499 500 501
APPENDIX 16: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO PATIENTS THROUGH THE REGISTRY BY TELEPHONE APPENDIX 2a: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2b: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2c: IRB APPROVAL – UNIVERSITY OF SALFORD APPENDIX 3a & 3b: CONSENT FORM (ENGLISH)	ERVICES - 497 CONTACT 498 499 500 501 502
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO PATIENTS THROUGH THE REGISTRY BY TELEPHONE APPENDIX 2a: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2b: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2c: IRB APPROVAL – UNIVERSITY OF SALFORD APPENDIX 3a & 3b: CONSENT FORM (ENGLISH) APPENDIX 3b: CONSENT FORM - ARABIC	ERVICES - 497 CONTACT 498 499 500 501 502 504
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO PATIENTS THROUGH THE REGISTRY BY TELEPHONE APPENDIX 2a: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2b: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2c: IRB APPROVAL – UNIVERSITY OF SALFORD APPENDIX 3a & 3b: CONSENT FORM (ENGLISH) APPENDIX 3b: CONSENT FORM - ARABIC APPENDIX 4a: PATIENT INFORMATION SHEET - ENGLISH	ERVICES - 497 CONTACT 498 499 500 501 502 504 506
APPENDIX 16: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO PATIENTS THROUGH THE REGISTRY BY TELEPHONE APPENDIX 2a: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2b: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2c: IRB APPROVAL – UNIVERSITY OF SALFORD APPENDIX 3a & 3b: CONSENT FORM (ENGLISH) APPENDIX 3b: CONSENT FORM - ARABIC APPENDIX 4a: PATIENT INFORMATION SHEET - ENGLISH APPENDIX 4b: PATIENT INFORMATION SHEET – ARABIC	ERVICES - 497 CONTACT 498 499 500 501 502 504 504 506 511
APPENDIX 15: GENERAL CONSENT FOR HEALTHCARE SI HIGHLIGHTED APPENDIX 1c: EMAIL CONFIRMING AUTHORIZATION TO PATIENTS THROUGH THE REGISTRY BY TELEPHONE APPENDIX 2a: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2b: IRB APPROVAL – TAWAM HOSPITAL APPENDIX 2c: IRB APPROVAL – UNIVERSITY OF SALFORD APPENDIX 3a & 3b: CONSENT FORM (ENGLISH) APPENDIX 3b: CONSENT FORM (ENGLISH) APPENDIX 4a: PATIENT INFORMATION SHEET - ENGLISH APPENDIX 4b: PATIENT INFORMATION SHEET – ARABIC APPENDIX 5a: CANCER SURVIVORS' NEED SURVEY – ENGLISH	ERVICES - 497 CONTACT 498 499 500 501 502 504 504 506 511 515



APPENDIX 6: DATA COLLECTION FORM CANCER REGISTRY	531
APPENDIX 7: RISK ASSESSMENT FORM	533
APPENDIX 8: RISK ASSESSMENT FORM	537
APPENDIX 9: SURVIVORSHIP CARE PLAN	539
APPENDIX 10: QUANTITATIVE DATA EXTRACTION FILES	541
APPENDIX 11a: QUALITATIVE INTERVIEWS – COMPILED TRANSCRIPTS	589
APPENDIX 11b: SELECTED FIELD NOTES FROM PARTICIPANTS INTERVIE	EWS
	639
APPENDIX 11c: CODING & THEMES	641
APPENDIX 12: GLOSSARY OF TERMS USED (ABBREVIATIONS)	744
APPENDIX 13: MAYO CLINIC APPROVAL FOR THE USAGE OF SURVEY	747



LIST OF FIGURES

Figure 1: Map of United Arab Emirates24
Figure 2: United Arab Emirates Population Pyramid25
Figure 3: Maslow's hierarchy of needs in breast cancer context
Figure 4: Potential long-term and late effects of treatment and how they affect cancer survivors
Figure 5: Phases of Cancer Survivorship
Figure 6: Understanding Cancer Pathway90
Figure 7: Key domains of Breast Cancer Survivorship97
Figure 8: Effective patient – provider communication & shared decision making
Figure 9: Peltonen, 1996 three level of social support system
Figure 10: Explanatory Design140
Figure 11: Flowchart of Sequential Explanatory mixed method design
Figure 12: Phases of thematic analysis (adapted from Braun & Clark's 2006)174
Figure 13: Relationship between demographic factors & psychosocial concerns
Figure 14: Relationship between psychosocial concerns & overall QOL231
Figure 15: Thematic Map: Psychosocial needs of Breast Cancer Survivors
Figure 16: Creation of Themes
Figure 17: Theme One: Living experience with Breast Cancer - Thematic map



LIST OF TABLES

TABLE 2.1: Comparison of Psychometric properties, design, strengths & weaknesses ofcommonly used instruments to assess cancer survivors' needs
TABLE 1: summary results of demographic analysis & clinical characteristics of breast cancer survivors
TABLE 3: Breast cancer survivors' preferred learning source about social concerns 188
TABLE 4: Breast cancer survivors' preferred learning source about emotional concerns 190
TABLE 5: Breast cancer survivors' preferred learning source about spiritual or religious concerns
TABLE 6: Breast cancer survivors' preferred topic of learning interest about healthy living
TABLE 7 : Breast cancer survivors' preferred topic of learning interest about financial concerns 196
TABLE 8 : Breast cancer survivors' preferred topic of learning interest about enhancing communication
TABLE 9 : Breast cancer survivors' preferred topic of learning interest about health screening
TABLE 10 : Breast cancer survivors' primary source of strength
TABLE 11: Summary Results of breast cancer survivors most prevalent psychosocial concerns 202
TABLE 12 : Breast cancer survivors most prevalent concerns by year of diagnosis
TABLE 13 : Severity of breast cancer survivors' Physical concerns 205
TABLE 14: Severity of breast cancer survivors' information concerns 207
TABLE 15: Severity of breast cancer survivors' social & financial concerns 208
TABLE 16: Severity of breast cancer survivors' emotional concerns 209
TABLE 17: Severity breast cancer of survivors' religious/spiritual concerns 210
TABLE 18a : Breast cancer survivors' age by nationality 212
TABLE 18b : Hypothesis 1 – Chi-Square Tests 212
TABLE 18c : 18c: Hypothesis 1 – phi Coefficient
TABLE 19a : Breast cancer survivors stage by nationality



TABLE 19b: Hypothesis 2 – Z TEST
TABLE 19c: Hypothesis 2 – phi Coefficient
TABLE 20a: Summary of differences in breast cancer survivors psychosocial concerns scores between Emiratis and expatriate 218
TABLE 20b : Difference in physical concerns between emiratis and expatriates breast cancer survivors
TABLE 20c: Difference in information concerns between emiratis and expatriates breast cancer survivors 219
TABLE 20d : Difference in social/financial concerns between emiratis and expatriates breast cancer survivors
TABLE 20e: Influence of nationality on emotional concerns breast cancer survivors220
TABLE 20f: Difference in spiritual/religious concerns between emiratis and expatriates breast cancer survivors
TABLE 21 : Correlation between quality of life and psychosocial concerns scoring of breast cancer survivors 222
TABLE 22a: Relationship between demographic factors and physical concerns of breast cancer survivors 225
TABLE 22b: Relationship between demographic factor and information concerns of breast cancer survivors
TABLE 22c : Relationship between demographic factors and social & financial concerns of breast cancer survivors 229
TABLE 22d : Relationship between demographic factors and emotional concerns of breast cancer survivors 230
TABLE 22e : Relationship between demographic factors and religious or spiritual concerns of breast cancer survivors
TABLE 23a : Relationship between psychosocial concerns and overall quality of life of breast cancer survivors 233
TABLE 24: Breast Cancer Survivors' demographics in Qualitative Phase 236



ACKNOWLEDGEMENTS

First and foremost, I would like to thank my supervisors Dr. Linda Dubrow-Marshall and Dr. Rod Dubrow-Marshall for the opportunity to learn throughout the PhD experience and to be the student again during this academic journey. I learned so much more during this research than my entire learning career in the past. I am very grateful for the knowledge they shared with me, the wisdom and the constructive guidance to help me learn and improve constantly to support and train me as a researcher. I would like to greatly appreciate my supervisors for the encouragement, support and constant feedback that they provided regardless of time differences between UAE and UK. Thank you for believing in my project and trusting me that I can add value to mankind and to be the voice of vulnerable cancer patients. Some of the patients were listened to for the very first time; thank you for helping me to be their voice. Your trust in me was a great motivation for me to work hard and find ways to remind myself that it is not just a PhD, it is something I will continue to work on in coming years and will continue try to make a difference in healthcare.

I would like to thank my husband who is not just my friend but a mentor too who has always respected my passion for the PhD and ensured that I do well providing me the time and resources that I needed to continue my studies. You have motivated me the most that anyone could have. Thank you for supporting me in whatever way possible to move this journey forward. Thank you for being there and for accomplishing my dreams. Your support and care have been the greatest push for me to move forward. You made this process gratifying rather daunting. Thank you for your trust and support; you have listened to me the most and helped me in whatever way during many crises. Your presence is no less than a blessing. I counted on you throughout this time. Thank you for being there for me every single day and part of this special journey in my life.

Thank you to my three children (Zoya Aamir, Zara Aamir and Zyan Ali) who understood the fact that mum works everyday even on weekends to keep her commitments to achieve her goals; thank you for putting up with my busy schedule. I owe them their time and their sheer understanding. It was quite challenging for all of us, but my family supported me truly without complaining. It wasn't possible without their understanding, support and love.



Most special thanks to my sister Sheena Hassan, you always have encouraged me spiritually and religiously especially towards the end so I could finally achieve my goals. Thank you for the love and support. Your words, advice and wisdom, mainly religious motivation, have been the most influential that I believe in myself, and I believe that goodness exists on earth. Thank you for listening to me and my issues while making me laugh every day.

Final thanks to all my colleagues at work who helped me in whatever way possible. Without the support of the above-mentioned individuals, it would not have been possible to manage my studies, work, and life and to achieve my goals. I thank you all from the bottom of my heart. I also would like to thank all the participants who were generous enough to give me time and talked truthfully about their concerns. I really hope this work could add a positive difference in all cancer patients' lives.



DEDICATIONS

For my mother from whom I have inherited learning and education.

For my father, from whom I have inherited determination and sustainability to work hard and achieve goals.

For my sister, who inspired me to be kind.

For my husband, for his patience, his advice and his faith in me, because he always understood me.

For all cancer patients, who taught me how to be resilient in life.



DECLARATION

I declare that the work in this dissertation was carried out in accordance with the requirements of the University of Salford's Regulations and Code of Practice for PhD Degree Programmes and does not incorporate without acknowledgment any material previously submitted for any other academic award in any university; and that to the best of my knowledge and belief it does not contain any material previously published or written by another person except where indicated by specific references in the text. This thesis is the candidate's own work and any work done in collaboration with, or with the assistance of, others, is indicated as such. Any views expressed in the dissertation are those of the author.

Signed: Maria Aamir

Date: 15th September 2021



ABSTRACT

Cancer is considered as a chronic disease which requires high-quality, long-term, posttreatment care (Drayton, et. al., 2012; Phillips & Currow, 2010). However, long-term survivorship care has become a growing healthcare burden requiring numerous resources including psychosocial support, late and long-term side effects monitoring, follow-up care to screen cancer progression, recurrence or newly developed or secondary cancers (Howell, et. al., 2011; McCorkle, 2011; Morgan, 2009). Breast cancer is the most frequent cancer in the world and in the United Arab Emirates (UAE); yet there is little known about breast cancer survivors' psychosocial concerns in the UAE. Research shows that meeting the full range of psychosocial needs significantly contributes to survivor's wellbeing and potentially elevates the quality of the patient's life (Holland & Reznik, 2005; Institute of Medicine, 2008; Culbertson, et. al., 2020). Thus, it is important to understand and meet the needs of the country's diverse population to help cancer patients deal with the range of psychosocial issues they may experience. The aim of this study was to investigate the psychosocial concerns of breast cancer survivors in a hospital setting in the UAE and the association with cultural factors related to two groups: Emiratis and expatriates.

A two-phase mixed methods study was conducted involving a cross-sectional quantitative survey to examine survivors' concerns and semi-structured interviews to develop an in-depth understanding of their needs. Among 205 breast cancer survivors who completed Phase One, twenty six percent were Emiratis and eighty seven percent were expatriates with the mean age of 49 years (both groups). Sixty percent of participants were diagnosed in 2018 or after that period. Around seventy four percent of survivors had regional stage disease and thirty eight percent of survivors had multimodality treatment. Fifty nine percent were on treatment whereas forty one percent had their treatment complete or had no treatment. Seventy two percent coded their quality of life as "good". The severity scores of each psychosocial domain were calculated based on the participants' reported concerns using mean scores. Information concerns were the highest reported concerns with the mean score 4.3. Emotional needs were the second most concerning reported with a mean score of 3.4. Physical needs were reported by the survivors with some level of concerns, mainly pain and fatigue (p=0.031). Survivors had a significant level of social and financial concerns (p<0.001). Regression analysis t-test results indicated no



significant differences in information and emotional needs between Emiratis and expatriates. However, a significant difference was found in physical and social & financial domains in two groups. There were no significant religious or spiritual concerns reported by survivors in both groups. A Chi-square test showed no association "between nationality and age" (Sig=0.287) and "between nationality and stage" (Sig=0.083) of the disease. Results also demonstrated significant positive association between physical concerns and received treatment whereas information concerns were significantly associated with age and type of the treatment received. The association between quality of life and psychosocial concerns was also explored using correlation analysis techniques. There was a negative correlation found between quality of life and the scores of psychosocial concerns including physical, social, emotional as well as spiritual concerns (p<0.01) and a significant correlation was found between quality of life and information concern (p<0.01). Multiple regression results showed significant positive association between physical concerns and received treatment, whereas information concerns were significantly associated with age and type of the treatment received. Physical and emotional concerns were found to make the strongest contribution to explain quality of life (QOL) (p<0.001, and p=0.001 respectively). In phase two, thematic analysis revealed three broader themes including "living experience with breast cancer, survivors' psychosocial concerns" and "survivors' experience with healthcare providers which revealed in-depth concerns amongst cancer survivors about addressing their physical, informational, social, financial, emotional, and spiritual needs related to living with cancer.

Cancer survivors continue to face challenges and symptoms even after their treatment is completed (Tian, Cao & Feng; 2019; Rutten, 2005; Siemsen, 2001), however, culturally tailored psychosocial support would likely improve their survivorship experience. In order to do that, health providers need to facilitate the development of comprehensive and integrated cancer services to meet the ongoing psychosocial needs of cancer patients. The study has indicated several gaps and barriers in the provision of high-quality cancer care such as lack of routine assessment of survivors' psychosocial concerns. It also highlights the need for further research in psychosocial needs and cancer survivorship care particularly in the region.

KEY WORDS: Breast cancer, psychosocial needs, expatriate in the United Arab Emirates, cultural dimensions of cancer care, cancer trajectory.



Chapter 1: INTRODUCTION

"All patients with cancer and their families should expect and receive cancer care that ensures the provision of appropriate psychosocial health services. It is not possible to deliver good-quality cancer care without addressing ... psychosocial health needs (Institute of Medicine, 2008, pg. 1)".

1.1 RESEARCH QUESTION

What are the psychological needs of Emirati and Expatriates in the United Arab Emirates?

1.2 RESEARCH PROBLEM

Breast cancer survivors not only experience multiple long-term adverse effects of treatment (Lovelace, et. al., 2019; Haidinger & Bauerfeind; 2019; Ewertz & Jensen., 2011) but many face a challenge to address the pre-existing comorbidities of their therapies which often lead to multiple other issues (Bodai & Tuso., 2015, Senkus, 2011). Cancer survivors are at risk for recurrence, secondary cancers, comorbidities, functional decline, and poor quality of life (Kline, et. al., 2018). There is no standard psychological or emotional response to a cancer diagnosis, but many patients experience the emotional consequences to be the hardest to deal with (Towers, 2016). Women's experience of breast cancer is unique and a complex feature of survival which impacts individuals varyingly. For some, cancer may be experienced as a single event with a defined start and end point whereas for others, it could be a chronic spectrum of events from pre-diagnosis, through diagnosis, treatment and follow-up. Follow-up may encompass different outcomes and consequences from long-term survivorship, recurrence of cancer or death due to it. It is important to assess the prevalence and magnitude of difficulties and concerns that cancer survivors may experience throughout the cancer trajectory. To improve cancer survivorship experience and to help identify the services needed by the cancer survivors, it is crucial to identify and address their needs, concerns and benefit of having lived with, through and beyond a cancer diagnosis.

The needs of cancer survivors are wholly a unique interpretation for each individual which is heavily influenced by several psychosocial factors. Cultural norms and beliefs also play an



important role. The disparities in cancer survivorship may occur due to the complex relationship of such factors. Thus, women of different cultural backgrounds may have different survival experiences from the disease and their needs could be distinct. Kagawa-Singer and colleagues, (2010) noted the impact of culture on cancer outcomes to be bidirectional in a way that one's culture influences the meaning assigned to the disease and their response to both disease and the healthcare system. Understanding sociocultural values has become the important aspect to detect potential determinants that could improve coordination of care and dynamics and meet culturally unique needs (IOM, 2013). In the absence of such understanding, cancer care strategies are less likely to be efficient to meet cancer patients' needs and to provide optimal cancer care. Additionally, large differences in cancer incidence and mortality have been consistently observed among social groups worldwide (Maskarinec, Sen & Koga, 2011; Ward, Cockkinides & Singh, 2004) yet, the underlying causes are not well understood (Ellis, et al., 2014). Studies have revealed that disease and illness are perceived individually in different cultures based on their beliefs and values (Morris, 2001; Resnicow, et. al., 1999). Moreover, the type and extent of the disease, the quality of care and psychological support services available to cancer survivors differ enormously across diverse settings (Andrykowski, et. al., 2008). Ethnic identification, religion, material realities, beliefs and behaviours lead to cultural diversity (Garrett-Rucks, 2016; Leavitt, 2002; Betancourt & Lopez, 1993).

Despite the high prevalence of cancer, there is a paucity of published data and limited understanding of the unmet needs of cancer patients in the Middle East (Nair, et. al., 2018). To date, relatively little is known about the psychosocial needs and its impact on cancer survivorship in the UAE. Majority of psychosocial issues and survivorship concerns have been studied in the Western world and awareness of these needs is relatively scarce in the Gulf region (Hulbert-Williams, et. al., 2019; Nair, et. al., 2019, 2018; Shapiro, 2018; Rowland & Bellizzi, 2008; Gilbert, et. al, 2008, Feuerstein, 2007, Ganz, 2003). Moreover, there is no structured survivorship program either at national or local level in the country. Psychosocial support services are generally provided upon request and referrals are made as needed. Knowledge of their specific psychosocial needs and if these needs are currently met, is also not known (Nair, et. al., 2018). In addition, there are only a few studies that have been conducted quite recently to assess cancer patients' needs in the diverse multicultural society (Nair, et. al., 2019; Nair. et. al., 2017; Elbarazi, 2016). The findings showed that women



in the UAE have significantly higher psychological unmet needs suggesting improvement in psychosocial care is an urgent priority.

1.3 AIMS AND OBJECTIVES

The aim of this study was to identify and investigate the psychosocial needs of Emirati and expatriate (UAE) residents who are not nationals, non-Emiratis) breast cancer survivors in the United Arab Emirates. Identification of cancer survivors' needs can indicate gaps in the service delivery and also identify utmost needs. The lack of literature for Emiratis breast cancer patients' needs provide the highest justification to pursue this study. This study is the first in the United Arab Emirates to address the breast cancer patients' needs. This research has identified concerns that breast cancer survivors deal with or fail to deal with. Thus, the study will aid in the development of culturally tailored cancer interventions that could improve survivorship care in the country and offers opportunities to redesign cancer services towards patient-centred care.

The study objectives were to:

- Identify the **physical**, **informational**, **social or financial**, **emotional and spiritual or religious needs** of cancer survivors residing in the UAE.
- Examine the **influence of nationality** on breast cancer survivors' needs.
 - *Hypothesis 1:* There is no association between breast cancer survivors' nationality and age
 - *Hypothesis 2:* There is a significant association between nationality and stage of breast cancer.
- Investigate **disparities in needs** of local Emiratis and expatriates utilising sociodemographics and clinical factors.
 - *Hypothesis 3:* There is a significant relationship between demographic factors and psychosocial concerns of breast cancer survivors.
- Identify factors to aid the development of culturally tailored **cancer survivorship programs.**
 - *Hypothesis 4:* There is a significant relationship between psychosocial concerns of breast cancer survivors and their overall quality of life.



1.4 SIGNIFICANCE OF THE STUDY

The widely accepted World Health Organisation (1948) definition of health encompasses physical, mental as well as social aspects of well-being which necessitates the use of multidimensional assessment of cancer patients' needs (Clinton-McHarg, et. al., 2010). The measurement of cancer patients' needs identifies their individual concerns and the extent of help likely to be required to address them (Rainbird, Perkins & Sanson-Fisher, 2005; Bonevski, et. al., 2000). By addressing the gaps in cancer patients' needs in accordance with different domains and in association with culture and sociodemographic factors, effective health interventions could be developed and tailored to meet the unique needs of each cancer survivor (Mukherjee, et. al., 2018; Setyowibowo, et. al., 2018; Douglas, et. al., 2016).

Understanding the full impact of unmet needs of the cancer survivors on their quality of life is crucial and clinically of prime importance throughout their continuum of survivorship to offer timely effective interventions. In addition, a robust body of knowledge to guide survivorship care is essential to optimally understand the healthcare needs of cancer survivors, identify evidence-based surveillance and follow-up strategies, identify and address unmet needs and determine ways to reduce occurrence and severity of treatment-related adverse effects among cancer survivors (Haylock & Curtiss, 2019). Several studies have found the direct association of greater supportive care with survival and its impact on better quality of life (Park & Hwang, 2012; Park & Hwang, 2012; Epplein, et. al., 2011; Uchida, et. al., 2011).

Improving the quality of life of breast cancer survivors requires addressing the unmet supportive care needs of the breast cancer patients (Lam, et. al., 2011). Investigating the needs of breast cancer patients offers a direct measure of the patients' support preference as well as gaps in the current services (Carlson, Waller & Mitchell, 2012). The identification of needs clarifies the areas where actions and resource allocation are needed in the cancer care setting to help cancer patients overcome their difficulties also result in reducing healthcare utilisation and cost by prioritising the urgent needs, and potentially enabling problems to be prevented or reduced through appropriate evidence-based early interventions (Akechi, et. al., 2011; Shim, et. al., 2006; Carlson & Bultz, 2004). Addressing breast cancer survivors' needs in the early stage of their survivorship provides guidance for new strategies in healthcare settings that could potentially reduce the burden of the disease and treatment in the long run thereby improving



their quality of life (Edib, et. al., 2016). Research in this area is important to ensure the provision of culturally relevant and appropriate cancer care and to improve cancer outcomes for the cancer survivors residing in the UAE. It is paramount to gain a more comprehensive understanding of the psychosocial needs and concerns that cancer patients deal with to ensure their optimal management and care. Developing and evaluating culturally tailored psychosocial care interventions for UAE cancer survivors would likely ensure their optimal management and care.

This thesis is apparently the first to investigate breast cancer survivor's needs and survivorship concerns and to foster informed intervention strategies towards cancer survivorship programs and research in the region. Additionally, cultural myths associated with cancer are not directly addressed in many countries particularly in Middle Eastern countries (Daher, 2012), hence, it is the first to identify cultural disparities in the residing population to address their survivorship issues encompassing their psychosocial needs. The study guides the implementation of appropriate psychosocial services as part of routine standard cancer care which may likely have a positive impact on cancer survivors' wellbeing and their quality of life.

This study successfully bridges the research gaps in the context of culture and diversity in the United Arab Emirates. This study offers valuable direction for future research and evidence to inform the optimal cancer care and provision of psychosocial cancer care for all cancer survivors in the UAE.

1.5 ORGANISATION OF THESIS

This dissertation is organised into eight chapters that allows developing the material as specified by the School of Health and Society at University of Salford. Chapter one provides the introduction encompassing an overview of the research problem, aims and objectives, rationale of this research, personal experience of conducting this study, and overview of thesis organization. Chapter two provides a literature review which explores the scientific evidence on breast cancer survivors' needs. Chapter three explains the methodology that guided the achievement of the research's objectives. Chapter four presents the results and analyses from both quantitative and qualitative methods. Chapter five contains discussion and the conclusion of the research including dissemination strategies to address the problem to healthcare leaders.



Chapter six presents an overall conclusion extrapolated from the study. Chapter seven provides a list of references whereas Chapter eight contains appendices providing a list of materials, tools and resources used to conduct this research.



1.6 BACKGROUND

1.6.1 UNITED ARAB EMIRATES POPULATION

The UAE is a small country situated in the Middle East region of Asia between Saudi Arabia and Oman. It comprises seven emirates which include Abu Dhabi, Dubai, Sharjah, Ajman, Umm Al Quwain, Ras Al Khaimah, and Fujairah. The Emirate of Abu Dhabi is the capital and the largest, equivalent to 86.7% of the country's total areas excluding the islands (Kenneth & Miklos, 2017).



Figure 1: Map of United Arab Emirates

The UAE is considered as one of the richest countries in the world with GDP per capita \$67,696 (World Economic Forum, 2017). It has the most diverse population in the Middle East due to a large number of expatriates. Over 75% of the UAE population is expatriate. The expatriates come from different nationalities and have different socioeconomic and cultural backgrounds than local Emiratis. About 50% of the UAE population is from South Asia mainly from India, Pakistan and Bangladesh. Those from other parts of Asia include Filipinos, Iranians and



Chinese contributing one million of the population. There are approximately 500,000 westerners from Europe or America in the country.



Figure 2: United Arab Emirates Population Pyramid

The population has a skewed gender distribution consisting of twice as many males than females (United Nations, 2019). Eighty eight percent of the UAE population is urban and the illiteracy rate (1975-2019) among youth is over 90% (The World Bank, 2021). Islam is the largest and the official religion of the country among both expatriates and UAE nationals (UAE Government Portal, 2019; AlMazrouei & Pech, 2015). Although Life expectancy is 77.8% (UAE Government Portal, 2021), the population of the UAE is still relatively young, with 25.3% of the population under 15 years and 90.4% of the total population under 45 (Whitman & Aldinger, 2009). Healthcare insurance is mandatory for UAE residents and mainly provided by employers.



1.6.2 UAE HEALTHCARE SYSTEM

UAE is actively expanding its national healthcare system to meet the growing needs of its people and support economic diversification with leading medical centres, corporations and academic institutions playing vital roles in the process. All seven emirates provide healthcare services to their citizen and are rapidly building their healthcare infrastructure inclusive of hospitals and clinics.

The U.A.E. has a number of healthcare regulators at both the federal and emirate levels, each with their own geographic or functional purviews. U.A.E healthcare regulation has recently been shaped by two broad trends: (1) an expanding remit for the Ministry of Health and Prevention and (2) increasing coordination among regulatory authorities. At the federal level, the most important regulatory authority in the U.A.E healthcare sector is the Ministry of Health and Prevention (MOHAP) while others include the Insurance Authority and the UAE Nursing and Midwifery Council. While healthcare provision in some emirates (Northern emirates) is regulated entirely by MOHAP, certain emirates or free zones also have their own regulatory authorities, most notably The Department of Health – Abu `Dhabi (DOH) and Dubai Health Authority (DHA) which regulates the healthcare sector in Abu Dhabi and Dubai respectively. Whereas, Dubai Healthcare City Authority – Regulatory (DHCR), regulates Dubai's healthcare free zone: Dubai Healthcare Operators in the free zone. On the other hand, Sharjah Health Authority (SHA), regulates Sharjah's healthcare system in the emirate of Sharjah.

1.6.3 ABU DHABI HEALTHCARE SYSTEM

The Department of Health Abu Dhabi (DoH) is the regulatory body of the healthcare sector in the Emirate of Abu Dhabi that defines the strategy for the health system and monitors and analyses the status of population and performance of the healthcare system. The DoH has a comprehensive cancer control plan that includes prevention, early detection, diagnosis, treatment and palliative care. DoH has established the breast cancer management standards with the help of experts that include screening standards as well as genetic counselling. Yet,



much work is needed towards raising awareness and implementing cancer survivorship strategies.

The Abu Dhabi Health Services Company (SEHA) is an independent public joint stock company that owns and operates all public hospitals and clinics across the Emirate of Abu Dhabi. SEHA claims to be is one of the largest healthcare networks in the U.A.E., consisting of 12 hospitals, 46 primary healthcare clinics, 10 disease prevention and screening centres, 3 mobile clinics, 1 school clinic, 2 blood banks, 4 dental centres, 2 employee healthcare centres, and 1 vaccination centre. SEHA's business entities are – in decreasing order of patients served – Tawam Hospital, Sheikh Khalifa Medical City, Al Ain Hospital, Al Gharbiya Hospitals, Al Mafraq Hospital, Al Rahba Hospital, Sheikh Shakhbout Medical City and Corniche Hospital (the U.A.E.'s largest maternity hospital). The Al Gharbia Hospitals business entity notably consists of six hospitals and four primary healthcare centres that serve more than 200,000 people living in the 60,000-square-kilometer area of Abu Dhabi known as the Western (Al Gharbia) Region.

In addition, Government-linked entities such as Mubadala Healthcare, a division of the prominent and diverse Abu Dhabi government investment vehicle Mubadala Investment Company, are supporting the development of a world-class, commercially sustainable private healthcare sector in the U.A.E. Mubadala Healthcare works with prominent international partners including Cleveland Clinic Abu Dhabi, Healthpoint, Imperial College London Diabetes Centre and Abu Dhabi Telemedicine Centre. Other Mubadala Healthcare assets include the Abu Dhabi Knee & Sports Medicine Centre, Capital Health Screening Centre, Tawam Molecular Imaging Centre, and the National Reference Laboratory.

Over the last decade, the U.A.E. has moved toward mandating health insurance for all citizens and residents. Healthcare insurance is mandatory for Abu Dhabi residents mainly provided by the employers whereas Abu Dhabi (now Department of Health) provides comprehensive health coverage to all U.A.E. nationals living in that emirate at no cost through the "Thiqa" program. However, despite breast screening being covered under health insurance, many women do not get checked. A recent study established that only 13.95% of females over the age of 40 in the UAE had a mammogram in a two-year period (So, et. al., 2019). A mammogram cost about



DHS 300 dirhams (\pounds 62); however, the major barriers to breast health seeking behaviour have been identified as cultural rather than economic.

1.6.4 CANCER DEFINITION AND BIOLOGY

Cancer can be defined as a disease in which a group of abnormal cells grow uncontrollably by disregarding the normal rules of cell division. Normal cells are constantly subject to signals that dictate whether the cell should divide, differentiate into another cell or die. Cancer cells develop a degree of autonomy from these signals, resulting in uncontrolled growth and proliferation. Abnormal cell growth (neoplasia) is the biological end point of the disease. If this proliferation is allowed to continue and spread, it can be fatal. Almost 90% of cancer related deaths are due to tumour spreading – a process called metastasis (Hejmadi, 2010). Initiation and progression of cancer depends on both external and internal factors within the cell such as (tobacco, chemicals, radiation and infectious organisms in the environment or inherited mutation, hormone or immune conditions or mutation that occur from metabolism within the cell). These factors can act together or in sequence, resulting in abnormal cell behaviour and excessive proliferation. As a result, cell masses grow and expand, affecting surrounding normal tissue and can also spread to other locations in the body which is called metastasis. Seyfried & Huysentruyt (2013) explained that tumour cell invasion of surrounding tissues and spread to distant organs is the primary cause of morbidity and mortality of cancer patients.

1.6.5 CANCER DIAGNOSIS

The diagnosis and development of cancer is a traumatic and life-changing event (Smith, et. al., 2016). It is a dreadful experience which disrupts a life of even the strongest individuals resulting in psychosocial consequences mainly, emotional distress, sadness, depression, anxiety and to the extent that families could break down too. Many women go through the intensive and challenging cascade of treatment with a diverse range of interventions. Cancer treatment often involves multiple options which can be toxic, exorbitant, intense and long-term. In addition, responses to cancer treatments could be quite variable thus predicting the potential risks and benefits of various treatment options for the individual cancer patients could be difficult. Because of the life-threatening nature of the disease and its emotional consequences



along with limitations and complexity of treatment choices, it is often difficult for cancer survivors to make decisions about their care and to cope with the illness due to the fragmented nature of the cancer care system. The transition from the various stages during the cancer trajectory is generally challenging and adjustment to survivorship could be daunting (Chadder, et. al., 2018; Costanzo, et. al., 2011; Kantsiper, et. al., 2009; American Society of Clinical Oncology, 2007). Cancer diagnosis has deep effects not only on psychological state but also on the cognitive, emotional, spiritual and social features of the individual (Çürük, et. al., 2020; Gorman, 2018; Wei, et. al., 2016). Cancer survivorship remains a challenge because, in most cases, patients do not return to a precancerous state after the completion of treatment. Diagnosis and treatment of breast cancer are stressful events and may result in various psychosocial concerns which are more commonly experienced following treatment (Liao, et. al., 2014). Breast cancer survivors need not only medical treatment but also psychosocial support (Garlick, et. al., 2011) which results in delaying the frequency, duration and intensity of late and long-term effects of cancer treatment (Liao, et. al., 2010).

1.6.6 CANCER – A GLOBAL PROBLEM

Cancer has emerged as one of the most serious health issue globally threatening human health and lives both. Cancer incidence and mortality are rapidly growing worldwide. The reasons are complex and reflect both aging and increase in population (Bray, et. al., 2018). Cancer is the second leading cause of death globally responsible for an estimated 9.6 million deaths in 2018 (World Health Organisation, 2021, 2018a). Globally one in six deaths is due to cancer. Approximately 70% of deaths are reported from lower- and middle-income countries where late-stage presentation and inaccessible diagnosis and treatment are reported as main concerns. According to the World Health Organisation (2018b), around one third of deaths from cancer are due to behavioural and dietary risks. In addition, only one in five low- and middle-income countries have the necessary data to drive cancer policy. Data from GLOBOCAN (2018a) indicated that 43.8 million people diagnosed with cancer were alive within 5 years of a cancer diagnosis. Survivor population represents a diverse range of psychosocial issues emerging from cancer. Understanding cancer survivorship issues can help the public health community meet the needs of this expanding and heterogeneous population. Likewise, the burden of cancer in the United Arab Emirates is high which is ranked the second leading cause of noncommunicable disease related mortality in the country (Ferlay, et. al., 2015; Loney, et. al.,



2013). Radwan and colleagues (2018) reported that in the GLOBOCAN (2012) reports, cancer incidence and mortality rates in UAE reached 92.5 and 58 per 100,000 people, respectively.

1.6.7 BREAST CANCER EPIDEMIOLOGY

Globally, breast cancer is the second most common cancer and the most frequent cancer (11.6% of all cancers) among women with an estimated above 2.1 million new cancer cases diagnosed in 2018 (Globocan, 2018a). It ranks as the fifth leading cause of death (627,000 deaths, 6.6%) because the prognosis is relatively favourable, at least in more developed countries. According to the World Health Organisation (2015), it is also the leading cause of cancer death in women (15.0%). WHO's projected death rate from breast cancer is predicted to rise to 805,116 deaths per year by 2030, accounting for a 43% increase in the absolute number of deaths from the disease (World Health Organisation, 2017a). For breast cancer mortality, the UAE ranks 131 out of 172 countries (World Life Expectancy, 2017a).

Breast cancer in the UAE is the most common cancer affecting women regardless of nationality or gender (International agency for Research on Cancer, 2020). It constitutes almost 20% of all cancer cases annually (Ministry of Health & Prevention, 2017). In 2014, it accounted for 38% of all cancers in women (Kenneth & Miklos, 2017). According to the Ministry of Health, UAE (2014) latest cancer report, there were 819 new cases diagnosed with breast cancer in the UAE in 2014 of which 94% were malignant and 6% were in-situ tumours. Among malignant cases, 77% were expatriates and 23% UAE nationals, reflecting overall UAE population distribution. The age group 45-49 was the most prevalent with 18% of all cases, followed by 40-44 & 35-39 (14%) and 50-54 (14%). Reported by the Department of Health, Abu Dhabi, (2017), 12% of deaths occurred due to breast cancer in the Emirates. Approximately over thousand women are newly diagnosed with breast cancer (International agency for Research on Cancer, 2020). Women in the Gulf region are more likely to develop breast cancer at least a decade earlier than western countries (Najjar & Easson., 2010). As per 2010 report by the Centre of Arab Genomic studies, Arab women are most likely to develop breast cancer at an early age with the median age at diagnosis 10-15 years younger than North America and Europe (Friend of Cancer Patients, 2018). In the UAE, around 15% of breast cancers are metastatic (Globocan, 2015). This statistic is quite alarming as in younger women, breast cancer is generally more aggressive and less likely to have sustained response to treatment than in older women



(Bendardaf, et. al., 2020; Friend of Cancer Patients, 2018; Elobaid, et. al., 2016; Maarraoui, et. al., 2006).

Despite an increase in the number of women in the UAE taking up breast cancer screening and having access to the national screening programmes, only around 75% seek medical advice after experiencing signs and symptoms (Cardoso, et. al., 2018). A previous study in the UAE addressed that almost 49% of women never had a breast exam and 44% never had a mammography, expressing a lack of knowledge about the existing screening programs in the country (Elobaid, 2014). The study concluded that such healthcare choices are often considered as being associated with the complex religiously dominated culture of the UAE. Therefore, culture has a strong association with women's decision-making in the UAE society (Elobaid, et. al., 2016).

1.6.8 CANCER SURVIVOR AND CANCER SURVIVORSHIP

"What patients see through the glass is not a world outside cancer, but a world taken over by it—cancer reflected endlessly around them like a hall of mirrors." -- Siddhartha Mukherjee, The Emperor of All Maladies

A cancer patient is considered to be a survivor **from the time of diagnosis until the end of life** (National Cancer Institute, 2021a).

The concept of survivorship has been widely debatable and in the past few years a variety of other definitions of cancer survivorship have been outlined. The definition has been expanded to include family, friends, and caregivers (National Cancer Institute, 2021b; Bell & Ristovski-Slijepcevic, 2013)

American Society of Clinical Oncology (2017) defines cancer survivorship as the process of living with, through and beyond cancer. By this definition, cancer survivorship begins when a person is diagnosed (Doyle, 2008; Mullan, 1985).

Although the term survivor refers to an individual who is diagnosed with cancer from the time of initial diagnosis, cancer survivorship rather focuses on the health and life of a person with cancer **mainly post treatment until the end of life.** According to Bugos (2015), the conceptual difference between the phase of survivorship beginning at the time of diagnosis and



survivorship following curative treatment creates conflict in defining the best time and ways to provide survivorship care, especially to people who are challenged by incurable cancers.

Survivorship covers the physical, psychosocial, and economic issues of cancer, from diagnosis until the end of life. It focuses on the health and life of a person with cancer beyond the diagnosis and treatment phases. Survivorship includes issues related to the ability to get health care and follow-up treatment, late effects of treatment, second cancers, and quality of life. Family members, friends, and caregivers are also part of the survivorship experience (National Cancer Institute, 2021). According to the Institute of Medicine (2006), survivorship begins at diagnosis, the moment the battle with cancer begins. Cancer survivorship describes the many experiences and emotions that are part of living life as a cancer survivor (Livestrong foundation, 2011; Lance Armstrong Foundation, 2004).

The definition of survivorship could differ around the scope of the population covered. Some refer to the term "survivorship" to encapsulate individual experience through the cancer journey whereas others apply the term to a distinct phase in the cancer trajectory between primary treatment and cancer recurrence or end of life (Rowland, et al., 2006; IOM & NRC, 2006). The later definition is widely accepted and commonly used to discuss cancer survivorship in general. In this particular study, the survivorship concept has been applied considering the former definition while broadening the scope of population: *all cancer survivors from the time of initial diagnosis through the cancer trajectory*. The approach will avoid missing out the opportunity to capture the population which is generally considered as reserved and modest.

1.6.9 CANCER AND QUALITY OF LIFE (QOL)

Cancer can lead to severe distress as survivors face several complex issues through transition to survivorship (Melchior, et. al., 2013). Effects of cancer and its treatment and quality of life perception both are subjective perceptions and how cancer survivors perceive them in day-to-day life is quite an individual experience which is unique to each person. Although accepting the new norm, the majority of cancer survivors have difficulty adjusting to the trade-offs of survival whether it is long term or late effects or treatment. The ability of maintaining a sense of control greatly influences survivors' psychological wellbeing most commonly resulting in anxiety and depression (Gorman, 2018). Although breast cancer care today involves state-of-



the-art biomedical treatment, it can fail to address the broader psychosocial issues associated with the transition to breast cancer survivorship (Culbertson, et. al., 2020). Physical symptoms, such as pain, infection, lymphedema and impaired arm mobility are known symptoms of breast cancer which adversely affect both physical functioning and psychological health, subsequently resulting in decreased QOL (Taghian, et. al., 2014; Fu, et. al., 2013). The QOL in the healthcare context is defined and measured by the perception of individuals where they stand in their lives geared by their culture and value system in which they live in relation to their goals, standards and concerns (WHO, 2012). Grant & Dean (2003) identified the domains of quality of life to include physical health, emotional state, level of independence, social relationships, environment, and spiritual state. In healthcare, physical, emotional, mental and spiritual elements of life are intricately woven together and interact to support a strong and healthy person (King, Smith & Gracey, 2009). Any change in one domain affects the other domain. For instance, fertility concerns affect the emotional, spiritual as well as social domains. Since breast cancer brings a multitude of problems in women, healthcare providers should take a holistic approach to identifying and addressing their psychosocial needs and adjustments of patients. Meeting cancer survivors' needs can lead to many positive patient outcomes, including better ability to cope with disease symptoms and treatment side-effects, improved physician-patient communication and better adherence to treatment regimens (Okediji, et. al., 2017; Brédart, et. al., 2012).

1.6.10 UNDERSTANDING CANCER AND CULTURE

"Experiences of cancer are culturally shaped and although it is impossible to know about all specific beliefs, we must recognise that differences exist. Cultural beliefs play an important role in whether people seek early diagnosis, how they understand their disease, the kind of support they are willing to accept, decisions about treatment and how they experience the process of getting well and dying (Cancer Forum, 1991).

Cancer is much more than a biological disease. The experience of illness cannot be separated from the cultural contexts in which it occurs. Cultural factors are involved at every stage in the journey through cancer from prevention to palliative care. Even the definition of what constitutes cancer varies according to the cultural context. For instance, in some cultures, only late-stage cancer is considered as real cancer that causes death and other non-fatal conditions



are not perceived as cancer therefore screening is not widely used (Dein, 2007). The relationship between culture and cancer can be described as complex and multifaceted and might be expected to contribute to variability of the cancer burden across different cultural groups (Schmitz, et. al., 2013). In cancer patients, the perception of illness largely determines the quality of life and sense of well-being (Ma, et. al., 2018; Kus, et. al., 2017; Ashley, et. al., 2015). Levental's self-regulatory model provides a framework for understanding the impact that illness beliefs and emotions have on how an individual copes, adapts and responds to illness (Leventhal, et. al., 1984, 1980). According to this model, patients react to an illness based on a conceptual idea which is developed both cognitively and emotionally through a collection of personal experiences (Iskandarsyah, et. al., 2013; Kaptein, et. al., 2013). This model postulates that disease coping mechanisms depend on patients' perceptions about it which greatly influence their adherence to treatment, disease outcomes as well as sense of wellbeing (Ma, et. al., 2018; Kaptein, et. al., 2013; McCorry, et. al., 2013). Women of different ethnic backgrounds have different disease outcomes and survival (Tsai, et. al., 2011; Maskarinec, et. al., 2011). There is substantial evidence of the influence of ethnicity towards illness perception; for instance, people with darker skin colour perceive illness differently and have varied response to disease outcome compared to whites (Wang, et. al., 2013). Additionally, religion and spirituality have a positive influence on mind, body and soul which resulted in better health outcomes (Bowie, et al., 2003; Koenig, 2000; Tix & Frazier, 1998). The incongruence in cancer survivorship may also occur due to multifarious factors including socio-economic position, social and cultural injustice (Kingsley, 2010; Cancer Health Disparities, 2008). Thus, identifying the cultural determinants and cultural differences in cancer survivorship is critical for improving care (Lam, et. al., 2011).

Cultural issues have become an increasingly important consideration in cancer care. Culture plays a vital role in establishing perceptions of disease and health. Cultural determinants particularly beliefs about health and illness, societal values, family role, religious beliefs and disclosure of cancer information have been addressed for their association with cancer and cancer disparities (Gómez-de-Regil, 2014; Chaturvedi, et. al., 2014, Marshall, et. al., 2011). However, beliefs vary across different regions of the world which are largely shaped by religious values and are associated somehow to the essential elements of culture (Martsolf, 1997). It is often obscure whether those beliefs originated from religious affiliation or are associated with cultural background (Miller, 1995). People's beliefs about health and illness



can clearly affect their behaviour (Gómez-de-Regil, 2014). Their perception of illness may influence the probability of seeking treatment (Kugbey, et. al., 2020; Agbokey, et. al., 2019; Fronda, 2017; Elobaid, et. al., 2016), treatment adherence (Nair, et. al., 2018; Lee, et. al., 2019), use of medical services (Qian-Xin, et. al., 2017; Meacham, et. al., 2016; Torres, et. al., 2016; Walijee, et. al., 2011) and quality of life (Lee, et. al., 2019; Hopman & Rijken, 2015). In some cultures, disease is considered God's punishment (Boyes, et. al., 2009, Wiggers, et. al., 1990) or one's 'karma' (fate) (Chaturvedi & Chandra, 1998). Illness and disease are considered to be weaknesses in Arab culture where families with ill individuals are perceived weak and vulnerable. In order to preserve Arab cultural values, disease is not discussed in societies and is considered a family issue which must be kept in their homes (Kane, et, al., 2020). Cancer is considered a bad word (Surbone, 2008) and individuals, particularly in Eastern cultures, have fatalistic views about living with passivity. Also, it is not uncommon that women's quality of life is partially shaped by gender hierarchies within some cultures particularly Arabs (Salem & Daher-Nashif, 2020) and is mainly described as the ability to fulfil their family roles as mothers and wives (Hwang, et. al., 2017; Hammoudeh, et. al., 2017; Elobaid, et. al., 2016; Jassim & Whitford, 2014; Taha, et. al., 2012). On the contrary, in Western cultures, most people involve themselves in fighting a battle against cancer and to win their health back (Chaturvedi, et. al., 2014).

How patients make healthcare decisions, may also be influenced by the cultural background. Some cultures place a strong value on the community while others encourage individual independence (Givler & Maani, 2020). Cultural geography influences an individual's needs, attitudes and response to cancer and can be targeted to improve comprehensive cancer care (Corner & Bailey 2008). Culture and dynamic belief, lifestyle and values systems pass through generations and shape people's definition of health and well-being (Kagawa-Singer & Kassim-Lakha, 2003; Angel & Thoits, 1987). Such influences are complex and may vary across different cultures (Fielding, et. al., 2013).

Multidimensional cultural factors such as language, environment, social structure, religion and economy influence health status, disease perception and medical care (Lopez-Class, et. al., 2011). There are strong social conventions which permit or prohibit the public expression of emotion. In some cultures, anger is a common response to cancer diagnosis whereas in others such as in Arabs, the expression of anger is seen as lack of faith in God. It is important for



healthcare providers to understand the modes of emotional expression so that they are not labelled as unacceptable or pathological (Dein, 2007). Additionally, socio-cultural factors and family values are associated with receiving a late diagnosis and known to be constituted barriers to cancer screening, early diagnosis and treatment (Elobaid, at. Al., 2016; Australian Institute of Health and Welfare, 2014).

The possible reasons for a worse prognosis of some cancer in certain cultural groups have been addressed as late presentation, lack of treatment, treatment non-compliance, use of traditional healing and cultural insensitivity among health professionals (Salem & Daher-Nashif, 2020; Elobaid, at. al., 2016, Shahid, et. al., 2016; Kagawa-Singer, et. al., 2010; Dein, 2007; Brach & Fraser, 2000). Cancer has also been associated with the "Evil Eye" and believed to be resolved on its own or through home remedies such as rubbing the lump with olive oil (Salem & Daher-Nashif, 2020).

There are also variations in the way that a cancer diagnosis is conveyed to the patients. In non-Western cultures mainly in Arabs it is the family who are told about the diagnosis not the patient. This can cause major problems in Western healthcare setting where the ethos is towards open disclosures. Psychosocial issues may also vary between cultures because cultures view and value illness, treatment and caregiving differently (Shahid, et. al., 2011; Chen, et. al., 2011; Kumar, et. al., 2010). Similarly coping strategies can also be influenced by such view about illnesses and cultural beliefs (Arman, et. al., 2004; Arman & Rehensfeldt, 2003; Rodgers & Cowles, 1997). For instance, in Muslim societies, it is believed that the stronger their faith in Allah (God), the greater the chance of a cure (Al-Azri, 2013). Therefore, women in these societies tend to follow their religious beliefs and practices to cope up with the diagnosis including listing and reciting the Qur'an, praying and reciting religious verses (Green, et. al., 2018; Gleeson, et. al., 2016; Al-Azri, 2014).

It is crucial to understand the differences between societies in which their culture shapes individuals' reaction to it. Cultural elements and traditions including gender hierarchies, family values, tribal mentality, cultural practices and perceptions may shape individual beliefs, social attitude, behaviours, perception, experience, needs as well as reaction and desired outcome of the disease and its treatment (Elobaid, et. al., 2016). The provision of high-quality cancer care to cancer survivors and their families require attention to their psychosocial needs and concerns



which may be related to cultural aspects of survivors' experience. Cancer survivors' needs should be met on their own terms. The provision of holistic care necessitates understanding of cultural factors and its influence on survivors' cancer experience and needs play an important role in their comfort throughout the disease process and help them deal with the life-threatening illness (Wright, 2019).

1.6.11 CANCER SURVIVORSHIP CARE IN THE UAE

Although we have a greater knowledge about the ongoing challenges of cancer survivorship, fragmented cancer care in the country has hampered effective and coordinated delivery of needed psychosocial care to cancer survivors. Unfortunately, there is no institution or facility in the UAE which offers a structured cancer survivorship programme so far. Also, there is a lack of evidence-based, culturally relevant and supportive care programs for cancer survivors in the country. Although the number of cancer survivors are increasing worldwide (Kimberly, et. al., 2019; Bray, et. al., 2018; Parry, et. al., 2011), the lack of data on survival rates in the UAE further overshadows the predicted need of survivorship programmes that are uniquely designed to meet the needs of local communities. Furthermore, there is no evidence available in the UAE to quantify the association of sociodemographic and clinical factors with breast cancer patients' needs. The concept of cancer survivorship has received more attention in some developed countries, yet the survivorship issues are less recognised in developing countries including the UAE (Grunfeld, 2006; Hewitt, et. al., 2006). The UAE health authorities have recognised the chronic nature of cancer yet the existing healthcare policies for the cancer care continuum remain primarily focused on raising awareness, early detection and standardising cancer care. Attention to cancer survivorship and long-term care needs are unevaluated and lacking in this country. There is a gap in cancer services to provide systematic follow-up care for the growing number of survivors. Cancer survivorship issues in the UAE are more complexed since the key challenge is that the complete spectrum of care is not fully covered by insurance companies and the basic insurance scheme for unskilled and lower paid workers may not cover cancer treatment expenses, although they may be covered by charitable organizations (Global Health Delivery-Dubai, 2018). Thus, many cancer survivors do not pursue follow-up after completing their cancer treatment due to sporadically planned hospitalbased follow-up care which could be a further challenge for expatriates as they need to pay out-of-their pockets.


Based on these findings, healthcare authorities in the UAE should establish and implement cancer services that include cancer survivorship programs to reflect the full range of psychosocial needs experienced by cancer patients from diagnosis to cure, or terminal care. Authorities need to ensure that health professionals understand the expectations of cancer patients and appreciate that psychosocial needs vary by patients' socio-demographics to help them deal with a range of their survivorship issues.



1.7 PERSONAL EXPERIENCE

Since I have been working in the Oncology field for the past 14 years, the desire to conduct this research initially emerged from my work experience. My role in the Oncology service line and cancer information services requires analysis of cancer data and reporting cancer information including patients' outcomes. In this role, I regularly review hundreds of cancer patients' records and read about their cancer experience through different healthcare facilities and with many different providers. I realise how cancer patients navigate through different systems and struggle to cope with the disease process and may get stressed during the transition from the various stages in the cancer trajectory. It is apparent that many cancer patients in this region keep the disease confidential and some do not even disclose it to their family members. Most patients wait to seek cancer care at their advanced stages probably due to cultural reasons as cancer has been considered as a social stigma in the UAE and in other Arab countries. I learned that most cancer patients seem to be less informed about the relationship and impact of continuity of care and their health outcomes. Furthermore, some cultural attributes such as the unobtrusive nature of the local population in the UAE are socially introverted in discussing cancer openly as well as in addressing their needs to health providers which may pose a challenge to provide patient-centred care. I empathize with the pain and suffering that cancer patients experience through their survivorship trajectory. The limited view of survivorship care and the lack of coordination in the currently fragmented healthcare system have led me to investigate the unexplored area of cancer patients' needs and concerns through cancer survivorship and the gaps in the current practices in this multicultural community. This research has provided me the opportunity to be the voice of cancer patients and I hope this study can make a positive difference in improving cancer services and cancer patients' lives in the UAE.



Chapter 2: LITERATURE REVIEW

This chapter begins with a presentation of the literature relevant to cancer survivors' needs and survivorship concerns including the identification of the gaps in the current knowledge. The process of conducting the search has been explained. The conceptual framework of survivorship needs has been described which forms the basis of the research question to highlight survivors' needs and experience of having cancer. The sub-sections explore the existing literature on distinct types of cancer survivors' needs in a detailed manner which includes concerns about physical, information or educational, social & financial, emotional, spiritual or religious concerns. **The cultural influence** has been studied in detail **to identify disparities** among different races and cultures to further understand how needs are influenced by certain socio-demographics. The last section provides a summary conclusion of the information gathered through the literature search.

2.1 LITERATURE SEARCH STRATEGY

An initial search was conducted on "cancer patient needs" through the Salford Library and EBSCO host search engines in English utilising all resources regardless of time frame. The initial query resulted in 167,247 resources which were further narrowed down to 833 with the availability of full text, resources type including articles, text resources and electronic books, selecting breast cancer only, and by selecting databases or resources including: Elsevier, MEDLINE, PubMed Central, Springer, Sage, BMJ, Wiley, JSTOR, BMJ, PMC, Ovid, Taylor & Francis online, ScienceDirect & EThOS. The Google search engine was also explored for the relevant search utilising combination of keywords: cancer, breast cancer, survivors, cancer survivorship, cancer patients, needs, perceived needs, needs assessment, supportive care needs, psychosocial needs, ethnicity, race, culture, expatriate, Emiratis, Arabs, United Arab Emirates, Middle East & Muslims. The term *cancer patient* was used as synonymous with cancer survivors. Additionally, "TRIP" database (Turning Research into Practice, 2018, 2017, 2005) which is a free access medical database was searched to maximise the outcome of the latest literature utilising the same search terms which resulted in 12,881 results that includes all resources. The latter search included all studies which assessed needs of cancer patients or described aspects of need with breast cancer and presented the results with cross-cultural



disparities. All searches utilised the above-mentioned methodology. Excluded studies were those that resulted in other types of cancer or did not particularly specify breast cancer or provided abstracts only. Newspaper articles, websites & audio-visual resources were also excluded from the search. There was no time frame considered for the published literature. The abstracts were reviewed initially to examine the relevance of the literature to the research. The references were also used to identify additional searches if relevant to the similar subject discussed in the source materials. The relevant studies were reviewed to assess their research design, sampling methods and findings; hence both qualitative and quantitative studies were reviewed. The below section will discuss the theoretical framework used in this study.

2.2 THEORETICAL FRAMEWORK

2.2.1 BACKGROUND

The general objective of this research is to investigate the multidimensional needs of breast cancer survivors so opportunities for improvement in cancer care could be identified. Many studies have shown that breast cancer diagnosis and treatment have considerable negative effects on the physical, psychological, and social health of patients and can reduce their quality of life (Haidinger & Bauerfeind, 2019; Kline, et. al., 2018; Jacobs, et. al., 2017; Hofsø, et. al., 2012; Waljee, et. al., 2011; Byar, et. al., 2006). The psychosocial domain in cancer care focuses on cancer patients' unique responses and reactions to cancer diagnosis and its treatment; thus, quality cancer care strives to minimise the patients' cancer treatment-related effects and symptoms. Minimising these effects may reduce distress and suffering of cancer patients to enable them to understand and cope with the disease, its treatment and the consequences. Comprehensive and high-quality cancer care needs to be delivered in a context of mutually respectful patient-provider relationship and must incorporate physical, psychosocial, emotional and spiritual dimensions (Aghaei, et. al., 2021; World Health Organization, 2018c; IOM, 2013, Epstein & Richard, 2007). As per Holland (2002), cancer survivors seek a comforting philosophical, spiritual or religious belief system to help them give a meaning to the new world of illness since the diagnosis of cancer, with the fears and threat attached to it result in a complex set of issues including physical and psychological reactions to individuals and their family as well. It is critical that healthcare providers need to individualise cancer care by using



knowledge of the patients' experiences, behaviours, feelings and perceptions (Radwin & Alster, 2002).

2.2.2 CONCEPT OF "NEED"

Breast cancer survivors' psychosocial concerns and needs are the main focus of the theoretical framework of this study. A 'need' can be defined as the requirement of some action or resource that is necessary, desirable or useful to attain optimal wellbeing in cancer patients (Koh, 2013).

Need is an idiosyncratic concept which compasses a subjective interpretation. Needs can also be viewed as a theoretical construct which consists of general principles of human existence (Holmes & Warelow, 1997). In literature, no consensus has been established about the meaning of "need" in health (Asadi-Lari, et. al., 2004). According to Robinson & Elkan (1996), need is a subjective concept that is both relative to time and place therefore it is important to understand the meaning of individual needs objectively. Murray (1938) and Carver & Scheier (2008) explain need as an internal state or condition, a lack of something that is necessary for well-being and motivates behaviour.

Bradshaw's (1972) taxonomy provides a useful way of considering different perspectives that may emerge while exploring the needs related to a particular issue or population group from different angles. According to Maslow (1943), human needs arrange themselves in "hierarchies of pre-potency" such that the appearance of one depends upon prior satisfaction of more prepotent needs. The needs are composed of physiological, safety, love/belonging, esteem and finally self-actualisation. In his seminal model of the Hierarchy of Needs pyramid, Maslow organized human needs in five levels forming a pyramid with more basic needs (physiological needs) at the bottom of the hierarchy and the need for self-actualization at the apex of the pyramid. Individuals will usually be motivated to meet higher need levels when lower order needs are satisfied therefore, we all must satisfy our need for security if we hope to achieve our highest potential which is called self-actualisation. However, Maslow (1943) later clarified that satisfaction of a need is not an "all-or-none" phenomenon, admitting that his earlier statements may have given "the false impression that a need must be satisfied 100 percent before the next need emerges" (Maslow, 1987).



The quality of life and well-being of cancer patients is determined by how well their physical, social, psychological, emotional, and spiritual needs are being met. Ng., et. al., (2017) used Maslow's hierarchy of needs pyramid in the context of breast cancer with reference to Institute of Medicine (US) (2008), Schmid-Büchi, S. et, al., (2008), NCCS Breast Cancer Survivorship Programme (2016) (Figure, 3) and posited that the distinction between the hierarchy of needs may be difficult for cancer patients as their needs are often interlinked. Therefore, all aspects of cancer patients needs must be assessed and managed concurrently (Ng, et. al., 2017). Cancer survivors' psychosocial needs can be explained using Maslow's hierarchy of needs in the figure below in the context of breast cancer patients' psychosocial concerns.



Figure 3: Maslow's hierarchy of needs in breast cancer context



Maslow's hierarchy of needs (1948) reflects on the key areas of everyday life and function that people with threatened health need such as cancer in order to continue their lives.

The first level which is the basic or physiological needs are of prime importance for the survival of the individual such as need for air, food, water and sleep; clothing and shelter and sexuality for the long-term survival of the human beings. If these needs are not met, the human body cannot function optimally. Maslow described the physiological needs as the most important since all the other needs become secondary until these needs are met. However, most humans effectively master physiological need-satisfying skills and then are freer to begin mastery of the skills required to satisfy the second level of needs that is "Safety needs". Cancer may threaten fulfilment of even those basic needs. Physiological needs are the most pre-potent of Maslow's needs pyramid and if these are not met as may be in the case of terminal cancer patients, other needs become meaningless and non-essentials. Cancer and cancer treatment may influence the patients' basic biological and physiological needs and patients may have a need for help managing the effects of cancer and cancer treatment (Sanson-Fisher, et. al. 2000; Ferrell & Hassey, 1997)

At the next level, the needs for safety and security such as safe circumstances, stability and protection becomes important. People want to experience structure, order, predictability and control of their lives. These needs can be fulfilled by the family and society (e.g., police, schools, business and medical care). For example, emotional security, financial security (e.g., employment, social welfare), law and order, freedom from fear, social stability, property, health and wellbeing (e.g., safety against accidents and injury). While five-year survival is quite common with most of the cancer types (Allemani, 2015), still many people perceive cancer diagnosis as a death sentence. Cancer patients also feel loss of control over their life. Cancer raises fear, anxiety and insecurity that accompanies the diagnosis. It may cause emotional feelings of anxiety, distress and uncertainty, fear of loss of control, changes in daily routine and stability and fear of recurrence and it is important to feel free from such fears to satisfy the needs for safety and security and for one's physiological health issues. Understanding these needs throughout their continuum of survivorship is crucial and clinically of high importance to offer survivors timely and effective interventions since challenges may exist beyond the initial diagnosis and treatment period as well (Stanton, et. al., 2015; Rowland & Bellizzi, 2014; Gilbert, et. al., 2008).



After physiological and safety needs have been fulfilled, the third level consists of '**belonging needs**'. Humans as social beings have a need for affectionate relationships and love, to belong to a group, to have friends, a family and a sense of community. Cancer has an impact on interpersonal relationships and sexuality which may lead to feelings of social isolation and loneliness or even depression (Adams, et. al., 2017).

Esteem needs constitute the fourth level which he classified into two categories: (a) **esteem** for oneself (dignity, achievement, mastery, independence) and (b) the desire for reputation or respect from others (e.g., status, prestige). The lower esteem is the need for respect from others, recognition, reputation, and status. The higher one is self-esteem. Humans have a need for self-respect. They want to feel good about themselves, valuable, competent, confident, independent and free. Maslow indicated that the need for respect or reputation is most important for children and adolescents and precedes real self-esteem or dignity. Many women experience a loss of confidence after having breast cancer and might avoid engaging socially due to low self-esteem due to changes and disfigurement in the body particularly after treatment such as surgery scars, changes in weight, skin, hair and the way one feels about her body. It is not uncommon for cancer patients to feel social rejection due to altered physical appearance for instance; after surgery (Arroyo & Lopez, 2011).

Maslow (1968) named the lower four levels of needs (physiological, security and safety, belongingness and esteem needs), deficiency needs or D-needs. Satisfying these needs is important in order to avoid unpleasant feelings or consequences

The fifth and highest level is **self-actualisation** which is based on a need to grow, to become what one is meant to be, and to develop one's potentials and capacities. These needs refer to the realization of a person's potential, self-fulfilment, seeking personal growth and peak experiences. Maslow (1943) describes this level as the desire to accomplish everything that one can, to become the most that one can be. As per Maslow (1968), growth is defined as the various processes which bring the person towards ultimate self-actualisation. Individuals may perceive or focus on this need very specifically. For instance, one individual may have a strong desire to become an ideal spouse, another may desire to be expressed economically, academically or athletically. For some, creatively, paintings, pictures or inventions are ways to



express self-actualization. However, reversing the order of importance of needs is the exception rather than the rule. Although mastery of lower-level needs frees a person to invest more effort in satisfying the next high-level needs, if there is too much deprivation of the low-level needs, the person will become reinvested in the satisfaction of that lower-level needs. Cancer may pose existential questions about life and death for instance purpose of life or self-actualization with an opportunity for personal growth or self-reformulation (Mayan, et. al., 2006).

2.2.3 CANCER PATIENTS' NEEDS – AN OVERVIEW

Need in health care is caused by some form of "disturbance in health and well-being" (Donabedian, 1973). According to the World Health Organisation (1948), "Health is a state of complete physical, psychological, and social wellbeing and not simply the absence of disease or infirmity." This definition of health has not been amended since 1948. Cohen (1996) described healthcare as a basic human need. He further mentioned that if one does not receive this care, he or she will greatly suffer from it. Current medical literature accentuates the importance of health dimensions described by WHO from the perspective of patients to understand and assess their needs and improve their quality of life. Needs assessments are designed to identify needs, measure their importance, determine how well they are met and the factors that influence them (Lipscom, Gotay & Snyder; 2005). However, identifying the unique needs of the growing number of cancer survivors has been identified as a challenge (Alfano & Rowland, 2006). Cancer is a complex and devastating disease both physically and emotionally. Cancer experience depends on one's perspective and can be framed from various perspectives whether it is a patient, healthcare provider, researcher or a family member (Seyfried, 2012). Cancer survivors' needs can be influenced by a number of psychosocial issues for instance socioeconomic status, stage at the time of diagnosis or cultural norms and beliefs. Some of these factors have been reported as influencing the survival rate of breast cancer in some countries (World Health Organisation, 2015; Seedhom & Kamal, 2011).

Patients' needs differ individually throughout cancer trajectory and vary with the situation that cancer survivors experience and disease outcome. Folkman & Geer (2000) illustrated that the individual's response varies based on one's capacity to perceive how one thinks and judges the outcome of an event on his well-being. Assessment of cancer survivors' varying needs offers an opportunity to explore the individual preferences, identify areas of unmet needs and the gaps



in cancer care services (Schouten, et. al., 2019; Cossich, Scholfield & McLanchlan, 2004). Such needs reflect coping demands of affected individuals throughout their cancer journey (Li, et. al., 2013) and ominously associated with psychological distress and poor outcome (Adler & Page, 2008). Hewitt, Herdman & Holland (2004) illustrated that the level of needs and distress varies individually and over the course of disease. Zimmermann (2015) further addressed the aspects of cancer survivorship that pertain to how psychological factors and the surrounding social environment influence the wellbeing of cancer survivorship strategy.

A person's health status is widely considered to be closely associated with subjective wellbeing (Ngamaba, et. al., 2017). For many, health is viewed in biomedical terms as the absence of disease and disability and is determined primarily by physiological factors. A behavioural perspective, on the other hand, acknowledges the importance of this medical model but sees health as being influenced by the way in which people live their lives and therefore recognises behavioural as well as physiological determinants. An additional perspective on health is socioeconomic approach which sees health as being primarily influenced by the social and economic environment within which people live and the constraints and opportunities such as structural factors create (Fines, Saunders & Forbes-Burford; 2013).

Most doctors will consider needs in terms of healthcare services that they can supply. Patients, however, may have a different view of what would make them healthier, for example, a job, a bus route to the hospital or health centre, or decent housing (Wright, Williams & Wilkinson; 1998). The government perspective of health needs is inherently political. It requires trade-offs and priority setting of health services while incorporating societal values and goals and compromising among stakeholders including the population since resources are scarce. Having said that, developing basic services and financial protection such as universal health coverage to all may results in major improvements in outcomes (American Cancer Society, 2021c). Since UAE is a high profile sustained country, prioritizing survivorship programming in a systematic way is imperative, given that, there are ongoing partnerships with some of the top ranked healthcare entities worldwide for instance, Johns Hopkins International, Cleveland Clinic and Mayo Clinic.



Cancer survivorship is the major aspect of cancer care that gives rise to multiple needs for cancer survivors (Walsh, et. al., 2011). Cancer is associated with substantial physical, emotional, social, and financial difficulties for affected individuals and their families (Niedzwiedz, et. al., 2019; Woźniak, & Iżycki, 2014; IOM, 2008). In the majority of cases, the diagnosis of cancer is either preceded by a period of gradual, nonspecific symptoms (NCI, 2004; Derogatis, et. al., 1983 or discovered by routine screening, and individuals are then thrust into a series of diagnostic testing, a range of procedures and treatments that are invasive and complicated with very little warning or opportunity to adjust their circumstances. Derogatis, et. al., (1983) reported that 68% of cancer patients have adjustment disorder during transition to cancer survivorship. According to the National Cancer Institute (2004), adjustment disorder is the most commonly diagnosed emotional disorder in cancer patients. Such adjustments of a person facing a life-threatening illness such as cancer are truly complex. Often, a multidisciplinary treatment approach is necessitated, leading patients to engage with multiple healthcare professionals from several different specialties and at several different locations. It is important that healthcare professionals facilitate patients' transition into care and navigate their cancer journey to minimize their distress and maximize their clinical outcomes (Giunti, 2013; Naylor, et. al., 2011, 2009, 2005; Naylor & Keating, 2008).

According to the Institute of Medicine & National Research Council (2006) and Aziz (2007), cancer survivors face unique short-term and long-term challenges to physical and mental health, family functioning and maintenance of a healthy lifestyle. For cancer patients, survivorship is a continuous struggle due to the consequences of complex treatment process and its multitude of long-term and late emerging side effects that have significant impact on physical, emotional, sexual, social and sometimes financial disturbances throughout the post-treatment phase (Tian, Cao & Feng; 2019; Rutten, 2005; Siemsen, 2001). Jacobs & Shulman (2017) have illustrated the potential long-term effects of treatment on cancer survivors in a circular diagram which illustrates the interplay between physical and emotional symptoms and how these are filtered by psychological factors, such as spiritual and existential needs in their impact upon the cancer survivor (Figure 4). These long-term and late effects of cancer treatment may arise during or after treatment and may persist throughout a patient's lifetime. Several studies have reported that cancer patients mostly suffer from physical, emotional, social, financial and psychological disturbances to a greater extent and unmet needs were highest in the post-treatment phase compared to the other phases of cancer continuum



(Thorsen, et. al., 2011; Wessels, et. al., 2010; Wong, Yee & Turner; 2008; Echlin & Rees; 2002). On the other hand, several studies mentioned the early stages of cancer survivorship had detrimental effects on the quality of cancer patients' lives (Wulff; 2012; Jenkinson, Coulter & Bruster; 2002).

Many cancer patients suffer unnecessary pain and distress and face major impediments to good end-of-life care (Silveira, et. al., 2000; Bretsher, et. al., 1999). In fact, *In Approaching Death*, the IOM (1997) identified key patient needs such as fears of abandonment and protracted death and called for reliable, respectful care that promotes "norms of decency". The IOM (1997) reports identified as a significant deficit the lack of data on patient needs and data on needs, especially the paucity of longitudinal data of diverse populations. Literature also identified inadequate understanding of needs among diverse populations as a major problem (Nair, et. al., 2018; Herschbach, et. al., 2008; Shane, et. al., 2004; IOM, 1997).

According to Minshull, et. al., (1986), unmet needs can lead to disorientation, instability and mental crisis. Cancer patients may need help to gain a sense of control, psychological stability, hope, confidence and a positive view for the future (Soothill, et. al., 2003; Bonevski, et. al., 2000; Ferrell & Hassey, 1997). Cancer patients may experience changes in interpersonal relationships, sexuality and everyday activities which may result in feelings of social isolation. If such belonging needs are not addressed, it may result in loneliness, rejection, friendlessness, rootlessness and alienation (Minshull, et. al., 1986). Moreover, cancer may result in changes in body appearance and body image thus an individual may develop concerns about attractiveness and fear of social rejection resulting in increased vulnerability and feelings of personal inadequacy (Hosseini & Padhy, 2020; Pawijit, et. al., 2017; Holland & Reznik, 2005). In order to integrate the cancer experience into life, cancer patients may develop the need to satisfy their esteem needs and to improve self-efficacy for their adjustment process (Morse & Carter, 1996). However at least 50% of all cancer patients require emotional support and develop an ongoing psychiatric complication; 1 in 3 patients experience severe psychological distress at any one time (Ryan, et. al., 2011; Breitbart & Alici, 2009). These conditions are often not detected by health professionals. To combat psychiatric issues, distress screening in the clinical environment can be used to detect persons who may benefit from additional emotional support.







Challenges exist at almost each stage of the cancer trajectory. For instance, patients' ability to understand and retain information may vary (Butow, et. al., 1998) and upon being told of a diagnosis of cancer, patients as well as their carers will have many information needs but may not always be able to articulate adequately (Laxmi & Khan., 2013) or express their questions or retain the facts presented during the stress of receiving such devastating news (Balogh, et.



al., 2011; Bender, et. al., 2008; Okamura, et. al., 1998). Pitceathly & Maguire (2003) also found the association between lack of medical information and the development of anxiety and depression. In addition, patients' preferences and language needs when providing information, can affect the success of communicating information about cancer (Tay, Hegney & Ang; 2011). Many cancer patients do not discuss their needs with their health providers and when they do, they often omit important psychosocial concerns such as sexual concerns (Maguire; 1999; Suchman, et. al., 1997). Patients who seem unwilling to discuss their disease or feelings, showing a preference to seek support from family or friends, and those patients who give only subtle indications about their information needs (sometimes based on cultural norms and beliefs) also deter effective interchange (Hagerty, et. al., 2005). However, partners and family members of an individual diagnosed with cancer who provide crucial support, they also are at risk for serious psychosocial difficulties and psychiatric morbidity such as, lack of support network of their own, relationship difficulty with the patient, failure to reveal their own worries or reluctance to seek help (Woźniak, K., & Iżycki., 2014; Pitceathly & Maguire., 2003). Young siblings may struggle with feelings of concern for the sick sibling who is battling with lowered self-esteem, fear for their own health or feelings of being less valued (Houtzager, et. al., 1999). Therefore, generous attention and caring support should be offered to support them with the care process. Some patients do not share their needs because they believe that pain, grief and suffering are inevitable with cancer. On the other hand, some patients believe that clinicians do not want to address those needs in the absence of clinician inquiry or concerns required to develop a meaningful dialogue (Bober, et. al., 2016; IOM, 2008; Brédart, et. al., 2005; Gretchen, 2005; Wen & Gustafson, 2004; Schulz & Beach; 1999; Ford, et. al., 1996). The growing trend could be the outcome of pressure to increase productivity, contain cost or patient volume (Babbott, et. al., 2014; Fiscella & Epstein, 2008; Mechanic, 2003; Ross, Steward & Sinacore; 1993) or it could be a fundamental health system problem that do not truly understand the needs of patients (Lipscomb, Gotay & Snyder, 2005; Wen & Gustafson; 2004). There is substantial evidence of lack of information needs and support affecting patient care (Schulz & Beach; 1999; Vitaliano; 1997; Peters-Golden; 1982).

Cancer patients' needs change as their condition changes, for instance, as the patients return to their normal routine, some of the initial intense reactions tend to decrease, with intermittent periods of increased intensity (Gorman, 2018). A cancer patient is concerned about higher needs such as fear of the unknown or of newly diagnosed cancer as long as basic needs are met.



If the higher needs are met, one can seek esteem needs such as needs which are meaningful for an individual. Most individuals began examining and reviewing their lives soon after diagnosis while looking for meaning in what happened to them. This period of intense existential analysis can be difficult and painful for patients and their families. Everyone needs to view life as existentially meaningful; however, cancer undermines this effort (Northouse & Northouse, 1996).

Studies further indicated that cancer patients' needs depend on which phase they are in the cancer care trajectory (Kendall, et. al., 2006; Squiers, et. al., 2005; Rutten, et. al., 2005). Thorsen, et. al., (2011|) suggested that cancer patients' needs are also influenced by the type of treatment they receive and presence of other comorbidities. The coexistence of comorbidities has substantial implications for treatment decisions and treatment outcomes (IOM, 2013; Geraci, et. al., 2005; Extermann, 2000) Cancer patients' needs also vary according to demographic and clinical patient characteristics (Østerbye, et. al., 2010; Wessels, et. al., 2010).

The assessment of needs for cancer care is a critical step in providing high quality cancer care and achieving satisfaction of cancer patients and their families (Wen & Gustafon, 2004). Identifying and addressing their needs is crucial due to their significant association with both distress and the quality of life which results in increased medical utilisation and cost to both (Lam, et. al., 2011). It is critical to understand that cancer survivors' needs are not restricted to healthcare needs but include wider psychosocial and environmental determinants of health such as physical, information, social, financial, emotional and spiritual concerns. For instance, although forgiveness and reconciliation may facilitate peace of mind during end-of-life care in cancer patients (Renz, et. al., 2020). Some cancer patients may want pain relief, but others may not appreciate the importance of dealing with forgiveness before death. It is important to identify cancer patients who have the highest needs while not ignoring those who are doing their best to cope with their disease. Meeting their needs will help them regain emotional stability, social adjustment, cognitive function, body image, future perspective and physical strength (Draborg, et. al., 2009; Hove, et. al., 2006; Squiers, et. al., 2005).

In summary, there is an increase in the number of individuals who are being diagnosed and surviving with cancer (The Cancer atlas, 2021; Arnold, et. al., 2019; Allemani, et. al., 2018). According to the Cancer Atlas (2021), in 2018, there were approximately 43.8 million cancer



survivors diagnosed within the previous 5 years. Due to advancement of technology and cancer research, treatments are becoming more complex, effective yet costly. Prolonged physical or mental fatigue, cognitive limitations, difficulty mobilizing support, difficulty managing stress and anxiety, difficulty coping with a new self-image are some crucial physical and mental health consequences of cancer and its treatments that impact cancer survivorship (Hardy, et. al., 2018; Weber, et. al., 2017; Yi & Syrjala., 2017; Fingeret, et. al., 2014), and therefore healthcare providers should address not only the patients' physical needs but also other psychosocial concerns including informational, emotional, social, financial or spiritual needs. By creating partnerships with patients to consider all their needs will result in a positive outcome. Routine assessment of psychosocial symptoms and functional impairments would allow for referrals to a multidisciplinary team of providers. For instance, referrals to exercise professionals and dieticians to promote health behaviour change in survivors have the potential to help decrease the risk of mortality from comorbid diseases, control ongoing symptoms such as fatigue and depression and improve physical functioning (Cheville, et. al., 2017; Mishra, et. al., 2012; Denmark, et. al., 2005), thus shifting care from point of care to point of need. Also, timely intervention and routine follow-up assessment to prevent cancer-related problems or to treat symptoms to avoid progression and disability will likely improve survivors' functioning and well-being (Alfano, et. al., 2019). This includes surveillance for recurrence, screening for new cancers, assessment and management of physical and emotional long-term and late effects, health promotion and care coordination, etc., (Alfano, et. al., 2019). Routine monitoring of chemotherapy-related symptoms will likely improve quality of life, reduce emergency department visits, drug adherence and improve survival. (Basch, et. al., 2017; 2016). Moreover, psychosocial interventions can also help survivors maintain employment contributing to better financial outcomes (Alfano, et. al., 2017).

2.2.4 CANCER NEED ASSESSMENT: THE PROCESS, ITS IMPORTANCE, BENEFITS & CHALLENGES

Understanding of health needs assessment of cancer patients requires a clear definition of need which particularly implies the capacity to benefit from an intervention. Need assessments are designed to identify needs, measure their importance and determine how well they are met. It is a systematic method of identifying the unmet health and healthcare needs of a population and making changes to meet these unmet needs. Needs assessment addresses a comprehensive



appraisal of care needs of individuals including physical, emotional, information, social, financial, spiritual and health care needs and can help identify whether or not the individuals need support and provide insights into the extent of that need (Prue, Santin & Porter; 2015). Health needs assessment is used to improve health and other service planning as many patients do not communicate concerns to their clinicians (Frosch & Kaplan, 1999; Guadagnoli & Ward, 1998; Suchman, et. al., 1997), priority setting and policy development (Pencheon, et. al., 2006).

Need assessment can be approached in much the same way as doing a jigsaw puzzle, so that different pieces are put together to give a complete picture of local health. It is evident that improvements in healthcare are unlikely to occur without much better understanding of patients' needs and factors that influence them (Wen & Gustafson; 2004). Need assessments offer a number of advantages by enabling direct assessment of patients' perceived needs, thus allowing more specific identification of resources that are needed. Additionally, needs assessment can help to prioritise the allocation of services and resources to where there are most needed (Gustafson, et. al., 1993; Rose, 1990; Houts, et. al., 1988; Fletcher, et. al., 1983). UAE has free public healthcare for Emirati nationals; however, expatriates are covered for basic health needs only and pay significantly higher fees for treatment which are not covered, such as cancer. No country in the world including the UAE can provide health services to meet all the possible needs of the population. Multiple criteria play an important role in the process of priority setting of health needs for a population. The process of priority setting is often complex and requires interventions to be chosen mainly to maximize general population health, to reduce health inequalities or to deal with life-threatening situations considering both practical and budgetary constraints (Baltussen & Niessen., 2006). The disease control priorities (DCP3) and universal health coverage (UHC) are examples of person-centred approaches that aim to improve the global evidence and resources to achieve equity and to support best practices in priority setting and development of health benefit packages (Department of Global Health, UoW, 2018; Shelton, 2018). The UHC model for cancer survivors in the UAE could be designed to aim all individuals and communities receive the health services they need without suffering financial hardship by including the full spectrum of essential, quality health services from health promotion to prevention, treatment, rehabilitation, and palliative care. According to the Ministry of Health and Prevention, UAE (2021), UAE's strategic participation to the international community aims to achieve universal health coverage and sustainable development goals.



When it comes to studying cancer morbidity outcomes, three primary methodologies are applied: quality of life, patient satisfaction and need assessment (Brédart, et. al., 2013; Ware, Davies-Avery & Stewart, 1978). Quality of life evaluations assess the events of the disease for different aspects of patients' experience including patients' priorities as well as satisfaction (Haslam, et. al., 2020; Gustafson 1993; Gustafson, et. al., 1991) whereas the focus of patient satisfaction surveys is perceived quality of care issues (Al-Abri & Al-Balushi, 2014; Prakash, 2010; Ware, Davies-Avery & Stewart, 1978). Needs assessment encompasses both quality of life and quality of care concerns when the impact of disease on patients' needs is assessed (Bonevskie. et. al., 2000). Regardless of the content, the common goal is to acquire information about health outcomes pertinent to assisting in the improvement of the quality of patient care and the reduction of cancer-related morbidity (Bonevskie. et. al., 2000; IOM, 1999). Modern medicine has also slowly recognised the importance of the perspective of the patient in health care and has commended more investigations to understand the importance of the interrelationships among health needs, satisfaction, and quality of life (Asadi-Lari, Tamburini, & Gray, 2004). The shift toward providing high value cancer care has placed an increasing importance on meeting patients' needs and preferences.

Literature has identified patients' supportive care needs at different phases pf cancer treatment and follow-up, for instance, at the start and end of palliative care, while receiving chemotherapy and radiation therapy and following cancer treatment (Ullrich, et. al., 2021; Sharma, et. al., 2019; Amatya, et. al., 2014). According to Sanson-Fisher, et al. (2000), time since diagnosis and the time since last admission for treatment correlated with some need in the physical and daily living domain. Nevertheless, working with patients at all stages to meet their current needs helps to optimize quality of life throughout the disease trajectory since concerns and needs continue to fluctuate based on the phase of the diagnosis and individualized patient and family priorities and needs (Payne & Murphy-Ende, 2019).

To address cancer survivors' needs, one should understand that cancer survivors are a heterogeneous group. No two human beings react in quite the same way – if we ask them what they have perceived they will give very diverse answers (Adler, 1954; 1927). As cancer differs from one another and person to person, similarly, the individual responses to the disease, cancer treatments and their adverse effects as well as their needs are unique and vary from person to person. Alder (1927) believed the individuality and uniqueness of people is explained in what



they perceive and how they perceive it. Cancer patients strive to ensure their needs are fulfilled; however, when disease occurs, there may be times their daily needs are more likely to change as the course of the disease changes (Dudas & Carlson, 1988). This new situation creates new demands with anxiety and uncertainty of the future. Cancer survivors may need new information or new skills to meet their needs and to deal with the illness. If these needs remain unmet, an individual may continue to feel emotional distress and difficulties (Fitch, 2008).

Richardson, et. al., (2007) has argued that a rigorous and systematic needs assessment is the crucial first step in integrated and patient-centred cancer care. The assessment of needs for cancer care is a critical step in providing high quality care and achieving cancer patients' and families' satisfaction (Wen & Gustafson; 2004). Needs assessment in cancer patients is an ongoing process which is recommended to be carried out from pre-diagnosis to cure, progressing disease or death into bereavement (Richardson, et. al., 2007).

It is important that cancer survivors' need assessment involve active, explicit and systematic identification of their unique needs rather than an ad-hoc implicit response to demand. The assessment of cancer patients' needs can be made clearer by differentiating the issues into needs, demands and supply (Pencheon, et. al., 2006). While assessing cancer survivors' unique needs, it is critical to make distinction between individual needs and the wider needs of the community. Hence, if individual needs are ignored then there is a danger of a top-down approach to providing health and other cancer services, reflecting what a few people perceive to be the needs of the population rather than what they are (Pencheon, et. al., 2006). It is also empirical to understand that cancer survivors' needs are not static, and any health assessment only provides a snapshot of the needs of the local population. These needs are always changing (Wright, Williams & Wilkinson; 1998) and it is important to evaluate the impact of these assessments by reviewing and updating the process of these evaluations.



Benefits of Need Assessment related to Cancer

Accurate and effective needs assessment among cancer patients can assist in prioritizing care needs, allocating resources to the areas and individuals that need them most, developing more appropriate and cost-effective cancer care strategies to improve cancer survivorship experience and quality of cancer patients' lives (Bonevski, et. al., 2000). Sines, et. al., (2009) highlighted some of the key benefits of need assessment related to cancer as listed below:

- Better understanding of cancer care and cancer services issues faced by cancer survivors.
- Better understanding of cancer equalities within a community.
- Evidence based planning and development of interventions such as Cancer survivorship programs.
- Evidence-based prioritisation of cancer care issues.
- More effective and equitable allocation of cancer resources.
- Create shared understanding of needs and priorities between cancer survivors and cancer care providers.
- Cancer patients' involvement in decision making.
- Development of public skills of cancer professionals such as reflective skills, planning skills, project management, communication and research skills etc.

Challenges of Need Assessment

Healthcare need assessment is a complex process. Although there are clear benefits resulting from the improvement in healthcare by the need assessment, there exists a danger that the target- driven culture of health services with pre-determined priorities may limit the opportunities to implement relevant decisions associated with findings emerged from the need assessments. Commitment to provide these resources may not always be forthcoming and timescales may not match management priorities and deadlines (Macdowell, et. al., 2006; Cavanagh & Chadwick, 2005). There are other major obstacles associated with cancer patients' need assessments. For instance, many healthcare providers often overlook the needs of family members and do not include the full set of patients and family needs as they under-appreciate the role of family or care workers who help patients cope up with cancer. Moreover, many



assessments do not document the changes in needs over time. Most need assessments identify how well needs have been met (typically some variant of a Likert scale) but few offer guidance on how to meet the needs. In addition, there is little understanding of how best to use needs assessment data to improve health care systems. Need assessment requires reflection and expects professionals to critically evaluate existing issues, perceptions, services, practices and norms. For conservative societies and cultures such as UAE, the process itself could be challenging and resultant findings arising from those assessments may require a fundamental shift and may be difficult to implement since cultures and norms can also act as a constraint to change (Row & Carey, 2004). Moreover, most cultural processes lead to the expectation that behaviour reinforces the mechanisms by which culture influences individual level psychological mechanisms such as needs, perceptions and belief (Kitayama, 1992).

2.2.5 NEED ASSESSMENT INSTRUMENT FOR CANCER

Unfortunately, very little information is available on how instrument developers went about identifying needs. Most instruments employ a self-reported format often sacrificing the richness of detail obtained, but on the other hand, they are relatively inexpensive and require much less time than interviews (Holland & Zittoun; 2012). The Likert-type scale is widely applied to measure importance and satisfaction of needs; however, it may produce "ceiling-effects," in which most needs are rated as very or somewhat important. This low variability among highly rated needs results in low separation of the needs and provides no guidance on where scarce resources should be concentrated (Wen & Gustafson; 2004). On the other hand, interviews or semi-structured interview approaches reduce missing data and might be employed in a clinical setting. Lipscomb, Gotay & Snyder (2005) cited the need to be more specific about the structure and process that is used in semi-structured interviews so that they facilitate the identification of needs by asking about specific areas rather than expecting participants to identify needs that they may not have reflected upon until asked.

Asking a cancer patient to think back to specific stages in their disease experience and to critical incidents is equally challenging. Retrospection may interfere with accurate assessment of their needs, given the challenges faced during the times of stress. For instance, a breast cancer patient was asked to think back to the time when she first felt the lump and was first told about the bad



news however, she said that the surgeon talked to her for an hour, but she remembered nothing and all she wanted was to go home (Lipscomb, Gotay & Snyder; 2005). This experience might lead to the recommendation that patients have time to pull themselves together in a safe environment before taking the next steps. Attention to such techniques must be given while assessing cancer patients' needs.

Ideally cancer patients' need assessment must address the trade-offs between meeting different needs, variation in needs among different patients according to their demographics such as culture, ethnicity or age (Chaturvedi, et. al., 2014; Ashing-Giwa, et. al., 2004; Aziz & Rowland, 2002; IOM, 2008; Lehmann, et. al., 1978; Craig, Comstock & Geiser, 1974) and changes over time in patients' needs during progression of the disease and/or treatment (Chua, et. al., 2019; Shahidi, et. al., 2014; IOM., 2008; Mor, Guadagnoli & Wool, 1987; Lehmann, et. al., 1978).

Cancer patients' needs change as their condition deteriorates, for instance, a need for assistance with personal care with advanced cancer patients may increase during chemotherapy treatment. Factors other than events such as personal preferences can also influence when a need becomes important enough to seek help for it. For instance, some people tend to avoid conflict while others prefer to confront it (Lipscomb, et. al., 2005). Therefore, the implementation of interventions must be modified according to the individual. The author also argued that research to understand issues such as when and why needs become more or less important is also limited.

According to Wen & Gustafson (2004), few instruments have been designed to assess the needs of cancer patients during survivorship and bereavement stages; thus, more work should be undertaken for cancer survivorship needs assessments. In a review of instruments used to assess the needs of people with cancer and their families during palliative care, it was found that most instruments developed failed to address all aspects of palliative care and that these instruments were often for research purposes and not adapted to the clinical setting (Osse, et. al., 2007). Gough (2004) and Alkire (2005) state that needs and the way they are met are often determined by experts and not people who they are supposed to benefit. These instruments often do not include the beneficiaries for patients themselves and assess the needs that healthcare workers feel are important (Richardson, et. al., 2005). In a review of assessment instruments to assess cancer patients' needs, Richardson, et. al., (2007) argued that the process of their development



and psychometric properties were reasonably well documented in the literature; however, data on how feasible they were to use in practice was scarce. He further stated that none were found to be complete for all dimensions of needs assessment and most were not sufficiently well tested for use in routine care.

Moreover, future research should determine how needs change overtime with consideration of key events across different stages of cancer across the disease continuum. More emphasis must be placed on how needs assessments have good test-retest reliability, are sensitive enough to detect differences and focus on a set of needs that are mutually exclusive and exhaustive. Research should be directed at defining needs and ways of meeting needs more operationally and move need assessment beyond simply rating "extent of need" to identifying evidencebased ways to meet needs (Lipscomb, Gotay & Snyder; 2005). To improve cancer care, health systems are required to meet cancer survivors' needs and therefore health systems should identify these needs and find a way to prioritize these. Slade, et. al., (2004) emphasised on a wider approach to care with an assessment of the individual's needs as an integral part of the care process. In order to provide adequate patient care, health systems need to assess the health needs of their populations, and this should include the social, financial or cultural factors that can be potential barriers (World Health Organization (WHO), 2007; Kay, Bengoa, & Assal, 2006; Bobadilla, 1996). There is a need to continue to develop and test instruments that could provide the holistic view of patients' psychosocial needs and have the necessary attributes for effective operation of cancer care delivery and cancer research.

The above section discussed the concept of need in general, health needs with a focus on cancer survivors' needs, health need assessment with benefits and challenges associated with the process. The next section provides an overview of gaps in the current literature regarding cancer care and cancer survivors' needs.

2.3 GAPS IN THE CURRENT LITERATURE

Cancer remains a significant scientific, clinical and societal problem across the globe. With the advanced strategies to detect cancer early and treat it effectively along with the aging population, the number of individuals diagnosed with cancer are living longer and thus continuing to rise (Aziz, 2006; Aziz & Rowland, 2003; Aziz, 2002a). According to the



American Cancer Society (2021b), 68% of adults diagnosed with cancer can expect to be alive in 5 years in the absence of other competing causes of death. However, research reveals that the quality of survival is influenced by medical and sociocultural factors such as psychos ocial and behavioural interventions, active screening behaviours and healthier lifestyles may also play an integral role in the overall survival (Demark-Wahnefried, et. al., 2005; Demark-Wahnefried, et. al., 2000). The goal of this research is to focus on cancer survivors' concerns and needs encapsulating their health and life experience beyond the acute diagnosis and treatment phase and throughout survivorship and to understand the factors that influence their survivorship experience.

The following section of the document discusses the issues relevant to cancer care focusing on survivors' concerns, needs and survivorship, while exploring the emerging research needs and gaps in different aspects of cancer care including evidence based-high quality cancer care, survivorship care, cancer patients' psychosocial needs, instruments to assess cancer survivors' needs, cultural association with needs and the current research gaps in the United Arab Emirates with regards to cancer survivors' needs and in the field of cancer survivorship which could be articulated into further knowledge and research.

2.3.1 EVIDENCE-BASED HIGH-QUALITY CANCER CARE

Literature provides considerable evidence of gaps between scientific evidence and clinical practice in many fields of healthcare (Cochrane, et. al., 2007; Buchan, 2004; McGlynn, et. al., 2003 & Grol, 2001). Grol (2001) quantified the percentage of patients as 40% who fail to receive treatments that are shown to be effective. McGlynn, at. al., (2003) further supported the fact that 20% to 25% receive unnecessary treatment which are not needed or those are potentially harmful. The IOM report (2008 & 2012) further recognised the lack of evidence-based decision making in cancer care stating that the increase in the complexity of cancer has impeded the ability of clinicians, patients and their families to formulate plans of care with the necessary speed, precision and quality. However, given increased acknowledgement of the need to address evidence-based practice gaps in cancer care, it is also expected that efforts to ensure effective translation of knowledge into clinical practice also have increased over time. However, Bensing (2000) & Groopman, (2007) suggested that evidence-based care focuses primarily on groups of patients rather than individuals and often fails to consider subgroup



analyses, incorporation of patient values and preferences. It is critical to develop evidencebased standards of care and clinical practice guidelines to improve the quality of cancer survivorship. Research is also essential to design new cancer therapies that could result in low recurrence rate and reduce the treatment related adverse effects. However, the appropriate timing and frequency of screening for new cancers and complications of cancer treatments are still unclear (Haylock & Curtiss., 2019). It is important to design and conduct guidelines which are evidenced-based and have been evaluated for effectiveness and impact.

According to Haylock & Curtiss (2019), quality survivorship care is not clearly defined. In 1999, the IOM released the report 'Ensuring Quality Cancer Care in which the US National Cancer Policy Board stated that 'reasons for failure to deliver high-quality care have not been studied adequately'. The report made recommendations that clinical practice guidelines be developed and implemented to ensure optimal care is provided to patients and that the quality of care provided be measured and monitored. The report further stated that the lack of evidence-based practices impedes clinicians' adoption and reporting of adherence to the standard guidelines.

Evidence-based practice is crucial to ensure accountability, quality of care and patient's and improved survival, mainly quality of life. Reducing the gap between evidence based and clinical practice results in reduction in patient morbidity and mortality (Saslow, et. al., 2012; McCullough, et. al., 2011; Komajda, et. al., 2005) as well as lowering healthcare costs (Shapiro, et. al., 1993). However, clinical practice and guidelines can differ greatly in terms of quality and applicability. This is due to several reasons, for example, variation in methodology of development with different quality standards and massive delays in the development cycles which results in guidelines that are already outdated at the moment of publication (Classen & Mermel, 2015). Moreover, there is lack of evidence on specific topics or the focus on followup care and recurrence detection rather than on the whole care process for instance, breast cancer (Spronk, 2017). Spronk (2017) further suggested the need for development and adoption of high-quality evidence-based guidelines on site specific to support cancer survivorship care. Risk-based screening for cancer survivors helps ensure that those at high risk are followed more closely than others (Dood et. al., 2018). Yet, little research exists to understand the personal factors that affect an individual's susceptibility to long-term and late effects (Nekhlyudov, et. al., 2017). Clinicians must weigh the risks of second cancers and



complications from treatment to determine screening schedules and must use established evidence-based associations between therapeutic exposures and late effects to identify high-risk populations that benefit from frequent screening measures (Bhatia, Armenian, & Landier, 2017). This knowledge could identify preventive interventions and improve risk assessment strategies and outcomes (Linschoten, et. al., 2018; Myers, et. al., 2017).

The IOM (2013) recommends high-quality cancer care across the care continuum from diagnosis and treatment to maintaining the health of survivors and providing end-of-life care consistent with patients' needs, values, and preferences. According to Zapka, et. al., (2003), an opportunity to improve the quality of cancer care exists in all steps of care delivery, as well as in the transitions between the types of care. The provision of patient-centred care planning, palliative care; the prevention and management of long-term and late effects of cancer treatment; and family caregiver support should span the cancer care continuum from diagnosis through end-of-life care. The full cancer care continuum must also include the domains of prevention and risk reduction and screening of psychosocial concerns. However, despite acknowledging psychosocial assessment and management as important priorities for high-quality cancer services, unfortunately, in some areas of cancer care delivery such as palliative care, the dissemination of psychosocial care guidelines is lacking. Health providers are not mindful to conduct full psychosocial assessment and tend to only partially endorse spiritual and cultural assessment and management (Bickel, et. al., 2016).

According to the Institute of Medicine (IOM, 2011), there are several challenges associated with the provision of high-quality cancer care. For instance, the fragmented nature of cancer care delivery is generally not integrated across the continuum from diagnosis to end of life. In addition, continuing geographic, financial, and social barriers prevent patients from seeking and receiving multidisciplinary care planning and comprehensive cancer care. There are ongoing disparities, including late-stage diagnoses and poorer outcomes for racial and ethnic minorities with cancer. Many cancer patients are not informed about their treatment options and their preferences are not elicited. Many cancer patients have no understanding of their treatment goals. Moreover, quite often they do not know what treatments they have received or the consequences of their treatments for their future health. Furthermore, high-quality evidence is lacking for effectiveness in many applications of cancer care including personalised medicine (Ngai, Lucas & Jason, 2018).



The knowledge about cancer patients' preferences is critical to provide patient centred care (Laryionava, et. al., 2019) and the IOM (2001) have called for shared decision-making and accommodation of patient preferences to improve overall health care quality. Yet, the current literature does not provide an overview on cancer patients' preferences (Petersen, Knudsen & Vinter, 2015). Additionally, great variability in cancer patients' preferences has been reported regarding the content and format of communication from their physicians to assess patients' experience (Elkin, et. al., 2007; Helft, 2005 & Butow, et. al., 1997). Russo, et. al., (2019) argued that an overview of psychological dimensions and instruments used in patients' preferences and health-related decision studies is lacking and till date, most attention focused on how to elicit preferences with less attention to why patients prefer certain choices and decisions. Further challenges exist with the increase in cancer incidence and survival placing new demands on the cancer care delivery systems. According to the National Cancer Institute (2013; 2012; 2007), most people diagnosed with cancer, die from it and the majority of survivors are adults 65 year or older.

The IOM (2009) reported that the oncology workforce may soon be limited to care for the growing number of people diagnosed with cancer; this could threaten cancer care delivery and result in a major structural crisis. In addition, the escalating cost of cancer treatment makes cancer care less affordable for patients and their families resulting in disparities in access and the high-quality of cancer (IOM, 2013; Stump, et. al., 2013; Sullivan, et. al., 2011). Many cancer patients do not receive the supportive care to manage their cancer related symptoms and side effects. Most frequently this occurs because of the clinicians' lack of knowledge about how to provide this care or to make an appropriate referral based on patients' psychosocial needs.

The IOM (2013) noted that most healthcare providers do not identify psychosocial care management as an integral component of high-quality cancer care. However, despite these findings and their implications for clinical care, psychosocial problems in cancer are still minimised and underestimated (Grassi, et. al., 2017). Researchers and clinicians favour a multidimensional concept of health-related quality of life to be implemented for cancer survivors; however, the complexity of cancer treatments renders the adaptability of such models into clinical practice. Quality care improvement process and outcome metrics must be considered in oncology practice along with the capacity to build multidisciplinary partnerships



of different healthcare providers to develop and implement interventions and supportive care services to promote cancer survivors' well-being and improve survivorship.

2.3.2 CANCER SURVIVORSHIP CARE

Although cancer specialists understand the obligation to offer long-term care for their patients, yet cancer patients feel abandoned as their treatment is over and face lack of resources to seek the care they need. Primary care physicians or General Practitioners lack confidence to deal with cancer related issues and feel they are not confident to provide specialised care as it is not their area of expertise. These healthcare system issues are universal for cancer care yet understudied. Due to fragmented nature of cancer care delivery, majority of the cancer care studies have focused on specific areas of cancer care mainly diagnosis and initial treatment leaving the gap to study cancer outcomes or survivors' needs after the completion of treatment (Hsieh, Chou, & Guo., 2018; Chua, Ten & Gandhi, 2018; Stanton, et. al., 2015; Fatima, et. al., 2013).

Due to the philosophical shift in treating cancer, the focus of cancer research stretches beyond treatment, representing the shift away from the medical deficit-dysfunction model toward a multidisciplinary approach (Aziz, 2007). However, cancer survivorship research studies should incorporate curing the disease and controlling the adverse and late effects of cancer as well as psychosocial domains in the conceptual models and research designs of cancer survivorship studies. Since cancer survivorship research today had added an incredible success in early detection and cancer treatment, resulting in an increased number of survivors (Aziz, 2006; Aziz & Rowland, 2003), there is a desire and a need to amalgamate the underlying mechanism, biology along with psychosocial factors influencing an individual's cancer experience.

It is important to understand the concept of survivorship to fully understand cancer survivors' needs during the different phases of cancer trajectory as it guides the development of comprehensive cancer survivorship programs. Yet, the need for research in all aspects of cancer survivorship is fertile, wide, and diverse (Haylock & Curtiss., 2019). There were 43.8 million cancer survivors diagnosed within the past 5 years across 185 countries (Bray, et. al., 2018). As the number of survivors are increasing worldwide propelled by advances in early detection and treatment and the aging population., it is placing a global challenge for survivors and their families, employers, healthcare systems and the governments. The rising number of cancer



survivors highlight the public health magnitude of cancer survivorship as well as the importance of understanding and addressing the needs of cancer survivors. According to Canadian Partnership Against Cancer (2018), the hardest part of the cancer journey is after cancer treatment; however, most patients diagnosed with cancer experience delays in survivorship care in terms of lack of continuity, guidance and resources (Nekhlyudov, et. al., 2017). The lack of survivorship care can affect patients as they transition to a normal life following treatment (Bilodeau., Tremblay & Durand; 2019)

The science of cancer survivorship care is evolving yet in its infancy and little is known to guide best practices for studying groups of survivor populations, models of survivorship care outside of academic centres, health disparities or long-term biopsychosocial and spiritual consequences of cancer and cancer treatment (Haylock & Curtiss., 2019). In 1985, cancer survivor Fitzhugh Mullan, published an article about his own cancer experience called "Seasons of Survival: Reflections of a Physician with Cancer" in The New England Journal of Medicine which became the foundation for recognizing the stages of the cancer journey. Mullan identified the "seasons" of cancer survivorship as:

- Acute survivorship, which involves diagnosis and initial treatment.
- Extended survivorship, a time of watchful waiting with celebration, uncertainty, and transition.
- Permanent survivorship, the season of gradual confidence that there would be a future free of cancer.

According to Mullan (1985), the availability and widespread use of curative and effective treatments would lead to a low likelihood of recurrence and longer survival times; however, the potential impact of late and long-term adverse physiologic and psychosocial effects of treatment was not described. Consistent with the shift in new knowledge emerging from the scientific bodies and further advancement, it is necessary to incorporate the evolving paradigm of cancer survivorship research into Mullan's original description of the survivorship experience (Aziz, 2007).

Following the establishment of the US National Coalition for Cancer Survivorship in 1986, National Cancer Institute Office of Cancer Survivorship in 1996 and the formation of National Cancer Policy Board within the Institute of Medicine (IOM) of the National Academies the



next year, many studies have been conducted with a significant upsurge on cancer survivorship (Urquhart, et. al., 2021; Fitch, 2019; Takahashi, 2016; Rowland, et. al., 2013; Harrop, et. al., 2011, Aziz, 2007, 2002). Besides the increase in the number of studies on survivorship, more research is needed primarily to address optimal survivorship care delivery. The American Society of Clinical Oncology (ASCO) established the Cancer Survivorship Committee in 2011 to provide leadership for its survivorship activities. In 2013, the committee recommended a comprehensive document with the statement Achieving High-Quality Cancer Survivorship Care outlining ASCO's survivorship agenda. The given statement proposed the creation of an inventory of existing research activities to guide strategic research planning (Jacobsen, et. al., 2016).

In 2016, a survey was conducted to identify the research studies conducted on cancer survivorship from American Society of Clinical Oncology and National Cancer Institute (Jacobsen, et. al., 2016). Out of 1,517 surveys, 699 responded and reported 714 studies of which 65% were observational and 35% were interventional. The most common age group was reported as 40-60 years in both observational and interventional studies and the time since diagnosis was less than 2 years. Although the findings may not be generalised but suggested several key gaps in the survivorship research. First, most studies on survivorship were conducted on breast cancer with a less attention to other common sites which were consistent with the findings of previously conducted studies (Jacobsen, et. al., 2016; Harrop, et. al., 2011).

Highly prevalent primary cancer sites such as colorectal, gynaecologic, head and neck, prostate and lung continue to be understudied with respect to medical outcomes (Aziz, 2007). Secondly, research on older cancer survivors, that is above 65 years has been limited. This gap is of concern because 60% of cancer survivors are reported to be 65 years of age or older (Jacobsen, et. al., 2016). In addition, most cancer survivors have reported a long-term survival rate (\geq 5 years) (DeSantis, et. al., 2014; de Moor, et al., 2013), yet research on long-term (\geq 5 years) cancer survivors does not seem to have been addressed (Richardson, et. al., 2011). Jacobsen, et. al., (2016) identified several key gaps.in cancer survivorship care research. Interventional studies on younger populations (age < 21 years) also seem to be lacking. Majority of the studies have focused on quality-of-life issues for instance emotional well-being both in observational and interventional research and there are many studies that have addressed physical concerns such as fertility and health behaviours for instance, exercise. On the other hand, only few



studies have included biologic or genetic components which are related to recurrence and adverse effects yet are poorly understood.

Clinical trials for cancer survivors also present challenges since small groups of survivors and long latency of complications often prevent randomized controlled trials from determining what modality or timing would be superior in early detection of complications or reduction in morbidity and mortality (Bhatia, et. al., 2017). Jacobsen, et. al., (2016) also highlighted the need of research into practice and addressed identified problems with the delivery of highquality cancer care in diverse settings. Moreover, substantial gaps exist regarding the process and outcomes of survivorship care planning. An integrative review of survivorship care plan studies revealed limited evidence of improved outcomes and only sporadic implementation of care planning (Mayer, et. al., 2015). Mayer also indicated massive variation in research on survivorship care plan content, implementation and dissemination processes, and outcomes. A systemic review by Jacobsen, et. al., (2018), reported the lack of evidence that survivorship care plans improve health outcomes and health delivery; however, the review found evidence of increased patient satisfaction with the amount of information received and delivery of care.

2.3.3 PSYCHOSOCIAL NEEDS OF CANCER SURVIVORS

Cancer care is a continuum process and survivor needs are also ongoing. Survivors' needs should be monitored and served at all phases along the cancer journey. However, many studies have mainly focused on the diagnosis and treatment phase only (Tamburini, et. al., 2003; Rutten, et. al., 2005). Wen & Gustafson (2004) highlighted many issues which exist to assess cancer survivors' needs including lack of clarity on best practices in initially identifying needs; determining possible levels of performance on each need; measuring the importance of needs and employing needs assessment data in design and improvement and evaluation efforts.

Researchers are also unable to identify the most critical period of the high levels of needs during the trajectory of illness (Pauwels, et. al., 2013; Adams, et. al., 2009; Schmid-Buchi, et. al., 2008; Hodgkinson et. al., 2007). Corner (2008) argued that the literature not only fails to highlight the content but also the timing of the highest needs. A study concluded that the highest but varied needs have been observed during the treatment, yet the majority of the cancer survivors have only been able to reveal their needs after the treatment (Harrison, et. al., 2009). Whereas Knobf, et. al., (2012, 2000), reported that the end of treatment is a vulnerable time as



patients transition away from active treatment and have less contact and support from health providers. Whelan, et. al., (1997) recognized the need to provide more comprehensive services to all cancer patients at any stage of their illness. Moadel, et. al., (2007) stated that although cancer patient's needs have been identified by researchers, it is unclear whether these needs are applicable to the population as a whole. Additionally, little evidence is available on the influence of age on psychosocial needs beyond the acute phase of treatment (Thewes, et. al., 2004). Harrison, et. al., 2009 and den Bakker, et. al., 2018 reported similar levels of needs post-treatment.

Another research issue with regards to survivors' need in different phases of cancer survivorship is to understand the late versus long-term effects of cancer treatment since most of the cancer survivors today face persistent and the late effects of cancer therapies on their health, longevity and quality of life. Long-term effects refer to any side effects or complications of treatment for which a cancer patient must compensate; also known as persistent effects, they begin during treatment and continue beyond the end of treatment.

On the other hand, late effects, in contrast, appear months to years after the completion of treatment. Some researchers classify cognitive problems, fatigue, lymphedema and peripheral neuropathy as long-term effects while others classify them as late effects (Aziz, 2007). Despite the increase in the number of survivors and their longevity, we have limited knowledge about their health status and quality of life, particularly those who have been post-treatment for longer time periods. For instance, what are the most common late effects of the treatment? Who are the survivors at risk of developing recurrence or new malignancies and how to prevent it? In addition, it is important to consider what represents optimal surveillance and who are responsible to conduct the follow-up care? All these queries necessitate the need for continued research in the growing cancer survivors (Demark-Wahnefried, et. al., 2005; Aziz, 2002; Demark-Wahnefried, et. al., 2000; Mullan, 1985). Conclusions about effectiveness are also limited because of few randomised controlled trials thus more research using more rigorous research designs is needed including use of longer follow-up periods (Helgeson, 2005). For instance, to understand the outcome of psychotropic drugs to treat depression and other mental health conditions of patients with cancer. What needed is, more trials using larger populations that are studied over a longer period in order to accurately assess the drug efficacy. Studies are



also needed to compare one drug with another and with the drug plus a psychosocial intervention.

Despite the known importance of need assessment, its potential appears not to have been met (Foot, 1996; Spiegel, 1994). Similarly, the above evidence, although supports the healthcare benefits of providing survivorship care effectively, yet there are still many unanswered questions. Key questions remain about how to address certain psychosocial health problems most effectively as well as how to deliver cancer services most efficiently to the various cancer survivors who need them. Several need assessments studies have been conducted on the development of different interventions to measure these needs however have not been adapted widely (Tian, et. al., 2019; Pauwels, et. al., 2013). A study conducted in the United State reported the availability of many support services for cancer survivors exist in Connecticut state however, those are often unknown or underutilized by the populations for which they are designed (Knobf, et. al., 2012). The author suggested that these findings could be utilized to target services and resources in the future, exchange of information and communication between patients and cancer care professionals.

According to the IOM (2008), the provision of psychosocial services to all who need them is hindered in part by limitations of the knowledge base. The challenge with the psychosocial research studies is that cancer is not a single disease; each cancer site's disease and pattern is different and even for a particular cancer site, individuals' specific psychosocial health care needs may vary (for instance; in early stage versus advanced disease). Since there are more than one hundred specific cancer types, it is therefore difficult to generalize about the benefits of a particular psychosocial intervention. Thus, evidence that an intervention works effectively for some populations does not necessarily mean that the same intervention is effective for or is generalizable to all people in all situations.

Another challenge with the psychosocial research is that most of these studies have focused on homogeneous samples of patients, making interpretation of outcomes more outstanding. Additionally, for some psychosocial problems, research has not resulted in any identified effective services to resolve them. Moreover, it is unclear that a given service which is effective does not exclude the possibility that another service is more effective for the same problem or equally effective at lower cost. All these issues have contributed to necessitate the need of



psychosocial research towards a larger population covering all psychosocial domains. Questions about the effectiveness of many psychosocial services have evolved from addressing whether given services are effective to addressing whom and under what circumstances specific services are needed yet effective (Helgeson, et. al., 2005; Stanton, 2005; Cohen, 2004; Zebrack & Zeltzer, 2003).

It is also needed that understanding must be raised about the psychosocial issues of cancer survivors such as poor quality of life, fear of recurrence, poor self-esteem, anxiety and depression, relationship difficulties, body-image concerns, employment and financial hardships (Ganz, et. al., 2002; Schnoll, et. al., 2001; Gotay & Muraoka, 1998). However, there are great heterogeneity effects of cancer in individuals; older adults have been reported as having a greater need for psychosocial services (IOM, 2008). On the contrary, a Qatar study found no effect of age, educational level and other demographics on psychosocial support (Alagraa, et. al., 2015). Ganz, et. al., (1993) described the association of other socio-demographics factors such as marital and financial status in addition to age and ethnicity with survivors' needs however the extent to which the selected subgroups for example, age or stage of cancer may benefit from different types of psychosocial interventions remains unclear (Fors, et. al., 2011).

Most psychosociology studies tends to focus on a specific site mainly including breast (Ashing-Giwa, et. al., 2004; Thewes et al., 2004) limiting the scope of findings to the larger population. Moreover, most studies on psychosocial needs of breast cancer have focused on an older population (Scotté; et. al., 2018; Shahrokni; et. al., 2017; Posma; et. al., 2009; Velengtas & Dailing, 1994) therefore, findings could not relate to a younger population. Workplace discrimination and financial worries have also been reported in the same context (Fenn, et. al., 2014; Conti, 1995). However, the perceived experience of stress including more subjective measures and the extent to which cancer survivors may experience psychological stress is also unclear (Schoemaker; et. al; 2016; Andrykowski, et. al., 2008).

Additionally, couple-based psycho-social intervention to improve the quality of cancer survivors appears to be as efficacious for couple communication. However, it has not been adopted widely (Pauwels; et. al., 2012; Regan, et. al., 2012). In addition, studies on cancer patients' needs have predominantly conducted on Americans and Europeans as compared to



Asians which have reported different need patterns (White-Means & Osmani, 2017; Edib, et. al., 2016; Davey; et. al; 2016; Fielding, et. al., 2013; Wenzel; et. al., 2012; Im, E. O., 2006).

Another research gap is, cancer is recognised as having large impact on family members, they are rarely the subject of or included in research on psychosocial healthcare (Helgeson, 2005). More commonly, research guides towards the effectiveness of specific psychosocial services but offers limited evidence about whether a broad spectrum of patients including their family members benefit equally from those services in all situations (IOM, 2008). Helgeson (2005) stressed on studying the impact of variables that moderate treatment effects, for instance, many cancer survivors report manageable psychosocial distress that resolves over time without the need of formal care. On the other hand, other researchers have found that patients with higher levels of distress often show the greater reduction in symptoms when formal psychosocial care is provided (Andrykowski & Manne, 2006; Antoni, et. al., 2006).

Cohen (2004) found that the effectiveness of different types of psychosocial support in individuals with varying availability of social services can depend on the nature and extent of those supports. Due to such variability of the effectiveness of different psychosocial care, it is needed that a new generation of information must be developed with careful consideration of the nature of the sample used, type of interventions and resulting outcomes (Stanton, 2005). Another concern with measuring the effectiveness of psychosocial support is to develop a standard outcome measure by which the effectiveness of psychosocial services can be measured since 'effectiveness' is often measured using dimensions of quality of life, however there are various measures of quality of life, what one study finds effective may not be interpreted as effective by others. Research testing the effect of the receipt of psychosocial health services on physiological and clinical outcomes also could help build a conceptual framework underpinning required psychosocial support to guide, develop and offer new interventions to those who are most vulnerable (Patenaude & Kupst, 2015; Thacker, et. al., 2007). Such research should address links between different types of stress and immune system functioning and the effect of psychosocial support.

Information needs for cancer survivors have been described as high in several previous studies (Faller, et. al., 2016; Beckjord, et. al., 2008). There is also clear evidence showing the damaging effects of insufficiently addressing information and supportive care needs (Okuhara; et. al.,



2018; Trenchard; et. al., 2016). Patients who are satisfied with the information provided are better prepared for the treatment (Husson, mols & de Poll-Franse, 2011). Although it is known that the treatment adherence may be affected due to anxiety and can leave a negative impact on coping strategies (Skarstein, et. al., 2000) yet, there is still a scarcity of prospective studies exploring the longitudinal relationship between satisfaction with information and levels of anxiety (Husson, mols & de Poll-Franse, 2011; Skarstein, et. al., 2000).

As both satisfaction with information and anxiety may have significant effects as such, it seems even more important to explore their interrelation (Goerling, et. al., 2020). Moreover, most of the longitudinal studies addressing the information needs have been conducted on breast cancer (Faller, et. al., 2017a; Halkett, et. al., 2012; Vogel, Bengel & Helmes, 2008) therefore further studies are needed capturing different aspects such as patients' specific concerns regarding both content and mode of information delivery, patients' information processing styles, influences of cultural background and communication strategies and various sources of information used by the patients (Goerling, et. al., 2020). The literature also suggests the need for information related to rehabilitation, disease recurrence, health promotion and available support services particularly with regards to the survivorship period (Lewis, et. al., 2009; Mayer et. al., 2007). With regards to mode of delivery and quality of information, (Cline & Haynes, 2001) estimated that more than seventy thousand websites disseminate health information to more than fifty million people seeking such information. The authors also highlighted growing concerns about people's ability to access good quality information online since most cancer patients use the internet for health information. However, variable quality of information on the web also can create greater confusion particularly when the internet is the only source used by cancer patients (Lorence & Greenberg, 2006).

Additionally, while the final consequences of cancer have received some acknowledgment there is no research on the impact of initiatives designed to deliver financial support (Moffatt, et. al., 2012). According to the American Cancer Society (2016), 28.7% of survivors reported at least one financial problem resulting from cancer diagnosis, treatment, or long-term side effects of treatment. A cancer diagnosis can be devastating for people who are experiencing financial hardship thereby influencing livelihood making it difficult for them to participate in cancer treatment (Callahan & Brintzenhofeszoc, 2015). Many cancer treatments are often


costly and even well-insured patients can face high out-of-pocket costs putting both patients and families at risk of financial distress.

Research is needed to measure the extent of financial toxicity and to characterise financial distress in cancer patients and understand how it affects their quality of life (de Souza, & Wong, 2013). Research is also lacking on financial stresses and the consequences of this for the households, despite widespread reporting of financial strain (Hanratty, et. al., 2007). Moreover, the financial burden among cancer survivors is often overlooked in survivorship care planning. Pisu, et. al., (2014) suggested that addressing the survivors' perspectives on how to deal with this financial burden is a first crucial step to identifying the means to provide this supportive care; however, little data are available to address financial issues among them.

Several studies have suggested that financial disparities may lead to disparities in quality of life as well (Ell, et. al., 2008; Coons, et. al., 2007; IOM, 2005). Crawford (2018) reported the result of a survey which revealed that out of 1,054 study participants, approximately one-third were uninsured. Uninsured patients with metastatic cancer were more likely to identify as racial/ethnic minority, have lower income and work full-time compared to insured respondents which more often resulted in refusing or delaying treatment due to cost. Therefore, it is pivotal to gain further understanding of financial influence on cancer survivorship from different aspects such as different ethnic or cultural backgrounds

Likewise, financial concerns, spirituality and spiritual distress are also individual experiences, and both require an individual approach (Martins & Caldeira, 2018). Spirituality can help lower the effects of cancer and its treatment thus can improve quality of life. According to the National Comprehensive Cancer Network (2020), cancer patients who are spiritual, tend to be more compliant with treatment and live a healthier lifestyle. Deborah K. Davis, the director of field education at Princeton Theological Seminary, who was a chaplain at Princeton Medical Centre in New Jersey for many years stated that the spirituality is a chance to be reconnected to God, a religious tradition and a community that provides hope and strength for the cancer patient (National Comprehensive Cancer Network, 2020).

NCCN has also observed that cancer patients who are already religious often become more deeply religious being diagnosed with cancer, whereas others who were not religious sometimes seek spirituality and a connection to a power outside themselves after the diagnosis.



Cancer patients who rely on their faith or spirituality tend to experience increased hope and optimism, positive thoughts and attitudes, satisfaction and inner peace and meaning of life (Moosavi, et. al., 2019). On the other hand, spiritual distress is labelled in the existing literature as existential suffering, spiritual anguish, spiritual pain and spiritual struggle (Bates, 2016; Wilt, et. al., 2016; Exline, et. al., 2013; Delgado-Guay, et. al., 2013; Chaves, et. al., 2010). Several studies have supported the existence of spiritual distress in cancer patients during different phases of cancer survivorship (Caldeira, et. al., 2017; Gielen, et. al., 2016; Hui, et. al., 2011). Skalla & Fernell, (2015), reported that cancer patients seem more susceptible to spiritual distress when they are diagnosed during progression of the disease and at the end-of-life. Likewise, in other psychosocial domains, spirituality plays a major but ill-defined role in cancer survivorship, yet the subject has not been given ample consideration in the literature (Lee; 2019; Forouzi; et. al., 2017; Polit & Beck., 2014; Rasool, 2000).

Irrespective of studies providing scientific evidence on spiritual distress, a gap in implementing spiritual care also remains due to lack of training, overlooking or misunderstanding the spiritual aspect or meaning of spirituality and its role in cancer survivorship (Balboni, et. al., 2014; Rushton, 2014; Tiew, et. al., 2013 & Nascimento, et. al., 2010). Further research is pivotal to identify patients' spiritual needs to promote and provide holistic cancer care (Guerrero, et. al., (2011). Moreover, the literature consists of mostly qualitative studies to identify cancer patients' spiritual needs therefore the generalizability of those studies is limited. The question is whether the concept of spirituality can only be applied to religious cancer patients or those who have no religious background can also gain benefits with spirituality?

Further research with a larger population coverage and with different religions will help understand the influence of spirituality in a diverse population to target tailored interventions and to meet individual cancer patients' needs. Moreover, the expectations regarding spiritual needs in the context of cancer may vary by cultural background, this aspect of cancer patients' needs must be explored further with greater sensitivity (Astrow, et. al., 2016). While the psychosocial support including physical, psychological, informational and social support needs have been identified for cancer survivors, health care providers still lack the data necessary to help meet the needs of specific cancer survivor populations for instance; by age and ethnic background (Knobf, et. al., 2012). Therefore, additional data are needed to develop and appropriately target interventions as well as facilitate access to resources for cancer survivors



to improve overall health and optimal quality of life. Nevertheless, questions remain as to how best to describe the experience of living with, through and beyond cancer, as well as how to provide the necessary supportive care to achieve optimal quality of life for given populations (McGrath & Holewa, 2012).

2.3.4 CANCER NEED ASSESSMENT INSTRUMENTS

It is also not very clear which screening instruments should be used to assess psychosocial health needs of cancer patients or whether there is a minimum set of domains that should be considered. What is the optimal or standardised care? Are there any existing instruments that could be improved to achieve greater utility? These questions could be answered if there is a valid psychosocial screening instrument that could be used to psychosocial health of cancer patients.

Wen & Gustafson (2004) conducted the systemic appraisal of needs assessment instruments for their reliability, validity, burden and psychometric properties and found several problems including wide variation in the needs addressed, inconsistency in domains within the instrument and in items included in similarly named domains, a lack of evidence of sensitivity to change over time, failure to address the period after initial treatment for cancer and failure to examine reading levels. The authors also expressed their concerns in whether one instrument could be developed to address all areas of interest and represents an ideal assessment tool. They also suggested developing a common set of domain terms to be adopted to form the core of need assessment and building a consensus on some items to be placed in the domains.

In an oncology setting, the ability of the survey to assess the magnitude of needs may be especially important in clinical oncology contexts (Bonevski, et. al., 2000). A 1996 review of the psychometric properties of needs assessment instruments used with oncology populations revealed that perhaps one of the reasons need assessment has not been commonly used is lack of quality instruments (Foot, 1996). The review of 71 articles found few needs assessment instruments met the criteria rated against 6 principles: 1) it measures the multidimensional impact of cancer on patients' needs; 2) it directly and comprehensively assesses subjective health-related needs for help; 3) it measures outcomes within a defined temporal context; 4) it demonstrates acceptable reliability and validity of assessment instruments and methods; 5) it is user-friendly; and 6) it is system-friendly. Another MEDLINE search of literature between



1996 and 1998 failed to reveal any advances in this area (Bonevski, et. al., 2000). Another literature review conducted in 2007, the researchers identified and appraised 15 instruments which were developed from 1984 to 2004 based on their validity, reliability, responsiveness and feasibility (Richardson, et. al., 2007). The findings showed that no instrument meet all the above criteria and were a good fit for the use in a clinical setting. Moreover, some instruments primarily focused on assessing other aspects of cancer care such as prevalence or severity of symptoms rather than assessing cancer patients' needs.

There have been recent advancement in developing new instruments such as Supportive Care Needs Survey-Short Form (SCNS-SF), Cancer Survivors' Unmet Needs Measure (CaSUN), Survivors Unmet Needs Survey (SUNS), and Needs Based Biopsychosocial Distress Instrument for Cancer Patients (CANDI) (Lowery, et. al., 2012; Campbell, et. al., 2010; Boyes, Girgis, Lecathelinais., 2009; Hodgkinson, et. al., 2007). Yet, their psychometric properties have not been systematically reviewed and compared (Tian, Cao & Feng., 2019). Lately, several cancer care need assessment instruments have been developed yet most are lacking the quality criteria of comprehensive and systematic appraisal of measurement properties of cancer-specific needs assessment instruments recommended by Terwee, et. al., (2007). This suggests the need for higher methodological quality studies on instrument development for cancer-specific care needs. While evaluating the psychometric properties of 20 instruments, a systemic review of identified 37 studies reported several inconsistencies and the need for future studies with the following recommendations (Tian, Cao & Feng., 2019):

- Development of site-specific modules for site specific tumours
- Clear identification of perceived symptoms and the needs of receiving care to remove the ambiguity about whether the individuals want assistance
- Need of higher methodological quality studies on instrument development for cancerspecific care needs assessment.
- In future studies, it is beneficial to incorporate measurement error as one of the measurement properties.
- Ability of instruments to detect clinically important change over time in the construct to be measured.
- Testing of criterion validity to assess an adequate gold standard for comparison in care needs among cancer patients.



- Test-retest reliability for the assessment of instrument stability with different time intervals.
- Development of multiple hypotheses regarding correlations or mean differences before data collection to improve the methodological quality of hypothesis testing.
- Cross-cultural adaptability and feasibility of the recruited tool and cross-cultural validation using a more appropriate factor analysis method (CFA) and inclusion of more samples.
- Further improvement of already existing and promising measurements.
- Selection of the most appropriate instrument for need assessment among cancer patients to early identify and effective cancer care management.

Needs-based assessment is a vital step in achieving optimal supportive care (Hall, et. al., 2014; Wen & Gustafson, 2004). Several cancer-specific needs assessment tools have been developed of which some of the most commonly used are "Supportive Care Needs Survey-Short Form (SCNS-SF), Cancer Survivors' Unmet Needs Measure (CaSUN), Survivors Unmet Needs Survey (SUNS), Supportive Care Needs Survey (SCNS), Distress Thermometer (DT) with Problem List (PL), Cancer Survivors Survey of Needs and Needs Based Biopsychosocial Distress Instrument for Cancer Patients (CANDI)". However, no previous multidimensional needs assessment instrument has been psychometrically evaluated for its utility in a different healthcare setting (Tian, et. al., 2019). Thus, further research is needed to address the obstacles for practicing the need assessments and to establish the relative importance and significance of identified needs. Below table (2.1) provides is a comparison of psychometric properties and the use of few available instruments that are most commonly used to assess cancer patients' needs (Al-Shaaobi, et. al., 2021; Tian, et. al., 2019; Jiao, et. al., 2015; Ehmke, et. al., 2016; Iskandarsyah, et. al., 2013; Hughes, et. al., 2011; Schlairet, et. al., 2010).



TABLE 2.1: Comparison of Psychometric properties, design, strengths & weaknesses of commonly used instruments to assess cancer survivors' needs

PSYCHOMETRIC PROPERTIES OF COMMONLY USED INSTRUMENTS TO ASSESS CANCER SURVIVORS' NEEDS

0 = no results reported, - = no evidence in favour, + = limited evidence in favour, ++ = some acceptable evidence in favour, but some aspects fail criteria or not reported, +++ = acceptable evidence in favour

Factors	Survivor Unmet Needs Survey - (SUNS)	Short Form Survivor Unmet Needs Survey - (SF- SUNS)	Cancer Survivors Unmet Needs - (CaSUN)	Distress Thermometer (DT) with Problem List (PL)	Supportive Care Needs Survey (SCNS)	Cancer Survivors Survey of Needs	
Content validity	+++	+	+++	++	+++	+++	
Construct	++	++	+	+	++	-	
Internal Consistency	+++	+++	+++	++	+++	+	
Cross-cultural	+	0	0	++	++	+	
Hypothesis testing	+	+	++		+++	+	
Responsivenes s	-	-	0	0	0	0	
Acceptability	_	+	++	++	++	++	
INSTRUMENT DESIGN							
Design	First section 89 tick box questions covering 5 domains	A shortened version of the SUNS Contains 30 items derived	Contains 35 unmet need items, 6 positive change items	Patients rate their distress level from a range of 0 (none) to 10	Contains 59 items, while the recommende d short form	Contains 50 questions covering five domains, including	



	Second section closed questions	from the original 89 SUNS items	and an open- ended question	(extreme) in the past 7 days on the DT and tick off relevant problems on the PL	SCNS-SF-34 contains 34 items encompassin g 5 domains: psychological , health system, physical & daily living, patient care & support, and sexuality	physical, emotional, social, spiritual, and other domain. It also contained open-ended questions about primary source of strength their primary concern regarding their healthcare needs. Overall QOL in the past 1 week prior to the survey in addition to demographic s and clinical information about cancer are part of the survey.
		KE	Y STRENTGH	S		
Strengths	Used and validated in a large number of mixed and haematologica l cancer survivors. It appears to have high acceptability, internal consistency, content and construct validity.	Used and validated in mixed early cancer survivor patients. It appears to have good construct validity and internal consistency. Convenient tool that may have better acceptability in clinical care due to its shortened form.	Used and validated in cancer survivor. Appears to have good internal consistency evidence of face, content and construct validity	Used in cancer patients with the Problem List and is recommended by the NCCN for cancer patients at all stages of their disease including survivorship. Simple and fast to use and can easily be implemented in clinics. It has good sensitivity and specificity in cancer patients. An objective, structured and consistent way to screen for emotional	Used in cancer survivors. Convenient too to be utilised in clinical setting.	Used in cancer survivors. Convenient too to be utilised in clinical setting. Provide awareness and insight into the needs of cancer survivors. Provide leaders with valuable information needed to provide appropriate resources and plan



University of	
Salford	
MANCHESTER	
	80

distress and triage to supportive care services

supportive programs.

LIMITATIONS

Limitations	It contains 89 items. It was validated in cancer survivors 1–5 years post- diagnosis. No questions on reproductive, sexual and spousal issues. Does not contain questions on physical symptoms. No studies on validation and implementatio n in clinical practice or patient outcome	Validated in cancer survivors 1–5 years post- diagnosis. Derived from SUNS but content validity of the short form tool was not completely reassessed. Does not contain questions on physical symptoms. No studies on validation and implementatio n in clinical practice or patient outcome	Developed and validated mainly in studies included females (81%) and had breast cancer (59.2%) or gynaecological cancer (16.9%). The average age was older (average 59.6 years) and had been diagnosed with cancer 2.3 years earlier (range 1–15 years). Limited test– retest reliability. No studies on validation and implementatio n in clinical practice or patient	May not reliably identify distress in cancer survivors. Does not measure level of unmet need. Items may need to be modified or added according to patient demographics. Usefulness and relevance of the tool is dependent on how it is used and communicated . Must be accompanied by clinical review	Developed and validated in cancer patients receiving treatment. Potentially missing needs relevant to cancer survivors. Limited questions on social & practical needs	Limited usage in research and clinical setting. It was validated by members of the cancer patient education network at Mayo Clinic.
			patient			

There is limited research assessing the clinical use of the need assessment instruments. (Jiao, et. al., 2015). It is difficult to assess the validity of the needs assessment instruments as there is no gold standard tool for comparison. Thus, future research in the clinical setting is required to assess effectiveness in assessing needs. There is room for further refinement of the instrument that are currently used, for instance, reduction of its length, continued testing with more diverse populations along with the translation into other languages and ability. There is also a need to continue to develop and test the instruments that have necessary attributes for effective practice and research (Richardson, et. al., 2007). Healthcare providers should employ a needs assessment tool based on the relevance of the scale to their specific



patient population and consider the psychometric rigour and practicality of the measure. Additionally, needs assessment instruments, especially checklists, may not identify all patient concerns (Taylor, et. al., 2011) thus healthcare providers should be aware of patient needs that may not be included in the needs assessment tool. Besides other highlighted issues, good listening and communication skills are required to establish a good relationship and trust with the patient to effectively identify needs including issues they may otherwise be reluctant to discuss. Patients should be included in the needs assessment process as active participants in their own care. Nevertheless, more research is required to evaluate the most effective and appropriate process before any firm recommendations are made based on the need assessments (Jiao, et. al., 2015). The author also emphasized that future research requirements of large longitudinal studies in the clinical setting to capture all cancer survivor groups, reassess validity and reliability and assess acceptability and feasibility. Structured and well-designed instruments can reduce Allgar, inequalities as they prompt discussion between patients and clinicians; however only few are designed for the use by clinicians while attention must be given to train them how to use it (Allgar, et. al., 2018). It is thus concluded that there is a need to develop and test instruments that have the attributes necessary for effective practice and research their outcome and effects to improve the quality of cancer care and patients' life (Richardson, et. al., 2007).

2.3.5 CANCER ASSOCIATION WITH CULTURE

Research shows that a number of studies have focused on addressing cancer patients' needs (Pauwels; et. al., 2012; Lam, et. al., 2011; Sanders, et. al., 2010; Harrison, et. al., 2009), however, disparities of diverse needs associated with different ethnicities or cultures are still poorly understood, inconsistently defined and ineffectively used in practice and research (Kagawa-Singer, et. al., 2010).

While cultural knowledge is recognized as key to reducing health disparities, yet the concept of culture and its relation to cancer remain poorly understood (Kagawa-Singer, et. al., 2010). Cancer related stigma and myths about cancer are critical survivorship issues that need to be addressed which are different from country to country (Daher, 2012). Likewise, contrary to the knowledge about long-term cancer survivorship, fewer studies have focused on the transitional journey from different stages from diagnosis to survivorship which is accompanied by impaired



quality of life (Pauwels, et. al., 2013). This clearly indicates the variations in reported patterns of needs both geographically and culturally. Literature indicates that patients' needs differ according to socio-demographics characteristics of population (Coronado, et. al., 2017; Costa-Requena, et. al., 2015; Fife, Young, et. al., 2014; Skrzypczak, et. al.; 2012; Goldzweig, et. al., 2009, Lehto, et. al., 2005; Girgis, 2000; Kennedy & Robinson, 1994). Shea-Budgell, et. al., (2014) further emphasized on the need to understand the diverse information needs of this growing population of long-term survivors. Understanding of the individual's cultural needs came with experience, which helps to provide culturally respectful care (Huang, et. al., 2009).

To treat cancer holistically, it is critical to identify and develop culturally tailored interventions to meet cancer patients' unique psychosocial needs since such interventions have proven to be beneficial in improving the quality of life of cancer survivors (Ramírez-Perdomo, et. al. 2019; Kim, Chang, & Lee, 2018; Chee, et. al., 2017; Meyer & Mark, 1995; Devine & Westlake, 1995). Edib, et. al., (2016) further reinforced rendering timely and appropriate psychological and physical rehabilitation programmes in healthcare settings. Author suggested that these interventions ought to be the highest priority to support cancer survivors in the long-term adjustment of survivorship and ensure a better quality of life.

Human behaviour can have a profound impact on how cancer is managed and may also affect disease-free or overall survival, we are not currently taking this information into consideration in the systematic delivery of care. There is a growing understanding of the role that sociocultural and behavioural factors play in cancer care including perceptions about cancer, cancer outcomes, decision-making, adherence to treatment, post-treatment health maintenance behaviours and willingness to adopt appropriate surveillance (Aziz, 2007). However, there is little known about cancer outcomes among culturally diverse groups since this aspect of cancer care has been under-studied (Lwin, et. al., 2014; Palmer, et. al., 2013; Phillipson, et. al., 2012; Kagawa-Singer, et. al., 2010). In addition to recognition of psychosocial services needed for cancer survivors, it is important that the effectiveness of those services must also be evaluated in different populations. Effectiveness research on psychosocial health services has most often focused on women with breast cancer at the middle or upper idle socioeconomic levels without regards to the amount of psychosocial stress they experience (IOM, 2008). The IOM (2008) recommends that such interventions must be tested in patients from different cultural and socioeconomic backgrounds, in patients with cancer sites other than breast, across different



stages of cancer and different levels of psychosocial needs. With regards to the effectiveness and impact of any psychosocial intervention, particular attention should be given to the socially disadvantaged population, examining the effect of socioeconomic status and ethnicity on the risk for psychosocial problems since the effectiveness of intervention is often considered to be linked with the quality of life. Nevertheless, to understand the influence of culture on cancer, it is first necessary to unravel the misperception of defining race, ethnicity and culture to carefully examine the impact of these on cancer patients' lives (Kagawa-Singer, et. al., 2010). The author also concluded that the health literature lacks clear, standardized definitions for culture, thus, further clarity of cultural factors will increase the scientific basis for the clinical practice and research among diverse groups of cancer survivors and will remove associated cultural health disparities.

Proficiency in cultural sensitivity is becoming increasingly important in cancer care as one comes from his or her own culture with the established set of norms, attitudes, beliefs and behaviours. It is important that health care providers not just become more culturally competent to understand one's culture but also their own, to understand how culture can impact survivors' as well as health care providers' reaction to cancer. It is important that prospective studies measure the association between cultural cancer-related behaviours and its impact on cancer survivorship. Studies must utilise rigorous research design that yields the magnitude of an intervention to better understand its effectiveness which attempt to address the unique psychosocial needs of cancer survivors while applying multicultural health principles of participation to recognise the needs of the particular culture.

There are significant gaps that bar the oncology care community in providing comprehensive cancer care including their need assessment and intervention for the whole person and for all persons. Moreover, the cancer patient and survivor population are increasingly diverse (Ashing, et. al., 2016). Therefore, it is needed that future studies include different models of care that include cancer survivors' distinctive needs incorporating their cultural beliefs, understandings and preferences for instances, addressing stigma and shame, religious beliefs, outcome expectancies, patient and providers' communication, rights of individuals to access to appropriate health information and quality health services that recognise their cultural needs.



2.3.6 PSYCHOSOCIAL NEEDS OF CANCER SURVIVORS IN THE UNITED ARAB EMIRATES

Despite the high prevalence of cancer in the United Arab Emirates, there is yet a limited number of studies and published data on cancer survivors' psychosocial needs and supportive care (Nair, et. al., 2019; Nair. et. al., 2018, Elbarazi, 2016). Nair, et. al., (2018) stated that there is a dearth of published data but also a limited understanding of unmet needs of cancer patients in the Middle East region, particularly in the Gulf Cooperation Council (GCC) countries, consisting of Kuwait, Qatar, Oman, Saudi Arabia, Bahrain and the United Arab Emirates (UAE). Nair, et. al., (2018) argued that the assessment of unmet needs of cancer patients in the UAE will improve the supportive care services and thereby positively impact the overall quality of life of cancer survivors in the country.

Keefe, et. al., (2016), emphasized on encouraging more physicians to specialise in supportive care along with specifying the crucial role of supportive care in medical education to improve the quality of life of cancer survivors and their satisfaction about treatment. The author also suggested medication associations to strengthen guidance and the integration of supportive care in oncology services highlighting them as the unique challenges of the Middle East region. In addition, it was also recommended that pharmaceutical industries and academia work together in research to develop fast-track drug discovery, evidence-based guidelines and commercialization. Despite cancer increasing in the UAE, the public health programs in the country have focused more on prevention and pharmaceutical responses, neglecting the psychosocial clinical aspect of cancer care (Silbermann, et. al., 2012, 2013). Researchers have more distinctly reported the scarcity of qualitative studies in cancer research in this region (Elbarazi, 2016; Nair, 2012; Aw, et. al., 2011; Tadmouri & Al-Sharhan, 1985). Most research studies in the region have used a quantitative approach that lacks the exploratory aspect of cancer patients' needs (Elbarzazi, 2016). Author also reported the dearth of research in cancer patients' education as well as information needs in the UAE. Assaf, Holroyd & Lopez (2017), further confirmed that in the UAE, there are no extensive health-awareness programs due to diversity of nationalities and languages highlighting health education as a challenge due to the cultural fear of spreading bad news or reluctance of publicly discussing health issues in this sensitive culture.



2.4 CANCER DIAGNOSIS & SURVIVORSHIP

Cancer diagnosis is a life changing experience which restricts an individual's life psychologically, physically and socially (Azhar & Bruera; 2018; Usta, 2012; Stein, et al., 2008; IOM, 2008). Many people equate cancer as an early death (Straker, 2013; Röing, Hirsch & Holmstrom, 2006) however, if detected and treated earlier, cancer no longer serves a death sentence (Bush & Gorman; 2018; WHO, 2017; IOM, 2002). Receiving a cancer diagnosis is devastating news for an individual and for the family (Bumb, et. al., 2017; Woźniak & Iżycki; 2014). Prior to diagnosis, most patients do not perceive themselves to be at higher risk for the disease thus when patients or families are told that they have a diagnosis of breast cancer, their responses clearly vary; most of them experience fear, disbelief or existential crisis (Coronado, et. al., 2017; LeSeure, et. al., 2015; Wronski, 2015; Hewitt, 2004). A cancer diagnosis can influence many aspects of a person's life and disrupt one's physical and psychological state by inducing significant level of stress (Grassi, et. al., 2017; Wang, et. al., 2013). There is a complex relationship that exists between cancer and psychosocial factors that influence the survivorship experience and their quality of life (Coggin & Shaw-Perry, 2006; Carreira, et. al., 2018; Patsou, et. al., 2018; Usta, 2012;). Research shows that cancer survivors experience poor physical and mental health outcomes and poor quality of life (Carreira, et. al., 2018; Clough-Gorr, et. al., 2010; Foster, et. al., 2009; Fu, et. al., 2009).

An IOM (2006) report revealed that survivors' follow up needs are being unmet. The patients' experience with advanced cancer illustrates the implications of not designing the delivery system around a clear understanding of patients' needs (Wen & Gustafson, 2004). Patient-centred care is unlikely to occur without good understanding of patient needs and factors that influence them (Zucca, et. al., 2014).

According to the Office of Cancer Survivorship of the National Cancer Institute (2014), an individual becomes a **cancer survivor** from the time of diagnosis for the rest of her life (Riker, 2015; Denlinger, et. al., 2014). The psychosocial needs of each cancer survivor are unique and need to be identified and addressed during the different phases of care. The literature search in this study focuses on the needs and concerns of cancer patients through all phases of cancer survivorship (Figure 5). The pioneer model of survivorship was initially described by Mullan (1985) as seasons of survival with unique sets of concerns including *acute survivorship* (the



experience of diagnosis and treatment), *extended survivorship* (a time for "watchful waiting"), and then *permanent survivorship* (long-term remission). This model was later elaborated by Hewitt, et. al., (2006) proposing a period of transition named transitional survivorship reflecting transition from active treatment to different forms of adaptation throughout as following:

- Acute Survivorship: The period that includes the diagnosis of cancer and its treatment. In this phase, patients typically experience a physically and emotionally difficult time.
- **Transitional Survivorship:** This begins at the completion of acute treatment. Cancer survivor experience transitioning back to previous responsibilities including work and family.
- **Extended Survivorship:** The period of remission after surgery, radiation or chemotherapy. This is a time of careful observation and surveillance.
- **Chronic Survivorship:** This entails people who are living with cancer as a chronic disease for instance women with metastatic breast cancer.
- **Permanent Survivorship:** This period recognises the heterogeneity in the permanent survivorship. This is a cancer free period, yet one might live cancer free for many years and develop late or long-term side effects or die from natural causes or die after a late recurrence or develop a second primary.



Figure 5: Phases of Cancer Survivorship

Adapted from Miller, K., 2012. Excellent Care for Cancer Survivors



2.5 BREAST CANCER SURVIVORSHIP

According to the American Cancer Society (2021), the average 5-year survival rate for people with breast cancer is 90%. Besides having better prognosis than other cancers, many breast cancer survivors suffer from long-term psychological distress because recurrence of cancer is mostly incurable (Chiriac, et. al., 2018; Tatsuo, et. al., 2015; Peled, et. al., 2008). Fear of recurrence, physical symptoms for instance, fatigue, pain, disruption of body image, stigma of disability, sexual dysfunction, treatment related issues & anxiety, family & relative's relationship issues have been reported as the major psychosocial needs in women with breast cancer (Sabiston, et. al., 2018; Zimmermann, 2015; Turnbull Macdonald, et. al., 2012). Additionally, advancements in breast cancer disease in terms of biology, diagnostic approaches, predicting prognostic factors and treatment decision making, have contributed to the complexity of the experience and influence of women's decision-making processes and psychological adjustment towards it (Holland, et. al., 2015; Pierce, 1988). An individual's emotional wellbeing and psychological adjustments towards disease can be positively or negatively influenced by factors such as personality attributes, preferences of information or choices of treatments, family or partner's relationship, sexuality concerns, religious or spiritual beliefs, patient-provider communication and degree of physical symptoms (Lamkin & Slavick, 2014; Paludi, 2014).

There is a variation in how women deal with cancer diagnosis and treatment and adapt to the uncertainty of life after it (Hajian, et. al., 2017; IOM & NCR, 2004; Ashing-Giwa, et. al., 2004). Fitch (2008) explained how a cancer patient may enter the cancer care system at various points and move through the variety of experiences at different paces and along different pathways (Figure 6). Some may just enter at the point of screening and may never proceed or at the point of diagnosis and move through different phases of treatment. Some become disease free and may have follow-up care without any symptoms while others may experience cancer recurrence. Some require palliative care, whereas some finally result in death as a cancer outcome. Each phase of the cancer journey is associated with unique needs for instance during the diagnosis phase, upon completion of treatment, survivorship phase when they re-enter their normal living, uncertainty of having recurrence of a disease or about the progression of cancer. Not everyone copes with these issues in a similar way (Fieldler & Koppelman, 2012). Tatsuo, et. al., (2015) suggested that different care for supporting breast cancer survivors after



diagnosis should be in place. Literature shows that primary care providers often do not know how best to care for the specific concerns and needs of cancer survivors (Nekhlyudov, et. al., 2012). Thus, the IOM (2008) stressed that it is critical for healthcare providers to be aware and supportive towards the range of psychosocial needs at each stage of the continuum of care (Denlinger, et. al., 2014; Ogce, et. al., 2007). Moreover, at each stage survivors' concerns and the needs must be understood in the context of other related issues, for instance, cultural or financial issues which may influence patients' quality of life and wellbeing (Ng, et. al., 2017; Costa-Requena, et. al., 2015).







Fitch, M. I., (2008). Supportive care framework



2.6 BREAST CANCER SURVIVORS' CONCERNS

2.6.1 SURVIVORSHIP CONCERNS AT DIAGNOSIS

Payne (2014) described cancer as a universally stressful experience. Receiving a cancer diagnosis is associated with immense negative feelings and distress for many (Nail, 2001). Although distress is highly prevalent and diverse at all stages of cancer (Holland & Alici; 2010) yet anxiety, uncertainty and psychological distress are universal elements reported during the diagnosis phase particularly before and after the mammography and after the pathological findings (Ghaemi, et. al., 2019; Ramírez-Perdomo, et. al., 2018; Jennifer & Yingling., 2018; Geyikci, et. al; 2018; Carreira, et. al., 2018; Holland, et. al., 2015). Ganz & Stanton (2015) argued that the most common psychological denominator of cancer regardless of any stage of the disease is the experience of life threat and uncertainty. Cancer diagnosis can disrupt one's life by threatening a person's sense of security (Bush & Gorman; 2018) and can create negative consequences on physical and psychological well-being (Epplein, et. al., 2011). Most women acutely experience fear and disbelief after being diagnosed with breast cancer (Tsaras, et. al., 2018; Coronado, et. al., 2017; Wronski, 2015; IOM & NRC, 2004). Apart from accepting the diagnosis, breast cancer survivors also grapple with self-esteem issues (Türk, et. al., 2018; Landry, et. al; 2018; Ng, et. al., 2017). According to Holland (2002), no disease has sustained as strong as a negative stigma as cancer.

Spittler, et. al., (2012) found that after being diagnosed with breast cancer, women were exposed to numerous healthcare professionals to discuss the disease process, prognosis and treatment options. For many women this process to acquire new information could be overwhelming resulting in making hasty decisions about choice of treatment as well as treatment centres (Bush & Gorman; 2018). The decision about choice of treatment can be quite challenging (Levit, et. al., 2013). In addition, where to have these treatments builds up further anxiety as one must deal with the little knowledge or information about the disease (Behboudifar, et. al., 2018; Hauffman, et. al., 2017; Katz, 2015; Canil, et. al., 2012). The ongoing uncertainty about the future affects how the individual views the future (de Vries & Stiefel, 2014).



Delivering bad news is one of the most challenging tasks oncology professionals must undertake (Buckman, 2005). Good communication is the key aspect of patient care thus, physicians should be sensitive toward disclosing the diagnosis considering the patient's emotional state by hoping for the best and preparing for the worst (Prip, et. al., 2018; Sarafis, et. al., 2013; Dias, et. Al., 2003). Considering the importance of communication, many medical schools have implemented training programs delivering news to the patients in their curricula (Bousquet, et. al., 2015; Reed et al., 2015). Nevertheless, how compassionate or skilled the health professionals are in delivering bad news, patients may experience intense emotional reactions (Shell & Kirsch, 2001). Healthcare professionals, including oncology nurses, need to develop skills in accurately and empathetically presenting information (Azadi, et. al., 2018). In doing so, they can maintain hope regardless of the patient's prognosis.

When cancer occurs in an individual's life, it also enters the lives of family members (Bush & Gorman; 2018). According to Zaider & Kissane (2015), cancer diagnosis marks a major transition in the family's life. Nevertheless, family member's support plays an integral part in cancer patients' psychological and mental wellbeing (Laryionava, et. al., 2018; Segrin, et. al., 2005) and play a key role in the support system for most patients (Bush & Gorman; 2018).

2.6.2 SURVIVORSHIP CONCERNS DURING TREATMENT

Once diagnosed with breast cancer, the initial decision-making lies with the choice of therapy for most patients, either to choose local tumour removal with radiotherapy or mastectomy with or without reconstruction (Manning & Sacchini, 2016). Prophylactic contralateral mastectomy which is defined as the removal of a healthy breast in a patient undergoing mastectomy for ipsilateral cancer is often considered as part of the initial plan of treatment (Boughey, et. al., 2016).

For early-stage breast cancer (0 to IIIA), radiation therapy for six weeks is the standard approach (American Cancer Society, 2019b; Holland, et. al., 2015). Fatigue is the most common side effect associated with radiotherapy in addition to skin and sensation reaction as well as swelling of breast (Nicholas, 2016). Black women have been reported with more frequent and severe skin changes associated with pain and psychological distress resulting in disruption of daily functions (Schnur, et. al., 2011; Ryan, et. al., 2007). In younger women, incision related asymmetry, long term hypopigmentation and fibrosis resulting from implanted



brachytherapy have been reported influencing women's psychological state (Walijee, et. al., 2008).

Axillary Lymph node dissection (ALND) has been a standard procedure for staging to measure the extent of disease which often results in surgical morbidity including sensory changes, infection risk and other serious adverse effects including pain, functional limitations particularly lymphedema which demand lifetime adjustments to avoid infection in addition to psychological distress, body image concerns and alteration of clothing which significantly impacts all domains of quality of life (Hulett & Armer, 2016; Paskett, et al., 2012).

Women who consider mastectomy as a treatment choice are often offered reconstruction, of which implant reconstruction is the most common (30%) reconstructive choice globally (Dellacroce & Wolfe, 2012). Reconstruction may require use of tissue expander for better aesthetic outcome; however, the late effects of expander usage may result in discomfort or pain with a need of re-surgery for the placement of permanent implant (Adkinson, et. al., 2015). Moreover, the implant reconstruction may fail and could lead to extrusion or capsular extraction (Holland, et. al., 2015). However, these events are uncommon with 40% risks reported in a review of published literature (Rabey, et. al., 2013). Higher acceptance rate of reconstructions has been reported in white, younger and higher socioeconomic status; is further supported by educational level, financial issues and the decision-making process (Nelson, et. al., 2012).

Standard adjuvant chemotherapy commonly takes about four to six months and is associated with symptoms including nausea, fatigue, peripheral neuropathy, alopecia, menopause, weight gain, joint pain, skin changes and insomnia and others (American Cancer Society, 2019c; Nicholas, 2016; Holland, et. al., 2015). Endocrine therapy is prescribed for pre-and post-menopausal women having estrogenic dependent tumours which later results in increasing endometrial cancer and thromboembolic risks (Schiavon & Smith, 2014). Aromatase inhibitors (AI) are recommended for postmenopausal women. Both these drugs have adverse symptoms such as hot flushes, night sweats, sleep alterations, vaginal discharge or dryness and dyspareunia (Warren, et. al., 2000). Sometimes patients need to discontinue AI therapy as it may result in moderate to severe musculoskeletal symptoms (Henry, et. al., 2012). Approximately up to 30% of patients receiving Herceptin for having Her-2/neu overexpression



of tumours. In the UAE, younger women are more likely to have breast cancer at an earlier age with a higher incidence of HER2 positive and ER/PR negative tumours, which are known to be associated with worse disease outcome (Bashir, et. al., 2012). Although there have been significant improved survival rates noted with adjuvant Herceptin, cardiac toxicity is the major concern associated with it (Dang, et. al., 2016; Onitilo, et. al., 2014).

Health challenges such as pain, fatigue, functional disability, sexual dysfunction, cognitive changes, fear of relapse, depression, anxiety or lack of social support can be a result of or exacerbate cancer treatment (Wei, et. al., 2016). It is important to address concerns to avoid patients and their families experiencing negative effects and to improve the delivery of healthcare. Psychological distress along with physical symptoms have been rated as moderate to severe in women more commonly resulting in difficulty in coping in those who did not feel prepared or had inadequate information (Knobf, 2008). Social isolation or mental illness must also be treated to avoid additional emotional distress and to improve patients' ability to adhere to their treatment (Grassi, et. al., 2017; Weinstein, et. al., 2016). Healthcare systems and providers need to consider how to integrate psychosocial factors within the management of care (Mukherjee, et. al., 2018; Grassi, et. al., 2017; Recklitis, et. al., 2017). Psychosocial health services can enable patients with cancer, their families and healthcare providers to manage the disease more effectively resulting in improved quality of life (Advisory Board., 2018; Wei, et. al., 2016; Paraskevi, 2012; Galway, et. al., 2012).

2.6.3 SURVIVORSHIP CONCERNS AFTER TREATMENT

Literature suggests that the majority of breast cancer survivors cope well with the disease (Rah, et. al., 2019; Committee on Appropriations, U. S, 2002). However, negative consequences of breast cancer may persist for many years even after the acute phase (Bodai & Tuso, 2015). The end of treatment and the beginning of survivorship are often observed with diminished support from providers and health systems (Tompkins, et. al., 2016; Knobf, 2013; Cappiello, et. al., 2007). This phase continues with emotional highs and lows and vulnerability of being "new" normal or back to normal (Holland, et. al., 2015). How a woman adjusts into survivorship phase depends on the degree of symptoms she experiences, the adequate level of information she acquires to manage disease and late effects, psychological therapy focused on empowerment and increased self-care and coping skills (van Londen, et. al., 2013).



Fear of recurrence, uncertainly of reintegration into normal life or fear of mortality with other physical symptoms negatively affects patient's wellbeing (van de Wal, et. al., 2018; IOM, 2008). Younger age limited social support, poor communication with the providers and preexisting psychosocial problems have been identified factors that could challenge women's recovery (Holland, et. al., 2015; Fillon, 2013; Zhou, et. al., 2010). On the other hand, Andersen, et. al., (1989) and Ganz, et. al., (2002) argued that several studies have reported few differences in long-term physical and emotional wellbeing when compared breast cancer survivors to healthy, age matched populations. A study of long-term adjustment of women 20 years after treatment reported 18% women with post-traumatic stress, 27% with lymphedema and 20% with numbness as persistent problems (Kornblith, et. al., 2003).

Continuous support from healthcare providers is essential to help cancer survivors cope with the cancer transitions that occur during the course of illness (IOM, 2006; Maguire, 1999). As cancer treatment ends, many survivors face abandonment from the care providers and experience the need to adjust to temporary or permanent physical and emotional changes (IOM, 2006). Cancer survivors need ongoing support throughout their lives to ensure they have appropriate medical care, including screening for new primaries and recurrences, access to services they need particularly emotional concerns and opportunities to engage them as active participants in cancer decision making (Philip & Merluzzi, 2016; Ganz, 2007).

For many survivors, coordination of their care is often observed as sub-optimal (Miller, 2012). There is a clear need for health systems to ensure a seamless patient experience throughout the cancer journey (Chadder, et. al., 2018). Treatment and summary care plans (Appendix 9) should be provided by healthcare providers to ensure appropriate coordination among the members of the extended care team and to support effective communication of a survivor's health status and long-term care needs. Survivorship programs can help them navigate the complexities of care and to bridge the care coordination as for the survivors, coordination of their care is often observed as sub-optimal (Miller, 2012).

Unfortunately, there is no survivorship program in the UAE cancer care system which can help cancer survivors navigate the complexities of care and to bridge the care coordination. Moreover, lack of evidence and literature on cancer survivorship in the UAE pose a great challenge to understand the issues associated with the local community.



2.7 KEY DOMAINS OF BREAST CANCER SURVIVORS' CONCERNS

Needs of human beings are not limited to physical aspects as there are various domains of needs, for instance, emotional, psychological, social and spiritual. Human beings take actions throughout the day to ensure these needs are met (Maslow, 1998). Dudas & Carlson (1998) argued that the usual way of meeting daily needs becomes ineffective as life events change; thus, a person's needs cannot be met in the same manner applying the routine approaches. The new situation may require new demands with the feeling of vulnerability leading to anxiety, uncertainty and a sense of loss of control over the situation. The individual may require learning new skills, new information or seek additional help to fulfil his needs. These circumstances could be challenging and add an additional burden during the time of distress (Ganz & Stanton, 2015; Vivar, Whyte, & McQueen, 2010; Taylor, 1983). A person may continue to feel vulnerable and feel emotional distress if these needs remain unmet (Fitch, 2008).

Cancer diagnosis is a universally stressful experience (Payne, 2014) which is influenced by a variety of factors; thus, needs emerge and change accordingly and overtime. The diagnosis and treatment of cancer can be conceptualized as an unexpected life event or series of events across a spectrum of experience with a disease (Fitch, 2008)

Throughout the illness, practical changes impact patient's needs which do vary from person to person (Fitch, 2000). Cancer and its treatment have an impact on individuals that can be felt in many ways. In addition to physical changes, cancer may evoke emotional, social, financial, information and spiritual changes (Figure 7). Following are the five main domains in which the key concerns of breast cancer survivors are discussed in greater detail:

- 1. Physical Concerns
- 2. Information or Educational Concerns
- 3. Social or Financial Concerns
- 4. Emotional Concerns
- 5. Spiritual or Religious Concerns





Figure 7: Key domains of Breast Cancer Survivorship

2.7.1 PHYSICAL CONCERNS OF BREAST CANCER SURVIVORS

Breast cancer diagnosis is critical in its physical implications and may turn into a crisis for the patient at an emotional level (Kirby, et. al., 2017; Ray & Baum, 1985). Physical needs encompass but are not limited to needs for physical discomfort and freedom from pain,



optimum nutrition, ability to carry out one's usual day to day functions and activities of daily living (Fitch, 2008; Hewitt, et. al., 2003). Physical discomfort or disability may impede an individual's capability to fulfil his or her needs (Zabora, et. al, 1997). Physical impairment such as fatigue, pain, post-surgical and persistent upper-quadrant issues, chemotherapy-induced peripheral neuropathy, lymphedema, cardiotoxic effects of chemotherapy and radiotherapy, weight gain, bone health challenges, and arthralgias are the most common concerns that survivors experience during or after the course of treatment which leads to limitation of one's daily activities (Schmitz, et al., 2012).

The IOM (2008) reported that people with cancer or with a history of cancer more frequently have poor health (30%), other chronic medical conditions (42%), one or more limitations in the ability to perform activities of daily living (11%) and other functional disabilities (58%). The National Health Interview Survey (NHIS) data from 1998, 1999 and 2000 indicated that a medical history of cancer at least doubles an individual's likelihood of poor health and disability. Hewitt, et. al., (2003) reported that the individuals with a history of cancer have significantly higher rates of other chronic illnesses resulting in 5 to 10% higher disability rate than expected.

Fatigue is the most frequently reported symptom of breast cancer (Bower, 2014; Narayanan & Koshy, 2009; Bardwell, et. al., 2008). which has been reported as major obstacle in maintaining their routine daily activities and has serious impact on quality of life (Rao & Cohen, 2004; Portenoy, & Itri, 1999). Cancer fatigue is generally not alleviated by sleep or rest, is typically of greater duration and severity, is often associated with high levels of distress, and is disproportionate to the level of exertion (Bower, 2014; Bardwell & Ancoli-Israel, 2008). In the literature, the proportion of patients experiencing cancer-related fatigue varies widely between 40% and 100% of the overall number of patients with cancer either due to lack of diagnostic criteria and assessment instruments and the effect of disease stage and status on fatigue (Savina & Zaydiner, 2019). A 2002 review by the Agency for Healthcare Research and Quality (AHRQ) found that mechanisms of cancer-related fatigue have been poorly explored, and current treatment options for fatigue are also limited (Carr, et. al., 2002). Ness, et. al., (2013) reported that the highest mean scores overall (rating 0–5) within the physical domain were fatigue, sleep disturbance, memory and concentration, and loss of strength. However, when excluding those who rated a score of zero, and only reporting on items rated 1–5, the most



prevalent items were sexual issues, peripheral neuropathy (numbness and tingling), hair and skin care, and fertility.

Pain resulting from the cancer and its treatment is experienced by one third to one-half of the patients (66%) and is often moderate to severe intensity in multiple sites (Caraceni & Shkodra, 2019; IOM, 2008). There is substantial evidence that the management of cancer pain is still often suboptimal (IOM, 2008; Kwon, 2014; Greco, et. al., 2014) and often undertreated, thus, remains despite the administration of analgesics and other therapies and continues to be a concern even in the absence of signs of cancer (IOM, 2008).

Disability, fatigue and pain can result even when there are no longer any signs of the disease which results in added emotional distress and leads to social problems such as inability to work (IOM, 2008).

Cognitive impairment has also been reported in some cancer survivors who were treated with chemotherapy for instance, subtle decline in cognitive functioning mainly language skills, short-term memory and spatial abilities (Lange & Joly, 2017; Adler & Page, 2008). There is a likelihood of infertility in younger women in addition to acute toxicities or radiation and chemotherapy treatments. The symptomatic burden of premature menopause can also result in mood disorders and complaints of cognitive impairment (Warga, 2000).

Anxiety and depression may be triggered by the events of cancer while at the same time lumpectomy or mastectomy and/or other treatment could lead to additional stress (Elobaid, 2014; O'Connor, et. al., 2011; Golden-Kreutz & Andersen, 2004). One study reported that 42% of women found the most stressful part of the cancer experience was the period immediately after finding the lump (Ray & Baum, 1985).

Parizadeh & Abbadi, (2012) defined Body Image as the attitudes and perceptions of individuals towards their appearances and the belief with respect to their body. Among Arab women, great cultural value is placed on body image; thus, anxiety about cancer detection and fear of breast loss consequently led them to resist seeking help. Yam (2013) and Anstett (2016) noted that body image is highly associated with cultural perceptions. Many Arab women tend not to have a mastectomy and would wish to keep their breast as they fear that being diagnosed with the disease and breast removed will result in negative consequences with family relationships



(Donnelly, et. al., 2017). Rezaei, et. al., (2016) and Boing, et. al., (2019) found that women with a low level of education suffer more from discomfort over their body image.

Sexual issues due to treatment could have destructive effects on women's relationship with their spouses or partners (Henson, 2012). Removal of the breast or other treatments could lead to sexual problems, for instance, dryness of vagina which could result in sexual dysfunction and adverse effects on women's sexuality leading to additional psychological trauma (Bakht & Najafi, 2010). After mastectomy, femininity of women may be considered lessened after losing an important sexual organ (Cordero, et al., 2015). Since the breast is a symbol of femininity and sexuality, any damage to it could lead to distress and could harm a women's whole life (Rezaei, et al., 2016). These physical changes triggered psychological trauma resulting in physical and emotional stress eventually undermine one's self-confidence (Almeida, et. al., 2012; Cohen, et. al., 2012). Yeung (2009) and Mckee & Schover (2001) concluded that most patients receive little or no assistance in dealing with the effects of cancer treatment may have on sexual intimacy. After cancer treatment, many patients continue to have unmet needs for information about restoring sexual function (Schover, et. al., 2014). Sexual dysfunction is not something that is generally identified by the health **providers**, and neither is sexual counselling routinely provided (Medndoza, et. al., 2017; Shankar, et. al., 2017 Vaziri, & Lotfi, 2012). Health systems should define and standardise the screening procedures to identify individual perceived needs. For instance, questions about sexual difficulties and intimacy have been recommended as part of the initial evaluation of patients with cancer (Avasthi & Sathyanarayana, 2017; IOM, 2008; McKee & Schover, 2001). In addition, sexual counselling has been recommended both early and continuing through treatment and recovery for those who need it (Barbera, et. al., 2017; Higano, et. al., 2016; Dizon, et. al., 2014).

Hospitalisation and surgery could be petrifying events for a breast cancer patient. Gruendemann (1965) has described surgery as a planned physical assault on the body. This observation was further supported by (Everly, 1989; Clarke, 1984; Masterson, 1979) stating that surgery, regardless of the specific nature of the operation, is purported to be a stressful experience. Cancer treatments including surgery can have direct physical effects on the patient and often lead to substantial permanent impairment of several organ systems with resultant disability (Aziz & Rowland, 2003). A Tunisian study showed a significant deterioration in physical, cognitive and social functioning among patients who underwent treatment, while an



Egyptian study reported marked change in physical and social functioning both pre-and post-treatment (Haddou, et. al., 2016).

Physical recovery after breast cancer surgery may be impaired in women with greater comorbidity (Lash and Silliman, 2000), which may contribute to greater psychological distress. Arab women are known to experience a greater impact of treatment type on their quality of life, particularly if treatment had a multimodality approach (El-Sharkawi, et al., 1997). The fragmented approach of treatment delivery for instance surgery, radiation, chemotherapy and attending survivorship clinics in different entities or different healthcare providers can lead to patients getting "lost" in the system, thereby failing to access appropriate services and adding additional barriers to a coherent approach to cancer care (Walsh, et. al., 2010; IOM and National Research Council, 2006).

A tailored treatment that meets cancer patients' medical and psychosocial needs may result in improved pain management and functional ability (Fischer, et, al., 2010). Physical activity for cancer survivors appears to be associated with an improved quality of life (Acton, 2011). This finding was strengthened by another study which concluded that regular physical activity shows promise as an aid to alleviating poor sleep and an increase in fatigue which can be consequences of having cancer (Humpel & Iverson, 2010). In order to alleviate physical symptoms, survivors must seek expert advice provided by the cancer care team. Pikler & Winterowd (2003) described cognitive coping strategies, social support and adaptations as the most effective interventions to reduce the impact of disease-related issues with women's body image.

It is imperative for health providers to work with patients to develop a personalised follow-up care plan (Daudt, et. al., 2014). The establishment of a survivorship program as part of cancer programs is considered a mandatory requirement by the Commission on Cancer, standard 3.3 (Stricker & O'Brien, 2014). Additionally, targeted exercise programmes for breast cancer survivors aid to prevent stiffness of body parts and reduce the chances of developing lymphedema (American Cancer Society, 2020). On the other hand, non-targeted exercises enhance patients' psychological and physical wellbeing; for example, yoga or weightlifting etc. which have been found to be effective in reducing anxiety, depression and fatigue in cancer survivors (Ng, 2017).



2.7.2 INFORMATION OR EDUCATIONAL CONCERNS OF BREAST CANCER SURVIVORS

Information needs have consistently been addressed as the top unmet need of cancer survivors in the literature (Wang, et. al., 2018; Chiesi, et. al., 2017; Fong & Cheah, 2016; Fiszer, et. al., 2014; Campbell, et. al., 2009; Harrison, et al., 2009). The benefits of information provision to cancer patients include but are not limited to an improved coping ability, reduction in anxiety and patient readiness to cope up with future events of illness. Several studies have explored the information and educational needs of breast cancer patients and modes to deliver such information (Findik, 2017; Shea–Budgell, et. al., 2014; Li, et. al., 2011; Wallberg, et. al., 2000).

Literature suggests that breast cancer patient's information needs are higher and directly associated with longer survival and well-being (Kimiafar, et. al., 2016; Tsuchiya & Horn, 2009). Treatment related information such as available treatments options have been the most frequent type of information reported in the literature followed by information seeking about the cancer type (Okuhara, et. al., 2018; Chua, et. al., 2018; Rechis, et. al., 2014; Squiers, et. al., 2005; Rutten, et. al., 2005; Bader & Theofanos, 2003).

In Eastern cultures, families act as bearers of bad news and strongly demand a "do not tell" approach (Zakri & Karim, 2015; Salem & Salem, 2013). Patients reported feeling upset after hearing the devastating news of cancer diagnosis and such anxiety may result in making it more difficult to provide further information (Konstantis & Exiara, 2015; Stoll, 1986). The delivery of bad news continues to be stressful for health providers and a traumatic experience for cancer patients and their families (Günüşen, et. al., 2018; Stiefel, 2006; Fallowfield & Jenkins, 2004).

Although no communication intervention can eliminate the impact of cancer diagnosis, yet the effective delivery of bad news can at least help patients become better informed, to increase patients' trust, to improve the quality of communication and to be able to better navigate the healthcare system (Sobczak, et. al., 2018, Mostafavian & Shaye; 2018; Back, et. al., 2005). Bellamy (2004) argued that patient education must be a core component of medical school curriculum. A large body of literature has recommended courses for students and residents for delivering such sensitive news, yet clinicians often deliver diagnostic information without proper training or support (Azadi, et. al., 2018; Kissane, et. al., 2017; Dresser, 2012; Eggly, et al., 1997). It is even more complex task for physicians working in Muslim countries who need



to require culture-specific training in order to break bad news; they are convinced of wholeheartedly and truthfully informing their patients about a bad diagnosis or a disastrous morbidity (Salem & Salem, 2013).

A cross-sectional study in Qatar, reported that a majority of physicians would disclose a cancer diagnosis, yet 66% would make exceptions in rare cases (Arnott-Smith & Keselman, 2015). Only Arab or Muslim physicians, who are studied in the region, reported that their usual policy is not to tell however it is also believed that their practice might change with the result of empirical research (Rodriguez, et. al., 2012). It is not uncommon in other parts of the world to deal with the disclosure of cancer news with patients as the most difficult scenario in a clinical setting as some patients may show denial and may experience great fear of death and disabilities while others may experience various levels of anxiety that leads to depression (Coulter, et. al., 2008; Eapen & Revesz, 2003). Many physicians too have reported having felt overwhelmed and traumatised by their early experience at delivering the bad news (Mostafavian & Shaye; 2018; Sobczak, et. al., 2018; Salem & Salem, 2013; Orlander, et. al., 2002).

Effective communication skills are the key to good cancer care (IOM, 2013; Stiefel, 2006). Inadequate communication may cause much distress for patients and their families, who often require considerably more information that is usually provided (Bush, et. al., 2018; Silistraru, 2018; Fallowfield & Jenkins, 1999). Not surprisingly, many cancer patients have been reported dissatisfied with much of the communication that takes place within hospitals (Avestan, et. al; 2019; Tu, et. al., 2019; Mahapatra, et. al., 2016; Zamanzadeh, et. al., 2014; Epstein & Street, 2007). Moreover, satisfaction rates associated with the information provided by healthcare providers have also been reported as relatively low in South Asian countries including UAE (Elbarazi, 2015; Au, et. al., 2013; Liao, et. al., 2012) which necessitates the access to accurate and timely information as well as educational needs.

Effective patient-clinician communication and shared decision making are key components of patient-centred care (Delaney, 2018; Newell & Jordan, 2015; IOM, 2013, King & Hoppe, 2013). In patient-centred care, the patient, clinician and healthcare system dynamically interact to influence patient centred care which leads to improved communication and health outcomes (Jardien-Babooa, et. al., 2016; IOM, 2013). Literature suggested that patient-centred communication should be sensitive to patients' vulnerability, degree of education and ability



to understand medical terms, concepts, risks, and benefits, possible reluctance to assert preferences and ask questions, preferences related to treatment and cultural and spiritual outlook (IOM, 2011). Oskay-Ozcelik, et. al., (2007), Kash, et. al., (2005) and Shilling, et. al., (2003) suggested that the patient centredness of cancer care can be improved in several ways, such as more supportive care, better communication, better information supply and better cooperation amongst physicians. Institute of medicine (2008) illustrated a model of psychosocial care delivery emphasizing on effective communication between patients and providers with shared decision making as below:





Figure 8: Effective patient – provider communication & shared decision making

IOM, (2011). Patient centred cancer treatment planning: Improving the quality of oncology care, Pg 20



Patient centred care includes good communication between patient and healthcare providers, delivering the evidence-based information to inform patients and practicing shared decision making (Nekhlyudov, et. al., 2014). The main dimensions of patient centred care are to recognize each patient as a unique person, to respect the patient's values and beliefs, and to respond to the patient's individual needs and preferences (Delaney, 2018; Entwistle & Watt, 2013; Pelzang, 2010; IOM, 2001). When it comes to treatment goals, shared decision-making and patient's preferences are particularly important, such as opting for less aggressive treatment options to maintain high quality of life (Glatzer, et. al., 2018; IOM 2013; Fraenkel, 2013; Berman, 2012; IOM, 2011)

Noguera, et. al., (2014) observed that most patients preferred a shared or active decision making and wanted accurate information about their diagnosis and preferences. On the contrary, a Greek study reported 71 percent of women's preference to play a passive role in the treatment decision making (Almyroudi, et. al., 2011). Another study has also shown that supporting patients to be active participants in their health journey from diagnosis to long-term survival can offer patients an increased level of autonomy in their health and may enhance overall quality of life (Jordan, et. al., 2008; Arora, et. al., 2002; Luker, et. al., 1995).

Post, et. al., (2001) argued that ethnic differences play a huge part in the communication process between patients and health providers. Not all patients are eager to learn what health providers want and some are not able to understand the complex teaching (Freda, 2004). A Saudi study reported various socio-cultural factors that play a role in deciding how much information should be given to cancer patients. For instance, families are key participants in medical related decision-making (Al-Amri, 2009). Obeidat & Khrais (2015) emphasized healthcare providers to assess patients' preferences about individuals' particular needs and disclosure of cancer diagnosis and related information. Elbarazi (2015) reported the higher needs to improve patient-physician communication taking into consideration the cultural and language barriers.

It is equally important to understand that information needs of cancer patients do not remain constant and may change over the course of illness (Hsieh, et. al., 2018; Al-Qadire, 2014; Ankem, 2005; Hewitt, et. al., 2004). It is important to understand what type, amount and quality of information delivery is necessary to achieve better health outcomes (Rood, et. al., 2018;



Degner, 1997). Rutten, et. al; (2005) suggested that the stage of cancer determines the information needs of the patients.

A study found the needs of information declined from initial consultation to follow-up nevertheless remained high whereas increased or remained stable by the time of follow-up (Douma, et. al., 2012). Information needs related to rehabilitation, recurrence and access to available support services have been reported higher during the follow-up phases of cancer care (Lewis, et. al., 2009; Squiers, et. al., 2005). Van Mossel, et. al (2014, 2012) and Matsuyama, et. al. (2013) affirm the need to identify the information needs at each stage of the cancer in order to accommodate specific unmet needs at each phase of cancer journey.

Additionally, patients may perceive information differently from one another as an individual's personality influences their perception of the information (Cowan & Hoskins, 2007). As information needs vary based on severity of disease and type of treatment (Shea–Budgell, et al., 2014; Meredith, et. al., 1996), researchers have also identified differences in information needs based on patient demographics and ethnicity (Al-Amri, 2013; Matsuyama, et. al., 2011 & Rutten, et al 2005). Ankem (2006) reported factors such as age, gender and education level influencing the information needs. The study found that younger patients had more needs than older patients and preferred a more active role in decision making. Wallberg, et al., (2000) and Papadakos, et. al., (2014) also found higher information needs in younger patients and in those with a higher education background.

A Jordanian study also reported that younger and higher education patients are more likely to prefer the truthful disclosure of breast cancer diagnosis (Obeidat & Khrasi, 2015). On the other hand, Recio-Saucedo, et. al., (2016) found the process of decision making in younger women quite challenging thus recommended the efforts to provide information regarding sexuality, body image, reconstruction, fertility and likelihood of familial predisposition. Meredith, et. al., (1996) noted that affluent patients required more information than those from deprived areas. Other demographics for instance, gender and religion were also found to be associated with the need for information.

Cultural and religious beliefs might be such factors impeding the information seeking behaviour mainly in Arabs (El-Obaid, et. al., 2014; Ortashi, et. al., 2012; Hatefnia, et. al., 2010; Al-Amri, 2009; Bener, et. al., 2002). Donnelly, et. al., (2013) found that women in Qatar, who



believed that cancer is God's punishment or bad luck, were significantly less likely to practice any kind of breast cancer program compared to participants who wanted to know if they have cancer.

Information and educational needs have become the most important part of cancer care to inform patients about their treatment plans and effectively cope up with the disease (Chua, et. al., 2018; IOM, 2011). Berger, et. al., (2018) addressed that well informed patients are more satisfied with care, have a better sense of control of their total situation, and report a better quality of life. Frenkel & Cohen, (2014) further illustrated that effective communication is the key to improved health outcomes. What information and when patients need to know and from whom has become the pivotal to ensure the delivery of patient-centred care (NIH, 2017; Rutten, et. al., 2005).

Most patient educational programs in the UAE are unstructured and fragmented (El Obaid, 2014; Aw, et al., 2011;). Availability to health educators is rare and the majority of the health education is delivered within the community (Sharif & Blair, 2011). Educational programmes should start with proper needs assessment and goal setting to meet those unique needs of the targeted population (Johnson, 2014; Ratzan, 2009). Fallowfield & Jenkins (1999) explained that multidisciplinary and comprehensive education programs are part of the process of patient empowerment (Chen, et. al., 2016). Less education or lack of literacy would delay seeking medical support in the presence of symptoms, thus, likely to be delayed for screening, resulting in progress or advanced stage at the time of diagnosis. (Al-Naggar, et. al., 2012). Several studies have reported a significant association of poor quality of life with lower levels of education and awareness and lower socio-economic status (Musarezaie, et. al., 2012; Sprague, et. al., 2011; Salonen, et al., 2009; Esbensen, et al., 2004). Thus, raising awareness and empowering women with knowledge and information could heighten their ability to avoid associated risk, understand the nature of the disease and to comply with the treatment regime that is most suitable for her.

Cancer patients should receive accurate information throughout their cancer journey to alleviate their fears and to cope with illness (Lawler, et. al., 2016; Rutten, et. al., 2005; Girgis, et. al., 2000; Leydon, et. al., 2000). Tailoring information as per patients' needs result in empowering patients to be part of decision-making and can ensure that the appropriate type and amount of information is received (Shea-Budgell, et. al., 2014). It could also guide



healthcare providers to deliver the most appropriate care through which patients could greatly benefit. According to Garza, et. al., (2005) and Fitzpatrick (2016), culturally designed interventions result in increased information access, reduce barriers, improve screening and survivorship.

Viewed optimistically, increased partnership and diminished power differentials between patients and their providers have resulted in optimal outcomes, enhanced patient autonomy and meeting unique patient's needs more than ever before (Chen, et. al., 2016). Empowered and informed patients have transformed the face of the current healthcare system (Lenert, 2010). Patients must be encouraged to develop a bond with their health providers. Cancer patients should be provided information about the available support services, helpline, books, trusted websites, secure patient information portals or material. Healthcare providers should enable their patients to work towards shared goals through optimal education processes (Paterick, et. al., 2017), such as by improvising their engagement in the decision-making process, by encouraging those to take ownership of their disease and by providing information that helps them integrate the illness into daily life and to navigate through the healthcare system.

Kissane, et. al., (2017) argued that it is imperative to make patients feel less tense and every attempt should be made to reduce their fear while providing the information they seek or need. Physicians should disclose cancer diagnosis in a personal setting and discuss the diagnosis and treatment options as well as adverse effects of each treatment with patients in a sensitive manner considering cultural variations among cancer patients (Figg, et. al., 2010). Additionally, appropriate training programs for health professionals involved in cancer management will help them to deliver adequate information to patients through professional-patient communication.


2.7.3 SOCIAL AND FINANCIAL CONCERNS OF BREAST CANCER SURVIVORS

The concept of social support in the literature is multidimensional and provides several definitions which also are vague and circular (Hardan-Khalil & Mayo; 2015; Rodriguez & Cohen, 1998). Marmot (2005) argued that social support is influenced by social structural constraints and mainly addressed the individual links of networks in terms of social cohesion. In the literature, the term is generally referred to as the mechanisms by which interpersonal relationships support and protect people from the negative consequences of stress (Pietromonaco & Collins, 2017; Bertero, 2000). Soylar & Genc (2016) defined social support as the help provided for the individual who is under stress or in a difficult situation by people around the individual. Taylor (2011) defines social support as the experience that one is part of a mutually supportive social network which has beneficial effects on mental and physical health. Despite the various interpretations of social support, it has been claimed to be associated with improved survival and have positive outcomes on physical and mental wellbeing (Gallant, 2013; Whitford, et. al., 2008; Astrow, et. al., 2007; Wortman, 1984). There is a large body of evidence indicating that social support affects physical and emotional health positively by meeting fundamental social needs such as love, compassion and belonging to a group, and is a significant help for the person in coping with difficulties in life (Pietromonaco & Collins; 2017; Soylar & Genc, 2016; Cohen & Wills, 1985).

The transition from the various stages in the cancer trajectory is a challenge to women with breast cancer. As needs of survivors shift from disease management to recovery, the adjustment is often not seamless and can be overwhelming, with many challenges including psychological and social well-being (Alexander, et. al., 2019; Chadder, et. al., 2018). Rosedale (2009) observed that isolation and loneliness are common among women who have experienced breast cancer. Wei, et. al., (2016) remarked that breast cancer and its treatment alter social roles and limit the social or daily activities of patients. According to IOM (2008), the physical and psychological problems that emerge from cancer can be exacerbated by or produce significant new social problems.

Social support has long been recognized as an important component of physical and mental health (Snyder & Pearse, 2010). Older women are more likely to face loss of a spouse or reduction in economic status (Carr & Bodnar-Deren, 2009). Many women with breast cancer



managed to deal with their illnesses and treatments based largely on social support they received; however, patients are often unaware of resources available to help them overcome psychosocial problems (Bayliss, et. al., 2003). And even if they are informed of available resources, problems with mobility, fatigue, poor understanding of disease, depression, lack of support, negative emotions or financial problems etc. prevent them from taking advantage of these resources (Jerant, et al., 2005). These problems are more apparent in particularly vulnerable and disadvantaged populations such as those with lower socioeconomic background or lower levels of literacy. When these resources are not available, the ability to manage cancer and its outcome is also decreased (IOM, 2008).

The importance of support from support groups and various other sources is well recognised (Adorna, et. al., 2015; Morris, et. al., 2014; Matthews, et. al., 2004; Leadbeater, 2004; Till, 2003). The social support system could be evolved at three levels: the primary level which includes close family and friends, the secondary level which includes other friends, relatives, neighbours and work colleagues and the third level which comprises groups and organisation (Peltonen, 1996; seeFigure 9). Peltonen explained that the primary level is most suitable for giving emotional and practical support when needed as lack of support or support which is not timely could lead to negative consequences. Dedeli, et. al., (2008) also stated that a large part of social support consisted of the patient's own family support. Peltonen (1996) associated the second and third levels of support to cultural, political and religious ideologies and other social values.





Figure 9: Three levels of social support system (Peltonen, 1996)

Family has the highest impact compared with significant other support with social support resources (Vrontaras, 2018; Snyder & Pearse, 2010; Bertero, 2000). The person receives support from their family or significant people in their life when they feel their abilities are inadequate; however, when such support is lacking, the psychosocial symptoms and issues are aggravated. On the other hand, social networks and expanding social networks post diagnosis are critically helpful resources for any cancer survivors since cancer experience sometimes may become an ordeal and could pose a challenging situation when family and friends are unavailable or unsupportive (Ginter & Braun, 2017; Eakin & Strycker, 2001). Social support becomes vitally important for patients who are lacking informal support that one desires from the closed circle or relationships.

It has been observed that Arab women with breast cancer suffer in silence over many things; for instance, due to stigma and prejudice, women with breast cancer often prefer to stay home as they do not want their families to suffer with guilt and shame just because one has breast cancer (Islam, et. al., 2017; Elobaid, et. al., 2016; Baider & Goldzweig, 2016; Hwang, et. al., 2015; Surbone, et. al., 2013). Women are hugely under pressure and do not ruin their family



image by disclosing that breast cancer runs in the family. Often, they are worried if their husband would still want to produce offspring with women with breast cancer or whether they are still able to produce children. Cancer is not something that one can openly talk about in Arab culture and is a topic surrounded by shame and silence (Elobaid, 2014). Moreover, in the Arab world, the role of women is often defined by marriage and children and fulfils multiple responsibilities for their families and society. Breast cancer diagnosis exposes women to disfigure this image and role that fulfils multiple responsibilities for their families and society. Women do not go and seek any social or medical help unless they are in extreme pain (Elobaid, 2014).

Isolation from society is another cultural myth linked with breast cancer as people do not want to visit a woman with breast cancer as they are afraid of the evil eye or bad luck which is an Arabic interpretation of envy and jealousy from a sick person towards a healthy one (Elobaid, at. al., 2016; Donnelly, et. al., 2013; Soskolne, 2007). In Arabs, women with breast cancer are commonly being labelled as guilty of having cancer due to the mistakes and sins that she carried out during her life. The lack of knowledge about the disease keeps them unaware of disease consequences and seeks any support when they have relevant signs and symptoms of breast cancer. Religious and cultural restriction of seeking medical help from male physicians to examine or treat her adds additional obstacle to health seeking behaviour for Muslim women (Donnelly, et. al., 2017; Elobaid, 2014). Shirazi, et al. (2013) argued that many Muslim women are socially or linguistically isolated thus support groups from their own culture or society may be best resource to accommodate their specific needs. The concepts of evil eye and cancer linked with sins may be a cultural dimension and not necessarily an Islamic dimension. According to Islam, a person should not lose hope and make every effort to seek possible treatment yet being thankful to Allah. The Islamic perspective of the disease has been discussed in detail in section 2.7.5 on Spiritual concerns.

In addition to social support needs, economic burden has also been found to have a significant impact on breast cancer survivorship (Landier, et. al., 2020; Lorgelly & Neri, 2018; Hrishikesh, et. al., 2016; Meneses, et. al., 2011). Financial stability is an integral part of good quality cancer care. Financial stress from low income, the cost of healthcare or lack of health insurance or reduced employment or income can result in substantial stress and can hinder receiving the appropriate care. The adverse effects of no or inadequate health insurance are well documented



and include poor health prior to receipt of care, delayed or no treatment, failure to get needed prescribed medications and worse outcomes of medical treatment (IOM & NRC, 2006). A study found that financial needs were most likely to be unmet and that for some of those experiencing financial hardship, 'this aspect of living with cancer was almost worse than the disease itself' (Soothill, et. al., 2001).

Over 75% of the population in the UAE are expatriates and their insurance is covered generally by their employers. Problems with health insurance are affected by problems with employment. If an individual loses their job, they also run the risk of losing health insurance coverage and income. Not having insurance is a catastrophe for an expatriate who is diagnosed with cancer in the UAE as the treatment is quite expensive. Cancer as a chronic disease is not usually covered by the basic insurance which is provided by the employers. The struggle for cancer patients is not just financial but once they are ready to return to work, they face two challenges — landing a job and coping with the chronic symptoms of the disease and some simply lose their jobs or resign from work. Bradley, et. al., (2002) observed that breast cancer may impose an economic hardship because it affects women's ability to work.

Expatriates who do not have the financial means to support their treatment in the UAE also do not have the financial means to receive it in their own country. Most of them do not even disclose to their families that they're sick. It becomes both a financial and psychological struggle. Furthermore, the prohibitive costs of cancer treatment demonstrate that many people do not end up seeking treatment and may terminate the care halfway. There are no current health policies in the UAE that consider coverage of cancer for expatriates thus cancer patients need to make a hard choice either to leave the country or suffer the illness if they remain in the country. Nevertheless, there are authorised charities, non-profit organisations and support groups in the country that may help cancer patients to provide financial assistance for instance, Pink Caravan, Breast Cancer Arabia, Friends of Cancer Patients, Emirates Cancer Foundation, Al Jalila Foundation and Rahma and UAE Government mandates. The majority of these groups provide psychosocial counselling, informational and emotional support and minimum financial help to all UAE residents regardless of their nationality.

As part of patient-centred care, it is critical for health professionals to screen cancer patients for social and financial difficulties. For instance, for some patients, transportation has been



observed as a paramount concern by cancer care teams (Matthews, et. al., 2004). Policies must be in place to enhance systematic healthcare reform that promotes patient-centred cancer treatment and to have quality metrics and incentives to ensure that patient-centred care is delivered. In addition to promoting patient centred cancer care policies, there is a clear need for health policies to balance out disparities in cancer care in the UAE by sustainable and affordable health insurance coverage for all residents. There is also a need for employment related policies with regards to cancer to ensure job security and to eliminate any discrimination at work due to cancer.



2.7.4 EMOTIONAL CONCERNS OF BREAST CANCER SURVIVORS

Distress is a common concern across the cancer trajectory, beginning at diagnosis and extending to the post-treatment phase of cancer and long term (Howell & Oslen, 2011). Distress is defined as a multifactorial unpleasant emotional experience of a psychological (cognitive, behavioural, emotional), social, and/or spiritual nature that may interfere with the ability to cope effectively with cancer, its physical symptoms, and its treatment (Ng, et. al., 2017). According to the IOM and National Research Council (2006), cancer patients may experience a generalised feeling of worry, sadness, fear of future, indecisiveness, uncertainty, higher sense of vulnerability, social isolation, fear of reproductive ability and sexual functions and changes in the individual's role or relationship. The emotional stress of cancer, for instance, long term treatment, fear of recurrence or new cancer, anxiety, mood disturbance, concerns about body, distress due to daily physical concerns can lead to or exacerbate pre-existing distress for cancer survivors and their families (IOM, 2008). Studies have found higher levels of depression among cancer patients (Hegel, et. al., 2006; Spiegel & Giese-Davis, 2003). According to the Agency for Healthcare Research and Quality (AHRQ, 2002), the prevalence rate of having major depressive disorders among cancer patients is four times higher than in the general population.

A study of 215 cancer patients among whom half of the patients had breast cancer revealed that 47% had psychiatric disorders with symptoms of reactive anxiety, depression or mix of both (Derogatis et al., 1983). Although there is some evidence that distress is greater in cancers with poor prognosis, yet it is not fully known whether women with breast cancer have higher rates of distress than others (Zabora, et. al., 2001). Emotional distress and psychiatric syndromes have been consistently reported by the breast cancer population at large (Izci, et. al., 2016; Uchida, et. al., 2011; Hegel, et al., 2006); however, the literature reveals that sensitivity to patients' emotional needs seems to be deficient in current practice (Bramley & Matiti, 2014; Oguchi et. al., 2011; Sheldon, et. al., 2009; Adler & Page, 2008).

Younger breast cancer survivors seem to have higher emotional needs (Knox, et. al., 2017; Fiszer, et. al., 2014; Vivar & McQueen, 2005; Epping-Jordan, et. al., 1999) showed that young adults with advanced cancer became socially isolated because they felt misunderstood and alienated from the rest of the world. The experience could be even more stressful as some may



be thinking about planning a family or having more children. Young patients with breast cancer treated with chemotherapy can experience ovarian failure, which can lead to chemotherapy-induced menopause impacting the quality of life (Passildas, et. al., 2019). For some women, chemotherapy can cause early menopause which can be very distressing itself. Such factors contribute to greater psychological distress in younger women (Bloom & Kessler, 1994). Warga (2000) mentioned that amenorrhea and premature menopause due to hormonal changes which are associated with breast cancer can be linked to mood disorders and cognitive impairment in younger women. In addition, women with no husbands or intimate partners are more likely to suffer from emotional distress and dysphoria and have concerns about her femininity, attractiveness, reproduction and future relationship or partner (Bloom and Kessler, 1994).

On the other hand, older women may face a new illness in addition to other pre-existing problems such as being the eldest caregiver of the family. Any pre-existing condition could add additional distress and influence on women's psychosocial health. For instance, several studies have indicated that women with comorbid conditions or impaired performance status report higher levels of psychological distress after a breast cancer diagnosis (Sun, et. al., 2019; Ganz, et. al., 1993; Lansky, et. al., 1985). Greenfield, et al., (1987) suggested that there is a likelihood of greater comorbidity at diagnosis with an increase in age. Moreover, physical recovery after breast cancer surgery can be impaired in women with comorbidities (Lash & Silliman, 2000).

Although emotional response to treatment has been extensively studied, most patients receive little or no assistance in dealing with the effects of cancer treatment on sexual intimacy (Eunyoung & Hyunjin, 2018; Mckee & Schover, 2001). Young women diagnosed with breast cancer need more emotional support at and after diagnosis, along with more informational support on fertility and cancer-related sexual dysfunction (Lewis, et. al., 2012). Sexual function changes after breast cancer in most women. A study reported that ninety percent of women suffer from some type of sexual dysfunction (Cobo-Cuenca, et. al., 2018).

Silence or 'quite' acceptance about breast cancer coping is much pronounced in Muslim women (El-Obaid, 2014; Al-Amoudi, et. al., 2013; Singh-Carlson, et al., 2013; Sheppard, et. al., 2010; Banning, et. al., 2009). For many Muslims, the secrecy of illness and coping



mechanisms is very personalised and involves a spiritual experience. For instance, one study conducted in Muslim women found out that spirituality is the primary source of psychological and emotional support among participants (Harandy, et al., 2008). Another study among Muslim women described similar findings of cultural influence on women's emotional attitude, *"initially I did not tell anyone, and I hide it from my family, I went for spiritual treatment, and I felt better"* (Banning, et al., 2009). It is important that future researchers focus on the interrelationship of survivors' emotional needs from different socio-demographic backgrounds to explore social or cultural determinants. Cultural sensitivity conferred with breast cancer requires further research which could help in developing effective culturally sensitive health interventions.

There is no right or wrong way to manage emotional distress but only what is right and workable for the individual. It is natural to be anxious since women are coping with a life-threatening illness. To manage stress and anxiety, strategies such as distraction which involves diverting focus on learning new things of interest or using a hobby, relaxation or meditation to reduce tension in mind and body, regular physical activity such as brisk walk, yoga and adapting healthy living can help them cope at a difficult time and navigate through the cancer experience by levelling down the stress and anxiety. In addition, counselling could identify feelings such as anger, anxiety, grief or other sources of emotional disturbance that could be related to cancer diagnosis. Cognitive behaviour therapy can also help changing patterns and thinking and behaviour to improve patients' current mindset (Cully & Teten, 2008). Healthcare professionals may recommend anti-depressant however the anticipated result could take longer to achieve full benefits (Machado-Vieira, et. al., 2010; IQWiG., 2006). Family, friends and neighbours are more likely to provide the most needed emotional support; however, speaking to local health providers may benefit particularly for an appropriate referral (Recklitis & Syrjala., 2017).

There is no standardised psychological response to cancer diagnosis, but many patients experience the emotional consequences to be the hardest to deal with. (Towers, 2016). Breast cancer patients can experience depression at any level of the disease from diagnosis to the end of the treatment or even post-treatment. It is also important to recognise depression at the earliest possible time. Patients must seek support either through their close circle of family or friends or professional counselling (Gordon, et. al., 2010). The available local support groups



could provide confidential and non-judgemental emotional assistance that is most relevant to the patient (Matthews, et. al., 2004; Eakin & Strycker, 2001). Most cancer patients were found to receive inadequate emotional and social support from their health providers (Wang, et. al., 2018; Cox, et. al., 2016; McKee & Schover, 2001; Schover, 1999). A study observed that only 22% of oncologists provided an empathetic response to their patients (Pollak, et. al., 2007). Cancer may be a lifelong process and require adjustment over time (Naus, et al., 2009) thus interventions to address emotional needs must meet the needs of newly diagnosed breast cancer patients as well as long term survivors to adjust the ongoing survivorship issues.



2.7.5 SPIRITUAL CONCERNS OF BREAST CANCER SURVIVORS

Spirituality is an inherent component of being human and is subjective, intangible, and multidimensional (Tanyi, 2002). According to Wright (1999), spirituality is the summation of our values which determines the process of how we interact with the world. It is worth noting that not everyone has faith in God; this means that some people explore spirituality through other means for instance; significant relationships, self-chosen values and goals (Tanyi, 2002).

Religion and spirituality were reported to have a positive influence on mind, body and soul which resulted in better health outcomes (Bowie, et al., 2003; Koenig, 2000; Johnson, et. al, 1999; Tix & Frazier, 1998). Studies have found a well-connected link between religious coping with spiritual growth which resulted in better mental health and positive outcomes following stressful life events (Ano & Vasconcelles, 2015; Pargament, 1997). Crouch & Meurier (2005) argued that meaningful care includes care that meets the spiritual needs of patients. The heightened distress experienced by the cancer survivors may have a spiritual dimension which significantly influences patients' self-esteem and personal relationships (Hatamipour, et. al., 2015; Maxwell & Aldredge-Clanton, 1994).

Cancer patients, being ready to help others increases the meaning and hope in their life and yet brings hope to others (Samson & Zerter, 2003). Stephenson, et. al. (2003) explored the experience of spirituality in the lives of hospice patients and illustrated that 93% of sampled patients with cancer expressed that spirituality helped them to sustain their hopes. Rahnama, et. al., (2012) further stated that spiritual and religious resources can lead to an overall sense of hope and optimism toward life. In a small study, Moch (1998) reported that breast cancer women found the meaning and purpose in their lives through their connectivity with the environment, self and others. Hulett, et. al., (2017) studied a group of breast cancer survivors and found that positive spiritual beliefs were associated with neuroendocrine-mediated peak cortisol awakening response activity. Another study by Yanez, et al. (2009), examined the relationship between spiritual well-being and psychological adjustment in 418 breast cancer patients and found a decline in depressive symptoms and increase in vitality within a 12-month period.

A study on Muslims' beliefs on breast cancer reported that "It is God who determines who develops breast cancer (and other diseases) and who is cured" (Johnson, et. al., 1999. Khayat



(1997) also noted that Muslims believe that disease and calamities are sent to people to shed their sins and are tested for their patience and faith. On the contrary, seeking treatment for illness is not regarded as a sign of conflict with reliance on Allah for a cure. According to Prophet Mohammad's (Peace Be Upon Him), "There is no disease that Allah has created, except that He also has created its treatment". The holy book Qur'an contains guidance for Muslims that every disease has a remedy for instance; "Say: It is for those who believe, a guide and a healing" [Fussilat 41:44]". "And We send down of the Quran that which is a healing and a mercy to those who believe" [al-Israa' 17:82]. "Muslims' views towards health and illness incorporate the notion of receiving illness and death with patience, meditation and prayers (Rasool, 2000). The belief is that a Muslim person should accept his afflictions with patience and should pray Allah to reduce his suffering. Another study on Arab women emphasized the importance of spirituality and religion to cope with the disease (Haddou, et. al., 2016).

It is critical to understand the link between religious values and beliefs which shape patients' notion of health and illness. In Islam, health is considered as a form of blessings from God (Allah) and illness or suffering is viewed as a test from Him (Lovering, 2012; Rasool, 2000). Muslims view God as the sole controller of health and illness (DeShaw, 2006; Ypinazar & Margolis, 2006). For Muslims, it involves connectedness with Allah through remembering and surrendering oneself to the Will of Allah (Ahmad, et. al., 2010). At the same time, Muslims are obligated to make every effort to seek the best possible treatment as well for an illness.

Spirituality is an essential element of personalised care. Spirituality as a whole provides a purpose of life as many cancer patients often reach a point of uncertainty and loss of hope which affects not just the body but the soul too (Villagomeza, 2005). In addition to other physical and psychosocial concerns, the treatment plan must also include spiritual needs to gain optimism and positive feelings toward health and life and strengthen one's relationship with Allah (God). Survivorship guidelines from the American Society of Clinical Oncology (ASCO) also acknowledge the effects of primary cancer therapy of spiritual nature. Two of the four survivorship guidelines (National Comprehensive Cancer Network & American Society of Clinical Oncology) recommend management suggestions including consideration of chaplaincy or spiritual interventions. In addition, the National Comprehensive Cancer Network Distress Guideline recommends chaplaincy care for distress management, including spiritual



assessment and counselling (Swenson, 2016). Integrating spirituality as part of the care process will alleviate the physical and psychological symptoms and will result in better health outcomes. Nevertheless, further research is required to design standardised measures on spiritually based interventions and to understand how to best implement these interventions into a patient's treatment.

The next section will discuss the disparities in cancer survivors' needs and survivorship focusing on Arabs Muslim community in the cultural and ethnicity framework.



2.8 DISPARITIES IN BREAST CANCER SURVIVORSHIP

ARAB MUSLIMS CULTURE & ETHNICITY

Psycho-oncologist Holland, J. C., (2010) said that culture plays a big role in how cancer is discussed particularly regarding the truth about a diagnosis, how the illness and its treatment is perceived and how patients and families can cope up with it (Mendoza-Dreisbach, & Dreisbach, 2018; Benowiltz, 1999). Culture is defined as a set of shared and socially transmitted ideas about the world that are passed down from generation to generation (Daher, 2012), whereas ethnicity is a social construct which is defined by individuals themselves (Dewis & Gribbin, 2009). It represents shared cultural values, beliefs and behaviours (Morre & Spiegel, 2004). In US literature, 'ethnicity' has often been considered a proxy for socio-economic status (Meer, 2014; Brawley, 2002). The association of ethnicity and socioeconomic position on cancer survival has also been acknowledged (Shen, et. al., 2007; Merkin, et. al., 2002). Schmitz, et. al., (2013) also described ethnicity as an independent predictor of disease-free survival in breast cancer patients.

Ethnic differences have been observed in cancer biomarker levels. For instance, a study showed that the Ki-67 labelling index, a biomarker of invasiveness in breast cancer, was higher in Arab/Moroccan patients compared with European individuals (Preat, et. al., 2014). Researchers have also reported ethnic variations in lifestyle factors such as diet and body, tumour biology as well as in tolerability and treatment response among breast cancer patients (Bhoo-Pathy, et al., 2012; Tellj, et al., 2011; Wallace, et. al., 2011; Ma, et. al., 2010). Survivors' needs differ according to demographics and clinical characteristics (Søndergaard, et. al., 2013). Ethnic and cultural disparities in mortality and survivorship among breast cancer patients have also been addressed with significant differences found in almost all domains of needs (Im, et. al., 2008). However, little attention has been directed towards identifying needs associated with ethnicity or underserved populations (Moadel, Morgan & Dutcher, 2007).

Culture and ethnicity both influence an individual's needs, attitudes and response to cancer and can be targeted to improve comprehensive cancer care and to enhance health equity among cancer survivors (Surbone & Halpern, 2016; Kagawa-Singer, et. al., 2010; Corner & Bailey 2008). Such influences are complex and may vary across different cultures (Fielding, et. al., 2013). For instance, in western culture, patients like to discuss and demand more knowledge



about different treatment options as well as western medical literature tends to prioritize patient autonomy and corresponding truth telling (Rosenberg, 2017; Benowiltz, 1999). In faith-based communities, prayer is often used to cope with disease (Aschenbrenner & Veneable, 2009). Wang, et. al., (2013) observed substantial evidence of influence of ethnicity towards cancer for instance, people with darker skin colour have found to perceive illness in different ways and have varied response to disease outcome compared to whites. Alden, et. al., (2018) further suggested that health providers should avoid East–West cultural stereotypes.

Apparently, there is a lack of evidence on how cultural backgrounds interact with cancer care delivery and supportive needs of breast cancer survivors. Particularly, the literature on Arab cancer survivors' needs in the UAE and Gulf region is limited. Nevertheless, the physical and information needs are more commonly addressed in most of the cultures (Tariman, et. al., 2014; Binkley, et. al., 2012; Bilodeau & Degner, 1996; Luker, et. Al., 1995). The concept of dependency has been uniquely reported in Egyptian Arabs (Ali, et. al., 1993). The language, communication, information and the need to get relief from dependency (being dependent or being a burden on the family) are the most frequently reported unmet needs among Arabs (Alananzeh, et. al., 2016; Shaw, et. al., 2013). An Australian study found that access to an interpreter did not eradicate the relevant need of information and support for migrant Arab patients (Butow, et al., 2011). Arabs in the western world have reported higher needs of additional "high quality" information to be available in their native language (Alananzeh, et. al., 2016). Shaw, et. al., (2015) suggested that information needs appear to be pivotal for the Arab community; however, little is known about the extent of information they have been provided and how it has been utilised. Strong association of information needs were also linked to Asian cancer patients as reported by Levesque, et. al., (2015). Another study found that 26% of Arab patients did not answer items related to sexual needs while conducting functional assessment of cancer therapy and 18% skipped these items suggesting that sexual needs are not commonly disclosed among Arab culture and thus could be under-reported (Bell, et al., 2013).

In the Arab world, cancer is often deemed as a social stigma (Mellon, et. al., 2013; Bou Khalil, 2013; Donnelly, et. al., 2013; Daher, 2012). There are several myths in the Arab community regarding cancer (Habib, et. al., 2010) for instance, it is contagious and if removed, it can spread. It is often perceived as a death sentence or God's will or punishment (Padela, et. al., 2014; Daher, 2012; Walaa, et. al., 2012). The complex cultural beliefs and health behaviours in



Arabs have a huge influence on how cancer is discussed and perceived (Walaa, et. al., 2012). Arab women, in general, tend to keep the disease confidential and seek medical facilities only when they feel ill or symptoms appear (Amin, et. al., 2009). Lack of knowledge about breast cancer has also been reported with the majority of them having been found to have false beliefs regarding treatment as well as prevention (Al-Sharbatti, et. al., 2014). These beliefs may serve as a key barrier to healthcare access in many Asian and Middle Eastern populations (Meiser, et. al., 2001).

In many Arab countries including the UAE, cancer is generally referred to as a "sign of shame" for a woman as well as for her family (Elobaid, et. al., 2016; Kawar,2013). Amongst Arabs, shame and honour are highly emphasized within the context of kinship as personal ill health not only dishonours the individual but also the entire family. Breast cancer represents a cultural stigma to be dealt with in utmost secrecy; thus, women prefer to be silent sufferers who tolerate considerable distress and seek help at advanced stages (Ravichandran, et. al., 2011; Abdel, 2000).

Breast cancer and mental illness both are viewed as a matter of shame and kept hidden as community will view the condition as hereditary defect which might affect one's social standing and eligibility of marriage among their tribes or families as most Arabs marriages are inter-family or inter-community (Hamamy, 2012; Dhami & Sheikh, 2000). Arab marriages are often arranged to secure the wealth with a family thus the consanguinity concept is socially acceptable in the Arab world (Bromfield, 2014). The age of marriage for women is also very low compared to other countries; most women marry in their teens.

Cultural barriers could be viewed with regards to women's relatively early age at the diagnosis and later stages at presentation in Arab countries (Elobaid, 2014; Mehdi, et. al., 2014; Zidan, et. al., 2012). Many Muslim women bear more household duties which limits their time to take care for themselves for instance raising children is considered to be main duties of Arabian women (Ezenkwele & Roodsari, 2013). Many Arab women keep their own cancer diagnoses secret from their children in order to protect them (Elobaid, 2014; Mellon, 2013). Although studies have addressed women's difficulty in communicating with children due to fear of causing them emotional harm and informing them of their possibly hereditary risk of cancer, no research has been conducted so far on how Arab women communicate about their cancer illness with their children (Mellon, et. al., 2013).



Family is the foundation of Arab culture. The Arab's notion of family goes beyond the immediate relatives and extends to all members of a tribe. Muslim women's health priorities are particularly aligned with the patriarchal family structures and Islamic religious values, for instance, perception of cancer as punishment and death, sterility from cancer leading to rejection and divorce, genetic family illness causing dishonour and stigma or no physical exposure to physicians (Baider & Goldzwig, 2016; Anstett, 2016). An Arab American study suggested that the low use of genetic testing is likely due to structural and cultural barriers that reduce access to preventive healthcare (Mellon, et. al., 2013). Moreover, obtaining an accurate family history could be another challenge to effectively operate screening services in the community which eventually result in underestimation of hereditary cancer risk.

Religious beliefs and cultural differences have also been identified influencing Muslim women's choices in healthcare utilisation (Fitzpatrick, 2016). Arab women tend to refuse further diagnostic workup after being diagnosed as having suspicious malignancy (Elobaid, 2014). The refusal is based on Arab women's belief that if others come to know about it then her daughter would lose the eligibility of marriage and would be considered as undesirable to other families. She may pursue further workup with a reassurance from health providers to keep the information completely confidential. Participation in any healthcare activity would lead to a possible early diagnosis which could conflict with Muslim traditional duties (Remennick, 2006). Moreover, patriarchal family structures, family interference, male approval and women's role and commitment are such factors which have been observed conflicting with the health seeking behaviours. For instance, Arab men may not allow their wives to consult male physicians as exposure of the female body misperceived as forbidden by Islamic religion as well as parents may not allow their daughters to consult physicians about breast cancer concerns (Kawar, 2013).

The families in Arab culture often dictate the care including the extent of the care. Families are viewed as the principal decision-makers. Physicians frequently discuss major ethical decisions with the family without the patient's involvement (Young, 2014). Kawar (2013) also reported that there are cultural consequences when a woman gets breast cancer as the man will be advised by his family to leave her and to marry another woman. Thus, most women are likely to keep the illness a secret and avoid seeking medical support. Halligan (2006) reported that family involvement is a significant contributor to the emotional, social, and psychological well-



being of the patient. Mataoui & Sheldon (2016) also emphasized on understanding the gender role in Muslim communities to consider when planning care and sharing information.

Moreover, cancer could result in sexual discomfort or dissatisfaction; however, these concerns are not revealed or discussed with care providers due to conservative cultural views (Elobaid, at. al., 2016). Marital breakdown is not more frequent among breast cancer survivors in Arabs, yet sexual dissatisfaction could be recognised as fairgrounds for divorce both religiously and by law (Ashrof, 2005). The belief that husbands abandon wives who have breast cancer is not uncommon in this community which could aggravate stress and deteriorate emotional wellbeing of women particularly in younger women (Elobaid, et. al., 2016).

Increased awareness and sensitivity of cultural diversity is critical to understand cancer patients' concerns and needs and is a prerequisite to design effective need assessments (Kagawa-Singer, et al., 2010; Andrulis & Brach, 2007). Kai (2003) points out, effective cross-cultural communication means being sensitive to an individual's culture in its broadest dynamic sense including ethnicity, socio-economic background, education and religion.

Healthcare systems should be designed to serve an ethnically diverse patient population, providers need to consider various patient perspectives, values, and behaviours about their health. Failure to do so will result in healthcare inequalities and poor health outcomes (Agency for Healthcare Research and Quality, 2015). However, to design culturally tailored effective cancer care interventions, a set of congruent practices and policies are crucial in addition to awareness of the health beliefs and effective communication with patients (Truong, et. al., 2014). Baty, et. al., (2003) further stated that information instruments tailored to patients' culture have been successfully implemented in other ethnic minorities in health improvement programs. Such methods could be effective and result in improvement of Arab women's knowledge and understanding of cancer disease and treatments.

Identifying the determinants for ethnic disparities is critical for improving care (Lam, et. al., 2011), yet little is known about the influence of culture and beliefs about breast cancer, and its implications on preventive health behaviour among the UAE population. It is critical to consider a multidimensional contextual framework and the role of the culture while understanding breast cancer patients' needs in this region so appropriate and effective interventions could be developed with cultural knowledge and sensitivity.



The above literature review provided a richer understanding of what are the concerns of breast cancer survivors, the contexts in which they are generated and how the patients and providers respond to those needs.

Tower (2016) described cancer diagnosis as a dreadful experience which disrupts the life of even the strongest individuals resulting in psychosocial consequences mainly, emotional distress, sadness, depression, anxiety and to the extent that families could break down too. Cancer diagnosis is not necessarily a life-threatening condition; however, it often results in physical and psychological burdens. In most cases, cancer survivors may not return to "normal life" after treatment. Cancer and its treatment impact an individual's life from the day of diagnosis through the long-term survival; therefore, the term 'survivor' which has been used in this study recognises the cancer patient as cancer survivor right from the identification of diagnosis through the various stages of cancer journey.

Many women go through the intensive and challenging cascade of treatment with a diverse range of interventions. Many experience long-term side effects and other challenges for years afterwards. Others with incurable cancer may remain on treatment for the rest of their lives. Short- and long-term effects of cancer treatment can affect every aspect of daily life. Transition at the end of the treatment is generally challenging and adjustment to survivorship could be daunting. Physical issues include but are not limited to fatigue, sleep disturbance, bowel problems, weight changes and incontinence. Emotional and psychological effects include shock, fear of recurrence, distress, depression and low self-esteem whereas common social issues include lack of support, altered relationships, loss of identity and fear of being a burden on family. However, what constitutes a psychosocial need is wholly a unique interpretation of each survivor.

The primary goal of this research was to explore and understand the unique needs of breast cancer survivors through the cancer trajectory to improve our understanding and management of the physical and psychological sequelae associated with the cancer diagnosis and treatment also known as survivorship care. The IOM (2008) emphasised that the continual assessment of cancer survivors' needs is critical to improve their health outcomes and to maximise the quality of their lives.



Although recent advances in the treatment of breast cancer have reduced the morbidity rates (American Cancer Society, 2019a) yet cancer can be a threat to a person's life (Gorman, 2018). Healthcare systems are still lacking to provide the holistic care planning and approach encompassing patient's psychosocial needs to help cancer survivors to cope through the illness (Cadet, et. al., 2016). Likewise, there is no structured cancer survivorship program in the UAE at any cancer centre. Cancer services in the UAE need establishment and implementation of cancer survivorship programs integrating multidisciplinary teams to reflect the full range of needs experienced by cancer patients from the diagnosis to cure or terminal care. UAE health authorities need to ensure that health professionals understand the expectations of their cancer patients and appreciate that psychosocial needs vary by patients' socio-demographics and ethnicity and cultural background. UAE healthcare facilities need to meet the needs of its diverse population to help patients deal with a range of psychosocial issues and if unavailable, appropriate referrals should be made through their network to address patients' specific needs.

To understand breast cancer survivors' concerns and to establish the appropriate survivorship program, local need assessment is the prerequisite. Meeting the full range of psychosocial needs will significantly contribute to survivor's wellbeing and eventually will elevate the quality of patients' life. Cancer survivors are often left in a gap between fragmented cancer care delivery which leads to poor health outcomes due to the lack of survivorship care. Survivors are exposed to the risk of experiencing diverse physical and psychosocial symptoms and seek to enhance daily functioning and quality of life through different interventions. Survivorship care programs are necessary to enhance patients' health, improve survival and to prevent further illness (recurrence) and side effects.

Cancer patients are often uncertain about how to address their needs, and little is known about their specific needs in the cultural context and interventions necessary to meet them. Many people living with cancer have reported their healthcare needs are not well addressed in their care process and showed dissatisfaction at the most fundamental level throughout the cancer journey including diagnosis, treatment and post-treatment. Cancer patients must not only live with such challenges and risks posed to their psychosocial health but also should overcome the obstacles that can interfere with their healthcare and diminish their health and function. Unfortunately, the current healthcare system often fails to adequately recognise and integrate psychosocial needs and resources or offer appropriate referrals.



Understanding and addressing cancer patients' needs must be an integral part of the cancer programs or cancer care however a significant number of patients do not receive the assessment of their needs even though the IOM (2008) has described it as prerequisite for ensuring optimal cancer care and achieving cancer patients' and their families' satisfaction. The post-treatment phase of cancer is largely neglected although better coordination of care among health providers and communication with patients could improve the situation with the provision of critical clinical information flow between health care providers with the use of treatment summary and survivorship care plan as recommended by the Institute of Medicine (Shulman, et. al., 2015). According to the IOM (2008), all components of the healthcare system that are involved in cancer care should explicitly incorporate attention to psychosocial needs into their policies, practices and standards addressing clinical health care and to provide such services to all patients who need them. The IOM (2006) report on survivor care suggests recognising the impact of unaddressed psychosocial problems and recommends more attention to these factors in the design and delivery of health care considering psychosocial needs assessment as an integral part of cancer care.

At large, the needs of people who survive cancer have not been prioritised (Cardoso, et. al., 2013; Ganz, et. al., 2013). This is because the focus of healthcare providers has been on survival, rather than the quality of lives of those who have survived. There is a call for both addressing the specific needs throughout the cancer journey and further ensuring those needs are met. The specific aim of survivorship care is to optimise the patients' physical, psychological and social functioning while minimising the limitations resulting from the side effects of cancer treatment. Limited research has been conducted on the complexities of cancer survivorship needs in the context of culture and ethnicity.

The purpose of conducting this research is not only to identify and address breast cancer survivors' needs but also to identify deficiencies and to recognise resources that may already be available and those which are necessary but are not in place to meet survivors' unique needs. Formalised and systematic assessment of survivors' needs is prerequisite for ensuring sufficient support through their cancer trajectory. Thus, the exploration of the unique needs of breast cancer survivors will help determine which model of care delivery is most feasible for the UAE's diverse population. This is thought to be the first study to investigate and describe the needs and concerns of breast cancer survivors and provides plans as informed by patients



at different stages of cancer including the time of diagnosis, during and after the treatment. Furthermore, the study explored the differences in unique needs of survivors by their demographics and clinical factors in order to understand the systems of care that would best meet their specific needs. Since there is no formal survivorship program for cancer patients in the UAE, the healthcare system has no obligation to identify cancer patients' needs as part of their care. The situation underpins an urgent need for a systemic survivorship program encompassing four essential components of survivorship including prevention, surveillance, intervention and coordination based on needs assessment of cancer survivors. Coordination among different healthcare providers will be needed to establish such an intervention ensuring the high-quality cancer care and support that every cancer patient deserves.

The next chapter describes the methodological approach and justification of the specific chosen methods citing relevant literature.



Chapter 3: METHODOLOGY

This chapter starts with describing how a sequential mixed methods study design is considered to achieve the aim and objectives of this study. The chapter starts with the brief description and justification of the selected design, followed by a description of the study design including participants and setting. An integration of the findings from two phases is discussed, which forms the key component of a mixed methods study. Following that, detailed descriptions of the methodology of the study's quantitative and qualitative components is presented. This chapter also describes the status of ethical approval and permission to conduct the study following IRB approval. The below section provides the philosophical overview of research design and describes details about a mixed methods design which was adapted to conduct this research.

3.1 PHILOSOPHICAL ORIENTATION OF RESEARCH DESIGN

The philosophy of science can be divided into two major areas: (a) Ontology, or the science of being, poses questions about *'what is'* and (b) Epistemology, the science of knowledge, is concerned with *'how we know'*. Ben Letaifa (2006) argued that epistemological questions correspond, necessarily, to our ontological assumptions about the reality under investigation. In other words, questions of how we know intersect with questions concerning what sorts of things there are to know about (Masadeh, 2012). In the context of healthcare, it is critical to understand how certain research methods and certain approaches to data collection have emerged from vastly different traditions, and more importantly, how they produce different understandings of the social world. In the healthcare research context, a researcher's paradigmatic positioning relates to their understanding of reality (their *ontological standpoint*) and the nature of knowledge (*their epistemological standpoint*).

According to Guba & Lincoln (1994), paradigms are superior to methods of enquiry in research. Quantitative research predominantly assumes a positivist world view (Henn, et. al., 2006) which are called paradigms and tied to research techniques firmly (Hughes, 1990). Quantitative research paradigm emphasizes the importance of generalizability and reliability (Henn, et. al., 2006). However, positivism imposes certain constraints on results and may



neglect important findings while capturing complex phenomena in a single (or a few) controlled quantifiable variable(s) which can be misleading (Weber, 2004). Sociologists argued that people's experiences are complex, subjective and embedded in specific social and historical context (Silverman, 2001).

Positivist approach does not provide sufficient attention to a person's lived experience (Rubin & Rubin 2005; Bryman, 2001) and lacks a detailed explanation of causes and processes of a research phenomenon and fails to acknowledge that the world is fragmented with disorganised units which must be observed distinctly through interactions (Balarabe Kura, 2021). Hammersley (2013) further argued strongly that it could be impossible to measure phenomena related to intention, attitudes, and thoughts of a human because these profound concepts may not explicitly be observed or measured with sense experience or without evidence. To overcome such limitations with positivism, social scientists need to endeavour the integration of supplementary descriptive methods along with the positivist approach such as the interpretivist methodology (Barbara Kura., 2012; Guba & Lincoln., 1994).

Interpretive methodology focuses on investigating interaction among individuals as well as the historical and cultural contexts which people inhabit (Creswell, 2009). In interpretivist paradigm, the fundamental endeavour is to understand the subjective world of human experience (Guba & Lincoln, 1989). The key principle of the interpretivist paradigm is that reality is socially constructed (Bogdan & Biklen, 1998). Interpretivism uses research methods to understand details of interaction in humans' context with the belief that social reality is based on subjective interpretation of actions. Interpretivists adopt a relativist ontology in which a single phenomenon may have multiple interpretations; therefore, researchers tend to gain a deeper understanding of the phenomenon and its complexity in its unique context rather than generalising the understanding to a larger population (Creswell, 2007).

Reflecting on the value of interpretive paradigm, there are shortcomings of the approach that although the interpretive paradigm is sensitive to individual meanings that can be neglected within broader generalizations (Cohen, et. al., 2011; Samdahl, 1999). Mack (2010) criticized that the ontological view of interpretivism tends to be subjective rather than objective. Knowledge produced by this paradigm has limited transferability as it is usually fragmented and not unified into a coherent body (Otoo, 2020, Scotland, 2012). Hence, it raises a question about verifying validity and usefulness of research outcomes with using scientific procedures



(Hesse-Biber, & Johnson, 2015; Scotland, 2012. However, there is no real consensus was noted among qualitative researchers on appropriate methods of data collection and analysis (Bryman, 2001). The author claimed that qualitative researchers often adopt convenience sampling which results in samples that could be seen biased through the researchers' assumptions and respondent self-selection strategies.

Mixed methods that combine both quantitative and qualitative approach in a single study, correspond to another paradigm called **"Pragmatism"**. Pragmatism is based on the belief that theories can be both contextual and generalizable by analysing them for "transferability" to another situation thus maintaining both subjectivity and objectivity in the research process (Shannon-Baker, 2015). According to Creswell & Clark (2011), pragmatism assumes that there can be single or multiple realities that are open to empirical inquiry.

A major underpinning of pragmatist philosophy is that knowledge and reality are based on beliefs and habits that are socially constructed (Yefimov, 2004) but some versions of those social constructions match individuals' experiences more than others (Morgan, 2014). Pragmatism is typically associated with abductive reasoning that moves back and forth between deduction and induction thus select the research design and the methodology that are most appropriate to address the research question (Kaushik & Walsh, 2019). In this approach, researchers emphasize on the research problem and use all approaches available to understand the problem rather than focusing on methods (Rossman & Wilson, 1985). The pragmatist researchers look to the 'what' and 'how' to research based on the intended consequences where they want to go with it.

Truth is what works at the time. It is not based on a duality between reality independent of the mind or within the mind. Thus, in mixed methods research, researchers use both quantitative and qualitative data because they work to provide the best understanding of a research problem.

3.2 OVERVIEW OF MIXED METHODS

Choosing the right methodology for a given research project can be a challenge. Creswell, (1994) and Newman & Benz, (1998) highlighted that the selection of appropriate research tools requires meticulous understanding of the type of research question or phenomenon under study or investigation. In addition, careful consideration of the process at each stage must be paramount to select the right tool and must consider factors that may influence the study and



thus must be tailored. According to Burrell & Morgan (1979), subscribing to a given research paradigm can be somehow seen as an act of faith. In some scenarios, the question under study may call upon for a combination of different methodologies (Yin, 1993); Newman & Benz, (1998); Tashakkori & Teddlie, 1998), all of which is dictated by the larger research paradigm to some extent which in turn can be influenced by the research process itself. It is important that the research question must be designed in a way that lends itself to a mixed methods design, ideally one in which using a method alone (qualitative or qualitative) would be insufficient.

The term "mixed methods" refers to an emergent methodology of research that advances the systematic integration, or "mixing," of quantitative and qualitative data within a single investigation (Wisdom & Creswell; 2013). The use of mixed methods allows for a deeper understanding of a phenomenon of interest than the use of either a quantitative or qualitative approach alone, especially when the phenomenon is complex (Fawcett, 2015; Doorenbos, 2014). Positive power and synergy of using mixed methods complements one's research findings.

Mixed-method research is currently being accepted as a 'new research paradigm' with an intention not only to re-establish the prevailing paradigms but to mitigate the weaknesses of individual research approaches (Creswell, 2013). This method can be particularly useful in healthcare research as only a broader range of perspectives can do justice to the complexity of the phenomena studied (Foss & Ellefsen, 2002; Steckler, et. al., 1992; Clarke & Yaros, 1988).

Kroll & Neri, (2009) characterised mixed methods research as one which involves integration of the qualitative and quantitative findings at some stage of the research process, either during data collection, analysis or at the interpretative stage of the research. Mixed methods research includes both quantitative and qualitative approaches. Quantitative methods are based on numbers to claim objectivity, whereas qualitative methods generate theories relying on subjectivity (Kaur, 2016). The approach draws on potential strengths of both quantitative and qualitative methods allowing researchers to explore diverse perspectives and uncover relationships that exist between intricate layers of multifaceted research questions (Shorten & Smith, 2017).

Mixed methods researchers use and often make explicit diverse philosophical positions. These positions often are referred to as dialectal stances that bridge postpositivist and social



constructivist worldviews, pragmatic perspectives, and transformative perspectives (de Salazar & Villar, 2018). For example, researchers who hold different philosophical positions may find mixed methods research to be challenging because of the tensions created by their different beliefs (Greene, 2007). Yet, it presents an opportunity to transform these tensions into new knowledge through a dialectical discovery.

Johnson & Onquegbuzie, (2004), clarified that mixed methods research can be viewed as an approach which draws upon the strengths and perspectives of each method recognising the existence and importance of the physical natural world as well as the importance of reality and influence of human experience. It allows combining macro and micro levels of a study by enriching the understanding of the measures and causal processes, constituting an interactive process, both in design and analysis and interpretation of data (Creswell, 2013). Onwuegbuzie & Leech (2004) described the advantages of using mixed methods as combining quantitative and qualitative research enables evaluation researchers to be more flexible and holistic in their investigative techniques, as they endeavour to address a range of complex research questions that arise. Moreover, it allows researchers to develop a framework to validate quantitative results by linking the information extracted from the qualitative phase of the study and to construct indices from qualitative data that can be utilized to analyse quantitative data (Madey, 1982). There is a general consensus that the quality of mixed methods study can be enhanced when the limitations of using a single approach can be counterbalanced (Cheng, 2012). It provides researchers, across research disciplines, with a rigorous approach to answering research questions (Aramo-Immonen, 2013).

The strengths and weaknesses of this mixed-methods design have been widely discussed in the literature (Creswell, 2003, 2005; Moghaddam, Walker & Harre, 2003; Green & Caracelli 1997; Creswell, Goodchild, & Turner, 1996). The key strengths of mixed methods research are words, pictures, and narrative that are used to add meaning to numbers. This methodology is particularly beneficial and appropriate when results arising from a quantitative study are unexpected (Morse, 1991). The main limitations of this design are lengthy time and feasibility of resources to collect and analyse both types of data. Since the research utilises both quantitative and qualitative methods, it can answer a more broad and complete range of research questions since it is not confined to a single method. Thus, use the strengths of an additional method to overcome the weaknesses in another method in a single study. It can



provide stronger evidence for a conclusion through convergence and corroboration of findings and add insights and deeper understanding that might be missed when a single method is used.

The method can be used to increase the generalizability of the results and to assess the validity of quantitative finding by using the qualitative data. Therefore, mixed methods approach guarantees greater validity of the research results (Ben Letaifa, 2006; Sale, et. al., 2002; Threlfall, 1999; Tashakkori & Teddlie, 1998). Quantitative data can also be used to help generate the qualitative sample or explain findings from the qualitative data. Moreover, the qualitative inquiry can inform the development or refinement of quantitative instruments or interventions or generate hypotheses in the qualitative component for testing in the quantitative component (O'Cathain, Murphy & Nicholl, 2010). Succinctly, mixed methods produce additional knowledge necessary to inform theory into practice. Its advantages include straightforwardness and opportunities for the exploration of the quantitative results in more detail. As claimed by Tashakkori & Teddlie (2003), mixed methods research eventually has become the standard methodological approach in both social and behavioural sciences.

Nevertheless, there are weaknesses of mixed methods research as the design can be a challenge for a single researcher especially if two or more approaches are expected to be used concurrently thus it may require a research team. It places high demands on researchers who may require additional skills and learning about utilising multiple methods and approaches to understand how to integrate them into a single study appropriately. Although there are many potential gains from data integration in mixed methodology, the extent to which mixed methods studies implement integration remains limited (Lewin, Glenton & Oxman 2009; Bryman 2006). Bazeley (2009) also highlighted the most concerning issue, is the guidance on the analysis and the integration of qualitative and the quantitative data in the mixed methods research. Pragmatic guidance is needed since the current literature is lacking such understanding (Östlunda, et. al., 2011; Creswell & Clark., 2011; Curry, Nembhard, & Bradley., 2009).

The lengthy nature of mixed methods research generally makes them unsuitable for publication in journals (Creswell, 1994). It is relatively more expensive and more time consuming and labour intensive. Although, it is easy to implement and enables the focus of the research to be maintained, as a result of one set of data building upon the other yet the challenge lies in the selection of participants in order that pertinent information is available (Almalki, 2016), hence,



the recruitment for the qualitative phase could be problematic as the participants may no longer be available due to a longer timeframe required to undertake the study. The other challenge of this design is the fact that it is generally an emergent design where the second phase cannot be fully developed until the first phase is completed. This may result in a second ethics or institutional review board (IRB) approval for the qualitative phase (Creswell & Clark, 2011).

Additionally, as the need and the use of mixed methods research is rapidly growing, it is also becoming increasingly important to consider the issue of quality (O'Cathain, et. al., 2008, 2007). The author also observed that the reporting of mixed methods studies is quite challenging in terms of clarity and transparency. Erzberger & Kelle, (2003) suggested that using triangulation as methodological metaphor and by explicitly showing the types of data used as the basis of interpretation, the lack of quality issue in reporting the result of mixed methods research could be tackled. Some research studies may require some details of mixed methods research to be worked out fully by a research methodologist for instance the problems of paradigm mixing such as how to qualitatively analyse the quantitative data and how to interpret conflicting results.

It is essential that mixed methodologist meet two primary goals which were developed by the American Educational Research Association (2006): Mixed methods researchers should (1) conduct and report research that is warranted or defensible in terms of documenting evidence, substantiating results, and validating conclusions; and (2) ensure that the conduct of research is transparent in terms of clarifying the logic underpinning the inquiry (Wisdom, et. al., 2012). It has been concluded that researchers' devotion, persistence and a lot of rigor is required in order to apply mixed methods for valid and reliable findings.

Several frameworks have been published to guide the rigorous conduct and evaluation of mixed methods research (Creswell & Clark, 2011; Tashakkori & Teddlie, 2010; Curry, Nembhard, & Bradley, 2009; Collins, Onwuegbuzie, & Sutton 2006). The frameworks recommend explicit rationales for all decisions regarding study design including the justification of employing both qualitative and quantitative methods. It is advised that study procedures should be reported transparently, including sufficient detail to allow the reader to make inferences about study quality (Curry, Nembhard, & Bradley 2009; Leech, et. al. 2009; Teddlie & Tashakkori, 2009; Armstrong, et. al. 2008; Creswell, 2008). Further advising that each component of both methods should be conducted rigorously in accordance with established principles in its field



while keeping the methodological reporting utterly transparent. For instance, sampling design must be specified (Onwuegbuzie & Collins, 2007), the level of mixing methods whether fully or partially, time orientation whether sequential or concurrent components of research and emphasis of an approach (if any whether equal importance or one more dominant) should be described (Leech & Onwuegbuzie, 2009).

3.3 JUSTIFICATION OF SELECTED METHOD

To conduct this study, mixed methods design was considered as the more appropriate method to address the research questions. Since mixed methods research holds greater potential to address these complex questions by acknowledging the dynamic interconnections that traditional research methods have not adequately addressed (Hesse-Biber & Crofts, 2008; Hesse-Biber & Leavy, 2006), mixing both qualitative and quantitative data was considered as neither quantitative nor qualitative methods were deemed sufficient to capture the details of survivors' needs. By combining qualitative and quantitative findings, the overall findings were verified, which was not possible by using a singular approach (Bryman, 2007). When used in combination, quantitative and qualitative methods complemented each other and allowed for a more robust analysis, taking advantage of the strengths of each (Tashakkori & Teddlie, 1998; Green & Caracelli, 1997; Miles & Huberman, 1994; Green, Caracelli & Graham, 1989).



Figure 10: Explanatory Design

Source: Adapted from Figure 3.2(b): Creswell and Plano Clark (2011: 69)



To conduct this mixed methods research, "Explanatory sequential design" was chosen which is two-phase mixed method design that starts with the collection and analysis of quantitative data followed by the subsequent collection and analysis of qualitative data which follows from (or connects to) the results of the first quantitative phase (Creswell & Clark., 2006). Sequential design with the quantitative component first and the qualitative second, added a "value-added" understanding of the results from both methods (Hesse-Biber, 2010). The combination of mixed methods was chosen to explore some cultural meanings that might be challenging to identify merely through quantitative data, minimising errors that may arise from a single technique and maximising the meaning from the results of data interpretation (Tashakkori & Teddlie 2003; Patton, 2002). The flowchart of the procedures for the chosen design is presented in the figure 11 below.



Figure 11: Flowchart of Sequential Explanatory mixed method design

Adapted from the figure (7.4a) Creswell, Clark, Gutmann, and Hanson (2003, p. 180)

In this design, the data collection and analysis occurred sequentially. The investigation begins by a quantitative analysis followed by qualitative exploration of breast cancer survivors' needs. The quantitative results (what, who, where and when) were integrated with qualitative findings to explain the in-depth understanding (why and how). Based on findings from the survey,



survivors' needs were assessed in UAE cultural context along with sociodemographic and clinical indicators including stage of the illness and treatment received. The results of the quantitative phase guided the "development" of the qualitative phase and assisted in the development of the second phase by creating a synergic effect whereby the result from quantitative data helped develop and inform the qualitative phase (Greene, et. al., 1989).

Phase one findings from statistical data were used to contribute to the methodology of phase Two guiding the purposive sampling and developing a semi structured interview in phase Two where participants were selected as sub-sample from the quantitative phase who showed willingness to be interviewed and scored high, intermediate, or low on the survey for in-depth understanding of the differences in their utmost needs.

Another rationale to use mixed methods design in this research was "**complementarity**" - seeking elaboration, enhancement, illustration and clarification of the results from one method with the results from another' (Greene et al., 1989).

In-depth, semi-structured interviews with participants allowed clarifying and further exploring the different aspects of breast cancer survivors' multidimensional needs that could not be obtained by quantitative methods alone. It was achieved by utilising both quantitative and qualitative data and not just the numerical or narrative explanation alone to understand the social phenomenon as a whole. It was further used as a tool for the cross-validation to produce comparable data (Yauch & Steudel, 2003). Thus, it allowed gaining a fuller understanding and clarifying the findings.

Triangulation was another reason for using this design to add credibility to this study as it used different sources of data and different approaches to analyse the data thus it allowed convergence, corroboration and correspondence of results from different methods (O'Cathain, et. al., 2010; Greene, et. al., 1989) in order to seek corroboration between quantitative methods which provided an overview of the breast cancer survivor's need and qualitative methods to further explore the highest needs during cancer survivorship. Such integration of findings in both methods provided convergent results with in-depth and comprehensive understanding of the research phenomenon which could not be achieved employing a single method.



Rovai, et. al., (2014) suggested that while employing the mixed method research there must be a purposeful and carefully implemented sequence to the study which must be conscientiously documented and evaluated. To establish the **priority and sequence** of the study, whether quantitative or qualitative (or both) to implement first, all the nuances of the study design were

visually represented for conceptual purposes as well as to provide its better comprehension. In solving the issues of priority, the decision-making process was guided by the purpose of the study and its research questions, as well as by the methodological discussions in the literature (Creswell, et. al., 2003; Morgan, 1998; Tashakkori and Teddlie 1998; Morse, 1991).

As this is a preliminary study on an unexplored topic, for the first phase, it was considered that the quantitative data would inform the choice of interview questions and reveal what areas of survivors' concerns are required to explore in a more in-depth approach. Moreover, quantitative initial data collection was deemed easy to administer in the conservative UAE culture since talking about cancer is a sensitive and confidential subject in this society (Younis, et. al., 2016; Elobaid, at. al., 2016, 2014). In the second phase, qualitative semi-structured interviews were conducted to probe deeper understanding of unique needs of breast cancer survivors as perceived by them. Qualitative data illuminated quantitative findings by putting 'meat on the bones' of dry quantitative data (Bryman, 2006). Thus, it helped by featuring the similarities and differences between particular aspects of a survivors' unique needs utilising both methods (Bernardi, et. al., 2007) allowing individuals' subjective experiences to be tak en into account.

It involved the collection and analysis of both quantitative and qualitative data in a single study in which the data were collected sequentially and integrated at later stages in the process of research (Hanson, et. al., 2005; Creswell, et. al., 2003) for the purpose of gaining a better understanding of the research problem (Creswell, 2005; Tashakkori & Teddlie 2003) and to increase the breadth and depth of understanding of survivors' needs (Johnson, Onwuegbuzie & Turner, 2007). This approach was considered as the more appropriate method to address the research question since quantitative data and their subsequent analysis provided a general understanding of the research problem and qualitative data and their analysis refined and explained those statistical results by exploring participants' views in more depth (Creswell, 2003; Tashakkori & Teddlie 1998; Rossman & Wilson1985).



The overall purpose of this design is that qualitative data helps explain or build upon initial quantitative results (Creswell et. al., 2003). Morse (1991) explained that this design is well suited to a study in which qualitative data was used to explain significant (or nonsignificant) results or surprising results. Finally, the quantitative and qualitative findings of the study were integrated by comparing and contrasting quantitative and qualitative data in the interpretation phase to further complement study conclusions. Thus, combining qualitative and quantitative methods yielded more useful and valid results, offering different perspectives on the survivors' needs offering greater overall understanding of the research.

3.4 DATA INTEGRATION

Integration refers to the stage or stages in the research process where the mixing or integration of the quantitative and qualitative methods occurs (Creswell, et. al., 2003; Tashakkori & Teddlie 1998). The integration of qualitative and quantitative approaches in any study is a considerably more complex undertaking and continues to be one of much debate (Bryman, 2004; Morgan, 2007; Onwuegbuzie & Leech, 2005). There are relatively few guidelines about 'how, when and why different research methods might be combined (Bryman, 1988). In particular, the different epistemological and ontological assumptions and paradigms associated with qualitative and quantitative research have had a major influence on discussions on whether the integration of the two is feasible, let alone desirable (Morgan, 2007; Sale, et. al., 2002).

Data integration is the fundamental concept of mixed methods research. Without integration, findings cannot be correlated and will remain independent leading to incomplete understanding of survivors' needs. Data integration can be performed at different stages starting from data collection to data interpretation. Teddlie & Tashakkori (2003), mentioned the possibility of mixing of quantitative and qualitative methods at the initial stages of the study while formulating the purpose of the study and the research questions as opposed to Teddlie & Tashakkori (2003), who suggested the integration to occur at the interpretation stage of the study. On the other hand, Hanson, et. al., (2005), informed that in the mixed methods sequential designs, the quantitative and qualitative are connected in the intermediate stage when the result of quantitative data analysis informs the qualitative data collection. Creswell, et. al., (2003), argued that in the sequential explanatory design, the two phases are typically connected in a way that quantitative results from the first phase guide the selection of participants for the



qualitative follow-up analysis. Another strategy might be the development of qualitative data collection protocols, grounded in the results from the first (quantitative) phase, to investigate and analyse those results in more depth.

In this study, integration occurred at the methods level by connecting the sample frame (Fetters, et. al., 2013) with cancer survivors who completed the survey also participated in the interviews to enhance the quantitative findings. The quantitative and qualitative phases were connected during the research process while selecting the participants for the qualitative study from those who responded to the survey in the quantitative phase based on their numeric scores. Later, the data from two phases were analysed separately and integrated at the interpretative stage to explain the study phenomenon, which is to explore the survivors' needs. This technique was chosen so the two data types could be compared, contrasted and later combined in a way that integrity of data remain preserved while exerting more focus on the potential to attain deeper understanding from the combination of findings of both datasets.

The use of integration leads to three consequences: divergence, convergence, and complementarity (Tashakkori & Teddlie, 2003). If the mixed methods applied in the study lead to divergent results (i.e., the qualitative and quantitative results contradict each other), two possible explanations exist: either the divergence is the result of methodological mistakes or inaccurate initial theoretical assumptions (Erzberger & Kelle, 2003). Thus, one could have an opportunity to revise theoretical assumptions as a consequence of divergent findings carefully so newly developed hypotheses must increase the empirical content of the initial theoretical assumptions without diminishing their consistency. Divergence occurs in cases where qualitative and quantitative results are inconsistent; thus, re-examination of the designs of the two approaches is needed.

If the inconsistency is due to methodological design, then it should be addressed under study limitations. If the quantitative and qualitative methods lead to convergent results (i.e., the qualitative and quantitative methods lead to the same results), then the integration may provide good arguments for the quality of the inferences and strengthen the initial theoretical assumptions (Erzberger & Kelle, 2003; Teddlie & Tashakkori, 2003). As in this research, if a mixed methods approach leads to **complementary results** (i.e., the qualitative and quantitative results relate to different objects or phenomena but may complement each other), then the



integration provides a more complete picture of the empirical domain under study (Erzberger & Kelle, 2003).

The quantitative and qualitative approaches were integrated while interpreting the main outcomes of the study. The qualitative data provided additional complementarity support for the quantitative evidence, hence, further validating the acquired assumptions (Teddlie & Tashakkori, 2003). This mutual verification process enhanced the validity of the findings, which is the primary objective of the mixed methods design. In this study, the consistent and compatible results generated from the two approaches were examined carefully and were reconciled to achieve a coherent interpretation and explanation of the breast cancer survivors' needs. During the interpretation and reporting, integration was applied while explaining the results where the qualitative and quantitative findings were presented separately and discussed in a single section in "Results" Chapter.

Nevertheless, the challenges associated with integration techniques have been well established. Proponents of mixed methods research suggest that the purist view, that quantitative and qualitative approaches cannot be merged, poses a threat to the advancement of science (Onwuegbuzie & Leech, 2005) and that while epistemological and ontological commitments may be associated with certain research methods, the connections are not necessarily deterministic (Bryman, 2004). Maxwell (1990) suggested that the theoretical debate about combining methods has prevented us from seeing the different ways in which researchers are actually combining methods. Researchers may encounter short-term inconvenience while integrating the results, in many cases the reappraisal and reanalysis required can reap long-term analytical rewards: alerting the researcher to the possibility that issues are more multifaceted than they may have initially supposed and offering the opportunity to develop more convincing and robust social explanations of the social processes being investigated (Deacon, et. al., 1998).

3.5 PHASE ONE: QUANTITATIVE METHODOLOGY

3.5.1 ETHICAL APPROVAL

Institution Review Board approval was obtained from the local IRB to conduct both phases (Appendices 1 & 2) and from the University of Salford Health Research Ethics Panel.


Ethical issues were given careful consideration, particularly patient safety and wellbeing, through the study with the availability of support groups and appropriate referrals as needed and to anyone who experienced distress. During the entire study, no patient experienced distress or asked for further support. The survey was conducted in a very safe manner within the centre in a n allocated room with a safe environment. All participants were informed about the purpose, procedure of the study and their voluntary participation by providing the "Patient Information Sheet" (Appendix 4). Participants were fully informed about their right to withdraw at any time in case they felt any anxiety or stressful event. Participants were fully informed of the purpose of the research. For phase one, all participants signed a written consent form (Appendix 3) prior to the survey. Data confidentiality protocols were maintained considerably during the research. To protect confidentiality, each participant used an allocated numerical code instead of their names. The coding list was kept in a locked cabinet, which was accessible only to the researcher.

Following the local IRB approval (Appendix 1), participants were identified and contacted through the cancer registry by telephone in phase one. Patients previously signed the general hospital consent which specifies usage of patients' health data for research purposes (Appendix 1) as customary practice within the hospital. Patients were contacted by the researcher and asked if they were interested in the study by providing a brief overview and aim of the study over the telephone. Agreed participants were given a copy of the participant information sheet (Appendix 4) and the consent form (Appendix 3) upon their due follow-up visit which they were attending at the hospital not particularly for the specific research session only. Those who were interested and had no further questions were given the opportunity to sign the written consent form to participate in the study by completing the survey (Appendix 5).

Prior to each patient's visit, patients' medical records were reviewed to perform pre-screening to collect demographics and specific clinical details including type of cancer, stage of the disease and treatment information with the purpose to gather accurate information during the survey. During the survey, health professionals were available to provide emotional support at no cost to the patients who might have found the survey to be an emotionally stressful experience, yet no referral was made. A research assistant trained in confidentiality and data protection issues was also available for participants who had difficulty with communicating in



English. Additionally, the researcher was also present throughout the survey to answer their questions if participants had any concern about any aspect of this study.

The survey was conducted at the follow-up clinics at Tawam Breast Cancer Centre where an area was allocated for the participants. The expected time of completion lasted for approximately 30 minutes and refreshments were available in appreciation for their time. Confidentiality was maintained by coding the surveys with a subject identification number. The survey was translated into Arabic and verified by the two physician members of the Oncology team at Tawam Hospital who were competent in both English and Arabic languages. The surveys were provided in both English and Arabic languages and were completed by the participants using pen and paper. A hospital translator was present to translate answers for any participants who preferred to complete the surveys in their native language; these translators were trained by the Cancer Centre to consider the sensitive nature of the survey and were committed to following data confidentiality protocols throughout the survey process. The survey was used in accordance with the UAE context by modifying the information that reflects the particular setting, for instance Nationality was recorded as the representative of the passport held by an individual.

3.5.2 PARTICIPANTS & SETTING

The proposed study was conducted in Tawam Breast Care Centre in Al Ain city. The justification for conducting the study at Tawam Hospital is, the hospital is considered as a national referral centre and the centre of excellence in Oncology providing cancer care services to the local community across the region and neighbouring gulf countries. Participants were recruited from Tawam Hospital who were identified initially by the Cancer Registry which is located within the Hospital. At patients' registration, all patients served by the hospital are required to provide an initial consent to allow the usage of their health data for research (Appendix 1) as part of hospital routine policy. The research was conducted after obtaining the ethical approval through the Hospital Research Office (Appendices 2 & 3). Participants were informed that the participation in the survey was not compulsory, and they could withdraw from the study at any time without any consequences. They were also assured that their responses would not affect their care process at the hospital. The survey could be completed



during their follow-up visit either in the waiting area or in clinic rooms available in the Breast Cancer Centre at Tawam Hospital.

Target Population

This study population consisted of all women in the UAE who were diagnosed with breast cancer in the year 2018. Tawam Hospital as national referral centre serves most of the cancer patients in the country and the accessible patients from the hospital could assume to be generalised as target population for the study.

Accessible Population

A total of 368 patients who were diagnosed or treated with breast cancer in the year 2018 were selected for the study

Inclusion Criteria

Women with all ages who were diagnosed with breast cancer and who were currently residing in the UAE. There were almost 368 new cancer cases diagnosed in the year 2018 at Tawam as reported by the hospital Cancer Registry. Based on the previously mentioned criteria, a total of 218 (59.2%) patients were included in this study which provides a good sample size based on the number of new breast cancer cases diagnosed each year.

Exclusion Criteria

Patients were excluded from the study based on the below mentioned criteria:

- Three (0.8%) male breast cancer survivors diagnosed in 2018
- Thirty-nine (10.5%) women with multiple primary cancers who have two or more different types of cancer)
- Thirty-eight women (10.3%) who lost to follow-up or deceased
- Seventy (19%) patients had their stage data missing or unknown whose essential data was missing



A total of 150, (40.7%) were excluded, thus, 218 (59.2%) patients were included in this study which provides a good sample size based on the number of new breast cancer cases diagnosed each year.

The quantitative study aimed to relate the findings to be applied to the general population of breast cancer patients; thus, careful consideration was given to the selection of a sample representative of the population (Delice, 2010). The optimal sample size was determined based on yearly new cases registered in the cancer registry which was between 300 -350. In this study, census sampling was used which often an actual national population (Lavrakas, 2008). The census sampling is defined as the subset of the population selected to represent the entire group, in all its characteristics. The objective of the census sampling is to collect data in relation to every member of the population under study to truly represent the whole population (McLennan, 1999). Since Tawam Hospital serves as national referral centre, the source provided access to all possible members of the population within a reasonable time with almost full enumeration of the target population.

The main disadvantage of a census sampling is access to the population within a reasonable time however in this study, all women were approached during their follow-up visits with no difficulty to reach them. This sampling technique was both cost as well as time efficient. The relevance of bias and sampling error using census sampling was predicted as not being high as all new breast cancer patients were identified through the registry and approached thus, the census sampling method was found more appropriate for the setting. After applying inclusion criteria, a total of **218** patients were recruited to be part of the study.

3.5.3 RESPONSE RATE OF THE SURVEY

The survey response rate was 94% considering the refusal of 13 (3.5%) patients out of a total of 218 patients who participated in the survey.

3.5.4 INSTRUMENT

Despite several needs assessments instruments available to assess the needs of cancer survivors, **Cancer Survivors Survey of Needs** from the Pearlman Mayo Clinic (Appendix 5) was selected to conduct the quantitative phase of the study. The main reason for using this



survey was it was convenient to conduct in the specific patient population. The survey was considered well formulated, relevant to the purpose of the study encompassing all psychosocial domains in addition to areas that might be of potential learning interest for cancer survivors. It was less time consuming compared to other commonly used surveys and was pilot tested on both Emiratis and expatriates in the current setting, thus, considered to be the most effective and appropriate need assessment instrument for this study. Official permission was obtained through email confirmation to utilise the survey in Tawam Hospital setting from the Cancer Education Program, Mayo Clinic Cancer Centre in Rochester, Minnesota, US prior to the conduct phase one (Appendix 13). The Mayo Clinic Cancer Centre developed this survey with the help of an interdisciplinary team of clinicians, administrators and investigators with support from the Mayo Survey Research Centre (consultation and feedback on instrument development, pilot sampling, and data analysis). Additional review of the survey tool was provided by medical, educational, and behavioural researchers. Survey questions were chosen based on current literature recommendations (IOM, 2006; Vachon, 2006; Baker, et. al., 2005) as well as from the developers' experience which was reviewed by members of the cancer patient education network. Data collected from this tool is intended to direct appropriate resources and plan supportive programs for cancer survivors (Jiao, et. al., 2015). The content validity of this survey was established by the author through the review by members of the Cancer Patient Education Network. Single-item linear analogue scales, including Likert scales, have been validated as general measures of global QOL dimensional constructs in several settings (Locke, et. al., 2007; Grunberg, et. al., 1996; Gudex, et. al., 1996; Hyland & Sodergren, 1996; Wewers & Lowe, 1990; Sriwatanakul, et. al., 1983). Reliability was assessed with the Cronbach alpha coefficient of 0.96 calculated across all subscales.

3.5.5 DATA COLLECTION

The data was collected on 69 items out of which 52 items used to rate psychosocial needs on a six-point Likert scale, with 0 indicating no concerns and 5 indicating extreme concerns. It had six domains: physical effects, information issues, social issues, emotional aspects, spirit ual issues and other issues. The survey contained additional questions on educational needs, preferred learning on topics and what specific topics the patient is interested in learning about. Overall, the survey contained eleven sections aimed to provide awareness and insight into the



needs of cancer survivors so those could be evaluated to guide the development of a survivorship program model to address these needs.

- 1. Demographics (7 items)
- 2. Quality of Life Assessment (1 item)
- 3. Physical concerns (20 items).
- 4. Information/Educational Concerns (7 items)
- 5. Social & Financial Concerns (6 items)
- 6. Emotional Concerns (14 items),
- 7. Spiritual Concerns (4 items)
- 8. Other concerns (1 items)
- 9. Mode of delivery assessment (4 items)
- 10. Identification of specific areas of survivor's interest (4 items)
- 11. Assessment of patient's drive through cancer experience (1 item)

Socio-demographic and clinical characteristics

The **first two sections** were based on patient demographics followed by the assessment of survivor's quality of life range from **0 to 10 (scoring from bad to good).** The demographics and cancer information including diagnosis and treatment were also extracted from the Tawam Cancer Registry to further validate survivor's responses. Demographic and clinical variables included age at diagnosis, nationality, year of cancer diagnosis, type of cancer, stage of cancer, type of treatment received and whether currently on treatment or not. In case of any conflicts, responses were verified by reviewing the clinical notes and by further verifying the patients' demographic as well as clinical data which was usually registered by the front desk staff at the hospital and was more prone to errors. **The second section** contains one question which was designed and studied to measure overall quality of life (QOL) using an 11-point Likert scale from 0 (as bad as it can be) to 10 (as good as it can be). This item reflected their quality of life in the past one week prior to the survey.

In sections 3-7, *patients rated their concerns* as of today from 0 to 5 (No concerns to extreme concerns) *to indicate the level of need* using a Likert-type scale which allowed quantifying the needs in each section. This section of the questionnaire assessed the critical need domains as perceived by the survivors.



The last 2 sections of the survey contained three open-ended questions to identify the source of strength during the cancer experience, their topmost concern and any additional comments that survivors may have.

To complete the survey, survivors were asked to indicate whether they were experiencing a need or concern and to rate the level of it. The total summed score (Item 3-7) was used to indicate a greater level of needs and perceptions of greater concerns whereas the lower score represented lower needs and thus represented less concerns. The applied threshold value for excluding incomplete surveys was set at 3 missing sections from the total of 11 items.

3.5.6 QUANTITATIVE DATA ANALYSIS

Quantitative data analysis was conducted using SPSS software. The statistical analysis includes descriptive statistics of socio-demographics and clinical characteristics (extracted from the Cancer Registry) and survivor's needs. To compare the socio-demographics and clinical characteristics of two groups (Emiratis and expatriates), a Chi-square test for categorical variables was used for statistical significance and Student *t*-test for continuous variables (Byrne, 2007). Differences between the Emiratis and expatriates were analysed to see if they are significant, and a significant mean difference was investigated between these two groups. Multivariate multiple regression was applied to compare the differences in each psychosocial need domain in order to adjust for possible confounding effects by sample differences by adjusting the effect of demographic and clinical characteristics since higher needs may be associated with the extension of disease or tumour characteristics).

Following the quantitative analysis, these results were used to inform the development of semistructured interviews in the qualitative phase of the study, particularly those which needed deeper understanding or could lead to multiple interpretations by the patient.

3.6 PHASE TWO: QUALITATIVE METHODOLOGY

3.6.1 ETHICAL APPROVAL

Phase two qualitative part was initiated with the **pre-obtained ethical approval** taken from the local IRB (Appendices 1 & 2) to conduct the mixed methods study.



This study considered all core areas of ethical concerns as described by Willig (2008) and the British Psychological Society Code of Human Research Ethics (2014). It includes informed **consent concerns, confidentiality and consequences of the research** with regards to possible risks as well as expected benefits of participating in the study.

A written consent was obtained from patients who met study inclusion criteria and agreed to take part in the qualitative study. During phase two of the study, all participants were asked to sign the **informed consent** which indicated information about the purpose and the procedure of the study (i.e., 'The interviews will be audio-recorded') and were explained to the participants what was required of them. They were also assured of confidentiality and privacy of their information. They were also informed to have access to the final results of the study. All participants were informed about their **rights to withdraw** from the study, that they could decide to terminate the interviews any time without any ramifications or judgement, and they have the right to choose what information they want to disclose to avoid discomfort. Participants were also assured that there will be no right or wrong statements because the researcher would only be interested in hearing their perspectives and concerns about the cancer experience. **Debriefing sessions** were pre-arranged, and participants were informed prior to commencing the interviews that some aspects of interviews may potentially result in psychological discomfort for which participants would be offered a debriefing session or if needed, would be referred to the required psychosocial support services.

To ensure the **confidentiality** of the interview data in phase two, the audio-interview files were stored on the secure university One Drive where only the researcher had access to the data. The identity of all participants was protected through the study using pseudonyms for instance, all names and personal data were removed and any information that had a potential to compromise the participant's anonymity was altered appropriately to avoid disclosure of the patients' identification. Participants were informed that their interviews would be audio recorded and transcribed after that and would be erased and other data records would be destroyed three years after the final thesis would be submitted to the university. Moreover, it was also declared that all interviews, transcriptions, records and accompanying material would be securely protected using password encrypted electronics on a secure university drive and no printed material would hold any identifiable information about the participants.



3.6.2 PARTICIPANTS & SETTING

All participants in the first phase of the study (quantitative) were asked about their interest to take part in phase two (qualitative). Once potential participants who met the criteria for the phase two were identified and who agreed to participate were contacted over the phone and asked for the convenient time to attend the interviews during their follow-up visit at Tawam Hospital Breast Care Centre. According to Creswell (2012), in any qualitative study, it is important to select people or sites that can best help understand the central phenomenon, thus, the interviews were held at Tawam Hospital Breast Cancer Centre which is known as a national cancer referral centre. Participants were interviewed in the nursing consultation room of the centre as it provided a quiet and safe environment. On the interview day, participants were briefed about the study and were given the oral and written explanations about the purpose of the study. After that their informed consent was obtained.

Sampling in qualitative research is contingent on many considerations since it is not a matter of representative opinions, but a matter of information richness (Guetterman, 2015). Participants for phase two were purposively selected from the sample in phase one. The appropriateness of sampling was achieved through the **purposive sub-sampling** of participants because the purposively selected participants could provide rich in-depth information on their multidimensional needs. Portney & Watkins (2009) described the purposive sampling as a method of non-probability sampling where the investigator selects the study participants based on specific criteria and excludes those who do not fit the criteria. Coyne (1997) & Patton (2001) suggested that this type of sampling usually aims to capture a wide range of the participants' perspectives on the complex phenomenon being studied and mainly to enhance data validity. Smith, et. al. (2009) stated that "samples which are selected purposely (rather than through probability methods), can offer a research project insight into a particular experience".

There is no straightforward answer to the question of 'how many' and that sample size is contingent on a number of factors relating to epistemological, methodological and practical issues (Baker & Edwards, 2012). Morse & Field (1996) suggested that sampling in the qualitative study was guided by two principles: **appropriateness (relevance) and adequacy**. Appropriateness of sampling addresses the quality of data obtained from identifying and



utilizing participants who best represent or have knowledge of the research topic whereas adequacy of sampling refers to the sufficiency of the data gathered to provide a full and rich description of the research phenomenon (Cheng, 2012).

A total of ten participants (five expatriates and five Emirati nationals) were chosen based on their scores on the survey to represent high, intermediate or low scores that allowed maximum variation in their perceptions of the needs which justified the appropriateness of the sample (Coyne, 1997). During the interview process, the participants were encouraged to talk about their most worrisome concerns about cancer survivorship which were handled in a respectful and purely non-judgmental manner by the researcher. The sample size was determined based on informational needs and study requirement which was guided with the principle of adequacy to attain sufficient data collection to reach the data saturation, that is, sampling to the point at which no new information is obtained and redundancy is achieved (Polit & Beck, 2014).

Creswell (1998) and Morse (1994) have recommended some guidelines to develop the sample size ranging from 20 to 50. Polkinghorne (1989), recommended interviewing 5 to 10 participants for phenomenological research who have all experienced similar events (phenomenon) so the commonality of their experiences can be captured and interpreted. Wertz (2005) indicated that the number of participants is dependent on the research problem. The selection of these participants should reflect and represent the homogeneity that exists among the participants' sample pool as it provides a better gauge and a 'better understanding' of the overall perceptions among the participants' 'lived experiences' (Alase, 2017). Stebbins (2001) described the conventional approach for determining sample size, theoretical saturation, as the point at which no new information is produced from additional cases. In qualitative research, the data is collected until data saturation has occurred.

Glaser, et. al. (1968), also suggested the concept of saturation for accomplishing an appropriate sample size in qualitative studies. Guest, et. al., (2006) proposed the use of the grounded theory concept of the theoretical situation. Saturation, although, is commonly considered as the "gold standard" for determining sample size yet it might not be the optimal indication of an adequate sample. There are no definitive criteria or rules for the sample size needed to reach data saturation. Additionally, saturation is varyingly defined or is sometimes undefined. There are various views on how many cases must be considered to achieve data saturation. The number



of respondents for data saturation depends on how homogeneous the sample is and on the qualitative design. The more homogeneous the sample, the fewer cases needed for data saturation. The more heterogeneous the sample, the more cases needed for data saturation.

These different views, thus, raise a number of problematic conceptual and methodological concerns (O'Reilly & Parker, 2013; Dey, 1999; Bowen, 2008). O'Reilly & Parker., (2013) argued the relevance of theoretical or thematic saturation beyond grounded theory methodology and claimed for more transparency in achieving saturation. Although Safman & Sobal (2004) quantified the need of less than 30 participants using in-depth interviews to obtain data saturation, yet the recommendation was challenged by the researchers who suggested that such recommendations reflect the researcher's personal experience and that data saturation is actually determined by a number of factors such as the quality of data collected, methods of data collection, study design and the nature of the enquiry (Ritchie, et. al., 2003; Sandelowski, 1995).

3.6.3 INTERVIEW GUIDE

The following were the key topics and questions in the interview guide and semi-structured interviews:

Part 1: Introduction

The following were described to the participants as part of the introduction of the study:

- Aim and objectives
- Personal (researcher's) interest in the study
- Contribution to the public or community

Part 2: Semi-structured Interview

- 1. Breast cancer occurrence:
 - a. Can you tell me what has happened since the moment you were diagnosed with breast cancer?



b. What were your thoughts and feelings when you were told you had breast cancer?

2. Breast cancer experience/Living with Cancer:

- a. What are your experiences with having breast cancer, its treatment, side effects and care that you received at the hospital?
- b. Tell me about the treatment choices and life changes you made after the treatment; how breast cancer impacted your life.

3. Utmost Needs or Concerns during the survivorship:

- a. Can you tell me what your main concerns or biggest challenges were; (prompt: focusing physical, information, social, emotional, spiritual or any other)
- b. How those have influenced your breast cancer experience? (Prompt: health, family, work, relationships).
- c. How did you manage to deal with them or from where you received the support?

Expectations about care and service delivery:

- d. What do you think and what elements of the service delivery and care at the hospital you were satisfied or not satisfied with?
- e. What change or improvement you would expect if possible or what kind of support you need? (Prompt: what comes to your mind, what do you want in terms of support that can make you feel better or to improve your experience).

3.6.4 DATA COLLECTION

Semi-structured face-to-face interviews

According to Smith, et. al., (2009) "A qualitative research interview is often described as 'a conversation with a purpose' which must be implicitly informed by the research question. The selected method of data collection was face to face semi-structured interviews because it facilitated the building of rapport, and the interviews allowed the opportunity to gather rich information in greater depth than the quantitative survey already used in this research. One-to-one interviews provide a powerful way to acquire insight into issues via gaining understanding of the experiences of people and makes the participant feel that their story is important



(Seidman, 1991). Denzin & Linclon (1998) affirmed that information gathered via a one-toone interview avenue of inquiry is one of the most powerful ways to understand individual experiences. In addition, face to face semi-structured interviews involve numerous predetermined topics to be discussed with each participant thus maintaining consistency. Moreover, the semi-structured interviews provide freedom to the participants to freely expand upon their answers. It is also worth mentioning the decision to choose individual interviews for the methodology rather than focus groups for the purpose to attain as much information from individuals without participants being influenced by ideas or perceptions of what other people hold or say.

In this phase, semi-structured interviews were considered as the more appropriate means of interviewing compared to unstructured interviews as it allows individuals to express their experiences, thoughts and attitudes related to a sensitive topic without group influence (Erdogan, 2014). It provides freedom to elicit new responses while maintaining the data consistency as it involves several predetermined topics to be discussed with each participant (Berg, 2009). Moreover, it allows 'gaining information on the perspectives, understanding, and meanings constructed by people regarding the events and experiences of their lives' (Grbich, 1999). The interview questions were based on the results from the quantitative study as well as the theoretical framework to explore cancer survivors' needs.

Interview Schedule

Each interview lasted for 30 to 45 minutes. Semi-structured interviews were recorded using a voice recorder and transcribed manually for analysis within a few days after the interview. At the beginning of each interview session, participants were welcomed and valued by expressing gratitude towards their participation. Some rapport was established with each participant to make them feel relaxed and comfortable. Following introductions, participants were told the purpose of the study. Efforts were made to establish trust and ease in order to obtain rich information from them about their concerns related to breast cancer. They were informed about the expected length of the interview and how the interviews would be recorded. Participants were reassured that they could repeat or clarify questions if needed and that there were no right or wrong answers. They were informed that they could take a break or stop at any time if needed. They were encouraged to ask any preliminary questions they might have before the



start of the interview. The aims and objectives were clarified with a provision of brief description of the research, and it was explained why their involvement is needed and how their insights would add significance that could lead to improved healthcare services. Privacy was assured throughout the interviews to enable participants to talk freely about their concerns without any disruption (Berg, 2009).

Data Collection Procedures

A total of ten interviews were conducted and were audio-recorded. Each interview was assigned a unique code, for example "Participant 1, 6th May 2020". If more than one interview took place on the same date, the interviews were identified by an assigned number that represents a unique participant for instance, "Participant 2, 18th May 2020".

Prior to each interview, written consent was obtained from each participant. Each interview began with a general, open question for instance, "Can you tell me what has happened since the moment you were diagnosed with breast cancer?" which was followed by specific questions to elicit particular concerns about breast cancer such as, "What are your experiences with having breast cancer, its treatment, side effects and care that you received at the hospital?" General questions were used to allow the participants to give as much detail as they would like about their unique experiences without restricting their responses by using questions that would be too specific or limiting. Follow-up probes such as "in what ways? or "why do you feel that way? - could you tell me more about this?" were used to encourage them to elaborate more on their responses. Such probing prompts deeper responses from the participants and adds richness to the data.

A semi-structured **interview topic guide** was used as a layout to probe further on issues with regard to the objectives of the study in order to help direct the conversation toward the topics and issues highlighted in the phase one and were relevant to the research questions. **The interview guide was constructed on the bases of the items in the questionnaire which are considered less deeply explored in the scientific literature or could lead to multiple interpretations by the patients, thus requiring further understanding. To ensure that the topic areas would be reflective of the participants' experience as well as being appropriate and understandable, the questions were pre-tested with two breast cancer patients and two oncologists. Pre-testing allowed refinement of the interview questions based on their current**



practice as well as permitted direct comparisons between participants' answers providing them the freedom to speak openly about significant aspects of particular questions. A few changes were made according to the feedback received from both oncologists and the patients, mainly to modify the language or tone of the questions as well as to consider probing questions if needed. For example, based on oncologist' suggestions and to broaden up the survivors' view to allow survivors to speak more broadly about their perceptions instead of asking "can you tell me what is happening to you due to your illness", the question was modified as "can you tell me what has happened since the moment you were diagnosed with breast cancer". Another example is, the question which was originally worded "What are your experiences with having breast cancer" was modified based on patients' suggestions to cover all aspects and to highlight treatment as being the most challenging and life changing aspect of having cancer to: "What are your experiences with having breast cancer, its treatment, side effects and care that you received at the hospital? "Tell me about the treatment choices and life changes you made after the treatment; how breast cancer impacted your life".

Participants had the freedom to express their views in their own words, and to raise any issues of concern. However, patients were also allowed to raise any issues of concern. Use of this interview guide allowed direct comparisons between participants' answers while simultaneously permitting their' freedom of response and giving them the ability to speak to personally significant aspects of particular questions (Rose, 1994).

The interview process was engaging for both researcher and participants in that the researcher was actively listening to what the participants were saying and asked follow-up questions, including for clarification of any points where needed. All questions in the semi-structured interviews were organised systematically; however, not necessarily were asked in the same sequence during the interview. The questioning sequence of each interview was adjusted based on the participant's responses. Also, not all the questions from the interview guide were asked. The number of questions and duration of interviews varied among each participant as there was an interchange of views between two persons about a theme of mutual interest. In addition, initial questions were also modified in the light of participants' responses to enquire about any related relevant experiences.



The questions were open-ended, encouraging participants to fully describe their experiences according to their own perceptions and in their own words, and follow-up questions used to ensure there was no risk of loss of meaning because of imposing a standard way of asking questions. Greater latitude was allowed as some interviews moved away from the planned questions and entered an area that was not predicted or unprompted to provide maximum autonomy to the participants to express their concerns. However, the researcher was careful not to let the conversation stray too much for the interview questions so as not to include irrelevant material. Questions were asked one at a time, and minimal probes were used. Most patients talked about these topics without needing much probing, but probing was used based on careful listening to what the participants appeared uncomfortable with a particular line of questioning, another attempt was made to ask the questions more gently or to back off this line of questioning altogether. The questions were "directed to participants' experience, feelings, concerns or needs and beliefs about the theme in question (Welman & Kruger, 1999).

The interviews were conducted until the data reached saturation, where there was no new information contributed during the interview based on the questions and domains mentioned in the interview guide (Guest, et. al., 2006). In order to minimize the participants' vulnerability, the researcher avoided confirmatory gestures or judgmental expressions.

3.7 QUALITATIVE DATA ANALYSIS METHODS – AN OVERVIEW

Grbich (2007) argued that there is diversity on how to process the preliminary data analysis and that it is an individual process based on the research and the researcher. In this study, **thematic analysis** was employed which is described as **a "method for identifying, analysing and reporting patterns (themes) within data"** (Braun & Clarke, 2006). Thematic analysis has been widely used in qualitative research (Braun & Clarke, 2006), although there is no clear consensus about what thematic analysis is and how you go about it (Tuckett, 2005; Attride-Stirling, 2001). Braun & Clarke (2006) argued that thematic analysis should be a foundational method for qualitative analysis and should be considered as a standalone method (Braun & Clarke, 2006; King, 2004; Thorne, 2000). Since it provides core skills to conduct many other forms of qualitative analysis and is used by many qualitative methods thus many authors argued



that it should not be considered as a separate method and should be used to assist researchers in analysis (Holloway & Todres, 2003; Ryan & Bernard, 2000; Boyatzis, 1998).

Content analysis is another method that can be used to identify patterns across the qualitative data; however, it tends to focus more at micro level and often provides (frequency) counts (Wilkinson, 2000) and allows for quantitative analyses of initially qualitative data (Ryan & Bernard, 2000). Boyatzis (1998) suggested that thematic analysis can be used to transform qualitative data into quantitative form and be subjected to statistical analyses. However, in thematic analysis themes tend not to be quantified, (Braun & Clarke, 2006).

Both **interpretative phenomenological analysis (IPA) and grounded theory** seek patterns in the data but are theoretically bounded. IPA's theoretical underpinnings stem from the phenomenology which originated with Husserl's (1931) attempts to construct a philosophical science of consciousness, with hermeneutics (Biggerstaff & Thompson, 2008) which allows deeper insight of people's everyday experience of reality to gain an understanding of the phenomenon in question (McLeod, 2001). Likewise, the goal of a grounded theory analysis is to generate a theory of the phenomenon which is grounded in the data (McLeod, 2001). On the contrary, in thematic analysis, there are no theoretical commitments of grounded theory to be subscribed by the researcher if they do not intend to produce grounded theory analysis (Braun & Clarke, 2006).

On the other hand, the term, **thematic discourse analysis**, is used to refer to a wide-range of pattern-type analysis of data encompassing thematic analysis within social constructionist epistemology to form analysis similar to the interpretative repertoire form of discourse analysis (Clarke, 2005). Whereas **thematic decomposition analysis** is a form of thematic discourse analysis which identifies patterns within data and theorises language as constitutive of meaning as social (Braun & Clarke, 2006).

This research was conducted using **realist** research paradigm in a subjective manner to examine the experiences, needs, meanings and the reality of breast cancer survivors since there is a direct relationship between experiences and meanings and language. (Potter & Wetherell, 1987; Widdicombe & Wooffitt, 1995).



3.7.1 RATIONALE FOR USING THEMATIC ANALYSIS

Given the scarcity of research conducted in the UAE on cancer survivors' psychosocial needs and lack of evidence based psychosocial interventions in this region, Braun & Clarke's (2006) thematic analysis seemed to be most appropriate method for this qualitative research since this approach allows for a deeper understanding of participants' lives and provides the flexibility allowing for complex analysis without dependence on theory that is required given the exploratory nature of this study. Additionally, by gathering 'deep' information and perceptions through the inductive method, the approach valued the phenomenological viewpoint of how participants perceived and expressed their needs.

The rationale of using an inductive thematic analysis of latent information was, it was open to participant's experiences, rather than seeking views on themes informed by the literature. This helped in avoiding assumptions and biases in the literature being perpetuated (Braun & Clarkes, 2006). It also limited personal influence and pre-existing belief as a researcher that added true knowledge about the individuals to this study. However, this does not imply that one can completely get rid of his/her theoretical "epistemological responsibilities". It is for this reason that inductive thematic analysis was believed to be an appropriate method of analysis.

This data driven approach was also beneficial as it allowed identification of unexpected themes in the data rather than themes that were developed based on prior theories or preconceptions. Additionally, literature shows that several studies have employed the Braun and Clarke's (2006) method of thematic analysis to explore cancer survivors' needs in other settings (Naidoo, et. al., 2020; Wan-Tzu, Patsy & Raymond., 2018; Dsouza, et. al., 2018; Mazanec, et. al., 2017; Ness, et. al., 2013; Maura, Heddon & Griffis., 2010).

Another reason to use thematic analysis, was its breadth of scope since it can be broader and include social and cultural phenomenon as well. Additionally, it allows greater flexibility with regard to theoretical framework and with the use of analytical tools.

Succinctly, thematic analysis was deemed appropriate, as the purpose of the analysis was to examine the ideas, issues and thoughts regarding psychosocial needs mentioned by the participants.



3.7.2 THEMATIC ANALYSIS – PROCESS OVERVIEW

Thematic analysis is defined broadly as "a way of seeing" and "making sense out of seemingly unrelated material" (Boyatzis, 1998). Braun, Clarke & Terry (2015), described the procedure as a personal and even emotional experience. In this research, the process of applied thematic analysis was organic and reflexive that involved researcher's engagement and intuition to be part of the analysis. It helped researchers establish what cancer survivors consider important, how they perceived their needs and cancer experience and categorized them, and how various categories are associated with each other (Herzog, Handke & Hitters, 2019).

According to Braun & Clarke (2006), thematic analysis provides a highly flexible approach due to its theoretical freedom and can be modified as needed, providing rich and detailed yet complex data. It is a useful method to examine the participants' perspectives, highlighting similarities and differences to generate deeper understanding and insight. Whereas interpretative phenomenological analysis or discourse analysis provides a more accessible form of analysis (Braun & Clarke, 2006) and makes it easier for researchers to employ the procedure with easy grasped understanding and learning.

Thematic analysis can be conducted within several other research paradigms yet the focus to conduct the analysis would be different for different paradigms. Thematic analysis can also use a **constructionist** paradigm that assumes that meaning and experiences are socially developed and not due to an individual perspective (Burr, 1995). Therefore, the thematic analysis with the constructionist framework leans more towards the socio-cultural phenomena and structural context rather than subjective factors that arise from the individual data (Majumdar, 2019). Thematic analysis can also be a **contextualist** method characterised by theories such as critical realism (Willig, 1999), which acknowledge the ways individuals make meaning of their experiences and the ways the society impinges on those meanings while maintaining focus on the material and the other limits of "reality" (Braun & Clarke, 2006). It is important to make clear the theoretical position of a thematic analysis.

For the analysis purpose, the recorded texts were assumed as the data and codes were developed by closely examining the texts as salient themes that were generated inductively from the texts.



The essence-capturing codes consisted of words as well as short phrases (Saldaña, 2016) which were modified as needed flexibly throughout the analysis (King, 2004).

The process of thematic analysis involved searching across a dataset a repeated pattern of meaning. Thematic analysis proceeded by first identifying codes then themes and eventually patterns in qualitative data. Themes captured something important related to the research topic and represented a sort of patterned response within the dataset. What was important to understand was the judgment about the theme as some were given considerable space in some data items and little or none in others or some appeared relatively little of the dataset. Thus, the flexibility of thematic analysis allowed determining themes in a number of ways. Considerable attention was given to the prevalence of themes so a rich thematic description of the entire dataset could be attained in order to derive the important themes which accurately reflect the entire dataset. Coding and theme development were driven by the goal of retaining considerable details in the data items in a way to interpret credible meanings from the complex data.

Themes within the data were identified through an **inductive (bottom up)** approach (Braun & Clarke, 2006) which were associated with the data itself and were not driven by the researcher's theoretical interest. An **inductive approach** is a process of coding of the data without trying to fit it into a pre-existing preconception; thus, this form of thematic analysis is data-driven. However, it must be noted that researchers cannot detach themselves of their theoretical or epistemological commitments and data are not coded in an epistemological vacuum (Braun & Clarke, 2006). Herzog, Handke & Hitters, (2019) supported the same notion that in thematic analysis, verbal expressions are analysed without relying on any pre-existing theoretical concepts and instead the interpretations reflect the judgment of the researcher in each individual case. The analysis identified latent (interpretative) themes which provide more detailed and nuanced analysis across the whole data set, moving beyond the semantic (surface) level of analysis to identify the implicit meaning of what the survivors expressed and underlying beliefs.

The latent inductive analysis was conducted where the identified themes reflected the surface of the meanings of the data. This includes describing the data, summarising it followed by interpreting the data in relation to the broader meanings and implications (Patton, 1990).



Boyatzis (1998) explained that all research aims at developing theory; thus, a phenomenon can be learned by drawing knowledge from the previous studies to establish something new. Boyatzis (1998), pointed out that 'you do not have to reinvent the wheel every time you want to get somewhere.

Phases to conduct Thematic Analysis – Process Demonstration

To conduct the process of thematic analysis, Braun & Clarke's (2006) six-phased linear method is demonstrated below (Figure, 12) although there were steps which were iterative and reflexive that developed over time (Ely, et. al., 1997). The whole process involved a constant moving back and forth as needed, throughout the phases (Nowell, et. al., 2017).

Phase 1: Creating transcripts from audio recordings

The data was collected by interactive semi-structured face to face interviews through audio recordings. Data was analysed with some knowledge which was gained from quantitative results and some analytic interests of interview questions (Tuckett, 2005).

Analysis of the data began after all interviews were transcribed manually with the aim to enhance the quality of the information gathered in each interview. Since transcription is a representational process immersed in the subjective interpretation of the researcher (Bucholtz, 2000), each transcript was printed and reviewed and reread several times for the actual representation of interviews, their accuracy and understanding of the correct wording and to get familiar with the data and expressions of the participants about their concerns with regards to breast cancer.

Audio recordings were carefully heard and thoroughly understood to attain deeper familiarity with the depth and breadth of the content of data as well as to creating thick descriptions from them. Afterwards, the entire dataset was read repeatedly to search meanings, patterns and other key words or phrases, to draw ideas and identification of possible patterns (Brain & Clarke, 2006). Repeated reading of thick descriptions was time consuming but was performed to build the foundation for the rest of the analysis.

During this phase, the entire dataset was broken down to read through to be actively engaged with it by searching for the patterns of meaning. Parallelly, data analysis was also started while



searching for patterns of meanings by linking and making notes of statements of cancer survivors' interviews. Since thematic analysis is applied in relation to a dataset in a written form, the transcripts were created during this phase. The verbal texts from interviews were transformed into written texts in a manner that retains the true nature of key information from the verbal account (Braun & Clarke, 2006). Transcribing the verbal texts was quite a time-consuming activity, however meticulously performed as it informed the early stages of analysis and continuous further understanding of data by checking back against the original audio recording for accuracy (Braun & Clarke, 2006). Moreover, all ideas for coding or searching patterns were documented to revisit in subsequent phases if needed to develop a more formal coding process which continued throughout the entire analysis (Braun & Clarke, 2012; 2006).

Since language is the fundamental tool for creating and validating knowledge in qualitative research (Taylor & Francis; 2013), all interviews were translated into English in case conducted in Arabic. Most of the patients were from Arabic speaking backgrounds regardless of different nationalities and most of them were good at speaking in English; non-Arabic people like Filipinos were interviewed in English for most of the communication. The interviews of Arabic speaking participants were translated by the Arabic speaking physicians who helped to maintain the original meaning of the text as described by the participants to check the accuracy of translated statements. The translators were further involved in the data analysis as well to reduce the chance of translational and conceptual assumptions, improve the clarity of meaning and understand how the breast cancer survivors described their concerns. Significant time was allocated during the data analysis process to identify the meanings of words in Arabic and also to understand the meaning of the data collected. All efforts were made to quote the participants during the transcription process to avoid any bias yet at some stages, participants' view was interpreted for the better understanding; for instance, some unfamiliar words were rephrased to gain complete sense of the statements which were rechecked by participants themselves to maintain the accuracy.

Participants were contacted as needed to fill the missing gaps found during the process of data analysis as well as to ensure the accuracy of information (Stuckey, 2014) provided during the interviews. This allowed co-construction of the knowledge by providing participants with the opportunity to engage with and add to interview and interpret the data (Birt, et. al., 2016). All efforts were made to conduct the process of transcriptions not necessarily with neutrality but



responsibility (Bucholtz, 2000). The key challenges associated with only a few transcriptions include the volume and the length of the interview recordings, the strong Arabic accent, incomplete sentences or omissions (Bailey, 2008; McLellan, et. al., 2003) which were dealt meticulously by member checking as described above.

Phase 2: Generation of initial codes

Savage (2000) explained qualitative coding as a process of reflection and a way of interacting with and thinking about data. The second phase involved the production of initial codes after reading and being familiarised with the data, having an initial list of ideas of what is in the data and what is interesting about them.

The entire dataset was organised into meaningful groups and codes were manually assigned to the data segments that provided the information from the raw data in a meaningful way (Kelle, 2004; Seale, 2000; Boyatzis, 1998). First, each transcript was read thoroughly several times and notes were taken if needed to familiarise with the data. During reading and re-reading, aspects of the data which were found interesting were noted down. This was performed manually on the transcripts. Line by line analysis of data was conducted and identified statements were highlighted and initial data-derived codes were generated at latent level to present participants' experience in a more detailed way. The latent codes provided the opportunity to bring new identified data-driven codes. Although quite time consuming, this ensured that nothing was missed out while performing the coding process. During this phase, A template was created to justify the inclusion of each code to assist in interpretation and clear evidence to add credibility to the research (King, 2004) and to enable the reader to judge whether the final outcome is rooted in the data generated (Ryan, et. al., 2007). Data was coded based on notes that were written and highlighted to indicate potential themes.

Following coding, the entire data was collated together within each code that formed the basis of themes (Braun & Clarke, 2006). Themes were created from the data itself without having any presumption in mind.



Phase 3: Searching for themes

A theme is an abstract entity that brings meanings and identity to a recurrent experience and its variant manifestations (DeSantis & Ugarriza, 2000). A theme captures something important in relation to the overall research question (Braun & Clarke, 2006), and unifies the nature or basis of experience into a meaningful whole (Nowell, et. al., 2017). Phase 3 begins after coding and collation of data with a creation of a list of different identified codes across the data.

This process involved bringing together ideas or experiences which may seem meaningless if viewed alone (Aronson, 1994). Upon producing the list of the codes, they were sorted out into initial themes. Themes were identified as salient by participants' responses based on something important to them as well as perceived frequent if the number of participants raised the similar theme not on the basis of frequency of statements made. This phase involved sorting and collating all potentially relevant coded data extracts into themes (Braun & Clarke, 2006). During this phase, codes were first analysed by organising and visualising them to conceptualise patterns and relationships within the data (Attride-Stirling, 2001).

Each code was written with a description and mind-maps were created to form an overarching theme. Mapping helped to develop an understanding of the significance of the themes (Herzog, Handke & Hitters, 2019). Some initial codes formed main themes whereas some were assigned as sub-themes, and some were discarded. Themes were generated inductively from the raw data (Boyatzis, 1998) thus were strongly linked to the data themselves (Nowell, et. al., 2017). Data coding was performed without trying to fit it into pre-existing ideas or preconceptions (Braun & Clarke, 2006).

Themes were created with a good number of predefined codes to help guide analysis. Most relevant themes were identified in relation with the research questions (King, 2004). Detailed notes were taken about the development and theme creation that helped establish confirmability in the audit trail (Halpren, 1983). The phase ended up with the collection of relevant themes and sub-themes and coding of all data extracts without leaving any area of data neglected.



Phase 4: Reviewing of themes

The fourth phase began after devising a set of themes to refine them more (Braun & Clarke, 2006). This phase involves two levels: reviewing and refining themes by developing candidate themes into final themes. During this phase, the entire data was re-read to assure whether the themes and sub-themes are coherent with the dataset and accurately represent that data. Also, to code any additional data within themes that might have been missed during the initial coding since recoding was considered ongoing (Braun & Clarke, 2006).

At level one, coded data extracts were reviewed for each theme to consider the coherent pattern. Some themes didn't have enough data to support them, some candidate themes were not really themes, some collapsed into each other, and some were broken down into separate themes (Braun & Clarke, 2006).

After achieving the coherent patterns, the second level involved refining of identified themes that are relevant to the data extract. The potential themes were aggregated to achieve a more manageable set of significant themes. (Attride-Stirling, 2001). Data within themes were cohered together in a way that clear and identifiable distinctions could be made between themes keeping internal homogeneity and external homogeneity (Patton, 1990). In the second level of analysis, the validity of individual themes was assessed to determine whether the themes accurately reflect the accurate representation of the dataset as whole (Braun & Clarke, 2006). In this phase, inadequacies in the initial coding and themes were identified and amendments were made accordingly (King, 2004).

After rereading, recoding, possibilities of identifying new themes were explored to generate any possible themes till the refinements added nothing substantial, the process was halted. At the end of the process, all transcripts were compared separately to identify significant statements that represent similar findings. Likewise, differences were collated across each interview to draw further conclusions based on that. Referential adequacy was tested by trailing back to the raw data and comparing it to the developed themes to make sure all conclusions were firmly grounded in the data (Lincoln & Guba, 1985).



Phase 5: Defining and Naming Themes

In this phase, a thematic map was developed. At this point themes were defined and redefined by identifying the essence of what they are about and analysed to determine which aspect of the data is captured by each theme (Braun & Clarke, 2006). Themes were reviewed, refined and defined with a concise name and theme were considered in relationship with each other in a 'thematic map' (Fig, 17). The analysis appeared to reach thematic saturation at a point when no additional themes were found from the reviewing of successive data regarding each investigated category (Guest, Bunce, & Johnson, 2006, Glaser & Strauss, 1967).

A detailed analysis for each individual theme was conducted to identify the story that each theme tells in relation to the research question(s). Part of the refinement was to identify subthemes which are essentially themes within themes which are useful to structure large and complex data and to demonstrate the hierarchy of meaning within the data. Themes were not finalised until all the texts were included which were relevant to the research questions, added to that, investing sufficient time to develop the themes increases the probability of developing credible findings.

At the end of this phase, themes were clearly identified. All themes were given concise names to the identified themes to provide a quick sense about what the theme is all about.

All stages of the analysis were peer reviewed and participants' statements were subject to rechecking by a reviewer as well as with the relevant participant as suggested by Robson (2002). The rechecking with the participant was carried out either by phone or voice notes as preferred by the participant. Finally, the original interview transcripts were re-read several times to ensure the final themes were supported by the original data. Although the stages used in the analysis of the data seemed sequential, they were iterative and built on the previous stage, making the analysis typically a recursive process with a back-and-forth movement between different phases (Braun & Clarke, 2006).

Phase 6: Producing the Report

The final phase was dedicated to the writing up of the analysis. It began after the establishment of a set of fully worked-out themes for final analysis and write-up of the report (Braun &



Clarke, 2006). Findings were developed in a way that is accessible to a reader in a way that claims made in relation to the dataset are rendered credible and believable. All methodological notes were taken to the audit trail to add trustworthiness and credibility (Halpren, 1983).

A reflexive journal was maintained throughout the research to ensure findings and conclusions were interpreted in a trustworthy manner and whether the literature is in line with the conclusion or not (Polit & Beck, 2008; Halpren, 1983). The aim was to articulate the meanings and implications of each theme as well as the assumptions that underpin it (Braun & Clarke, 2006). Also, to build a valid argument for themes selection and to add credibility to the discussion by including all relevant whether unanticipated outcomes that did not correspond to the research (Côté & Turgeon, 2005). The final results were aimed to be shared with the participants for their feedback to allow linking between respondents' views and representations (Tobin & Begley, 2004; Lincoln & Guba, 1985). The entire process of thematic analysis was recursive during which the writing up and analysis were executed parallelly in a way where assumptions were applied congruently with conceptualization of the research questions (Reicher & Taylor, 2005).





Figure 12: Phases of thematic analysis (adapted

from Braun & Clark's 2006)

 $https://www.researchgate.net/figure/Phases-of-Thematic-Analysis-Adapted-from-Braun-Clarke-2006-p-87_fig1_275952540$

3.7.3 TRUSTWORTHINESS AND REFLEXIVITY

Conducting a naturalistic inquiry is not a straightforward task. Qualitative studies are complex in many ways compared to quantitative research which follows a structured and rigid design. Apparently, it seems like there are many similarities among various qualitative methods and the nature of qualitative enquiry is subjective (Creswell, 2013; Polit & Beck, 2012; Ryan, et. al., 2007). Qualitative research has tried to address methodological concerns by offering



guidelines for trustworthiness and reflexivity in order to assure that the method is rigorous (Cypress, 2017). Rigor was addressed for the first time by Lincoln and Guba (1985) in their model of trustworthiness of qualitative research.

Trustworthiness is used as the central concept in their framework to appraise the rigor of a qualitative study. Trustworthiness has been applied to quality, authenticity and truthfulness of findings and can be described as the degree of trust described in many different ways by the researchers. It relates to the degree of trust or confidence that readers have in results (Schmidt & Brown, 2015). Trustworthiness is used as a criterion to evaluate the quality of a research design (Yin, 1994) and addresses methods that ensure that the research process has been carried out correctly (Guba & Lincoln, 1989). Trustworthiness is based on the concepts of reliability and validity and objectivity (Manning, 1997; Seale, 1999). Lincoln & Guba (1985) clearly established four criteria as benchmarks for quality based on the identification of four aspects of trustworthiness that are relevant to both quantitative and qualitative studies, which are *truth value, applicability, consistency* and *neutrality*. In this study, the concept of trustworthy in terms of **credibility, transferability, dependability and confirmability** was used to determine the validity and reliability of study findings (Lincoln & Guba's, 1985).

Credibility refers to the "fit" between participants' views and the researcher's representation of them (Tobin & Begley, 2004). It is defined as the truth of the study findings including the structural description of the phenomenon being studied (Anney, 2014; Ryan, et. al., 2007). Credibility was achieved by spending enough time and prolonged engagement with participants to gain deeper understanding of participants' responses, establishing rapport and trust with them. Also interviewing participants until data saturation was reached (Anney, 2014; Polit & Beck, 2012; Baxter & Eyles, 1997). Other ways to achieve credibility included **data triangulation** which was achieved by cross-checking the data and interpretations of each area described by the participants, by two other researchers who were experts in qualitative research methods. Moreover, **purposive sampling** for the qualitative part of the study (different age, stage, survival and treatment), peer review for an **external check** on the research process as well as **referential adequacy** to check preliminary findings against the raw data were added strategies to add credibility to the current study. This was carried out during the dissertation while writing up the findings and final analysis (Baxter & Eyles, 1997).



"Memoing" is an important data source in qualitative research that was considered while conducting interviews (Miles & Huberman, 1984). According to Lofland & Lofland, (1999) field notes (memoing) which are the secondary data storage method in qualitative research, are crucial to retain data gathered as the human mind tends to forget quickly. In this study, field notes were taken upon hearing, seeing, experiencing or thinking in the course of collecting and reflecting on the process as it is crucial to be aware of descriptive notes and reflective notes such as presentiments, impressions, expressions or feelings etc. Morgan (1997) suggested that since field notes involve interpretation thus, they are part of data analysis rather than the data collection or a step toward data analysis. Field notes were taken to trail back to the raw data. Findings were also shared with the participants to check the accuracy of interpretations to add further credibility to the study (Baxter & Eyles, 1997; Lincoln & Guba, 1985). It was ensured that personal bias did not interfere with the study at any stage thus notes taking was considered during data collection and data analysis to examine and avoid personal engagement. Field notes were made to capture non-verbal expressions and behaviours observed during interviews. This was an essential step that contributed to the richness of data and to interpret verbal information (Berg, 2009).

After making sense of the data, all efforts were made to identify and illuminate emerging themes and descriptions. Credibility was finally sought by coordinating best the **methodological and analytical materials utilised** in the entire research.

Transferability refers to the generalisability of inquiry. In qualitative research, this concerns only case-to-case transfer (Tobin & Begley, 2004). In this research, transferability was achieved by the **purposive sampling** recruiting participants from different emirates and with different nationalities (Anney, 2014; Creswell, 2013; Polit & Beck, 2012) and **data saturation** to obtain rich with a wide possible range of information and thick description of the phenomenon who had experienced it (Creswell, 2013; Polit & Beck, 2012). Data was collected from a single source and thus could result in lack of transferability; however, this was overcome by obtaining detailed descriptions of the participants' experience of the phenomenon in an attempt to know (Baxter & Eyles, 1997). Special consideration was given to the process of data collection, data identification and data analysis throughout the course of the study. The audio records were carefully transcribed after numerous rereading for future scrutiny and all



attempts were made to document all aspects of analysis in a way to make sense of the data and to write the final draft as accurately as possible (Merriam, 1995).

Dependability is the degree to which the study findings could be stabilised over time. It was established by an audit trail and data triangulation which involve recording of all the original data, reviewing and re-reading all transcripts, field notes, audio recordings and keeping notes, feedbacks and comments from the assessors to ensure the research process is logical, traceable and clearly documented (Anney, 2014; Tobin & Begley, 2004; Baxter & Eyles, 1997). According to Lincoln & Guba (1985), readers are better able to judge the dependability of the research if they can examine the research process. In this study dependability was established by an audit trail (Koch, 1994) as well as data triangulation which includes documenting and reviewing the original data, maintaining and reviewing all transcripts, field notes and audio recordings, refining the work upon feedbacks and comments from the supervisors and assessors who were expert in thematic analysis (Anney, 2014). All ten interviews were transcribed and then reviewed and correctly translated to **ensure accuracy** of documentation. Transcripts were repeatedly reviewed and reread to determine the meaning units (themes and subthemes) to understand the deeper meanings. Finally, the findings of thematic analysis were reviewed and assessed by two qualitative researchers experienced in thematic analysis from the UAE University Tawam Campus in the audit trail confirming that all possible codes and themes were identified, ensuring accuracy of the interpretations (Mays & Pope, 1995). Further discussions were carried out to discuss the datasets and to resolve discrepancies such as whether to consider treatment effects such as hair loss, skin or weight changes under "physical concerns" or "emotional concerns". Finally, full alignment was obtained by ironing out the discrepancies to achieve unified understanding. Findings were also shared with the participants to validate the results over the telephone (member checking) in order to check whether the findings were reported true or not (Ziebland & McPherson, 2006).

Confirmability is the degree to which the study findings can be confirmed by the other researchers (Baxter & Eyles, 1997). It is concerned with establishing that the researchers' interpretations are clearly derived from the data. According to Guba & Lincoln (1989), confirmability is established when credibility, transferability and dependability are all achieved. In this study, **conclusions and interpretations** were clearly demonstrated to establish confirmability (Tobin & Begley, 2004). In addition, the reasons for theoretical,



methodological and analytical choices were highlighted throughout the study with the aim to build understanding to other readers about how and why decisions were made (Nowell, et. al., 2017).

Reflexivity is central to the audit trail; (Nowell, et. al., 2017). Tobin & Begley (2004) suggested that researchers should keep a self-critical account of the research process, including internal and external dialogues. This can be used to document the methodological decision, rationales and to record researcher's **personal reflections** of their values, interest and insight information about self (Lincoln & Guba, 1985).

In this study, all internal and external communication was maintained to self-evaluate the research process (Tobin & Begley, 2004). A reflexive journal was used to record the daily documentation of the research, methodological decisions and reasons to record personal reflections, interests and insights information about self (Lincoln & Guba, 1985). In addition, the research process and the particular epistemological stance was examined to figure out if it might have influenced any aspect of data analysis (Willig, 2008). For instance, having a certain stance on UAE culture and women's status, it was presumed that women might not be able to talk openly about cancer and share their views about cancer experience yet, all women encouraged the researcher that their voices were heard for the first time. It was also discovered during qualitative interviews that UAE men were supportive toward their families who suffered from cancer. Most Emirati women showed gratitude towards their spouse and families for the continuous support and mentioned religion as the highest source of strength and coping with illness which falsified the set notion and pre-conceived ideas about the UAE culture, women's modesty and the dominant role of Muslim men in decision making that might be considered hindering in seeking treatment for breast cancer or resulting in fears as presented in the existing literature (ElObaid, 2016, 2014; Donnnelly, et. al., 2013; Bou Khalil., 2013; Daher, 2012; Bener, et. al., 2001; Milaat., 2000).

Eventually, the findings of the qualitative study were integrated with the quantitative study in order to address the research objectives. This chapter has provided an overview of the used methodology and methods pursued and applied to process and analyse the data. The next chapter provides the results and findings of both quantitative and qualitative phases of the study.



Summary

The methodology of the study was presented in this chapter. A sequential, explanatory mixedmethods (quantitative phase followed by qualitative phase) approach was used in the current study to identify the psychosocial needs of breast cancer survivors in the UAE. A crosssectional design was used in the quantitative phase, followed by a semi-structured interview in the qualitative phase. The procedures for recruitment of a sample, data collection methods and data analysis approaches were described in details.

In quantitative phase one, **Cancer Survivors Survey of Needs** was used to identify the psychosocial needs of breast cancer survivors. Census sampling was used with the aim to reflect the entire population under study. The data was collected on 205 patients utilising quantitative survey with 94% response rate. Data analysis approaches was described to mention the use of SPSS software to provide descriptive statistic of survivors' demographics and clinical characteristics, the significant mean differences between two groups (Emiratis and expatriates) using chi-square test and t-test and multivariate multiple regression to compare the differences in each domain of psychosocial needs of survivors.

In qualitative second phase, ten semi-structured interviews were conducted to gain in-depth understanding of the needs of breast cancer survivors who were selected through purposive subsampling from phase one. Data collection procedures were described including the creation of an interview guide which was constructed on the basis of items in the questionnaire to gain further understanding. The procedure of data analysis was explained which utilised Braun and Clarke's (2006) inductive thematic analysis methodology demonstrating its six phases. The chapter ended with the provision reflexivity in this research and the methods used to achieve trustworthiness which included transferability, dependability and confirmability, dependability and confirmability.



Chapter 4: RESULTS

Chapter four presents the result and overall finding of the study. This chapter is organised by presentation of the demographics of the sample along with the findings of phase one of the study followed by the demonstrating phase two findings that includes describing process and techniques used for coding, creation of categories and the development of themes. Further explanations of findings revealed from both phases and results were integrated with the aim to understand comprehensively the needs and concerns of breast cancer survivors which were obtained by analysis and integrating findings from both methods. The summary has also been provided at the end of each section to present a succinct view of both quantitative and qualitative findings.

This chapter consists of two major sections, that is section 4.1 and 4.2. The first section (4.1) presents the quantitative results of the study. The statistical analyses were performed using version 17.0 of SPSS. **Quantitative results have been organised as follows:**

- Descriptive data analysis
- Application of Chi-square test for independence to compare the socio-demographics and clinical characteristics of two groups (Emiratis and expatriates) for statistical significance.
- Significant mean differences were investigated between two groups.
- Application of t-test to explore any significant differences between nationality (Expats and Emiratis) and psychosocial concerns.
- **Pearson Correlation Coefficient** was used to measure the correlation between psychosocial needs and demographics factors (age & QOL) and
- **Multiple regression** testing to determine the significance of the effect of the independent variables on the dependent variables.

Section 4.2 provides qualitative results which were later integrated in order to obtain a complete and synergistic utilisation of the data than results that were achieved separately from quantitative and qualitative data collection and analysis.



4.1 DESCRIPTIVE DATA ANALYSIS

The section below includes a descriptive statistics analysis including **frequency distribution**, **means, standard deviation (SD), medians, and proportions** which were calculated for all socio-demographic variables.

Demographic and clinical variables included **age at diagnosis**, **nationality**, **year of cancer diagnosis**, **type of cancer**, **stage of cancer**, **type of treatment received and whether currently on treatment or not**, were also analysed.

With regards to **psychosocial concerns** of cancer survivors, **physical concerns, information or educational concerns, social & financial concerns, emotional concerns and spiritual concerns were rated by the patients from 0 to 5 (No concerns to extreme concerns)** which indicates the level of need using a Likert-type scale. This allowed quantifying the concerns in each domain of psychosocial needs. These factors assessed the critical need domains as perceived by the survivors.

Data was cleaned and manipulated in order to create the scores for each psychosocial domain for instance physical concern (20 items) describing the mean, minimum, maximum and the standard deviation.



4.1.1 DEMOGRAPHIC CHARACTERSTICS OF BREAST CANCER SURVIVORS

Nationality distribution of cancer survivors

A total of 205 women, 54 (26.3%) cancer survivors were UAE nationals and 151 (**73.7%**) were **Expatriates**. This is also in line with the current population as expatriates contribute to approximately 75% of the UAE population.

Age group of cancer survivors

The mean age of the women was 49 years (SD = 11.864, range = 22–86 years). The median age of breast cancer survivors was 44-54 years. There were 12 (6 %) women who developed breast cancer at an earlier age ranging from 22-32. The majority of the women (124 = 60 %) diagnosed with breast cancer were **between** the **33-54** age category. Alternatively, it could be concluded that 23.4 % of survivors were between 25-39 years, 64.9 % of them between 40-64 years. Only 20 (10 %) women were diagnosed with breast cancer who were above the 65 years age group.

Year of initial diagnosis of cancer survivors

Out of a total of 205 participants, 80 (39%) of the breast cancer survivor patients were initially diagnosed with cancer between 2012 to 2017. Whereas the highest number of breast cancer survivors (122, **59.5%)**, were diagnosed either in or after 2018. Only 3 breast cancer survivors were diagnosed in 2011 or earlier as shown in Table 1.

Stage at diagnosis of cancer survivors

The highest number of survivors (n=105, 51.2%) diagnosed with breast cancer stage II, regional without lymph node involvement followed by stage III, regional with lymph node involvement (n=48, 23.4%). It is evident that the majority of breast cancer survivors are diagnosed with regional stage cancer at the time of diagnosis. There were no survivors who presented with advanced disease at the time of diagnosis. Only 13 (6.3%) survivors were diagnosed with the in-situ stage, which denoted the lowest level of disease when there's no


cancer, only abnormal cells with the potential to become cancer. No woman out of 205 women presented with late advanced **stage IV** in the study sample.

Type of treatment received by cancer survivors

Majority of the expatriate women with breast cancer (n=78, 38%) received multimodality treatment with combination of surgery, radiation, chemotherapy, hormonal and other treatment whereas 26 women (12.7%) were treated with the combination of surgery, radiation and hormone therapy. Only 1 (0.5%) cancer survivor received chemotherapy followed by surgery. 6 women (2.9%) were given chemotherapy along with other treatment.

Type of treatment received by Nationality

Majority of the expatriate women with breast cancer (n=55, 36.4%) received multimodality treatment with combination of surgery, radiation, chemotherapy, hormonal and other treatment whereas the highest number of Emirati women (n=12, 22.2%) went for the combination of surgery, radiation and chemotherapy. Only 1 (1.9%) expatriate cancer survivors received chemotherapy followed by surgery, likewise only 1 (1.9%) Emirati woman had chemotherapy along with other treatment.

Overall quality of life rated by cancer survivors

Out of 205 participants 149 (72.2%) of breast cancer survivors described their quality of life during the past week of the study as good. The mean score for overall quality of life for both group Emirati and Expatriates was 9.40 (SD = 1.262) which represents a good quality of life. Over 86% of patients rated their quality of life as 9/10 on a scale of 1 to 10.

The table 1. below provides the summary results of breast cancer survivors' demographics and their clinical characteristics of breast cancer survivors.



 Table 1: Summary results of demographic analysis & clinical characteristics of breast

 cancer survivors

Breast Cancer Survivors Demographic Factors (n=205)		
DEMOGRAPHICS	n	%
NATIONALITY		
Emirati	54	26.3
Expatriate	151	73.7
AGE		
Mean= 48.84; SD=11.864; Min=22; Max=86		
22-32	12	5.9
33-43	64	31.2
44-54	64	31.2
55-65	45	22.0
Above 65	20	9.8
CLINICAL CHARACTERSTICS		
Year of diagnosis		
2000-2005	2	1.0
2006-2011	1	0.5
2012-2017	80	39.0
>=2018	122	59.5
Type of cancer		
Breast	205	100
Stage of cancer		
I	39	19.0
П	105	51.2
III	48	23.4
In-situ	13	6.3
Currently receiving treatment (Treatment Received)		
No	84	41.0
Yes	121	59.0
Overall OOL	Mean=9.40. SD=1.262	



4.1.2 BREAST CANCER SURVIVORS' (a) PREFERRED LEARNING SOURCES, (b) TOPICS OF INTEREST & (c) PRIMARY SOURCE OF STRENGTH (SECTION 9-11)

4.1.2.1a CANCER SURVIVORS' PREFERRED LEARNING SOURCES

Breast cancer survivors were asked to indicate their **preferred source of learning** about the **psychosocial concerns including:**

- 1. Physical
- 2. Social
- 3. Emotional
- 4. Spiritual.

The options included written materials, class, video, healthcare specialist, other cancer survivors or internet.

BREAST CANCER SURVIVORS' PREFERRED LEARNING SOURCES ABOUT PHYSICAL CONCERNS

The table 2. below presents breast cancer survivors' preferred learning sources about physical concerns. It shows that out of a total of 205 breast cancer survivors, almost **59%** breast cancer survivors opted to consult "only' the **healthcare specialist** for physical issues. **8%** of cancer survivors preferred learning through **videos and health specialists**, **7%** wished to learn through **video**, **3%** showed their interest to learn physical issues through other **cancer survivors** and another **3%** preferred learning through **written materials, class, video as well as through healthcare specialists**. Whereas, only 0.5% of survivors preferred the internet as a learning resource about their physical concerns.

Thus, it was concluded that the majority of the breast cancer survivors preferred to consult with and learn from the health specialist for their physical concerns whereas internet resources were the least favourable choice of cancer survivors.



Learning sources about physical concerns	Frequency	Percent
Cancer Survivor	6	2.9
Cancer Survivor, Written Materials, Class, Video, Healthcare Specialist	1	.5
Class	1	.5
Class, Healthcare Specialist	1	.5
Class, Healthcare Specialist, Cancer Survivor	1	.5
Class, Video	2	1.0
Class, Video Specialist	4	2.0
Healthcare Specialist	120	58.5
Healthcare Specialist, Cancer Survivor	3	1.5
Healthcare Specialist, Class	2	1.0
Healthcare Specialist, Video	8	3.9
Healthcare Specialist, Video Survivor	1	.5
Healthcare Specialist, Video Materials	1	.5
Healthcare Specialist, Written Materials	1	.5
Healthcare Specialist, Written Materials, Video	1	.5
Video	14	6.8
Video, Healthcare Specialist	16	7.8
Video, Healthcare Video, Healthcare Materials	1	.5
Video, Internet	1	.5
Written Materials	2	1.0
Written Materials, Class	1	.5
Written Materials, Class Specialist	1	.5
Written Materials, Class Specialist, Video	1	.5
Written Specialist, Video	1	.5
Written Materials, Class, Video, Healthcare Specialist	6	2.9
Written Materials, Class, Video, Healthcare Specialist, Cancer Survivor	3	1.5
Written Materials, Healthcare Specialist	4	2.0
Written Materials, Healthcare Specialist, Cancer Survivor	1	.5
Total	205	100.0

Table 2: Breast cancer survivors' preferred learning sources about physical concerns



BREAST CANCER SURVIVORS' PREFERRED LEARNING SOURCES ABOUT EMOTIONAL CONCERNS

The table 3. below presents breast cancer survivors' preferred learning sources about social concerns. It shows that out of a total of 205 breast cancer survivors, almost **68%** breast cancer survivors opted to consult the **healthcare specialist only** whereas 7% preferred learning through **videos only** for their social issues whereas **7%** of cancer survivors preferred learning through video and healthcare specialist both. 3% preferred to learn from other cancer survivors. Only 2% wished to learn through **healthcare specialists and other cancer survivors** and another 2% preferred learning about their social concerns through **healthcare specialists and video**.

Thus, it was concluded that the majority of the breast cancer survivors preferred to consult with and learn from the health specialist as well as from videos for their social concerns whereas using the internet to learn emotional concerns was the least favourable choice among those patients.

Table 3: Bre	east cancer sur	vivors' pre	ferred learr	ning source a	about social	concerns

Learning sources about social concerns	Frequency	Percent
Cancer Survivor	6	2.9
Cancer Survivor, Written Materials, Class, Video, Healthcare Specialist	1	.5
Class	1	.5
Class, Cancer Survivor	1	.5
Class, Healthcare Specialist, Cancer Survivor	1	.5
Class, Video	1	.5
Class, Video Specialist	2	1.0
Healthcare Specialist	139	67.8
Healthcare Specialist, Cancer Survivor	4	2.0
Healthcare Specialist, Cancer	1	.5
Healthcare Specialist, Video	4	2.0
Healthcare Specialist, Video	1	.5
Healthcare Specialist, Video Materials, Cancer Survivor	1	.5
Healthcare Specialist, Written Specialist, Written Survivor	1	.5
Healthcare Specialist, Written Materials, Video	1	.5
Internet, Video	1	.5
Video	14	6.8
Video, Healthcare Specialist	14	6.8



Video, Internet	1	.5
Written Materials	1	.5
Written Materials, Class	2	1.0
Written Materials, Class, Video, Healthcare Specialist	3	1.5
Written Materials, Class, Video, Healthcare Specialist, Cancer Survivor	1	.5
Written Materials, Healthcare Specialist	3	1.4
Total	205	100.0



BREAST CANCER SURVIVORS' PREFERRED LEARNING SOURCES ABOUT EMOTIONAL CONCERNS

The table 4. below presents breast cancer survivors' preferred learning sources about emotional concerns. It shows out of a total of 205 breast cancer survivors, almost **69%** breast cancer survivors opted to consult the **healthcare specialist only** whereas **8%** preferred learning through the **video** for their emotional issues. **6%** of cancer survivors preferred learning through **video and healthcare professionals. 3%** wished to learn through **other cancer survivors** and another **3%** showed learning from **healthcare specialists and other cancer survivors** about their emotional concerns. Only **1.5%** of survivors preferred **class** as a source of learning about emotional concerns.

Thus, it was concluded that the majority of breast cancer survivors preferred to consult with and learn from the health specialist as well as from video for their emotional concerns whereas the internet was the least favourable choice among cancer survivors.

Learning sources about emotional concerns	Frequency	Percent
Cancer Survivor	7	3.4
Cancer Survivor, Healthcare Specialist	1	.5
Cancer Survivor, Healthcare Specialist, Video Materials	1	.5
Class	3	1.5
Class, Cancer Survivor	1	.5
Class, Healthcare Specialist, Cancer Survivor	1	.5
Class, Video	1	.5
Class, Video Specialist	1	.5
Healthcare Specialist	142	69.3
Healthcare Specialist, Cancer Survivor	6	2.9
Healthcare Specialist, Video	2	1.0
Healthcare Specialist, Video	2	1.0
Video	16	7.8
Video, Class	1	.5
Video, Healthcare Specialist	13	6.3
Video, Internet	1	.5
Written Materials	1	.5
Written Materials, Class	1	.5
Written Specialist, Video	1	.5

Table 4: Breast cancer survivors' preferred learning source about emotional concerns



Written Materials, Class, Video, Healthcare Specialist, Cancer Survivor	1	.5
Written Materials, Healthcare Specialist	2	.9
Total	204	100.0



BREAST CANCER SURVIVORS' PREFERRED LEARNING SOURCES ABOUT SPRITUAL/RELIGIOUS CONCERNS

The table 5. below presents breast cancer survivors' preferred learning sources about spiritual or religious concerns. It shows that out of a total of 204 breast cancer survivors (excluding blank, n=1), almost **41%** breast cancer survivors opted to consult the **healthcare specialist only** whereas **24%** preferred learning through **classes only** for their spiritual or religious needs. **6%** of cancer survivors preferred learning through **video only**. Only **4%** wished to learn through **video and healthcare specialists** about the spiritual or religious concerns.

Thus, it was concluded that the majority of the breast cancer survivors preferred to consult with and learn from the health specialist as well as classes for their spiritual or religious concerns whereas Internet and written material were observed as the least favourable choices among cancer survivors.

Learning sources about emotional concerns	Frequency	Percent
Cancer Survivor	7	3.4
Cancer Survivor, Class	3	1.5
Cancer Survivor, Healthcare Specialist, Cancer	1	0.5
Class, Healthcare Specialist	5	2.5
Class, Healthcare Specialist, Cancer Survivor	1	0.5
Class, Video	6	2.9
Class, Video Specialist	3	1.5
Class, Video Specialist, Class	1	0.5
Class, Video Survivor, Internet	1	0.5
Class	49	24.0
Healthcare Specialist	84	41.2
Healthcare Specialist, Cancer Survivor	3	1.5
Healthcare Specialist, Class	5	2.5
Healthcare Specialist, Class Survivor	1	0.5
Internet	1	0.5
Internet, Video	1	0.5
Survivor, Internet	1	0.5
Video	12	5.9
Video, Class	9	4.4
Video, Healthcare Specialist	9	4.4
Video, Internet	1	0.5

 Table 5: Breast cancer survivors' preferred learning source about spiritual or religious concerns



Cancer Survivor	7	3.4
Cancer Survivor, Class	3	1.5
Cancer Survivor, Healthcare Specialist, Cancer	1	0.5
Class, Healthcare Specialist	5	2.5
Class, Healthcare Specialist, Cancer Survivor	1	0.5
Class, Video	6	2.9
Class, Video, Healthcare Specialist	3	1.5
Total	204	100.0



4.1.2.1b BREAST CANCER SURVIVORS' PREFERRED TOPIC OF LEARNING INTEREST

Breast cancer survivors were asked to indicate their **preferred choices** about four **areas including Healthy living choices, Financial Concerns, Enhancing communication and Health Screening.** The options included for each area are mentioned below:

(A) Healthy living choices including nutrition, safe exercise and smoking cessation.

(B) Financial Concerns including living wills and disability.

(C) Enhancing communication including enhancing communication with either doctors or with a spouse or a partner whereas

(D) Health Screening including enhancing communication Health screening included cancer and heart screening.



BREAST CANCER SURVIVORS' PREFERRED TOPIC OF LEARNING INTEREST ABOUT HEALTHY LIVING

The table 6. below provides breast cancer survivors' preferred topic of learning interest about healthy living. It shows that out of a total of 196 breast cancer survivors (excluding blank, n=9), **46%** of breast cancer survivors mentioned that they were interested in learning more about **nutrition and safe exercise** with regards to **healthy living choices**. Whereas **40%** showed their interest in learning about **nutrition only. 12%** of survivors preferred learning about **safe exercise**. Whereas only 0.5% of cancer survivors showed their interest in **smoking cessation and nutrition** both to achieve healthy living choices.

Thus, it was concluded that the majority of breast cancer survivors preferred to learn about nutrition and safe exercise to achieve a healthy lifestyle. Smoking cessation was the least interesting topic for learning and only 1% of the survivors opted for this option.

Table 6: Breast cancer survivors' preferred topic of learning interest about healthy living

Learning interest about healthy living	Frequency	Percent
Nutrition	79	40.3
Nutrition, Safe exercise	91	46.4
Nutrition, Smoking cessation	1	0.5
Safe exercise	24	12.2
Smoking cessation, Nutrition	1	0.5
Total	196	100.0



BREAST CANCER SURVIVORS' LEARNING INTEREST ABOUT FINANCIAL CONCERNS

The table 7. below shows breast cancer survivors' preferred topic of learning interest about financial concerns. With regards to **financial concerns**, **79%** (**n=162**) of survivors did not provide an answer about their interest to learn more about financial concerns. Majority of the cancer survivors (67%) mentioned that they were interested in **living will**, which is a written document in which a person specifies the type of medical treatment they would like if they became so ill that they are certain to die and are unable to communicate their wishes about their treatment. Only 7% survivors showed their interest in learning more about **disability** and equally 7% mentioned **living will and disability both**.

Thus, it was concluded that the majority of breast cancer survivors preferred not to answer about financial concerns which could be their inability to manage their finances. However, some survivors showed their interest in learning more about living wills and very few were interested in knowing more about disabilities.

 Table 7: Breast cancer survivors' preferred topic of learning interest about financial concerns

Learning interest about financial concerns	Frequency	Percent
Disability	7	16.2
Living wills	29	67.4
Living wills, Disability	7	16.2
Total	43	100.0



BREAST CANCER SURVIVORS' LEARNING INTEREST ABOUT ENHANCING COMMUNICATION

The table 8. below shows breast cancer survivors' preferred topic of learning interest about enhancing communication. With regards to **enhancing communication**, **55%** (n=113) of survivors did not provide an answer about their interest in enhancing communication with either of the options given. Whereas, 18% showed their interest in doctors or spouse communication enhancement. **73%** survivors mentioned their interest in enhancing communication with doctors and only 4% mentioned their interest in improving communication only with their **spouse or significant other**.

Thus, it was concluded that the majority of the breast cancer survivors preferred not to answer about interest in enhancing communication. However, those who answered, showed more interest in approaching their doctors only for enhancing communication compared to their spouse or significant other which were only few.

 Table 8: Breast cancer survivors' preferred topic of learning interest about enhancing communication

Learning interest about enhancing communication	Frequency	Percent
With your doctors	67	72.8
With your doctors, With your spouse or significant other	17	18.4
With your spouse or significant other	8	8.6
Total	92	100.0



BREAST CANCER SURVIVORS' LEARNING INTEREST ABOUT HEALTH SCREENING

The table 9. below shows breast cancer survivors' preferred topic of learning interest about health screening. With regards to **Health Screenings**, approximately **8%** (**n=16**) **of patients** did not answer for their interest to learn about health screenings. However, **64% of survivors** mentioned that they are interested to learn about **cancer screening** whereas 36% showed their interest in **health screenings including both cancer and heart**.

Thus, it was concluded that the majority of the breast cancer survivors were interested in learning more about health screening particularly in cancer screening in addition to some interest in knowing more about health screening in order to attain health living.

 Table 9: Breast cancer survivors' preferred topic of learning interest about health

 screening

Learning interest about health screening	Frequency	Percent
Cancer	121	64.0
Cancer, Heart	68	35.9
Total	189	100.0



4.1.2.1c BREAST CANCER SURVIVORS' PRIMARY SOURCE OF STRENGTH

The breast cancer survivors' primary source of strength was explored during their cancer experience including God, God & Family and Family & friend.

The table 10. below presents breast cancer survivors' primary source of strength during the course of disease. Out of a total of 193 breast cancer survivors (excluding blank, n=12), **79%** specified **God** as the only primary source of strength during their cancer experience. **12 % of patients** considered **God and family** both as the sources of strength. Approximately, **9%** mentioned **family and friends** as their primary sources of strength during the cancer journey and 6% of patients decided not to answer and left this question blank in the survey.

Table 10: Breast cancer survivors' primary source of strength

Primary source of strength	frequency	Percent
Family & Friends	17	8.8
God	153	79.2
God & Family	23	11.9
Total	193	100.0



4.1.3 MOST PREVELANT PSYCHOSCOIAL CONCERNS OF BREAST CANCER SURVIVORS'

Breast cancer survivors concerns were explored by their severity level within six psychosocial domains, as mentioned below. (See Appendix 5A, Section 3):

- 1. **Physical concerns** (Pain; fatigue; sleep disturbance; memory and concentration; nausea and vomiting; poor appetite; trouble swallowing; dental or mouth problems; weight changes; balance, walking, and mobility; loss of strength; peripheral neuropathy; lymphedema; osteoporosis and bone health; hair and skin care issues; body changes; bowel or bladder changes; hot flashes; sexual issues; and fertility issues.
- 2. **Information/Educational** Concerns (Being fully informed about their cancer and treatment, potential side effects and long-term adverse effects of the treatment, any familial risk associated with the disease, any symptom appear (recurrence/new primary), who to contact if any symptom appear (recurrence/new primary), next follow-up date, place and physician, the Psychosocial support and resources to other support services).
- 3. Social & Financial Concerns (Managing household activities, caring for family members, talking about cancer with family and friends, returning to work, health insurance, financial concerns, and debt from medical bills).
- 4. Emotional Concerns (Defining a new sense of normal; managing difficult emotions such as anger, fear, sadness, depression, guilt, and anxiety; coping with grief and loss; living with uncertainty; fear of recurrence; managing stress; isolation and feeling alone; intimacy issues; looking for the bright side with gratitude, forgiveness, love, happiness, and contentment; having a sense of well-being; changing relationships with spouse, family, and others; finding support resources; connecting to counselling services; and genetic counselling which includes concerns and fear about your children getting cancer).
- 5. **Spiritual Concerns** (Religious or spiritual support, religious or spiritual distress, loss of faith and end-of-life concerns).
- 6. **Other concerns** (Staying connected with the medical system, knowing who to call for medical problems, keeping your primary care physician informed of your cancer



treatment and risk of recurrence, use of complementary and alternative therapies, and concern about long-term effects of treatment).

Mean severity concern scores were calculated based on the entire sample. In addition, the data was analysed to explore survivors' major concerns using the sample of 205 patients as a whole as well as by time elapsed since diagnosis.

The individual severity scores of all concern items were categorised into three groups including: low concerns (score: 0 to 1), moderate concerns (score: 2 to 3) and extreme concerns (score: 4 to 5).

Correlational and regression analysis to evaluate the relationship between independent participants' characteristics and severity of concerns (dependent variables) was planned and undertaken.

The independent variables used in the modelling process for regression were gender, age, nationality, treatment of treatment and time since diagnosis. On the other hand, the dependent variable was set as an extreme concern with a rating of 4 to 5. Multivariate linear models were created, and the relationship was also evaluated between overall quality of life scored and having an extreme level of concern (score: 4-5).

Table 11 below provides the overall summary of descriptive statistics of breast cancer survivors' most prevalent concerns.

- The breast cancer survivors had, on average, low or some concerns about their physical needs (mean = 6.14, SD= 12.27).
- The breast cancer survivors had, on average, more **extreme concerns about their** information needs (mean = 27.9, SD= 11.0).
- The mean score of **social and financial concern** is approximately 7. This indicates that the breast cancer survivors had, on average, **some social and financial concern** (mean = 7.13, SD= 6.32).
- The mean score for **emotional concern** is apparently one which indicates that the breast cancer survivors had, on average, **some level of emotional concerns** (mean = 15.05, SD= 10.89).



• The mean score of **spiritual concerns** is less than one which denotes, on average, **no or very low concerns either regarding their spiritual or religious needs** (mean = 0.20, SD= 0.915).

Therefore, it is concluded that the informational needs were, on average, the highest worry for the majority of breast cancer survivors than any other domain of psychosocial concerns.

It should also be noted that the data on the total scores for the different psychosocial concerns and overall quality of life is significantly skewed (see Appendix 10) and therefore that the results involving these variables, including the multiple regression analysis, should be interpreted with caution.

Table	11: \$	Summary	Results of	most p	orevalent	psychosocial	concerns of	breast cancer
		•				1 0		

Score of Psychosocial needs	Mean	S. D
Physical Concerns	6.14	12.273
Information Concerns	27.99	11.020
Social & Financial Concerns	7.13	6.323
Emotional Concerns	15.05	10.896
Spiritual/Religious Concerns	.20	0.915



4.1.4 MOST PREVALENT PSYCHOSOCIAL CONCERNS OF BREAST CANCER SURVIVORS BY YEAR OF DIAGNOSIS

The data about breast cancer survivors' concerns was divided into three timeframes based on the time elapsed since initial diagnosis which are 0 to 1 year, 2 to 5 years and 6 and earlier years since initial diagnosis in order to identify the most prevalent extreme concerns described by the participants within certain timeframes.

Within the first year after breast cancer diagnosis (2018 and onwards), respondents reported extreme concerns regarding information with the highest mean score (32) and then emotional concern with a mean score of 13.89.

Between 2-5 years after breast cancer diagnosis (2012-2017), respondents reported the same concerns as they reported after a year of initial diagnosis which were **information concerns** with a mean score approximately 30 and emotional concern with a mean score of 17.

Above 6 years and earlier after breast cancer diagnosis (2011 and earlier), **information concerns** remained the most prevalent as reported by the participants in other two timeframes with mean score of 32; however, social and financial concerns were reported high from six years since the initial diagnosis with mean score of 10. Table 12 below provides the breast cancer survivors' most prevalent concerns during three different time frames.

Most Prevalent Concerns by Year of diagnosis	n	mean	SD
0–1 year since diagnosis, $>=2018$ (n =122)			
Physical	122	7	13.3
Information	122	26.6	11.5
Social and financial	122	6.30	6.2
Emotional	122	13.89	10.6
Spiritual and religious	122	6.23	1.02
2-5 years since diagnosis, 2012-2017 (n = 80)			
Physical	80	4.31	8.7
Information	80	29.9	10
Social and financial	80	8.3	6.4
Emotional	80	17	11.3
Spiritual and religious	80	0.16	0.75
6 and earlier since diagnosis (n = 3)			

Table 12: Breast cancer survivors' most prevalent concerns by year of diagnosis



Physical	3	4.31	33.2
Information	3	32	5.2
Social and financial	3	10	7.2
Emotional	3	9.3	4.04
Spiritual and religious	3	0	0



4.1.5 SEVERITY OF BREAST CANCER SURVIVORS' PSYCHOSOCIAL CONCERNS

The severity of breast cancer survivors' concerns' scores within each psychosocial domain was calculated as below. The most extreme concern reported by the breast cancer survivors among all psychosocial domains was **"being fully informed about potential Psychosocial needs and long-term adverse effects"** with a mean of **4.34**. That indicated that **information was one of the most severe concerns** reported from the majority of the participants.

SEVERITY OF BREAST CANCER SURVIVORS' PHYSICAL CONCERNS

Within the physical domain, the highest mean scores overall (rating 0-5) were **fatigue, pain, hair and skin concerns and sleep disturbance** however, when excluding those who rated a score of 0 (No concern) and only reporting on items rated 1 to 5 (as a concern), the most prevalent concerns reported were hot flashes, hair and skin care, weight changes, and fertility concerns. Utilising the entire sample, the result indicates that the mean score is less than one indicating that the **breast cancer survivors had no major physical concern** about their health. Table 13. below presents the severity of breast cancer survivors' physical concerns.

PHYSICAL CONCERNS		SCORI	SCORE 0-5		SCORE 1-5	
a Range = $0-5$ b Excludes participants who selected 0 responses (range = $1-5$)		Severit	y a	Severit	ty b	
		Mea n	SD	Mea n	SD	
PHYC1	Pain	.64	1.274	2.40	1.355	
PHYC2	Fatigue	.69	1.283	2.24	1.376	
РНУС3	Sleep disturbance	.38	1.044	2.44	1.413	
PHYC4	Memory and concentration	.25	.757	1.93	1.072	
РНҮС5	Nausea / Vomiting	.25	.757	1.93	1.072	
PHYC6	Poor appetite	.22	.803	2.30	1.380	
PHYC7	Trouble swallowing	.25	.757	1.93	1.072	
PHYC8	Dental or mouth problems	.14	.622	2.07	1.328	
РНҮС9	Weight changes	.35	.992	2.57	1.230	
РНҮС1 0	Balance / Walking / Mobility	.29	.908	2.46	1.318	
PHYC1 1	Loss of strength	.28	.889	2.38	1.345	

Table	13: S	everity o	f breast	cancer	survivors'	Physical	concerns
		•				•	



PHYC1 2	Tingling or numbness in feet & hands (neuropathy)	.19	.630	1.73	1.032
PHYC1 3	Swelling in legs or arms (lymphedema)	.15	.587	1.72	1.127
PHYC1 4	Osteoporosis / Bone health	.27	.899	2.33	1.465
PHYC1 5	Hair and skin care concerns	.46	1.161	2.64	1.397
PHYC1 6	Body changes	.36	.983	2.35	1.305
PHYC1 7	Bowel or bladder changes	.33	.963	2.31	1.417
PHYC1 8	Hot flashes	.32	1.068	3.00	1.633
PHYC1 9	Sexual concerns	.09	.404	1.58	0.669
PHYC2 0	Fertility concerns	.21	.786	2.53	1.281

SEVERITY OF BREAST CANCER SURVIVORS' INFORMATION CONCERNS

In the information domain, the most prevalent concern with a highest means score (4.34) was" being fully informed about potential psychosocial needs and long-term adverse effects" (n=205) followed by concerns related to "being fully informed about any symptom appear (recurrence/new primary)" (n=202) and "being fully informed about any familial risk" (n=201), with more than half (above 50%) of the participants reporting concerns in all but one of these areas. When excluding those who rated a score of 0 (No concern) and only reporting on items rated 1 to 5 (as a concern), the most prevalent concerns reported were being fully informed about potential psychosocial needs and long-term adverse, being fully informed about the Psychosocial support and resources to other support services and being fully informed about who to contact if any symptom appears (recurrence/new primary). The overall results indicate that the mean score is approximately between 3 and 4 (approximately 4) which denotes that the breast cancer survivors apparently had more concern about the information needs. Table 14. below presents the severity of breast cancer survivors' information concerns.



INFORMATION CONCERNS		SCORE	E 0-5	SCORE 1-5		
a Panao - () 5	Severit	Severity a		y b	
a Range = $0-5$ b Excludes participants who selected 0 responses (range = $1-5$)		Mea n	SD	Mea n	SD	
INF_1	Being fully informed about my cancer and treatment	3.89	1.829	4.502 8	1.0451 4	
INF_2	Being fully informed about potential Psychosocial needs and long-term adverse effects	4.34 6	1.481 0	4.714 3	0.7941 7	
INF_3	Being fully informed about any familial risk	3.96	1.765	4.511 1	1.0274 9	
INF_4	Being fully informed about any symptom appear (recurrence/new primary)	4.00	1.749	4.555 6	0.9700 7	
INF_5	Being fully informed about who to contact if any symptom appears (recurrence/new primary)	3.93	1.812	4.573 9	0.9228 5	
INF_6	Being fully informed about next follow-up date, place and physician	3.94	1.793	4.533 7	0.9923 4	
INF_7	Being fully informed about the Psychosocial support and resources to other support services	3.93	1.808	4.579 5	0.9099 0	

Table 14: Severity of breast cancer survivors' information concerns

SEVERITY OF BREAST CANCER SURVIVORS' SOCIAL AND FINANCIAL CONCERNS

In the social and financial domain, the two most prevalent concerns with the highest mean score for each 1.96 were health Insurance and financial issues. The result indicates that the mean score is approximately one. This represents that breast cancer survivors had less concern about the social and financial needs. When excluding those who rated a score of 0 (No concern) and only reporting on items rated 1 to 5 (as a concern), the most prevalent concerns reported were also with 0 concern and the mean score was 3.0455. Table 15. below presents the severity of breast cancer survivors' social & financial concerns.

Table 15: Severity of breast cancer survivors' social & financial concerns

SOCIAL & FINANCIAL CONCERNS		SCORI	SCORE 0-5		1-5	
a Danco -	0.5	Severit	Severity a		Severity b	
a Range = $0-5$ b Excludes participants who selected 0 responses (range = $1-5$)		Mea n	SD	Mea n	SD	
SFC1	Managing household activities	.85	1.204	1.822 9	1.1516 1	
SFC2	Caring for family members	.86	1.168	1.791 7	1.0849 9	



SFC3	Talking about cancer with family and friends	.61	1.052	1.739 1	1.0934 6
SFC4	Returning to work	.94	1.420	2.345 7	1.3149 2
SFC5	Health insurance	1.96	1.860	3.045 5	1.4349 2
SFC6	Financial concerns	1.96	1.860	3.045 5	1.4349 2

SEVERITY OF BREAST CANCER SURVIVORS' EMOTIONAL CONCERNS

In the emotional domain, "fear of recurrence" (n=205) and "genetic counselling" (worries about children getting cancer (n=203) were the most prevalent concerns with a highest mean score of 3.4 for each item. When excluding those who rated a score of 0 (No concern) and only reporting on items rated 1 to 5 (as a concern), the most prevalent concerns reported were also as with 0 concern and the mean score was around 4.33. The overall result in this domain represents that the mean score is approximately one, indicating that the breast cancer survivors had **some concerns about the emotional needs except factors including fear of recurrence and genetic counselling.** This explains that they had concerns about cancer reappearing and worries about their offspring getting cancer inherently. Table 16. below presents the severity of breast cancer survivors' emotional concerns.

EMOTIONAL CONCERNS		SCORE	0-5	SCORE 1	1-5
a Range = 0-	5	Severity	a	Severity	r b
b Excludes pa	irticipants who selected 0 responses (range = $1-5$)	Mea n	SD	Mean	SD
EC1	Defining a new sense of normal	1.04	1.380	2.161 6	1.23476
EC2	Managing difficult emotions: anger, fear, sadness, depression, guilt, anxiety)	.78	1.200	1.939 0	1.15841
EC3	Coping with grief and loss	.40	.905	1.840 9	1.07710
EC4	Living with uncertainty	1.41	1.828	3.042 1	1.49407
EC5	Fear of recurrence	3.44	1.966	4.304 9	1.05306
EC6	Managingstress	.97	1.246	1.960 4	1.09472
EC7	Isolation / Feeling alone	.40	.911	1.844 4	1.06506
EC8	Intimacy issues	.26	.718	1.709 7	0.97275

TT 11 1/	G • 4		• •		
Table 16	 Severity of 	hreast cancer	SURVIVORS'	emotional	concerns
I abic 10	· Devenity of	of cast cancer	Suivivois	cinotional	concerns



EC9	Looking for the bright side: gratitude, forgiveness, love, happiness, contentment	.59	1.079	2.016 7	1.04948
EC10	Having a sense of well being	1.04	1.537	2.696 2	1.28457
EC11	Changing relationships with spouse, family, and others	.28	.797	1.933 3	1.08066
EC12	Finding support resources	.45	1.095	2.384 6	1.31019
EC13	Connecting to counselling services	.54	1.140	2.265 3	1.23787
EC14	Genetic counselling (worry about your children getting cancer)	3.44	1.988	4.331 3	1.05442

SEVERITY OF BREAST CANCER SURVIVORS' RELIGIOUS/SPIRITUAL CONCERNS

Religious or spiritual concerns were neither prevalent nor significant and did not score high as compared to other psychosocial domains. However, for those who did rate some level of concern (rating 1 to 5), end of life concerns was reported by 13 participants (6%) with a mean of 0.16, whereas the majority of the patients above 95% showed no spiritual or religious concerns in any of the item apart from end-of-life care. The domain results show that the **spiritual or religious needs were less concerning** as the mean score was less than one. In the other domain, no participant reported any concerns. Table 17. below presents the severity of breast cancer survivors' religious or spiritual concerns.

Table	17: Seve	rity of br	east cancer	• survivors	' religious/	spiritual	concerns
						~p	••••••

RELIGIOUS/SPRITITUAL CONCERNS		SCORE 0-	5	SCORE 1	-5
$a \operatorname{Ranga} = 0.5$		Severity a		Severity b	
b Excludes	participants who selected 0 responses (range = 1–5)	Mean	SD	Mean	SD
SRC1	Religious or spiritual support	.02	.279	4.0000	
SRC2	Religious or spiritual distress	.02	.279	4.0000	
SRC3	Loss of faith	.00	.000		
SRC4	End of life concerns	.16	.738	2.5385	1.61 325



4.1.6 ASSOCIATION BETWEEN BREAST CANCER SURVIVORS' (A) NATIONALITY & AGE (B) NATIONALITY & STAGE OF CANCER

Literature shows that the breast cancer occurrence in Emirati women is at earlier ages unlike other world population or nationalities (Al-Shamsi & Alrawi, 2018; FOCP, 2018; Najjar & Easson; 2010; Elobaid, 2010; Tadmouri, & Al-Sharhan, 1985). Additionally, there is a late presentation with an advanced stage of cancer in Emiratis than other nationalities (Younis, et. al., 2016; Elobaid, 2010). Thus, an association between nationality and stage of breast cancer was explored using Chi-Square test as follows:

4.1.6a ASSOCIATION BETWEEN BREAST CANCER SURVIVORS' NATIONALITY & AGE

HYPOTHESIS 1:

H0: There is no association between age and nationality

H1: There is an association between age and nationality

It is observed that the breast cancer occurrence in Emirati women is at earlier ages unlike other world population or nationalities thus an association between nationality and age was explored as below:



Breast cancer survivors' age group by nationality

Table 18a presents the percentage of breast cancer survivors by age groups in both Emiratis and expatriates:

- 91.7% of survivors aged [22-32] were expatriate, while 8.3 % were Emirati.
- 68.9 % of survivors aged [33-43] were expatriate, while 31.3 % % were Emirati.
- 21.9% of survivors aged [44-54] were Emirati, while 78.1 % were expatriates.
- 33.4% of survivors aged [55-65] were Emirati, while 66.7 % were expatriates.
- 20.0% of survivors aged above 65 were Emirati, while 80.0 % were expatriate

Table 18a: Breast cancer survivors' age by nationality

Age group * Nationality Crosstabulation

Each subscript letter denotes a subset of Nationality categories whose column proportions do not differ significantly from each other at .05 level.

			Nati	onality	
			Emirati	Expatriate	lotal
	22.22	Count	1a	11a	12
22-32		Expected Count	3.2	8.8	12.0
		Count	20a	44a	64
	33-43	Expected Count	16.9	47.1	64.0
-	44 54	Count	14a	50a	64
Age group	44-54	Expected Count	16.9a	47.1a	64.0
	55 (S	Count	15	30	45
55-65		Expected Count	11.9	33.1	45.0
	. (5	Count	4a	16a	20
	>05	Expected Count	5.3	14.7	20.0
Tatal		Count	54	151	205
Total		Expected Count	54.0	151.0	205.0



Hypothesis 1 – There is no association between breast cancer survivors' nationality and age

Table 18b provides the result of Chi-square test conducted for hypothesis one showing that the proportion of breast cancer survivors' ages, those who were Emirati was not significantly different from the proportion of survivor's ages, who were expatriates. Therefore, it can be concluded that **there was no significant association between age and nationality** (X2 = 5.007, p=0.287).

Chi-Square Tests					
	Value	Df	Asymptotic Significance (2-sided)		
Pearson Chi-Square	5.007 ^a	4	.287		
Likelihood Ratio	5.479	4	.242		
Linear-by-Linear Association	.069	1	.792		
N of Valid Cases	205				
a. 1 cells (10.0%) have expected count less than 5. The minimum expected count is 3.16.					

Table 18b: Hypothesis 1 – Chi-Square Tests

There are a number of effect size statistics available in the crosstabs procedure. For 2 by 2 tables, the most commonly used one is the **phi coefficient**, which is a correlation coefficient and can range from 0 to 1. The higher value indicates a stronger association between the two variables. Table 18c provides the Phi value in the symmetric measures which was calculated as **0.156**. This considers a **very minimal effect** using Cohen's (1988).

Table 18c: Hypothesis 1 – phi Coefficient

		Value	Approximate Significance
X7 · 11 X7 · 1	Phi	.0156	.287
Nominal by Nominal	Cramer's V	.056	.287
N of Valid Cases		205	

Conclusion: "A Chi-Square test for independence indicated no significant association between age and nationality, Pearson chi Square (df=4, n=205) =5.007, Sig=0.287, phi=0.156 from what would be expected by chance".



4.1.6b ASSOCIATION BETWEEN BREAST CANCER SURVIVORS' STAGE OF CANCER & NATIONALITY

HYPOTHESIS 2:

H0: There is no association between nationality and stage of breast cancer survivors

H1: There is an association between nationality and stage of breast cancer survivors

It is observed that there is a late presentation with an advanced stage of cancer



Breast cancer survivors' stage of cancer by nationality

Table 19a presents the percentage of breast cancer survivors by stage groups in both Emiratis and expatriates:

- **In-situ stage:** In total of 13 patients with In-situ stage, 61% of survivors were expatriate and 39 % were Emirati.
- **Stage I:** In total of 39 patients with stage I, 69% of survivors were expatriate and 31 % were Emirati.
- **Stage II:** In total of 105 patients with stage II, 70% of survivors were expatriate and 30 % were Emirati.
- **Stage III:** In total of 48 patients with stage III, 88% of survivors were expatriate and 12 % were Emirati.

Breast can	cer Stage by I	Nationality Crosstabula	tion		
			Nationality		
			Emirati	Expatriate	Total
	T 6"4	Count	5 _a	8a	13
	In-Situ	% Within Stage	38.5%	61.5%	100.0%
	Ct T	Count	12 _a	27 _a	39
Breast	Stage I	% Within Stage	30.8%	69.2%	100.0%
Cancer Stage	G4 II	Count	31 _a	74 _a	105
Stage	Stage 11	% Within Stage	29.5%	70.5%	100.0%
	64 111	Count	6 _a	42 _b	48
	Stage III	% Within Stage	12.5%	87.5%	100.0%
T ()		Count	54	151	205
Total		% Within Stage	26.3%	73.7%	100.0%
Each subscri each other a	ipt letter denotes t the .05 level.	a subset of Nationality cate	gories whose column proportion	s do not differ sign	ificantly from

Table 19a: Breast cancer survivors stage by nationality



Hypothesis 2 – There is no association between stage of breast cancer and nationality

The below tables 19b and 19c, Chi square Z test show that there is no association in most part between nationality and cancer stage however the Z test shows a significant association for stage 3 with significantly higher numbers for expatriates and lower numbers for Emirati than would be expected by chance.

Chi square Z test shows that there is no association in most part between nationality and modified stage however the Z test shows a significant association for stage 3 with significantly higher numbers for expatriates and lower numbers for Emirati than would be expected by chance. Because we have not violated the assumption of Chi-Square concerning the minimum expected value frequency, which should be 5 or greater (or at least 80 per cent of the total value. The value 12.5 % has an expected count less than 5, thus considered Asymptotic sig. (2-sided), Sig=0.083 showing that **our result is not significant.** This me**ans that the proportion of breast cancer survivors from different stages was not significantly different** from what would be expected by chance with the exception of stage 3 as noted above from the Z tests.

Chi-Square Tests				
	Value	df	Asymptotic Significance (2-sided)	
Pearson Chi-Square	6.666ª	3	.083	
Likelihood Ratio	7.344	3	.062	
Linear-by-Linear Association	4.949	1	.026	
N of Valid Cases	205			
a. 1 cells (12.5%) have expected count less than 5. The minimum expected count is 3.42.				

Table 19b: Hypothesis 2 – Z TEST

Since the cross table above is larger than 2 by 2, the value to report is Cramer's V, which takes

into account the degrees of freedom. The Cramer's V in the symmetric measures is 0.180, which is considered as a weal effect using Cohen's (1988). Thus, the result of this hypothesis can be presented as: "A Chi-Square test for independence indicated a non-significant association between stage of cancer and nationality, Pearson chi Square (df=3, n=205) =13.661, Sig=0.083, Cramer'V=0.256".



Table 19c: Hypothesis 2 – phi Coefficient

Symmetric Measures			
		Value	Approximate Significance
	Phi	.180	.083
Nominal by Nominal	Cramer's V	.180	.083
N of Valid Cases		205	



4.1.7 INFLUENCE OF NATIONALITY ON PSYCHOSOCIAL NEEDS SCORES OF BREAST CANCER SURVIVORS

An independent Sample t-test (Welch's test) was conducted to test if nationality has an influence on scoring rated by breast cancer survivors on each domain of psychosocial concerns.

From the below table 20a., the following conclusion can be made:

- A significant difference was found between Emirati and expatriate breast cancer survivors in physical domain (t=-2.180, p< 0.05) as well as social & financial domain (t=-4.997, p<0.05)
- There were no significant differences between Emirati and expatriates in other domains of psychosocial concerns including information, emotional and spiritual or religious concerns.

 Table 20a: Summary of differences in psychosocial concerns scores between Emiratis and

 expatriates breast cancer survivors

Score	T-test	p-value	Decision
Physical Concerns	-2.180	0.031	S
Information Concerns	-0.986	0.326	N/S
Social & Financial Concerns	-4.997	0.000	S
Emotional Concerns	1.676	0.097	N/S

S= Significant, N/S= Not Significant

Welch's t test was used because of unequal variances between the groups arising from unequal sample sizes.



Influence of nationality on physical concerns of breast cancer survivors

T-tests were performed to explore the differences in **Physical Concerns** between two groups, Emiratis and Expatriates.

Result: There was a statistically significant difference in Physical Concerns between Emiratis and Expatriates (3.48). Physical Concerns were higher in Expatriates (7.05) than Emiratis (3.57). **t=-2.153**, **p-value=0.033**.

 Table 20b: Difference in physical concerns between emiratis and expatriates breast

 cancer survivors

	Nationality	Ν	Mean	SD	Std. Err/Me an	t-value	p-value
Physical	Emirati	54	3.57	8.878	1.208	2 1 5 2	0.022
Concerns	Expatriate	151	7.05	13.183	1.073	-2.153	0.033

Influence of nationality on information concerns of breast cancer survivors

T-tests were performed to explore the differences in **Information Concerns** between two groups Emiratis and Expatriates.

Result: There were no differences in information needs between Emiratis and Expatriates. **t=-1.020**, **p-value=0.311**.

 Table 20c: Difference in information concerns between Emiratis and expatriates breast

 cancer survivors

	Nationali ty	Ν	Mean	SD	Std. Err/ Mean	t-value	p-value
	Emirati	54	26.41	14.395	1.959		
Information Concerns	Expatriat e	151	28.56	9.524	0.775	-1.020	0.311

Influence of nationality on social/financial concerns of breast cancer survivors



T-tests were performed to explore the differences in **Social & Financial Concerns** between two groups Emiratis and Expatriates.

Result: There were higher differences in Social & Financial Concerns between Emiratis and Expatriates (4.8). Social and financial Concerns were higher in Expatriates (8.39) than Emiratis (3.59). **t=-5.067**, **p<0.001**.

 Table 20d: Difference in social/financial concerns between Emiratis and expatriates

 breast cancer survivors

	Nationality	Ν	Mean	SD	Std. Err/Mean	t-value	p-value
Social	Emirati	54	3.59	3.839	0.522	5.077	-0.001
Concerns	Expatriate	151	8.39	6.561	0.534	-5.06/	<0.001

Influence of nationality on emotional concerns of breast cancer survivors

T-tests were performed to explore the differences in **Emotional Concerns** between two groups Emiratis and Expatriates.

Result: There were no differences in emotional needs between Emiratis and Expatriates. t=1.671, p-value=0.096.

 Table 20e: Difference in emotional concerns between emiratis and expatriates breast

 cancer survivors

	Nationality	Ν	Mean	SD	Std. Err/Mea n	t-value	p-value
Emotional	Emirati	54	17.17	10.056	1.368	1 (71	0.000
Concerns	Expatriate	151	14.29	11.116	0.905	1.6/1	0.096

Influence of nationality on religious spiritual concerns of breast cancer survivors

T-tests were performed to explore the differences in spiritual/**religious concerns** between two groups Emiratis and Expatriates.


Result: There were no differences in spiritual/religious concerns between Emiratis and Expatriates. **t=0.727**, **p-value=0.468**.

				_		-
Table 20	f. Difforona	a in aninitu	al/moligiona	aanaanna hatwaan	aminatic and	wnothiotog
I able 20	n: Difference	e ili suiritu	ai/rengious	concerns between	emmatis and (expairates

	Nationality	Ν	Mean	SD	Std. Err/Mean	t-value	p-value
Religious/Spiritual	Emirati	54	0.28	0.960	0.131	0.707	0.460
Concerns	Expatriate	151	0.17	0.900	0.073	0.727	0.468



4.1.8 CORRELATION BETWEEN QUALITY OF LIFE & PSYCHOSOCIAL CONCERNS OF BREAST CANCER SURVIVORS

Correlation **between quality of life and psychosocial concerns** was also explored to find out whether there is any positive or negative correlation between the two factors.

There was a significant **negative correlation found between quality of life and the scores of psychosocial concerns including physical, social, emotional as well as spiritual concerns** (p < 0.01). The results indicated that the higher the psychosocial concerns (including physical, social, emotional as well as spiritual or religious concerns), the lower the quality of life (one increases the other decreases) as in line with the literature (DiSipio, et. al., 2009; Avis & Crawford, 2005; Edib, et. al., 2016).

A positive correlation was found between quality of life and information concern (p < 0.01). In other words, it could be concluded that if quality of life scores increase, the information concerns also increase along with it. Therefore, an assumption can be made that those cancer survivors who scored higher in the information domain also reflected higher scores in quality-of-life assessment which is quite consistent with the previous studies (Park & Hwang, 2012; Carey, et. al., 2012; Uchida, et. al., 2011).

Table 21: Correlation between	quality of life and	psychosocial	concerns	scoring	breast
cancer survivors					

Relationship	Correlation (R-value)	Significance level (P-value)
Overall QL and physical concerns	-0.481	< 0.001
Overall QL and information concerns	0.208	0.003
Overall QL and social concerns	-0.271	< 0.001
Overall QL and emotional concerns	-0.366	< 0.001
Overall QL and spiritual concerns	-0.166	0.017



4.1.9 RELATIONSHIP BETWEEN (a) DEMOGRAPHIC FACTORS & PSYCHOSOCIAL CONCERNS, (b) PSYCHOSOCIAL CONCERNS & QUALITY OF LIFE OF BREAST CANCER SURVIVORS

Multiple regression was used to measure the **degree of variance between demographic factors and psychosocial concerns** (physical concerns, information concerns, social and financial concerns, emotional concerns, and spiritual or religious concerns) **and between psychosocial concerns and overall quality of life.**

Multiple regression is a technique used **to explore the relationship between the dependant continuous variable such as physical concerns score and a number of independent variables** or predictors (continuous or categorical variables) for instance, age, stage of cancer, or currently receiving treatment to identify the coefficient of variation of the independent variables in the dependent variable. Also, to observe the coefficients, whether these are significant or not. Although multiple regression is based on correlation, it allows a more sophisticated exploration of the interrelationship among a set of variables and predictor variables can be categorical as well as continuous with categories predicting greater or less proportions of variance in scores the dependent measure. This makes it ideal for the investigation of more complex real-life, rather than laboratory-based, research questions.

Multiple regression is used to address a variety of research questions. It can inform how well a set of variables is able to predict a particular outcome. Multiple regression provides information about the model as a whole and the relative contribution of each of the variables that make up the model. The stepwise multiple regression was utilised in the analysis. This type of regression uses the highest independent variable that is correlated with the dependent variable and puts it into an equation. The process was continued until a model was reached to obtain all significant factors and excluded all non-significant factors.



4.1.9a RELATIONSHIP BETWEEN DEMOGRAPHIC FACTORS & PSYCHOSOCIAL CONCERNS OF BREAST CANCER SURVIVORS

Demographic Factors: Age Stage of cancer Receiving treatment **Psychosocial needs:**

Physical Concerns Information Concerns Social & Financial Concerns Emotional Concerns Religious/Spiritual Concerns

Figure 13: Relationship between demographic factors & psychosocial concerns of breast cancer survivors

The relationship has been explored between the scores of different domains of psychosocial concerns and the independent variables including age, stage of cancer and currently receiving treatment. The figure above shows the conceptual model for testing the following hypothesis.

HYPOTHESIS 3:

There is a significant relationship between demographic factors and psychosocial concerns of breast cancer survivors.



RELATIONSHIP BETWEEN DEMOGRAPHIC FACTORS & PHYSICAL CONCERNS OF BREAST CANCER SURVIVORS

The below table 22a. shows that correlation between **each of the demographic variables and physical concern was not significant**, so multiple regression was not conducted since demographic factors were not significantly related to the physical concerns (for SPSS output see Appendix 10, table 34).

Table 22a: Relationship between demographic factors and physical concerns of breast cancer survivors

Relationship	Correlation (R-value)	Significance level (P-value)
Age & physical concerns	-0.114	0.104
Stage & physical concerns	-0.041	0.562
Receiving treatment & physical concerns	0.117	0.095



RELATIONSHIP BETWEEN DEMOGRAPHIC FACTORS & INFORMATION CONCERNS OF BREAST CANCER SURVIVORS

The below table 22b. shows that there was a significant positive correlation between age and information concern, r=0.182, p=.009 and a significant positive correlation between information concern and receiving treatment, r=0.179, p=.010. The relationship between stage of cancer and information concerns was not significant, r=.044, p=.528 and therefore stage of cancer was not included in the multiple regression below.

Multiple linear regression (using the stepwise method with p<.05 for entry and p>.10 for removal) was conducted to assess if the two significant correlates with information concerns are also significant predictors (for SPSS output see Appendix 10, table 35).

In the first model, with age as the sole predictor, R=.182, R-square = .033 and adjusted R square = .028, F (df=1, 203) =6.956, p=.009, which shows that age is a significant predictor of information concerns (confirming the significant correlation reported above) accounting for 3.3% of the variance of information concern.

In the second model by adding a second predictor of currently receiving treatment, R=.252, R -square=.063, and adjusted R-square = .054, F (df=2, 202) =6.832, p=.001, which shows that age and currently receiving treatment together are significant predictors of information concerns and account for 6.3 % of the variance of information concern. It is furthermore the case that in the second model that R-square change=.03, F-change (df=1, 202) =6.591, p=.011, which shows that when currently receiving treatment is added to the equation that explains a significant extra amount of the variance of information concern (with a similar F and level of significance as for when age was entered as the sole predictor variable).



 Table 22b: Relationship between demographic factor and information concerns of breast

 cancer survivors

Relationship	Correlation (R-value)	Significance level (p-value)
Age and information concerns	0.182	0.009
Stage of cancer and information concerns	0.044	0.528
Receiving treatment and information concerns	0.179	0.010

The below table confirms that Age is a significant predictor of information concerns when it is the sole predictor in model 1, Beta=.182, t=2.637, p=.009. In model 2, when currently receiving treatment is also entered into the equation, then Age remains a significant predictor of information concerns, Beta=.177, t=2.602, p=.010 and currently receiving treatment is also a significant predictor of information concerns, Beta=.174, t=2.553, p=.011. Both predictors appear to be of similar size and significance, as also shown by the similar F and F-change results reported above for model 1 and model 2.

Model Summary

					Significance level (p-value)
Model 1	Age		.182	2.637	.009
	Age		.177	2.602	.010
Model 2	Currently treatment	receiving	.174	2.553	.011



RELATIONSHIP BETWEEN DEMOGRAPHIC FACTORS AND SOCIAL & FINANCIAL CONCERNS OF BREAST CANCER SURVIVORS

In the correlation table 22c., there was a positive significant correlation between stage of cancer and social financial concern, r=0.157, p=.025 and also a significant positive correlation between receiving treatment and social financial concern, r=0.246, p<.001. There was a non-significant relationship between age and social financial concerns, r=-0.060, p=.394 and therefore age factor will not be included in the multiple regression below.

Multiple linear regression (using the stepwise method with p<.05 for entry and p>.10 for removal) was conducted to assess if the two significant correlates with social financial concerns are also significant predictors (for SPSS output see Appendix 10, table 36).

In the first model, with currently receiving treatment as the sole predictor R=.246, R-square = .061 and adjusted R square = .056, F (df=1, 203) = 13.118, p=.000 which shows that currently receiving treatment is a significant predictor of social and financial concerns (confirming the significant correlation reported above) accounting for 5.6% of the variance of social and financial concerns.

In the second model by adding a second predictor of cancer stage, R=.286, R -square=.082, and adjusted R-square = .073, F (df=2, 202) =6.832, p=.001, which shows that currently receiving treatment and cancer stage together are significant predictors of social and financial concerns and account for 7.3 % of the variance of social and financial concern. It is furthermore the case that in the second model that R-square change=.03, F-change (df=1, 202) =4.625, p=.011, which shows that when cancer stage is added to the equation that explains a significant extra amount of the variance of social and financial concerns (with a similar F and level of significance as for when currently receiving treatment was entered as the sole predictor variable).



 Table 22c: Relationship between demographic factors and social & financial concerns of

 breast cancer survivors

Relationship	Correlation (R- value)	Significance level (P-value)
Age and Social financial concerns	-0.060	0.394
Stage of cancer and Social financial concerns	0.157	0.025
Receiving treatment and Social financial concerns	0.246	< 0.001

The below table confirms that currently receiving treatment is a significant predictor of social and financial concerns when it is the sole predictor in model 1, Beta=.246, t=3.622, p=.000. In model 2, when cancer stage is also entered into the equation, then currently receiving treatment remains a significant predictor of social and financial concerns, Beta=.239, t=3.545, p=.000 and cancer stage is also a significant predictor of social and financial concerns, Beta=.145, t=2.151, p=.033. Both predictors appear to be of similar size and significance, as also shown by the similar F and F-change results reported above for model 1 and model 2.

Model Summary

Model	Predictor variable	Standardised regression coefficient beta	t-value	Significance level (p-value)
Model 1	Receiving treatment	0.246	3.622	0.000
Model 2	Receiving treatment	0.239	3.545	0.000
	Stage of cancer	0.145	2.151	0.033



RELATIONSHIP BETWEEN DEMOGRAPHIC FACTORS & EMOTIONAL CONCERNS OF BREAST CANCER SURVIVORS

In the correlation table 22d., there was only a significant positive correlation between receiving treatment and emotional concerns, r=0.240, p=.001. There was also a non-significant relationship between age and emotional concerns, r=0.133, p=.058, although this approaching statistical significance and stage of cancer and emotional concerns, r=0.037, p=.602 and therefore age factor and stage of cancer will not be included in the regression below.

Linear regression (using the stepwise method with p<.05 for entry and p>.10 for removal) was conducted (for SPSS output see Appendix 10, table 37) with currently receiving treatment as the sole predictor R=.240, R-square = .058 and adjusted R square = .053, F (df=1, 203) = 12.417, p=.001 which shows that currently receiving treatment is a significant predictor of emotional concerns (confirming the significant correlation reported above) accounting for 5.3% of the variance of emotional concerns.

Table 22d: Relationship between demographic factors and emotional concerns of breast cancer survivors

Relationship	Correlation (R-value)	Significance level (p-value)
Age and emotional concerns	0.133	0.058
Stage of cancer and emotional concerns	0.037	0.602
Receiving treatment and emotional concerns	0.240	0.001

The below table confirms that currently receiving treatment is a significant predictor of emotional concerns when it is the sole predictor with Beta=.240, t=3.524, p=.001.

Model	Predictor variable	Standardised regression coefficient beta	t-value	Significance level (p-value)
Model	Receiving treatment	0.240	3.524	0.001

Model Summary



RELATIONSHIP BETWEEN DEMOGRAPHIC FACTORS & RELIGIOUS OR SPIRITUAL CONCERNS OF BREAST CANCER SURVIVORS

In the correlation table 22e. there was no positive significant correlation between spiritual or religious concerns and all the demographics factors. Thus, multiple regression was not calculated since demographic factors were not significantly related to associated with spiritual or religious concerns.

Table 22e: Relationship between demographic factors and religious or spiritual concerns of breast cancer survivors

Relationship	Correlation (R-value)	Significance level (p-value)
Age and religious or spiritual concerns	0.045	0.523
Stage of cancer and religious or spiritual concerns	-0.069	0.325
Receiving treatment and religious concerns	-0.024	0.734



4.1.9b RELATIONSHIP BETWEEN PSYCHOSOCIAL CONCERNS & OVERALL QUALITY OF LIFE OF BREAST CANCER SURVIVORS



Figure 14: Relationship between psychosocial concerns & overall QOL of breast cancer survivors

The relationship was explored between the scores of different domains of psychosocial concerns and the overall quality of life. The figure above shows the conceptual model for testing the following hypothesis.

HYPOTHESIS 4:

Psychosocial concerns of breast cancer survivors are significantly related to their 'Overall Quality of Life'.



RELATIONSHIP BETWEEN PSYCHOSOCIAL CONCERNS AND OVERALL QUALITY OF LIFE OF BREAST CANCER SURVIVORS

The below table 23a. shows that there was a significant negative correlation between quality of life and psychosocial concerns including physical concerns r=-0.481, p<.001, social and financial concerns, r=-0.271, p<.001emotional concerns, r=-0.366, p<.001 and spiritual or religious concerns, r=-0.166, p<.001. whereas information concerns were significantly positively correlated with quality of life, r=0.208, p<.003.

 Table 23a: Relationship between psychosocial concerns and overall quality of life of breast cancer survivors

Relationship	Correlation (R-value)	Significance level (P-value)
Quality of life & physical concern	-0.481	<0.0001
Quality of life & information concern	0.208	0.003
Quality of life & social financial concern	-0.271	< 0.0001
Quality of life & emotional concern	-0.366	< 0.0001
Quality of life & spiritual concern	-0.166	< 0.0001

Multiple linear regression (using the stepwise method with p<.05 for entry and p>.10 for removal) was conducted to assess if the two significant correlates with information concerns are also significant predictors (for SPSS output see Appendix 10, table 39).

In the first model, with physical concerns as the sole predictor, R=.483, R-square = .234 and adjusted R square = .230, F (df=1, 203) =61.848, p<0.001, which shows that physical concerns is a significant predictor of quality of life (confirming the significant correlation reported above) accounting for 23% of the variance of quality of life.

In the second model by adding a second predictor of emotional concerns, R=.521, R - square=.271, and adjusted R-square = .264, F (df=2, 202) =37.582, p<0.001, which shows that physical concerns and emotional concerns together are significant predictors of quality of life and account for 26.4 % of the variance of quality of life. It is furthermore the case that in the second model that R-square change=0.038, F-change (df=1, 202) =10.440, p=.001, which shows that when emotional concerns is added to the equation that explains a significant extra amount of the variance of quality of life.



In the third model by adding a third predictor of information concerns, R=.540, R -square=.292, and adjusted R-square = .281, F (df=3,201) =27.625, p<0.0001, which shows that physical concerns, emotional and information concerns together are significant predictors of quality of life and account for 28.1 % of the variance of quality of life It is furthermore the case that in the second model R-square change=0.021, F-change (df=1, 201) =5.892, p=.016) which shows that when information concerns is added to the equation that it explains a significant extra amount of the variance of quality of life.

Model	Predictor variable	Standardized regression coefficient (Beta)	t value	p (Significance level)
Model 1	Physical Concerns	-0.483	-7.864	< 0.001
Model	Physical Concerns	-0.405	-6.259	< 0.001
2	Emotional Concerns	-0.209	-3.231	< 0.001
Model	Physical Concerns	-0.337	4817	< 0.001
	Emotional Concerns	-0.261	-3.868	< 0.001
5	Information Concerns	0.160	2.427	0.016

Model Summary



4.2 QUALITATIVE DATA ANALYSIS

"We are not ourselves when nature, being oppressed, commands the mind to suffer with the body — William Shakespeare"

4.2.1 INTRODUCTION

This section presents the findings of the qualitative component of the current study to gain deeper insight and rich understanding of the survivors' needs that might not be identified through a single procedure. Ten semi-structured interviews were conducted to thoroughly examine the psychosocial needs of breast cancer survivors. The 10 participants who agreed to conduct interviews were selected from the 205 participants who completed the questionnaire in phase one of the study based on their responses. Ten consenting participants were purposely selected who had scored as low, intermediate and high; five Emiratis and five expatriates. To maintain confidentiality, all women were assigned unidentifiable numbers (Table 24).

This phase entails contextualizing the quantitative findings to better understand breast cancer survivors' needs and concerns during the survivorship. At the start of each interview, participants received a summary of the significant quantitative findings from Phase One. They were asked to review the findings and provide their comment on that. No cancer survivor was surprised by the findings, confirming that the results were consistent with what they experienced.

The below table 24. presents ten breast cancer survivors who were interviewed and their demographics (age, nationality) and their clinical characteristics (histology and stage group). Out of a total of ten women, three were between 31-39 age group, three from 41-46 age group and four women had their age between 50 to 62 age group. Within the 31-39 age group, 2 women were expatriates, and one was Emirati. Among the 41-46 age group, one woman survivor was an expatriate and two were Emiratis and in the most aged group which was 50-62 years, there were two participants in each group, meaning two Emiratis and 2 expatriates.

With regards to participants' nationality, ten out of five participants were Emirati nationals and five were expatriates from different countries including Egypt, South Africa, Russia,



Philippines and Iraq. Nine participants' religion was Islam and only one participant was Christian.

To describe clinical characteristics (Histology and Stage) of the breast cancer survivors, seven out of ten had infiltrating duct carcinoma which is the most common histology of breast cancer and three had infiltrating lobular carcinoma which occurs only in 5-10% of population yet both ductal and lobular types are aggressive in nature. Four out of five expatriates had infiltrating duct carcinoma and one had infiltrating lobular type whereas three Emirati women had infiltrating duct carcinoma and two had infiltrating lobular type.

Seven out of ten breast cancer survivors presented with stage II cancer which denotes regional spread means cancer spreads into lymph nodes, under the arm on the same side as breast cancer; three of them were expatriates and four women were Emiratis. Two survivors including one Emirati and one expatriate, diagnosed with localised stage I showing cancer spreading beyond its original location and into the surrounding breast tissue with no lymph node involvement or any other body areas. Only one participant who was an expatriate, had stage III cancer at diagnosis. Stage III is a locally advanced stage showing cancer spreading to adjacent lymph nodes, organs or tissues and in some cases, it spreads beyond its organ of origin. In breast cancer patients, stage III denotes cancer extending to the chest wall and or the skin of the breast and cancer cells may be found in nearby lymph nodes but not to the distant organs. There was no participant presented with stage IV cancer which represents distant metastasis for instance lung, bone, liver or brain

Participant	Age	Nationality	Religion	Histology	Stage
CS034 – Participant 1	46	Philippines	Muslim	Infiltrating duct carcinoma, NOS	2B
CS040 – Participant 2	31	Egypt	Muslim	Infiltrating duct carcinoma, NOS	3A

Table 24: Breast Cancer Survivors	' demographics in Qualitative Phase
--	-------------------------------------



CS048 – Participant 3	36	Russia	Christian	Infiltrating duct carcinoma, NOS	2A
CS070 – Participant 4	50	South Africa	Muslim	Infiltrating lobular carcinoma, NOS	2A
CS109 – Participant 5	62	Iraq	Muslim	Infiltrating duct carcinoma, NOS	1A
CS1333 – Participant 6	41	UAE	Muslim	Infiltrating lobular carcinoma, NOS	2B
CS141 – Participant 7	39	UAE	Muslim	Infiltrating duct carcinoma, NOS	1A
CS159 – Participant 8	44	UAE	Muslim	Infiltrating duct carcinoma, NOS	2A
CS009 – Participant 9	50	UAE	Muslim	Infiltrating duct carcinoma, NOS	2A
CS016 – Participant 10	54	UAE	Muslim	Infiltrating lobular carcinoma, NOS	2B



4.2.2 THEMATIC ANALYSIS

There were three broad areas used to categorise and understand the utmost concerns of breast cancer survivors aiming to achieve the research objectives. The categories were **survivors' living experience with breast cancer**, **psychosocial concerns** including *physical, information, social & financial, emotional and spiritual or religious concerns* and the third was **survivors' expectations of healthcare delivery or support needed**. The emerging themes were identified for each category. Category 1 was living experience with breasts: diagnosis, acceptance, fear, anger. Category 2 included **psychosocial** concerns with sub themes as **"Physical:** complications of treatment and impact on daily life; **Information:** access, type, sources; **Social & Financial:** insurance coverage, financial coverage, charity sources, access to support groups; **Emotional:** relationships support, fear, anger, hope, acceptance, isolation; **Spiritual or Religious:** 'faith value, reliance on Allah or God".

4.2.2.1 Theme 1: Living experience with Breast Cancer

- 4.2.2.1b Sub-theme 2: Living with cancer/difficulties
- 4.2.2.1ac Sub-theme 3: Difficulty in access to health care
- 4.2.2.1d Sub-theme 4: Health decision-making

4.2.2.2 Theme 2: Breast Cancer Survivors' Concerns

4.2.2.2a	Sub-theme 1: Physical concerns
4.2.2.2b	Sub-theme 2: Information concerns
4.2.2.2c	Sub-theme 3: Social & Financial concerns
4.2.2.2d	Sub-theme 4: Emotional concerns
4.2.2.2e	Sub-theme 5: Spiritual/Religious concerns

4.2.2.3 Theme 3: Patient experience with healthcare providers

- 4.2.2.3aSub-theme 1: Patient Satisfaction
- 4.2.2.3b Sub-theme 2: Patient Expectations





Figure 15: Thematic Map: Psychosocial needs of Breast Cancer Survivors







Figure 17: Theme One: Living experience



4.2.2.1 THEME 1: LIVING EXPERIENCE WITH BREAST CANCER

This theme encapsulates breast cancer survivors' living experience with breast cancer whose lives were affected by the cancer. The main theme was categorised into four sub-themes and identified how survivors dealt with the cancer diagnosis and how cancer shaped their living experience with and beyond cancer, difficulties in accessing healthcare and their decision making towards disease identification and disease management.

4.2.2.1a Sub-theme 1: Diagnosis of Cancer

The word cancer can create extreme disruption in the life of almost any individual and can be a threat to one's sense of security that none is ever prepared for such an event. Although the majority of cancers are treatable, for most survivors, the word cancer is a truly unanticipated event that is experienced as a shock, which represents pain, suffering and fear of death that no individual anticipates. Even though one foresees the diagnosis, most individuals' initial response is shock and disbelief (Gorman, 2018).

Participant 1: "It was by sudden"

Participant 2: "Something different I feel it"

Participant 3: "When I was told I had cancer; I was sort of shocked, like I paused. It was not what I was expecting, or I thought I would ever face, but of course I later accepted it as part of the reality. I was not angry or sad but shocked you can say as it hit me from nowhere, something unexpected that you never thought of it"

Participant 6: "I was shocked and devastated to hear that I have cancer"

Participant 7: "It was a complete surprise for me that such a thing happened to me"

Participant 10: "I was shocked, I was shocked really. The doctor told me that I needed to remove my breast; I had no other choice. There was a shock in the beginning when the doctor told me to remove my breast completely"

Some individuals initially respond with denial, which is a protective response against the threat. It is a common reaction which decreases over time. Although some individuals are able to deal



with the situation as they research about the disease and treatment options, an emotional reaction can become evident at any time.

Participant 5: "It was a sense of denial; I never thought myself being in that situation"

In Muslims' culture, it is not uncommon that cancer survivors consider the disease as a will from Allah and accepted as their destiny. The spiritual connection with God does help believers in dealing with the diagnosis of cancer with hope and positivity while providing a strong coping mechanism through faith.

Participant 4: "I was surprised but it was my destiny, it's from Allah and it took it easy"

Participant 8: "When I heard the first time, I was shocked, sure it was a shock for me but later I said it's ok Alhamdulillah, I will pray"

It is not unlikely that some cancer survivors not just react but also feel normal and get on with their life as normal as they were used to in the past. For instance,

Participant 9: "In Oct 2018, I diagnosed with breast cancer with stage II cancer. When I heard the news, I didn't feel bad, I was positive, and I felt normal"



4.2.2.1b Sub-theme 2: Living with cancer/difficulties

The sub-theme 2 describes the impact of cancer on survivors' daily life and provides insight into one's perception about continuing life and carrying on a new normal. It also entails the experience of living with cancer and difficulties an individual faces during daily living. Normally after treatment most cancer survivors return to normal life the way they used to have or ready to do a fresh start at something new yet at the same time feeling sad and worried is something one cannot eradicate so quickly, and it can take time to recover. Many are uncertain about how to come to new terms with the new normal. Adjusting to new circumstances and making changes accordingly can be both physically as well as emotionally challenging.

For some cancer survivors, **one of the hardest things is dealing with the difficulties during survivorship**. For some, it is a drastic change hence, the most difficult to deal with however, the majority of them find a way out to deal with cancer issues with different coping strategies in order to adapt to the new normal.

Participant 1: "It was difficult; it was challenging to deal with the problems; there were some difficulties especially throughout the mechanical treatment; there were difficulties, but I managed all hamdullila"

Participant 8: "After the treatment, everything changed for me; everything changed for me hair, body, everything"

Participant 4: "I had difficulties, but I was able to manage my daily life and tasks as normal"

Participant 3: "I just accepted that it happened, what else I could do; it was sure not easy and sometimes pain and other disturbances were too much for me to handle but I managed somehow"

Cancer patients may see themselves and life in a different way or find that others think of them differently after cancer. Most survivors find that this transition period after being diagnosed with cancer is about finding out what is meaningful for them now. Some become more considerate about their family and loved ones and seek a new purpose of life. It could be a change of perspective about themselves, about others, about values and priorities or about



their own life. Material and little things may be less important for them than concerns about family and loved ones.

Participant 7: "If you compare me one and half year back and now it is different; when I had the treatment, I understood it's my life now"

Participant 3: "I felt my life changed to slow and I became more thoughtful and considerate towards my family than myself"

Participant 6: "I had no idea, it is gonna be that tough; my life has changed including me, my appearance, my feelings, my priorities. Everything has changed to the extent that it is all just new. It has changed my life completely. My quality of life was just, I just lived for the weekend so I could just stay in bed. I just did the most essential groceries in my house and put them like noodles, easy things for my daughter to make for herself. Your life just will not be the same again"

Participant 6: "It's such a big thing, it's like a paradigm shift, it's never the same. Like before, I would like, oh look at this line, you know that was the main concern, health wise. It was so shallow; you got other things now. I couldn't be bothered anymore"

Breast cancer treatment can have an impact on one's body, psyche as well as selfconfidence. Depending on the type of treatment, one's body can change drastically in a short period of time. It is not unlikely that a survivor may discover that nothing in the closet fits right and finding new clothes that make one feel confident becomes a difficult task.

Participant 6: "At this point of my life, as you can see, even the small things will affect your life. You know what I mean. It affects how you dress. Like inside my apartment, I would go without my Sheila (scarf) but even in the balcony, I am gonna cover myself because I just don't want everybody to see"

Participant 5: "Until now, you know I cannot wear a bra. It's very hard for me to wear the bra because when I wear the bra, my pain increases. For example, until now when my nieces and nephews come and hug me, my breast is like somebody hit me or something like this slightly and I feel pain"



Expressing sexuality remains important to all of us including people with cancer. Cancer affects how a woman perceives her sexual identity, for instance, an altered body image. Along with that cancer impacts intimate relationships and it is an issue that many people find it difficult to address. Body changes can occur immediately after treatment or over a long period of time. It can affect the way one thinks and feels about her body image and self-confidence to the extent that one could worry that people will treat a person with cancer differently because of the change to the body image (Fitch, 2020; Rosenberg, 2020; Diaz-Montes, 2010). These concerns about the body could be challenging to deal with (Bolton, et. al., 2010) and can occur at any time (Rosenberg, 2020).

Participant 6: "I have become more comfortable not having breasts but it's difficult. Every aspect of your life changes drastically. Because it's like I would have a conversation with my gardener; I can't do this anymore. This is too difficult. Because my view of myself changed and it was going to change immensely after and coming to terms with my new body, I can tell you at this point that I do not have a partner in my life, but this is gonna impact me having one again"

In many cases, cancer affects a person's work life. Whether a person is able to work during or after treatment depends on several things, for instance, the type and stage of cancer, the type of treatment one has and its side effects, how one feels during treatment and the kind of work a person does. Many people return to "normal life" including work after treatment ends, yet cancer can change this plan for many reasons. Sometimes cancer treatments cause massive fatigue and physical disability or memory problems that make a person unable to work. The decision of how much time off a person needs depends on the support and flexibility of both health providers as well as employers. Most people who want to keep working during treatment are able to do so in some capacity either by adjusting their work hours or working part time whereas some take a break (Lawless, 2009., National Institutes of health. U. S.,1993). Each person's situation is different as not everyone with the same type of cancer can make the same decision about their work. **Continuing work could be very stressful and challenging** for some people as cancer experience is unique for each individual and could be quite challenging. It is best to consider what one feels right for herself and what one feels normal. It is important that health providers must discuss and support employment and work issues with regards to a



person's health since being able to take time off work does not appear to be a legitimate option equally available to all cancer survivors.

Participant 5: "Still I am suffering that I have to go to work, and I still don't have a reason that could stop me to go the work; they just gave me sick leave for just six months and then they let me go and said I am fit but I was not fit; They said, you are fine, and this is normal. What's normal? Normal is when people can do and practice daily life. They can go and come as they wish. This is not normal!

I told my doctor this and she said, you are just saying...of course I am saying, I am not doing something from my mind that I am sick or something. Doctors have to believe what patients are saying. I am not creating these things. How could you know that I am having this, or I don't have one? You won't see inside me!"

Many people after being diagnosed with cancer begin a new chapter in their lives. **For most people, getting back to a new normal does not mean going back to life exactly as it was**; in some areas of life, the new normal is very different (Katz, 2011; National Cancer Institute, 2019). Having said that, there are some for whom things go back to exactly what they used to be.

Participant 9: "I would say none was challenging for me, my life didn't change much; I have no problems as such in daily tasks of daily living, it's normal".

Participant 10: "I would say it didn't change for me even one percent, I went to my work the same after the surgery. I am doing my life activity as before, nothing changed for me at all; everything was like the past. I can't believe it has been one year and half years since I did it, it was so easy Alhamdullila"

Participant 2: "I have been alright during and after the treatments hamdullila"



4.2.2.1c Sub-theme 3: Difficulty in access to health care

The sub-theme 3 describes the issues and challenges that most cancer patients face, particularly newly diagnosed. The link between poor access to care and poor health outcomes is well acknowledged however, the reasons for inadequate access are not well understood. Accessing appropriate cancer care is the most common barrier to cancer patients. Access, as defined by the Institute of Medicine (1994, pg. 4), is the timely use of personal health services to achieve the best possible health outcomes. This definition of access encapsulates both the use of health services as well as the quality of such services. This sub-theme mainly focuses on difficulties faced by cancer survivors in accessing cancer care as well as concerns about care coordination with healthcare providers. This includes attributes of the healthcare providers, for instance, lack of knowledge about cancer care or treatment. Identification of such barriers is critical in order to address and implement strategies that could help on both individual as well organisational level.

Some patients travel from a far distance to attend scheduled appointments for the consultation with their doctors or to receive treatment; this adds additional stress and burden as fatigue and pain are the most common concerns among cancer survivors. The remoteness and the lack of availability of specialised services could be the major obstacle in accessing the appropriate cancer care which either result in delayed diagnosis or could add both physical and emotional toll on particularly vulnerable cancer patients.

Participant 1: "I am very far from the hospital (Al Ain); Sometimes I cannot reach the doctor".

Participant 8: "You know, I am driving from Abu Dhabi to Al Ain and it's considerably very far for a patient".

According to the literature, physicians' treatment recommendations could be influenced by a number of factors including physician age, gender, specialty and his or her belief in efficacy of care (Mosadeghrad, 2014; 2013; Cooper-Patrick, et. al., 1999; Levy, 1985). The content of physician communication also varies according to patient characteristics including age, income, education, race or ethnicity and expected prognosis. Variation in offering different



type of treatments can be explained by physician specialty, training and experience while delaying in adoption or lack of compliance with practice guidelines can limit access to recommended cancer care. Accessing appropriate cancer care requires establishment of trust between doctors and patients as well as clear communication. The quality of doctor-patient relationship is critical and can affect clinical outcomes, adjustment to cancer, patients' quality of life and satisfaction with care thus communication gap can result in lack of trust and could be highly worrisome to some patients.

Participant 2: "They did a wrong operation on me, wallah. They didn't remove the lymph node under my arm. "They took like half from my breast and the doctor said, khalas, you are ok." "My doctor, Dr. XYZ was so much angry because it's wrong operation and they should have removed everything."

"They took too much flesh from my left side"

Participant 5: "All doctors are like take it...take it...how can I take it? Even I asked them, did you have a patient like me?" This happened, and they said let's try. What's let's try?"



4.2.2.1d Sub-theme 4: Health decision-making

Sub-theme 4 describes the decision-making process by cancer patients about their treatment choices. Decision-making is one of the most complex skills required in oncology associated with decision making by both doctors and patients (Glatzer, 2020). Literature suggests that there are many psychological factors involved when patients consider treatment options, including prior experience, quality of life during or after treatment, life expectancy, opinion of their care provider and preference of the patient's family (Glatzer, 2020; Jin, et. al., 2008; Adachi, et. al., 2007). Each patient brings her own personal values and beliefs to the decision-making process. One big concern for patients diagnosed with cancer is whether there is time to research about choosing an oncologist and treatment facility or getting a second opinion before starting the treatment. Similarly, sometimes patients prefer to participate in deciding whether to undergo different treatment for instance, which chemotherapy protocol is likely to benefit the patient. Some patients go to private facilities to do additional tests or to get a second opinion and pay to get an assurance that the choice of treatment is right.

Participant 1: "I went to Al-Zahra hospital in Sharjah (Private facility) before coming to Tawam"; "I decided to go to Tawam Hospital to complete my treatment". "But now I stopped chemo". "I finished chemotherapy, I finished surgery, I finished the radiation and now I am on hormonal treatment"

Participant 2: "I had an operation and radiation"; "I didn't take chemotherapy, only radiation".

Participant 4: "I had chemotherapy"; "I had surgery"; "I took radiation for a month"

Participant 5: "I did the radiation; I had the surgery; I was taking this hormone therapy; "we need to do the hormone therapy which I didn't complete"; "I had a radiation; surgery and I started this hormone therapy then I stopped it"

Participant 10: "If I didn't remove it, it may have gone to another place and I didn't know what would happen, so I am really happy that it was removed. I don't regret anything"



Most cancers are symptomatic. Patient pathways to presentation of those symptoms to healthcare professionals and its initial management are key determinants of outcomes in cancer (World Health Organisation, 2017b). It is possible to improve the prognosis and reduce the need for extensive and aggressive treatment of cancer patients who seek timely care and detect the symptoms at early stages (World Health Organisation, 2017c). There could be a number of psychological or socio-cultural factors that influence patient decision making about seeking or delaying timely care (Tetteh & Faulkner, 2016; Elobaid, 2014). In addition, sometimes, family situations could have an effect on an individual's health priorities. Understanding these factors are important to address and illustrate the underlying problems in patient delay (Saeed, et. al., 2021).

Participant 2: "Then I said, it happened but it will go; But it took time and there was my sister's wedding and then I said not now till my sister's wedding is over. You know then after that I went to see the doctor here in Sharjah"

It could also be socio-cultural context or a behavioural delay that some patients just do not take the symptoms seriously until it becomes distinctive or aggressive.

Participant 2: "When I went there to see the doctor without an appointment and thought they would see me, and they will say nothing, and I will return"

Participant 3: When I was taking care of him, I knew I had a lump but when he passed away after that I decided to go and see the doctor; "Before that I was ignoring my condition"

People facing serious illnesses such as cancer have a great stake in the decision-making process. Patients' trust in their physicians is the major concern when it comes to decision-making. It is considered essential for high quality and effective cancer care, and it is well acknowledged that trusting patients are more satisfied and more involved in the diction-making process (Paduca, et. al., 2021; Ruhnke, et. al., 2020; Mahlich, et. al., 2019; Xu & Wong., 2017; Krist, et. al., 2017). Informed decision making is central to patient-centred care where the physician provides the information to the patient and then the patient makes the decision. Patient preferences and values must be considered and respected by the healthcare providers with regard to the outcomes of options.



Participant 2: "I went to Tawam to see Dr. XYZ and she did the surgery for me". "She (Doctor) said it's better to remove all and the second one; even if you don't have cancer in that, remove it". "I said remove both (breasts). She said you don't have cancer in the other one and I said, it's better for me to do it both"

Participant 10: "I didn't take any chemo; I just had my surgery done". "The best thing that I did it Alhamdullila because if I didn't do it, I would be suffering from it." "You know, if I had to choose chemotherapy, I would take the tablet (hormone therapy) not chemotherapy, so it was good for me. I didn't take chemo or radiation, so it was just fine with me. It was just the best that I didn't do these two treatments. I am satisfied that I am on tablets/hormonal treatment, it is easier."

Participant 8: "I took my treatment, it was concentrated chemo; I changed my treatment and when I changed my treatment, I took radiation and my treatment together. I had surgery, I removed my breast and it's difficult for me, but I accepted so I agree for everything."

"I told you about the medicine, the hormonal treatment, I didn't feel anything. It's difficult for me."

It is not uncommon that many patients feel that they are not active participants in developing their cancer treatment plans (Institute of Medicine, 2011, pg. 7). Few people realise that they can ask questions from the health providers and even fewer know what questions to be asked during patient-doctor consultations. Some consider support groups and seek support through forums where they can find other cancer survivors to share their experience. Most cancer patients need extensive and detailed information to recognise and appreciate the effect treatment has on their lives. Thus, it is important that healthcare providers understand patients' priorities and perceptions of their treatment which may change over time and include them in discussions during decision-making.

Participant 6: "So, if there is a forum, if there are people who have gone through the same things, they can relate. There should be a safe place to speak about it"



Technological advancement has provided cancer patients an interactive platform to change experiences of living with breast cancer. There are growing numbers of cancer survivors who want to connect with other people sharing similar experiences and support each other. Forums and community support groups are created to provide a safer place for cancer patients to discuss their concerns so survivors could connect with one another and support each other.

Participant 6: I put myself into the forum mainly because of the effects chemotherapy has had on me". "I associated myself with the forum."

Being diagnosed with breast cancer can be both confusing and daunting particularly at earlier stages when less is known about the disease. And at that stage peer to peer information sharing becomes very meaningful (Allicock, et. al., 2017). People may feel more comfortable questioning their peers within the informal environment than health professionals within time constrained consultations in a hospital setting. Doctors' time is also limited thus not all concerns could be addressed by the patients and many could be left unanswered. Limited time and resources also tend to shift the care of patients to alternate health professionals, or they could be left unattended.

Participant 6: "When you are first diagnosed, if you speak to a doctor in her office, her time is gonna be limited because there will be more patients waiting outside her room"

"There was a lady, and she was on her first session when I was on my 3rd or 4th session. She came from Fujairah to Tawam. She asked me how it feels now where you are and what you need to look for and I told her, listen, look for this and this and this and then after a couple of sessions down the line, we landed up together in the same room and she said the same thing to her. She said she did on Wednesday. She said from Sunday to Thursday she couldn't do anything so what I had told her so basically prepared her for that, so we need that"

In these forums people share information on issues that most health professionals have no direct experience of such as how it feels when one undergoes a treatment and what it feels like to live with a change in body image after surgery or chemotherapy treatment. This first-hand knowledge of experience of breast cancer is the key to empower cancer survivors, support decision-making and help them to regain some control.



Participant 6: "When those people who have cancer, if they can come around together and tell, listen, this is what you need to look for and this is how you're gonna feel"



4.2.2.2 THEME 2: BREAST CANCER SURVIVORS' CONCERNS

Theme 2 encapsulates psychosocial issues as described by the breast cancer survivors. This theme was sub-categorised into five domains including survivors' concerns about physical, information, social & financial, emotional and spiritual or religious domains.

4.2.2.2a Sub-theme 1: Physical concerns

The sub-theme 2 summed up physical concerns associated with the disease and its treatment described by breast cancer survivors. Women with breast cancer face many physical challenges associated with disease and mainly treatment such as changes in weight and in body compositions, menopausal symptoms, persistent fatigue as well as psychological symptoms associated with the alteration in body image resulting from either surgery, adjuvant therapy or side effects of chemotherapy or hormonal therapy. The majority of the women experienced higher physical needs and had more concerns, the most challenging to deal with particularly pain after surgery and fatigue (Bardwell & Ancoli-Israel, 2008; Amiel, et. al., 2016).

Participant 1: "My major concern was physical appearance like, no hair, skin change".

Participant 5: "Pain, physical concerns for sure. Pain was the most difficult."

"This was the biggest challenge because the pain, pain on the chest cannot let you sleep."

"They told me that pain will go away after one year. It has been more than one year, and I am having that pain, seriously unbearable pain."

Participant 6: "Physical concerns and emotional concerns were the most important issues I had to deal with cancer".

Participant 8: "Physical concerns were the highest".

Participant 10: "Maybe the physical concerns were more challenging in terms of like surgery made something happen to me, it was dangerous to my life, I thought like this".



Participant 2: "The difficult part was after the operation. I had pain and sometimes I was about to vomit. I suffered from this part so mainly physical".

Participant 3: "As I said, I did not have much difficulty but of course after surgery I suffered pain. Pain was the main problem".

Participant 4: "I got pain in my bones and body; I had pain in the area where I had surgery, but it wasn't that much". "Pain in terms of I do not take the tablet, the yellow tablet, I do not take one". "The intensity of pain and cramps; having all your periods and childbirth is just nothing like that swear to God.". "Other was fatigue, I couldn't leave my pillow...that was the level of fatigue"

For some women, their ability to carry out activities of daily living such as driving or daily jobs becomes very challenging because of the effects of the medication they take to manage their disease. They feel weaker to walk or drive after chemotherapy sessions. For some women employment or particular type of job becomes a struggle.

Participant 8: "After I had a session, I drove back to Abu Dhabi to Al Ain. OK, I felt so much pain and tiredness, but I did manage". "I have too much pain and you know I work in sales, and I should stand for hours but hamdullila. I try to manage not that I can manage but I try to manage every day."

Cancer affects each one differently and every one-person's experience is unique. Although pain was the most commonly reported concern among breast cancer survivors, yet the level of intensity was described differently by few. Some women suffered with mild pain only and their physical concerns disappeared with time or were not extreme.

Participant 7: "Mainly fatigue was the biggest challenge. I didn't have any other problems from any treatment. I was very lucky"

Participant 9: "I have had mild pain in my knee, not too much".

Participant 10: "There was pain but after surgery two to three weeks, everything was fine. It was so easy, no difficulty like I didn't do anything." "Like two months, there



was little difficulty in activities, say for walking, sitting, you know but after that it was ok, nothing changed".

Some women described unique physical concerns other than pain such as problems with eyesight, or issues with the immune system which affected their daily lives to the extent, for instance.

Participant 4: "My eyesight dropped, and I couldn't see particularly at night. Mainly the left eye. At night I had all blurry vision"

Participant 5: "I am having the flu frequently. Most days I have this flu. There is no day when I don't have congested nose. My immune system became down, and I told them, and they said everything is fine. How? I wasn't like this!"

Sleep disturbance, poor sleep or insomnia are common concerns of breast cancer patients associated with psychological and medical consequences. In breast cancer patients, the phenomenology of sleep is that the challenges faced by cancer patients may contribute to or result in sleep disturbances or could increase a risk of insomnia. Pain has been well acknowledged as one of the eye's most frequent negative symptoms experienced by cancer yet other comorbid conditions could largely result in difficulty in sleeping in breast cancer survivors.

Participant 3: "I was having a sleep disturbance"

Participant 5: "Whole night I couldn't sleep because of my congested nose, how could I do daily work? Also, I am having asthma, so it affects my chest so until now I am suffering from this"

Participant 6: "Inability of sleep, I was just awake. I would sleep one hour and then awake"

Most breast cancer survivors experience body image dissatisfaction. After diagnosis and surgery most of them experience changes in their appearance and become dissatisfied with their bodies for many reasons, for instance altered body image could affect women's sexuality and could be quite traumatic psychologically. Hair fall, skin and weight changes, all these are


the most common symptoms experienced by breast cancer survivors. Such symptoms could have a significant emotional impact on the survivors' quality of life. Beauty is often based on the outer appearance and when it is altered, it can have a negative impact on one's life. Losing hair could be losing self-worth. For some women loss of hair and skin changes are bigger concerns as it affects their self-confidence and could be quite devastating to deal with (Iddrisu, et. al., 2020; Rezaei, et. al., 2016; Preston, 2010).

Participant 6: "It affects your sexuality"

Participant 4: "My hair fell"

Participant 8: "Everything changed for me, hair, body, everything".

Participant 1: "The main problem I faced was the outlook and changes in my face and my skin. My eyebrows started falling during the treatment. There were no eyebrows, no hair"

"All the side effects that I faced were in the chemo. The main problems I faced with Chemo treatment were mainly physical and outlook change"

"The side effects were on my skin...darkness on my face, on my arms, on my neck"

Participant 6: My hair fell off. I was just showing my nephew the picture of my pillow after my second chemo. And my pillow, it was just full of hair, that was the amount of hair I lost. I just went to the bathroom, I didn't have the shaver, I just put the Veet all over my head"

Weight changes after breast cancer diagnosis may have a negative impact on quality of life (Chen, et. al., 2011). Weight changes, either weight loss or weight gain are common during cancer treatment (American Cancer Society., 2021a). Weight loss is typically due to poor appetite or nauseas which could be the side effects of chemotherapy. Some weight loss could be quite drastic yet most women with breast cancer gain weight during treatment due to change in hormone levels or menopause which could result from chemotherapy.



Participant 1: "My appetite was very bad, I couldn't eat, I lost weight, I lost around five kilos"

Participant 4: "I lost weight".

Participant 6: "Because of the side effects of chemo I lost 7 kilos".

Participant 9: I have become fatty. Before I was sixty-five kilos and now, I am seventy-two.

Many survivors have to deal with the effects of cancer treatment. Some adverse effects last a short time and have little influence on survivors' quality of life, yet others can cause long-term complications and add to increased suffering. Some women described their health as "very bad" whereas others mentioned the problems and suffering of daily living due to adverse effects of cancer treatment. Nausea, vomiting and headache are some symptoms that result in treatment being needed immediately.

Participant 5: "I had this period for a long time", "I was having my periods more than twenty days, so my health became very bad"

Participant 6: "The side effects of the chemo, to use a bad word...just terrible it is. Side effects are long, it's like, one time I found myself crawling to the bathroom like I have diarrhoea". "That was so like today I am going to the bathroom and after that I couldn't go back to it and it escalated so much the first time, I was just floating. My whole face showed it. I was just in my bed; I couldn't do anything, just couldn't do anything"

"I did start my chemo on Wednesday, and it was only on Sunday when I woke up, I started vomiting and diarrhoea and my stomach was so tortured"

"And taste in mouth, anything I eat, it tastes off, I couldn't eat anything and whole six months of chemo because of that"

Participant 8: "I also had lots of difficulty with my stomach, my hand, my mouth but after sessions I went to work". "My hand got too much black and my colleague told me



(participant name) why your hand is like this, you want to wash your hand and I told them No, it's from the treatment"

"In the first session, I was tired for sure. After three days, I was a little bit OK but because I had so much problem with my stomach, I was vomiting, also my mouth, for sure my hair fell"

"Sometimes I went to emergency, three times because I had too much headache and vomiting"

"I had some days, I couldn't stand because I had too much vomiting, I took water, I vomited, I had too much headache and I went to emergency maybe three times"

Participant 9: "I had lots of side effects mainly vomiting but I managed"



4.2.2.2b Sub-theme 2: Information concerns

Cancer patients have better health outcomes when they are more informed about their disease and more involved in their treatment choices. Information provided to cancer patients relieve anxiety and help them make informed treatment decisions. The amount and the type of information cancer patients want depends on their unique information needs which vary according to stage, age, culture or beliefs. In addition, information satisfaction is a prediction of quality of life. Breast cancer survivors were asked whether they had any concerns, or they faced any difficulties in obtaining information. The sub-theme 2 presents those concerns described by the breast cancer survivors during their interview process.

There was a mixed response when participants were asked about their information concerns; most women were fully satisfied with the information provided to them and mentioned that there was no difficulty in obtaining information, yet few showed their concerns about information delivery from healthcare providers.

Participant 6: "Yes, I was fully informed". "I was fully informed about the side effects of chemo; I remember my doctor told me all"

"I was fully informed about all the symptoms and side effects"

"I was also informed of all the services available if needed"

Participant 7: "I was fully informed, even more I needed to know".

Participant 9: "I was given all sorts of information and I had no issues with that".

Participant 10: "The doctor told me everything, every single detail about the treatment, everything to do. All the information I got and the details also"

One of the participants appreciated today's world where all sorts of information is easily accessible over the internet. From disease to treatment and their side effects could be researched using Google or YouTube. However, some information is specific to an individual for which they need expert's help.



Participant 1: "There is no difficulty if you want to know anything. Just google and YouTube and you will know everything. The Internet helps me"

"What should we eat and what we shouldn't eat? That is what I needed most"

Participant 7: We are living in the world of information. We can open Google and check on whatever we want.

Some women informed their close interaction with the healthcare providers and showed their doctors' involvement to keep them up-to-date and maintain regular follow-up with them. Most women mentioned they were fully informed by their doctors. They found the information clear and had no difficulty in obtaining the information. They showed their satisfaction by appraising the healthcare providers.

Participant 2: "Dr. XYZ, she explained everything to me and also Dr. XYZ a ". "I had no problems with obtaining information at all"

Participant 3: "I was fully informed multiple times about the side effects of medications". "I was fully informed of the symptoms I had"

"The Healthcare department was calling me regularly and asking me how I am doing, how I am progressing"

Participant 4: "They were clear, and I had no difficulty in obtaining information. I was told everything"

Not just the type of information, it is also equally important for some women from whom the information is provided. Some only consider the information from the doctors as authentic. It is important that healthcare providers must provide detailed, accurate and satisfactory information to their patients as they rely on them more than anyone.

Participant 7: "I didn't know anything and you even offered me all the information which is enough for me but it all must be for my doctor".



"If I know something, it will change my life but no, it will not change my life. If my doctor doesn't know some information, it will affect me but if I don't know some information, I don't think it will change my life"

On the other side, not all were happy with the information provided. Some did mention their concerns either about lack of information or information they received was not sufficient to fulfil their knowledge or needs.

Participant 8: "In the session, they gave me some medication and I slept. I was not informed"

One participant had a different opinion than the rest where she showed her disappointments towards her doctor that although they provided information, their answers were not sufficient for her. The doctor was not able to answer all her queries and consequently her health deteriorated. She complained further that the doctor only informed what they learned but their knowledge was not broad enough to satisfy her queries. In summary, some cancer patients seek specific information, for instance whether the condition is unique to the patient or others may have the similar consequences of the disease. This information may help an individual to prepare in advance to adjust to the outcomes of cancer whereas a lack of information may negatively impact their quality of life.

Participant 5: "They think I don't have anything, so their answers were not convenient to me at all".

"Lack of information from the doctors, this effect"

"Some words were related to me but others, not especially my health was going down and they didn't answer me"

"I am asking them to give me information. Did it happen before, is it something naturally happening? Do you have other cases or not? So, in that situation, they just want to do what they learned"

Cancer diagnosis brings with it many challenges, anxiety and uncertainty which adversely affects a person's emotional and psychological well-being. Unfortunately, it is not uncommon



that while cancer patients are generally well treated medically for their disease, their healthcare providers fail to address cancer related psychosocial or emotional aspects in a helpful manner or worse they are not screened routinely for this aspect of cancer.

Participant 6: "They go very efficient in everything but the psychological and emotional side, nobody said you are gonna be depressed afterwards"

"If you are gonna see the mirror for the first time, you are gonna break down in tears. Nobody prepared you for that"

Some women do individual research and build their own knowledge through various channels, yet they are not fully prepared to face unforeseen circumstances. Some are given information in bits and pieces thus while encountering an adverse effect, they have no idea what to do since they were not fully informed or fully prepared for that.

Participant 6: "Nothing prepared me for that. Yes, I prepared myself for that but not to that level that will float me"

"I had read, I had researched but I did not know that it was such a fine balance that you had to run between diarrhoea and normal bathroom needs that need to be there.... that was not there"

Some cancer survivors may have very unique queries particularly when they are fully informed about their circumstance for instance about the course of the disease however, there are unanticipated situations where a person is not prepared.

Participant 6: "Having my double mastectomy, when do I stop it? Yes, I was told, and I remember everything that the doctor said at that time but there was a time when I was like what do I do now?

"Once I had to come to the ER on Friday evening. After 9, I had to call my friend, I had a fever. I just showed that, so I went to him. But it wasn't covered where to go or whom to speak to if I have any social needs"



Participant 7: "I don't remember but I don't think I was told if I face such and such issues, I can go to this person or that person, but I think this is something you need".

It is also important that healthcare providers provide clear and detailed information regarding all aspects of disease in order to prepare patients for expected as well as for unexpected events during the course of the disease. One woman appreciated the information that was shared by her doctor about the disease and possible side effects of treatment, yet she was not prepared to what extent she had to deal with side effects. She also emphasized that she wanted to know more about what could go wrong instead of what will go right. She stressed that she needed clarity on such issues from her doctor as a survivor could have more queries which cannot be answered in a limited consultation with doctors considering their time constraints.

Participant 6: "My doctor, my oncologist, she explained to me what's going to happen but as I said I didn't own it. I didn't realise that's gonna be you that's gonna happen to you like other people". "Side effects, yes gonna be there but to what levelI cannot say questions answered but addressed"

"Tell me these things, I don't want to hear that you are gonna be OK cause you are not going to be OK. So, be clear like my doctor said chemo will not but the side effects throw you"

"You know all my questions were answered, I have to say that but like I have had time to think about it and I have more questions so who's gonna follow that, it cannot be the doctor because she has other patients to see"

Being an expatriate, itself could be a challenge for a cancer patient as they are not quite familiar with the resources available in that country. Healthcare providers must ensure that all cancer patients are fully informed about the available resources including groups and charities that could help expatriates. One woman talked about this important aspect and shared her concern of being an expatriate.

Participant 6: "Because I am not from here, I do not know, I can go there or go there".



4.2.2.2c Sub-theme 3: Social & Financial concerns

Cancer not only takes a physical and emotional toll, but it also has a huge social impact on many aspects of life such as personal relationships, social interactions, finances or employment etc (Banegas, et. al., 2019). The physical and psychological problems can be exacerbated by or result in serious social problems. The cost of health care, lack of insurance, low income or employment issues all could contribute to major financial stress. However, social and financial issues are beyond the capabilities of healthcare providers yet addressing these issues are an integral part of a good quality healthcare system (Golics, et. al., 2013; Goldman. 2010). The sub-theme 3 describes the impact of social and financial difficulties or concerns on women with breast cancer.

Financial worries were mentioned mostly by expatriates since all Emirati women were covered by national insurance called "Thiqa". The UAE government provides Emiratis full health coverage; thus, no financial worries or concerns were found among Emirati cancer survivors.

Participant 1: "I had no financial problems". "It is covered"

Participant 2: "No, I had no financial or social issues; is covered by insurance alhamdullilah".

Participant 3: "No financial problem; my healthcare was covered by insurance"

Participant 4: "No, I had no issues with finance, it is covered, and they did everything"

Participant 5: "No, I didn't have any financial problem because this is all from the government"

There were two women who did not mention any financial concerns with regards to insurance or health coverage as their insurance provided them full health coverage.

Participant 9: "I had no problems financially because my treatment is covered by Daman (insurance)".

Participant 10: "I had my insurance, and it was totally fine. I had no problems anything alhamdulilla"



According to UAE law, providing health coverage is the employer's responsibility which could either be basic coverage or enhanced coverage to cover chronic health conditions with or without co-existing issues. It also reflects the employment status of an individual; most highly paid jobs provide enhanced health coverage. Average employment does not cover entire health issues thus an expatriate could face serious health problems especially when one has extensive disease or other comorbidities (Dubai Health Authority, 2020; Jones, 2000).

Participant 6: "Dr. XYZ will tell you the issues I had with NEXtCare which is my provider from the Ministry of Education".

"They said that I have to pay ten thousand and I said I can't do that...I can't afford it"

Some patient's health coverage was not covered by the insurance thus they were supported by a mandate issued by the UAE government to support cancer patients (mainly expatriates) by providing financial support to those who could not afford and have limited financial resources.

Participant 6 also mentioned her difficulties in delaying her treatment session because of delays in the approval process from the insurance which sometimes resulted in her missing sessions until approval was granted. She mentioned her difficulties and frustration in chasing up and following healthcare providers for approvals so she can get the treatment.

Participant 6: "Sometimes I would come, and the approval will not go through entirely so a couple of sessions I had to miss. When I had started my chemotherapy, the first session had to be postponed because they didn't approve it. Lot of the time, initially the first few weeks of my chemo, I didn't do it. After that it had to be postponed to next Wednesday because it wasn't approved. I have had a lot of issues in the beginning".

"Every time I come for chemo; I have learned that the day before I had to call them to ask them if it was approved. The ladies here, they are putting the request and then tell me that I put it in ten days ago and say they are still working on it. It is ridiculous! They said go to the social worker and speak to the social worker and they asked if I had insurance and I said yes, I have insurance and they said then you cannot have it, go to the principal"



Participant 8: "My insurance is very bad, now I took treatment under mandate because my insurance doesn't cover anything"

"For my surgery I had money problems and I couldn't manage the money for surgery"

Another participant mentioned that having a mandate helps cancer patients and provide financial support, yet it only covers the existing condition. She had another cancer and the mandate only covered one condition thus it was somehow helped but not fully and she had to bear expenses to cover other health problems besides breast cancer which added to her financial burden. She also mentioned the ease of being treated by one physician at a single facility"

Participant 7: "I have a SEHA mandate. It is really hard for me for example in my case"

"Because I do not have a mandate approved from Abu Dhabi for gynaecological cancer. They said, "If you have breast cancer, we can help you with that, but we cannot cover this gynaecological cancer care"

"It is most comfortable; if I get the gynaecologist here, I cannot pay two thousand dirhams for a gynaecologist. I did this once here, but this is, in my job, it is quite expensive. So, this is the challenge I faced here, because I would like to be here, but I cannot afford this amount"

Having financial support is a big relief for expatriates, particularly those who have average employment status. Even some of the support made a difference in their lives. Women with breast cancer expressed their gratitude towards the UAE government for such support and appreciated the consideration as it saved their lives. Some healthcare facilities also helped getting cancer patients their health coverage by making the patients eligible to receive the required care for instance, admitting them as in-patients as out-patients are not eligible for some of the essential treatment and coverage is only reserved for outpatients.

Participant 7: "My gynaecology is not covered here

"It's not going to affect me as such. I actually really appreciate all that has happened to me, the chemo, the doctors, all covered, it's really big help and it saved my life"



Participant 6: "They (insurance) did not cover because that insurance wanted to block patients so they kind of wangle it so it would be inpatient so they (hospital) can give it to me in the hospital and I didn't have to pay for it. So, the hospital was very accommodating"

"But to be honest, I would be at home, and it would have been coming out of my pocket, I am extremely grateful that I am here in the UAE"

Fortunately, most women manage their psychosocial distress quite well provided they have an available support system, including spouse, family or friends. In addition, professional support from healthcare providers with intensive professional guidance and referral to appropriate resources or support groups are critical to avoid psychosocial distress that unfortunately comes along with breast cancer. Most women anticipate positive relationships with their loved ones to gain strength and to maintain their psychological wellbeing.

Family support plays a critical part to deal with cancer. Most people manage the circumstances and find a way to deal with cancer because of the strength they receive from their families. Talking to family members openly reduces cancer distress, fear and anxiety caused by cancer (Vrontaras, 2018; National Cancer Institute, 2021)

Participant 1: "I talked to my husband; That was from my husband, I was strong"

"I have found so much support from my family, from my relatives that's why they gave us strength hamdullila"

"I found a way to deal with it through my faith and my family support"

"I managed all with my family and my doctor"

Participant 2: "My two sons and daughter, they are with me. They told me, do like this, don't eat like this, put me in a good way you know for my food, my exercise. They were with me, and I managed"



"Mainly from my children, my family and my sisters. I had lots of support from my family. I managed well with their support"

Participant 3: "My sister-in-law had cancer, so I was comfortable talking to her and get the support I needed"

"I spoke to my family, and they were all with me to support me"

"My husband was very supportive though, and I had no issue talking to him about it"

Participant 3 described she was affected positively by the sister-in-law and was comfortable talking about it which gave her support.

Participant 4: "Everything and everyone was good and supportive. I had no emotional or social concerns hamdullila"

Participant 5: "I had all the support from my sisters"

Apart from family, having a good social circle and having good friends do not just ease emotional distress but some really understand the pain and difficulties in performing daily tasks and provide as much support as possible.

Participant 6: "I had lots of friends assisting me"

"Everybody was most helpful"

"I am not lying, I swear to God, every day I had lunch, I had dinner. It would be left at my door. Every day she sent somebody from the cleaning company, every day my apartment got cleaned because I cannot, I must not do anything. Alhamdulilla, she took me like her sister, and she did it for the whole of my period when I was recovering till I did it and I said Amina, I can do it and only then did she stop"

Participant 9: "I am happy with my family, and I managed all".

"My family has been very supportive, and they supported me all"

Participant 10: "You know my family; they were there for me".



"My daughter and my son, really. Maybe they encouraged me so much to do. They were more scared about it; I was like I will do it anyways, but they were scared about it. My husband passed away seven years ago, but I got all the support from my family"

Employment concerns are equally challenging as cancer impacts one's physical ability thus maintaining a job could be quite challenging. Support from employers is crucial. Employment concerns were only mentioned by two women but positively. Their supervisors or managers were empathetic and went a long way to support them, so their job was not negatively affected by their cancer (Sun, et. al., 2016., de Rijk, et. al., 2020).

Participant 4: "My work was OK and flexible, and they knew my condition and allowed me to leave at twelve because I couldn't stay more as I got tired. They told me; you can leave early"

Participant 6: "My principle was assisting me. I am a teacher, so she took away some of my classes, so it won't drain me out. She put another teacher with me in the class so if I can't do anymore, she takes over. I am blessed with a very good principal alhamdulilla and she has been immensely supportive".

Participant 7: "My director in the school, he knows what happened to me. Actually, he gave me one month off when I started my chemo so I can continue my job"

Participant 8: "Because my manager told me, you can sit no problem; your salary is OK but for me I said, No, I should go to work".

Participant 5: "I didn't have any support except the people at my work who helped me".

Participant 5 appreciated her only support from her workplace yet mentioned her disappointment to receive appropriate support from healthcare providers.

"The doctors didn't support me in this"

Yet, every person's experience with healthcare providers is different and some receive quite a good support from healthcare providers.



Participant 7: "I do remember, the first chemo, there was a Syrian nurse, she did my first chemo. I cried; I don't know how to explain. I felt there was something in my body, I don't know, I felt like I wouldn't wake up. She was standing right next to me and said why are you crying, are you missing someone. Everything will be OK"

"All the doctors, nurses, they have become more concerned than my friends"

Most of the women mentioned immense support from their family and friends yet every person's life experience is different. There were some participants who did not receive any support from their family or friends. They expressed their disappointments and difficulties due to lack of support from their very close people. Unfortunately, they mentioned adverse reactions from them.

Participant 7: "You know through this sickness, almost all my people changed, people whom I was with and thought they are close, and I thought they are my relatives"

Participant 8: "I have so my sister brought my mother. This gave me strength. My husband ...NO, he is X (showed the cross sign). he didn't stand up with me. From my husband, No.... there was no support. My family gave me this, not my husband. I have two kids and my husband, but he didn't support me emotionally, he didn't support me financially. It is difficult for me, my husband has money, he could have given me money for my surgery, but he didn't give me. He let me go to Hilal Ahmar (Charity) like this to manage my surgery. It was difficult for me"



4.2.2.2d Sub-theme 4: Emotional concerns

Breast cancer experience requires a person to consider an array of psychosocial and emotional demands, mainly emotional. The experience of breast cancer diagnosis and treatment may result in considerable emotional distress that brings fear and challenges about a woman's identity, self-esteem, body image as well as relationships. Although most women adjust well to these commonly felt difficulties particularly in the absence of advanced symptoms and if they do not experience recurrence, the negative impact of breast cancer can be significant. Family, relationships and other support groups are protective factors to avoid emotional distress. Understanding and addressing emotional concerns is crucial among these vulnerable populations as the risk of distress can seriously impact their quality of life (Karunanithi, et. al., 2018). The sub-theme 4 describes emotional concerns among women with breast cancer and their intrusive thoughts about their breast cancer experience and how it had an impact on their well-being.

Although breast cancer diagnosis affects a person from many different psychosocial aspects, emotional concerns are the highest and the most challenging among this population (Conley, et. al., 2016).

Participant 7: "Dealing with emotions, maybe, was the biggest challenge for me because you need lots of strength and it is very important for you to have supportive people"

Participant 6: "Emotional concerns were the most important issues I had to deal with".

Participant 8: I would say physical and emotional.

Being hit with news of having breast cancer can be completely overwhelming. A diagnosis of cancer itself can be traumatic and can trigger a variety of emotions. It can affect the emotional health of patients and their families. At first, it may be hard to accept the news and to believe it that some may not feel real or feel blank or stunned while others feel frightened, angry, sad, worried or express mixed emotions. Dealing with emotions may not be easy at first. It is difficult to identify and accept. For some it is hard to accept the reality and some just accept it right from the time they got the news.



Participant 1: "I was not happy because there were some worries and things to be worried about but you"

Participant 2: "I was thinking all the time if they would tell me something bad, what I will do"

Participant 3: "I was a little depressed of course it's not good news but later I accepted the reality"

Participant 4: "I was worried and of course unhappy"

Participant 6: "It also had a psychological effect; I became very depressed". It was like I was devastated"

"I just sit on my muslaah (prayer mat) and cry... cry in the middle of the night"

Participant 7: First, I remember, I was sitting down and crying". "Of course, it was angry, like sadness, fear, it was like no emotions at all. So, all, emotions or mixed emotions"

"Before I prayed, and I didn't cry but this time I pray, and I cry so that I can say it's different"

Participant 8: "It's difficult to accept"

Some women feel alone or isolated after cancer diagnosis and treatment. Sometimes people do not know how to deal with cancer and the emotions it brings up. Feeling of being isolated could be for many reasons such as one tries to avoid people to pretend everything is fine and to carry on normal life while one could feel pressure to be positive, making it hard for one to discuss how one feels after cancer. One may think she is experiencing something that others haven't been through. Yet, loneliness can be worse if one finds it difficult to talk about her feelings as this can help patients find ways to support them.

Participant 3: "I do not have many friends. I wasn't going out much and I was a bit down, so I was always with my family"



Participant 6: "I didn't feel like I wanted to go to my friends and colleagues to tell them like I feel this, like they have enough of them. "I feel like I cannot go to them all the time, I feel like I cannot vent out my story"

Some may have family history or members who had cancer and they are prepared in advance. This could be a learning experience. Although cancer can impact one's life negatively, yet most women deal with the situation and build strength to think and live positively in order to fight the cancer battle.

Participant 4: "I didn't experience any psychological or emotional issues with family or with my friend or colleagues. I had no emotional or social concerns hamdullila. In our family, there are lots of people who have cancer. My sister also has breast cancer, so it wasn't that much"

Participant 2: "How you think, it's very important. Think in a good way"

Participant 6: "I tried to get myself as positive as I could".

Participant 8: "My colleague told me why you told her about it, and I said no problem, it's time and InshAllah it will pass. I didn't think too much about depression and I told myself, ``now I am taking treatment, let's live now"

Most women share the news of their diagnosis with their closest family members, others may hide the news from their family members for various reasons.

Participant 10: "I didn't tell anyone about it because you know, they would be worried about me. I didn't want that, so I kept it to myself. And you know when people hear about it, they will be shocked; oh, how would you do it, it is dangerous. I didn't want that negativity"

Breasts are important for female beauty, identity and sexuality. It creates women's' body image and can affect relationships between couples. It is not always about appearance but sometimes loss of sensation or sexuality. Dealing with body image concerns after breast cancer diagnosis and treatment is a process rather than an event. For few women, physical changes after treatment make some of them less comfortable with their bodies. While discussing her concerns



about breast cancer, one of the women said that the look after surgery is affecting her psychologically and she is quite unhappy about it (Jayarajah, U., & Samarasekera., 2017; Sarwer & Spitzer., 2012).

Participant 5: "The look of the surgery is very bad. It is affecting me psychologically"

Other women mentioned that looking at herself in the mirror was an emotional experience.

Participant 10: "But there was difficulty like when I saw myself in the mirror, it was a bit psychological or say emotional"

Participant 6: "The psychological and emotional impact it had on me; I didn't know how this would affect my life like..."

"I don't wanna share as I said I have a daughter, I didn't want her to let her see because first I had to own it for myself, I had to look at it and I didn't after the operation. I would just not look into the mirror. And I had a mirror, the one full body mirror which I put in my room and now I have just a small round one and I needed to speak to someone"

She talked about her feelings towards having no breast after surgery and mentioned that the decision to remove both breasts was itself an emotional experience and she wanted to talk to somebody about it.

"The decision to remove both of my breasts, it was such an immense step; I needed someone to speak to"

On the Contrary, for some women, physical appearance was the least priority since they have other important priority such as family and children, etc.

Participant 2: "I had no issues with my physical appearance.....I don't care this is the last thing I care about.

Fear is the predominant emotion in cancer patients (Singer, et. al., 2015) and it could be both physiological and subjective experience resulting from the presence of a threat (Mazzocco, et. al., 2019). Some women described fear as an affectively charged emotional response towards



cancer including fear of recurrence, fear of the unknown or fear of losing a job because of cancer consequences. Fear of cancer relapse is associated with a poor quality of life and contributes to more distress. Fear of cancer returning or fear of getting a new cancer was commonly described as emotional concern in breast cancer survivors.

Participant 9: "I was scared"

Participant 6: "The fear of recurrence that it will come back"

Participant 3: "I had a fear of recurrence; I was always having this fear that I might get another cancer"

Participant 7: "What I have to do. And of course, when you don't know what chemo is, how does it affect you? This scares you because you don't know anything"

Participant 6: "That is always at the back of my mind, tomorrow I can lose my job and then what do I do?"



4.2.2.2e Sub-theme 5: Spiritual/Religious concerns

Spirituality has been recognised as an essential part of health and a person's well-being (Puchalski., 2001) whereas religion is an aspect of a person's faith that is used to cope with major stressor such as breast cancer (Hebert, et. al., 2009). The sub-theme 5 describes how breast cancer women used their religion as the major resource to overcome fears, finding hope and meaning in their lives. A person's strong faith that God knows and cares about them helped breast cancer survivors to accept their circumstances and provided them invisible support and strength to fight with cancer.

Participant 1: "Muslim always has faith, so I said alhamdulilla". "It worked for me so I should say alhamdulilla as I have faith"

"When we have faith, it helps us pass through all the difficulties hamdullila" "Islam is a gift; it is really a gift. It helps us to pass everything. I found a way to deal with it through my faith"

"Most importantly with my faith in Allah, there were difficulties, but I managed all hamdullila"

Participant 2: "When he told me, I felt that God gave me something like power".

"If this is what, God wants it". "Allah loves me"

Participant 3: "I managed it all with my faith in Allah". "Allah made it easy for me"

"I found more peace and courage with my strong faith in Allah and in my religion"

Most women mentioned their relationship with Allah (God) has flourished and they found themselves even closer to God than ever before. After being diagnosed with cancer, their faith in God became even more stronger.

Participant 4: "When I was diagnosed with cancer, since then I got much closer to Allah"



"During my disease and while I was receiving my treatment, I found myself closer to Allah".

Participant 5: "This disease made my faith into God stronger".

"Because the way God stands with me, nobody can be with me. In this situation, you will feel how God is closest to you"

Participant 6: "I have gone closer to my God"

Participant 7: "I think I got more closer to the God"

"You know I think God is with me cause the treatment did not affect me that I had to stop the work."

Participant 8: "I told my god I agree with everything. Alhamdullila, I got more closer to Allah. From the beginning, I accepted, I prayed, and I told my Allah, I agree with all I have"

Participant 10: "It made me closer to God"

While sharing the importance of having or following religion, participant 7 appreciated the fact of being Muslim as it provided strength to her. She emphasized that without religion, one has no direction. She found herself being a more committed Muslim after being diagnosed with cancer. Another woman shared that positive aspect of the religion as it removed negativity and fears.

Participant 6: We need religion like something to hold onto. I have always been a Muslim but now I am gonna be a strong Muslim. I made a lot of Tahajjud (night prayer) and salah during the Ramadan. I have done more sunna prayers. Because if you don't have religion, you are lost.

Participant 7: "You know when you pray, it releases your fears". "Through prayers, the words you read in a particular way, it already gives your brain a different world and of course the feelings, when you are in this situation, and you really are going to think about this.



Participant 7 also expressed her thoughts about God and her relationship as whichever emotional state you are in, you find yourself as close to God. She also showed gratitude that she was blessed that she suffered no side effects from cancer treatment.

"When you are happy, you are close to God, if you are sad, you are close to God"

"Really this is a miracle that I really didn't get any bad effects from the treatment, and I think it is from God"

Most women showed their reliance on God as well as the future in God's hand as they have faith that God loves them and there is always a reason in God's plan. They acknowledged that the support from God gave them the strength to deal with the life-threatening disease.

Participant 8: "I will put tomorrow up to God. Because I accepted all from the beginning"

"I got the main support from my God. Hamdullila, it is from my God"

Participant 9: "My God and my family really supported me. My religious belief really helped me through the disease and to cope up"

Participant 10: "I believe if anything happens to anyone, it is for a reason so Allah... He knows what is better for you, so it was totally fine"



4.2.2.3 THEME 3: PATIENT EXPERIENCE WITH HEALTHCARE PROVIDERS

A woman with breast cancer tries to balance her role as a patient, daughter, sister, mother, friend or as a colleague. Each of these role changes after cancer which alters her needs, expectations as well as her relationships in various different positions. Theme 3 describes breast cancer survivor's expectations and kinds of support they need to balance her position and role after breast cancer diagnosis.

To cope with breast cancer requires different kinds of support at different times. Getting support and being satisfied with the care they receive is important for breast cancer survivors' wellbeing and adjustment to the new normal. Almost all women showed a good level of satisfaction with the care they received from their health providers. They mentioned that they were treated in a polite way and found the hospital staff quite helpful yet there were some expectations from them towards healthcare providers to improve their services in certain areas.

4.2.2.3a Sub-theme 1: Patient Satisfaction

Patient satisfaction is an important indicator of quality of life (Jackson, et. al., 2001). The delivery of cancer services needs to aim to improve survivors' well-being and their quality of life (Elder, et. al., 2005). Unidentified and met psychosocial needs of cancer survivors may lead to predicted subsequent poor patient satisfaction (Lam, et. al., 2018). High patient satisfaction is associated with better practice guideline adherence and lower inpatient mortality rates (Glickman, et. al., 2010). It is imperative to acknowledge and reinforce the importance of patient satisfaction and patient-centred care models in cancer care delivery.

Faith, religious concerns and beliefs of cancer patients should be considered in providing more holistic cancer care, particularly in ethnically diverse cancer patients which may differ by cultural background since religiosity significantly relates to general satisfaction with life and feelings of hopelessness (Ringdal, 1966). Qualitative results in this study revealed that religious beliefs and faith that emerged from the cultural roots represent the practical relevance of an individual life. Results showed that women who expressed their belief found to be satisfied and seemed more content and positive.

Participant 1: "Muslim always has faith, so I said alhamdullila"



"It worked for me so I should say alhamdullila as I have faith"

Moreover, the psychological mechanisms arising from the spiritual beliefs may be assumed to be closely linked with many perceived positive life outcomes to regain and maintain balance, health, well-being and quality of life (Sharma, et. al., 2020). High levels of spiritual needs are known to be associated with lower levels of satisfaction and diminished perception of quality of care (Astrow, et. al., 2017).

Participant 8: "After I came here, hamdullila from my God, I feel I am comfortable".

Breast cancer carries many psychosocial consequences. It is important to analyse the relationship between patients' satisfaction, meaning in life, optimism, body image and depression. Meaning in life and optimism are positively correlated with life satisfaction where higher depression and body image disturbance appeared to be associated with lower life satisfaction (Fonseca, et. al., 2014). The study result showed that most patients treated were satisfied with the care and hence had minimum concerns with healthcare delivery.

Participant 1: "I was satisfied, and I am still satisfied especially with my doctor".

Participant 2: "Yes, I am satisfied with the care I received". "They (health providers) treated me a good way. All the people were good even when I had radiation, they were helpful people hamdullilah"

Participant 3: "I don't think there is anything missing from the healthcare department. I received all the updates I needed from my doctors and nurses and the doctor was calling me to check on me how I am doing so I got sufficient support from the healthcare department"

"I was fully satisfied with the care I received. Everything was good and the doctor kept on checking about everything"

Participant 4: "I was satisfied with the care I received. I didn't have any issues really"

Participant 5: "You have very good treatment so you cannot say that you didn't have good care".



Participant 6: "I was satisfied with the care"

Participant 7: "I am so happy about the nurses and researchers. I can say that they are with us in this. It's their job and people can do their job in different ways, see here it's really helpful"

Participant 9: "I was fully satisfied with the care I received. For me it's all good, the service Tawam Hospital provided. Today I am a survivor because of all the doctors and staff at Tawam Hospital Oncology department"

Participant 9: "But it is all good and I think they are doing all fine"

Participant 10: "The care I received was perfect. I was satisfied with everything. I got the care that I need and even more really"

4.2.2.3b Sub-theme 2: Patient Expectations

The gap between patients' and physicians' expectations has been a difficult problem in cancer care (Takeuchi, et. al., 2016). Adequate cancer care and treatment outcomes are crucial for every cancer patient yet could be challenging for health providers to meet their expectations. Once the diagnosis of cancer has been established, treatment choices and decisions take the central place in cancer patients' lives (Warning, 2000). All cancer patients' clearly express their expectations regarding these choices which may differ by age, gender and education level (Hoofman, et. al., 2004). Knowing that patient satisfaction is subjective and may not be determined from the satisfaction of the healthcare providers (Leser, et. al., 2021), careful consideration needs to be given to understand and meet cancer patients' expectations and to relate their experience to enhance their quality of life.

The qualitative result showed that there were some unique concerns or expectations shared to improve the current services such as adding *Nutritionist in treatment planning, improved pharmacy services, enhanced or expanded mandate to cover comorbidities, improvement in information delivery, emotional and physical support provision by doctors.*



Some patients expressed their concerns that services needed to be improved and be more efficient; for example, one participant wanted to have sessions with a nutritionist while another was concerned about the slow delivery of pharmaceutical services.

Participant 1: "I really needed a Nutritionist. You just need someone to advise you on how one can deal with appetite. What should we eat and what we shouldn't eat? That is what I needed most"

Participant 2: "Tawam pharmacy services are very slow; I think that should be improved for faster services and better delivery"

Furthermore, some participants felt that healthcare professionals needed to provide more information and another person pointed out that they should provide emotional support as well.

Participant 5: "Lack of information from the doctors, this affects"

Participant 6: "There should be an improvement in information delivery"

Participant 8: "Maybe if doctors can provide emotional support. I prefer if I receive more support from health providers about the physical issues I have. Maybe doctors can also provide emotional support"

To foster an open patient-provider communication, it is imperative to understand the factors that influence patient's expectations regarding care delivery and health system. Policies need to be appropriately framed to emphasise the goal to meet patients' expectations to enhance the quality of care rather than reducing the cost. Health insurance coverage is critical to increase patients' access to cancer care (Goss, et. al., 2009; C-Change, 2008). Although health insurance is covered for all UAE residents (mostly by employers), expatriate cancer patients continue to experience problems accessing the care they need. Insurance coverage expansion and enhancement of current mandates have been the most visible concerns expressed by the expatriates in this study.

Participant 9: "If hospitals can provide enhanced coverage for those who cannot afford the treatment, this would be very helpful"



Participant 7: "Because see I have two ways of my surgery, so plastic doesn't come under mandate approved treatment but, in my case, better to go for this operation. So, for some patients, like in my case like something to do with the increase or enhanced coverage. Expand the mandate, this will help me of course but will help my doctor as well"

"I really appreciate SEHA or whoever created this program for expatriates to get this treatment"



4.2.3 SUMMARY OF QUANTITATIVE & QUALITATIVE RESULTS

Chapter 4 was divided into two parts; the first part included the quantitative result followed by qualitative findings answering the research questions in relation to explore the psychosocial concerns of breast cancer survivors. The quantitative results showed that the most prevalent and most rated concerns by breast cancer survivors were **"information concerns"** followed by **"emotional concerns"**. There was some level of needs mentioned by the survivors with regards to social and financial concerns. Survivors showed much lower **"Physical concerns"** compared to other concerns such as information or emotional concerns which were rated as extreme concerns in the survey, whereas no major concerns were reported in the "**spiritual or religious"** domain.

The overall results in this study demonstrated that women with breast cancer regardless of their nationalities experienced a range of survivorship concerns related to breast cancer experience. After exploring the concerns mentioned by breast cancer survivors in the quantitative phase, further understanding was developed with face-to-face interviews to explore those concerns in greater detail and to gain further understanding about **survivors' living experience** with breast cancer encompassing their typical living challenges and consequences of the breast cancer diagnosis on their lives. Almost all women experienced difficulties in transitioning their lives as their priorities as well as perspective of life shifted through cancer survivorship. Psychosocial concerns of cancer survivors covering five domains were explored n quantitative results and further understood individually as five sub-themes including physical concerns, information concerns, social & financial concerns, emotional concerns and spiritual or religious concerns.

For most women the first reaction to **breast cancer diagnosis** was shock while they gradually came to terms with the diagnosis. For some it was denial and trouble believing or accepting the fact that they have cancer. Denial may be helpful as it gives time to adjust to the diagnosis and most women, they work through it. With regards to **access to care**, some women experienced difficulties in accessing healthcare and mentioned their concerns about **trusting their healthcare** providers. Seeking care and getting the right treatment was another concern as cancer treatment is complex and involves multiple modalities which could relate to socio-



cultural contexts as well as doctor-patient relationship and trust in healthcare providers. Although breast cancer survivors showed no major concerns with regards to physical needs in the quantitative analysis, the interviews gave survivors the opportunity to explain in more depth how their lives were impacted and transformed due to physical difficulties that survivors encountered due to cancer and its treatment effects, mainly fatigue, pain, sleep disturbance and body changes etc. The quantitative results significantly highlighted the differences in physical concerns between Emiratis and expatriates where expatriates showed higher concerns than Emiratis. Some differences in the severity of concerns were noted, for instance, physical concerns were the lowest rated in the survey whereas the qualitative evaluation of the data showed that almost all women complained about their higher physical concerns. This could be explained with an understanding that cancer is a complex human phenomenon which is interpreted uniquely by all individuals. While qualitative data should not be generalised, it is nevertheless very important in helping to gain deeper understandings about the women's experiences, specifically the 'how' and 'why' rather than the 'what'. Regardless of differences in severity found in both quantitative and qualitative data, the results yielded rich information about each psychosocial domain as complementary, where results were used to supplement each other. Additionally, a qualitative sample of a very small group cannot be representative of the overall sample or the larger population.

Information concerns were the most prevalent concern in phase one. Being fully informed about potential Psychosocial needs and long-term adverse effects, recurrence, familial risk, cancer and its treatment, follow-up and support services were the significantly raised concerns about the information needs. However, the qualitative data showed that almost all women were quite satisfied with the information delivery from their healthcare providers yet concerns about being more specific in terms of adverse effects and unique queries about their specific conditions were raised. The same rationale could be applied to understand the difference in outcomes in two forms of data collection as in the later phase the severity was further explained where data was not meant for generalisation but to acquire in-depth understanding. Both quantitative and qualitative results were integrated to facilitate an improved understanding of their needs, survivorship experience and to reflect survivors' point of view to ensure the study findings are grounded in survivors' experience which provided a more complete story than either method would alone.



It was also revealed by reviewing the qualitative data that patients must be fully informed at all levels through the cancer survivorship as needs and circumstances change along the course of the disease. Few survivors were not given any information about what could happen next and what they could face which added to their distress. They were not informed about their specific queries as everyone could have different questions which must be answered in advance so they can cope appropriately upon facing the situation. It was also evident that most doctors do not address the psychosocial aspect of cancer and focus on treatment to get rid of cancer from the body. Expatriate women could have more information needs, particularly about available resources and support groups. It was clear that breast cancer survivors prefer information from their health providers than from any other source yet they acknowledged the need to search for the best information about treatment as it was lacking by their health providers.

Considering the UAE's financial status and healthcare system coverage for nationals, it was expected that there would be no major **financial concerns** among Emirati women as they had full health coverage through national insurance. Also, expatriate women were covered either by their insurance or they had a mandate from the government which covered their cancer treatment. Nevertheless, a few women who mentioned financial difficulties were expatriates. Social and financial concerns were reported higher in expatriates than Emiratis. They mentioned their challenges faced with their insurance providers, mainly approval of procedures as they sometimes had to postpone their treatment. With regards to **social concerns**, almost all patients showed their appreciation towards their family as they received the appropriate support from their family members. No women raised concerns about their work performance and instead, some of them mentioned the support from their colleagues and employers to carry out their jobs and set appropriate work adjustments.

In the quantitative phase, **emotional concerns** were the second most commonly rated concerns by the survivors following information concerns. Dealing with the news of having a breast cancer diagnosis itself was an emotional issue for almost all the women. Most women were unhappy, depressed, worried, angry or having mixed emotions at the time they were diagnosed with the cancer. For some it was difficult to accept the reality which made them very emotional. Feeling isolated and not sharing their feelings about breast cancer affected few survivors quite seriously yet some showed positive attributes and tried to stay positive. Qualitative data further



explained that concerns about body image especially after surgery were both psychologically and emotionally challenging experience for the women and for some it was loss of identity and sexuality. Fear of recurrence, future worries mainly about family and children and living with uncertainty were the utmost emotional concerns revealed during both phases. Fear of recurrence and uncertainty such as losing the job were some of the factors that were expressed by survivors as most concerning. The qualitative results revealed a rich understanding about how breast cancer survivors feel and deal with their utmost concerns. They expressed several concerns about their body image, appearance, hair fall, sleep disturbance, weight, appetite and other health issues mainly from the side effects of cancer treatment which affected them psychologically as well as emotionally.

None of the women had any **religious or spiritual concerns** rather almost all found it a way of strengthening themselves to fight cancer. During the survivorship, getting close to God immensely helped cancer survivors in building their faith and positivity. Some had their utter belief that it happened by Allah's (God) will and Allah will take care of that. God was the primary source of strength for almost all cancer survivors regardless of their nationality.

There were no differences in information, emotion, spiritual or religious domains in Emiratis and expatriates. With regards to breast cancer survivors' **experience with healthcare providers**, most women were satisfied with the care they received yet few provided suggestions to improve current services based on their unique needs such as adding nutritionists to advise, enhanced mandate's coverage and improvement in delivery of information by the doctors to their patients.

The data from both phases were integrated to strengthen the rigor, enrich the analysis and findings and acquire deeper and more meaningful understanding of cancer survivors' experiences, concerns and challenges.

The next chapter presents the detailed discussion on both sets of findings in the light of available literature in line with the conceptual framework of the breast cancer survivors' concerns.



Chapter 5: DISCUSSION

The aim of this chapter is to discuss the findings and identify the concerns of breast cancer exhibited by the women and to address the research question in relation to findings from the study and available literature. This chapter discusses the findings of the study to address the four objectives including: identifying the psychosocial needs of breast cancer survivors, examining the influence of nationality on survivors' needs, investigate disparities in Emiratis and expatriates utilising the sociodemographic and clinical characteristics of the participants and to identify the factors to aid the development of culturally tailored cancer survivorship program.

Breast cancer is the most frequently diagnosed cancer in the world and the most frequent cause of death among women (Ferlay, et. al., 2019; World Health Organization, 2018). Similarly, in Arab countries, breast cancer is the most common cancer in females (Abulkhair, et. al., 2010) representing an estimated 17.7% to 19% of all new cancers in 2018 (Global Cancer Observatory, 2018; Globocan, 2018b). With a growing number of women surviving breast cancer each year and despite the high prevalence of cancer in the country, the literature on the needs of cancer survivors is rudimentary. There is limited research conducted to report the needs of breast cancer survivors in the United Arab Emirates. To the best of my knowledge, this is the first study that explores and addresses the psychosocial needs of breast cancer survivors in the UAE and highlights the existent gaps in the literature that is the lack of studies on exploration of psychosocial needs of breast cancer survivors in this region. In general, the concerns and needs expressed by the breast cancer survivors in this study were consistent with survivorship issues that have been addressed in the literature (Connell, et. al., 2006; Vos, et. al., 2004; National Breast Cancer Centre, 2003; Marlow, et. al., 2003; National Health and Medical Research Council, 2000). However, this study provides some unique and rich findings with new discoveries which are discussed in detail in the chapter below.



5.1 DEMOGRAPHICS AND CLINICAL CHARACTERISTICS OF BREAST CANCER SURVIVORS

With regards to participants' demographics and clinical characteristics, 205 women with breast cancer were surveyed to explore their psychosocial needs among whom 26% were Emiratis and 74% were expatriates in line with the country's population structure. The mean age of the participants was 49 years. More than half of the participants were between 35 to54 years of age. Over half of the survivors who were surveyed, diagnosed either in or after 2018 whereas 37% had their diagnosis established between 2012 and 2017. Majority of the surveyed women (74%) had regional disease whereas 25% presented with an early or localised stage at the time of diagnosis. The survey result showed that the majority (72%) of the women with breast cancer in the UAE perceived they had a good quality of life (Table 1, section 4.1.1).

To explore the **relationship between nationality and patients' demographics and clinical factors,** the two associations were explored based on the current observations and literature findings: "nationality and participants' age" and "nationality and patients' stage of the disease" (Section 4.1.6). The results showed **no significant association between survivors' nationality and their age or stage of cancer at the time of diagnosis**. The result has added valuable evidence to the existing literature where Arab physicians' impression about women has been noted as having it at an earlier age and at a more advanced stage than in western world (Hashim, et. al., 2018; Aldosari, 2017; Elobaid, 2014; Chouchane, et. al., 2013; Najjar & Easson, 2010; El Saghir, et. al., 2007; AbdelHadi, 2006). The result has confirmed that such an observation is not scientifically valid, and results could differ among different Arab populations. The result also showed that there are **significant differences in domains of physical needs and social and financial needs between Emiratis and expatriates (Table. 20a)**. As UAE had a larger number of expatriates compared to nationals reflective of the sample of the study. The sample of the study was unequal leading to unequal variance of groups. Therefore, the differences were signified by using Welch's test instead of t-tests.



5.2 DIAGNOSIS AND CANCER EXPERIENCE

The diagnosis and treatment of cancer can provoke serious psychosocial disruption in a manner that anxiety and depression can elevate significantly in comparison to normal individuals (Jacobsen & Andrykowski, 2015). Diagnosis of cancer is disruptive news for an individual and for the family. When people have cancer, life can change for them and for the people who care about them which can be difficult to handle. Cancer diagnosis is an emotional experience and can bring a number of different emotions and sometimes confusing emotions may appear. Literature shows that the moment of receiving a breast cancer diagnosis resulted in much fear and anxiety for Arab women (Hammoudeh, et. al., 2017; Obeidat, et. al., 2013a; Montazeri, 2008).

The finding in this study shows that for most patients, diagnosis of cancer is a shock for which they are unprepared; uncertainty and fear are common reactions (Participant 1, 3, 6, 7, 10, section, 4.2.2.1a). One may feel shocked, angry or depressed; such feelings may vary from hope to despair or from courage to fear. These are normal reactions for cancer patients which is a disturbing change in their lives. Although openness about the diagnosis enables patients to think more realistically yet healthcare providers need to create a balance between providing information without alienating the patients which can be difficult (Carroll-Johnson, et. al., 2006). Most individuals are able to adjust to the diagnosis over time, religious beliefs help too; however, concerns about the future, medical tests, hospital stays and medical bills are common. Patients usually are better able to adjust their emotions and cope up with the circumstances if they can talk openly about their illness and their feelings with their families (Hagag, 1999). Healthcare providing helpful information considering cultural context as sometimes the patients may not be the first to know the diagnosis.

To better understand the breast cancer concerns, during the interviews, women were asked to share their **living experiences of cancer**. Each cancer patient responds differently to cancer and its treatment. The outcome of cancer differs as well so their experience, concerns and needs as well. Some may find it difficult to accept that they have the disease (Participant 3, 8, section 4.2.2.1a) some may find it easy to have conversation about it whereas others may shun the discussion when the topic arises (Participant 6, 10, section 4.2.2.2d). Some may become



anxious and depressed and for some the impact of cancer is not as high and they can get on with their life as normal as it was before the diagnosis. Each cancer survivor interprets cancer differently from their own individual perspective. The findings illustrated that cancer did impact the lives of breast cancer survivors in various ways that included changes in their perspective, behaviour and lifestyle.

Cancer threatens a persons' well-being which cannot be overemphasized (World Health Organisation, 2014; Eilertsen et al., 2012). In addition to the array of emotions, coping with cancer can be physically demanding and can lead to physical and mental health impairment which together can lead to major social problems (IOM, 2008). Findings show that diagnosis of breast cancer and its treatment is associated with the range of symptoms that impact an individual's physical, emotional as well as financial situation (Lehto, et. al., 2005). Changes in the body could affect a woman's self-esteem and cognitive functioning which can have profound impact on her well-being and quality of life which indicates which facets of an individual's life are most affected by disease and treatment (World Health Organisation, 2012).

Literature also reports an association between compromised quality of life and clinical outcome including survival (WislØff, et. al., 1997). Consistent with other studies, results of this study suggest that cancer changes every part of an individual life (Participant 2, 3, 6, 7, section 4.2.2.1b). Study found that psychosocial and emotional impact of cancer is so profound that patients had adjustment and stress issues mainly related to treatment (Participant 2, 6, 10, section 4.2.2.2a; participant 6, section 4.2.2.2d) as mentioned in studies carried out earlier (Von Essen, et. al., 2000; Hudson, et. al., 2003; Li, et. al., 2013). The physical changes influence not just a person's way of living but also the priorities and meaning of life resulting in a complete paradigm shift. Surviving cancer becomes a way of life so one adapts to live the rest of the life with changes that cancer brings during the survivorship. Finding shows (Participant 6, section 4.2.2.1b; Participant 1, 4, 5, 6, 8, section 4.2.2.2a; Participant 2, 4, 6, 8, section 4.2.2.2d) that participants experienced serious psychosocial symptoms emphasizing emotional and physical side effects such as body changes, pain, fatigue, hair loss, weight changes that impact their ability to carry out the daily living, roles and responsibilities (Jassim & Whitford, 2014).

In this region, responsibilities to fulfil the needs of families take priority over individual needs compared to western regions (Jassim & Whitford, 2014; Surbone, 2008). Participants


(Participant 4, 5, 6, 7, section 4.2.2.1b) endured the adverse effects of cancer treatment in order to carry out their role as a mother, continued employment and housework for their family similar to the findings earlier reported among Arab women in the literature (Al-Azri et al., 2014; Jassim & Whitford, 2014). Most Muslim women believed that the health and disease come from Allah thus accepted the reality and became more appreciative towards life and gain a new perspective to seek new normality and relied on God for their future and health (Table 10; Participant 1, 2, 3, 8, 10, section 4.2.2.2e). Previous research has shown that religious practices and faith are means of coping with the illness which shows their trust in God (Hammoudeh. et. al., 2017; Assaf, et. al., 2017; Al-Azri et al., 2014). On the other hand, in this study, there were participants who expressed no impact on their lives (Participant 2, 10 section 4.2.2.1b; Participant 10, section 4.2.2.2a; Participant 9, section 4.2.2.1a), and they simply returned to normal living following treatment suggesting that cancer is a unique experience of each individual that determines the level of impact that breast cancer can have on each person's life.

5.3 ACCESS TO CARE

To prevent new or recurring cancer and its late- and long-term psychosocial consequences, cancer survivors require ongoing timely access to care to ensure early intervention and coordination between healthcare providers (National Research Council, 2006). Unfortunately, there are numerous barriers to timely and equitable access to cancer care that exist across the cancer continuum which negatively impact cancer outcomes. To improve access to care and reduce disparities in cancer outcomes requires identifying, understanding and addressing those barriers which differ by location and population. These barriers could be structural, geographical, sociocultural, personal or financial.

This study demonstrates that structural barriers including geographical location, need of transport, insufficient diagnostic or treatment services or insufficient trained workforce in the place of residence were some of the barriers to access quality cancer care for some survivors (Participant 1, 8, section 4.2.2.1c). Personal barriers such as mistrust of the health system, poor delivery of care or lack of trust in healthcare providers result in greater psychosocial effects on cancer survivors and impact their lives negatively. Hence, identification of target population and such barriers can also help identification of the best intervention for improving access to



care for instance, programs to improve structural barriers emphasize on multidisciplinary approaches, guidelines and standards for referrals and quality metrics to track down timely access from presentation with breast cancer concerns to diagnosis and time from diagnosis to treatment (Pan American Health Organisation, 2016; IOM, 2002a).

Programs to reduce structural and personal barriers ideally involve community groups as well as breast cancer survivors in educational or supportive services. Improving access and utilisation of healthcare services can impact breast cancer incidence and survival both. Thus, strategies for improving equitable access to care must be customised to meet the needs of the targeted population (Anderson, et. al., 2017). Literature also suggests that patients who require more complex interventions must have access to trained professionals within routine cancer services (Hackett, et. al., 2018; Grassi & Watson, 2012). A document published by Ministry of Health, New Zealand (2010) on guidance to improve supportive care for adults with cancer suggested that the best-practice service approaches must ensure that adults with cancer and their families must have access to the supportive care needed throughout the stages of cancer, from diagnosis onwards (Ministry of Health, New Zealand, 2010).

With regards to **access to care**, most women understood the significance of their symptoms straightaway yet **delayed consulting a doctor** about their breast symptoms (Participant 2, 3 4.2.2.1d; Fallowfield & Clark, 1991). Previous studies in the region also confirmed the similar findings that despite advancement in breast cancer treatment, treatment coverage and well-established screening services, there is a considerable lack of awareness about breast health mainly due to sociodemographic behavioural and cultural factors (Elobaid, et. al., 2021, 2016, 2014; Arafa, et. al., 2020; Younis, et. al., 2016). Findings from this study, also revealed a lack of trust on the part of healthcare providers which led to **second opinions** or visits to other providers or abroad in order to seek expert help (participant 2, section, 4.2.2.1c; participant 1, 2, section, 4.2.2.1d). Patients need to trust that their healthcare providers will be able to care for them appropriately, being experts in the field of Oncology.

The IOM (2013) reported that there is a need for converse acceptance of responsibility emphasizing the importance of building trust with patients and maintaining therapeutic relationship with them over the entire extent of their treatment is crucial. Interestingly, neither the first opinion nor second opinion is an absolute truth (Payne, et. al., 2014; Axon, et. al.,



2008) Second opinion may benefit patients as they bring more certainty; however, if the second opinion differs from the first, it could result in increased uncertainty which may lead to further opinions at higher cost (Hillen, et. al., 2017). Preliminary evidence suggests that patients' motivations to second opinion are more often based on impairment of physician patient relationship rather than on patients' doubts about their physicians' medical competence (van Dalen, et. al., 2001; Payne, et. al., 2014). Literature suggests that the main reasons to seek second opinion are either patients feel the need of confirmation, lack of trust or dissatisfactory communication with the health providers or a need for more personalised information (Hillen, et. al., 2017). Therefore, if patients desire a need for more certainty or express the need to gain more information, the healthcare providers need to either refer them to the available resources which could offer such treatment or offer an extra time to provide more information as some patients are hesitant to admit the lack of trust or dissatisfaction, in that case, health providers can explore patients' need or difficulties (Back, et al., 2005). Oncologists may refer the patient to a colleague within the same facility rather than seeking for a second opinion in case the patient's relationship is damaged due to poor or undesired outcomes (Hillen, et. al., 2017).

5.4 HEALTH DECISION-MAKING

Many women with breast cancer face psychosocial challenges at the time of cancer diagnosis and treatment. **Health decision-making** has a critical importance in cancer care particularly since the introduction of the patient-centred approach which emphasises the relationship between patients and doctors to promote a shared decision-making model (Mazzocco, et. al., 2019). Cancer treatment decision-making is often complex and based on a multidisciplinary team's opinion to opt for the best available evidence-based treatment choices (Institute of Medicine (2013), this approach falls on physicians' decision and is less likely to involve patients and consider patients' desires. As a result, these women can often be lost in transition and may experience distress or other psychological symptoms at a critical juncture during the cancer care pathway.

In contrast to the existing literature which informs that women often perceive themselves as peripheral to any treatment decisions and sometimes undergo treatment against their wishes (Obeidat & Lally., 2014; Jassim & Whitford, 2014; Obeidat, et. al., 2013b; Doumit, et. al., 2010), this study results show that despite being informed about the appropriate treatment, it



was often the women who made decisions and choices about the treatment recommended by their doctors (Participant 1, 5, section, 4.2.2.1d). Thus, it was observed that women in this region regardless of their nationalities, are not passive recipients nowadays as stated in the literature, they rather actively take part in the treatment decision-making. They also showed their desire to be more aware and increase their knowledge about treatment and their side effects (Table 14, section 4.1.5; participant 7, section, 4.2.2.2b).

Nevertheless, their decision-making could be affected by various cultural factors such as knowledge and awareness of possible side effects, fear of recurrence, trust in their own judgement, family responsibilities or priorities etc., (Participant 2, section 4.2.2.1d). It was also noted that surgical decision-making was driven by fear of recurrence (Participant 10, section 4.2.2.1d; Fernandes-Taylor, et. al., 2015). **Shared decision-making** allows patients and their providers to make better decisions, taking into account patients' values and preferences with the best scientific evidence available. It was also noted that the majority of the patients preferred to share decision-making and improved communication with their physicians (Table 8; Nies, et. al., 2017). With regards to autonomy in the decision-making process, the study results show that women who were certain and did take part in their treatment choices at their comfort level demonstrated confidence and greater treatment satisfaction (Participant 2, 5, 8, 10, section, 4.2.2.1d). Katz, et. al., (2014) reported similar results in his study in the US population suggesting that more informed and engaged patients are more satisfied and have better outcomes.

Women may experience severe psychosocial symptoms at the time of diagnosis (section 4.2.2.1a) and treatment decision-making as anxiety and depression are more commonly experienced around that time (Fiszer, et. al., 2014). Some women took up the primary role in treatment decision-making instead of letting their physicians decide. They benefited themselves with this active approach as they outweighed and managed the emotional burden of breast cancer while deciding among treatment choices taking the benefits of it against adverse effects (Participant 6, 4.2.2.2d; Fernandes-Taylor, et. al., 2015). Numerous studies have shown that some of the psychosocial factors can predominantly affect the breast cancer patients' therapeutic decision-making. For instance, support from healthcare providers, family and colleagues (Halford, et. al., 2001; Makabe, 1999; Stanton, et. al., 1998), degree of patients' involvement, their expectations, values and satisfaction with the treatment (Sato, 2005;



Petersen, et. al., 2001; Stanton, et. al., 1998; Morris & Ingham, 1988). It has been found that for some women who chose treatment voluntarily, it was a more important factor than treatment itself, for psychological adjustment afterwards (Participant 10, section, 4.2.2.1d; Stanton, et. al., 1998; Morris & Ingham, 1988). Hence, it is important that offered treatment meets the patients' expectation to ensure her psychological well-being and adjustment (Asadi-Lari, et. al., 2004; Lateef, 2011). Therefore, offering flexibility instead of rigid discipline by healthcare providers may help in building trust and confidence in patients advocating partnering not paternalism (Dawson-Rose, et. al., 2016; Saha, et. al., 2008).

5.5 THE ROLE OF SUPPORT FORUMS

A 'Support forum' is a formal environment which offers its members an opportunity to reduce stress and isolation and share information and experiences among other cancer survivors. In such a place, patients can share their fears and concerns with other people with similar experiences who understand what one is going through since they have been in such a situation. The results of this study show that some women expressed their deep desire to gain some personalised information about their disease by associating with support forums, in which information can be sought anonymously which would aid those patients who are reluctant to ask due to concerns about physician time being limited to due to personal shyness about talking directly about their health concerns.

Support forums may enhance the communication, understanding and knowledge about breast cancer and its treatment, diet and nutrition, health behaviours, coping and adjustment and available resources to cancer survivors to provide social and psychological support (Hillen, et. al., 2017). However, the quality of information and readability are important factors to consider while gaining information from online communities and forums. A study found that although breast cancer websites have good readability, they may still not be understood by the average patient as, to be understood by 75% of the population, readability should be at a sixth-grade level (Arif, & Ghezzi, 2018). Nevertheless, the available information on online forums have been evaluated with 90% accuracy (Bender, et. al., 2013).



5.6 PSYCHOSOCIAL CONCERNS OF BREAST CANCER SURVIVORS

In this study, psychosocial concerns of breast cancer survivors were explored to identify their needs in five domains including physical, information, social and financial, emotional, religious or spiritual. It has been shown that cancer survivors experience various difficulties in their daily lives including physical, emotional, social or economic difficulties (Amir, et. al., 2012; Boyes, et. al., 2012). These concerns include but are not limited to adjusting to new normality (Participant 5, 8, section 4.2.2.1b), adjustment with stress and other psychological issues, managing expectations, dealing with emotions and several other concerns (Participant 3, 4, 6, section 4.2.2.1b).

The results of this study strengthened the finding from the previous study conducted by Nair and colleagues (2018) on cancer survivors' needs in the UAE and demonstrated that breast cancer patients require a wide range of psychosocial care in the UAE. The result of the study showed that breast cancer survivors had their concerns and needs which were prevalent in both groups including UAE nationals and Emiratis (Table 20a).

Information and emotional needs were highly rated concerns in the survey (Table 20a) whereas physical needs were mostly specified during the qualitative interviews (4.2.2.1b; 4.2.2.2a). Patients showed their desire to have more information to manage treatment side effects and to deal with cancer that had an impact on their daily functioning consistent with the needs described in the literature (Participant 6, section 4.2.2.2b; Participant 5, 6, section 4.2.2.3b). The needs of information and being well-informed commence first before any other needs. Secondly, breast cancer survivors showed that the need to have emotional support and to speak to someone who can empathise is critical to manage the disease during the survivorship, consistent with previous studies (Participant 6, section 4.2.2.2d; Connel, et. al., 2006; Thewes, et. al., 2004; Marlow, et. al., 2003; Gray, et. al., 1998). Broader literature also suggests that Arab women with breast cancer who remain symptomatic with limited access to reliable information may struggle to transition to a new normal even several years after the treatment (Fearon, et. al., 2019). These factors could be cultural, thus culturally-based intervention must be considered for the targeted population.



5.6.1 PHYSICAL CONCERNS OF BREAST CANCER SURVIVORS

Despite the awareness of **physical impairments** that can occur after breast cancer treatments, there is little known how best to manage those (Harris, et. al., 2012). Participants described how they adapted to living with pain and other physical difficulties resulting from disease and mainly treatment (Participant 1, 6, 8, section 4.2.2.1b). Physical concerns were much more highlighted as major concerns by the survivors during the qualitative interviews (section 4.2.2.2a). The integrated result (Table 13; section, 4.2.2.2a) revealed that **pain and fatigue** were the most concerning aspects of physical concerns described by the participants in both groups consistent with previous studies (Jensen, et. al., 2010; Bardwell & Ancoli-Israel., 2008). Jensen, et. al., (2010) also noted that although pain is a common problem among women with breast cancer, its impact and predictors have not been studied at large.

Pain has been more commonly reported in breast cancer survivors than any other survivors' group such as those with a history of cancer of other sites for instance colorectal or prostate (Jiang, et. al., 2019; Boland & Ahmedzai, 2017; Glare, et. al., 2014; Brown, et. al., 2014; Deimling, et. Al., 2007). The potential impact of pain on patient functioning and quality of life has also been reported in previous studies; researchers found that survivors who reported pain also reported several functional difficulties required to carry out daily activities compared to those who did not report any pain (Peuckmann, et al., 2009; Deimling, et. al., 2007; Gulluoglu, et al., 2006).

In this study, the participants described the intensity of the pain to variable degree, some described it as manageable, others found it difficult to manage with work responsibilities due to their pain which affected their living too (Participant 2, 3, 4, 5, section 4.2.2.2a). The intensity of pain described by the breast cancer survivors in the literature as mild where moderate to severe pain was very rare (Jensen, et. al., 2010) which is consistent with the findings in this study (Participant 9, 10, section 4.2.2.2a). Finding showed that pain was a great contributor (mean=.64, SD=1.274) that impacted survivors' daily living, but the adverse effects of the treatment affected psychologically as well as negatively (Participant 1, section 4.2.2.2a; participant 6, section 4.2.2.2d). Significant associations have been found between both the presence and severity of pain and psychological functioning and physical functioning both (Jensen, et al., 2010).



Fatigue was another common symptom reported by most women in this study (mean=.69, SD=1.283). Fatigue has been recognised as a distressing, persistent, subjective sense of physical, emotional and/or cognitive tiredness or exhaustion related to cancer or cancer treatment that is not proportional to recent activity and interferes with usual functioning (National Comprehensive Cancer Network, 2010). Fatigue is a multidimensional concept that affects patients differently. This cause of fatigue can be disease itself, treatments, nutritional issues, pain, drugs, sleeping problems or other psychosocial problems such as anxiety or depression. Therefore, the management of fatigue in cancer patients is generally theoretical and medical approaches focus on treatments rather than focus on the symptoms of fatigue (Aydiner, et. al., 2019). Cancer related fatigue and sleep disturbances are reported to have common aetiology and are related to pain, depression, concentration and cognitive functional loss (Roussi, et. al., 2007).

Apart from pain and fatigue, several other physical concerns were expressed by the survivors for instance, body image, appearance, hair fall, sleep disturbance, weight, appetite and other health issues mainly due to the side effects of cancer treatment which affected them both psychologically as well as emotionally (Participant 1, 3, 4, 6, 8, section 4.2.2.2a; Jassim & Whitford, 2014; Nizamli et al., 2011; IOM, 2008). The results of this study demonstrate that loss of breasts or breast tissue to minimise the risk of having breast cancer or to treat cancer, resulted in body image concerns in most of the women which affected the survivor's psychological health (Participant 8, section 4.2.2.2a; participant 5, 6, 7, section 4.2.2.2d). Finding confirmed that altered body image and concerns about femininity and sexuality can remain silent issues in cancer care particularly in Arab women due to reserve culture (Participant 6, 10, section 4.2.2.1b). However, these issues can remain either silent or arise long-after treatment that many people may feel they don't know where to turn for information and support.

Asymmetry in appearance after removal of breast tissue resulted in great difficulties both physically and psychologically. Survivors explained that their living changed and lifestyle too; for instance, lack of choices in finding clothes that could fit them was another concern. Owning the new body shape was a psychologically difficult process as reported by the survivors. For some women, change in physical appearance was least concerning since other priorities were



more important for them. On the contrary, for some, **the body or self-image** was related to their self-confidence (Participant 6, section 4.2.2.1b).

Body image after surgery had a great impact on women's lives and their self-confidence as other researchers found the negative correlation between body image and self-esteem (Morales-Sánchez, et. al., 2021; Prates, et. al., 2017; Bolton, et. al., 2010; Sertöz, et. al., 2004; Al-Ghazal, et. al., 2000). Literature shows that the only factor differed significantly between groups of women who had different type of surgeries for breast cancer was the body image (Cohen, et al., 2000). In this study, other than pain, fatigue and body image, survivors also discussed **sleep disturbance and insomnia** which became worse with other conditions for instance, chest pain or other comorbidities. A study reported that women who experience neuropathic pain which is sometimes severe, increases with the movement and interferes with sleep as found in more than 50% of cancer survivors (Aydiner, et. al., 2019).

In this study, qualitative findings further discussed that **adverse effects of chemotherapy** were the worst aspect of the physical concerns due to its unpleasant and unexpected side effects (Participant 6, section 4.2.2.1d). Total or partial **hair loss** caused by chemotherapy and hair regrowth issues were commonly described by women who had chemotherapy (Participant, 1, 4, 6, 8, section 4.2.2.2a; Choudhury, et. al., 2018; IOM 2008). **Skin changes, mainly** discoloration, was also noted by some women (Participant, 1, section 4.2.2.2a). Weight gain and weight loss both after treatment were also reported in some women, particularly weight gain in women who had hormone therapy (Participant, 1, 4, 9, section 4.2.2.2a). These findings were commonly reported in the literature earlier and show that women may continue to experience these physical concerns throughout cancer survivorship (IOM, 2008; Cappiello, et. al., 2007).

By losing such feminine physical characteristics, some women developed social stigma which induced stress and a physical burden of the disease in them (Participant, 1, section 4.2.2.2a; participant 5, 10, section 4.2.2.2d). This can further reduce the opportunities in life and increase isolation and social rejection which greatly impact them with poor health outcomes (Kagawa-Singer, et. al., 2010; Freeman, 2004). These physical issues may limit the ability of breast cancer women to carry out their roles and responsibilities which may result in the loss of confidence and control leading to decreased autonomy and thus become dependent on others



(Luoma, et. al., 2004). Yet, most women showed that they were able to adapt and continued their roles to look after their children or families as their common traditional role, mainly Muslims or Arab women, even after transitioning from caretaker to care receiver (Kawar, 2013; Taha, et. al., 2012). It is crucial to understand that some breast cancer survivors' outcomes depend not only on their prognosis but also on other factors such as cancer treatment and its adverse effects of cancer treatment (Du, et. al., 2021; Odle, 2014). Therefore, breast cancer survivors must be monitored and then treated when any physical symptoms are indicated (Bardia, et. al., 2012).

Participants also noted that their **physicians did not try to understand the patients' difficulties** in terms of physical issues thus left unattended or unaddressed appropriately which may lead to maladaptive behaviours, poor prognosis and low quality of life (Participant 2, 5, section 4.2.2.1c; Participant 5, section 4.2.2.1b; participant 5, section 4.2.2.2c; Reyes-Gibby, 2012). Healthcare providers must consider that breast cancer patients may lose the physical traits they face weakness and emaciation thus, they may not be able to take care of themselves and do personal care routines as prior to disease. As a result, they may not be recognised as the same person by others. This can make a person feel less loveable. Maintaining cancer survivor's dignity, respecting modesty and supporting personal care all are important supportive care measures that must be considered while assisting with breast cancer survivors' physical concerns (Gorman, 2018).

This study also assessed cancer survivors' preferences about learning sources and topics. Fifty nine percent participants showed their **preferred learning source** with regards to physical concerns was only healthcare specialists (59%) (Table 2). For **preferred learning topics**, 46% of participants showed they had interest in learning more about nutrition and safe exercises whereas 40% wanted to know about nutrition only suggesting that participants understand that nutrition plays an important role towards adapting to healthy living (Table 6). Over half of the women (64%) wanted to know more about cancer screening and 36% showed their interest in health screening (Table 9). 73% showed their interest in learning about enhancing communication with their doctors, 18% mentioned both doctors and their spouses, suggesting satisfied communication patterns with their healthcare providers (Table 8).



5.6.2 INFORMATION CONCERNS OF BREAST CANCER SURVIVORS

In this study, results showed that breast cancer survivors had *extreme concerns* about the "information needs" which scored highest in rating (mean=27.99, SD=1.020). Information needs were the major worry for most breast cancer survivors than any other psychosocial domain. Information concerns have been reported as of highest need than any other psychosocial concerns in numerous studies that were conducted earlier to identify cancer survivors' needs (Legese, et. al., 2021; Kuruppu, et. al., 2020; Fong & Cheah, 2016; Shea–Budgell, et. al., 2014).

Participants showed their strong desire to gain more knowledge about breast cancer issues; they felt they had insufficient information about breast cancer and its management (Participant 5, 6, section 4.2.2.3b). From their physicians, they received only the basic information such as adverse effects of chemotherapy; however, what else could go wrong and what else could be anticipated apart from side effects were important aspects that were missing from the information delivery (Participant 6, section 4.2.2.2b). Elbarazi (2016) argued that multicultural communities pose challenges to healthcare providers in providing the appropriate information to meet their unique needs appropriate with the cultural values. It is a critical aspect of cancer care, yet, little is known about the information needs of breast cancer survivors in the UAE. One possible reason for this lack could be the practicing model of physician centred care rather than patient-centred care which is rare in this region (Elbarazi, 2016). Another obvious reason which is also consistent with previous studies is, physicians mainly focus on biological or physical symptoms or medical strategies and usually do not address or provide information about the psychosocial impact of breast cancer (Eaton, et. al., 2020; Smith, et. al., 2011; Henkel, 2005; IOM & NRC, 2004).

Survivors opted to be linked with online forums, searching the internet for the specific information (Mancini, et. al., 2006; Jenkins, et. al., 2001) or talked to other survivors to educate themselves since some survivors found physicians' limitations with the information delivery which were either time constraints (Obeidat & Khrais, 2015), lack of accurate information or communication issues mainly trust (Dean, et. al., 2017; Kaiser, et. al., 2011). Qualitative findings further strengthen the quantitative findings and explained that information concerns



remain exist throughout cancer survivorship and must be addressed at all levels of cancer trajectory as circumstances and needs change overtime or vary at different points in the disease process (Participant 5, 6, section 4.2.2.2b; Participant 5, 6, section 4.2.2.3b; Lu, et al., 2020; Parker, et. al., 2007). Also, the level of need can differ by socio-demographics or cancer characteristics (Playdon, et al., 2016).

Communication is the core requirement for setting healthcare goals. Although the majority of the patients in qualitative interviews showed satisfaction with obtaining the information from their healthcare providers (Participant 1, 2, 3, 7, 9, 10, section 4.2.2.2b) yet some were not confined with the amount as well as the content of information by their physician and perceived that there was no professional or healthcare setting available for patients to get the specific information about their specific issues (Participant 5, 6, section 4.2.2.2b). Majority of the participants showed the **preferred source** to provide cancer information must be their doctors (Table 2, 3, 4, 5, section 4.1.2; participant 7, section 4.2.2.b; Maloney, et. al., 2015). Findings confirmed that the major issue with the information needs was not the delivery or how information was delivered but what was delivered as most women had access but were not confined with the amount of information provided or to gain personalised information to meet information their specific needs (Participant 5, 6, section 4.2.2.b).

Participants also showed their desire for **communication with peers** and need to be prepared for upcoming treatments by talking to other survivors and sharing experiences as they had been through the same and could educate and inform other patients in advance (Participant 6, section 4.2.2.1d; Allicock, et. al., 2014). It was concluded that information concerns were the greatest need expressed by both Emiratis and expatriates and there were no differences found in information needs between these two groups (Table 20a) suggesting that **information needs are the key concern of cancer survivors regardless of nationality**. *Being fully informed about potential psychosocial needs and long-term adverse effects were the most alarming* information concerns reported by the survivors (mean=4.346). Consistent with this study results, Mills & Sullivan (1999) also expressed that information about treatment and side effects, the extent of the disease, prognosis and self-care were the most relevant for cancer survivors (Table 14). Other information concerns include, *being fully informed about any symptom appearing (recurrence/new primary), being fully informed about any familial risk*.



next follow-up date, place and physician, who to contact for new symptoms, about psychosocial support as well as about cancer and its treatment (Table 14). This suggests that breast cancer survivors' biggest worries and uncertainties are associated with the treatments' adverse effects, fear of recurrence, support needed to manage the disease, risk to family, follow-up on disease. Survivors' information needs are continuous throughout the cancer trajectory. Unavailability of psychosocial resources, structured cancer survivorship programs, lack of support groups and rehabilitation programs are serious considerations and a call for actions to improve cancer care and survivors experience and their quality of life.

Additionally, cancer survivors' most prevalent concerns were identified and assessed for three different timeframes that are 0-1 year since diagnosis (2018 or above), 2-5 years of diagnosis (2012-2017) and 6 years of initial diagnosis or earlier. The results indicated that **information concerns were the most extreme concerns reported in all three timeframes suggesting the constant need of information among cancer survivors** throughout the survivorship (Table 12). Furthermore, a significant positive correlation was found between quality of life and information concern (p < 0.01) means, with the increase in information needs, quality of life scores will also increase suggesting those who scored higher in the information domain also had higher scores in quality-of-life evaluation (Table 21).

Previous studies have identified that cancer patients' information needs differ with gender, age, stage and type of cancer (Davison & Breckon, 2012; Bernstein, et. al., 2011; Rolnick, et. al., 2007; Jenkins, et. al., 2001). Chua and colleagues (2018) argued that there is, however, little understanding on the type and amount of information that patients require. This study results show that cancer patients desire to attain the most authentic information (Participant 7, section 4.2.2.2b). Accurate and reliable information can empower cancer survivors, relieve anxiety and depression, hence, can improve quality of life (Chua, et. al., 2018). Having said that, many studies have not addressed the different dimensions of the information needs such as patients' satisfaction or mode of delivery of information which is also part of the worry of some survivors reported in this study.

Providing information to all cancer survivors helps them with treatment choices, decisionmaking and cope with disease or treatment side effects, reduce anxiety, fears, depression thus leads to patient satisfaction, improves communication and quality of life (Miyashita, et. al.,



2015; Davison & Breckon., 2012). There is a clear need to conduct research on cancer patients' information needs in the UAE (Elbarazi, 2016). Meeting cancer survivors' information needs is a fundamental aspect of patient-centred care (Davis, et. al., 2005) and patient satisfaction with the information provision is a key indicator of quality cancer care.

5.6.3 SOCIAL AND FINANCIAL CONCERNS OF BREAST CANCER SURVIVORS

In this study, social and financial concerns were expressed and rated as the highest (mean=1.96) mainly in expatriates which indicated 'some' level of needs. **Health insurance and financial concerns** were the main worry for expatriates as Emirati women had full national coverage by the government (p<0.05, table 20a,). UAE mandates compulsory health insurance for all UAE citizens. The UAE government provides full medical coverage to all nationals whereas for expatriates, the insurance is provided by the employers for which the extent of coverage is determined by their salary, designation etc. Studies show that financial stress is associated with poor quality of life (Salem & Daher-Nashif, 2020; Carroll-Johnson, et. al., 2006).

In this study, few expatriates expressed their challenges with the insurance approval process as they had to either wait and postponed their planned treatment (Participant 6, section 4.2.2.2c), yet no patients declared major financial or insurance issues as those who had no or lack of insurance coverage, they utilised government mandate which issued from the local government to facilitate patients who cannot afford the treatment (Participant 7, section 4.2.2.2c). Financial concerns were not reported very high in this study (Table 15); thus, it can be concluded that unlike many other countries, in UAE, the financial concerns of breast cancer survivors were not the most distressing concerns affecting their quality of life (Delgado-Guay, et. al., 2015; Zafar & Abernethy, 2013; Sharp & Carsin; 2013; Klimmek, et. al., 2010). Additionally, employers and colleagues were equally supportive which helped them sustain their jobs. Qualitative results showed that most of the patients managed their employment with the support of colleagues and employers yet, some mentioned problems such as having needs for time off from work due to treatment (Lim, et. al., 2017).

Healthcare providers can assist cancer patients through directing them to any available financial support programs to cover the cost of cancer treatment for those who can't afford it.



Literature (Fenn, et. al., 2014) indicates that increased financial burden due to cancer treatment cost is the strongest independent predictor of poor quality of life which affect it negatively and lead to other psychological symptoms for instance, anxiety, depression and distress among cancer survivors (Fenn, et. al., 2014; Sharp, et. al., 2013). Therefore, early identification of financial concerns, appropriate referrals, support services and other early interventions such as problem-solving, cognitive behavioural therapy and family group support may help lessens these concerns and could be effective ways to reduce the depression and anxiety induced by financial burden for those who are at risk (Semin, et. al., 2020; Azzani, et. al., 2015; Cagle & Loscalzo, 2015; Damaskos & Parry, 2015).

This study also assessed social concerns of cancer survivors and findings showed that almost all participants had received a good level of support from their husbands and other family members which helped them cope up with their disease (Participant, 1, 2, 3, 4, 5, 9, 10, section 4.2.2.2c). For most women, family was an important source of strength to manage the disease, showing the importance of the role of family in both diagnosis and treatment stages. Although the current literature presents gender domination in Arab and Muslim culture where men are considered dominant and quite conservatives about their women to the extent that they object their partners and their families to be seen by clinicians and expose their body to other than husband (Salem & Daher-Nashif, 2020; Ziadeh, et. al., 2018, Doumit, et. al., 2010; Almegewly, et. al., 2018; Donnelly, et. al., 2017; Hwang, et. al., 2017; Saca-Hazboun, 2017; Elobaid, et. al., 2016; Haddou, et. al., 2016).

Breast cancer survivors were affected by the presence or absence of family support (Snyder & Pearse, 2010). Family and husband both were the source of support for the women throughout the course of the disease, differing the existing literature which demonstrates men dominance over women in Arabs in terms of health decision making. Women were open to talk about their disease and for some women, these were their husbands who encouraged them to go and see doctors (Participant 1, 2, section, 4.2.2.2c). This could be due to increased availability of information on the internet, increased awareness about breast cancer and screening by health authorities or social media information in the region. Although the findings of this study show no major social concerns yet; it is important for healthcare providers to understand that cancer survivors, when they are diagnosed or even when they are cured, often continue to experience psychosocial difficulties associated with the disease, such as cancer stigma and changes in



relationships with their acquaintances, which are medically difficult to solve (Knapp, et. al., 2014; Lebel & Devins, 2008; Burg, et. al., 2015) therefore preventive interventions including couple-based interventions (Zimmermann, 2015) or psycho-oncology interventions (Faller, et. al., 2013) could be more effective to benefit patients and their spouses to gain advantage of it as they both adjust to cancer together.

5.6.4 EMOTIONAL CONCERNS OF BREAST CANCER SURVIVORS

Breast cancer has been acknowledged as a possible traumatic stressor, researchers have documented that dealing with breast cancer could result in poor psychological outcomes for instance, post-traumatic stress disorder or enhanced appreciation of life known as post-traumatic growth (Aydiner, et. al., 2019). These psychological issues can remain a concern for many survivors and these patients have an increased risk of major depression (Ewertz & Jenson, 2011). Many women report the need for emotional support for recovery (Thewes, et. al., 2004) and for well-being (Bloom, et. al., 2004).

In addition to information concerns, **the second most prevalent extreme concerns for breast cancer survivors were emotional concerns** (mean=15.05). Findings show that emotional needs which ranged from concerns associated with the cancer diagnosis itself, dealing with emotions and mixed emotions, fear of recurrence, genetic counselling, worrying about future and family particularly children, fear of losing job and loneliness were contributing factors which affected those women emotionally (Participant 2, 3, 4, 6, 7, 8, 9, section 4.2.2.2d). Fear of recurrence and *genetic counselling* were the most prevalent emotional concerns with a mean of 3.4 for each factor in quantitative results (Table 16). Also, there were **no significant differences** (**p>0.05**) in emotional needs between Emiratis and expatriates (Table 20). However, the multiple regression model explained the increase in emotional concerns with the increase in receiving treatment suggesting multiple treatment will add more to women's emotional burden (Table 22d).

Diagnosis of cancer is itself an emotional experience which brings other fears along with it. These fears relate to treatment, recurrence of cancer, getting through the process, being exposed to others or fear that other members of the family will also get the cancer. The results of this study reveal that most participants became sad and cried when they were told they had cancer (Participant 1, 3, 4, 6, 7, 8, section 4.2.2.2d). Women showed denial and worry at the time of



initial diagnosis wondering whether their symptoms were serious or whether they really had cancer; most were not expecting the disease and were shocked and in a state of denial initially (Participant 8, section, 4.2.2.2d; Participant, 1, 3, 4, 5, 6, 7, 8, 10, section 4.2.2.1a; Price, 2019; Baio, 2017; Aydiner, et. al., 2019). They found it hard to accept that they had cancer (Participant 8, section 4.2.2.2d). Fear of recurrence continues after diagnosis in most survivors (Participant 3, 6, section 4.2.2.2d; Vickberg, 2001; Ferrell, et al., 1998). It was a distressing problem in most of survivors (Participant 3, 6, 7, section, 4.2.2.2d; Ziner, et. al., 2012). Fear of recurrence was also found a contributing factor to anxiety (Participant 3, section 4.2.2.2d; Aydiner, et. al., 2019).

Secondary problems such as pain and discomfort due to changes in the body were commonly reported after surgery which led to depressive symptoms (Participant 8, section 4.2.2.1b; Participant 4, 10, section 4.2.2.2a; Cordero, et. al., 2015; Bower, 2008). Surprisingly, some women had no issues at all with the physical appearance as physical appearance was the last thing that mattered to them, and they were rather worried about their family and children and describing their feelings that they must be strong as they have families and responsibilities towards them (Participant 2, section 4.2.2.2d; Castillo, et. al., 2019; Carroll-Johnson, 20006). It is more concerning to Arab women whose main role in the society and family is to be a care provider and protector of their children, husband and families as mothers, wives and daughters (Fearon, et. al., 2020).

Feelings of permanent and temporary loss were commonly depicted as concerns by the survivors associated with depression, anxiety and feelings of grief as women perceive their breast as something that makes them complete thus feel demoralised when they lose their breast (Alhusban, 2019; Mermer, et. al., 2016). Results showed that surgery not only creates an intensified sense of loss for some women, but it has the potential to unleash psychological reactions since it damages the symbolic connection of the breast to femininity and sexuality (Participant 6, 4.2.2.2a). The results of this study demonstrate that breast cancer impacted women' functions, body image, perception, psychological state and relations with those around them (Participant 5, 6, 10, section 4.2.2.2d). Furthermore, it provoked various concerns and fear including fear of recurrence and passing it to family, difficulties for family mainly children, sexuality and loss of self-esteem etc.



One of the biggest worries found was fear of disease recurrence (Participant 3, 6, 9, section 4.2.2.2d). Although women in the UAE do not talk about sexuality or issues with sexual relationships which could be due to reserved culture however, they expressed their embarrassment of exposing the body, discomfort seeing and showing the scars or overall body changes (Fingeret, et. al., 2014; Collins, et. al., 2011). The first study on psychological reactions associated with surgery was conducted in 1952 by Renneker & Cutler which drew attention to the existence of grief defined by depression, anxiety, insomnia, guilt and defined the relationship between this state and organ loss. After that mastectomy practice has become a significant area of research to understand the relationship between cancer, organ loss and psychopathology (Özkan, et. al., 2010; Özkan, et. al., 2003).

Social isolation or changes in social interaction occurred with some of the survivors as they felt isolated and lost their interest in maintaining relationships with colleagues and friends (Participant 3, 6, section 4.2.2.2d). This either could be due to grief of cancer, lack of self-confidence, concerns that change of appearance may reveal a breast cancer diagnosis to others or to avoid the conversation on this topic. It is also worth considering that in Arab communities, family members tend to hide illness and disease from the community since cultural values require them to keep family issues within the home (Kane, et. al., 2020).

Ferrel, et. al., (1998) explained that breast cancer survivors continue to experience fatigue, pain, fear of breast cancer recurrence, fear of breast cancer in female relatives, and the stress of living with uncertainty and maintaining hope. Findings show that accepting that one has cancer was not easily digestible on any scale, it was overpowering and scary but most Muslims women with breast cancer eventually arrived at the stage of acceptance believing it is from Allah (God) (Participant 3, 4, 8 section 4.2.2.1a; Participant 3, 8, section 4.2.2.1d). Acknowledging one has a disease is half the battle. Accepting that one has a disease really allows a person to get in the fight towards beating cancer (Baio, 2017). Findings showed that most survivors did hope for a positive future while focusing on the present, possibly due to their stronger faith in almighty Allah (Participant 1, 3, 10, section 4.2.2.2e).

Some patients showed their concerns about the absence of not having psychologist and emotional support services available during their care process to receive appropriate psychological or emotional care (Participant 6, 7, section, 4.2.2.2b; Participant section, 6,



4.2.2.1d). However, this is usually the case in advanced countries where patients can request for a clinical psychologist (Giesler, et. al., 2019). These services are however, not usually part of the routine cancer care. Breast cancer survivors require holistic care from healthcare providers (Iddrisu, et. al., 2020). It is important that at every stage along the cancer continuum, the care delivered must address physical aspects of the illness in addition to the mental health and coping strengths of the patient and family (Carroll-Johnson, et. al., 2006).

5.6.5 SPIRITUAL OR RELIGIOUS CONCERNS OF BREAST CANCER SURVIVORS

Spiritual or religious concerns can be identified as concerns about a relationship with a higherbeing, concerns about personal religious practices and concerns about meaning or purpose of life (Holland, et. al., 2015). Although spiritual or religious concerns are individuals' essential needs in all places and time (Yousefi & Abedi., 2011) but for cancer patients, spiritual or religious concerns are likely to reflect on existential issues due to uncertainty of their future (Nixon, et. al., 2013). Spiritual or religious concerns are highly correlated with depression, anxiety and a measure of overall distress (Wells-Di Gregorio, et. al., 2013).

In this study, participants shared their perceptions about their breast cancer experience and how their psychosocial needs had an impact on their lives. They described how they face daily life challenges due to cancer so life could be managed, how they learned to adapt to a new normal and manage their living with physical, emotional, social issues and how their faith helped them through the difficulties (Section 4.2.2.1) and ease their struggles and religion became the source of strength during cancer survivorship (Table 10, section 4.1.4). Thus, no spiritual or religious concerns found in this study (mean = 0.20, SD= 0.915), rather survivors' faith and belief in Allah (God) helped them accept the disease and suffering of cancer (Section, 4.2.2.2e).

The findings of this study are consistent with the research conducted earlier while presenting some new discoveries related to the unique UAE population (Assaf, 2017; Elobaid, at. al., 2016). The study findings confirmed that religion was a fundamental aspect of women living with breast cancer in this region (Jassim & Whitford, 2014; Lazenby & Khatib, 2012; Ahmad, et. al., 2011). It was also evident that religion was the source of strength for almost all survivors regardless of their nationality (Table, 10). Religion also was the central aspect for almost all



survivors as they relied on their faith and religion more than any other source to get through the course of the disease and to attain hope.

Participants believed that religion is the way of life for all issues of Muslim's life (Nabolsi & Carso, 2011) and everything including disease, cure, life and death is in Allah's hand and it is Allah's decision too. In general, Muslims believe that Allah predetermines everything that happens to them in their lives including diseases (Khater & Alkwiese, 2013; Obeidat et al., 2013a; Alqaissi & Dickerson, 2010). The holy book "Quran" states that Allah is the disposer of affairs (Surat Al-Zumar 39:62); "And when I am ill, it is He who cures me" (Surat ash-Shu`ara` 26:80). Therefore, Muslims believe that Allah is the ultimate healer. Muslim breast cancer survivors believed that cancer is a test from Allah for their patience and tolerance thus they seek help through their faith as it is believed that disease expatiate their sins in this life (Khater & Alkwiese, 2013; Taha, 2012).

All participants believed that cancer has strengthened and has added more of their faith (tawakkal) in Allah (God) (Jassim & Whitford, 2014; Nizamli, et. al., 2011). They also described that their faith and religion was the source of strength, comfort and hope in line with the holy verse in Quran. "It is (the Qur'ãn) for those who believe, a guide and a healing" (Surat Fussilat 41:44). Their trust in Allah eased their experience and gave them the ability to cope up and deal with their worries and difficulties, suffering, pain and uncertainties and more importantly to accept their disease (Jassim & Whitford, 2014). Participants mentioned that they have become closer to Allah than ever before and have increased their prayer practice by investing their time more toward recalling and reminding Allah and reading the Quran - It is also mentioned in Surat AlBaqarah 2:155, "We will test you with something of fear, hunger and loss of wealth, and lives and vegetation. And give glad tidings to those who have patience".

These beliefs helped the Muslims to surrender to Allah and make them contend with whatever difficulties cancer brings to their lives (Jassim & Whitford, 2014; Doumit, et. al., 2009). Literature suggests that cancer survivors who practiced religious activity more frequently daily had better health behaviours for instance, greater likelihood to follow their physicians' advice, an association that was mediated by self-assurance. In contrast, higher reports of religious struggle for instance, feeling abandoned or punished by God were associated with poor health behaviour, an association that was mediated by guilt (Park, et. al., 2009). Thus, it is important



to screen cancer patients for their religious or spiritual distress and possibly for their beliefs that may influence cancer related health behaviour negatively such as beliefs that may compromise the adherence to guidelines and treatment. The quantitative finding also showed that **end of life** concerns were the most prevalent in the spiritual domain and there were no significant differences in spiritual or religious concerns in both groups, that is Emiratis and expatriates.

Spiritual or religious needs related to patients' disease can affect their mental health and failure to meet these needs may impact their quality of life (Arrey, et. al., 2016), which is an important concept of cancer care. Musick, et. al., (1998) observed that the use of spiritual or religious coping may be more commonly observed in black race since they face more discrimination and prejudice in general. Earlier studies also have reported association between religious or spiritual practices and well-being by race with higher levels of spiritual well-being in blacks than whites, yet the question remains if these differences in religious or spiritual practices are associated with differences in higher quality of life or to health disparities (Canada, et. al., 2013). Ell & Nishimoto (1989) found some notable differences in coping between Non-Hispanic, White and Hispanic cancer patients where Hispanics tend to have more difficulty accepting cancer than other groups and were more likely to rely on religion than Whites.

Nevertheless, further research is needed on whether religious or spiritual practices or cultural practices remain constant or changes across cancer trajectory. It is also important to be aware of cultural and religious differences when looking at patterns of adjustment to cancer. A knowledge of the role that religion and spirituality play in the patients' life may enable healthcare professionals to understand how religious or spiritual beliefs affect patients' response to cancer diagnosis and decisions about treatment. This study explains how survivors' faith in Allah (God) permeates all aspects of cancer survivorship, believing cancer is Allah's will and almighty is the healer. Their strong faith in Allah strengthened and guided them to live through the life-threatening disease. Therefore, while assessing cancer patients' spiritual or religious needs, healthcare providers must understand that western cultures use different coping strategies than individuals from non-western cultures (Barg & Gullatee, 2001).



5.6.6 QUALITY OF LIFE OF BREAST CANCER SURVIVORS

In this study, all participants expressed their satisfaction with the care received at the cancer facility stating they were fully satisfied with the treatment received and the staff. Yet, they made suggestions to improve cancer services stating that the presence of nutritionist and psychosocial services while receiving definitive cancer treatment would improve the current cancer care. Additionally, information delivery was one of the aspects of care that was highlighted for improvement. Other services such as efficient pharmacy services and enhanced coverage for mandate to include other comorbid conditions for those who cannot afford cancer care were few expectations from the cancer survivors who were interviewed that were suggestive of enhancing their cancer experience.

The World Health Organization (2021) defined quality of life as individuals' perceptions of their position in life in the context of the culture and value system in which they live and in relation to their goals, standards, and concerns. The results of this study show that the majority of cancer survivors deemed their quality of life as good, representing the majority of women who possessed a good quality of life (Mean=9.40). There was a **negative correlation between quality of life and the scores of psychosocial concerns including physical, social, emotional as well as spiritual concerns** (p < 0.01). The results indicated that the higher the psychosocial concerns (including physical, social, emotional as well as spiritual or religious concerns), the lowest the quality of life as in line with the literature (Edib, et. al., 2016; DiSipio, et al., 2009; Avis & Crawford, 2005). However, there was a **positive correlation between quality of life** and information concerns also increase along with it suggesting those cancer survivors who scored higher in the information domain also reflected higher scores in quality-of-life assessment which is quite consistent with the previous studies (Park & Hwang, 2012; Carey, et. al., 2012; Uchida, et. al., 2011).



5.7 REFLECTIONS

"For the story of my life is always embedded in the story of those communities from which I derive my identity....the self has to find its moral identity in and through its membership in communities such as those of the family, the neighbourhood, the city, and the tribe....without those moral particularities to begin from there would never be anywhere to begin; but it is in moving forward from such particularity that the search for the good, for the universal consist – Alasdair Macintyre, 1984 - 221"

As I reflect on my journey writing this dissertation, I think about how much I have learned and experienced personally and professionally over the last few years. The experience of being a researcher pushed me to explore my own strengths and weaknesses in a unique way both personally as well as academically which I have learnt how to overcome diligently. I feel proud of the work I have done and became the voice of vulnerable people with cancer who may have spoken for the first time since in my opinion, this was the first mixed method study on psychosocial needs exploration of cancer survivors in the United Arab Emirates.

I began my PhD because of my sheer passion in Oncology which I developed during my employment and dealing with cancer patients records and reading their stories and gathering information for healthcare providers so they could do the needful to provide the best possible care. While working in the biggest Oncology centre in the country, I have seen how breast cancer patients suffer from depression besides the consequences of illness and its treatment. I was concerned about the care survivors receive which is quite fragmented. I have raised this concern in several Oncology task force meetings during my employment. This triggered my compassion and curiosity both towards cancer patients who often are treated for their disease not as a whole.

These concerns renewed numerous times even while conducting the research when I was speaking to the cancer survivors about their concerns. My passion guided me to bridge the observed gaps between oncology practice and survivorship and I am extremely grateful for the opportunity to share my knowledge that I gathered through this research with others to make a positive impact in cancer survivors' lives. I remember, survivors overwhelmingly showed the highest gratitude and thanked me for addressing their unidentified needs and acknowledging



the existence of their ongoing and unidentified problems. I feel very proud and fortunate to take this opportunity and become the voice of the cancer survivors. As a principal investigator my study allowed me to critically explore the challenges associated with breast cancer in the cultural context. The current oncology practice in the UAE focuses mainly on treating the disease not a person. As a result, most patients either go unnoticed or accept the disease as it is, do not question local medical practices and some seek treatment abroad if they can afford to which is most likely the case. However, cancer is not a static disease and it can continue throughout a person's life. Those who return to the country need to seek or continue treatment locally. I believe very strongly that it is time to reflect and improve current local practices and incorporate survivorship care as part of the standard oncology practice so individuals could be screened for psychosocial needs associated with cancer rather than just receiving the treatment. While working in Tawam Hospital and having my workplace in the middle of the oncology ward made me even more determined to conduct this research and I felt great compassion when I used to see patients in waiting room waiting for their chemotherapy sessions. Most of them used to look fatigued and struggling with their disease. Most patients used to come for treatment sessions only. Physicians' time was also a challenge as it was mainly to allocate the right protocol as per defined cancer staged not by their psychosocial needs. I was very much aware of most of the common issues and challenges that are faced by breast cancer survivors in the UAE in terms of disease as well as healthcare practices which became the main driving force for me to conduct this research and wanting to make a difference. During the research, I personally was very touched by cancer patients and built a real connection with them as I came to understand their emotional needs. The appreciation I received from the participants motivated me to pursue my research and attempt to guide policy makers and leaders to uncover the reality that is usually not investigated by clinicians. I realised that it is important to gain patients' trust to understand and treat them well and with due consideration of the whole person. I was surprised on many occasions to know how others perceive women's stance in the UAE, without an appreciation of how very knowledgeable women are and fully able to address their concerns and guide health providers about their choices if they are given an opportunity to express themselves. I also agree that health providers need to build a partnership with their patients with shared autonomy of decision-making process and meeting their unique needs.

With regards to methodological reflections, verification of face validity of the survey and to ensure that the survey items were well formulated, it was pilot tested on five Emiratis and five



expatriate cancer survivors during their waiting time in the Radiotherapy unit at Tawam Hospital assisted by the Oncologist. The pilot testing assured the consciousness of cultural sensitivity, prior adjustments and the mutual interest of cancer survivors to take part in the research. Additionally, the opinion from the oncology physician regarding usage of appropriate wording in the survey poised the survey readiness. The survey results were further verified by reviewing their health records in the registry to resolve any discrepancies in the responses whether demographics or clinical. The comparison showed that responses were accurately recorded without any conflicts. The selection of interview questions was a meticulous process which covered all psychosocial aspects of survivors' needs. Field notes were taken for each interview to relate findings with conscious observations of speech, pauses and gestures as well as my own perspectives. Distinctions were made between data and field notes to eliminate preconceptions through self-reflection.

Analytical reflections include peer and participants' review of all analytical stages and rechecking results and interpretations. Peer reviews include residents and physicians working in the Oncology unit at Tawam Hospital Breast Care Centre who were involved in the patient care, thus adding more authenticity of findings. Moreover, two qualitative research experts were contacted to provide suggestions, review the applied thematic analysis and to verify results from the Faculty of Medicine, UAE University which is located adjacent to Tawam Hospital. There were few suggestions on refinement and addition of sub-themes which were added at a later stage of writing. Results were also shared with participants over the phone to ensure that interviewed were interpreted and reported accurately and to gain true perspectives of cancer survivors on their needs.

Since the lack of awareness of survivorship care both at patients and healthcare setting level, the findings may fall short given that cultural norms and different backgrounds and regional settings may have masked the identification of survivors' needs. Thus, caution must be taken in applying these findings to other contexts while indicating the need of further research on cancer survivors' needs addressing psychosocial forces and cultural disparities and expectations to improve their cancer experiences and survivorship. The research also suggests opportunities for innovation in the public health practices to design cancer control framework encompassing survivorship as a recognised part of cancer care continuum, develop guidelines of survivorship care as part of routine practices and encourage and develop strategies and



interventions that engage survivors in their care processes promoting shared care and patient physicians partnership models that could aim to meet their needs and improve their quality of life.



5.8 STRENGTHS AND LIMITATIONS

This study possesses some unique values and strengths. With the best of the knowledge this is the first ever study in the United Ara Emirates as well as in the region that addresses the needs and concerns of breast cancer survivors (UAE's top most common cancer) utilising mixed methods approach and adds some great value by confirming the existing and producing some new knowledge towards cancer survivorship literature. The study demonstrates the strengths and advantages of combining both qualitative and quantitative approaches in a mixed-method study. The quantitative survey gathered the information from cancer survivors identifying their needs at different points of cancer trajectory. The rich and descriptive data obtained from the qualitative interviews offer evidence-based information on breast cancer survivors' concerns which will help provide valuable insight to guide policy makers and healthcare providers establishing evidence-based survivorship programs to meet their unique needs and to improve cancer care. It also addresses the key concerns to consider for prevention and remediation efforts. Since the study is apparently the first of its kind to examine the psychosocial concerns of people living with breast cancer, it makes a significant contribution in highlighting an important area for further study in the United Arab Emirates, and furthermore benefits from using mixed methods to give more comprehensive and meaningful results in this area.

Although this study provides insight into breast cancer survivors' concerns representing the larger cancer population in the UAE, it is not without limitations. A number of factors hampered the research process among which ethics approval from both Tawam Hospital and University of Salford was predominant as it took longer than anticipated timeframe which delayed the process of data collection. Delayed approval altered the timeframe resulting in the loss of some participants as they left the country for the preferred treatment. Bear in mind that Emiratis (local) patients are supported by the government to be treated outside the UAE on its expenses, thus most Emiratis prefer to be treated at world-class facilities in countries such as Singapore, Thailand, Korea, US or UK.

An additional view that was not considered in this study was whether or not the participants received all the care at another facility whether outside the country or at another clinical setting and attended Tawam Hospital for just follow-up as there is a possibility that their concerns with the current health providers either could have been missed if the current care was given at the



optimal grounds and they were all satisfied and revealed only the current needs. Switching healthcare providers or treatment facilities might have implied some degree of bias in the evaluation of survivors' needs and this factor was not considered in the current study. Moreover, it is not uncommon that some of the cancer patients prefer to seek treatment at other private facilities either due distance to their residence or to some extent due to an inaccessibility to afford treatment which could be due to lack of insurance coverage and never visited Tawam Hospital.

Also, it is important to consider that the concerns of breast cancer survivors were assessed for a recent timeframe of the past one week which gave critical insight into women's breast cancer needs and concerns at a potentially reflective point. The approach allowed identification of concerns at that particular phase which undoubtedly overlooked acute psychosocial concerns or needs that they might have confronted earlier. However, if these were asked about the concerns, they had years prior, this would introduce a recall bias. The data obtained at one point of the study from the selective group of cancer survivors might not be able to capture the changes in breast cancer survivors' concerns over time and changes over the trajectory of cancer survivorship. Although most women were able to speak English, it still needs to be taken into consideration that for those who did not, some material and nuance will inevitably have been lost in translation from Arabic to English.

According to Doyle (2008), cancer survivorship is a dynamic and evolving journey that begins with a diagnosis and continues through the remainder of a person's life however this cross-sectional study design does not favour the assessment of psychosocial constructs over time that is; pre-diagnosis, during treatment and post-treatment to examine and explore the changes in patterns of psychosocial functioning and health behaviour across the continuum of the cancer experience. A longitudinal study of cancer survivors' needs, and their concerns is therefore required to best understand the critical and gain complete insights on ever changing unique concerns of breast cancer survivors such as assessment of needs after the diagnosis phases, during the treatment and after the treatment for instance after five years of treatment.

Other limitations including methodological as well as organisational limitations inherited in the study that might affect the generalization of the results must be acknowledged before drawing implications from the study findings. Limitations of the qualitative phases included



the potential for bias in the selection of participants through purposive sampling. The participants were selected based on their availability and willingness to be part of the study, so it is possible that interviews with other participants would have yielded different results. The qualitative results were intended to add depth and breadth to understanding the lived experienced of cancer survivors, and should not be generalised to the general population. Although Tawam Hospital is a national referral centre for cancer patients yet; the selected sample size was not fully representative of the entire cancer population in the United Arab Emirates as some patients might have not visited to Tawam for the treatment and received treatment at other facilities or abroad.

Another inconvenience was the inaccessibility recruitment site, Tawam Hospital is considerably far for the majority of the cancer patients who mostly resides in other emirates which are considered distant, thus, arranging an interview during the follow-up visit was quite a challenge as patients were required to stay even longer during their farthest visit.

Participants were largely expatriate and among most of them had health coverage whereas Emiratis were just fully covered by national health insurance and were also entitled to get treatment abroad utilising the government funds thus financial concerns were hard to assess and were not fully representative of entire population and if assessed among a different sample, would have most likely yielded different study results.

Perhaps the biggest limitation was scarcity of research into psychosocial aspects of breast cancer in this region which might not allow comparison or integration of the study findings with other populations or groups. This is the first study to explore breast cancer survivors' concerns in the UAE.



5.9 IMPLICATIONS

The psychosocial impact that cancer causes is devastating, severe and enduring and affects individuals, their families and their loved ones. While delivering cancer care, it is essential to address not only the physical but mental health also to be taken into by routine screening for any psychological or emotional distress or other related symptoms. Additionally, patients and their family must also be educated for coping mechanisms to deal with the life-threatening illness.

This study indicates that psychosocial concerns and needs of cancer patients in the UAE are continuous throughout the cancer trajectory which should be met to improve cancer survivorship. The study addresses that a considerable portion of the women desire prolonged psychosocial intervention throughout the cancer trajectory. Data also indicates that cancer survivors in the UAE have different dimensions of needs where information, emotional and physical concerns are the highest that require serious attention.

The findings provide several opportunities to improve a number of key aspects of cancer care and toward meeting the psychosocial needs of cancer survivors that are usually missed during the routine cancer care as most physicians' treat cancer patients for the biological aspect of the disease missing out psychosocial concerns. Most important is to break the silence and hear the concerns of the patients. A key barrier to addressing psychosocial well-being of cancer survivors is that providers do not ask about it, and this is compounded by their reluctance to discuss their personal concerns. It is vital that providers include the routine screening and assessment of cancer survivors' psychosocial needs including anxiety and depression as a part of care across the cancer care trajectory. Future policies could support the proactive identification of at-risk survivors within the medical setting as a standard of practice. The findings inform that a holistic oncology model is needed that includes psychosocial and survivorship care as a distinctive part of it. The higher prevalence of information and emotional concerns among breast cancer survivors living in the UAE supports the need for both active screening and the use of a range of culturally tailored interventions to address and meet those needs timely and efficiently.



Identifying and recognizing that each cancer patient has unique cancer experience, and their needs are unique too, is an important element to remember while establishing and providing cancer services. Cancer care facilities should consider establishing a structured survivorship program which should include psychosocial services including all psychosocial domains to meet the needs of all cancer patients which are unique, specific and ongoing. A comprehensive and carefully considered approach to meet the challenges and barriers in survivorship care should be used in the delivery of cancer follow-up. Survivorship care plans, including written treatment summaries and plans for follow-up care, have been suggested as a means for providing adequate information and improving survivor outcomes. Thorough examination of survivors' needs, development and testing of survivorship care models, assessment of provider knowledge as well as future research is needed to establish understanding of the barriers and possible solutions necessary to deliver high-quality cancer care.

Cancer survivorship has ascended as an important aspect of cancer care and has been recognised as a crucial component of cancer care. Standard setting organizations for cancer care, such as American Society of Clinical Oncology (ASCO) and National Cancer Institute (NCI) advocate strongly to address the needs of cancer survivors by establishing and promoting standards for oncology and cancer care and to consider psychosocial and survivorship care as part of cancer program and routine services. The National Cancer Institute (US) has established the Office of Cancer Survivorship to create more awareness of survivorship issues experienced by cancer survivors, particularly post-treatment. Health organizations in the UAE are required to raise awareness of cancer survivors' needs and recognise cancer survivorship as a distinct phase of the cancer continuum that needs to be incorporated in the standard cancer care. Risk assessment and intervention should be considered both at diagnosis and follow-up so that healthcare providers can better understand how cancer treatment may affect a patient's life as a survivor and tailor treatment accordingly. Giving the right therapy to the right person at the right time should be the goal. High risk individuals need to be identified and addressed for second malignant neoplasms by primary prevention. Additionally, appropriate and personalised treatment will help in reducing morbidity, and to minimize late effects and optimize healthcare resources.

Also, there is an urgent need to integrate evidence-based psychosocial services into current cancer care delivery from the time of diagnosis and in follow-up focusing on prevention of late



and long-term effects. This requires collection of outcomes data on survivors and continuous follow-up care.

New care delivery models will require the development and implementation of quality measures focused on survivorship care. Hence, an adequately educated and trained workforce will be needed to identify, address and manage survivorship issues to meet cancer survivors' psychological needs. Healthcare providers need to develop the core knowledge about the impact of cancer diagnosis and treatment on cancer patients' lives. Likewise, cancer survivors need to be educated and given accurate and timely information about what to expect after the diagnosis, treatment and after the completion of treatment. Clinicians might not have the expertise that they need to address many possible issues reported by survivors therefore guidance may require to address and manage survivor's physical and emotional health, health behaviours and other psychosocial concerns since these remain throughout the cancer survivorship and if unmet can negatively affect quality of life. Survivors need to be fully informed about the psychological changes that may occur, the long-term physical effects of treatment, the impact of the financial or social issues on disease and the effect of the disease on an individual and their family.

Cancer survivors should be prepared for the anxiety that is associated with follow-up medical appointments, symptoms monitoring, self-care, the end of treatment, reactions when returning to work and disease related emotions to provide them support and reassurance. They must be encouraged to seek medical assistance for all aspects of their disease and must continue medical follow-up. They should be provided full information of resources and support groups available. Moreover, family members and their caregivers also need to be prepared and educated about the process of survivorship.

This study indicates that women in the UAE express a need for an immediate intervention right after diagnosis in order to get prepared for what may occur in future. Having said that, it is important to understand and accept that healthcare systems are increasingly depending on each individual for follow-up as breast care is rapidly changing. Resources to provide psychosocial care are no exception in the UAE as they continue to be a challenge to constantly push to meet cancer patients' needs and the need of psychosocial support is dire. Thus, healthcare professionals may need to alter their understanding about how to best provide psychosocial



support women with breast cancer to enhance their experience using cost-effective techniques that help survivors to help themselves.

This could be possible by providing them with an early emotional foundation on which they can stand more firmly throughout the disease trajectory. In order to achieve this, healthcare providers need to find ways to work smarter in a system where the level of psychosocial support provision becomes a routine, and this may be accomplished by focusing on better preparing cancer survivors for what lies ahead when they are diagnosed and when they complete their treatment. It is critical to assess the quality of psychosocial services once implemented to ensure their needs are met. Structured survivorship programs will ensure that all cancer patients receive appropriate survivorship care (IOM, 2008, 2006) to provide quality cancer care encompassing all aspects of psychosocial needs of cancer survivors. Results from this study can support and guide survivorship care services focusing on information, physical and emotional needs and setting up resources to help manage these concerns for instance, fear of recurrence, genetic counselling, pain, fatigue, emotional support etc. Since "no formal survivorship program" exists either locally or at national level, thus, this study can be the foundation to address the need for a structured survivorship program and guide facilities and authorities.

Cancer survivorship has been a neglected area with little awareness both among physicians and cancer survivors. Study findings call for further research since the subject has not been given much attention and studied despite having a high incidence of cancer and having the best and modern healthcare system and facilities in the UAE. Future psychosocial research needs to be the priority to find ways to support breast cancer survivors and provide immediate effective support. Longitudinal studies are imperative to understand changes in survivor's needs and concerns overtime. Also, more knowledge about symptom management will result in improving the survivor's well-being.

Approaches to develop measurement instruments to measure these needs to allow direct comparison of patients' needs should also be considered for future research. Additionally, more research is needed using different research designs including qualitative methods and mixed methods in this area as there are only few studies that have been conducted which mostly have utilised quantitative methodology. Using qualitative and mixed methods, psychosocial



concerns could be explored, and in-depth understanding can be gained. Although this study has identified key concerns of cancer survivors, yet further research is required to explore how these needs could be met. Finally, an establishment of a psychosocial registry should be considered to make predictions and prioritizations of psychosocial services and its allocation in the future.

SUMMARY

There are thousands of cancer survivors in the United Arab Emirates who are getting older and are receiving treatment across healthcare settings. Although, UAE health systems have made a remarkable progress lately, it is important to focus on improving the quality of comprehensive survivorship care. The study findings provides a comprehensive survivorship care road map to address and systematically improve the quality of psychosocial care for cancer patients that may be used across the cancer population in the United Arab Emirates. This study proposes several strategies for healthcare systems, policy, research and cancer survivors aiming to address gaps in knowledge and policy and to implement effective and evidence-based clinical survivorship care that meets the need of cancer patients in the UAE.

Implications for Healthcare Systems

- Psychosocial distress screening & survivorship care to be part of routine care
- Develop survivorship care guidelines for clinical and psychosocial needs
- Provision of 'Treatment/survivorship care plan' to each survivor
- Coordinated, unfragmented, comprehensive and culturally sensitive patient-centred cancer care
- Enhance physician patient partnership & patient empowerment
- Transition from medical paternalism to patient autonomy
- Develop evidence-based interventions
- Education & training of health care professionals

Implications for Policy



- Reimbursement for essential post-treatment survivorship care
- Enhance cancer mandate coverage
- Develop professional guidelines and training on cancer survivorship care
- Measures to address shared decision making and patients satisfaction in health promotion and disease management
- Develop evidence-based, culturally appropriate information and resources to meet the needs of survivors

Implications for Research

- Funding for survivorship research to increase evidence base
- Further research on cancer psychosocial care and survivorship care
- Education of policy makers to increase awareness of survivorship issues

Implications for Cancer Survivors

- Awareness about psychosocial concerns & survivorship care in general public
- Develop models for cancer survivorship to help empower survivors to manage their survivorship care

The final chapter of this dissertation presents conclusions that were drawn from this research and summarizes the viewpoints that were extrapolated from the study findings.



Chapter 6: CONCLUSIONS

"Becoming human is becoming individual, and we become individual under the guidance of cultural patterns, historically created systems of meaning in terms of which we give form, order, point, and direction to our lives (Geertz, 1973, pg. 13)".

There are thousands of adults and children in the UAE who are cancer survivors. Most of them are often relieved by the end of the treatment yet the transition to a new normal could be quite challenging which may impact them not just physically but also mentally and emotionally. It may require adjustment to a new world, new emotions, new concerns, new ways of support as their perspective to the world changes after being diagnosed with cancer.

Comprehensive cancer care delivery requires coordinated care and the provision of services that meets the unique needs of cancer survivors. Physicians-led care can be suboptimal; hence healthcare professionals need to work in partnership with cancer survivors to offer a holistic approach that addresses and meet cancer survivors needs throughout the cancer continuum by developing new models of shared care. This research emphasises on the critical and often neglected role of healthcare providers in addressing cancer survivors needs which may differ among the diverse UAE population. Additionally, it also underscores the increasingly important role of the breast cancer survivors

Cancer survivorship awareness and the rise of cancer survivors in the UAE has enabled the need of more coherent research in this region to acquire evidence-based knowledge about this unexplored area. Standard cancer care practices in the UAE need to implement standard guidelines for survivors' clinical and psychosocial needs which are often lacking in routine care. Each cancer survivor needs to be provided with a survivorship care plan while transitioning to post-treatment period as part of a standard oncology practice. for each survivor. There is a need to identify gaps in cancer survivorship research in the UAE including survivors' psychosocial needs, need of coordinated care, management of long-term and late psychosocial effects, need to understand ethno-cultural disparities among cancer survivors which may lead to the development an efficient healthcare system and delivery of more comprehensive, coordinated and evidence-based quality care to the UAE residents. Moreover, such applied


research can guide and form the basis of development of evidence-based guidelines for the provision of culturally sensitive and quality cancer care.

Nonetheless, without integrative care models, advocacy, proper reimbursement and placement of incentives for healthcare professionals to administer survivorship care provided limited resources and time constraints could be challenging. Thus, appropriate policies and advocacy efforts will be needed to ensure the implementation of person-centred compassionate quality cancer care in the UAE.

This research fills the existing knowledge gap regarding the breast cancer survivors' psychosocial needs in the United Arab Emirates. The mixed method design adds to the literature that breast cancer survivors experience a variety of psychosocial health effects resulting from cancer and its treatment which impact their lives in numerous ways. To navigate with these changes and adapting to a new normal is a dynamic process of adjustment to life changes. Survivors' needs could differ by age, gender, ethnicity and clinical presentation of the cancer; thus, it is important to design and implement culturally relevant intervention to meet their unique needs. Although this research was limited to breast cancer survivors only in one geographic region of the country, this study provides a conceptual model to develop an integrated care model to enhance physician patient partnership. Nevertheless, further understanding is needed to refine this model to address the dynamics and multifaceted aspects of breast cancer survivors needs by conducting further research which can serve as a framework to guide future oncology practice in the UAE.



Chapter 7: REFERENCES

Abdel, H. M., (2000). Breast cancer awareness among health professionals. Ann Saudi Med20:135-136

Acton, Q. A., (2011). Issues in Cancer Treatment: 2011 Edition. Published by Scholarly Editions.

AbdelHadi, M. S., (2006). Breast cancer awareness camping: Will it make a difference? Journal of Family Community Medicine, 13(3), 115-118. Retrieved from http://www-ncbi-nlm-nih-gov.ezproxyhost.library.tmc.edu/pmc/articles/PMC3410058/ [accessed on Sep 18, 2018].

Abulkhair, O. A., Al-Tahan, F. M., Young, S. E., Musaad, S. M. A. & Jazieh, A. R. M., (2010). The first national public breast cancer screening program in Saudi Arabia. Annual of Saudi Medicine, 30(5), 350–357

Adachi, K., Ueno, T., Fujioka, T., Fujitomi, Y., Ueo, H., (2007). Psychosocial factors affecting the therapeutic decision-making and postoperative mood states in Japanese breast cancer patients who underwent various types of surgery: body image and sexuality. Jpn J Clin Oncol. 2007 Jun;37(6):412-8. doi: 10.1093/jjco/hym041. Epub 2007 May 18. PMID: 17513308.

Adams, E., Boulton, M., & Watson, E. (2009). The information needs of partners and family members of cancer patients: Systematic literature review. Patient Education & Counseling, 77, 179–186.

Adams, R. N., Mosher, C. E., Rand, K. L., Hirsh, A. T., Monahan, P. O., Abonour, R., & Kroenke, K. (2017). The Cancer Loneliness Scale and Cancer-related Negative Social Expectations Scale: development and validation. Quality of life research: an international journal of quality-of-life aspects of treatment, care and rehabilitation, 26(7), 1901–1913. https://doi.org/10.1007/s11136-017-1518-4



Adkinson, J. M., Miller, N. F., Eid, S. M., Miles, M. G., Murphy, R. X., (2015). Tissue Expander Complications Predict Permanent Implant Complications and Failure of Breast Reconstruction. Annals of Plastic Surgery 2015 Jul;75(1):24-8. doi: 10.1097/SAP.00000000000142. DOI: 10.1097/SAP.0000000000142 [accessed on Mar 12, 2018].

Adler, A. (1927). The practice and theory of individual psychology (P. Radin, Trans.). New York: Harcourt, Brace & Company.

Adler, A. (1954). Understanding human nature (W. B. Wolf, Trans.). New York: Fawcett Premier. (Original work published 1927).

Adler, E. N., & Page, E. K. A., (2008). Cancer Care for the Whole Patient: Meeting Psychosocial Health Needs. Institute of Medicine (US) Committee on Psychosocial Services to Cancer Patients/Families in a Community Setting. Washington (DC): National Academies Press (US). Available at: https://www.ncbi.nlm.nih.gov/books/NBK4017/ [accessed on Mar 18, 2018].

Adorna, E. L., Morari-Cassol, E. G., Ferraz, N. M. S., (2015). Breast cancer and support group in the life of mastectomized women. ISSN 0103-5150 Fisioter. Mov., Curitiba, v. 28, n. 3, p. 597-602, July/Sept. 2015 Licenciado sob uma Licença Creative Commons DOI: http://dx.doi.org.10.1590/0103-5150.028.003.AO18.

http://www.scielo.br/pdf/fm/v28n3/0103-5150-fm-28-3-0597.pdf [accessed on Sep 18, 2018].

Advisory Board, (2018). Integrating Psychosocial Risk Factors into Ongoing Care How to identify patient needs and tailor care planning to optimize outcomes. RESEARCH REPORT. https://www.advisory.com/-/media/Advisory-com/Research/PHA/Research-

Briefings/2018/PHA-Psychosocial%20Risk%20Factors%20Brief.**pdf** [accessed on Sep 18, 2018].

Aghaei, M.H., Vanaki, Z. & Mohammadi, E., (2021). Inducing a sense of worthiness in patients: the basis of patient-centered palliative care for cancer patients in Iran. *BMC Palliat Care* 20, 38 (2021). https://doi.org/10.1186/s12904-021-00732-3



Agbokey, F., Kudzawu, E., Dzodzomenyo, M., Ae-Ngibise, K. A., Owusu-Agyei, S., & Asante, K. P. (2019). Knowledge and Health Seeking Behaviour of Breast Cancer Patients in Ghana. International journal of breast cancer, 2019, 5239840. https://doi.org/10.1155/2019/5239840

Agency for HealthCare Research and Quality, (2015). Consider Culture, Customs, and Beliefs Tool 10. Second Edition, available at http://www.ahrq.gov/literacy. https://www.ahrq.gov/sites/default/files/wysiwyg/professionals/quality-patient-safety/quality-resources/tools/literacy-toolkit/healthlittoolkit2_tool10.pdf [accessed on Sep 18, 2018].

Ahmad, F., & Binti, M & Abdullah, A., (2010). Religion and Spirituality in Coping with Advanced Breast Cancer: Perspectives from Malaysian Muslim Women. Journal of religion and health. 50. 36-45. 10.1007/s10943-010-9401-4.

Ahmad, F., Muhammad, M., & Abdullah, A. A. (2011). Religion and spirituality in coping with advanced breast cancer: Perspectives from Malaysian Muslim women. Journal of Religion and Health, 50(1), 36–45.

Australian Institute of Health and Welfare (2014). The measurement of patient experience in non-GP primary health care settings. Cat. no. WP 66. Canberra: AIHW.

Akechi, T., Okuyama, T., Endo, C., Sagawa, R., Uchida, M., Nakaguchi, T., et, al., (2011). Patient's perceived need and psychological distress and/or quality of life in ambulatory breast cancer patients in Japan. Psycho-Oncology. 2011;20(5):497–505.

Al-Abri, R., & Al-Balushi, A. (2014). Patient satisfaction survey as a tool towards quality improvement. Oman medical journal, 29(1), 3–7. https://doi.org/10.5001/omj.2014.02

Al-Amoudi, S, Cañas, J, Hohl, S. D., Distelhorst, S. R., Thompson, B., (2013). Breaking the Silence: Breast Cancer Knowledge and Beliefs Among Somali Muslim Women in Seattle, Washington. Health Care for Women International, 36:608–616, 2015. Taylor & Francis Group, LLC DOI: 10.1080/07399332.2013.857323Taylor & Francis Group. Available at: http://web.a.ebscohost.com.salford.idm.oclc.org/ehost/pdfviewer/pdfviewer?vid=1&sid=e334 8ee7-f777-4eb6-8ad3-5f5187e59fa5%40sessionmgr4009 [accessed on Sep 18, 2018].



Al-Amri., A. M., (2009). Cancer patients' desire for information: a study in a teaching hospital in Saudi Arabia. Eastern Mediterranean Health journal, 15(1). http://applications.emro.who.int/emhj/1501/15_1_2009_0019_0024.pdf [accessed on Dec 19, 2018].

Al-Azri MH, Al-Awisi H, Al-Rasbi S, Al-Moundhri M. (2013). Coping with a diagnosis of breast cancer among Omani women. J Health Psychol 2013 Mar;19(7): 836-846. doi: 10.1177/1359105313479813. Epub 2013 Mar 21. PMID: 23520353.

Al-Azri, M., Al-Awisi, H., Al-Rasbi, S., El-Shafie, K., Al-Hinai, M., Al-Habsi, H., & Al-Moundhri, M. (2014). Psychosocial impact of breast cancer diagnosis among Omani women. Oman medical journal, 29(6), 437–444. https://doi.org/10.5001/omj.2014.115

Alagraa, R., Abujaber, A., Chandra, P., Doughty, J., (2015). Evaluating psychosocial support needs of female cancer patients in the State of Qatar. Qatar medical journal, 2015, Vol.2015(1), pp.4. Available from: http://www.qscience.com/doi/10.5339/qmj.2015.4

Alananzeh, I., Levesque, J., Kwok, C., & Everett, B. (2016). Integrative Review of the Supportive Care Needs of Arab People Affected by Cancer. Asia-Pacific Journal of Oncology Nursing, 3(2), 148–156. http://doi.org/10.4103/2347-5625.177396 ali et al 1999 [accessed on Dec 19, 2018].

Alase, A., (2017). The Interpretative Phenomenological Analysis (IPA): A Guide to a Good Qualitative Research Approach. International Journal of Education & Literacy. Vol. 5 No. 2; April 2017. Australian International Academic Centre, Australia. doi: 10.7575/aiac.ijels. v.5n.2p.9

Alden, D. L., Friend, J., Lee, P. Y., Lee, Y. K., Trevena, L., Ng, C. J., Kiatpongsan, S., Lim, Abdullah, K., Tanaka, M., Limpongsanurak, S., (2018). Who Decides: Me or We? Family Involvement in Medical Decision Making in Eastern and Western Countries. Med Decis Making. 2018 Jan;38(1):14-25. doi: 10.1177/0272989X17715628. Epub 2017 Jul 8. PMID: 28691551.



Aldosari, H., (2017). The Effect of Gender Norms on Women's Health in Saudi Arabia. Policy Paper #1. https://agsiw.org/wp-content/uploads/2017/05/Aldosari_Womens-Health_Online-2.pdf [accessed on Dec 19, 2018].

Alexander, A., Kaluve, R., Prabhu, J. S., Korlimarla, A., Manjunath, S., Patil, S., & Kodaganur, G., (2019). The Impact of Breast Cancer on the Patient and the Family in Indian Perspective. Indian journal of palliative care, 25(1), 66–72. doi: 10.4103/IJPC.IJPC 158_18

Al-Ghazal, S. K., Fallowfield, L., Blamey, R. W., (2000). Comparison of psychological aspects and patient satisfaction following breast conserving surgery, simple mastectomy and breast reconstruction. Eur J Cancer. 2000;36(15):1938–43.

Allemani, C., Weir, H. K., Carreira, H., Harewood, R., Spika, D., Wang, X. S., Bannon, F., Ahn, J. V., Johnson, C. J., Bonaventure, A., Marcos-Gragera, R., Stiller, C., e Silva, G. A, Chen, W. Q., Ogunbiyi, O. J., Rachet, B., Soeberg, M. J., You, H., Matsuda, T., Bielska-Lasota, M., Storm, H., Tucker, T. C., Coleman, M. P., (2015). CONCORD Working Group. Global surveillance of cancer survival 1995-2009: analysis of individual data for 25,676,887 patients from 279 population-based registries in 67 countries (CONCORD-2). Lancet. 2015 Mar 14;385(9972):977-1010. doi: 10.1016/S0140-6736(14)62038-9. Epub 2014 Nov 26. Erratum in: Lancet. 2015 Mar 14;385(9972):946. PMID: 25467588; PMCID: PMC4588097.

Allemani, C., Matsuda, T., Di Carlo, V., Harewood, R., Matz, M., Nikšić, M., Bonaventure, A., Valkov, M., Johnson, C. J., Estève, J., Ogunbiyi, O. J., Azevedo E Silva, G., Chen, W. Q., Eser, S., Engholm, G., Stiller, C. A., Monnereau, A., Woods, R. R., Visser, O., Lim, G. H., ... CONCORD Working Group (2018). Global surveillance of trends in cancer survival 2000-14 (CONCORD-3): analysis of individual records for 37 513 025 patients diagnosed with one of 18 cancers from 322 population-based registries in 71 countries. Lancet (London, England), 391(10125), 1023–1075. https://doi.org/10.1016/S0140-6736(17)33326-3

Allgar, V. L, Chen, Hong., Richfield, Ed., Currow, D., Macleod, Una., & Johnson, M. J., (2018). "Psychometric Properties of the Needs Assessment Tool—Progressive Disease Cancer in U.K. Primary Care." *Journal of Pain and Symptom Management* 56.4 (2018): 602-12. Web.



Alkire, S. (2005). Valuing freedoms: Sen's capability approach and poverty reduction. Oxford, UK: Oxford University Press

Almalki, S., (2016). Integrating Quantitative and Qualitative Data in Mixed Methods Research—Challenges and Benefits. Journal of Education and Learning; Vol. 5, No. 3; 2016. Published by Canadian Center of Science and Education. doi:10.5539/jel.v5n3p288. URL: http://dx.doi.org/10.5539/jel.v5n3p288

Almazrouei, H & Pech, R., (2015). Working in the UAE: expatriate management experiences. Journal of Islamic Accounting and Business Research. 6. 73-93. 10.1108/JIABR-08-2013-0032.

Ali, N. S., Khalil, H. Z., Yousef, W., (1993). A comparison of American and Egyptian cancer patients' attitudes and unmet needs. Cancer Nurs. 1993 Jun; 16(3):193-203.

Allicock, M., Carr, C., Johnson, L. S., Smith, R., Lawrence, M., Kaye, L., Gellin, M., Manning, M., (2014). Implementing a one-on-one peer support program for cancer survivors using a motivational interviewing approach: results and lessons learned. J Cancer Educ. 2014 Mar; 29(1):91-8

Allicock, M., Haynes-Maslow, L., Johnson, L. S., Carpenter, W. R., Vines, A. I., Belle, D. G., Phillips, R., & Cherry, M. W. (2017). Peer Connect for African American breast cancer survivors and caregivers: a train-the-trainer approach for peer support. Translational behavioral medicine, 7(3), 495–505. https://doi.org/10.1007/s13142-017-0490-4

Alfano, C. M., & Rowland, J. H. (2006). Recovery issues in cancer survivorship: A new challenge for supportive care. Journal of Cancer, 12(5), 432-443

Alfano, C. M., Kent, E. E., Padgett, L. S., Grimes, M., de Moor, J. S., (2017). Making cancer rehabilitation services work for cancer patients: recommendations for research and practice to improve employment outcomes. PM R. 2017;9:S398-S406.

Alfano, C. M., Leach, C. R., Smith, T. G., Miller, K. D., Alcaraz, K. I., Cannady, R. S., Wender, R. C., Brawley, O. W., (2019). Equitably improving outcomes for cancer survivors and supporting caregivers: A blueprint for care delivery, research, education, and policy. CA



Cancer J Clin. 2019 Jan;69(1):35-49. doi: 10.3322/caac.21548. Epub 2018 Oct 30. PMID: 30376182.

Almegewly, W., Gould, D., & Anstey, S., (2018). Hidden voices: an interpretative phenomenological analysis of the experience of surviving breast cancer in Saudi Arabia. https://doi.org/10.1177/1744987118809482. Journal of research in nursing. Volume: 24 issue: 1-2, page(s): 122-132

Alhusban, R. Y., (2019). Changed body image as perceived by Jordanian women undergoing breast Cancer treatment. Asian Pac J Cancer Prev. 2019;20(3):767–73.

Almeida, S. N., Elliott, R., Silva, E. R., & Sales, C. M. D., (2018). Fear of cancer recurrence: A systematic review and meta-synthesis of patients' experiences. Clinical Psychology Review. Elsevier.

https://pure.strath.ac.uk/ws/portalfiles/portal/85300331/Almeida_etal_CPS_2018_Fear_of_ca ncer_recurrence.pdf

Almyroudi, A., Degner, L. F., Paika, V., Pavlidis, N., Hyphantis, T., (2010). Psycho-Oncology, August 2011, Vol.20(8), pp.871-879. DOI: 10.1002/pon.1798

Al-Naggar, R. A., & Bobryshev, Y. V., (2012). Practice and barriers of mammography among Malaysian women in the general population. Asian Pac J Cancer Prev. 2012; 13:3595–600.

Alqaissi, N. M., & Dickerson, S. S., (2010). Exploring common meanings of social support as experienced by Jordanian women with breast cancer. Cancer Nursing, 33(5), 353-361.

Al-Qadire, M., (2014). Jordanian cancer patients' information needs and information seeking behaviour: A descriptive study. European Journal of Oncology Nursing, 18(1) 46–51.

Al-Shamsi, H. O., & Alrawi, S., (2018). Breast cancer screening in the United Arab Emirates: is it time to call for a screening at an earlier age? Journal of Cancer Prevention & Current Research. 2018;9(3):123–126. Cancer Prev Curr Res. DOI: 10.15406/jcpcr.2018.09.00334. https://medcraveonline.com/JCPCR/JCPCR-09-00334.pdf



Al-Sharbatti, S. S., Shaikh, R. B., Mathew, E., & Al-Biate, M. A., (2014). Assessment of Breast Cancer Awareness among Female University Students in Ajman, United Arab Emirates. Sultan Qaboos University Medical Journal. 14(4): e522-529

Al-Shaaobi, A., Alahdal, M., Yu, S., & Pan, H., (2021). The efficiency of distress thermometer in the determination of supporting needs for cancer inpatients, Libyan Journal of Medicine, 16:1, DOI: 10.1080/19932820.2021.1957199

Amatya, B., A., Khan, F., Ng, L., & Galea, M., (2014). Supportive Care Needs following Cancer Treatment: A Comparison of Breast and Brain Cancer in an Australian Cohort. International Scholarly Research Notices, vol. 2014, Article ID 945472, 10 pages, 2014. https://doi.org/10.1155/2014/945472

Amiel, C. R., Fisher, H. M., & Antoni, M. H. (2016). Concerns about Breast Cancer, Pain, and Fatigue in Non-Metastatic Breast Cancer Patients Undergoing Primary Treatment. Healthcare (Basel, Switzerland), 4(3), 62. https://doi.org/10.3390/healthcare4030062

American Cancer Society, (2016). Financial Problems Lower Many Cancer Survivors' Quality of Life. https://www.cancer.org/latest-news/financial-problems-lower-many-cancer-survivors-quality-of-life.html [accessed on Dec 19, 2018].

American Cancer Society, (2019). Breast Cancer Facts & Figures 2019-2020. Atlanta: American Cancer Society, Inc. 2019. https://www.cancer.org/content/dam/cancer-org/research/cancer-facts-and-statistics/breast-cancer-facts-and-figures/breast-cancer-facts-and-figures-2019-2020.pdf [accessed on Feb 12, 2021].

American Cancer Society, (2019a). Cancer Treatment & Survivorship. Facts & Figures 2019-2021. Atlanta: American Cancer Society; 2019. [accessed on Feb 02, 2019].

American Cancer Society, (2019b). Treatment of Breast Cancer Stages I-III https://www.cancer.org/cancer/breast-cancer/treatment/treatment-of-breast-cancer-by-stage/treatment-of-breast-cancer-stages-i-iii.html. [accessed on Feb 02, 2019].



AmericanCancerSociety,(2019c).ChemotherapyforCancer.https://www.cancer.org/cancer/breast-cancer/treatment/chemotherapy-for-breast-cancer.html.[accessed on Feb 02, 2019].

American Cancer Society, (2020). For People at Risk of Lymphedema

https://www.cancer.org/content/dam/CRC/PDF/Public/8902.00.pdf . [accessed on Feb 12, 2020].

American Cancer Society, (2021). Survival Rates for Breast Cancer. https://www.cancer.org/cancer/breast-cancer/understanding-a-breast-cancer-diagnosis/breast-cancer-survival-rates.html. [accessed on Jan 0, 2021].

American Cancer Society, (2021b). Cancer Facts & Figures, 2021. https://www.cancer.org/content/dam/cancer-org/research/cancer-facts-and-statistics/annualcancer-facts-and-figures/2021/cancer-facts-and-figures-2021.pdf. [accessed on Jan 01, 2021].

AmericanCancerSociety(2021c).Universalhealthcoverage.https://canceratlas.cancer.org/taking-action/universal-health-coverage/

AmericanCancerSociety.,(2021a).WeightChanges.https://www.cancer.org/treatment/treatments-and-side-effects/physical-side-effects/eating-problems/weight-changes.html [Accessed on 25th May 2021]VeightVeight

American Educational Research Association., (2006)., "Standards for Reporting on Empirical Social Science Research in AERA Publications." Educational Researcher 35 (6): 33-40.

American Society of Clinical Oncology (ASCO), (2017). Survivorship Dictionary. https://www.cancer.net/sites/cancer.net/files/cancer_survivorship.pdf [accessed on Dec 19, 2018].

American Society of Clinical Oncology, (2007). Ensuring Continuity of Care Through Electronic Health Records. Recommendations from the ASCO Electronic Health Record



Roundtable. Journal of Oncology Practice 2007 3:3, 137-142. DOI: 10.1200/JOP.0733501 Journal of Oncology Practice 3, no. 3 (May 01, 2007) 137-142

Amin, T. T., Al Mulhim A. R. S., Al Meqihwi, A., (2009). Breast cancer knowledge, risk factors and screening among adult Saudi women in a primary health care setting. Asian Pacific Journal of Cancer Prevention. 10: 133–138.

Amir, Z., Wilson, K., Hennings, J., Young, A., (2012) The meaning of cancer: implications for family finances and consequent impact on lifestyle, activities, roles and relationships. Psycho-oncology. 2012;21(11):1167–74.

Andersen, B. L., Anderson, B., deProsse, C., (1989). Controlled prospective longitudinal study of women with cancer: II. Psychological outcomes. J Consult Clin Psychol. 1989 Dec;57(6):692-7. doi: 10.1037//0022-006x.57.6.692. PMID: 2600239; PMCID: PMC2719967.

Anderson, B. o., Dvaladze, A., Ilbawi, A., Luciani, S., Torode, J., Zujewski, J. A., (2017). Breast Cancer Initiatives Planning: Improving Access to Breast Cancer Care. (NCI). https://www.fredhutch.org/content/dam/www/research/divisions/public-healthsciences/epidemiology/bci-25/KSPDF/KS%20Planning%20Access%20030617.pdf

Andrulis, D. P., & Brach, C. (2007). Integrating Literacy, Culture, and Language to Improve Health Care Quality for Diverse Populations. American Journal of Health Behavior, 31(Suppl 1), S122–S133. http://doi.org/10.5555/ajhb.2007.31.supp.S122

Andrykowski, M. A., & S. L. Manne. (2006). Are psychological interventions effective and accepted by cancer patients? I. Standards and levels of evidence. Annals of Behavioral Medicine 32(2):93–97.

Andrykowski, M. A., Lykins, E., & Floyd, A. (2008). Psychological health in cancer survivors. Seminars in oncology nursing, 24(3), 193–201. https://doi.org/10.1016/j.soncn.2008.05.007

Angel, R., & Thoits, P., (1987). The impact of culture on the cognitive structure of illness. Cult Med Psych 11, 465–494 https://doi.org/10.1007/BF00048494



Anney, V. N., (2014). Ensuring the quality of the findings of qualitative research: looking at Trustworthiness Criteria. Journal of Emerging Trends in Educational Research and Policy Studies, 5(2), 272-281.

Ano, G. G., & Vasconcelles, E. B., (2005). Religious coping and psychological adjustment to stress: a meta-analysis. Journal of Clinical Psychology, 2005 Apr; 61(4):461-80.

Ankem, K., (2005). Types of information needs among cancer patients: a systematic review. Available at: http://www.libres-ejournal.info/wpcontent/uploads/2014/06/Vol15_I2_AnkemTextFinal-with-Tables.pdf

Ankem, K. (2006). Factors Influencing Information needs among cancer patients: a metaanalysis. Library & Information Science research, 28, 7-23.

Anstett, P., (2016). Breast Cancer Surgery and Reconstruction: What's Right for You Publisher: Rowman & Littlefield.

Antoni, M. H., Lechner, S. C., Kazi, A., Wimberly, S. R., Sifre, T., Urcuyo, K. R., Phillips, K., Glück, S., & Carver, C. S. (2006). How stress management improves quality of life after treatment for breast cancer. Journal of consulting and clinical psychology, 74(6), 1143–1152. https://doi.org/10.1037/0022-006X.74.6.1152

Arafa, M. A., Rabah, D. M., Farhat, K. H., (2020). Rising cancer rates in the Arab World: now is the time for action. East Mediterr Health J. 2020;26(6):638–640. https://doi.org/10.26719/emhj.20.073

Aramo-Immonen, H., (2013). Mixed Methods Research Design. In: Lytras M.D., Ruan D., Tennyson R.D., Ordonez De Pablos P., García Peñalvo F.J., Rusu L. (eds) Information Systems, E-learning, and Knowledge Management Research. WSKS 2011. Communications in Computer and Information Science, vol 278. Springer, Berlin, Heidelberg. DOI: https://doi.org/10.1007/978-3-642-35879-1_5

Arif, N., & Ghezzi, P., (2018). Quality of online information on breast cancer treatment options. Breast. 2018 Feb; 37:6-12. doi: 10.1016/j.breast.2017.10.004. Epub 2017 Oct 15. PMID: 29040893



Arman, M, & Rhensfeldt, A., (2003). Living with breast cancer: a challenge to expansive and creative forces. Eur J Cancer Care. 2003; 11:290-296.

Arman, M., Rehensfeldt, A., Lindholm, L., Hamrin, E., & Eriksson, K., (2004). Suffering related to health care: a study of breast cancer patients' experiences. Int J Nurs Pract. 2004; 10:248-256.

Armstrong, R., Waters, E., Moore, L., Riggs, E., Cuervo, L. G., Lumbiganon, P., & Hawe, P., (2008). "Improving the Reporting of Public Health Intervention Research: Advancing TREND and CONSORT." Journal of Public Health 30 (1): 103-9.

Arnold, M., Rutherford, M. J., Bardot, A., Ferlay, J., Andersson, T. M., Myklebust, T. Å., Tervonen, H., Thursfield, V., Ransom, D., Shack, L., Woods, R. R., Turner, D., Leonfellner, S., Ryan, S., Saint-Jacques, N., De, P., McClure, C., Ramanakumar, A. V., Stuart-Panko, H., Engholm, G., Walsh, P. M., Jackson, C., Vernon, S., Morgan, E., Gavin, A., Morrison, D. S., Huws, D. W., Porter, G., Butler, J., Bryant, H., Currow, D. C., Hiom, S., Parkin, D. M., Sasieni, P., Lambert, P. C., Møller, B., Soerjomataram, I., Bray, F., (2019). Progress in cancer survival, mortality, and incidence in seven high-income countries 1995-2014 (ICBP SURVMARK-2): a population-based study. Lancet Oncol. 2019 Nov;20(11):1493-1505. doi: 10.1016/S1470-2045(19)30456-5. Epub 2019 Sep 11. PMID: 31521509; PMCID: PMC6838671.

Arnott-Smith, C., & Keselman, A., (2015). Chapter 8: Health Information delivery outside the clinic in a developing nation: The Qatar Cancer Society in the State of Qatar. 8.5 Barriers to healthcare in Qatar. Pg 176.

Aronson, J., (1994). A pragmatic view of thematic analysis. The Qualitative Report, 2, 1–3. Retrieved from http://www.nova.edu/ssss/QR/BackIssues/QR2-1/aronson.html

Arroyo, J. M. G., & López, M. L. D., (2011). "Psychological Problems Derived from Mastectomy: A Qualitative Study", International Journal of Surgical Oncology, vol. 2011, Article ID 132461, 8 pages, 2011. https://doi.org/10.1155/2011/132461

Arora, N. K., Johnson, P., Gustafson, D. H., McTavish, F., Hawkins, R. P., Pingree, S., (2002). Barriers to information access, perceived health competence, and psychosocial health



outcomes: test of a mediation model in a breast cancer sample. Patient Education and Counseling. Vol.47(1):37–46.

Arrey, A. E., Bilsen, J., Lacor, P., Deschepper, R., (2016). Spirituality/Religiosity: A cultural and psychological resource among sub-saharan african migrant women with HIV/AIDS in belgium. PLoS One 2016;11:e0159488.

Asadi-Lari, M., Tamburini, M., & Gray, D., (2004). Patients' needs, satisfaction, and health related quality of life: Towards a comprehensive model. Health and Quality of Life Outcomes 2004, Vol.2:32. DOI: 10.1186/1477-7525-2-32

Aschenbrenner, S. A., & Venable, J. S., (2009). Chapter 12: Culture and Inherited Traits Consideration in drug therapy. Drug Therapy in Nursing, Third edition. Wolters Kluwer Health. Lippincott Williams& Wilkins. The Point.

Ashley, L., Marti, J., Jones, H., Velikova, G., Wright, P., (2015). Illness perceptions within 6 months of cancer diagnosis are an independent prospective predictor of health-related quality of life 15 months post-diagnosis. Psychooncology. 2015 Nov; 24(11):1463-70.

Ashing-Giwa, K. T., Padilla, G., Tejero, J., Kraemer, J., Wright, K., Coscarelli, A., Clayton, S., Williams, I., & Hills, D. (2004). Understanding the breast cancer experience of women: a qualitative study of African American, Asian American, Latina and Caucasian cancer survivors. Psycho-oncology, 13(6), 408–428. doi:10.1002/pon.750

Ashing, K. T., Loscalzo, M., Burhansstipanov, L., Wang, J. H & Napoles, A., (2016). Attending to distress as part of quality, comprehensive cancer care: gaps and diversity considerations Pages 257-259 Published online: 14 Jun 2016. https://www.tandfonline.com/doi/full/10.1080/23809000.2016.1195689

Ashrof, M. V. A., (2005). Islam and Gender Justice: Questions at the Interface. Chapter6: Who is an Ideal Muslim Woman? Kalpaz Publications.

Assaf, G. N. (2011). A Two-Stage Exploratory Study of the Coping Patterns of Arab Women with Breast Cancer Residing in the United Arab Emirates. Retrieved from A Two Stage Exploratory Study of the Coping Patterns of Arab Women with Breast Cancer Residing in the



Emirates

United Arab https://researchbank.rmit.edu.au/eserv.php?pid=rmit:160065&dsID=Assaf.pdf.

Assaf, G. N., Holroyd, E., & Lopez, V., (2017). Isolation and prayer as means of solace for Arab women with breast cancer: An in-depth interview study. Psycho-Oncology. Volume 26, Issue 11. https://doi-org.salford.idm.oclc.org/10.1002/pon.4402

Astrow, A. B., Wexler, A., Texeira, K., He, M. K., & Sulmasy, D. P., (2007). Is failure to meet spiritual needs associated with cancer patients' perceptions of quality of care and their satisfaction with care? Journal of clinical oncology: official journal of the American Society of Clinical Oncology. Vol.25(36):5753-7.

Astrow, A. B., Kwok, G., Sharma, R. K., Fromer, N., Sulmasy, D., (2016). Just what are spiritual needs of cancer patients? An empirical study in a diverse population. Journal of Clinical Oncology. Volume 34, Issue 15_suppl. American Society of Clinical Oncology. https://ascopubs.org/doi/abs/10.1200/JCO.2016.34.15_suppl.10005

Astrow, A. B., Kwok, G., Sharma, R. K., Fromer, N., Sulmasy, D. P., (2017). Spiritual Needs and Perception of Quality of Care and Satisfaction with Care in Hematology/Medical Oncology Patients: A Multicultural Assessment. Volume, 55, ISSUE 1, P56-64. DOI: https://doi.org/10.1016/j.jpainsymman.2017.08.009

Attride-Stirling, J., (2001). Thematic networks: an analytic tool for qualitative research. Qualitative Research, 1(3), 385-405.

Au, A., Lam, W., Tsang, J., et, al, (2013). Supportive care needs in Hong Kong Chinese women confronting advanced breast cancer. Psych oncology, 22, 1144-51

Avasthi, A., Grover, S., & Sathyanarayana Rao, T. S. (2017). Clinical Practice Guidelines for Management of Sexual Dysfunction. Indian journal of psychiatry, 59(Suppl 1), S91–S115. doi:10.4103/0019-5545.196977

Avestan, Z., Pakpour, V., Rahmani, A., Mohammadian, R., Soheili, A., (2019). The Correlation between Respecting the Dignity of Cancer Patients and the Quality of Nurse-



Patient Communication. Indian J Palliat Care. 2019 Apr-Jun; 25(2): 190–196. doi: 10.4103/IJPC.IJPC_46_18. https://europepmc.org/articles/pmc6504734

Avis, N. E., & Crawford, S., (2005). Manuel J. Quality of life among younger women with breast cancer. J Clin Oncol. 2005;23(15):3322–30.

Aw, T. C., Zoubeidi, T., Al Maskari, F., & Blair, I., (2011). Challenges and Strategies for Quantitative and Qualitative Field Research in the United Arab Emirates. Asian Pacific Journal of Cancer Prevention, 12, 1641-45.

Axon, A., Hassan, M., Niv, Y., Beglinger, C., Rokkas, T., (2008). Ethical and legal implications in seeking and providing a second medical opinion. Dig Dis. 2008;26(1):11-7. doi: 10.1159/000109379. Epub 2007 Dec 14. PMID: 18097139.

Aydiner, A., Igci, A., Soran, A., (2019). Breast Disease. Management and therapies, Vol. 2 Secon Edition. Springer Nature Switzerland

Azadi, A., Abdekhoda, M., Habibi, S., (2018). Medical Student's Skills in Notifying Bad News Based on SPIKES Protocol. J Liaquat Uni Med Health Sci. 2018;17(04):249-54. doi: 10.22442/jlumhs.181740587. https://www.lumhs.edu.pk/jlumhs/Vol17No04/pdfs/10.pdf

Azhar, A., & Bruera, E., (2018). Outcome measurement and complex physical, psychosocial and spiritual experiences of death and dying. Home / Vol 7, Supplement 3 (October 2018): Annals of Palliative Medicine (Outcome Measurement for Advanced Disease Patients). doi: 10.21037/apm.2018.07.04. http://apm.amegroups.com/article/view/20717/21263

Aziz, N. M., (2002a). Cancer Survivorship Research: Challenge and Opportunity, The Journal of Nutrition, Volume 132, Issue 11, November 2002, Pages 3494S–3503S, https://doi.org/10.1093/jn/132.11.3494S

Aziz, N. M., (2002). Long-term survivorship: Late effects. In: Principles and practice of palliative care and supportive oncology. Berger AM, Portenoy RK, Weissman DE, editors. 2nd ed. Philadelphia, PA: Lippincott Williams & Wilkins; 2002. p 101933.



Aziz, N. M., & Rowland, J. H., (2002). Cancer survivorship research among ethnic minority and medically underserved groups. Oncol Nurs Forum. 2002 Jun;29(5):789-801. doi: 10.1188/02.ONF.789-801. PMID: 12058154.

Aziz, N. M., & Rowland, J. H., (2003). Trends and Advances in Cancer Survivorship Research: Challenge and Opportunity. Seminars in Radiation Oncology, 13(3): 248-66.

Aziz, N. M., (2006). Long-term survivorship: Late effects. In: Chapter 71. Principles and practice of palliative care and supportive oncology. Berger AM, Portenoy RK, Weissman DE, editors. 3rd ed. Philadelphia, PA: Lippincott Williams & Wilkins; 2006. p 787804

Aziz, N. M., (2007). Cancer survivorship research: state of knowledge, challenges and opportunities. Acta Oncol. 2007;46(4):417–432. DOI: 10.1080/02841860701367878

Azzani, M., Roslani, A. C., & Su, T. T., (2015). The perceived cancer-related financial hardship among patients and their families: a systematic review. Support Care Cancer, 2015, 23(3): 889–898. 10.1007/s00520-014-2474-y

Babbott, S., Manwell, L. B., Brown, R., Montague, E., Williams, E., Schwartz, M., Hess,
E., Linzer, M., (2014). Electronic medical records and physician stress in primary care: results from the MEMO Study. J Am Med Inform Assoc. 2014 Feb;21(e1):e100-6. doi: 10.1136/amiajnl-2013-001875. Epub 2013 Sep 4. PMID: 24005796; PMCID: PMC3957395.

Back, A. L., Arnold, R. M., Baile, W. F., Tulsky, J. A., Fryer-Edwards, K., (2005). Approaching difficult communication tasks in oncology. CA Cancer J Clin. 2005 May-Jun;55(3):164-77. doi: 10.3322/canjclin.55.3.164. PMID: 15890639.

Bader, J. L., & Theofanos, M. F. (2003). Searching for cancer information on the internet: analyzing natural language search queries. Journal of medical Internet research, 5(4), e31. doi:10.2196/jmir.5.4.e31

Baider, L., & Goldzweig, G., (2016). A Brief Encounter with the Middle East: A Narrative of One Muslim Woman Diagnosed with Breast Cancer. Asia-Pacific Journal of Oncology Nursing, Vol.3(2): 205–210. doi: 10.4103/2347-5625.172485



Bailey, J. (2008). First steps in qualitative data analysis: Transcribing. Family Practice, 25(2), 127–131. doi:10.1093/fampra/cmn003

Baio, L. R., (2017). Cancer: Second Hand. Being the "Last one standing". Lulu Publishing Services. Chapter 6: The emotional stages of cancer.

Balarabe Kura, K. S. Y., (2012). Qualitative and Quantitative Approaches to the Study of Poverty: Taming the Tensions and Appreciating the Complementarities. The Qualitative Report, 17(20), 1-19. https://doi.org/10.46743/2160-3715/2012.1773

Balboni, M. J., Sullivan, A., Enzinger, A. C., Epstein-Peterson, Z. D., Tsend, Y. D., Mitchell, C., Niska, J., Zollfrank, A., VanderWeele, T. J., Balboni, T. A., (2014). Nurse and Physician Barriers to Spiritual Care Provision at the End of Life. Journal of Pain and Symptom Management 48: 400 –10

Balogh, E. P., Ganz, P. A., Murphy, S. B., Nass, S. J., Ferrell, B. R., & Stovall, E. (2011). Patient-centered cancer treatment planning: improving the quality of oncology care. Summary of an Institute of Medicine workshop. The oncologist, 16(12), 1800–1805. https://doi.org/10.1634/theoncologist.2011-0252

Baltussen, R., & Niessen, L., (2006). Priority setting of health interventions: the need for multi-criteria decision analysis. *Cost effectiveness and resource allocation: C/E*, *4*, 14. https://doi.org/10.1186/1478-7547-4-14

Banning, M., Hafeez, H., Faisal, S., Hassan, M., Zafar, A., (2009). The Impact of Culture and Sociological and Psychological Issues on Muslim Patients with Breast Cancer in Pakistan. Cancer Nursing, Vol.32(4), p.317-324. DOI: 10.1097/NCC.0b013e31819b240f.

Banegas, M. P., Schneider, J. L., Firemark, A. J., Dickerson, J. F., Kent, E. E., de Moor, J. S., Virgo, K. S., Guy, G. P., Jr, Ekwueme, D. U., Zheng, Z., Varga, A. M., Waiwaiole, L. A., Nutt, S. M., Narayan, A., & Yabroff, K. R. (2019). The social and economic toll of cancer survivorship: a complex web of financial sacrifice. Journal of cancer survivorship: research and practice, 13(3), 406–417. https://doi.org/10.1007/s11764-019-00761-1



Bener, A., Alwash, R., Miller, C. J., Denic, S., Dunn, E. V., (2001). Knowledge, attitudes, and practices related to breast cancer screening: a survey of Arabic women. J Cancer Educ. 2001, 16 (4): 215-220.

Bener, A., Honein, G., Carter, A. O., Da'ar, Z., Miller, C., & Dunn, E. V., (2002). The determinants of breast cancer screening behaviour: a focus group study of women in the United Arab Emirates. Oncology Nursing Forum, 29(9), E91-8.

Baker, F., Denniston, M., Smith, T., & West, M. (2005). Adult cancer survivors: How are they faring at about a year after diagnosis? Cancer, 104, 2565–2576. doi:10.1002/cncr.21488

Baker, S. E., & Edwards, R., (2012). How many qualitative interviews is enough? Expert voices and early career reflections on sampling and cases in qualitative research. National Centre for Research Methods Review Paper. 2012; http://eprints.ncrm.ac.uk/2273/4/how_many_interviews.pdf.

Bakht, S., & Najafi, S., (2010). Body image and sexual dysfunctions: comparison between breast cancer patients and healthy women. Procedia - Social and Behavioral Sciences. Vol.5:1493–7. doi: 10.1016/j.sbspro.2010.07.314.

Barbara Kura, S. Y., (2012). Qualitative and Quantitative Approaches to the Study of Poverty: Taming the Tensions and Appreciating the Complementarities. The Qualitative Report, 17(20), 1-19. Retrieved from https://nsuworks.nova.edu/tqr/vol17/iss20/4

Barbera, L., Zwaal, C., Elterman, D., McPherson, K., Wolfman, W., Katz, A., Matthew, A., & Interventions to Address Sexual Problems in People with Cancer Guideline Development Group (2017). Interventions to address sexual problems in people with cancer. Current oncology (Toronto, Ont.), 24(3), 192–200. https://doi.org/10.3747/co.24.3583

Bardia, A., Arieas, E. T., Zhang, Z., Defilippis, A., Tarpinian, K., Jeter, S., Nguyen, A., Henry, N. L., Flockhart, D. A., Hayes, D.F., Hayden, J., Storniolo, A. M., Armstrong, D. K., Davidson, N. E., Fetting, J., Ouyang, P., Wolff, A. C., Blumenthal, R. S., Ashen, M. D., Stearns, V., (2012). Comparison of breast cancer recurrence risk and cardiovascular disease incidence risk among postmenopausal women with breast cancer. Breast Cancer Res Treat. 2012 Feb; 131(3):907-14.



Bardwell, W. A., & Ancoli-Israel, S., (2008). Breast Cancer and Fatigue. Sleep medicine clinics, 3(1), 61–71. https://doi.org/10.1016/j.jsmc.2007.10.011

Barg, F. K., & Gullatte, M. M., (2001). Cancer support groups: Meeting the needs of African Americans with cancer. Seminars in Oncology Nursing, 17 (3): 171-178

Basch, E., Deal, A. M., Kris, M. G., Scher, H. I., Hudis, C. A., Sabbatini, P., Rogak, L., Bennett, A. V., Dueck, A. C., Atkinson, T. M., Chou, J. F., Dulko, D., Sit, L., Barz, A., Novotny, P., Fruscione, M., Sloan, J. A., Schrag, D., (2016). Symptom Monitoring with Patient-Reported Outcomes During Routine Cancer Treatment: A Randomized Controlled Trial. J Clin Oncol. 2016 Feb 20;34(6):557-65. doi: 10.1200/JCO.2015.63.0830. Epub 2015 Dec 7. Erratum in: J Clin Oncol. 2016 Jun 20;34(18):2198. Erratum in: J Clin Oncol. 2019 Feb 20;37(6):528. PMID: 26644527; PMCID: PMC4872028.

Basch, E., Deal, A. M., Dueck, A. C., Scher, H. I., Kris, M. G., Hudis, C., Schrag, D., (2017). Overall Survival Results of a Trial Assessing Patient-Reported Outcomes for Symptom Monitoring During Routine Cancer Treatment. JAMA. 2017 Jul 11;318(2):197-198. doi: 10.1001/jama.2017.7156. PMID: 28586821; PMCID: PMC5817466.

Bashir, M., Ali, N. A., Shaban, S.; Hassouneh, G.; Ashari, M. A., Jaloudi, M., (2012). Young Women in the UAE Have Higher Incidence of HER2 Positive Breast Cancer. Breast Journal, November 2012, Vol.18(6), pp.637-638

Bates, A. T., (2016). Addressing Existential Suffering. British Columbia Medical Journal 58: 268–73

Baty, B. J., Kinney, A. Y., & Ellis, S. M., (2003). Developing culturally sensitive cancer genetics communication aids for African Americans. American Journal of Medical Genetics Part A, 15 April 2003, Vol.118(2), pp.146-155. DOI: 10.1002/ajmg.a.10150

Baxter, J., & Eyles, J. (1997). Evaluating qualitative research in social geography: establishing 'rigour' in interview analysis. Transactions of the Institute of British Geographers, 22(4),505-525.



Bayliss, E. A., Steiner, J. F., Ferbald, S. H, Crane, L. A & Main, S; (2003). Description of barriers to self-care by persons with comorbid diseases. Annals of Family Medicine 1(1): 15-21.

Bazeley, P., (2009). Analysing mixed methods data. https://doi.org/10.1002/9781444316490.ch6

Beckjord, E. B., Arora, N. K., McLaughlin, W., Oakley-Girvan, I., Hamilton, A. S., Hesse,
B. W., (2008). Health-related information needs in a large and diverse sample of adult cancer survivors: implications for cancer care. J. Cancer Surviv, 2 (2008), pp. 179-189

Behboudifar, A., Heshmati Nabavi, F., Anvari, K., & Shakeri, M. T. (2018). Effect of pretreatment education on anxiety in patients undergoing radiation therapy for the first time: A randomized clinical trial. Cogent Psychology, 5(1), 1483612. https://doi.org/10.1080/23311908.2018.1483612

Bensing, J., (2000). Bridging the gap. The separate worlds of evidence-based medicine and patient-centered medicine. Patient Educ Couns 2000; 39:17–25.

Bell, K., & Ristovski-Slijepcevic, S., (2013). Cancer Survivorship: Why Labels Matter.JournalofClinicalOncology201331:4,409-411.https://ascopubs.org/doi/10.1200/JCO.2012.43.5891

Bell, M. L., Butow, P. N., & Goldstein, D., (2013). Informatively missing quality of life and unmet needs sex data for immigrant and Anglo-Australian cancer patients and survivors. Quality of life research: an international journal of quality-of-life aspects of treatment, care and rehabilitation, Vol.22(10):2757-60. doi: 10.1007/s11136-013-0392-y.

Bellamy, R., & Med, Teach., (2004). An introduction to patient education: theory and practice. Jun;26(4):359-65. DOI: 10.1080/01421590410001679398

Bender, J. L., Hohenadel, J., Wong, J., Katz, J., Ferris, L. E., Shobbrook, C., Warr, D., Jadad, A. R., (2008). What patients with cancer want to know about pain: a qualitative study. J Pain Symptom Manage. 2008 Feb;35(2):177-87. doi: 10.1016/j.jpainsymman.2007.03.011. Epub 2007 Dec 26. PMID: 18158232.



Bender, J. L., Jimenez-Marroquin, M. C., Ferris, L. E., Katz, J., Jadad, A. R., (2013). Online communities for breast cancer survivors: a review and analysis of their characteristics and levels of use. Support Care Cancer 21(5), 1253–1263 (2013)

Bendardaf, R., Saheb Sharif-Askari, F., Saheb Sharif-Askari, N., Yousuf Guraya, S., A AlMadhi, S., & Abusnana, S. (2020). Incidence and Clinicopathological Features of Breast Cancer in the Northern Emirates: Experience from Sharjah Breast Care Center. International journal of women's health, 12, 893–899. https://doi.org/10.2147/IJWH.S266335

Ben Letaifa, S., (2006). Compatibilité et incompatibilité des paradigmes et méthodes [online] available from: http://www.strategie-aims.com/actesateliers/Quali06/BenLetaifa.pdf [14 April 2019]

Benowiltz, S., (1999). To tell the truth: a cancer diagnosis in other cultures is often a family affair. Journal of the National Cancer Institute. Vol.91(22):1918. Available at:

Berger, O., Grønberg, B. H., Loge, J. H., Kaasa S., & Sand, K., (2018). Cancer patients' knowledge about their disease and treatment before, during and after treatment: a prospective, longitudinal study. BMC Cancer 18, 381 (2018). https://doi.org/10.1186/s12885-018-4164-5

Berg, B. (2009). Qualitative research methods for the social science (7th ed.). Boston: Allyn & Bacon.

Bernard, H. R. (2002). Research methods in anthropology: Qualitative and quantitative approaches (3rd ed.). Walnut Creek, CA:Alta Mira Press.

Bernstein, K. I., Promislow, S., & Carr, R., Rawsthrone, P., Walker, J. R., Bernstein, C. N., (2011). Information needs and preferences of recently diagnosed patients with inflammatory bowel disease Inflamm Bowel Dis 17 590–598 https://doi.org/10.1002/ibd.21363

Bertero, C. M., (2000). Types and Source of Social Support for People Afflicted with Cancer. Nursing and Health Science. 2000; 2:93–101. http://dx.doi.org/10.1046/j.1442-2018.2000.00044.x.



Berman, A., (2012). A Living life in my own way—and dying that way as well. Health Affairs (Millwood). Vol.31(4):871–874.

Bernardi, L., Kleim, S., von der Lippe, H., (2007). Social influences on fertility: a comparative mixed methods study in Eastern and Western Germany Journal of Mixed Methods Research, 1 (1) (2007), pp. 23-47

Betancourt, H., & Lopez, S. R., (1993). The Study of Culture, Ethnicity, and Race in American Psychology. American Psychologist Copyright 1993 by the American Psychological Association, Inc. 0003-066X/93/S2.00 Vol. 48, No. 6, 629-637. Available at: https://dornsifecms.usc.edu/assets/sites/376/docs/1993_Betancourt_Lopez.pdf

Bhatia, S., Armenian, S. H., & Landier, W. (2017). How I monitor long term and late effects after blood or marrow transplantation. Blood, 130, 1302–1314. https://doi.org/10.1182/blood-2017-03-725671

Bhoo-Pathy, N., Hartman, M., Yip, C-H., Saxena, N., Taib, N. A., Lim, S-E., Iau, P., Adami, H-O., Bulgiba, A. M., Lee, S-C., Verkooijen, H. M., (2012). Ethnic Differences in Survival after Breast Cancer in South East Asia. https://doi.org/10.1371/journal.pone.0030995

Bickel, K. E., McNiff, K., Buss, M. K., Kamal, A., Lupu, D., Abernethy, A. P., Border, M. S., Shapiro, C. L., Acheson, A. K., Malin, J., Evans, T., Krzyzanowska, M. K., (2016). Defining High-Quality Palliative Care in Oncology Practice: An American Society of Clinical Oncology/American Academy of Hospice and Palliative Medicine Guidance Statement. J Oncol Pract. 2016;12(9):e828–38. 10.1200/JOP.2016.010686

Bilodeau, K., Tremblay, D., & Durand, M. J. (2019). Gaps and delays in survivorship care in the return-to-work pathway for survivors of breast cancer-a qualitative study. Current oncology (Toronto, Ont.), 26(3), e414–e417. https://doi.org/10.3747/co.26.4787

Bilodeau, B. A., & Degner, L. F., (1996). Information needs, sources of information, and decisional roles in women with breast cancer. Oncology Nursing Forum, 23, 691-696.



Biggerstaff, D. L. & Thompson, A. R. (2008). Interpretative Phenomenological Analysis (IPA): A Qualitative Methodology of Choice in Healthcare Research. Qualitative Research in Psychology 5: 173 – 183 DOI: 10.1080/14780880802314304

Binkley, J. M., Harris, S. R., Levangie, P. K., Pearl, M., Guglielmino, J., Kraus, V. and Rowden, D. (2012). Patient perspectives on breast cancer treatment side effects and the prospective surveillance model for physical rehabilitation for women with breast cancer. Cancer, 118: 2207–2216. doi:10.1002/cncr.27469. Available at: http://onlinelibrary.wiley.com/doi/10.1002/cncr.27469/full

Birt, L., Scott, S., Cavers, D., Campbell, C., Walter, F., (2016). Member Checking: A Tool to Enhance Trustworthiness or Merely a Nod to Validation? Qual Health Res. 2016 Nov;26(13):1802-1811. doi: 10.1177/1049732316654870. Epub 2016 Jul 10. PMID: 27340178.

Bishop, F. L., (2015) Using mixed methods in health research: Benefits and challenges. British Journal of Health Psychology20(1): 1–4**Blaikie, N. (1993)**. Approaches to Social Enquiry. Cambridge: Polity Press

Blaikie, N., (1993). Approaches to Social Enquiry. Cambridge: Polity Press

Bloom, J. R., & Kessler, L., (1994). Risk and timing of counseling and support interventions for younger women with breast cancer. Journal of the National Cancer Institute. Monographs, 16:199–206.

Bloom, J. R., Stewart, S. L., Chang, S., Banks, P. J. (2004). Then and now: quality of life of young breast cancer survivors Psycho-oncology 13 pp.147-160

Bobadilla, J. L., (1996). Priority setting and cost effectiveness. In K. Janovsky (Ed.), Health policy and systems development: An agenda for research (pp. 43-60). Geneva, Switzerland: World Health Organization

Bober, S. L., Reese, J. B., Barbera, L., Bradford, A., Carpenter, K. M., Goldfarb, S., & Carter, J. (2016). How to ask and what to do: a guide for clinical inquiry and intervention



regarding female sexual health after cancer. Current opinion in supportive and palliative care, 10(1), 44–54. https://doi.org/10.1097/SPC.000000000000186

Bodai, B. I., & Tuso, P. (2015). Breast Cancer Survivorship: A Comprehensive Review of Long-Term Medical Issues and Lifestyle Recommendations. The Permanente Journal, 19(2), 48–79. http://doi.org/10.7812/TPP/14-241

Bogdan, R. C. & Biklen, S. K. (1998). Qualitative research in education: An introduction to theory and methods (3rd ed.). Needham Heights, MA: Allyn & Bacon

Boing, L., Pereira, G. S., Araújo, C., Sperandio, F. F., Loch, M., Bergmann, A., Borgatto, A. F., Guimarães, A.; (2019). Factors associated with depression symptoms in women after breast cancer. Revista de saude publica, 53, 30. doi:10.11606/S1518-8787.2019053000786

Boland, E. G., Ahmedzai, S. H., (2017). Persistent pain in cancer survivors. Curr Opin Support Palliat Care. 2017 Sep;11(3):181-190. doi: 10.1097/SPC.00000000000292. PMID: 28700361.

Bolton, M. A., Lobben, I. and Stern, T. A., (2010). "The Impact of Body Image on Patient Care." Primary Care Companion to the Journal of Clinical Psychiatry 12.2 (2010): Primary Care Companion to the Journal of Clinical Psychiatry, 2010, Vol.12 (2) https://doi.org/10.4088/PCC.10r00947blu

Boughey, J. C., Attai, D. J., Chen, S. L., Cody, H. S., Dietz, J. R., Feldman, S. M., Greenberg, C. C., Kass, R. B., Landercasper, J., Lemaine, V., MacNeill, F., Margenthaler, J. A., Song, D. H., Staley, A. C., Wilke. L. G., Willey, S. C., Yao, K. A., (2016). Contralateral Prophylactic Mastectomy Consensus Statement from the American Society of Breast Surgeons: Additional Considerations and a Framework for Shared Decision Making. Ann Surg Oncol. 2016 Oct;23(10):3106-11. doi: 10.1245/s10434-016-5408-8. Epub 2016 Jul 28. PMID: 27469118; PMCID: PMC4999472.

Bousquet, G., Orri, M., Winterman, S., Brugière, C., Verneuil, L., & Revah-Levy, A. (2015). Breaking bad news in oncology: A metasynthesis. Journal of Clinical Oncology, 33, 2437–2443. doi:10.1200/JCO.2014.59.6759



Bonevski, B., Sanson-Fisher, R., Girgis, A., Burton, L., Cook, P., Boyes, A., (2000). Evaluation of an instrument to assess the needs of patients with cancer. Supportive Care Review Group. Cancer. 2000; 88(1): 217–25. pmid:10618626. https://acsjournals.onlinelibrary.wiley.com/doi/epdf/10.1002/%28SICI%291097-0142%2820000101%2988%3A1%3C217%3A%3AAID-CNCR29%3E3.0.CO%3B2-Y

Bou Khalil, R., (2013). Attitudes, beliefs and perceptions regarding truth disclosure of cancerrelated information in the Middle East: A review. Palliat Support Care. 2013, 11 (1): 69-78. 10.1017/S1478951512000107

Bowen, G. A., (2008). Naturalistic inquiry and the saturation concept: a research note. Qual. Res. 2008; 8(1):137–152. doi: 10.1177/1468794107085301.

Bower, J. E., (2008). Behavioral symptoms in patients with breast cancer and survivors. Journal of clinical oncology: official journal of the American Society of Clinical Oncology, 26(5), 768–777. https://doi.org/10.1200/JCO.2007.14.3248

Bower, J. E., (2014). Cancer-related fatigue--mechanisms, risk factors, and treatments. Nature reviews. Clinical oncology, 11(10), 597–609. doi:10.1038/nrclinonc.2014.127

Bowie, J., Sydnor, K. D., Granot, M., (2003). Spirituality and care of prostate cancer patients: a pilot study, Journal of the National Medical Association, vol. 95(10), pg. 951-954

Boyatzis, R. E. (1998). Transforming qualitative information: Thematic analysis and code development. Thousand Oaks, CA: Sage

Boyes, A., Girgis, A., Lecathelinais, C., (2009). Brief assessment of adult cancer patients' perceived needs: development and validation of the 34-item Supportive Care Needs Survey (SCNS-SF34). J Eval Clin Prac.2009; 15(4): 602–6. pmid:19522727

Boyes, A. W., Girgis, A., D'Este, C., Zucca, A. C., (2012). Prevalence and correlates of cancer survivors' supportive care needs 6 months after diagnosis: a population-based cross-sectional study. BMC Cancer. 2012; 12:150.



Braun, V., & Clarke, V., (2006). Using thematic analysis in psychology. Qualitative research in psychology, 3(2), 77-101

Braun, V., Clarke, V., & Terry, G. (2015). Thematic analysis. In P. Rohleder & A. C. Lyons (Eds.), Qualitative research in clinical and health psychology (pp. 95–113). Houndmills, Basingstoke, Hamphshire, UK: Palgrave Macmillan.

Bray, F., Ferlay J., Soerjomataram, I., Siegel, R. L., Torre, L. A., Jemal, A., (2018). Global cancer statistics (2018): GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. https://doi.org/10.3322/caac.21492

Brach, C., & Fraser, I. (2000). Can cultural competency reduce racial and ethnic health disparities? A review and conceptual model. Medical care research and review: MCRR, 57 Suppl 1(Suppl 1), 181–217. https://doi.org/10.1177/1077558700057001S09

Bradley, C. J., Bednarek, H. L., Neumark, D., (2002). Breast cancer and women's labor supply. Health Services Research. Vol.37(5):1309–1328.

Bramley, L., & Matiti, M., (2014). How does it really feel to be in my shoes? Patients' experiences of compassion within nursing care and their perceptions of developing compassionate nurses. Journal of Clinical Nursing, 23(19-20), 2790–2799. http://doi.org/10.1111/jocn.12537

Brawley, O. W., (2002). Disaggregating the effects or race and poverty on breast cancer outcomes. Journal of the National Cancer Institute. Vol.94(7): 471–473.

Brédart, A., Bouleuc, C., & Dolbeault, S., (2005). Doctor-patient communication and satisfaction with care in oncology. Curr Opin Oncol. 2005 Jul;17(4):351-4. doi: 10.1097/01.cco.0000167734.26454.30. PMID: 15933466.

Brédart, A., Kop, J. L., Griesser, A. C., Zaman, K., Panes-Ruedin, B., Jeanneret. W., Delaloye, J. F., Zimmers, S., Jacob, A., Berthet, V., Fiszer, C., Dolbeault, S., (2012). Validation of the 34-item supportive care needs survey and 8-item breast module French versions (SCNS-SF34-Fr and SCNS-BR8-Fr) in breast cancer patients. Eur J Cancer Care. 2012;21(4):450–459. doi: 10.1111/j.1365-2354.2012.01356.x



Brédart, A., Kop, J-L., Griesser, A-C., Zaman, F. C., Panes-Ruedin, B., Jeanneret, W., Delaloye, J-F., Zimmers, S., Berthet, V & Dolbeault, S., (2013). Assessment of needs, health-related quality of life, and satisfaction with care in breast cancer patients to better target supportive care. Annals of Oncology24: 2151–2158, 2013doi:10.1093/annonc/mdt128 Published online 7 April 2013

Bretsher M, Rummons R, Sloan J, Kaur J, Bartlett A, Borkenhagen L, Loprinzi C. (1999). Quality of life in hospice patients. Psychosomatics. 1999; 40:309–313

Breitbart, W. S, & Alici, Y., (2009). Psycho-oncology. Harv Rev Psychiatry. 2009; 17:361-376.

British Psychological Society Code of Human Research Ethics (2021). Available at: https://www.bps.org.uk/sites/bps.org.uk/files/Policy%20-%20Files/BPS%20Code%20of%20Human%20Research%20Ethics.pdf Accessed on 19th Aug 2018

Bromfield, N., (2014). Interviews with divorced women from the United Arab Emirates: a rare glimpse into lived experiences. Families, Relationships and Societies. vol 3. 339–54. Policy Press 2014. http://dx.doi.org/10.1332/204674313X13842674567773

Brown, M. R., Ramirez, J. D., & Farquhar-Smith, P., (2014). Pain in cancer survivors. Br J Pain. 2014 Nov;8(4):139-53. doi: 10.1177/2049463714542605. PMID: 26516548; PMCID: PMC4616728

Bryman, A., (2001). Social Research Method. Oxford: Oxford University Press.

Bryman, A., (2004). Social Research Methods. University Press, Oxford.

Bryman, A., (2006). Integrating quantitative and qualitative research: how is it done? Qualitative Research, 6 (1) (2006), pp. 97-113

Bryman, A., (2007). Barriers to integrating quantitative and qualitative research. Journal of Mixed Methods Research 1 (1), 8–22.



Buchan, H., (2004). Gaps between best evidence and practice: causes for concern. Med J Aust. 2004;180(15):S47–S48.

Bucholtz, M., (2000). The politics of transcription. Journal of Pragmatics, 32(10), 1439–1465. doi:10.1016/S0378-2166(99)00094-6

Buckman, R. A., (2005). Breaking bad news: The S-P-I-K-E-S strategy. Community Oncology, 2, 138–142. doi:10.1016/S1548-5315(11)70867-1

Bumb, M., Keefe, J., Miller, L., Overcash, J., (2017). Breaking Bad News: An Evidence-Based Review of Communication Models for Oncology Nurses. Journal Club Article. CNE Article. CJON 2017, 21(5), 573-580 DOI: 10.1188/17.CJON.573-580

Burg, M. A., Adorno, G., Lopez, E. D., Loerzel, V., Stein, K., Wallace, C., Sharma, D. K., (2015). Current unmet needs of cancer survivors: analysis of open-ended responses to the American Cancer Society study of Cancer survivors II. Cancer. 2015;121(4):623–30.

Bugos, K. G., (2015). Issues in adult blood cancer survivorship care. Seminars in Oncology Nursing, 31, 60–66. https://doi.org/10.1016/j.soncn.2014.11.007

Burr, V., (1995). An introduction to social constructionism. Taylor & Frances/Routledge. https://doi.org/10.4324/9780203299968

Burrell, G., & Morgan, G., (1979). Sociological paradigms and organizational analysis. Farnham, UK: Ashgate Publishing

Bush, N. J., & Gorman, L., (2018). Psychosocial Nursing Care Along the Cancer Continuum(Third Edition). CHAPTER 1 Psychosocial Impact of Cancer on the Individual, Family, andSociety.PublishedBy:OncologyNursingSociety.Society.https://www.ons.org/sites/default/files/2018-10/Psychosocial%20Nursing%20Care%20Along%20the%20Cancer%20Continuum%20Sample%20Chapter%20%281%29.pdf

Butow, P. N., Maclean, M., Dunn, S. M., Tattersall, M. H., Boyer, M. J., (1997). The dynamics of change: cancer patients' preferences for information, involvement and



support. Ann Oncol. 1997. pp. 857–63. Available from http://www.ncbi.nlm.nih.gov/entrez/query.fcgi?cmd=Retrieve&db=PubMed&dopt=Cita tion&list_uids=9358935.

Butow, P., Brindle, E., McConnell, D., Boakes, R., Tattersall, M., (1998). Information booklets about cancer: factors influencing patient satisfaction and utilization. Patient Educ Couns. 1998; 33:129-141.

Butow, B. **M., Goldstein, D., Sze, M., Aldridge, L., Abdo, S., Eisenbruch, M., (2011).** Grappling with cultural differences: Communication between oncologists and immigrant cancer patients with and without interpreters. Patient Education and Counseling. 2011; 84: 398-405

Byar, K. L., Berger, A. M., Bakken, S. L., & Cetak, M. A., (2006). Impact of adjuvant breast cancer chemotherapy on fatigue, other symptoms, and quality of life. Oncol Nurs Forum, 33 (2006), pp. E18-E26

Byrne, G., (2007). A Statistical Primer: Understanding Descriptive and Inferential Statistics. Evidence Based Library and Information Practice. Evidence Based Library and Information Practice, Vol 2, Iss 1, Pp 32-47 (2007).

Cadet, T., Davis, C., Elks, J., & Wilson, P. (2016). A Holistic Model of Care to Support Those Living with and beyond Cancer. Healthcare (Basel, Switzerland), 4(4), 88. https://doi.org/10.3390/healthcare4040088

Cancer health disparities, (2008). Washington, DC: The National Cancer Institute, U.S.NationalInstitutesofHealth;2008.Mar11,(http://www.cancer.gov/cancertopics/factsheet/cancer-health-disparities/disparities). Availableat: http://ejournals.library.ualberta.ca/index.php/EBLIP/article/view/168/239

Cagle, J. G., & Loscalzo, M., (2015). Using cognitive and behavioral approaches throughout the cancer experience. In G. Christ, C. Messner & L. Behar (Eds.), Handbook of oncology social work: Psychosocial care for people with cancer (pp. 345–350). New York, NY: Oxford University Press



Caldeira, S., Timmins, F., de Carvalho, E. C., Vieira, M., (2017). Clinical Validation of the Nursing Diagnosis Spiritual Distress in Cancer Patients Undergoing Chemotherapy. Int J Nurs Knowl. 2017 Jan;28(1):44-52. doi: 10.1111/2047-3095.12105. Epub 2015 Jun 11. PMID: 26095541.

Callahan, C. & Brintzenhofeszoc, K,. (2015). Financial Quality of Life for Patients with Cancer: An Exploratory Study. J Psychosoc Oncol. 2015;33(4):377-94. doi: 10.1080/07347332.2015.1045679. https://www.ncbi.nlm.nih.gov/pubmed/25996976

Campbell, S. H., Sanson-Fisher, R., Taylor-Brown, J., Hayward, L., Wang, S. X., & Turner, D., (2009). The cancer Support Person's Unmet Needs Survey. Cancer, 115(14), 3351–3359. doi: 10.1002/cncr.24386.

Campbell, H. S., Sanson-Fisher, R., Turner, D., Hayward, L., Wang, X. S., Taylor-Brown, J., (2010). Psychometric properties of cancer survivors' unmet needs survey. Support Care Cancer. 2010; 19(2): 221–30. pmid:20099001

Canada, A. I., Fitchett, G., Murphy, P. E., Stein, K., Portier, K., Crammer, C., Peterman, A. H., (2013). Racial/ethnic differences in spiritual well-being among cancer survivors. J. Behav Med. 2013; 36:441-453

Canadian Partnership Against Cancer (cpac), (2018). Living with Cancer: A Report on the Patient Experience. Toronto, ON: cpac; 2018. p. 52

Canil, T., Cashell, A., Papadakos, J., Abdelmutti, N., & Friedman, A. J. (2012). Evaluation of the effects of pre-treatment education on self-efficacy and anxiety in patients receiving radiation therapy: A pilot study. Journal of Medical Imaging and Radiation Sciences, 43(4), 221–227. doi: 10.1016/j.jmir.2012.05.002

Cappiello, M., Cunningham R. S., Knobf M. T., (2007). Breast cancer survivors: Information and support after treatment. Clinical Nursing Research. 2007; 16:278–293. discussion 294–301.



Caraceni, A., & Shkodra, M., (2019). Cancer Pain Assessment and Classification. Cancers (Basel). 2019 Apr 10;11(4):510. doi: 10.3390/cancers11040510. PMID: 30974857; PMCID: PMC6521068.

Cardoso, F., Bese, N., Distelhorst, S. R., Bevilacqua, J. L., Ginsburg, O., Grunberg, S. M., Gralla, R. J., Steyn, A., Pagani, O., Partridge, A. H., Knaul, F. M., Aapro, M. S., Andersen, B. L., Thompson, B., Gralow, J. R., Anderson, B. O., (2013) Supportive care during treatment for breast cancer: resource allocations in low- and middle-income countries. A Breast Health Global Initiative 2013 consensus statement. Breast 2013; 22:593-605.

Cardoso, F., Spence, D., Mertz, S., Corneliussen-James, D., Sabelko, K., Gralow, J., Cardoso, M. J., Peccatori, F., Paonessa, D., Benares, A., Sakurai, N., Beishon, M., Barke, r S. J., Mayer, M., (2018). Global analysis of advanced/metastatic breast cancer: Decade report (2005–2015). Breast. 2018 Jun; 39:131-138. doi: 10.1016/j.breast.2018.03.002. Epub 2018 Apr 19

Carey, M., Lambert, S., Smits, R., Paul, C., Sanson-Fisher, R., Clinton-McHarg, T., (2012). The unfulfilled promise: a systematic review of interventions to reduce the unmet supportive care needs of cancer patients. Support Care Cancer. 2012 Feb;20(2):207-19. doi: 10.1007/s00520-011-1327-1. Epub 2011 Nov 17. PMID: 22089430; PMCID: PMC3244607.

Carreira, H., Rachael Williams, Martin Müller, Rhea Harewood, Susannah Stanway, Krishnan Bhaskaran., (2018). Associations Between Breast Cancer Survivorship and Adverse Mental Health Outcomes: A Systematic Review, JNCI: Journal of the National Cancer Institute, Volume 110, Issue 12, December 2018, Pages 1311–1327, https://doi.org/10.1093/jnci/djy177

Carlson, L.E., Bultz, B. D., (2004). Efficacy and medical cost offset of psychosocial interventions in cancer care: making the case for economic analyses. Psycho-Oncology. 2004;13(12):837–49.

Carlson, L. E., Waller, A., Mitchell, A. J., (2012). Screening for distress and unmet needs in patients with cancer: review and recommendations. Journal of Clinical Oncology. 2012: JCO. 2011.39. 5509



Carr, D., & Bodnar-Deren, S., (2009). International Handbook of Population Aging. Chapter 32: Gender, aging and widowhood. P. Uhlenberg (ed.), DOI 10.1007/978-1-4020-8356-3_32,
© Springer Science+Business Media B.V. 2009. Available at: https://sites.bu.edu/deborahcarr/files/2018/01/carrbodnar-deren_2009.pdf [Accessed on 29th July 2019].

Carr, D., Goudas, L., Lawrence, D., Pirl, W., Lau, J., DeVine, D., Kupelnick, B., Miller, K., (2002). Management of cancer symptoms: Pain, depression, and fatigue Management of cancer symptoms: Pain, depression, and summaries. In: AHRQ Evidence Report Summaries. Rockville (MD): Agency for Healthcare Research and Quality (US); 1998-2005. 61. Available at: https://www.ncbi.nlm.nih.gov/books/NBK11961/

Carroll-Johnson, R. S., Gorman, L. M., & Bush, N. B., (2006). Psychosocial nursing care along the cancer continuum. CHAPTER 1. The Psychosocial Impact of Cancer on the Individual, Family, and Society

Carver, C. S., & Scheier, M. F., (2008). Perspectives on Personality. Pearson International Edition, Boston, New York, San Francisco.

Castillo, A., Mendiola, J., & Tiemensma, J., (2019). Emotions and Coping Strategies During Breast Cancer in Latina Women: A Focus Group Study. Hispanic Health Care International. 2019;17(3):96-102. doi:10.1177/1540415319837680

C-Change., (2008). Increasing access to cancer care: An action guide for comprehensive cancer control coalitions. 2008. Available at: https://www.ceoroundtableoncancer.org/sites/default/files/2008_increasing_access_to_cancer _care_an_action_guide_for_comprehensive_cancer_control_coalitions.pdf [Accessed: 24th Sp 2019]

Cavanagh, S., & Chadwick, K., (2005). Health Needs Assessment. Health Development Agency, London.

Chadder, J., Zomer, S., Lockwood, S., Shaw Moxam, R., Louzado, C., Coronado, A., Rahal, R., Green, E., (2018). Understanding the Experiences of Cancer Patients as They Transition from Treatment to Primary and Community Care: A Pan-Canadian Study of Over



13,000 Cancer Survivors. Canadian Partnership Against Cancer, Toronto, Canada. DOI: 10.1200/jgo.18.24900 Journal of Global Oncology 4, no. Supplement

Chaturvedi, S. K., & Chandra, P. S., (1998) Palliative care in India. Support Care Cancer. 1998 Mar; 6(2):81-4.

Chaturvedi, S. K., Strohschein, F. J., Saraf, G., & Loiselle, C. G. (2014). Communication in cancer care: psycho-social, interactional, and cultural issues. A general overview and the example of India. Frontiers in psychology, 5, 1332. https://doi.org/10.3389/fpsyg.2014.01332

Chaves, E. C. L., Campos de Carvalho, E., & Hass. V. J., (2010). Validação do Diagnóstico de Enfermagem Angústia Espiritual: Análise por Especialistas. Acta Paulista de Enfermagem 23: 264–70.

Chee, W., Lee, Y., Im, E. O., Chee, E., Tsai, H. M., Nishigaki, M., Yeo, S. A., Schapira, M. M., & Mao, J. J. (2017). A culturally tailored Internet cancer support group for Asian American breast cancer survivors: A randomized controlled pilot intervention study. Journal of telemedicine and telecare, 23(6), 618–626. https://doi.org/10.1177/1357633X16658369

Cheng, H., (2012). Understanding breast cancer survivorship experience among mainland Chinese women: a mixed methods study. Chinese University of Hong Kong Graduate School. Division of Nursing. UMI Number: 3572621

Chen, C. H., Tang, S. T., Chen, C. H., (2011). Meta-analysis of cultural differences in Western and Asian patient-perceived barriers to managing cancer pain. Palliat Med. 2012 Apr;26(3):206-21. doi: 10.1177/0269216311402711. Epub 2011 Apr 7. PMID: 21474622.

Chen, J., Mullins, C. D., Novak, P., & Thomas, S. B. (2016). Personalized Strategies to Activate and Empower Patients in Health Care and Reduce Health Disparities. Health Education & Behavior: The Official Publication of the Society for Public Health Education, 43(1), 25–34. http://doi.org/10.1177/1090198115579415

Chen, X., Lu, W., Gu, K., Chen, Z., Zheng, Y., Zheng, W., & Shu, X. O. (2011). Weight change and its correlates among breast cancer survivors. Nutrition and cancer, 63(4), 538–548.



Cherryholmes, C. H., (1992). Notes on pragmatism and scientific realism. Educational Researcher, 13–17

Cheville, A. L., Mustian, K., Winters-Stone., K., Zucker, D. S., Gamble, G. L., Alfano, C. M., (2017). Cancer Rehabilitation: An Overview of Current Need, Delivery Models, and Levels of Care. Phys Med Rehabil Clin N Am. 2017 Feb;28(1):1-17. doi: 10.1016/j.pmr.2016.08.001.PMID: 27912990.

Chiesi, F., Bonacchi, A., Primi, C., & Miccinesi, G. (2017). Assessing unmet needs in patients with cancer: An investigation of differential item functioning of the Needs Evaluation Questionnaire across gender, age and phase of the disease. PloS one, 12(7), e0179765. doi: 10.1371/journal.pone.0179765

Chiriac, V. F., Baban, A., & Dumitrascu, D. L. (2018). Psychological stress and breast cancer incidence: a systematic review. Clujul medical (1957), 91(1), 18–26. doi:10.15386/cjmed-924

Choudhury, C. R., Bandopadhyay, A., Mandal, P., (2018). Title: Psychosocial implications of Hair loss following Chemotherapy Intervening with completion of Adjuvant Therapy for Breast Cancer. Journal of medical science and clinical research. Vol 06 issue 10 Oct 2018. DOI: https://dx.doi.org/10.18535/jmscr/v6i10.143

Chouchane, L., Boussen, H., & Sastry, K. (2013). Breast cancer in Arab populations: Molecular characteristics and disease management implications. The Lancet Oncology, 14(10), e417-e424

Chua, G. P., Tan, H. K., & Gandhi, M. (2018). What information do cancer patients want and how well are their needs being met? Ecancer medical science, 12, 873. doi:10.3332/ecancer.2018.873

Chua, G. k., Ng, Q. S., Tan, H. K., Ong, W. S., (2019). Concerns and Quality of Life of Cancer Survivors Across the Survivorship Trajectory, 26 November 2019, PREPRINT (Version 2) available at Research Square [https://doi.org/10.21203/rs.2.14780/v2]



Clarke, M., (1984). The constructs' stress 'and coping as a rationale for nursing activities. Journal of Advanced Nursing, 267-27s

Clarke, P. N., & Yaros, P. S., (1988). Research blenders: commentary and response. Transitions to new methodologies in nursing sciences Nursing Science Quarterly, 1 (4) (1988), pp. 147-151

Clarke, V., (2005). 'We're all very liberal in our views': Students' talk about lesbian and gay parenting. Lesbian & Gay Psychology Review, 6(1), 2-15.

Classen, D. C., Mermel, L. A., (2015). Specialty society clinical practice guidelines time for evolution or revolution? JAMA 314:871–87

Cline, R. J. W, & Haynes, K. M., (2001). Consumer health information seeking on the Internet: the state of the art. Health Education Research 16(6) 671–92.

Clinton-McHarg, T., Carey, M., Sanson-Fisher, R., Shakeshaft, A., & Rainbird, K. (2010). Measuring the psychosocial health of adolescent and young adult (AYA) cancer survivors: a critical review. Health and quality of life outcomes, 8, 25. https://doi.org/10.1186/1477-7525-8-25

Clough-Gorr, K. M., Ganz, P. A., (2010). Silliman RA. Older breast cancer survivors: factors associated with self-reported symptoms of persistent lymphedema over 7 years of follow-up. Breast J. 2010; 16:147–155.

Cobo-Cuenca, AI., Martín-Espinosa, N. M., Sampietro-Crespo, A., Rodríguez-Borrego, M. A., Carmona-Torres, J. M., (2018). Sexual dysfunction in Spanish women with breast cancer. PLoS One. 2018 Aug 31;13(8): e0203151. doi: 10.1371/journal.pone.0203151. PMID: 30169506; PMCID: PMC6118366.

Cochrane, L. J., Olson, C. A., Murray, S, Dupuis, M., Tooman, T., Hayes, S., (2007). Gaps between knowing and doing: understanding and assessing the barriers to optimal health care. J Contin Educ Health Prof. 2007;27(2):94–102. doi: 10.1002/chp.106.


Coggin, C., & Shaw-Perry, M., (2006). Breast Cancer Survivorship, Journal of Psychosocial Oncology, 24:4, 107-122, DOI: 10.1300/J077v24n04_06

Cohen & Wills, (1985). Stress, Social Support, and the Buffering Hypothesis. Psychological Bulletin 98(2):310-57. DOI: 10.1037/0033-2909.98.2.310.

Cohen, J., (1988). Statistical Power Analysis for the Behavioral Sciences. New York, NY: Routledge Academic

Cohen, J., (1996). Preferences, needs and QALYs. Journal of Medical Ethics, 22, 267-272

Cohen, L., Hack, T. F., de Moor, C., Katz, J., Goss, P. E., (2000). The effects of type of surgery and time on psychological adjustment in women after breast cancer treatment. Ann Surg Oncol. 2000;7(6):427–34.

Cohen, S., (2004). Social relationships and health. American Psychologist 59(8):676–684.

Cohen, L., Manion. L., & Morrison, K. (2011). Research methods in education. London: Routledge.

Cohen, M., Anderson, R. C., Jensik, K., Xiang, Q., Pruszynski, J., Walker, A. P., (2012). Communication between breast cancer patients and their physicians about breast-related body image issues. Plast Surg Nurs. 2012;32(3):101–105. doi: 10.1097/PSN.0b013e3182650994

Collins, K. M. T., Onwuegbuzie, A. J., Sutton, I. L., (2006). A Model Incorporating the Rationale and Purpose for Conducting Mixed Methods Research in Special Education and Beyond. Learning Disabilities: A Contemporary Journal. 2006; 4:67–100

Collins, K. K., Liu, Y., Schootman, M., Aft, R., Yan, Y., Dean, G., Eilers, M., & Jeffe, D. B., (2011). Effects of breast cancer surgery and surgical side effects on body image over time. Breast cancer research and treatment, 126(1), 167–176. https://doi.org/10.1007/s10549-010-1077-7

Committee on Appropriations, U. S, (2002). Breast Cancer Research and Development: Hearing Before a Subcommittee of the Committee on appropriations United States Senate. Prepared statement of Richard Klausner. US government printing office.



Connell, S., Patterson, C., Newman, B., (2006). Issues and concerns of young Australian women with breast cancer. Supportive Care in Cancer, 2006;14:419–426.

Conti, J. V., (1995). Job discrimination against people with a cancer history. Journal of Applied Rehabilitation Counseling, 26(2): 12-16.

Coons, S. J., Chongpison, Y., Wendel, C. S., Grant, M., Krouse, R. S., (2007). Overall quality of life and difficulty paying for ostomy supplies in the Veterans Affairs ostomy health-related quality of life study: an exploratory analysis. Med Care. 2007 Sep;45(9):891-5. doi: 10.1097/MLR.0b013e318074ce9b. PMID: 17712260.

Conley, C. C., Bishop, B. T., & Andersen, B. L. (2016). Emotions and Emotion Regulation in Breast Cancer Survivorship. Healthcare (Basel, Switzerland), 4(3), 56. https://doi.org/10.3390/healthcare4030056

Cordero, M. J. A., Villar, N. M., Sánchez, M. N., Pimentel-Ramírez, M. L., García-Rillo, A., & Valverde, G. E., (2015). Breast cancer and body image as a prognostic factor of depression: a case study in México City. Nutrición hospitalaria: Organo oficial de la Sociedad española de nutrición parenteral y enteral, 2015, Vol.31(1), pp.371-379.

Corner, J., (2008). Addressing the needs of cancer survivors: issues and challenges. Expert Rev Pharmacoecon Outcomes Res. 2008 Oct;8(5):443-51. doi: 10.1586/14737167.8.5.443.

Corner, J., & Bailey, C., (2008). Cancer Nursing: Care in Context. Part 5: Needs and Priorities in Cancer Care. Chapt 32: Ethnicity, differences and care. Published by Blackwell Publishing

Coronado, A. C., Tran, K., Chadder, J., Niu, J., Fung, S., Louzado, C., Rahal, R., & Collaboration with the System Performance Steering Committee and the Technical Working Group (2017). The experience of patients with cancer during diagnosis and treatment planning: a descriptive study of Canadian survey results. Current oncology (Toronto, Ont.), 24(5), 332–337. doi:10.3747/co.24.3782

Cossich, T., Schofield, P. E., & McLachlan, S. A., (2004). Validation of the Cancer Needs Questionnaire (CNQ) Short-Form version in an ambulatory cancer setting. DOI: 10.1023/B: QURE.0000037496. 94640.d9



Costanzo, E. S., Lutgendorf, S. K., & Roeder, S. L., (2011). Common-sense beliefs about cancer and health practices among women completing treatment for breast cancer. Psycho-Oncology Psycho-Oncology 20: 53–61 (2011) Published online 11 February 2010 in Wiley Online Library (wileyonline library.com). DOI: 10.1002/pon.1707

Costa-Requena, G., Ballester Arnal, R., & Gil, F., (2015). The influence of demographic and clinical variables on perceived social support in cancer patients. Revista de Psicopatología y Psicología Clínica, 20, 25-32, 2015 doi: 10.5944/rppc.vol.1.num.1.2015.14404

Côté, L., & Turgeon, J. (2005). Appraising qualitative research articlesin medicine and medical education. Medical Teacher, 27, 71–75. doi:10.1080/01421590400016308

Coulter, A., Parsons, S., & Aksham, J. (2008). Where are the patients in decision making about their own care? Health systems and policy analysis, World health Organisation, European Ministerial Conference on Health Systems.

Cowan, C., & Hoskins, R., (2007). Information preferences of women receiving chemotherapy for breast cancer. Eur J Cancer Care (Engl). 2007 Nov;16(6):543-50. doi: 10.1111/j.1365-2354.2007.00782. x. PMID: 17944771.

Cox, C. L., Zhu, L., Ojha, R. P., Li, C., Srivastava, D. K., Riley, B. B., Hudson, M. M., Robison, L. L. (2016). The unmet emotional, care/support, and informational needs of adult survivors of pediatric malignancies. Journal of cancer survivorship: research and practice, 10(4), 743–758. doi:10.1007/s11764-016-0520-0

Coyne, I. T., (1997). Sampling in qualitative research. Purposeful and theoretical sampling merging or clear boundaries. Journal of Advanced Nursing, 26, 623-630.

Craig, T. J., Comstock, G. W, Geiser, P. B., (1974). Quality of survival in breast cancer carecontrol comparison. Cancer 33: 1451-7

Crawford, K., (2018). Three Studies Reveal Cancer Patients Face Serious Financial Hardship, Do Not Discuss Treatment Costs with Physicians. https://www.asco.org/about-asco/presscenter/news-releases/three-studies-reveal-cancer-patients-face-serious-financial\



Creswell, J. W., (1994). Research design: qualitative & quantitative approaches. California: Sage

Creswell, J. W., Goodchild, L. F., & Turner., P. P., (1996). Integrated qualitative and quantitative research: Epistemology, history, and designs. In Higher education: Handbook of theory and research, ed. J. C. Smart, 90–136. New York: Agathon Press.

Creswell, J. W., (2003). Research design: Qualitative, quantitative and mixed methods approaches. 2nd ed. Thousand Oaks, CA: Sage

Creswell, J. W., Plano Clark, V. L., Gutmann, M., & Hanson, W. (2003). Advanced mixed methods research designs. In A. Tashakkori & C. Teddlie (Eds.), Handbook of mixed methods in social and behavioral research (pp. 209–240). Thousand Oaks, CA: Sage

Creswell, J. W., (2005). Educational research: Planning, conducting, and evaluating quantitative and qualitative approaches to research. 2nd ed. Upper Saddle River, NJ: Merrill/Pearson Education.

Creswell, J. W., & Clark, V. L. P., (2006). Designing and Conducting Mixed Methods Research. 3rd edition. CHAPTER 4: Choosing mixed methods design. Sage Publication. https://www.sagepub.com/sites/default/files/upm-binaries/10982_Chapter_4.pdf

Creswell, J., W., (2007). Research design. Qualitative and mixed methods approaches. London: Sage.

Creswell, J. W., (2008) Research Design: Qualitative, Quantitative and Mixed Methods Approaches, 3rd edn. Sage, Thousand Oaks, CA.

Creswell, J. W. (2009). Research design: Qualitative and mixed methods approaches. London: SAGE.

Creswell, J. W., & Plano Clark, V. L. (2011). Designing and conducting mixed method research (2nd ed.). Thousand Oaks, CA: Sage.

Creswell, J. W., (2013). Qualitative Inquiry and research design choosing among five approaches (3rd Ed). Thousand Oaks, CA: Sage Publications.



Creswell, J. W., (2013). Telling a complete story with qualitative and mixed methods research. https://www.youtube.com/watch?v=l5e7kVzMIfs

Crouch, A. T & Meurier, C., (2005). Health Assessment. Chapter 10: Cultural and spiritual health Assessment (pg 318).

Culbertson, M.G., Bennett, K., Kelly, C.M., Sharp, L., & Cahir, C., (2020). The psychosocial determinants of quality of life in breast cancer survivors: a scoping review. BMC Cancer 20, 948 (2020). https://doi.org/10.1186/s12885-020-07389-w

Cully, J. A., & Teten, A. L., (2008). A therapist's guide to brief cognitive behavioral therapy. Houston, TX: Department of Veterans Affairs South Central MIRECC.

Curry, L. A., Nembhard, I. M., & Bradley. E. H., (2009). "Qualitative and Mixed Methods Provide Unique Contributions to Outcomes Research." Circulation 119: 1442-52.

Çürük GN, Kartin PT, Şentürk A (2020) A Survey of social support and psychosocial compliance in patients with breast cancer. DOI: 10.17352/2581-4265.000043

Cypress, B. S., (2017). Rigor or Reliability and Validity in Qualitative Research: Perspectives, Strategies, Reconceptualization, and Recommendations. DOI: 10.1097/DCC.0000000000025. https://www.nursingcenter.com/wkhlrp/Handlers/articleContent.pdf?key=pdf_00003465-201707000-00007

Daher, M., (2012). Cultural beliefs and values in cancer patients. Annals of Oncology 23 (Supplement 3), pp66–69. DOI:10.1093/annonc/mds091

Damaskos, P., & Parry, C. (2015). Cancer survivorship: Concepts, interventions, and research. In G. Christ, C. Messner & L. Behar (Eds.), Handbook of oncology social work: Psychosocial care for people with cancer (pp. 57–64). New York, NY: Oxford University Press. https://media.cancercare.org/publications/original/366-hidden_cost.pdf

Dang, C. T., Yu, A. F., Jones, L. W., Liu, J., Steingart, R. M., Argolo, D. F., Hudis, C. A. (2016). Cardiac Surveillance Guidelines for Trastuzumab-Containing Therapy in Early-Stage



Breast Cancer: Getting to the Heart of the Matter. Journal of Clinical Oncology, 34(10), 1030–1033. http://doi.org/10.1200/JCO.2015.64.5515

Daudt, H. M. L., van Mossel, C., Dennis, D. L., Leitz, L., Watson, H. C., & Tanliao, J. J. (2014). Survivorship care plans: a work in progress. Current Oncology, 21(3), e466–e479. http://doi.org/10.3747/co.21.1781

Davey, M. P., Bilkins, B., Diamond, G., Willis, A. I., Mitchell, E. P., Davey, A., Young, F. M., (2016). African American Patients' Psychosocial Support Needs and Barriers to Treatment: Patient Needs Assessment. Journal of Cancer Education: The Official Journal of the American Association for Cancer Education, 01 Sep 2016, 31(3):481-487. DOI: 10.1007/s13187-015-0861-9 PMID: 26048632 PMCID: PMC4671828

Davis, K., Schoenbaum, S. C., & Audet, A. M., (2005). A 2020 vision of patient-centered primary care J Gen Intern Med 20 953–957 https://doi.org/10.1111/j.1525-1497.2005.0178.x PMID: 16191145 PMCID: 1490238

Davison, B. J., & Breckon, E. N., (2012). Impact of health information-seeking behaviour and personal factors on preferred role in treatment decision making in men with newly diagnosed prostate cancer. Cancer Nurs 35 411–418 https://doi.org/10.1097/NCC.0b013e318236565a

de Rijk, A., Amir, Z., Cohen, M., et. al., (2020). The challenge of return to work in workers with cancer: employer priorities despite variation in social policies related to work and health. J Cancer Surviv 14, 188–199 (2020). https://doi.org/10.1007/s11764-019-00829-y

Dean, L. T., Moss, S. L., McCarthy, A. M., & Armstrong, K. (2017). Healthcare System Distrust, Physician Trust, and Patient Discordance with Adjuvant Breast Cancer Treatment Recommendations. Cancer epidemiology, biomarkers & prevention: a publication of the American Association for Cancer Research, cosponsored by the American Society of Preventive Oncology, 26(12), 1745–1752. https://doi.org/10.1158/1055-9965.EPI-17-0479

Deacon, D., Bryaman, A., and Fenton, N., (1998). Collision or collusion? A discussion and case study of the unplanned triangulation of quantitative and qualitative research methods.



International Journal of Social Research Methodology. Volume 1, 1998 - Issue 1 https://doi.org/10.1080/13645579.1998.10846862

Dedeli, O., Fadiloglu, C., & Uslu, R., (2008). A survey of functional living and social support in patients with cancer. J Turk Oncol, 23, 132-39.

Degner, L.F., Kristjanson, L.J., Bowman, D., Sloan, J.A., Carriere, K.C., O'Neil, J., Bilodeau, B., Watson, P. & Mueller, B. (1997). Information needs and decisional preferences in women with breast cancer. Journal of the American Medical Association, 27(18), 1485-92.

Deimling, G. T., Bowman, K. F., Wagner, L. J., (2007). The effects of cancer-related pain and fatigue on functioning of older adult, long-term cancer survivors. Cancer Nursing 2007 ;30 (6):421–433

Dein, Simon., (2007). Culture and Cancer Care. Maidenhead: McGraw-Hill Education, 2007. Web. Chapter 12: Culture and cancer care

DellaCroce, F. J., Wolfe, E. T., (2012). Breast reconstruction. Surg Clin North Am. 2013 Apr;93(2):445-54. doi: 10.1016/j.suc.2012.12.004. Epub 2013 Jan 29. PMID: 23464695.

Delaney, L. J., (2018). Patient-centred care as an approach to improving health care in Australia. Collegian. Australian journal of nursing practice, scholarship and research. DOI: https://doi.org/10.1016/j.colegn.2017.02.005.

https://www.collegianjournal.com/article/S1322-7696(17)30042-2/fulltext

Delgado-Guay, M. O., De, La., Cruz, M. G., Epner, D. E., (2013). 'I don't want to burden my family': handling communication challenges in geriatric oncology. Ann Oncol. 2013 Oct;24 Suppl 7: vii30-35. doi: 10.1093/annonc/mdt263. PMID: 24001760.

Delgado-Guay, M., Ferrer, J., Rieber, A. G., Rhondali, W., Tayjasanant, S., Ochoa, J., Cantu, H., Chisholm, G., Williams, J., Frisbee-Hume, S., & Bruera, E., (2015). "Financial Distress and Its Associations with Physical and Emotional Symptoms and Quality of Life Among Advanced Cancer Patients." The Oncologist (Dayton, Ohio) 20.9 (2015): 1092-098. Web.



Delice, A., (2010). The Sampling Issues in Quantitative Research. Educational Sciences: Theory & Practice. Available at: http://files.eric.ed.gov/fulltext/EJ919871.pdf

Demark-Wahnefried, W., Aziz, N. M., Rowland, J. H., & Pinto, B. M. (2005). Riding the crest of the teachable moment: promoting long-term health after the diagnosis of cancer. Journal of clinical oncology: official journal of the American Society of Clinical Oncology, 23(24), 5814–5830. https://doi.org/10.1200/JCO.2005.01.230

Demark-Wahnefried W, Peterson B, McBride C., lipkis, I., Clipp, E., (2000). Current health behaviors and readiness to pursue life-style changes among men and women diagnosed with early-stage prostate and breast carcinomas. Cancer 2000; / 88: / 674 84.

de Moor, J. S., Mariotto, A. B., Parry, C., Alfano, C. M., Padgett, L., Kent, E. E., Forsythe, L., Scoppa, S., Hachey, M., Rowland, J. H., (2013). Cancer survivors in the United States: prevalence across the survivorship trajectory and implications for care. Cancer Epidemiol Biomarkers Prev. 2013 Apr;22(4):561-70. doi: 10.1158/1055-9965.EPI-12-1356. Epub 2013 Mar 27. PMID: 23535024; PMCID: PMC3654837.

den Bakker, C. M., Schaafsma, F. G., Huirne, J., Consten, E., Stockmann, H., Rodenburg, de Kekrk, C. J., Bonjer, H. J., & Anema, J. R., (2018). Cancer survivors' needs during various treatment phases after multimodal treatment for colon cancer - is there a role for eHealth? BMC cancer, 18(1), 1207. doi:10.1186/s12885-018-5105-z

Denlinger, C. S., Carlson, R. W., Are, M., Baker, K. S., Davis, E., Edge, S. B., Friedman,
D. L., Goldman, M., Jones, L., King, A., Kvale, E., Langbaum, T. S., Ligibel, J. A.,
McCabe, M. S., McVary, K. T., Melisko, M., Montoya, J. G., Mooney, K., Morgan, M.
A., O'Connor, T., ... Freedman-Cass, D. (2014). Survivorship: introduction and definition.
Clinical practice guidelines in oncology. Journal of the National Comprehensive Cancer
Network: JNCCN, 12(1), 34–45. https://doi.org/10.6004/jnccn.2014.0005

Denzin, N. K., & Lincoln, Y. S., (1998). Collecting and interpreting qualitative material. Thousand Oaks, CA: Sage.

Department of Global Health, University of Washington, (2018). http://dcp-3.org/translation



Department of Health, Abu Dhabi, DoH, AD., (2017). Cancer Facts & Statistics. 2017. Available at: https://www.haad.ae/simplycheck/tabid/140/Default.aspx Accessed September 29, 2017.

de Salazar, L. M., & Villar, R. C. L., (2018). Globalization and health inequalities in latin America. ISBN 978-3-319-67219-5. Chapter 15: Abet for the reduction of health inequalities in accordance with the conditions of the latin America region. Pg 286

DeShaw, P., (2006). Use of the emergency department by Somali immigrants and refugees. Minnesota medicine. 89(8):42-45.

de Souza, J. A., & Wong, Y. N. (2013). Financial Distress in Cancer Patients. Journal of medicine and the person, 11(2), 10.1007/s12682-013-0152-3. https://doi.org/10.1007/s12682-013-0152-3

Derogatis, L. R., Morrow, G. R., Fetting, J., Penman, D., Piasetsky, S., Schmale, A. M., Henrichs, M., Carnicke, C. L. Jr., (1983). The prevalence of psychiatric disorders among cancer patients. JAMA 249(6):751–757.

DeSantis, L., & Ugarriza, D. (2000). The concept of theme as used in qualitative nursing research. Western Journal of Nursing Research, 22, 351–372. doi:10.1177/01939459000220030

DeSantis, C. E., Lin, C. C., Mariotto, A. B., Siegel, R. L., Stein, K. D., Kramer, J. L., Alteri, R., Robbins, A. S., Jemal, A., (2014). Cancer treatment and survivorship statistics, 2014. CA Cancer J Clin. 2014 Jul-Aug;64(4):252-71. doi: 10.3322/caac.21235. Epub 2014 Jun 1. PMID: 24890451.

Dewis, R., & Gribbin, J., (2009). Breast Cancer: Diagnosis and Treatment - An Assessment of Need. NICE Clinical Guidelines, No. 80-81S. NHS National Institute for Health and clinical Excellence. Available at: http://www.ncbi.nlm.nih.gov/pubmed/21901869

Devine, E. C, & Westlake, S. K., (1995). The effects of psychoeducational care provided to adults with cancer: Meta-analysis of 116 studies. Oncol Nurs Forum 1995; / 22: / 136981.



de Vries, M., & Stiefel, F. (2014). Psycho-oncological interventions and psychotherapy in the oncology setting. In U. Goerling (Ed.), Psycho-oncology (pp. 121–135). doi:10.1007/978-3-642-40187-9_9

Dey, I., (1999). Grounding Grounded Theory: Guidelines for Qualitative Inquiry. San Diego: Academic Press.

Dhami, S., & Sheikh, A. (2000). The Muslim family: predicament and promise. Western Journal of Medicine, 173(5), 352–356.

Dias, L., Chabner, B. A., Lynch, T. J. Jr., Penson, R. T., (2003). Breaking bad news: a patient's perspective. Oncologist. 2003;8(6):587-96. doi: 10.1634/theoncologist.8-6-587. PMID: 14657537.

Diaz-Montes, T. P., (2010). Patients' guide to uterine cancerChapter 7. Surviving Uterine cancer – reenaging in mind and body health after treatment. Jones amd Bartlett Publishers Canada

DiSipio, T., Hayes, S., Newman, B., Janda, M., (2009). What determines the health-related quality of life among regional and rural breast cancer survivors? Aust N Z J Public Health. 2009;33(6):534–9.

Dizon, D. S., Suzin, D., & McIlvenna, S. (2014). Sexual health as a survivorship issue for female cancer survivors. The oncologist, 19(2), 202–210. doi:10.1634/theoncologist.2013-0302

Donabedian, A., (1973). Aspects of medical care administration: Specifying requirements for health care. Cambridge, UK: Harvard University Press.

Donnelly, T. T., Al Khater, A. H., Al-Bader, S. B., Al Kuwari, M. G., Al-Meer, N., Malik, M., Singh, R., Chaudhry, S. & Fung, T., (2013). Beliefs and attitudes about breast cancer and screening practices among Arab women living in Qatar: a crosssectional study. BMC Womens' Health, 13, 49.



Donnelly, T. T., Al-Khater, Al-Hareth, Al-K., Al-Bader, S. B., Al-Kuwari, M. G., Abdul Malik, M. A., Al-Meer, N., & Singh, R., (2017). Perceptions of Arab men regarding female breast cancer screening examinations—Findings from a Middle East study. PLoS ONE12(7): e0180696. https://doi.org/10.1371/journal.pone.0180696

Doorenbos, A. Z., (2014). Mixed methods in nursing research: an overview and practical examples. Kango Kenkyu. [Internet]. [cited 2016 May 01]; 47(3):207-17. Available from: http://www.ncbi.nlm.nih.gov/pmc/articles/PMC4287271/

Douglas, S. L., Daly, B. J., & Lipson, A. R. (2016). Relationship Between Physical and Psychological Status of Cancer Patients and Caregivers. Western journal of nursing research, 38(7), 858–873. doi:10.1177/0193945916632531.

Douma, K. F., Koning, C. C., Zandbelt, L. C., de Haes, H. C., Smets, E. M., (2012). Do patients' information needs decrease over the course of radiotherapy? Support Care Cancer. Vol:20:2167–76.

Doumit, M. A., El Saghir, N., Huijer, H. A. S., Kelley, J. H., & Nassar, N. (2010). Living with breast cancer, a Lebanese experience. European Journal of Oncology Nursing, 14(1), 42-48.

Doumit, M. A., El Saghir, N., Huijer, H. A. S., Kelley, J. H., & Nassar, N. (2009). Living with breast cancer, a Lebanese experience. European Journal of Oncology Nursing, (2009). DOI: 10.1016/j.ejon.2009.08.003

Dawson-Rose, C., Cuca, Y. P., Webel, A. R., Solís Báez, S. S., Holzemer, W. L., Rivero-Méndez, M., Sanzero Eller, L., Reid, P., Johnson, M. O., Kemppainen, J., Reyes, D., Nokes, K., Nicholas, P. K., Matshediso, E., Mogobe, K. D., Sabone, M. B., Ntsayagae, E. I., Shaibu, S., Corless, I. B., Wantland, D., ... Lindgren, T. (2016). Building Trust and Relationships Between Patients and Providers: An Essential Complement to Health Literacy in HIV Care. *The Journal of the Association of Nurses in AIDS Care: JANAC*, *27*(5), 574–584. https://doi.org/10.1016/j.jana.2016.03.001

Doyle, N., (2008). Cancer survivorship: evolutionary concept analysis. J Ad Nurs 2008; 62:499-509. Doi: 10.1111/j.1365-2648.2008.04617.x



Draborg, E., Kjær, T., Bech, M., Mollerup, M., (2009). Volume 5Denmark: Institute of Public Health, Health Economics, University of Southern Denmark, Dokumentationsrapport Spørgeskemaundersøgelse, PaRIS – Patientens Rejse i Sundhedssektoren [In Danish], Health Economics Papers

Drayton, M., Tamblyn, M., Mercer, C., (2012). Ontario Agency for Health Protection and Promotion, Public Health Ontario, Cancer Care Ontario. Taking action to prevent chronic disease: Recommendations for a healthier Ontario. Toronto, ON: Ontario Agency for Health Protection and Promotion; 2012. Canadian Electronic Library (Firm)

Dresser, R., (2012). Malignant – Medical ethicists confront cancer. Chapter3: Learning the Bad News. Oxford university press.

Dood, R. L., Zhao, Y., Armbruster, S. D., Coleman, R.L., Tworoger, S., Sood, A. K., & Baggerly, K. A. (2018). Defining survivorship trajectories across patients with solid tumors: An evidence-based approach. JAMA Oncology, 4, 1519–1526. https://doi.org/10.1001/jamaoncol.2018.2761

Du, R., Wang, X., Ma, L., Larcher, L. M., Tang, H., Zhou, H., Chen, C., Wang, T., (2021). Adverse reactions of targeted therapy in cancer patients: a retrospective study of hospital medical data in China. *BMC Cancer* 21, 206 (2021). https://doi.org/10.1186/s12885-021-07946-x

Dubai Health Authority., (2020). Dubai Health Insurance Corporation, Service Catalogue. Government of Dubai.

Dudas, S., & Carlson, C, E. (1988). Cancer rehabilitation. Oncology Nursing Forum, 15(2), 183-18

Dsouza, S. M., Vyas, N., Narayanan, P., Parsekar, S. S., Gore, M., & Sharan, K., (2018). A qualitative study on experiences and needs of breast cancer survivors in Karnataka, India. Clinical Epidemiology and Global Health. http://dx.doi.org/10.1016/j.cegh.2017.08.001 2213-3984/© 2017 INDIACLEN. Published by Elsevier



Eakin, E. G., Strycker, L. A., (2001). Awareness and barriers to use of cancer support and information resources by HMO patients with breast, prostate, or colon cancer: Patient and provider perspectives. Psycho-Oncology. Vol. 10(2):103–113.

Easterby-Smith, M., Thorpe, R., & Lowe, A., (1991). Management Research: An Introduction. London: Sage

Eaton, L. H., Narkthong, N. & Hulett, J. M., (2020). Psychosocial Issues Associated with Breast Cancer-Related Lymphedema: a Literature Review. Curr Breast Cancer Rep 12, 216– 224 (2020). https://doi.org/10.1007/s12609-020-00376-x

Eapen, V., & Revesz, T., (2003). Psychosocial correlates of paediatric cancer in the United Arab Emirates. Supportive Care in Cancer. Vol.11(3):185-9. Epub 2003 Jan 21. DOI: 10.1007/s00520-002-0423-7

Echlin, K. N., Rees, C. E., (2002). Information needs and information-seeking behaviors of men with prostate cancer and their partners: a review of the literature. Cancer Nurs. 2002, 25 (1): 35-41. 10.1097/00002820-200202000-00008.

Edib, Z., Kumarasamy, V., Abdullah, N. B., Rizal, A. M., Al-Dubai, S. A. R., (2016). Most prevalent unmet supportive care needs and quality of life of breast cancer patients in a tertiary hospital in Malaysia. Health and Quality of Life Outcomes (2016) 14:26 DOI 10.1186/s12955-016-0428-4

Eggly, S., Afonso, N., Rojas, G., Cardozo, L., & Robertson, R. S., (1997). An assessment of residents' competence in the delivery of bad news to patients. Academic Medicine: Journal of Association of American Medical Colleges. 1997; 72:397-399.

Ehmke, N. J., Biddle, H., Hunnicutt, B. A., Trenkner. S., (2016). SEASONS: A prospective study assessing the physical, psychosocial, spiritual, and financial needs of patients with breast and prostate cancer. DOI: 10.1200/jco.2016.34.3_suppl.4 Journal of Clinical Oncology 34, no. 3_suppl (January 20, 2016) 4-4. Published online January 20, 2016.

Eilertsen, MEB, Jozefiak, T, Rannestad, T. (2012) Quality of life in children and adolescents surviving cancer. European Journal of Oncology Nursing 16(2): 185–193.



Elbarazi, I., (2015). Doctorate thesis. Exploring knowledge and information needs among cancer patients in the United Arab Emirates during their cancer treatment journey. Available at: http://flex.flinders.edu.au/file/8c6d1d32-d1d5-4542-93d3-fa30226ccbfc/1/Iffat%20Elbarazi%20Dissertation%20Final%20Version.pdf

Elbarazi, I., (2016). Exploring knowledge and information needs among cancer patients in the United Arab Emirates during their cancer treatment journey

Elder, E. E., Brandberg, Y., Björklund, T., Rylander, R., Lagergren, J., Jurell, G., Wickman, M., Sandelin, K., (2005). Quality of life and patient satisfaction in breast cancer patients after immediate breast reconstruction: a prospective study. Breast. doi: 10.1016/j.breast.2004.10.008. PMID: 15927829.

Ell, K., Xie, B., Wells, A., Nedjat-Haiem, F., Lee, P. J., Vourlekis, B., (2008). Economic stress among low-income women with cancer: effects on quality of life. Cancer. 2008; 112(3):616–625.

Ell, K. & Nishimoto, R., (1989). Coping resources in adaptation to cancer-analysis of socioeconomic and racial differences, Social Services Review, 63: 443 – 6.

Elkin, E. B., Kim, S. H., Casper, E. S., Kissane, D. W., Schrag, D. (2007). Desire for information and involvement in treatment decisions: elderly cancer patients' preferences and their physicians' perceptions. J Clin Oncol. 2007. pp. 5275–80. Available from http://www.ncbi.nlm.nih.gov/entrez/query.fcgi?cmd=Retrieve&db=PubMed&dopt=Cita tion&list_uids=18024875.

Ellis, L., Woods, L. M., Estève, J., Eloranta, S., Coleman, M. P., & Rachet, B., (2014). Cancer incidence, survival and mortality: Explaining the concepts. International Journal of Cancer. Volume 135, Issue 8. ttps://doi.org/10.1002/ijc.28990

Elobaid, Y. E., (2014). Breast cancer presentation delay among women in the United Arab Emirates. (2014). Dissertations. https://scholarworks.uaeu.ac.ae/all_dissertations/31



Elobaid, Y., Aw, T-C; Lim, J. N. W., Hamid, S., Grivna, M., (2016). Breast cancer presentation delays among Arab and national women in the UAE: a qualitative study. Population Health, Vol.2, pp.155-163. https://doi.org/10.1016/j.ssmph.2016.02.007

Elobaid, E. Y., Aw, T-C., Grivna, M., & Nagelkerke, N., (2014). Breast Cancer Screening Awareness, Knowledge, and Practice among Arab Women in the United Arab Emirates: A Cross-Sectional Survey.

Elobaid, Y., Aamir, M., Grivna, M., Suliman, A., Attoub, S., Mousa, H., et al. (2021) Breast cancer survival and its prognostic factors in the United Arab Emirates: A retrospective study. PLoS ONE 16(5): e0251118. https://doi.org/10.1371/journal.pone.0251118

El Saghir, N. S., Khalil, M. K., E, T., El Kinge, A. R., Charafeddine, M., Geara, F., Seoud, M., Shamseddine, A. I., (2007). Trends in epidemiology and management ofbreast cancer in developing Arab countries: A literature and registry analysis. International Journal of Surgery (2007)5, 225e233. doi: 10.1016/j.ijsu.2006.06.015

El-Sharkawi, F. M., Sakr, M. F., Atta, H. Y., Ghanem, H. M., (1997). Effect of different modalities of treatment on the quality of life of breast cancer patients in Egypt. La revue de santé de la méditeranée orientale, 1997;3(1):68–81.

Ely, M., Vinz, R., Downing, M., & Anzul, M. (1997). On writing qualitative research: Living by words. London: Routledge/Falmer.Emanuel, L., Ferris, F. D., Gunten, C. F., Roenn, J. H., (2010). Cancer Survivorship: Role of Healthcare Providers - Medscape - Sep 21, 2010

Entwistle, V. A., & Watt, I. S. (2013). Treating patients as persons: a capabilities approach to support delivery of person-centered care. The American journal of bioethics: AJOB, 13(8), 29–39. doi:10.1080/15265161.2013.802060

Epplein, M., Zheng, Y., Zheng, W., Chen, Z., Gu, K., Penson, D., Lu, W., Shu, X., (2011). Quality of life after breast cancer diagnosis and survival. Journal of Clinical Oncology, 29, 406–412, doi:10.1200/JCO.2010.30.6951

Epping-Jordan, E. J., Compas, B. E., Osowiecki, D. M., Oppedisano, G., Gerhardt, G., Primo, K., and Krag, D. N., (1999). Psychological Adjustment in Breast Cancer: Processes



of Emotional Distress. Health Psychology 1999, Vol. 18, No. 4, 315-326. American Psychological Association, Inc. 0278-6133/99/S3.00.

Epstein, R. M., & Street, R. L. (2007). Patient-centered communication in cancer care: Promoting healing and reducing suffering. National Cancer Institute; 2007. NIH Publication No. 07-6225.

Erdogan, S., (2014). Qualitative Research: Nursing Research, Process, Practice and Critical. Edits: Erdogan, S., Nahcivan, N., Esin, N. M., Nobel Medicine Bookstore 2014, İstanbul, p:131-165.

Erzberger, C., & Kelle, U. (2003). Making inferences in mixed methods: The rules of integration. In A. Tashakkori & C. Teddlie (Eds.), Handbook of mixed methods in social and behavioral research (1st ed.) (pp. 457-490). Thousand Oaks, CA: Sage.

Esbensen, B. A., Osterlind, K., Roer, O., Hallberg, I. R., (2004). Quality of life of elderly persons with newly diagnosed Cancer. Eur J Cancer Care, 13, 443-53.

Etikan, I., Musa, A. A., Alkassim, R. S., (2016). Comparison of Convenience Sampling and Purposive Sampling. American Journal of Theoretical and Applied Statistics. 2016; 5(1): 1-4. Published online December 22, 2015 (http://www.sciencepublishinggroup.com/j/ajtas) doi: 10.11648/j.ajtas.20160501.11. ISSN: 2326-8999 (Print); ISSN: 2326-9006 (Online)

Eun-young, J., & Hyunjin, O., (2018). Changed Body and Marital Intimacy among Women with Breast Cancer in Korea. International Journal of Applied Engineering Research ISSN 0973-4562 Volume 13, Number 16 (2018) pp. 12516-12521. Research India Publications. https://www.ripublication.com/ijaer18/ijaerv13n16_15.pdf

Everly, G. S., (1989). A clinical guide to the treatment of the human stress response. Chapter 7: Control and the human stress response. Control and Recovery. Plenum Press, New York.

Ewertz, M., Bonde Jensen, A. (2011). Late effects of breast cancer treatment and potentials for rehabilitation Acta Oncologica 50 pp.187-193



Exline, J. J., Prince-Paul, M., Root, B. L., & Peereboom, K. S., (2013). The Spiritual Struggle of Anger toward God: A Study with Family Members of Hospice Patients. Journal of Palliative Medicine 16: 369–75.

Extermann, M., (2000). Measuring comorbidity in older cancer patients. Eur J Cancer. 2000 Mar;36(4):453-71. doi: 10.1016/s0959-8049(99)00319-6. PMID: 10717521.

Ezenkwele, U. A., & Roodsari, G. S. (2013). Cultural competencies in emergency medicine: Caring for Muslim-American patients from the Middle East. Journal of Emergency Medicine, 45, 168–174. doi: 10.1016/j.jemermed.2012.11.077

Faller, H., Schuler, M., Richard, M., Heckl, U., Weis, J., Küffner, R., (2013). Effects of psycho-oncologic interventions on emotional distress and quality of life in adult patients with cancer: systematic review and meta-analysis. J Clin Oncol. 2013 Feb 20; 31(6):782-93.

Faller, H., Koch, U., Brähler, E. Härter, M.; Keller, M., Schulz, H., Wegscheider, K., Weis, J., Boehncke, A., Hund, B., Reuter, K., Richard, M., Sehner, S., Szalai, C., . Wittchen, H.-U, Mehnert, A., (2016). Satisfaction with information and unmet information needs in men and women with cancer. *J Cancer Surviv* 10, 62–70 (2016). https://doi-org.salford.idm.oclc.org/10.1007/s11764-015-0451-1

Faller, H., Brähler, E., Härter, M., Keller, M., Schulz, H., Wegscheider, K., Weis, J., Boehncke, A., Reuter K, Richard, M., Sehner, S., Koch, U., Mehnert, A., (2017a). Unmet needs for information and psychosocial support in relation to quality of life and emotional distress: A comparison between gynecological and breast cancer patients. Patient Educ Couns. 2017 Oct;100(10):1934-1942. doi: 10.1016/j.pec.2017.05.031. Epub 2017 May 24. PMID: 28592366.

Fallowfield, L. & Clark, A., (1991). Chap 3: Finding a breast lump and hearing the diagnosis. https://ebookcentral.proquest.com/lib/salford/reader.action?docID=170001&ppg=8

Fallowfield, L., & Jenkins, V., (1999). Effective Communication Skills are the Key to Good Cancer Care (1999). European Journal of Cancer, Vol. 35, No. 11, pp. 1592-1597. Published by Elsevier Science Ltd.



Fallowfield, L., & Jenkins, V., (2004). Communicating sad, bad, and difficult news in medicine. Lancet. 2004; 363:312–319.

Fatima, N., Zaman, M. U., Maqbool, A., Khan, S. H., Riaz, N., (2013). Lower incidence but more aggressive behavior of right sided breast cancer in Pakistani women: does right deserve more respect? Asian Pac J Cancer Prev. 2013;14(1):43-5. doi: 10.7314/apjcp.2013.14.1.43. PMID: 23534768.

Fawcett, J., (2015). Invisible nursing research: thoughts about mixed methods research and nursing practice. Nurs Sci Quart [Internet]. 2015 [cited 2016 May 01]; 28(2):167-8. Available from: http://www.ncbi.nlm.nih.gov/pubmed/25805392

Fearon, D., Hughes, S., & Brearley, S. G., (2019). Experiences of breast cancer in Arab countries. A thematic synthesis. Quality of Life Research (2020) 29:313–324 https://doi.org/10.1007/s11136-019-02328-0

Fearon, D., Hughes, S. & Brearley, S. G., (2020). Experiences of breast cancer in Arab countries. A thematic synthesis. Qual Life Res 29, 313–324 (2020). https://doi.org/10.1007/s11136-019-02328-0

Fenn, K. M., Evans, S. B., McCorkle, R., DiGiovanna, M. P., Pusztai, L., Sanft, T., Hofstatter, E. W., Killelea, B. K., Knobf, M. T., Lannin, D. R., Abu-Khalaf, M., Horowitz, N. R., Chagpar, A. B., (2014). Impact of financial burden of cancer on survivors' quality of life. J Oncol Pract. 2014 Sep;10(5):332-8. doi: 10.1200/JOP.2013.001322. Epub 2014 May 27. PMID: 24865220.

Ferrell, B. R., & Hassey, D. K., (1997). Quality of life among long-term cancer survivors. Oncology 11, 565–576

Ferrell, B. R., Grant, M. M., Funk, B. M., Otis-Green, S. A., & Garcia, N. J. (1998). Quality of life in breast cancer survivors: Implications for developing support systems. Oncology Nursing Forum, 25, 887–895.



Ferlay, J., Soerjomataram, I., Dikshit, R., et. al., (2015). Cancer incidence and mortality worldwide: sources, methods and major patterns in GLOBOCAN. Int J Cancer 2015;5: E359–386.

Ferlay, J., Colombet, M., Soerjomataram, I., Mathers, C., Parkin, D. M., Piñeros, M., Znaor, A., Bray, F., (2019). Estimating the global cancer incidence and mortality in 2018: GLOBOCAN sources and methods. Int J Cancer. 2019 Apr 15;144(8):1941-1953. doi: 10.1002/ijc.31937. Epub 2018 Dec 6. PMID: 30350310.

Fernandes-Taylor, S., Adesoye, T., & Bloom, J. R. (2015). Managing psychosocial issues faced by young women with breast cancer at the time of diagnosis and during active treatment. Current opinion in supportive and palliative care, 9(3), 279–284. https://doi.org/10.1097/SPC.00000000000161

Fetters, M. D., Curry, L. A., & Creswell, J. W., (2013). "Achieving integration in mixed methods designs--principles and practices." Health Services Research, vol. 48, no. 6, Dec. 2013, p. 2134+. Gale Academic OneFile, https://link-gale-com.salford.idm.oclc.org/apps/doc/A353995444/AONE?u=salcal2&sid=AONE&xid=a32fe8 41. Accessed 2 June 2020. DOI: http://dx.doi.org.salford.idm.oclc.org/10.1111/1475-6773.12117

Feuerstein, M., (2007). "Defining cancer survivorship," Journal of Cancer Survivorship, vol. 1, no. 1, pp. 5–7, 2007

Fieldler, C; & Koppelman, K. L; (2012). Robbery and Redemption: Cancer as Identity Theft. Chapter 3: The Cancer Journey and Detours Along the Way. Baywod, Publishing Company

Fielding, R., Lam, T. W. W., Shun, C. S., Okuyama, T., Lai, Hur. Y., Wada, M., Akechi, T., (2013). Attributing Variance in Supportive Care Needs during Cancer: Culture-Service, and Individual Differences, before Clinical Factors. PLoS ONE, 8(5), e65099. http://doi.org/10.1371/journal.pone.0065099

Fife, B. L., Kennedy, V. N., & Robinson, L., (1994). Gender and Adjustment to Cancer. Journal of Psychosocial Oncology, 12, 1-21. DOI:10.1300/ J077V12N01_01



Figg, W. D., Smith, E. K., Price, D. K., English, B. C., Thurman, P. W., Steinberg, S. M., Emanuel, E., (2010). Disclosing a diagnosis of cancer: where and how does it occur? J Clin Oncol. 2010 Aug 1;28(22):3630-5. doi: 10.1200/JCO.2009.24.6389. Epub 2010 Jul 6. PMID: 20606078; PMCID: PMC2917319.

Fillon, M; (2013). Young-Adult Cancer Survivors Face Unique Challenges, JNCI: Journal of the National Cancer Institute, Volume 105, Issue 20, 16 October 2013, Pages 1517–1518, https://doi.org/10.1093/jnci/djt294

Findik, U. Y., (2017). The Information Needs of Women Who Have Undergone Breast Cancer Surgery in the West of Turkey. Findik, U.Y. J Canc Educ (2017) 32: 432. https://doi.org/10.1007/s13187-015-0965-2

Fines, D., Saunders, M., & Forbes-Burford, J., (2013). Community Health Care Nursing.4th Ed. Chapter 5: Health Assessment and the Community Nurse. Wiley-Blackwell

Fingeret, M. C., Teo, I., & Epner, D. E. (2014). Managing body image difficulties of adult cancer patients: lessons from available research. Cancer, 120(5), 633–641. https://doi.org/10.1002/cncr.28469

Fitch, M. I., (2000). Supportive care for cancer patients. Hospital Quarterly, 3(4), 39–46. doi:10.5737/1181912x181614

Fitch, M. I., (2008). Supportive care framework. Can Oncol Nurs J. 2008 Winter;18(1):6-24. English, French. doi: 10.5737/1181912x181614. PMID: 18512565.

Fitch, M. I., (2019). Exploring Experiences of Survivors and Caregivers Regarding Lung Cancer Diagnosis, Treatment, and Survivorship. Journal of Patient Experience. April 2020:193-199. doi:10.1177/2374373519831700

Fitch, M., (2020). Living with body image changes following completion of cancer treatment. Canadian oncology nursing journal = Revue canadienne de nursing oncologique, 30(3), 231–234.



Fiscella, K., & Epstein, R. M., (2008). So much to do, so little time: care for the socially disadvantaged and the 15-minute visit. Archives of internal medicine, 168(17), 1843–1852. https://doi.org/10.1001/archinte.168.17.1843

Fischer, D. J., Villines, D., Kim, Y. O., Epstein, J. B., Wilkie, D. J., (2010). Anxiety, depression and pain: Differences by primary cancer. Support Care Cancer. Vol.18(7): 801–810. doi: 10.1007/s00520-009-0712-5

Fiszer, C., Dolbeault, S., Sultan, S., Bredart, A., (2014). Prevalence, intensity, and predictors of the supportive care needs of women diagnosed with breast cancer: a systematic review, Psycho-oncology. Vol. 23 (pg. 361-74), 361–374.10.1002/pon. v23.4

Fitzpatrick, T. R, (2016). Treating Vulnerable Populations of Cancer Survivors: A Biopsychosocial Approach. Chapter 5 Survivorship issues among muslim women with cancer pg 57

Fletcher R. H., O'Malley, M. S., Earp, J. A., Littleton, T. A., Fletcher, S. W., Greganti, M. A. Davidson, R. A., Taylor, J., (1983). Patients' priorities for medical care. Med Care 1983; 21:234–42.

Folkman, S., & Greer, S., (2000). Promoting psychological well-being in the face of serious illness: when theory, research and practice inform each other. Psych oncology. Vol.9(1):11-9.

Fong, E. J., & Cheah, W. L., (2016). Unmet Supportive Care Needs among Breast Cancer Survivors of Community-Based Support Group in Kuching, Sarawak. International Journal of Breast Cancer. Volume 2016, Article ID 7297813, 9 pages. http://dx.doi.org/10.1155/2016/7297813.

https://www.hindawi.com/journals/ijbc/2016/7297813/

Foot, G., (1996). Needs assessment in tertiary and secondary oncology practice: a conceptual and methodological exposition [Ph.D. dissertation]. Newcastle: University of Newcastle, 1996

Ford, S., Fallowfield, L., Lewis, S., (1996). Doctor-patient interactions in oncology. Social Science and Medicine. 1996;42:1511–1519. doi: 10.1016/0277-9536(95)00265-0.



Fonseca, S., Lencastre, L., & Guerra, M., (2014). Life Satisfaction in Women with Breast Cancer. https://doi.org/10.1590/1982-43272459201403

Fors, E. A., Bertheussen, G. F., Thune, I., (2011). Psychosocial interventions as part of breast cancer rehabilitation programs? Results from a systematic review. Psycho-Oncology, 2011;20(9):909–918.

Forouzi, M. A., Tirgari, B., Safarizadeh, M. H., & Jahani, Y. (2017). Spiritual Needs and Quality of Life of Patients with Cancer. Indian journal of palliative care, 23(4), 437–444. doi: 10.4103/IJPC.IJPC_53_17Fors, E. A., Bertheussen, G. F., Thune, I., (2011). Psychosocial interventions as part of breast cancer rehabilitation programs? Results from a systematic review. Psycho-Oncology, 2011;20(9):909–918.

Foster, C., Wright, D., Hill, H., Hopkinson, J., Roffe, L., (2009). Psychosocial implications of living 5 years or more following a cancer diagnosis: a systematic review of the research evidence. European Journal of cancer care. 18(3):223–247. DOI: 10.1111/j.1365-2354.2008. 01001.x

Foss, C., & Ellefsen, B., (2002). The value of combining qualitative and quantitative approaches in nursing research by means of method triangulation. Journal of Advanced Nursing, 40 (2) (2002), pp. 242-248

Fraenkel, L., (2013). Incorporating patients' preferences into medical decision making. Medical care research and review: MCRR, 70(1 Suppl), 80S–93S. doi:10.1177/1077558712461283

Freda, M.C. (2004). Issues in patient education', Journal of Midwifery and Womens' Health, 49(3).

Freeman, H. P., (2004). Poverty, culture, and social injustice: determinants of cancer disparities. CA Cancer J Clin. 2004 Mar-Apr; 54(2):72-7.

Friend of cancer Patients, (2018). https://focp.ae/wp-content/uploads/2018/10/Decade-study-UAE-V4Final.pdf. https://www.thenational.ae/uae/health/younger-women-need-breast-cancer-checks-health-experts-warn-1.448028



Frenkel, M., & Cohen, L. (2014). Effective Communication About the Use of Complementary and Integrative Medicine in Cancer Care. Journal of Alternative and Complementary Medicine, 20(1), 12–18. http://doi.org/10.1089/acm.2012.0533

Fronda, C. R. A., (2017). Perceptions, Beliefs, and Behaviors Toward Breast Cancer Screening of Filipino Women in Saudi Arabia. Doctoral Study Submitted in Partial Fulfillment. Walden University. https://scholarworks.waldenu.edu/cgi/viewcontent.cgi?article=4664&context=dissertations

Frosch, D. L., & Kaplan, R. M., (1999). Shared decision making in clinical medicine: past research and future directions. American Journal of Preventive Medicine 1999, 17:285-294

Fu, M. R., & Rosedale, M., (2009). Breast cancer survivors' experiences of lymphedemarelated symptoms. Journal of Pain and Symptom Management. 38:849–859. doi: 10.1016/j.jpainsymman.2009.04.030.

Fu, M. R., Ridner, S. H., Hu. S. H., Stewart, B. R., Cormier, J. N., Armer, J. M., (2013). Psychosocial impact of lymphedema: a systematic review of literature from 2004 to 2011. Psycho-Oncology. 2013;22(7):1466–84.

Gallant, M. P., (2013). Social Networks, Social Support, and Health-Related Behavior. The Oxford Handbook of Health Communication, Behavior Change, and Treatment Adherence. Edited by Leslie R. Martin and M. Robin DiMatteo. DOI: 10.1093/oxfordhb/9780199795833.013.016

Galway, K., Black, A., Cantwell, M., Cardwell, C., Mills, M., & Donnelly, M. (2012). Psychosocial interventions to improve quality of life and emotional wellbeing for recently diagnosed cancer patients. Cochrane database of systematic reviews (Online), 2012(11), [CD007064]. DOI: 10.1002/14651858.CD007064.pub2

Ganz, P. A., (2003). "Why and how to study the fate of cancer survivors: observations from the clinic and the research laboratory," European Journal of Cancer, vol. 39, no. 15, pp. 2136–2141, 2003.

Ganz, P. A., (2007). Cancer Survivorship: Today and Tomorrow. Published by Springer



Ganz, P. A., Hirji, K., Sim, M. S., Schag, C. A., Fred, C., Polinsky, M. L., (1993). Predicting psychosocial risk in patients with breast cancer. Med Care 31(5):419–431

Ganz, P. A, Desmond, K. A, Leedham, B., Rowland, J. H., Meyerowitz, B. E., Belin, T. R., (2002). Quality of life in long-term, disease-free survivors of breast cancer: A follow-up study. J Natl Cancer Inst 2002; / 94: / 3949.

Ganz, P. A., Yip, C. H., Gralow, J. R., Distelhorst, S. R., Albain, K. S., Andersen, B. L., Bevilacqua, J. L., de Azambuja, E., El Saghir, N. S., Kaur, R., McTiernan, A., Partridge, A. H., Rowland, J. H., Singh-Carlson, S., Vargo, M. M., Thompson, B., Anderson, B. O., (2013). Supportive care after curative treatment for breast cancer (survivorship care): resource allocations in low- and middle-income countries. A Breast Health Global Initiative 2013 consensus statement. Breast 2013; 22:606-615.

Ganz, P. A., & Stanton, A. L., (2015). Living with metastatic breast cancer. Advances in Experimental Medicine and Biology, 862, 243–254. doi:10.1007/978-3-319-16366-6_16

Garlick, M., Wall, K., Corwin, D., Koopman, C., (2011). Psycho-spiritual integrative therapy for women with primary breast cancer. J Clin Psychol Med Settings. 2011 Mar; 18(1):78-90.

Garrett-Rucks, P., (2016). Intercultural Competence in Instructed Language Learning: Bridging Theory & Practice. Chapter 2: Defining Culture and Its Role in Foreign Language Learning. Information Age Publishing

Garza, M. A., Luan, J., Blinka, M., Farabee-Lewis, R. I., Neuhaus, C. E., Zabora, J. R., Ford, J. G., (2005). A culturally targeted intervention to promote breast cancer screening among low-income women in East Baltimore, Maryland. Cancer Control. 2005 Nov;12 Suppl 2:34-41

Geertz, C., (1973). From The Interpretation of Cultures - Selected Essays. Chapter 2. The impact of the concept of culture o the concept of man. Pg. 13. https://web.mit.edu/allanmc/www/geertz.pdf [Accessed 14 May 2020].



Geraci, J. M., Escalante, C. P., Freeman, J. L., & Goodwin, J. S., (2005). Comorbid disease and cancer: the need for more relevant conceptual models in health services research. Journal of clinical oncology: official journal of the American Society of Clinical Oncology, 23(30), 7399–7404. https://doi.org/10.1200/JCO.2004.00.9753

Geyikci, R., Cakmak, S., Demirkol, M. E., & Uguz, S. (2018). Correlation of anxiety and depression levels with attitudes towards coping with illness and sociodemographic characteristics in patients with a diagnosis of breast cancer: Journal of psychiatry and neurological sciences. Dusunen Adam, 31(3), 246-257. doi: http://dx.doi.org.salford.idm.oclc.org/10.5350/DAJPN2018310302

Ghaemi, S. Z., Keshavarz, Z., Tahmasebi, S., Akrami, M., Heydari, S. T., (2019). Conflicts women with breast cancer face with: A qualitative study. J Family Med Prim Care [serial online] 2019 [cited 2021 Feb 1]; 8:27-36. Available from: https://www.jfmpc.com/text.asp?2019/8/1/27/251155

Glatzer, M., Panje, C. M., Siren, C., Cihoric, N., & Putora, P. M., (2018). Decision Making Criteria in Oncology. Oncology and Informatics – Review. DOI: 10.1159/000492272. https://www.karger.com/Article/Pdf/492272

Glaser, B. G., Strauss, A. L., & Strutzel, E., (1968). The Discovery of Grounded Theory; Strategies for Qualitative Research. Nurs Res. 1968;17(4):364.

Glasman, J., (2020). Humanitarianism and the Quantification of Human Needs: Minimal Humanity. Published by Routledge, New York

Glickman, S. W., Boulding, W., Manary, M., Staelin, R., Roe, M. T., Wolosin, R. J., et. al., (2010). Patient satisfaction and its relationship with clinical quality and inpatient mortality in acute myocardial infarction. Circ Cardiovasc Qual Outcomes. 2010;3(2):188–95.

Gielen, J., Bhatnagar, S., Chaturvedi, S. K., (2016). Spirituality as an ethical challenge in Indian palliative care: A systematic review. Palliat Support Care. 2016 Oct;14(5):561-82. doi: 10.1017/S147895151500125X. Epub 2015 Oct 29. PMID: 26510891



Gilbert, S. M., Miller, D. C., Hollenbeck, B. K., Montie, J. E., & Wei, J. T. (2008). Cancer survivorship: challenges and changing paradigms. The Journal of urology, 179(2), 431–438. https://doi.org/10.1016/j.juro.2007.09.029

Ginter, a. C., & Baun, B., (2017). Social support needs of breast cancer patients without partners. Journal of Social and Personal Relationships. https://doi.org/10.1177/0265407517718390

Girgis, A., Boyes, A., Sanson-Fisher, R. W., & Burrows, S., (2000). Perceived needs of women diagnosed with breast cancer: rural versus urban location. Perceived needs of women diagnosed with breast cancer: rural versus urban location.

Giesler, J. M., Faust, T., Ernst, J., Kuhnt, S., Mehnert, A., Weis, J., (2019). Selected quality Characteristics of out-patient psychosocial Cancer counseling centers in Germany: a Nationwide analysis. Gesundheitswesen. 2019;81(4):336–43.

Giunti, A., (2013). Nurse Navigation and the Transition to Cancer Survivorship: A Review of Determinants Essential to Program Success. Revue interdisciplinaire des sciences de la santé - Interdisciplinary Journal of Health Sciences 3(2):26. DOI: 10.18192/riss-ijhs.v3i2.1342

Givler, A., Bhatt, H., Maani-Fogelman, P. A., (2020). The Importance of cultural competence in pain and palliative care. [Updated 2020 Dec 1]. In: StatPearls [Internet]. Treasure Island (FL): StatPearls Publishing; 2020 Jan-. Available from: https://www.ncbi.nlm.nih.gov/books/NBK493154/

Glare, P. A., Davies, P. S., Finlay, E., Gulati, A., Lemanne, D., Moryl, N., Oeffinger, K. C., Paice, J. A., Stubblefield, M. D., Syrjala, K. L., (2014). Pain in cancer survivors. J Clin Oncol. 2014 Jun 1;32(16):1739-47. doi: 10.1200/JCO.2013.52.4629. Epub 2014 May 5. PMID: 24799477; PMCID: PMC4031191.

Gleeson, H., Calderon, A., Swami, V., Deighton, J., Wolpert, M., Edbrooke-Childset, J., (2016). Systematic review of approaches to using patient experience data for quality improvement in healthcare settings. BMJ Open 6: e011907, 2016



Global Health Delivery–Dubai, (2018). Health System Strengthening and Breast Cancer Care in the Middle East. Proceedings of the Harvard Medical School Center for Global Health Delivery–Dubai. 2018, Dubai, United Arab Emirates. Chapter 2: Breast cancer in the Middle East: Harvard Research Study findings by country and regionally. https://ghddubai.hms.harvard.edu/files/ghd_dubai/files/breast_cancer_proceedings.pdf

Globocan, (2015). Fact Sheets by Population. N.p., n.d. Web. 25 Nov. 2015

Globocan, (**2018a**). The Global Cancer Observatory - All Rights Reserved, March 2019. Accessed 20th May 2019. http://gco.iarc.fr/today/data/factsheets/cancers/39-All-cancers-fact-sheet.pdf

Globocan. (2018b). World Health Organisation: East Mediterranean region. Retrieved April 1, 2019, from http://gco.iarc.fr/today /data/factsheets /population s/993-who-east-mediterranean-region-emro-fact-sheets.pdf

Global Cancer Observatory (2018).United Arab Emirates - Global Cancer Observatory.InternationalAgencyforResearchonCancer.2018.https://gco.iarc.fr/today/data/factsheets/populations/784-united-arab-emirates-factsheets.Pdf

Goerling, U; Faller, H; Hornemann, B; Hönig, K; Bergelt, C; Maatouk, I; Stein, B; Teufel, M; Erim, Y; Geiser, F; Niecke, A; Senf, B; Wickert, M; Büttner-Teleaga, A; Weis, J., (2020). Information needs in cancer patients across the disease trajectory. A prospective study. Patient Education and Counseling.Volume 103, Issue 1, January 2020, Pages 120-126. https://doi.org/10.1016/j.pec.2019.08.011

Golden-Kreutz, D. M., Andersen, B. L., (2004). Depressive Symptoms after breast surgery: Relationships with global, cancer-related and life event stress. Psycho-Oncology, 13(3), 211– 220. http://doi.org/10.1002/pon.736

Goldman M, Chang D., (2010). Maximizing the value of a breast center. Breast. 2010 Aug;19(4):253-9. doi: 10.1016/j.breast.2010.03.019. Epub 2010 Apr 18. PMID: 20400310



Hebert, R., Zdaniuk, B., Schulz, R., & Scheier, M., (2009). Positive and negative religious coping and well-being in women with breast cancer. Journal of palliative medicine, 12(6), 537–545. https://doi.org/10.1089/jpm.2008.0250

Goldzweig, G., Hubert, A., Walach, N., Brenner, B., Perry, S., Andritsch, E., & Baider, L., (2009). Gender and psychological distress among middle- and older- aged colorectal cancer patients and their spouses: An unexpected outcome. Critical Reviews in Oncology/Hematology, 70, 71-82.

Golics, C. J., Basra, M. K., Finlay, A. Y., & Salek, S. (2013). The impact of disease on family members: a critical aspect of medical care. Journal of the Royal Society of Medicine, 106(10), 399–407. https://doi.org/10.1177/0141076812472616

Gómez-de-Regil, L., (2014). Causal Attribution and Illness Perception: A Cross-Sectional Study in Mexican Patients with Psychosis. The Scientific World Journal / 2014 / Volume 2014 |Article ID 969867 | https://doi.org/10.1155/2014/969867

Gordon, B. L., Shaw, H. S., Kroll, A. J., Daniel, B. R., (2010). Breast Cancer Recurrence and Advanced Disease: Comprehensive Expert Guidance. Duke University Press.

Gorman, L. M., (2018). Psychosocial Nursing Care Along the Cancer Continuum (Third Edition). Chapter 1 - Psychosocial Impact of Cancer on the Individual, Family, and Society. Published by Oncology Nursing Society

Goss, E., Lopez, A. M., Brown, C. L., Wollins, D. S., Brawley, O. W., Raghavan, D., (2009). American Society of Clinical Oncology policy statement: Disparities in cancer care. Journal of Clinical Oncology. 2009;27(17):2881–2885

Gotay, C. C., & Muraoka, M. Y., (1998). Quality of life in long-term survivors of adult-onset cancers. J Natl Cancer Inst 1998; / 90: / 65667.

Gough, I., (2004). Human well-being and social structures: Relating the universal and the local. Global Social Policy, 4, 289-311.



Grant, M. M., & Dean, G. E. (2003). Evolution of quality of life in oncology and oncology nursing. In C.R. King & P.S. Hinds (Eds.), Quality of life from nursing and patient perspectives (2nd ed., pp. 3–27). Sudbury, MA: Jones and Bartlett.

Grassi, L., Spiegel, D., & Riba, M. (2017). Advancing psychosocial care in cancer patients. F1000Research, 6, 2083. doi:10.12688/f1000research.11902.1

Grassi, L., & Watson, M., (2012). Psychosocial care in cancer: an overview of psychosocial programmes and national cancer plans of countries within the International Federation of Psycho-Oncology Societies. Psycho-OncologyPsycho-Oncology21: 1027–1033 (2012). Published online in Wiley Online Library (wileyonlinelibrary.com). DOI: 10.1002/pon.3154. https://www.ipos-

society.org/resources/Documents/Federation/PON3154_IPOSFedn_Editorial_2012_Nov.pdf

Gray, R. E., Fitch, M., Greenberg, M., Hampson, A., Doherty M., Labrecque, M., (1998). The information needs of well, longer-term survivors of breast cancer. Patient Education and Counseling, 1998;33:245–255

Grbich, C. (1999). Qualitative research in health: An introduction. London: Sage Publications

Grbich, C. (2007). Qualitative Data Analysis: An Introduction. London: SAGE Publications.

Greco, M. T., Roberto, A., Corli, O., Deandrea, S., Bandieri, E., Cavuto, S., Apolone, G., (2014). Quality of cancer pain management: an update of a systematic review of undertreatment of patients with cancer. J Clin Oncol. 2014 Dec 20;32(36):4149-54. doi: 10.1200/JCO.2014.56.0383. Epub 2014 Nov 17. PMID: 25403222.

Green, M., Anderson, K., Griffiths, K, Garvey, G., Cunningham, J., (2018): Understanding Indigenous Australians' experiences of cancer care: Stakeholders' views on what to measure and how to measure it. BMC Health Serv Res 18: 982, 2018

Greene, J. C., & Caracelli, V. J., (1997). Advances in mixed-method evaluation: The challenges and benefits of integrating diverse paradigms. In New directions for evaluation, Vol. 74, CA: San Francisco: Jossey-Bass



Greene, J. C., Caracelli, V. J., & Graham, W. F. (1989). Toward a conceptual framework for mixed-method evaluation designs. Educational Evaluation and Policy Analysis, 11(3), 255–274.

Greene, J. C. (2007). Mixed methods in social inquiry. San Francisco: John Wiley & Sons.

Greenfield, S., Blanco, D. M., Elashoff, R. M., Ganz, P. A., (1987). Patterns of care related to age of breast cancer patients. JAMA 257(20):2766–2770.

Gretchen, H., (2005). Addressing Patients' Psychosocial Needs & Concerns. Oncology Times: May 10, 2005 - Volume 27 - Issue 9 - p 26-28. doi: 10.1097/01.COT.0000289248.78474.66

Grol, R., (2001). Successes and failures in the implementation of evidence-based guidelines for clinical practice. Med Care. 2001;39(8): II46–II54.

Groopman, J. (2007). How doctors think. Boston, Massachusetts, USA: Houghton Mifflin Company, 2007.

Grunberg, S.M., Groshen, S., Steingass, S., & Zaretsky, S., (1996). Comparison of conditional quality-of-life terminology and visual analogue scale measurements. Quality of Life Research, 5, 65–72.

Gruendemann, B. J., (1965). The impact of surgery on body image in Nursing clinics of North America, 10: 635-43

Grunfeld, E., (2006). Looking beyond survival: How are we looking at survivorship. Journal of Clinical Oncology, 32 (24), 5166-5169.

Guadagnoli, E., Ward, P., (1998). Patient participation in decision-making. Social Science and Medicine 1998, 47:329-339.

Guba, E.G., & Lincoln. Y.S., (1989). What is This Constructivist Paradigm Anyway? in Fourth Generation Evaluation, London: Sage Publications, 79-90.



Guba E. G., & Lincoln Y. S. (1994). Competing paradigms in qualitative research. In N. Denzin & Y. Lincoln (Eds.), Handbook of Qualitative Research (p. 105-117). Thousand Oaks, CA: Sage.

Gudex, C., Dolan, P., Kind, P., & Williams, A., (1996). Health state valuations from the general public using the Visual Analogue Scale. Quality of Life Research, 5, 521–531. doi:10.1007/BF00439226

Guerrero, G. P., Márcia M. F. Z., Namie, O. S., & Maria, H. P., (2011). Relação Entre Espiritualidade e Câncer: Perspectiva do Paciente. Revista Brasileira de Enfermagem 64: 53–59.

Guest, G., Bunce, A., & Johnson L., (2006). How many interviews are enough? An experiment with data saturation and variability. Field methods. 2006;18(1):59–82.

Guest, G., MacQueen, K. M., & Namey, E. E., (2012). Applied Thematic Analysis. Chapter1|IntroductiontoAppliedThematicAnalysis.DOI: https://dx.doi.org/10.4135/9781483384436.n1

Gulluoglu, B. M., Cingi, A., Cakir, T., Gercek, A., Barlas, A., Eti, Z., (2006). Factors related topost-treatment chronic pain in breast cancer survivors: The interference of pain with life functions. Int J FertilWomens Med 2006;51(2):75–82.

Günüşen, N. P., Üstün, B., Serçekuş, P & Besen. D. B., (2018). Secondary traumatic stress experiences of nurses caring for cancer patients. International Journal of Nursing Practice Volume 25, Issue 1. https://doi.org/10.1111/ijn.12717

Guetterman, T. C., (2015). Descriptions of Sampling Practices Within Five Approaches to Qualitative Research in Education and the Health Sciences. Volume 16, No. 2, Art. 25 – May 2015. Forum: Qualitative social research. http://www.qualitative-research.net/index.php/fqs/article/view/2290/3825

Gustafson, D. H., Tayor, J. O., Thompson, S., Chesney, P., (1993). Assessing the needs of breast cancer patients and their families. Qual Manage Health Care 1993; 2:6–17.



Gustafson, D. H., (1991). Expanding on the role of patient as consumer. QRB Qual Rev Bull 1991; 17:324–5

Habib, F., Salman, S., Safwat, M., Shalaby, S., (2010). Awareness and knowledge of breast cancer among university students in Al Madina Al Munawara region. Middle East Journal of Cancer. Vol.1:159–66.

Hackett, J., Bekker, H., Bennett, M. I., Carder, P., Gallagher, J., Henry, C., Kite, S., Taylor, S., Velikova, G., & Ziegler, L. (2018). Developing a complex intervention to support timely engagement with palliative care for patients with advanced cancer in primary and secondary care in the UK: a study protocol. *BMJ open*, 8(5), e022835. https://doi.org/10.1136/bmjopen-2018-022835

Haddou, R. B., El Rhazi, K., Ouasmani, F., Nejjari, C., Bekkali, R., Montazeri, A., & Mesfioui, A. (2016). Quality of life in Arab women with breast cancer: a review of the literature. Health and Quality of Life Outcomes, 14, 64. http://doi.org/10.1186/s12955-016-0468-9

Haidinger, R. & Bauerfeind, I., (2019). Long-Term Side Effects of Adjuvant Therapy in Primary Breast Cancer Patients: Results of a Web-Based Survey. Breast Care 2019; 14:111-116. https://doi.org/10.1159/000497233

Hagag, N. G., (1999). when cancer strikes. Chapter 1, when cancer strikes: A tribute to the family caregiver. Nova Science Publisher.

Hagerty, R. G., Butow, P. N., Ellis, P. M., Dimitry, S., Tattersall, M. H., (2005). Communicating prognosis in cancer care: a systematic review of the literature. Ann Oncol. 2005; 16:1005-1053.

Hajian, S., Mehrabi, E., Simbar, M., & Houshyari, M. (2017). Coping Strategies and Experiences in Women with a Primary Breast Cancer Diagnosis. Asian Pacific journal of cancer prevention: APJCP, 18(1), 215–224. doi:10.22034/APJCP.2017.18.1.215

Hall, A., D'Este, C., Tzelepis, F., Sanson-Fisher, R., Lynagh, M., (2014). The Survivor Unmet Needs Survey (SUNS) for haematological cancer survivors: a cross-sectional study



assessing the relevance and psychometric properties. BMC Health Serv Res 14, 211 (2014). https://doi.org/10.1186/1472-6963-14-211

Halford, W. K., Scott, J. L., Smyth, J., Schmaling, K. B., Sher, T. G., (2001). Couples of coping with cancer, The Psychology of Couples and Illness: Theory, Research, and Practice, 2001Washington, DCAmerican Psychological Association (pg. 135-70)

Halligan, P., (2006). Caring for patients of Islamic denomination: Critical care nurses' experiences in Saudi Arabia. Journal of Clinical Nursing, December 2006, Vol.15(12), pp.1565-1573. DOI:10.1111/j.1365-2702.2005.01525.x

Halkett, G. K. B., Kristjanson, L. J., Lobb, E., Little, J., Shaw, T., Taylor, M.; Spry, N., (2011). Information needs and preferences of women as they proceed through radiotherapy for breast cancer. Oxford: Elsevier Ireland Ltd. Patient education and counseling, 2011, Vol.86 (3), p.396-404

Halpren, E. S., (1983). Auditing naturalistic inquiries: The develop-ment and application of a model (Unpublished doctoral dissertation). Indiana University, Bloomington

Hamamy, H. (2012). Consanguineous marriages: Preconception consultation in primary health care settings. Journal of Community Genetics, 3(3), 185–192. http://doi.org/10.1007/s12687-011-0072-y

Hammersley, M. (2013). What is Qualitative Research? London and New York: Bloomsburry

Hammoudeh, W.; Hogan, D.; Giacaman, R., (2017). From a Death Sentence to a Disrupted Life.Qual. Health Res.2017,27, 487–496.

Hanratty, B & Holland, P & Jacoby, A & Whitehead, M. (2007). Review article: Financial stress and strain associated with terminal cancer—A review of the evidence. Palliative medicine. 21. 595-607. 10.1177/0269216307082476.

Hardan-Khalil, K., & Mayo, A. M., (2015). Psychometric Properties of the Multidimensional Scale of Perceived Social Support. Clin Nurse Spec. 2015 Sep-Oct;29(5):258-61. doi: 10.1097/NUR.00000000000148.



Hardy, S. J., Krull, K. R., Wefel, J. S., Janelsins, M., (2018). Cognitive Changes in Cancer Survivors. Am Soc Clin Oncol Educ Book. 2018 May 23; 38:795-806. doi: 10.1200/EDBK_201179. PMID: 30231372.

Hanson, W. E., Creswell, J. W., Plano Clark, V. L., Petska, K. S., & Creswell, J. D., (2005). Creswell, J. D. Published in Journal of Counseling Psychology 52:2 (2005), pp. 224–235; doi 10.1037/0022-0167.52.2.224

Harandy, T. B., Ghofranipour, F., Montazeri, A., Anoosheh, M., Bazargan, M., Mohammadi, E. Ahmadi, F., & Niknami, S., (2009). Muslim Breast Cancer Survivor Spirituality: Coping Strategy or Health Seeking Behavior Hindrance? Taylor & Francis Group. Health Care for Women International, 15 December 2009, Vol.31(1), p.88-98. http://dx.doi.org.salford.idm.oclc.org/10.1080/07399330903104516

Harris, S. R., Schmitz, K. H., Campbell, K. L., McNeely, M. L. (2012). Clinical Practice Guidelines for Breast Cancer Rehabilitation: Syntheses of Guideline Recommendations and Qualitative Appraisals Cancer April pp.2312-2324

Harrison, D. J., Young, M. J., Price, A. M., Butow, N. P., Solomon, J. M., (2009). What are the unmet supportive care needs of people with cancer? A systematic review. Support Care Cancer (2009) 17:1117–1128. DOI 10.1007/s00520-009-0615-5

Harrop, J. P., Dean, J. A., Paskett, E. D., (2011). Cancer survivorship research: A review of the literature and summary of current NCI-designated cancer center projects Cancer Epidemiol Biomarkers Prev 20:2042–2047,2011

Hashim, M. J., Al-Shamsi, F. A., Al-Marzooqi, N. A., Al-Qasemi, S. S., Mokdad, A. H., & Khan, G. (2018). Burden of Breast Cancer in the Arab World: Findings from Global Burden of Disease, 2016. Journal of epidemiology and global health, 8(1-2), 54–58. https://doi.org/10.2991/j.jegh.2018.09.003

Haslam, A., Herrera-Perez, D., Gill, J., & Prasad, V. (2020). Patient Experience Captured by Quality-of-Life Measurement in Oncology Clinical Trials. JAMA network open, 3(3), e200363. https://doi.org/10.1001/jamanetworkopen.2020.0363



Hatamipour, K., Rassouli, M., Yaghmaie, F., Zendedel, K., & Majd, H., (2015). Spiritual Needs of Cancer Patients: A Qualitative Study. Indian Journal of Palliative Care, 2015, Vol.21(1), p.61. doi: 10.4103/0973-1075.150190

Hatefnia, E., Niknami, S., Bazargan, M., Mahmoodi, M., Lamyianm, M., Alavi, N. (2010). Correlates of mammography utilization among working Muslim Iranian women. Health Care Women Int, 31, 499-514. Doi: 10.1080/07399331003725507.

Hauffman, A., Alfonsson, S., Mattsson, S., Forslund, M., Bill-Axelson, A., Nygren, P., & Johansson, B. (2017). The development of a nurse-led internet-based learning and self-care program for cancer patients with symptoms of anxiety and depression—A part of U-CARE. Cancer Nursing, 40(5), E9–E16. doi:10.1097/NCC.0000000000000402

Haylock, P., & Curtiss, C., (2019). Cancer Survivorship: Interprofessional, Patient-Centered Approaches to the Seasons of Survival. Chapter 1 - Evidence-Based Survivorship Care: Current and Future Challenges in Survivorship Research. Published By: Oncology Nursing Society

Hegel, M. T., Moore, C. P., Collins, E. Dale., Kearing, S., Gillock, K. L., Riggs, R. L., Clay,
K. F., & Ahles, T. A., (2006). Distress, psychiatric syndromes, and impairment of function in women with newly diagnosed breast cancer. Cancer. Vol.107(12):2924-31. DOI: 10.1002/cncr.22335

Hejmadi, M., (2010). Introduction to cancer biology. Chapter1. Defining cancer. Pg 6

Helft, P. R., (2005). Necessary collusion: prognostic communication with advanced cancer patients. J Clin Oncol. 2005. pp. 3146–50. Available from http://www.ncbi.nlm.nih.gov/entrez/query.fcgi?cmd=Retrieve&db=PubMed&dopt=Citation &list_uids=15860876. DOI: 10.1200/JCO.2005.07.003 Journal of Clinical Oncology 23, no. 13 (May 01, 2005) 3146-3150.

Helgeson, V. S., (2005), Recent advances n psychosocial oncology, Journal of consulting and clinical psychology 73(2): 268-271

Henn, M., Weinstein, M. & Foard, N. (2006). A short introduction to social research. London: Sage.



Henkel, G., (2005). Addressing Patients' Psychosocial Needs & Concerns, Oncology Times: May 10th, 2005 - Volume 27 - Issue 9 - p 26-28 doi: 10.1097/01.COT.0000289248.78474.66

Henry, N. L., Azzouz, F., Desta, Z., et. al., (2012). Predictors of aromatase inhibitor discontinuation as a result of treatment-emergent symptoms in early-stage breast cancer. J Clin Oncol 30:936-942, 2012

Henson, H. K., (2012). Breast Cancer and Sexuality. Sexuality and Disability, 2002, Volume 20, Number 4, Page 261. DOIhttps://doi.org/10.1023/A:1021657413409

Herschbach, P., Book, K., Brands, T., Keller, M., Lindena, G., Neuwohner, K., et al. (2008). Psychological distress in cancer patients assessed with an expert rating scale. Br. J. Cancer 99, 37–43. doi: 10.1038/sj.bjc.6604420

Herzog, C., Handke, C., & Hitters, E., (2019). Analyzing Talk and Text II: Thematic Analysis. In book: The Palgrave Handbook of Methods for Media Policy Research (pp. https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3068081) Publisher: Palgrave Macmillan.

https://www.researchgate.net/publication/321197933_Analyzing_Talk_and_Text_II_Thematic_Analysis

Hesse-Biber, S. N., & Leavy, P. (Eds.). (2006). Emergent methods in social research Thousand Oaks, CA: Sage.

Hesse-Biber, S. N., & Crofts, C. (2008). User-centered perspective on qualitative data analysis software: Emergent technologies and future trends. In S. N Hesse-Biber & P. Leavy (Eds.), Handbook of emergent methods (pp. 655–674) New York: Guilford Press

Hesse-Biber, S. N., (2010). Mixed Methods Research: Merging Theory with Practice, Guilford Publications, 2010. Chapter1: Introduction to Mixed Methods Research. ProQuest Ebook Central, http://ebookcentral.proquest.com/lib/salford/detail.action?docID=471119. Created from salford on 2020-05-27 15:27:49.

Hesse-Biber, & Johnson, B. (2015). *The Oxford handbook of multimethod and mixed methods research inquiry.* Oxford University Press.


Hewitt, M., Rowland, J. H., Yancik, R., (2003). Cancer survivors in the United States: Age, health, and disability. The journals of gerontology. Series A, Biological sciences and medical sciences, January 2003, Vol.58(1), pp.82-91

Hewitt, M., Herdman, R., & Holland, J., (2004). Institute of Medicine (US) and National Research Council (US) National Cancer Policy Board; editors. Meeting Psychosocial Needs of Women with Breast Cancer. Chapter: 3 Psychosocial Needs of Women with Breast Cancer. https://www.nap.edu/read/10909/chapter/5. https://doi.org/10.17226/10909.

Hewitt, M., Greenfield, S., & Stovall, E., (2006). From Cancer Patient to Cancer Survivor: Lost in Transition. Washington, DC: The National Academies Press.

Higano, C. S., Zarowski, C., Wassersug, R., & Elliott, S. (2016). Sexual Health After Cancer Therapy. Journal of oncology practice, 12(4), 305–306. doi:10.1200/JOP.2016.011536

Hillen, M. A., Medendorp, N. M., & Smets, E. M. A., (2017). Patient-Driven Second Opinions in Oncology: A Systematic Review. The Oncologist, 12 Jun 2017, 22(10):1197-1211. DOI: 10.1634/theoncologist.2016-0429 PMID: 28606972 PMCID: PMC5634767

Hodgkinson, K., Butow, P., Hunt, G. E., Wyse, R., Hobbs, K. M., & Wain, G. (2007). Life after cancer: Couples' and partners' psychological adjustment and supportive care needs. Supportive Care in Cancer, 15, 405–415.

Hofsø, K., Miaskowski, C., Bjordal, K., Cooper, B. A., Rustøen, T., (2012). Previous chemotherapy influences the symptom experience and quality of life of women with breast cancer prior to radiation therapy. Cancer Nurs, 35 (2012), pp. 167-177

Holmes, C. A., & Warelow, P. J., (1997). Culture, needs and nursing: a critical theory approach. J Adv Nurs 25, 463–470.

Holland, J. C., (2002). History of psychosocial oncology: Overcoming attitudinal and conceptual barriers. Psychosomatic Medicine, 64, 206–221. doi:10.1097/00006842-200203000-00004



Holland, J. C., & Reznik, I., (2005). Pathways for psychosocial care of cancer survivors. Cancer 104, 2624–2637.

Holland, J. C., & Alici, Y., (2010). Management of distress in cancer patients. Journal of Supportive Oncology, 8, 4–12

Holland, J. C., & Zittoun, R., (2012). Psychosocial Aspects of Oncology. Quality of life assessment in cancer clinical trials. Pg 104.

Holland, J. C., Breitbary, W. S., Butow, P. N., Jacobsen, P. B., Loscalzo, M. J & McCorkle, R., (2015). Psycho Oncology. 3rd Edition. Chapter 20: Breast Cancer. Pg 135

Holloway, I., & Todres, L., (2003). The status of method: flexibility, consistency and coherence. Qualitative Research, 3(3), 345-357

Hopman, P., & Rijken, M., (2015). Illness perceptions of cancer patients: relationships with illness characteristics and coping. Psycho-Oncology: 2015, 24(1), 11-18. DOI: 10.1002/pon.3591

Hosseini, S. A., & Padhy, R. K. B., (2020). Body Image Distortion. [Updated 2020 Jul 2]. In: StatPearls [Internet]. Treasure Island (FL): Stat Pearls Publishing; 2021 Jan-. Available from: https://www.ncbi.nlm.nih.gov/books/NBK546582/

Houts, P. S., Yasko, J. M., Harvey, H. A., Kahn, S. B., Hartz, A. J., Herman, J. F., Schelzel, G. W., (1988). Unmet needs of persons with cancer in Pennsylvania during the period of terminal care. Cancer 1988; 62:627–34.

Houtzager, B. A., Grootenhuis, M. A., Last, B. F., (1999). Adjustment of siblings to childhood cancer: a literature review. Support Care Cancer. 1999 Sep;7(5):302-20. doi: 10.1007/s005200050268. PMID: 10483815.

Hove, G., Heje, H., Olesen, F., Vedsted, P., (2006). DanPEP - Danske Patienter Evaluerer Almen Praksis [In Danish]., Research Unit for General Practice: Aarhus University

Howell, D., & Olsen, K. (2011). Distress-the 6th vital sign. Current oncology (Toronto, Ont.), 18(5), 208–210. doi:10.3747/co.v18i5.790



Howell, D., Hack, T., Oliver, T. K., Chulak, T., Mayo, S., Aubin, M., Chasen, M., Tompson, M., (2011). Survivorship services for adult cancer populations: A pancanadian guideline. Current Oncology. 2011;18(6):265–281.

Hrishikesh, P., Kale, M. S., Norman, V. C., (2016). Self-reported financial burden of cancer care and its effect on physical and mental health-related quality of life among US cancer survivors. ACS journals, American Cancer Society. https://doi.org/10.1002/cncr.29808

Hsieh, L. Y., Chou, F. J., & Guo, S. E. (2018). Information needs of patients with lung cancer from diagnosis until first treatment follow-up. PloS one, 13(6), e0199515. doi: 10.1371/journal.pone.0199515

Huang, Y. L., Yates, P., & Prior, D., (2009). Accommodating the diverse cultural needs of cancer patients and their families in palliative care. Cancer Nurs. 2009 Jan-Feb;32(1): E12-21. doi: 10.1097/01.NCC.0000343370. 16894.b7. PMID: 19104194.

Hudson, M. M., Mertens, A. C., Yasui, Y., (2003). Health status of adult long-term survivors of childhood cancer: A report from the Childhood Cancer Survivor Study. Journal of the American Medical Association 290(12): 1583–1592.

Hughes, J. (1990). The philosophy of social research. (Second Edition) London: Longman.

Hui, D., Cruz, M. de C., Thorney, S., Parsons, H. A., Delgado-Guay, M., & Bruera, E., (2011). The Frequency and Correlates of Spiritual Distress Among Patients with Advanced Cancer Admitted to an Acute Palliative Care Unit. The American journal of Hospice & Palliative Care 28: 264–70.

Hughes, K.L., Sargeant, H. & Hawkes, A.L., (2011). Acceptability of the Distress Thermometer and Problem List to community-based telephone cancer helpline operators, and to cancer patients and carers. BMC Cancer 11, 46 (2011). https://doi.org/10.1186/1471-2407-11-46

Humpel, N., & Iversion, D. C., (2010). Sleep quality, fatigue and physical activity following a cancer diagnosis. European Journal of Cancer Care, November 2010, Vol.19(6), pp.761-768. DOI: 10.1111/j.1365-2354.2009.01126. x.



Hulbert-Williams, N. J., Pendrous, R., Hubert-Williams, L., & Swash, B., (2019). Recruiting cancer survivors into research studies using online methods: a secondary analysis from an international cancer survivorship cohort study. ecancer 2019, 13:990 https://doi.org/10.3332/ecancer.2019.990

Hulett, J. M., & Armer, J. M. (2016). A Systematic Review of Spiritually Based Interventions and Psychoneuroimmunological Outcomes in Breast Cancer Survivorship. Integrative Cancer Therapies, 15(4), 405–423. http://doi.org/10.1177/1534735416636222

Hulett, M. J., Armer, M. Jane., Leary, R. E., Stewart, R. B., Mcdaniel, R. R., Smith, R., K., Millspaugh, R. R., Millspaugh, R. Joshua., (2017). Religiousness, Spirituality, and Salivary Cortisol in Breast Cancer Survivorship: A Pilot Study. Cancer Nursing, 2017, DOI: 10.1097/NCC.00000000000471.

Husserl, E., (1931). Ideas: General introduction to pure phenomenology (D. Carr, Trans.). Evanston, IL: Northwestern University Press

Husson, O., Mols, F., van de Poll-Franse, L. V., (2011). The relation between information provision and health-related quality of life, anxiety and depression among cancer survivors: a systematic review. Ann. Oncol., 22 (2011), pp. 761-772

Hwang, J. J., Donnelly, T. T., Ewashen, C., McKiel, E., Raffin, S., & Kinch, J., (2015). Sociocultural Influences on Arab Women's Participation in Breast Cancer Screening in Qatar. Qualitative Health Research 1–13. sagepub.com/journalsPermissions.nav DOI: 10.1177/1049732315619373

Hwang, J. J., Donnelly, T. T., Ewashen, C., McKiel, E., Raffin, S., Kinch, J., (2017). Sociocultural Influences on Arab Women's Participation in Breast Cancer Screening in Qatar. Qual Health Res. 2017 Apr;27(5):714-726. doi: 10.1177/1049732315619373. Epub 2016 Jul 10. PMID: 26631675.

Hyland, M.E., & Sodergren, S. C., (1996). Development of a new type of global quality-oflife scale and comparison and preference for 12 global scales. Quality of Life Research, 5, 469– 480.



Iddrisu, M., Aziato, L. & Dedey, F., (2020). Psychological and physical effects of breast cancer diagnosis and treatment on young Ghanaian women: a qualitative study. BMC Psychiatry 20, 353 (2020). https://doi.org/10.1186/s12888-020-02760-4

Im, E. O., (2006). A situation-specific theory of Caucasian cancer patients' pain experience. ANS Adv Nurs Sci. 2006 Jul-Sep;29(3):232-44. doi: 10.1097/00012272-200607000-00006. PMID: 17139205.

Im, E. O., Chee, W., Guevara, E., Lim, H. J., Liu, Y., Shin, H., (2008). Gender and ethnic differences in cancer patients' needs for help: an Internet survey. International Journal of Nursing Studies, 2008, Vol.45(8), pp.1192-1204. DOI: 10.1016/j.ijnurstu.2007.09.006. Available at: http://www.sciencedirect.com/science/article/pii/S0020748907002362

International agency for Research on Cancer, (2020). 784-united-arab-emirates-fact-sheets.pdf (iarc.fr)

IOM (Institute of Medicine), (1993). Access to Health Care in America. *Washington D.C.: National Academy Press.*

IOM (**Institute of Medicine**), (1997). In: Approaching death: Improving care at the end of life. Field M and Cassel C, editor. Washington, D.C.; 1997.

IOM (Institute of Medicine), (1999). National Cancer Policy Board, National Research Council. Defining and Assessing Quality Cancer Care." Ensuring Quality Cancer Care. Washington, DC: The National Academies Press. doi: 10.17226/6467.

IOM (**Institute of Medicine**), (2001). Institute of Medicine, 2001. Patient-Centered Cancer Treatment Planning, Improving the Quality of Oncology Care. A National Coalition for Cancer Survivorship and Institute of Medicine Workshop. Chapter 2, Page 49. 2017 The Nationals Academies Press

IOM (Institute of Medicine), (2002). Institute of Medicine: Cancer and the environment. National Academy Press



IOM (Institute of Medicine), (2005). From Cancer Patient to Cancer Survivor: Lost in transition. Washington, DC: National Academy Press; 2005.

IOM (Institute of Medicine), 2006. From Cancer Patient to Cancer Survivor: Lost in Transition. The National Academies of science engineer Medicine (NAP). (2006). https://www.nap.edu/read/11468/chapter/2

IOM (Institute of Medicine), (2008). Institute of Medicine (US) Committee on Psychosocial Services to Cancer Patients/Families in a Community Setting; Adler NE, Page AEK, editors. Cancer Care for the Whole Patient: Meeting Psychosocial Health Needs. Washington (DC): National Academies Press (US); 2008. 1, The Psychosocial Needs of Cancer Patients. Available from: https://www.ncbi.nlm.nih.gov/books/NBK4011/

IOM (Institute of Medicine), (2009). Ensuring quality cancer care through the oncology workforce: Sustaining care in the 21st century: Workshop summary. Washington, DC: The National Academies Press; 2009

IOM (**Institute of Medicine**), (2011). Patient-centered cancer treatment planning: Improving the quality of oncology care: Workshop summary. Washington, DC: The National Academies Press; 2011

IOM, Institute of Medicine., (2011). Patient-Centered Cancer Treatment Planning. Improving the Quality of Oncology Care: Workshop Summary. Patient Empowerment (Pg7).

IOM (**Institute of Medicine**), (2013). Delivering High-Quality Cancer Care: Charting a New Course for a System in Crisis. Chapter 3, Patient-Centered Communication and Shared Decision Making. Committee on Improving the Quality of Cancer Care: Addressing the Challenges of an Aging Population; Board on Health Care Services; Institute of Medicine; Levit, L., Balogh, E., Nass, S., & Ganz, P. A., editors. Washington (DC): National Academies Press (US).

IOM and NRC (Institute of Medicine) and National Research Council, (2006). From cancer patients to cancer survivor: Lost in transition. Hewitt, M., Greenfield, and Stovall, E. The National Academies Press



IOM & NRC, (Institute of Medicine (US) and National Research Council (US) National Cancer Policy Board); (2004). Hewitt M, Herdman R, Holland J, editors. Meeting Psychosocial Needs of Women with Breast Cancer. Washington (DC): National Academies Press (US); 2004. 3, Psychosocial Needs of Women with Breast Cancer. Available from: https://www.ncbi.nlm.nih.gov/books/NBK215940

IQWiG, (2006). Depression: How effective are antidepressants? InformedHealth.org [Internet]. Cologne, Germany: Institute for Quality and Efficiency in Health Care. Available from: https://www.ncbi.nlm.nih.gov/books/NBK361016/ [Accessed 18th June 2021].

Islam, N., Patel, S., Brooks-Griffin, Q., Kemp, P., Raveis, V., Riley, L., Gummi, S., Queano, N. P., Ravenell, J., Cole, H., Known, S., (2017). Understanding Barriers and Facilitators to Breast and Cervical Cancer Screening among Muslim Women in New York City: Perspectives from Key Informants. SM J Community Med. 2017; 3(1): 1022.

Ivankova, N., Creswell, J., & Stick, S. (2006). Using mixed methods in sequential explanatory design: From theory to practice. Field Methods, 18(3), 3-20.

Iskandarsyah, A., De Klerk, C., Suardi, D. R., Soemitro, M. P., Sadarjoen, S. S., Passchier, J., (2013). Satisfaction with information and its association with illness perception and quality of life in Indonesian breast cancer patients. Support Care Cancer. 2013;21(11):2999–3007. doi:10.1007/s00520-013-1877-5

Iskandarsyah, A., de Klerk, C., Suardi, D. R., Soemitro, M. P., Sadarjoen, S. S., Passchier, J., (2013). The Distress Thermometer and its validity: a first psychometric study in Indonesian women with breast cancer. PLoS One. 2013;8(2): e56353. doi: 10.1371/journal.pone.0056353. Epub 2013 Feb 13. PMID: 23418561; PMCID: PMC3571987

Izci, F., İlgün, A. S., Fındıklı, E., & Özmen, V. (2016). Psychiatric Symptoms and Psychosocial Problems in Patients with Breast Cancer. The Journal of Breast Health, 12(3), 94–101. http://doi.org/10.5152/tjbh.2016.3041

Jacobsen, P. B., Rowland, J. H., Paskett, E. D., Leeuwen, F. V., Moskowitz, C., Wollins, S. K., Robison, L. L., (2016). Identification of Key Gaps in Cancer Survivorship Research:



Findings. From the American Society of Clinical Oncology Survey. DOI: 10.1200/JOP.2015.009258 Journal of Oncology Practice 12, no. 3 (March 01, 2016) 190-193.

Jacobsen, P.B., DeRosa, A.P., Henderson, T.O., Mayer, D.K., Moskowitz, C.S., Paskett, E.D., & Rowland, J.H. (2018). Systematic review of the impact of cancer survivorship care plans on health outcomes and health care delivery. Journal of Clinical Oncology, 36, 2088–2100. https://doi.org/10.1200/JCO.2018.77.7482

Jacobsen, P. B., & Andrykowski, M. A. (2015). Tertiary prevention in cancer care: Understanding and addressing the psychological dimensions of cancer during the active treatment period. American Psychologist, 70, 134-145. 10.1037/a0036513

Jacobs, L. A., & Shulman, L. N., (2017). "Follow-up Care of Cancer Survivors: Challenges and Solutions." The Lancet Oncology 18.1 (2017): E19-29. Web.

Jackson, J. L., Chamberlin, J., Kroenke, K., (2001). Predictors of patient satisfaction. Soc Sci Med. 2001;52(4):609–20.

Jardien-Babooa, S., Rooyen, D., Ricks, E., & Jordan, J., (2016). Perceptions of patientcentred care at public hospitals in Nelson Mandela Bay. Health SA Gesondheid. Volume 21, December 2016, Pages 397-405.https://doi.org/10.1016/j.hsag.2016.05.002. https://www.sciencedirect.com/science/article/pii/S1025984816300114#!

Jassim, G. A., & Whitford, D. L. (2014). Understanding the experiences and quality of life issues of Bahraini women with breast cancer. Social Science & Medicine, 107, 189-195.

Jayarajah, U., & Samarasekera, D. N. (2017). Psychological Adaptation to Alteration of Body Image among Stoma Patients: A Descriptive Study. Indian journal of psychological medicine, 39(1), 63–68. https://doi.org/10.4103/0253-7176.198944

Jennifer, A., Klimek. Yingling., (2018). "Negotiating Emotional Order: A Grounded Theory of Breast Cancer Survivors." The Grounded Theory Review 17.1 (2018): The Grounded Theory Review, 2018-12-01, Vol.17 (1). Web.



Jenkinson, C., Coulter, A., Bruster, S., (2002). The Picker Patient Experience Questionnaire: development and validation using data from in-patient surveys in five countries. Int J Qual Health Care. 2002, 14 (5): 353-358. 10.1093/intqhc/14.5.353.

Jenkins, V., Fallowfield, L., & Saul, L., (2001). Information needs of patients with cancer: results from a large study in UK cancer centres. Br. J. Cancer, 84 (2001), pp. 48-51.

Jensen, M. P., Chang, H., Lai, Y., Syrjala, K. L., Fann, J. R., & Gralow, J. R., (2010). Pain in Long-Term Breast Cancer Survivors: Frequency, Severity, and Impact, Pain Medicine, Volume 11, Issue 7, July 2010, Pages 1099–1106, https://doi.org/10.1111/j.1526-4637.2010.00880.x

Jerant, A. F., Friedrichs-Fitzwater, M. M., Moore, M., (2005). Patients' perceived barriers to active self-management of chronic conditions. Patient Education and Counseling. 2005;57(1):300–307. DOI: 10.1016/j.pec.2004.08.004

Jiang, C., Wang, H., Wang, Q., Luo, Y., Sidlow, R., Han, X., (2019). Prevalence of Chronic Pain and High-Impact Chronic Pain in Cancer Survivors in the United States. JAMA Oncol. 2019 Aug 1;5(8):1224-1226. doi: 10.1001/jamaoncol.2019.1439. PMID: 31219506; PMCID: PMC6587142.

Jiao, M., Nolte, L., Hall, A., Jefford & M., Piper, A., (2015). Needs assessment tools for posttreatment cancer survivors: literature review. Australian Cancer Survivorship Centre, Peter MacCallum Cancer Centre, Melbourne. https://www.petermac.org/sites/default/files/page/downloads/Needs_Assessment_Tools_for_Survivors_Review_-_January_2016.pdf

Jin, J., Sklar, G. E., Min Sen Oh, V., & Chuen Li, S. (2008). Factors affecting therapeutic compliance: A review from the patient's perspective. Therapeutics and clinical risk management, 4(1), 269–286. https://doi.org/10.2147/tcrm.s1458

Johnson, A. E., (2014). Health literacy, does it make a difference? Australian Journal of Advanced Nursing, 31(3), 39-45.



Johnson, J. L., Bottorff, J. L., Balneaves, L. G., Grewal, S., Bhagat, R., Hilton, B. A., & Clarke, H., (1999). South Asian womens' views on the causes of breast cancer: images and explanations. Patient Education and Counseling. 1999; 37(3):243-254.

Johnson, R. B., & Onwuegbuzie, A. J., (2004). Mixed Methods Research: A Research Paradigm Whose Time Has Come. Educational Researcher, 33 (7), p 14-26.

Johnson, R. B., Onwuegbuzie, A. J., & Turner. L. A., (2007). "Toward a Definition of Mixed Methods Research". Journal of Mixed Methods Research 2:112-33.

Jones, S., (2000). Medical aspects of expatriate health: health threats. Occup. Med. Vol. 50, No. 8, pp. 572-578, 2000. Uppincott Williams & Wilkins for SOM Printed in Great Britain.

Jordan, J. E., Briggs, A. M., Brand, C. A., & Osborne, R. H., (2008). Enhancing patient engagement in chronic disease self-management support initiatives in Australia: the need for an integrated approach. The Medical journal of Australia, 17 November 2008, Vol.189(10 Suppl), pp. S9-S13

Kai, J. (2003). Ethnicity, Health and Primary Care. Section 2: Developing effective healthcare.2.6: Effective cross-cultural communication, Oxford: Oxford University Press

Kaiser, K., Rauscher, G. H., Jacobs, E. A., Strenski, T. A., Ferrans, C. E., & Warnecke, R. B. (2011). The import of trust in regular providers to trust in cancer physicians among white, African American, and Hispanic breast cancer patients. Journal of general internal medicine, 26(1), 51–57. https://doi.org/10.1007/s11606-010-1489-4

Kagawa-Singer, M., & Kassim-Lakha, S., (2003). A strategy to reduce cross-cultural miscommunication and increase the likelihood of improving health outcomes. Acad Med. 2003 Jun; 78(6):577-87.

Kagawa-Singer, M., Dadia, A. V., Yu, M. C., & Surbone, A., (2010). CA: Cancer, Culture, and Health Disparities: Time to Chart a New Course? A Cancer journal for Clinicians. DOI: 10.3322/caac.20051. Available at http://cajournal.org and http://cacancerjournal.org



Kane, T., Hammad, S., Nazmul, I., Alwattary, N., Clark, J., Daher-Nashif, S., (2020). Dementia Caregiving in the Middle East and North Africa. Transcult. Psychiatry 2020, in press.

Kantsiper, M., McDonald, E. L., Geller, G., Shockney, L., Snyder, C., & Wolff, A. C. (2009). Transitioning to breast cancer survivorship: perspectives of patients, cancer specialists, and primary care providers. Journal of general internal medicine, 24 Suppl 2(Suppl 2), S459–S466. https://doi.org/10.1007/s11606-009-1000-2

Karunanithi, G., Sagar, R. P., Joy, A., & Vedasoundaram, P. (2018). Assessment of Psychological Distress and its Effect on Quality of Life and Social Functioning in Cancer Patients. Indian journal of palliative care, 24(1), 72–77. https://doi.org/10.4103/IJPC.IJPC_104_17

Kash, K. M., Mago, R., & Kunkel, E. J., (2005). Psychosocial oncology: supportive care for the cancer patient. Semin Oncol 32:211–218. doi: 10.1053/j.seminoncol.2004.11.011

Kaptein, A. A., Yamaoka, K., Snoei, L., van der Kloot, W. A., Inoue, K., Tabei, T., Kroep, J. R., Krol-Warmerdam, E., Ranke, G., Meirink, C., Does, A., Nortier, H., (2013). Illness perceptions and quality of life in Japanese and Dutch women with breast cancer. J Psychosoc Oncol. 2013;31(1):83–102. doi:10.1080/07347332.2012.741092

Katz, A., (2011). Surviving After Cancer: Living the New Normal. Chapter 11: A blueprint of Health. Pg, 171. Rowman & Littlefield Publishers, Inc

Katz, A., (2015). Meeting the need for psychosocial care in young adults with cancer. Pittsburgh, PA: Oncology Nursing Society.

Katz, S. J., Belkora, J., Elwyn, G., (2014). Shared decision making for treatment of cancer: challenges and opportunities. J Oncol Pract. 2014 May; 10(3):206-8.

Kaur, M., (2016). Application of Mixed Method Approach in Public Health Research. Indian journal of community medicine: official publication of Indian Association of Preventive & Social Medicine, 41(2), 93–97. doi:10.4103/0970-0218.173495



Kaushik, V., & Walsh, C. A., (2019). Pragmatism as a Research Paradigm and Its Implications for Social Work Research. Soc. Sci. 2019, 8, 255; doi:10.3390/socsci8090255

Kay, P., Bengoa, R., & Assal, J. P. (2006). A citizen's guide to chronic disease care. Les Treilles, France: WHO Collaborating Centre for Reference and Research for Long-Term Follow-Up of Chronic Diseases.

Kawar, L. N., (2013). Barriers to breast cancer screening participation among Jordanian and Palestinian American women. European Journal of Oncology Nursing, February 2013, Vol.17(1), pp.88-94. DOI: http://dx.doi.org/10.1016/j.ejon.2012.02.004 Available at: http://www.ejoncologynursing.com/article/S1462-3889(12)00025-7/fulltext

Keefe, D., Al Garni, A., Villalon, A., Hassan, A., Kwong, D., Li, P., Tamura, K., Kim, J.
H., Yusof, M. M.m Javed, M. Z., Kanesvaran, R., Hsieh, R-K., Aapro, M., (2016).
Challenges in supportive cancer care: perspectives from the Asia Pacific and Middle East.
Support Care Cancer 24, 4479–4481 (2016). https://doi.org/10.1007/s00520-016-3381-1

Kelle, U., (2004). Computer-assisted analysis of qualitative data. In U. Flick, E. von Kardorff & I. Steinke (Eds.), A companion to qualitative research (pp. 276-283). London: Sage.

Kendall, M., Boyd, K., Campbell, C., Cormie, P., Fife, S., Thomas, K., Weller, D., Murray, S. A., (2006). How do people with cancer wish to be cared for in primary care? Serial discussion groups of patients and carers. Fam Pract. 2006, 23 (6): 644-650. 10.1093/fampra/cml035.

Kenneth, D. M., & Miklos, S., (2017). Global Perspectives on Cancer: Incidence, Care, and Experience [2 volumes]: Chapter 14: Abu Dhabi and United Arab Emirates: An Overview of Breast Cancer pg 268.

Khater, W. A., & Alkwiese, M. J. (2013). Predictors of hope among patients with cancer in Jordan. Journal of Hospice & Palliative Nursing, 15(8), 471-478.

Khayat, O., (1997). Health: an Islamic perspective, In: The Right Path to Health: Health Education through Religion, Alexandria: World Health Organization publications.



Kim, D., Chang, S. J., & Lee, H. O., (2018). The Effects of a Culturally Tailored, Patient-Centered Psychosocial Intervention in South Korean Cancer Survivors. https://doi.org/10.1177/1054773818755836

Kimberly, D. M., Leticia, N., Mariotto, A. B., Rowland, J. H., Yabroff, K. R., Alfano, C. Jemal, A., Kramer, A. L., (2019). Cancer treatment and survivorship statistics, 2019. First published: 11 June 2019. https://doi.org/10.3322/caac.21565

Kimiafar, K., Sarbaz, M., Shahid S., S., Esmaeili, M., Javame, G. Z., (2016). Breast cancer patients' information needs and information-seeking behavior in a developing country. The Breast, August 2016, Vol.28, pp.156-160. DOI: 10.1016/j.breast.2016.05.011.

King, A., & Hoppe, R. B. (2013). "Best practice" for patient-centred communication: a narrative review. Journal of graduate medical education, 5(3), 385–393. doi:10.4300/JGME-D-13-00072.1

King, N. (2004). Using templates in the thematic analysis of text. In C. Cassell & G. Symon (Eds.), Essential guide to qualitative methods in organizational research (pp. 256–270). London, UK: Sage.

King, M., Smith, A., Gracey, M., (2009). Indigenous health part 2: The underlying causes of the health gap. Lancet 374: 76 - 85, 2009

Kingsley, C., (2010). Cultural and Socioeconomic Factors Affecting Cancer Screening, Early Detection and Care in the Latino Population.

Kirby, B., Edward, Copeland, V. Suzanne, K., Gradishar, W., (2017). The Breast E-Book: Comprehensive Management of Benign and Malignant Diseases. Section XVII: Survivorship, follow-up care and rehabilitation of the breast cancer patient. 84: Psychosocial consequences and lifestyle intervention. Fifth edition. Elsevier

Kissane, D. W., Bultz, B. D., Butow, P. N., Bylund, C. L., Noble, S. & Wilkinson, S., (2017). Communication in Oncology and Palliative Care. Section B: A core curriculum for communication skills training for oncology and Palliative care. Oxford university press



Kitayama, S., (1992). Some thoughts on the cognitive-psychodynamic self from a cultural perspective. Psychological Inquiry, 3(1), 41-44

Klimmek, R., Snow, D., Wenzel, J., (2010). Insurance-related and financial challenges reported by managed care enrollees with breast cancer. Clin J Oncol Nurs. 2010 Oct;14(5):598-606. doi: 10.1188/10.CJON.598-606. PMID: 20880817.

Kline, R. M., Arora, N. K., Bradley, C. J., Brauer, E. R., Graves, D. L., Lunsford, N. B.,
McCabe, M. S., Nasso, S. F., Nekhlyudov, L., Rowland, J. H., Schear, R. M., & Ganz, P.
A. (2018). Long-Term Survivorship Care After Cancer Treatment - Summary of a 2017
National Cancer Policy Forum Workshop. Journal of the National Cancer Institute, 110(12),
1300–1310. https://doi.org/10.1093/jnci/djy176

Knapp, S., Marziliano, A., Moyer, A., (2014). Identity threat and stigma in cancer patients. Health Psychol Open. 2014;1(1):2055102914552281.

Knobf, M. T., (2000). Symptom distress before, during and after adjuvant breast therapy. Dev Support Cancer Care. 2000; 4:13–17.

Knobf, M. T., Ferrucci, M. L., Cartmel, B., Jones, A. B., Stevens, D., Smithm M., Salner, A., & Cowley, S., Bergen, A., Young, K., & Kavanagh, A., (2000). A taxonomy of needs assessment, elicited from a multiple case study of community nursing education and practice. Journal of Advanced Nursing, 31(1): 126-134. Available at: http://www.ncbi.nlm.nih.gov/pubmed/10632801

Knobf, M. T., (2008). "Coming to grips" with chemotherapy-induced premature menopause. Health Care Women Int. 2008 Apr;29(4):384-99. doi: 10.1080/07399330701876562. PMID: 18389434.

Knobf, M, T., Ferrucci, L.M., Cartmel, B., Jones, B. A., Stevens, D., Smith, M., Salner, A., & Mowad, L., (2012). Needs assessment of cancer survivors in Connecticut. J Cancer Surviv 6, 1–10 (2012). https://doi.org/10.1007/s11764-011-0198-2



Knobf, M. T., (2013). Being prepared: essential to self-care and quality of life for the person with cancer. United States: Oncology Nursing Society. Clinical journal of oncology nursing, 2013-06, Vol.17 (3), p.255-261

Knox, M. K., Hales, S., Nissim, R., Jung, J., Lo, C., Zimmermann, C., Rodin, G., (2017). Lost and stranded: the experience of younger adults with advanced cancer. Support Care Cancer. 2017 Feb;25(2):399-407. doi: 10.1007/s00520-016-3415-8. Epub 2016 Sep 27. PMID: 27678379.

Koch, T., (1994). Establishing rigour in qualitative research: The decision trail. Journal of Advanced Nursing, 19, 976–986. doi:10.1111/j.1365-2648.1994.tb01177.x

Koenig, H. G., (2000). Religion, spirituality and medicine: application to clinical practice, JAMA, 2000, vol. 284 pg. 1708

Koh, K. B., (2013). Somatization and Psychosomatic Symptoms. Chapter 17: Cancer in Psychosomatic perspective. Pg 229. Springer New York.

Komajda, M., Lapuerta, P., Hermans, N., Gonzalez-Juanatey, J. R., van Veldhuisen, D. J., Erdmann, E., Tavazzi, L., Poole-Wilson, P., Le Pen, C., (2005). Adherence to guidelines is a predictor of outcome in chronic heart failure: the MAHLER survey. Eur Heart J. 2005;26(16):1653–1659. doi: 10.1093/eurheartj/ehi251

Konstantis, A., & Exiara, T. (2015). Breaking Bad News in Cancer Patients. Indian Journal of Palliative Care, 21(1), 35–38. http://doi.org/10.4103/0973-1075.150172

Kornblith, A. B., Herndon, J. E., Weiss, R. B., Zhang, C., Zuckerman, E. L., Rosenberg, S., Mertz, M., Payne, D., Jane, M. M., Holland, J. F., Wingate, P., Norton, L., & Holland, J. C., (2003). Long-term adjustment of survivors of early-stage breast carcinoma, 20 years after adjuvant chemotherapy. Cancer. 2003 Aug 15;98(4):679-89. doi: 10.1002/cncr.11531. PMID: 12910510.

Krist, A. H., Tong, S. T., Aycock, R. A., & Longo, D. R. (2017). Engaging Patients in Decision-Making and Behavior Change to Promote Prevention. Studies in health technology and informatics, 240, 284–302.



Kroll, T., Neri, M., (2009). Designs for mixed methods research © Andrew, Sharon; Halcomb, Elizabeth J., Jan 02, 2009, Mixed Methods Research for Nursing and the Health Sciences Wiley-Blackwell, Hoboken, ISBN: 9781444316506.
16_Mixed_Methods_Kroll_Neri_2009b.pdf (bwgriffin.com) Oaks, CA: Sage

Kugbey, N., Asante, K. O., Meyer-Weitz, A., (2020). Illness perception and coping among women living with breast cancer in Ghana: an exploratory qualitative study. . BMJ Open 2020;10: e033019. doi:10.1136/bmjopen-2019-033019

Kumar, S., Shaikh, A. J., Khalid, S., & Masood, N., (2010). "Influence of patient's perceptions, beliefs and knowledge about cancer on treatment decision making in Pakistan," Asian Pacific Journal of Cancer Prevention, vol. 11, no. 1, pp. 251–255, 2010.

Kuruppu, D., Wijeyaratne, C., Gunawardane, N., & Amarasinghe, I., (2020). "Assessment of Information Needs of Patients with Breast Cancer: A Hospital-Based Study in Sri Lanka." Asia-Pacific Journal of Oncology Nursing 7.4 (2020): 375-81. Web.

Kus, T., Aktas, G., Ekici, H., Elboga, G., Djamgoz, S., (2017). Illness perception is a strong parameter on anxiety and depression scores in early-stage breast cancer survivors: a single-center cross-sectional study of Turkish patients. Support Care Cancer. 2017 Nov; 25(11):3347-3355.

Kwon, J. H., (2014). Overcoming Barriers in Cancer Pain Management. J. Clin. Oncol. 2014, 32, 1727–1733.

Lam, W., W.T., Kwong, A., Suen, D. et al. (2018). Factors predicting patient satisfaction in women with advanced breast cancer: a prospective study. BMC Cancer 18, 162 (2018). https://doi.org/10.1186/s12885-018-4085-3

Lam, T. W. W., Au, Y. H. A., Wong, F. H. J., Lehmna, C., Koch, U., Fielding, R., Mehnert, A., (2011). Unmet supportive care needs: a cross-cultural comparison between Hong Kong Chinese and German Caucasian women with breast cancer. Breast Cancer Research and Treatment, Volume 130, Issue 2, pp 531-541



Lamkin, D. M., & Slavich, G. M. (2014). Psychosocial factors and cancer. In H. L. Miller (Ed.), Encyclopedia of theory in psychology. Thousand Oaks, CA: Sage Publications. Available at: www.uclastresslab.org

Lance Armstrong Foundation, (2004). Centers for Disease Control and Prevention, Department of Health and Human Services, & Lance Armstrong Foundation. A National Action Plan for Cancer Survivorship: Advancing Public Health Strategies, 2004.p. 3-4.

Landry, S., Chasles, G., Pointreau, Y., Bourgeois, H., Boyas, S., (2018). Influence of an Adapted Physical Activity Program on Self-Esteem and Quality of Life of Breast Cancer Patients after Mastectomy. Oncology. 2018;95(3):188-191. doi: 10.1159/000489265. Epub 2018 May 30. PMID: 29847828.

Landier, W., Dai, C., Sparks, J., Anthony, K. R., Barrett, J.S., Hageman, L., Francisco, L., Rocque, G. B., Stringer-Reasor, E. M., Nabell, L., & Bhatia, S., (2020). Financial toxicity among breast cancer survivors with health insurance. Journal of Clinical Oncology 2020 38:15_suppl, 12073-12073

Lansky, S. B., List, M. A., Herrmann, C. A., Ets-Hokin, E. G., DasGupta, T. K., Wilbanks, G. D., Hendrickson, F. R., (1985). Absence of major depressive disorder in female cancer patients. Journal of Clinical Oncology 3(11):1553–1560

Lange, M. & Joly, F., (2017). How to Identify and Manage Cognitive Dysfunction After Breast Cancer Treatment. Journal of Oncology practice. https://doi.org/10.1200/JOP.2017.026286

Laryionava, K., Pfeil, T. A., Dietrich, M., Reiter-Theil, S., Hiddemann, W., & Winkler, E. C. (2018). The second patient? Family members of cancer patients and their role in end-of-life decision making. BMC palliative care, 17(1), 29. doi:10.1186/s12904-018-0288-2

Laryionava, K., Hartmann, M., Sklenarova, H., Heussner, P., Villalobos, M., Jäger, D., Winkler, E., C., (2019). Cancer Patients' Preferences for either Quality of Life or a Longer Life Determine Their Willingness to Talk about Forgoing Cancer-Specific Treatment. Oncol Res Treat 2019; 42:458-465. doi: 10.1159/000501129



Lash, T. L., & Silliman, R. A., (2000). Patient characteristics and treatments associated with a decline in upper-body function following breast cancer therapy. Journal of Clinical Epidemiology 53(6):615–622.

Lateef, F., (2011). Patient expectations and the paradigm shift of care in emergency medicine. *Journal of emergencies, trauma, and shock*, 4(2), 163–167. https://doi.org/10.4103/0974-2700.82199

Lawler, M., Banks, I., Law, K., et. al., (2016). The European Cancer Patient's Bill of Rights, update and implementation 2016. ESMO Open 2016;1: e000127. doi:10.1136/ esmoopen-2016-000127

Lawless, G. D., (2009). The working patient with cancer: implications for payers and employers. American health & drug benefits, 2(4), 168–173.

Laxmi, S., & Khan, J. A. (2013). Does the cancer patient want to know? Results from a study in an Indian tertiary cancer center. South Asian journal of cancer, 2(2), 57–61. https://doi.org/10.4103/2278-330X.110487

Lavrakas. (2008). Encyclopedia of Survey Research Methods. In *Encyclopedia of Survey Research Methods*. SAGE Publications. https://doi.org/10.4135/9781412963947

Lazenby, M., & Khatib, J., (2012). Associations among patient characteristics, health-related quality of life, and spiritual well-being among Arab Muslim cancer patients. Journal of Palliative Medicine, 15(12), 1321-1324.

Leadbeater, M., (2004). Evaluating use of a support group in secondary breast cancer care. Nursing Times 2004; 100:34-6.

Leavitt, R. L., (2002). Developing Cultural Competence in a Multicultural World: part 1. PT magazine; 10(12); 36-48. Available at: http://www.udel.edu/PT/current/PHYT600/2012/Lecture4Handouts/CES_25_CulturalCompe tence_012003%5B1%5D.pdf.



Lebel, S., & Devins, G. M., (2008). Stigma in cancer patients whose behavior may have contributed to their disease. Future Oncol. 2008;4(5):717–33.

Lederer, K., (1980). Needs methodology: The environmental vcase. In K. Lederer (Ed). Human needs: A contribution to the current debate (pp 259-278). Cambridge, MA: Oeleschlager, Gunn & Hain

Lee, Y. H., (2019). Spiritual Care for Cancer Patients. Asia Pac J Oncol Nurs 2019; Vol. 6: Page: 101-3. Available from: http://www.apjon.org/text.asp?2019/6/2/101/251292

Lee, Y., Baek, J. M., Jeon, Y. W., & Im, E. O. (2019). Illness perception and sense of wellbeing in breast cancer patients. Patient preference and adherence, 13, 1557–1567. https://doi.org/10.2147/PPA.S225561

Leech, N. L., & Onwuegbuzie, A. J., (2009). "A Typology of Mixed Methods Research Designs." Quality & Quantity: International Journal of Methodology 43: 265-75.

Leech, N. L., Dellinger, A. B., Brannigan, K. B., & Tanaka, H., (2009). "Evaluating Mixed Research Studies: A Mixed Methods Approach." Journal of Mixed Method Research4 (1): 17-31

Legese, B., Addissie, A., Gizaw, M., Tigneh, W., & Yilma, T., (2021). "Information Needs of Breast Cancer Patients Attending Care at Tikur Anbessa Specialized Hospital: A Descriptive Study." Cancer Management and Research 13 (2021): 277-86. Web

Lehmann, J. F., Delisa, J. A., Warren, C G., et. al. (1978). Relationship of race to functional status among breast cancer patients after surgery. Journal of surgical oncology 11:3333-9

Lehto, U. S., Ojanen, M., Kellokumpu-Lehtinen, P., (2005). Predictors of quality of life in newly diagnosed melanomaand breast cancer patients. Ann. Oncol.2005, 16, 805–816

Lenert, L., 2010. Transforming healthcare through patient empowerment. Stud Health Technol Inform. 2010; 153:159-75.

Leser, C., Tan, Y. Y., Singer, C. et al. (2021). Patient satisfaction after breast cancer surgery. Wien Klin Wochenschr 133, 6–13. https://doi.org/10.1007/s00508-020-01730-w



LeSeure, P., & Chongkham-Ang, S. (2015). The Experience of Caregivers Living with Cancer Patients: A Systematic Review and Meta-Synthesis. Journal of personalized medicine, 5(4), 406–439. doi:10.3390/jpm5040406

Leventhal, H., Meyer, D., Nerenz, D., (1980). The common-sense representation of illness danger. In: Rachman S, editor. Contributions to medical psychology. New York: Pergamon Press; 1980. p. 17–30.

Leventhal, H., Nerenz, D., Steele, D., (1984). Illness representations and coping with health threats. In: Baum A, Taylor S, Singer J, editors. Handbook of psychology and health. Vol. 4. Hillsdale (NJ): Erlbaum; 1984. p. 219–52

Levit, L., Balogh, E., Nas, S., et al., (2013). Committee on Improving the Quality of Cancer Care: Addressing the Challenges of an Aging Population; Board on Health Care Services; Institute of Medicine; Delivering High-Quality Cancer Care: Charting a New Course for a System in Crisis. Washington (DC): National Academies Press (US); 2013 Dec 27. 3, Patient-Centered Communication and Shared Decision Making. Available from: https://www.ncbi.nlm.nih.gov/books/NBK202146/

Levesque, J. V., Girgis, A., Koczwara, B., Kwok, C., Singh-Carlson, S., & Lambert, S., (2015). Integrative review of the supportive care needs of asian and caucasian women with breast cancer. Current Breast Cancer Reports. 2015; 7:127–42.

Levy, D. R., (1985). White doctors and black patients: influence of race on the doctor-patient relationship. Pediatrics.1985;75:639-643.

Lewin, S., Glenton, C., & Oxman, A. D., (2009). "Use of Qualitative Methods Alongside Randomised Controlled Trials of Complex Healthcare Interventions: Methodological Study." British Medical Journal 339: b3496.

Lewis, R. A., Neal, R. D., Hendry, M., France, B., Williams, N. H., Russell, D., Hughes, D. A., Russell, I., Stuart, N. S., Weller, D., Wilkinson, C., (2009). Patients' and healthcare professionals' views of cancer follow-up: systematic review. Br J Gen Pract. 2009 Jul; 59(564): e248-59.



Lewis, P. E., Sheng, M., Rhodes, M. M., Jackson, K. E., & Schover, L. R., (2012). Psychosocial concerns of young African American breast cancer survivors. Journal of psychosocial oncology, 30(2), 168–184. https://doi.org/10.1080/07347332.2011.651259

Leydon, G. M., Boulton, M., Moynihan, C., Jones, A., Mosman, J., Boudioni, M. & McPherson, K. (2000). Cancer patients' information needs and information seeking behaviour: in depth interview study. BMJ, 320:909.

Li, P. W. C; So, W, K. W; Fong, D. Y. T.; Lui, L. Y. Y.; Lo, J. C. K.; Lau, S. F., (2011). The Information Needs of Breast Cancer Patients in Hong Kong and Their Levels of Satisfaction with the Provision of Information. Cancer nursing, 2011-01, Vol.34 (1), p.49-57 United States: Lippincott Williams & Wilkins, Inc. doi: 10.1097/NCC.0b013e3181ef77a0.

Li, W. W., Lam, W. W., Shun, S-C., Lai, Y-H., Law, W-L., Poon, J., Fielding, R., (2013). Psychometric Assessment of the Chinese Version of the Supportive Care Needs Survey Short-Form (SCNS-SF34-C) among Hong Kong and Taiwanese Chinese Colorectal Cancer Patients. PLoS One. 2013 Oct 11;8(10): e75755. doi: 10.1371/journal.pone.0075755. eCollection 2013. https://doi.org/10.1371/journal.pone.0075755

Liao, M. N., Chen, S. C., Lin, Y. C., et al (2012). Changes and predictors of unmet supportive care

Liao MN, Chen SC, Lin YC, Chen MF, Wang CH, Jane SW. 2014. Education and psychological support meet the supportive care needs of Taiwanese women three months after surgery for newly diagnosed breast cancer: a non-randomised quasi-experimental study. Int J Nurs Stud. 2014 Mar; 51(3):390-9.

Liao, M. N., Chen, P. L., Chen, M. F., Chen, S. C., (2010). Effect of supportive care on the anxiety of women with suspected breast cancer. J Adv Nurs. 2010 Jan; 66(1):49-59.

Lim, B. T., Butow, P., Mills, J., Miller, A., & Goldstein, D. (2017). Information needs of the Chinese community affected by cancer: A systematic review. Psycho-Oncology, 26(10), 1-11. doi:10.1002/pon.4347

Lincoln, Y., Guba, E. G., (1985). Naturalistic inquiry. Newbury Park, CA: Sage.



Linschoten, M., Teske, A. J., Cramer, M. J., van der Wall, E., & Asselbergs, F. W. (2018). Chemotherapyrelated cardiac dysfunction: A systematic review of genetic variants modulating individual risk. Circulation: Genomic and Precision Medicine, 11, e001753. Retrieved from https://www.ahajournals.org/doi/10.1161/CIRCGEN.117.001753

Lipscomb, J., Gotay, C. C., & Snyder, C., (2005). Outcomes Assessment in Cancer. Chapter 15: Need assessment in Cancer. G 305. Cambridge University Press

Livestrong Foundation., (2004), Centers for Disease Control and Prevention, Department of Health and Human Services, & Lance Armstrong Foundation. A National Action Plan for Cancer Survivorship: Advancing Public Health Strategies, 2004.p. 3-4.

Locke, D. E., Decker, P. A., Sloan, J. A., Brown, P. D., Malec, J. F., Clark, M. M., Rummans, T. A., Ballman, K. V., Schaefer, P. L., & Buckner, J. C., (2007). Validation of single-item linear analog scale assessment of quality of life in neuro-oncology patients. Journal of pain and symptom management, 34(6), 628–638. https://doi.org/10.1016/j.jpainsymman.2007.01.016

Lofland, J., & Lofland, L. H., (1999). Data logging in observation: Fieldnotes. In A. Bryman & R. G. Burgess (Eds.), Qualitative research (Vol. 3). London: Sage

Loney, T., Aw, T., Handysides, D. G., Ali, R., Blair, I., Grivna, M., Shah, S. M., Sheek-Hussein, M., El-Sadig, M., Sharif, A. A., & El-Obaid, Y., (2013)."An Analysis of the Health Status of the United Arab Emirates: The 'Big 4' Public Health Issues." Global Health Action 6.1 (2013): 20100. Web.

Lorence, D. P., Greenberg, L, (2006). The Zeitgeist of Online Health Search. Journal of General Internal Medicine 21(2) 134–9.

Lorgelly, P. K., & Neri, M., (2018). "Survivorship Burden for Individuals, Households and Society: Estimates and Methodology." Journal of Cancer Policy 15 (2018): 113-17. Web.

Lopez-Class, M., Perret-Gentil, M., Kreling, B., Caicedo, L., Mandelblatt, J., & Graves, K. D. (2011). Quality of life among immigrant Latina breast cancer survivors: realities of culture and enhancing cancer care. Journal of cancer education: the official journal of the



American Association for Cancer Education, 26(4), 724–733. https://doi.org/10.1007/s13187-011-0249-4

Luoma, M. L., & Hakamies-Blomqvist, L., (2004). The meaning of quality of life in patients being treated for advanced breast cancer: a qualitative study. Psych-oncology, 13, 729-739.

Lovering, S. (2012). The crescent of care: A nursing model to guide the care of Muslim American patients. Diversity and Equality in Health and Care, 9, 171.

Lovelace, D. L., McDaniel, L. R., Golden, D., (2019). Long-Term Effects of Breast Cancer Surgery, Treatment, and Survivor Care. J Midwifery Womens Health. 2019 Nov;64(6):713-724. doi: 10.1111/jmwh.13012. Epub 2019 Jul 19. PMID: 31322834

Lowery, A. E., Greenberg, M. A., Foster, S. L., Clark, K., Casden, D. R., Loscalzo, M, et al., (2012). Validation of a needs-based biopsychosocial distress instrument for cancer patients. Psychooncology. 2012; 21(10): 1099–106. pmid:21830256

Lu, H., Xie, J., Gerido, L. H., Cheng, Y., Chen, Y., Sun, L., (2020). Information Needs of Breast Cancer Patients: Theory-Generating Meta-Synthesis. J Med Internet Res. 2020 Jul 28;22(7): e17907. doi: 10.2196/17907. PMID: 32720899; PMCID: PMC7420822.

Luker, K. A., Beaver, K., Leinster, S. J., Owens, R. G., Degner, L. F., Sloan, J. A., (1995). The information needs of women newly diagnosed with breast cancer. Journal of Advanced Nursing, July 1995, Vol.22(1), pp.134-141.

Lwin, Z., Broom, A., Cosman, R., Livingstone, A., Sawkins, K., Good, P., Kirby, E., Koh, E. S., Hovey, E., & COGNO CALD working group (2014). Culturally and linguistically diverse patient participation in glioma research. Neuro-oncology practice, 1(3), 101–105. https://doi.org/10.1093/nop/npu009

Ma, B. B. Y., Hui, E. P., Mok, T. S. K., (2010). Population-based differences in treatment outcome following anticancer drug therapies. Lancet Oncology 11: 75–84.



Ma, C., Yan, J., Wu, Y., Huang, W., (2018). Illness perceptions of Chinese women with breast cancer and relationships with socio-demographic and clinical characteristics. Int J Nurs Pract. 2018 Oct; 24(5):e12677.

Macdowell, W., Bonell, C., & Davies, M., (2006). Health Promotion Practice. Open university Press, Berkshire.

Machado-Vieira, R., Baumann, J., Wheeler-Castillo, C., Latov, D., Henter, I. D., Salvadore, G., & Zarate, C. A. (2010). The Timing of Antidepressant Effects: A Comparison of Diverse Pharmacological and Somatic Treatments. Pharmaceuticals (Basel, Switzerland), 3(1), 19–41. https://doi.org/10.3390/ph3010019

Maguire, P., (1999). Improving communication with cancer patients. European Journal of Cancer. 1999; 35:1415–1422. doi: 10.1016/S0959-8049(99)00178-1.

Mack, L., (2010). The philosophical underpinnings of educational research. Polyglossia, 19. Retrieved

http://en.apu.ac.jp/rcaps/uploads/fckeditor/publications/polyglossia/Polyglossia_V19_Lindsa y.pdf

Madey, D. L. (1982). Some benefits of integrating qualitative and quantitative methods in program evaluation, with some illustrations. Educational Evaluation and Policy Analysis, 4, 223-236.

Mahapatra, S., Nayak, S., & Pati, S. (2016). Quality of care in cancer: An exploration of patient perspectives. Journal of family medicine and primary care, 5(2), 338–342. doi:10.4103/2249-4863.192349

Mahlich, J., Schaede, U. & Sruamsiri, R., (2019). Shared Decision-Making and Patient Satisfaction in Japanese Rheumatoid Arthritis Patients: A New "Preference Fit" Framework for Treatment Assessment. Rheumatol Ther 6, 269–283 (2019). https://doi.org/10.1007/s40744-019-0156-4

Majumdar, A., (2019). Qualitative Techniques for Workplace Data Analysis. Thematic Analysis in Qualitative Research. DOI: 10.4018/978-1-5225-5366-3.ch009



Makabe, R. A., (1999). Descriptive study of social support among women with breast cancer: Focused on their preoperative states, J Jpn Soc Cancer Nurs, 1999, vol. 13 (pg. 43-52) (in Japanese with English abstract)

Mallmann, C. A., & Marcus, S. (1980). Logical clarification in the study of needs. In K. Lederer (Ed.) Human needs: A contribution to the current debate (pp. 163-185). Cambridge, MA: Oelgeschlager, Gunn & Hain.

Maloney, E. K., D'Agostino, T. A., Heerdt, A., Dickler, M., Li, Y., Ostroff, J. S., & Bylund, C. L. (2015). Sources and types of online information that breast cancer patients read and discuss with their doctors. Palliative & supportive care, 13(2), 107–114. https://doi.org/10.1017/S1478951513000862

Mancini, J., Noguès, C., Adenis, C., Berthet, P., Bonadona, V., Chompret, A., Coupier, I., Eisinger, F., Fricker, J. P., Gauthier-Villars, M., Lasset, C., Lortholary, A., N'Guyen, T. D., Vennin, P., Sobol, H., Stoppa-Lyonnet, D., & Julian-Reynier, C., (2006). Patients' characteristics and rate of Internet use to obtain cancer information. J Public Health (Oxf). 2006 Sep; 28(3):235-7.

Manning, A. T., & Sacchini, V. S. (2016). Conservative mastectomies for breast cancer and risk-reducing surgery: The Memorial Sloan Kettering Cancer Center experience. Gland Surgery, 5(1), 55–62. http://doi.org/10.3978/j.issn.2227-684X.2015.10.02

Manning, K., (1997). Authenticity in constructivist inquiry: methodological considerations without prescription. Qual Inquiry. 1997; 3:93-115.

Marlow, B., Cartmill, T., Cieplucha, H., & Lowrie, S., (2003). An interactive process model of psychosocial supportneeds for women living with breast cancer.Psycho-Oncology, 2003;12:319–330

Marmot, M., & Wilkinson, R. G., (2005). Social support and social cohesion. DOI:10.1093/acprof:oso/9780198565895.003.08

Maarraoui, A. M., Utamachandani, A., Mattamona, A. I., Lazaro, C. A., Mathews, L. S., Uttamchandani, V. S., Maarraoui, Y., Abdulatif, H., & Attia, N. M., (2006). Breast cancer



in United Arab Emirates: Younger patients and more aggressive histology. Journal of Clinical Oncology 24(18_suppl):10612-10612. DOI: 10.1200/jco.2006.24.18_suppl.10612

Marshall, C. A., Larkey, L. K., Curran, M. A., Weihs, K. L., Badger, T. A., Armin, J., & García, F. (2011). Considerations of culture and social class for families facing cancer: the need for a new model for health promotion and psychosocial intervention. Families, systems & health: the journal of collaborative family healthcare, 29(2), 81–94. https://doi.org/10.1037/a0023975

Martins, H., & Caldeira, S., (2018). Spiritual Distress in Cancer Patients: A Synthesis of Qualitative Studies. Religions 2018, 9, 285; doi:10.3390/rel9100285.

Martsolf, D. S., (1997) Cultural aspects of spirituality in cancer care. Semin Oncol Nurs. 1997 Nov; 13(4):231-6

Masadeh, M, A., (2012). Linking Philosophy, Methodology, and Methods: Toward Mixed Model Design in the Hospitality Industry. European Journal of Social Sciences ISSN 1450-2267 Vol.28 No.1 (2012), pp. 128-137 © EuroJournals Publishing, Inc. 2012

Maslow, A. H., (1943). A theory of human motivation. Psychological Review, 50(4), 370-96

Maslow, A. H., (1948). Higher and lower needs. J Psychol.; 25:433–436.

Maslow, A. H., (1968). Toward a psychology of being. Princeton: D. Van Nostrand Company.

Maslow, A. H., (1987). Motivation and personality (3rd ed.). Delhi, India: Pearson Education.

Maslow, A. H., (1998). Toward a psychology of being (3rd ed.) Mississauga, ON: John Wiley & Sons Canada Ltd.

Maslow, A. H., (2012). A theory of human motivation (Kindle edition). Start Publishiing LLC.

Masterson, B. J., (1979). Manual of Gynaecologic Surgery. Chapter 28. Psychosocial aspects pf gynaecologic surgery. Sringer-VerlagNew York Inc.



Matthews, B. A., Baker, F., & Spillers, R. L., (2004). Oncology professionals and patient requests for cancer support services. Supportive care in cancer 12 (10): 731-738

Matsuyama, R. K., Wilson-Genderson, M., Kuhn, L., Moghanaki, D., Vachhani, H. & Paasche-Orlow, M. (2011). Education level, not health literacy, associated with information needs for patients with cancer. Patient Education and Counseling, 85(3), 229-36.

Matsuyama, R.K., Kunn, L.A., Molisani, A. & Wilson-Genderson, M.C. (2013). Cancer patients' information needs the first nine months after diagnosis. Patient Education and Counseling, 90, 96-102.

Mataoui, F. Z., & Sheldon, L. K., (2016). Providing Culturally Appropriate Care to American Muslims with Cancer. Clinical Journal of Oncology Nursing, 2016, Vol.20(1), p.11(2). DOI: 10.1188/16.CJON.11-12. https://cjon.ons.org/cjon/20/1/providing-culturally-appropriatecare-american-muslims-cancer

Maskarinec, G., Sen, C., Koga, K., & Conroy, S. M. (2011). Ethnic Differences in Breast Cancer Survival: Status and Determinants. Women's Health (London, England), 7(6), 677–687. http://doi.org/10.2217/whe.11.67

Maura, S., Heddon, M. A., & Griffis, M., (2010). Piloting a needs assessment to guidedevelopment of survivorship program for a community cancer centre. Oncology Nursingforum.Vol.37,No.4.,July2010.DOI:http://dx.doi.org.salford.idm.oclc.org/10.1188/10.ONF.501-508

Maxwell, J.A. (1990) 'Response to "Campbell's Retrospective and a Constructionist's Perspective", Harvard Educational Review 60.

Maxwell, R., & Aldredge-Clanton, J., (1994). Survivor Gulit in Cancer Patients: A pastoral perspective, "Journal of Pastoral Care 48", no. 1 (spring 1994): 25-31.

Maxwell, J. A., & Mittapalli, K., (2010). Realism as a stance formixed methods research. In: Tashakkori A and Teddlie C(eds)Handbook of Mixed Methods in Social and BehaviouralResearch, 2nd edn. Thousand Oaks, CA: SAGE,pp. 145–167.



Mayan, M. J., Morse, J. M., Eldershaw, L. P., (2006). Developing the concept of self-reformulation, International Journal of Qualitative Studies on Health and Well-being, 1:1, 20-26, DOI: 10.1080/17452620500534587

Mays, N., & Pope, C., (1995). Rigour and qualitative research. BMJ, 311: 109 - 112.

Mayer, D. K, Terrin, N. C, Kreps, G. L, Menon, U., McCance, K., Parsons, S. K., Mooney, K. H., (2007). Cancer survivors information seeking behaviors: a comparison of survivors who do and do not seek information about cancer. Patient Educ Couns. 2007. Mar; 65(3):342-50.

Mayer, D. K., Birken, S. A., Check, D. K., & Chen, R. C., (2015). Summing it up: An integrative review of studies of cancer survivorship care plans (2006–2013). Cancer, 121, 978–996. https://doi.org/10.1002/cncr.28884

Mazanec, S. R., Gallagher, P., Miano, W. R., Sattar, A., & Daly, B. J., (2017). Comprehensive assessment of cancer survivors' concerns to inform program development. JCSO 2017;15(3):e155-e162. doi https://doi.org/10.12788/jcso.0338

Mazzocco, K., Masiero, M., Carriero, M. C., Pravettoni, G., (2019). The role of emotions in cancer patients' decision-making ecancer 13 914. https://ecancer.org/en/journal/article/914-the-role-of-emotions-in-cancer-patients-decision-making

McCorkle, R., Ercolano, E., Lazenbym M., Schulman-Greenm D., Schillingm L. S., Lorig, K., Wagner, E. H., (2011). Self-management: Enabling and empowering patients living with cancer as a chronic illness. CA: American Cancer Journal for Clinicians. 2011; 61:50–62. doi: 10.3322//caac.20093.

McCullough, M. L., Patel, A. V., Kushi, L. H., Patel, R., Willett, W. C., Doyle, C., Thun, M. J., Gapstur, S. M., (2011). Following cancer prevention guidelines reduces risk of cancer, cardiovascular disease, and all-cause mortality. Cancer Epidemiol Biomarkers Prev. 2011;20(6):1089–1097. doi: 10.1158/1055-9965.EPI-10-1173.

McCorkle, R., Ercolano, E., Lazenby, M., Schulman-Green, D., Schilling, L. S., Lorig, K., Wagner, E. H., (2011). Self-management: Enabling and empowering patients living with



cancer as a chronic illness. (2011). CA Cancer J Clin. 2011 Jan-Feb;61(1):50-62. doi: 10.3322/caac.20093. Epub 2011 Jan 4. PMID: 21205833; PMCID: PMC3058905.

McCorry, N. K., Dempster, M., Quinn, J., Hogg, A., Newell, J., Moore, M., Kelly, S., Kirk, S. J., (2013). Illness perception clusters at diagnosis predict psychological distress among women with breast cancer at 6 months post diagnosis. Psychooncology. 2013 Mar; 22(3):692-8.

McGlynn, E. A., Asch, S. M., Adams, J., Keesey, J., Hicks, J., DeCristofaro, A., Kerr, E. A., (2003). The quality of health care delivered to adults in the United States. N Engl J Med. 2003;348(26):2635–2645. doi: 10.1056/NEJMsa022615

McGrath, P., & Holewa, H. (2012). What does the term 'survivor' mean to individuals diagnosed with a haematological malignancy? Findings from Australia. Supportive Care in Cancer, 20, 3287–3295. https://doi.org/10.1007/s00520-012-1453-4

 McKee, A. L., & Schover, L. R., (2001). Sexuality rehabilitation. Cancer. American Cancer

 Society.
 https://doi.org/10.1002/1097-0142(20010815)92:4+<1008::AID-</td>

 CNCR1413>3.0.CO;2-2.

McLennan, W., (1999). An Introduction to Sample Surveys A User's Guide. Produced by the Australian Bureau of Statistics

McLellan, E., MacQueen, K. M., & Neidig, J. L. (2003). Beyond the qualitative interview: Data preparation and transcription. Field Methods, 15(1), 63–84. doi:10.1177/1525822X02239573

McLeod, J., (2001). Qualitative research in counselling and psychotherapy. London: Sage

Meacham, E., Orem, J., Nakigudde, G., Zujewski, J. A., Rao, D., (2016). Exploring stigma as a barrier to cancer service engagement with breast cancer survivors in Kampala, Uganda. Psychooncology, 25 (2016), pp. 1206-1211

Mechanic, D., (2003). Physician discontent: challenges and opportunities. JAMA. 2003 Aug 20; 290(7):94



Meer, N., (2014). Key concept in Race and Ethnicity. Chapter 11. Health and Ethnicity. Published by SAGE

Mehdi, I., Monem, E. A., Al Bahrani, B. J., Al Kharusi, S., Nada, A. M., Al Lawati, J., & Al Lawati, N. (2014). Age at diagnosis of female breast cancer in Oman: Issues and implications. South Asian Journal of Cancer, 3(2), 101–106. http://doi.org/10.4103/2278-330X.130442

Meiser, B., Eisenbruch, M., Barlow-Stewart, K., Tucker, K., Steel, Z., Goldstein, D., (2001). Cultural aspects of cancer genetics: setting a research agenda. J Med Genet 2001; 38:425–429.

Melchior, H., Büscher, C., Thorenz, A., Grochocka, A., Koch, U., & Watzke, B. (2013). Self-efficacy and fear of cancer progression during the year following diagnosis of breast cancer. Psycho-Oncology, 22(1), 39-45. doi: 10.1002/pon.2054

Mellon, S., Gauthier, J., Cichon, M., Hammad, A., & Simon, M. S. (2013). Knowledge, Attitudes, and Beliefs of Arab-American Women Regarding Inherited Cancer Risk. Journal of Genetic Counseling, 22(2), 268–276. http://doi.org/10.1007/s10897-012-9546-2

Medndoza, N., Molero, F., Criado, F., Cornellana, M. J., Gonzalez, E., (2017). Sexual health after breast cancer: Recommendations from the Spanish Menopause Society, Federación Espanola ~ de Sociedades de Sexología, Sociedad Espanola ~ de Médicos de Atención Primaria and Sociedad Espanola ~ de Oncología Médica Nicolás Mendozaa, Francisca Molero b, Fermín Criadoc, Ma Jesús Cornellana d, Encarna González e, On behalf of the Sexuality In Breast Cancer Survivors Group. http://dx.doi.org/10.1016/j.maturitas.2017.02.010 0378-5122/© 2017 Elsevier B.V.

https://seom.org/seomcms/images/stories/recursos/Sexual_health_after_breast_cancer.pdf

Mendoza-D. S., & Dreisbach, J. L., (2018). Female breast cancer as taboo: Cultural factors and awareness amongst patients and their families in the Philippines. GEOGRAFIA Online TM Malaysian Journal of Society and Space 14 issue 4 (201-211) © 2018, e-ISSN 2680-2491 https://doi.org/10.17576/geo-2018-1404-16.



https://www.academia.edu/37840577/Female_breast_cancer_as_taboo_Cultural_factors_and _awareness_amongst_patients_and_their_families_in_the_Philippines

Meneses, K., Azuero, A., Hassey, L., McNees, P., and Pisu, M., (2011). "Does Economic Burden Influence Quality of Life in Breast Cancer Survivors?" Gynaecologic Oncology 124.3 (2011): 437-43. Web.

Meredith, C., Symonds, P., Webster, L., Lamont, D., Pyper, E., Gillis, C.R. & Fallowfield, L. (1996). Information needs of cancer patients in West Scotland: cross sectional survey of patients' views. British Medical journal, 313(21), 724-26.

Mermer, G., Nazli, A., Ceber, E., Mermer, G., (2016). Social perceptions of breast Cancer by women still undergoing or having completed therapy: a qualitative study. Asian Pac J Cancer Prev. 2016;17(2):503–10.

Merriam, S., (1995). What can you tell from an N of 1? Issues of validity and reliability in qualitative research. PAACE J Lifelong Learn.1995;4:51<60

Merkin, S. S., Stevenson, L., Powe, N., (2002). Geographic socioeconomic status, race, and advanced-stage breast cancer in New York City. The American Journal of Public Health, Jan, 2002, Vol.92(1), p.64(7). Available at: http://www.ncbi.nlm.nih.gov/pubmed/11772763

Mertens, D. M., (2010). Transformative mixed methods research.Qualitative Inquiry16: 469–474

Meyer, T. A., & Mark, M. M., (1995). Effects of psychosocial interventions with adult cancer patients: A meta-analysis of randomized experiments. Health Psychol 1995; / 14: / 1018.

Milaat, W. A., (2000). Knowledge of secondary-school female students on breast cancer and breast self-examination in Jeddah, Saudi Arabia. East Mediterr Health J. 2000, 6 (2–3): 338-344.

Miller, M. A., (1995) Culture, spirituality, and women's health. J Obstet Gynecol Neonatal Nurs. 1995 Mar-Apr; 24(3):257-63.



Miller, L. E., (2012). Sources of uncertainty in cancer survivorship. J Cancer Surviv. 2012 Dec;6(4):431-40. doi: 10.1007/s11764-012-0229-7. Epub 2012 Jul 20. PMID: 22815086.

Miller, K., (2012). Excellent care for cancer survivors. A Guide to Fully Meet Their Needs in medical offices and in the community. Chapter 1. An overview of cancer survivorship.

Miles, M. B., & Huberman, A. M., (1984). Qualitative data analysis, a sourcebook of new methods. Newbury Park, CA: Sage.

Miles, M. B., & Huberman, A. M. (1994). Qualitative data analysis: An expanded sourcebook. Second edition. Thousand Oaks, CA: Sage Publications. Available at: https://vivauniversity.files.wordpress.com/2013/11/milesandhuberman1994.pdf

Mills, M. & Sullivan, K. (1999). The importance of information giving for patients newly diagnosed with cancer: a review of the literature. Journal of Clinical Nursing, 8, 631–42.

Ministry of Health, United Arab Emirates, (MOHAP) (2014). National Cancer Registry Report. Available at: http://www.mohap.gov.ae/en/OpenData/Pages/default.aspx#afc952d9-1f0a-4b49-9403-

085e53858790=%7B%22k%22%3A%22%22%2C%22r%22%3A%5B%7B%22n%22%3A% 22MOHEnOpenDataCategory%22%2C%22t%22%3A%5B%22%5C%22%C7%82%C7%82 53746174697374696373%5C%22%22%5D%2C%22o%22%3A%22and%22%2C%22k%22 %3Afalse%2C%22m%22%3Anull%7D%5D%7D

Ministry of Health & Prevention, (MoHAP) UAE, (2017). The Statistics & Research Dept.
National Disease Registry Section. National Cancer Registry (UAE) (moh.gov.ae)

Ministry of Health and Prevention, (MoHAP) UAE., (2021). https://mohap.gov.ae/en/media-center/news/13/12/2021/uae-hosts-who-celebration-ofuniversal-health-coverage-day

Ministry of Health, New Zealand., (2010). Guidance for Improving Supportive Care for Adults with Cancer in New Zealand. Wellington: New Zealand, Ministry of Health, 2010. (Available from: www.health.govt.nz/publication/guidance-improving-supportive-care-adults-cancer-new-zealand) accessed 28th Nov. 2020



Minshull, J., Ross, K., & Turner, J., (1986). The human needs model of nursing. J Adv Nurs 11, 643–649.

Mishra, S. I., Scherer, R. W., Snyder, C., Geigle, P. M., Berlanstein, D. R., Topaloglu, O., (2012). Exercise interventions on health-related quality of life for people with cancer during active treatment [serial online]. Cochrane Database Syst Rev. 2012;8:CD008465.

Miyashita, M., Ohno, S., Kataoka, A., Tokunaga, E., Masunda, N., Shein, T., Kawabata, K., Takahashi, M., (2015). Unmet information needs and quality of life in young breast cancer survivors in Japan Cancer Nurs 38 1–11 https://doi.org/10.1097/NCC.00000000000201

Moadel, A. B., Morgan, C., & Dutcher, J., (2007). Psychosocial needs assessment among an undeserved, ethnically diverse cancer patient population. Cancer, 109 (S2), 446-454

Moch, S. D., (1998). Health within illness: Concept development through research and practice. Journal of Advanced Nursing, 28 (2), 305-310.

Moffatt, S., Noble, E., White, M. DB, A., (2012). Addressing the Financial Consequences of Cancer: Qualitative Evaluation of a Welfare Rights Advice Service. Vol.7(8), p.e42979. https://doi.org/10.1371/journal.pone.0042979

Moghaddam, F. M., Walker, B. R., & Harre, R., (2003). Cultural distance, levels of abstraction and the advantages of mixed methods. In Handbook on mixed methods in the behavioral and social sciences, ed. A. Tashakkori and C. Teddlie, 51–89. Thousand Oaks, CA: Sage.

Montagu, M. F. A (1955). The direction of human development: Biological and social bases. New York: Harper & Brothers.

Montazeri, A. (2008). Health-related quality of life in breast can-cer patients: A bibliographic review of the literature from 1974 to 2007. Journal of Experimental and Clinical Cancer Research, 27(1), 32

Mor, V., Guadagnoli, E., Wool, M., (1987). An examination of the concrete service needs of advanced cancer patients. Journal of Psychosocial Oncology 5: 1-17.



Morales-Sánchez, L., Luque-Ribelles, V., Gil-Olarte, P., Ruiz-González, P., Guil, R., (2021). "Enhancing Self-Esteem and Body Image of Breast Cancer Women through Interventions: A Systematic Review" Int. J. Environ. Res. Public Health 18, no. 4: 1640. https://doi.org/10.3390/ijerph18041640

Morgan, D. L., (1997). Focus groups as qualitative research. Thousand Oaks, CA: Sage

Morgan, D., (1998). "Practical strategies for combining qualitative and quantitative methods: Applications to health research." Qualitative Health Research 8, (3) 362-376

Morgan, D. L., (2007). Paradigms lost and pragmatism regained: methodological implications of combining qualitative and quantitative methods. Journal of Mixed Methods Research, 1 (1) (2007), pp. 48-76

Morgan, M. A., (2009). Cancer survivorship: History, quality-of-life issues, and the evolving multidisciplinary approach to implementation of cancer survivorship care plans. Oncology Nursing Forum. 2009;36(4):429–436.

Morgan, D. L., (2014). Integrating Qualitative and Quantitative Methods: A Pragmatic Approach. Thousand Oaks: Sage

Moosavi, S., Rohani, C., Borhani, F., Akbari, M. E., (2019). Consequences of Spiritual Care for Cancer Patients and Oncology Nurses: a Qualitative Study. Year: 2019| Volume: 6 | Issue: 2 | Page: 137-144

Morse, J. M., & Carter, B., (1996). The essence of enduring and expressions of suffering: the reformulation of self. Sch Inq Nurs Pract 10, 43–60

Morre, J. R., & Spiegel, D., (2004). Cancer, Culture and Communication. Chapter 3: Quality of life in culturally diverse cancer patients. Kluwer Academic/Plennum Publisher. New York.

Morse, J. M. (1991). Approaches to qualitative-quantitative methodological triangulation. Nursing Research, 40, 120–123

Morse, J. M., (1994). Designing funded qualitative research. Handbook of qualitative research. 2nd ed Denizin NK, Lincoln YS, editors. California: Thousand Oaks.



Morse, J. M. & Field, P. A. (1996). Nursing research: The application of qualitative approaches (2nd ed.). Thousand Oaks, CA: Sage Publications

Morris, E. F., (2001). Clinical practices with African Americans: juxtaposition of standard clinical practices and Africentricism, Prof Psychol Res Prac , 2001, vol. 32 (pg. 563-572)

Morris, B., Lepore, S., Wilson, B., Lieberman, M., Dunn, J., Chambers, S., (2014). Adopting a survivor identity after cancer in a peer support context. Journal of Cancer Survivorship, 2014, Vol.8(3), pp.427-436

Morris, J., Ingham, R., (1988). Choice of surgery for early breast cancer: psychological considerations, Psychol Med, 1988, vol. 27 (pg. 1257-62)

Mosadeghrad, A. M., (2013). Obstacles to TQM success in health care systems. International Journal of Health Care Quality Assurance, 26(2): 147–173

Mosadeghrad, A. M., (2014). Factors Affecting Medical Service Quality. Iranian journal of public health, 43(2), 210–220

Mostafavian, Z., & Shaye, Z. A., (2018). Evaluation of physicians' skills in breaking bad news to cancer patients. J Family Med Prim Care 2018; 7:601-5. Available from: http://www.jfmpc.com/text.asp?2018/7/3/601/236850

Mukherjee, A., Mazumder, K., & Ghoshal, S. (2018). Impact of Different Sociodemographic Factors on Mental Health Status of Female Cancer Patients Receiving Chemotherapy for Recurrent Disease. Indian journal of palliative care, 24(4), 426–430. doi: 10.4103/IJPC.IJPC_64_18

Mullan, F., (1985). Seasons of survival: reflections of a physician with cancer. N Engl J Med 1985; 313:270-3. Doi: 10.1056/nejm198507253130421

Murray, H. A. (1938). Explorations in Personality. New York: Oxford University Press

Musarezaie, A., Naji. E. H., Momeni, G. T., & Karimian, J., & Ebrahim, A., (2012). The relationship between spiritual wellbeing and stress, anxiety, and depression in patients with breast cancer. Journal of Isfahan Medical School. 9/3/2012, Vol. 30 Issue 195, p1-10. 10p.



Musick, M.A., Koenig, H.G., Hays, J.C. and Cohen, H.J. (1998). Religious activity and depression among community-dwelling elderly persons with cancer – the modelling effect of race, Journal of Gerontology Social Sciences, 53: S218 – 227.

Myers, J. S., Koleck, T. A., Sereika, S. M., Conley, Y. P., & Bender, C. M. (2017). Perceived cognitive function for breast cancer survivors: Association of genetic and behaviorally related variables for inflammation. Supportive Care in Cancer, 25, 2475–2484. https://doi.org/10.1007/s00520-017-3654-3

Nabolsi, M. M., & Carson, A. M., (2011). Spirituality, illness and personal responsibility: the experience of Jordanian Muslim men with coronary artery disease. Scandinavian Journal of Caring Sciences, 25(4), 716-724.

Naidoo, J, Hayes, E., Teo, MY., Horgan, A., Calvert, P., O'Connor, M., (2020). An Irish breast cancer survivorship study: are we meeting our patients' needs? Irish Medical Journal (IMJ). http://hdl.handle.net/10147/303571

Nail, L. M., (2001). I'm coping as fast as I can: Psychosocial adjustment to cancer and cancer treatment. Oncology Nursing Forum, 28, 967–970.

Nair, S. C., Jaafar, H., Jaloudi, M., Qawasmeh, K., AlMarar, A., & Ibrahim, H. (2018). Supportive care needs of multicultural patients with cancer in the United Arab Emirates. Ecancermedicalscience, 12, 838. https://doi.org/10.3332/ecancer.2018.838

Nair, S. C., Hassen, W. A., 2, Sreedharan, J., Qawasmeh, K., Ibrahim, H., (2019). Comparison of perceptions of unmet supportive care needs between cancer patients and their oncologists. Support Care Cancer. 2019 Nov;27(11):4229-4235. doi: 10.1007/s00520-019-04705-4. Epub 2019 Mar 7. DOI: 10.1007/s00520-019-04705-4

Najjar, H., & Easson, A., (2010). Age at diagnosis of breast cancer in Arab nations, International Journal of Surgery, Volume 8, Issue 6, 2010, Pages 448-452, ISSN 1743-9191, https://doi.org/10.1016/j.ijsu.2010.05.012.

Narayanan, V., & Koshy, C. (2009). Fatigue in cancer: a review of literature. Indian journal of palliative care, 15(1), 19–25. doi:10.4103/0973-1075.53507


Nascimento, L. C., Santos de Oliveira, F. C., Moreno, M. F., & Machado da Silva, F., (2010). Spiritual Care: An Essential Component of the Nurse Practice in Pediatric Oncology. Acta Paulista de Enfermagem 23: 437–40.

National Breast Cancer Centre, (2003). National Cancer Control Initiative. Clinical Practice Guidelines for the Psychosocial Care of Adults with Cancer. Camper-down, NSW: National Breast Cancer Centre, 2003

National Cancer Policy Forum, (2014). Nass, S. J., Patlak, M., National Cancer Policy Forum; Board on Health Care Services; Institute of Medicine. Ensuring Patient Access to Affordable Cancer Drugs: Workshop Summary. 2014. Available at: http://www.nap.edu/catalog.php?record_id=18956.

National Cancer Institute., (2004). Normal adjustment, psychosocial distress, and the adjustment disorders (PDQ) Updated 2004 Apr 22; cited 2004 Jun 25. Available at: http://www.cancer.gov/cancertopics/pdq/supportivecare/adjustment/HealthProfessional.

National Cancer Institute, (2007). Cancer trends progress report—2007 update: Costs of cancer care. 2007. [May 13, 2013]. http://progressreport.cancer.gov/2007/doc_detail .asp?pid=1&did=2007&chid=75&coid=726&mid=

National Cancer Institute, (2012). SEER Stat Fact Sheets: All Sites. 2012. [April 19, 2018]. http://seer.cancer.gov/statfacts/html/all.html.

National Cancer Institute, (2013). Estimated U.S. Cancer Prevalence Counts: Method. 2013c. [April 19, 2013]. http://dccps.nci.nih.gov/ocs/prevalence/prevalence.html.

National Cancer Institute, (2019). Cancer Coping with Cancer: Survivorship: A new normal. https://www.cancer.gov/about-cancer/coping/survivorship/new-normal [accessed on 26th Aug, 2019].

National Cancer Institute, (2021a). NCI dictionary of cancer terms. Survivor: https://www.cancer.gov/publications/dictionaries/cancer-terms/def/survivor



National Cancer Institute, (2021b). NCI dictionary of cancer terms. Survivorship: https://www.cancer.gov/publications/dictionaries/cancer-terms/def/survivorship

National Comprehensive Cancer Network Guidelines, (2010). Clinical practice guidelinesinoncology:Cancer-relatedfatigue.Retrievedfrom.http://www.nccn.org/professionals/physician_gls/pdf/fatigue.pdf

National Cancer Institute, (2021). Helping Cancer Survivors Cope with Cancer-Related Anxiety and Distress. https://www.cancer.gov/news-events/cancer-currents-blog/2020/cancer-survivors-managing-anxiety-distress [Accessed 22nd May 2021]

National Comprehensive Cancer Network, (2020). Finding Comfort in Spirituality. https://www.nccn.org/patients/resources/life_with_cancer/spirituality.aspx

National Coalition for Cancer Survivorship, (1996). Imperatives for Quality Cancer Care: Access, Advocacy, Action, and Accountability. Clark EJ, Stovall EL, Leigh S, Siu AL, Austin DK, Rowland JH. Silver Spring, MD: NCCS.

National Institutes of health. U. S. (1993). Department of health and human services. Public health services. Radiation therapy and you. A guide to self-help during treatment. Chapter 23: Managing side effects. Pg 24

National Health and Medical Research Council, (2000). Psychosocial Clinical Practice Guidelines: Information, Support and Counselling for Women with Breast Cancer. Canberra: Commonwealth of Australia

National Research Council (2006). From Cancer Patient to Cancer Survivor: Lost inTransition. Washington, DC: The National Academies Press.https://doi.org/10.17226/11468

Naus, M. J., Ishler, M. D., Parrott., C. E. and Kovacs, S. A. (2009). Cancer survivor adaptation model: conceptualizing cancer as a chronic illness. Journal of Clinical Psychology, 65: 1350–1359. doi:10.1002/jclp.20622



Naylor, M. D., Bowles, K. H., McCauley, K. M., Maccoy, M. C., Maislin, G., Pauly, M.
V., Krakauer, R., (2011). Journal of Evaluation in Clinical Practice. 2011. High-value transitional care: Translation of research into practice.

Naylor, M. D., Feldman, P., Keating, S., Koren, M. J., Kurtzman, E. T., Maccoy, M., Krakauer, R., (2009). Translating research into practice: Transitional care for older adults. Journal of Evaluation in Clinical Practice. 2009; 15:1164–1170.

Naylor, M. D., Stephens, C., Bowles, K. H., Bixby, M. B., (2005). Cognitively impaired older adults: From hospital to home. A pilot study of these patients and their caregivers. American Journal of Nursing. 2005;105(2):40–49

Naylor, M. D., Keating, S. K., (2008). Transitional care: Moving patients from one care setting to another. State of the science: Supporting family caregivers. American Journal of Nursing. 2008;108(S9):58–63

Niedzwiedz, C.L., Knifton, L., Robb, K.A. Katikireddi, S. V., & Smith, D. J., (2019). Depression and anxiety among people living with and beyond cancer: a growing clinical and research priority. BMC Cancer 19, 943 (2019). https://doi.org/10.1186/s12885-019-6181-4

Nies, Y. H., Islahudin, F., Chong, W. W., Abdullah, N., Ismail, F., Bustan, R. S. A., Wong, Y. F., Saladina, J. J., Mohamed, S. N., (2017). Treatment decision-making among breast cancer patients in Malaysia. Patient Prefer Adherence. 2017; 11:1767-1777 https://doi.org/10.2147/PPA.S143611

Nelson, H. D., Zakher, B., Cantor, A., Fu, R., Griffin, J., O'Meara, E. S., Buist, D. S., Kerlikowske, K., van Ravesteyn, N. T., Trentham-Dietz. A., Mandelblatt, J. S., Miglioretti, DL., (2012). Risk factors for breast cancer for women aged 40 to 49 years: a systematic review and meta-analysis. Ann Intern Med. 2012 May 1; 156(9):635-48. doi: 10.7326/0003-4819-156-9-201205010-00006. PMID: 22547473; PMCID: PMC3561467.

Nekhlyudov, L., Aziz, N., Lerro, C. C., Virgo, K. S., (2012). Oncologists' and primary care providers' awareness of late effects of cancer treatment: implications for survivorship care [abstract] J Clin Oncol. 2012;30 (Suppl): Abstract 6008



Nekhlyudov, L, Levit, L., Hurria, A., Ganz, P. A., (2014). Patient-centered, evidence-based, and cost-conscious cancer care across the continuum: Translating the Institute of Medicine report into clinical practice. CA: A Cancer Journal for Clinicians. DOI: 10.3322/caac.21249

Nekhlyudov, L., Ganz, P. A., Arora, N. K., Rowland, J. H., (2017). Going beyond being Lost in Transition: a decade of progress in cancer survivorship. J Clin Oncol. 2017; 35:1978–81. doi: 10.1200/JCO.2016.72.1373.

Ness, S., Kokal, J., Fee-Schroeder, K., Novotny, P., Satele, D., Barton, D., (2013). Concerns across the survivorship trajectory: results from a survey of cancer survivors. Oncology nursing forum, January 2013, Vol.40(1), pp.35-42. DOI: 10.1188/13.ONF.35-42.

Newell, S., & Jordan, Z., (2015). The patient experience of patient-centred communication with nurses in the hospital setting: a qualitative systematic review protocol. JBI Database System Rev Implement Rep. 2015 Jan;13(1):76-87. doi: 10.11124/jbisrir-2015-1072. DOI: 10.11124/jbisrir-2015-1072

Newman, I., & Benz, C. R., (1998). Qualitative-quantitative research methodology: Exploring the interactive continuum. Carbondale: University of Illinois Press

Ngamaba, K. H., Panagioti, M., Armitage, C. J., (2017). How strongly related are health status and subjective well-being? Systematic review and meta-analysis, European Journal of Public Health, Volume 27, Issue 5, October 2017, Pages 879–885, https://doi.org/10.1093/eurpub/ckx081

Ng, Z. X., Ong, M. S., Jegadeesan, T., Deng, S., & Yap, C. T. (2017). Breast Cancer: Exploring the Facts and Holistic Needs during and beyond Treatment. Healthcare, 5(2), 26. http://doi.org/10.3390/healthcare5020026.

Ngai, C., Lucas, G., Jason, W. B., (2018), Evidence-based medicine and precision medicine: Complementary approaches to clinical decision-making, Precision Clinical Medicine, Volume 1, Issue 2, September 2018, Pages 60–64, https://doi.org/10.1093/pcmedi/pby009

Nicholas, D. R., (2016). Psychosocial care of the adult cancer patient. Evidence based practice in Psycho-oncology. Oxford university press 2016



NIH, National Cancer Institute, (2017). Patient Centred Care Chapter 6: Summary, Conclusions, and Future Directions. https://healthcaredelivery.cancer.gov/pcc/pccm_ch6.pdf

Nixon, A. V., Narayanasamy, A., Penny, V., (2013). An investigation into the spiritual needs of neuro-oncology patients from a nurse perspective. BMC Nurs 2013; 12:2.

Nizamli, F., Anoosheh, M., & Mohammadi, E. (2011). Experiences of Syrian women with breast cancer regarding chemotherapy: A qualitative study. Nursing & Health Sciences, 13(4), Kawar, 2013; 481-487

Noguera, A., Yennurajalingam, S., Torres-Vigil, I., Parsons, H. A., Duarte, E. R., Palma, A., Bunge, S., Palmer, J. L., Bruera, E., (2014). Decisional control preferences, disclosure of information preferences, and satisfaction among Hispanic patients with advanced cancer. Journal of Pain and Symptom Management, May 2014, Vol.47(5), pp.896-905.

Northouse, L.L., & Northouse, P.G. (1996). Interpersonal communication systems. In R. McCorkle, M. Grant, M. Frank-Stromborg, & S.B. Baird (Eds.), Cancer nursing: A comprehensive textbook (2nd ed., pp. 1211–1222). Philadelphia: Saunders.

Nowell, L. S., Norris, J. N., White, D. E & Moules, N. J. (2017). Thematic analysis: Striving to meet the trustworthiness criteria. International Journal of Qualitative Methods 16(1), 1–13

Obeidat, R. F., Dickerson, S. S., Homish, G. G., Alqaissi, N. M., & Lally, R. M., (2013a). Controlling fear, Jordanian women's perceptions of the diagnosis and surgical treatment of early-stage breast cancer. Cancer Nursing, 36(6), 484–492.

Obeidat, R. F, Homish, G. G, & Lally, R. M. (2013b). Shared decision making among individuals with cancer in non-western cultures: a literature review. Oncology Nursing Forum, 40(5), 454–463

Obeidat, R. F., & Lally, R. M., (2014). Health-related information exchange experiences of Jordanian women at breast cancer diagnosis. Journal of Cancer Education, 29(3), 548–554.

Obeidat, R., & Khrais, H. I., (2015). Information Needs and Disclosure Preferences among Jordanian Women Diagnosed with Breast Cancer. Journal of Cancer Education, 30, 94–99.



O'Cathain, A., Murphy, E., & Nicholl, J., (2007). Why, and how, mixed methods research is undertaken in health services research in England: a mixed methods study BMC Health Services Research, 7 (2007), p. 85

O'Cathain, A., Murphy, E., Nicholl, J., (2008). The quality of mixed methods studies in health services research. Journal of Health Services Research & Policy, 13 (2) (2008), pp. 92-98

O'Cathain A, Murphy E, Nicholl J., (2010). Three techniques for integrating data in mixed methods studies. BMJ 2010;341:c4587. 10.1136/bmj.c4587.

O'Connor, M., Christensen, S., Jensen, A. B., Møller, S., & Zachariae, R. (2011). How traumatic is breast cancer? Post-traumatic stress symptoms (PTSS) and risk factors for severe PTSS at 3 and 15 months after surgery in a nationwide cohort of Danish women treated for primary breast cancer. British Journal of Cancer, 104(3), 419–426. http://doi.org/10.1038/sj.bjc.6606073

Odle, T. G., (2014) Adverse effects of breast cancer treatment. Radiol Technol. 2014 Jan-Feb;85(3):297M-319M; quiz 320M-323M. PMID: 24395910

Ogce, F., Ozkan, S., & Baltalarli, B., (2007). Psychosocial Stressors, Social Support and Socio-demographic Variables as Determinants of Quality of Life of Turkish Breast Cancer Patients. Determinants of Quality of Life of Turkish Breast Cancer Patients. Asian Pacific J Cancer Prev, 8, 77-82

Oguchi, M., Jansen, J., Butow, P., Colagiuri, B., Divine, R. & Dhillon, H. (2011). Measuring the impact of nurse cue–response behaviour on cancer patients' emotional cues. Patient Education and Counseling 82(2): 163–8.

Okamura, H., Uchitomi, Y., Sasako, M., Eguchi, K., &Kakizoe, T., (1998). Guidelines for Telling the Truth to Cancer Patients, Japanese Journal of Clinical Oncology, Volume 28, Issue 1, January 1998, Pages 1–4, https://doi.org/10.1093/jjco/28.1.1

Okediji, P. T., Salako, O., Fatiregun, O. O., (2017). Pattern and predictors of unmet supportive care needs in cancer patients. Cureus. 2017;9(5):e1234.



Okuhara, T., Ishikawa, H., Urakubo, A., Hayakawa, M., Yamaki, C., Takayama, T., & Kiuchi, T. (2018). Cancer information needs according to cancer type: A content analysis of data from Japan's largest cancer information website. Preventive medicine reports, 12, 245–252. doi: 10.1016/j.pmedr.2018.10.014

Onitilo, A. A., Engel, J. M., & Stankowski, R. V. (2014). Cardiovascular toxicity associated with adjuvant trastuzumab therapy: prevalence, patient characteristics, and risk factors. Therapeutic Advances in Drug Safety, 5(4), 154–166. http://doi.org/10.1177/2042098614529603

Onwuegbuzie, A. J., & Leech, N. L., (2004). Enhancing the Interpretation of "Significant" Findings: The Role of Mixed Methods Research. The Qualitative Report Volume 9 Number 4 December 2004 770-792 770. http://www.nova.edu/ssss/QR/QR9-4/onwuegbuzie.pdf

Onwuegbuzie, A. J., & Leech, N. L., (2005). On becoming a pragmaticresearcher: the importance of combining quantitative and qualitative methodologies. International Journal of Social Research Methodology8 (5), 375–387.

Onwuegbuzie, A. J., & Collins, K. M. T., (2007). "A Typology of Mixed Methods Sampling Designs in Social Science Research." Qualitative Report 12: 281-316.

O'Reilly, M., & Parker, N., (2013). 'Unsatisfactory saturation': a critical exploration of the notion of saturated sample sizes in qualitative research. Qual. Res. 2013;13(2):190–197. doi: 10.1177/1468794112446106

Orlander, J. D., Fincke, B. G., Hermanns, D., Johnson, G. A., (2002). Medical residents' first clearly remembered experiences of giving bad news. Journal of General Internal Medicine, November 2002, Vol.17(11), pp.825-840.

Ortashi, O., Raheel, H., Shalal, M. & Osman, N. (2013). Awareness and Knowledge about Human Papillomavirus Infection and Vaccination among Women in UAE. Asian Pacific Journal of Cancer Prevention, 14(10), 6077-80.

Oskay-Ozcelik, G., Lehmacher, W., Konsgen, D., Christ, H., Kaufmann, M., Lichtenegger, W., (2007). Breast cancer patients' expectations in respect of the physician-



patient relationship and treatment management results of a survey of 617 patients. Ann Oncol 18:479–484. doi:10.1093/annonc/mdl456

Östlunda, U., Kidd, L., Wengström, Y., Rowa-Deward, N., (2011). Combining qualitative and quantitative research within mixed method research designs: A methodological review. International Journal of Nursing Studies. Volume 48, Issue 3, March 2011, Pages 369-383. https://www.sciencedirect.com/science/article/pii/S0020748910003639#Rbib0105

Osse, B. H., Vernooij-Dassen, M. J., Schade, E., & Grol, R. P., (2007). A practical instrument to explore patients' needs in palliative care: The Problems and Needs in Palliative Care Questionnaire short version. Palliative Medicine, 21, 391-399. doi:10.1177/0269216307078300

Østerbye, T., Nyby, T. K., (2010). Landsdækkende undersøgelse af kræftpatienters oplevelser [In Danish]. 2010, Copenhagen: Unit of Patient-Perceived Quality

Otoo, B. K., (2020). Declaring My Ontological and Epistemological Stance: A Reflective Paper. Journal of Educational Thought Vol. 53, No. 1, 2020, 67-88. https://www.researchgate.net/publication/344231306_Declaring_My_Ontological_and_Epist emological_Stance_A_Reflective_Paper/link/607a0bb32fb9097c0ceca7f0/download

Özkan, S., Özkan, M., Armay, Z., (2010). The burden of mastectomy. In: Preedy VR, Watson RR, editors. Handbook of disease burdens and quality of life measures. New York: Springer; 2010

Özkan, S., Yaklaşım, M. K. H. P., Kalitesi, Y., (2003). In: Topuz E, Aydıner A, Dinçer Mİ, editors. Meme kanseri. İstanbul: Nobel Tip Kitapevleri; 2003. p. 681–90.

Padela, A. I., Peek, M., Johnson-Agbakwu, C. E., Hosseinian, Z., & Curlin, F. (2014). Associations between religion-related factors and cervical cancer screening among Muslims in greater chicago. Journal of lower genital tract disease, 18(4), 326–332. doi:10.1097/LGT.00000000000026



Paduca, A., Arnaut, O., Beschieru, E. et al. (2021). Shared decision making and patients satisfaction with strabismus care—a pilot study. BMC Med Inform Decis Mak 21, 109 (2021). https://doi.org/10.1186/s12911-021-01469-y

Palmer, N, R., Giger, A. M; Felder, T. M; Lingyi, L., Case, L. D., Weaver, K. E., (2013). Washington, DC: American Public Health Association. American journal of public health (1971), 2013, Vol.103 (7), p.1306-1313

Paludi, (2014). Chap 10. Impact of Psychosocial factors on health outcomes of cancer in women.

Pan American Health Organisation, (2016). Early Detection (2 OF 3): Breast physiology and the clinical breast exam (CBE). Breast Health Awareness and Early Detection Strategies and Early Detection: Diagnosis and Screening with Mammography. https://www.paho.org/en/documents/early-detection-breast-physiology-and-clinical-breast-exam-cbe

Papadakos, J., Urowitz, S., Olmstead, C., Friedman, A.J., Zhu, J. & Catton, P. (2014). Informational needs of gastrointestinal oncology patients. Health Expectations, 17, doi: 10.1111/hex.12296

Parker, S. M., Clayton, J. M., Hancock, K., Walder, S., Butow, P. N., Carrick, S., Currow, D., Ghersi, D., Glare, P., Hagerty, R., Tattersall, M. H., (2007). A systematic review of prognostic/end-of-life communication with adults in the advanced stages of a life-limiting illness: patient/caregiver preferences for the content, style, and timing of information. J Pain Symptom Manage. 2007 Jul;34(1):81-93. doi: 10.1016/j.jpainsymman.2006.09.035. Epub 2007 May 25. PMID: 17531434.

Park, B. W., & Hwang, S. Y., (2012). Unmet needs of breast cancer patients relative to survival duration. Yonsei Med J. 2012;53(1):118–25

Park, C. L., Edmondson, D., Hale-Smioth, A, Blank, T. O., (2009). Religiouness/spirituality and health behaviours iin younger adult cancer survivors: does faith promote a healthier lifestyle? J Behav Med. 2009; 32:582-591



Parry, C., Kent, E. E., Mariotto, A. B., Alfano, C. M., & Rowland, J. H., (2011). Cancer survivors: a booming population. Cancer epidemiology, biomarkers & prevention: a publication of the American Association for Cancer Research, cosponsored by the American Society of Preventive Oncology, 20(10), 1996–2005. https://doi.org/10.1158/1055-9965.EPI-11-0729

Passildas, J., Collard, O., Savoye, A-M. Dohou, Ginzac, A., Thivat, E., Durando, X., **Kwiatkowski, F., Penault-Llorca, F., Abrial, C., Mouret-Reynier, M. A., (2019).** Impact of Chemotherapy-induced Menopause in Women of Childbearing Age with Non-metastatic Breast Cancer - Preliminary Results from the MENOCOR Study. Clin Breast Cancer. 2019 Feb;19(1): e74-e84. doi: 10.1016/j.clbc.2018.10.003. Epub 2018 Oct 23. PMID: 30448088.

Paskett, E. D., Dean, J. A., Oliveri, J. M., Harrop. J. P., (2012). Cancer-related lymphedema risk factors, diagnosis, treatment, and impact: a review. J Clin Oncol. 2012 Oct 20;30(30):3726-33. doi: 10.1200/JCO.2012.41.8574. Epub 2012 Sep 24. PMID: 23008299.

Patenaude, A. F., & M. J. Kupst. (2005). Psychosocial functioning in pediatric cancer. Journal of Pediatric Psychology 30(1):9–27.

Paterick, T. E., Patel, N., Tajik, A. J., & Chandrasekaran, K. (2017). Improving health outcomes through patient education and partnerships with patients. Proceedings (Baylor University. Medical Center), 30(1), 112–113.

Patsou, E. D., Alexias, G. T., Anagnostopoulos, F. G., & Karamouzis, M. V. (2018). Physical activity and sociodemographic variables related to global health, quality of life, and psychological factors in breast cancer survivors. Psychology research and behavior management, 11, 371–381. doi:10.2147/PRBM.S170027

Patton, M. Q., (1990). Qualitative evaluation and research methods (2nd ed.). Newbury Park, CA: Sage.

Patton, M. Q. (2001). Qualitative research and evaluation and methods (3rd ed.). Beverly Hills, CA: Sage



Patton, M. Q., (2002) Qualitative Research and Evaluation Methods, 3rd edn. Sage Publications, Thousand Oaks, CA **Park, B. W., Hwang, S. Y., (2012).** Unmet needs of breast cancer patients relative to survival duration. Yonsei Med J. 2012;53(1):118–25

Paraskevi, T. (2012). Quality of life outcomes in patients with breast cancer. Oncology Reviews, 6(1), e2. http://doi.org/10.4081/oncol.2012.e2

Pargament, K. I., (1997). The Psychology of Religion and Coping: Theory, Research, Practice. New York, NY: Guilford Press; 1997.

Parizadeh, H., Abadi Mashhadi, H., (2012). Comparison the effectiveness of existential therapy and reality group therapy on problem solving body image of women with mastectomy. Iranian J Obs. 2012;15(22):35–27.

Pauwels, E., De Bourdeaudhuij, I., Charlier, C., Lechner, L., & Hoof, E. V., (2012): Psychosocial Characteristics Associated with Breast Cancer Survivors' Intimate Partners' Needs for Information and Support After Primary Breast Cancer Treatment, Journal of Psychosocial Oncology, 30:1, 1-20. http://dx.doi.org/10.1080/07347332.2011.633982

Pauwels, J. E. E., Charlier, C., Bourdeaudhuij, D. I., Lechner, L., & Hoof, V. E., (2013). Care needs after primary breast cancer treatment. Survivors' associated sociodemographic and medical characteristics. Psycho-Oncology 22: 125–132 (2013). DOI: 10.1002/pon.2069.

Pawijit, Y., Likhitsuwan, W., Ludington, J., & Pisitsungkagarn, K., (2017). Looks can be deceiving: body image dissatisfaction relates to social anxiety through fear of negative evaluation. International Journal of Adolescent Medicine and Health. 2017; DOI: 10.1515/ijamh-2017-0031

Payne, J. K., (2014). State of the science: Stress, inflammation and cancer. Oncology Nursing Forum, 41, 533–540. doi: 10.1188/14.ONF.533-540

Payne, V. L., Singh, H., Meyer, A. N., Levy, L., Harrison, D., & Graber, M. L., (2014). Patient-Initiated Second Opinions: Systematic Review of Characteristics and Impact on Diagnosis, Treatment, and Satisfaction, Mayo Clinic Proceedings, Volume 89, Issue 5, 2014, Pages 687-696, ISSN 0025-6196, https://doi.org/10.1016/j.mayocp.2014.02.015.



Payne, J. K., & Murphy-Ende, K., (2019). Current Trends in Oncology Nursing. Second edition. Chapter 1: Theoretical Frameworks and Philosophies of Care. https://www.ons.org/sites/default/files/2019-02/0673_sample.pdf

Peters-Golden, H., (1982). Varied perceptions of social support in the illness experience. Social Science Medicine. 1982; 16:463–491. doi: 10.1016/0277-9536(82)90057-0.

Petersen, G. S., Knudsen, J. L., Vinter, M. M., (2015). Cancer patients' preferences of care within hospitals: a systematic literature review, International Journal for Quality in Health Care, Volume 27, Issue 5, October 2015, Pages 384–395, https://doi.org/10.1093/intqhc/mzv059

Peled, R., Carmil, D., Siboni-Samocha, O., & Shoham-Vardi, I. (2008). Breast cancer, psychological distress and life events among young women. BMC cancer, 8, 245. doi:10.1186/1471-2407-8-245

Peltonen, R., Socialt, S., & Livskontroll, O. H., (1996). Socialpolitiska institutionen, Abo Akademi, Abo 1996. (Social support, control of life and health).

Pelzang, R., (2010). Time to learn: understanding patient-centred care. British Journal ofNursing,2010,Vol19,No14.https://pdfs.semanticscholar.org/d9e7/b8e301ab774b129f0d83d33d7fa52151ad1a.pdf

Pencheon, D., Mezler, D., Gray, M., & Guest, C., (2006). Assessing health needs. Pg 38-47. Oxford Handbook of Public Health, 2nd edition. Part1: Options and decisions; 1.3: Assessing health needs. Pg 20. Oxford university Press NY. New York: Oxford University Press.

https://books.google.ae/books?id=cYnzLTV7aVEC&printsec=frontcover#v=onepage&q&f=f alse

Peuckmann, V., Ekholm, O., Rasmussen, N. K, Groenvold , M., Christiansen, P., Møller, S., Eriksen, J., Sjøgren, P., (2009). Chronic pain and other sequelae in long-term breast cancer survivors: nationwide survey in Denmark. Eur J Pain. 2009 May;13(5):478-85. doi: 10.1016/j.ejpain.2008.05.015. Epub 2008 Jul 16. PMID: 18635381Peters-Golden, H., (1982).



Varied perceptions of social support in the illness experience. Social Science Medicine. 1982; 16:463–491. doi: 10.1016/0277-9536(82)90057-0.

Petersen, S., Heesacker, M., & deWitt Marsh, R., (2001). Medical decision making among cancer patients. Journal of Counseling Psychology, 48(2), 239–244. https://doi.org/10.1037/0022-0167.48.2.239

Petersen, G. S., Knudsen, J. L., Vinter, M. M., (2015). Cancer patients' preferences of care within hospitals: a systematic literature review, International Journal for Quality in Health Care, Volume 27, Issue 5, October 2015, Pages 384–395, https://doi.org/10.1093/intqhc/mzv059

Pham, L. T. M., (2018). QUALITATIVE APPROACH TO RESEARCH. A review of
advantages and disadvantages of three paradigms: positivism, interpretivism and critical
inquiry.DOI:10.13140/RG.2.2.13995.54569.https://www.researchgate.net/publication/324486854

Phillips, J. L., & Currow, D. C., (2010). Cancer as a chronic disease. Collegian. 2010;17(2):47–50.

Phillipson, L., Larsen-Truong, K., Jones, S. & Pitts, L. (2012). Improving cancer outcomes among culturally and linguistically diverse communities: a rapid review of the literature. Australia: The Sax Institute.

Philip, E. J., & Merluzzi, T. V. (2016). Psychosocial issues in post-treatment cancer survivors: Desire for support and challenges in identifying individuals in need. Journal of psychosocial oncology, 34(3), 223–239. doi:10.1080/07347332.2016.1157716

Pierce, P., (1988). Women's experience of choice: Confronting the options for treatment of breast cancer. In J. K. Harness, H. A. Oberman, A. S. Lichter, D. Adler, & R. L. Cody (Eds.), Breast cancer collaborative management. Chelsea, MI: Lewis Publishing

Pisu, M., Martin, M. Y., Shewchuk, R., & Meneses, K. (2014). Dealing with the financial burden of cancer: perspectives of older breast cancer survivors. Supportive care in cancer:



official journal of the Multinational Association of Supportive Care in Cancer, 22(11), 3045–3052. https://doi.org/10.1007/s00520-014-2298-9

Pikler, V., & Winterowd, C. (2003). Racial and Body Image Differences in Coping for Women Diagnosed with Breast Cancer. Health Psychology, 22(6), 632-637. http://dx.doi.org/10.1037/0278-6133.22.6.632

Pietromonaco, P. R., & Collins, N. L. (2017). Interpersonal mechanisms linking close relationships to

Pitceathly, C., Maguire, P., (2003). The psychological impact of cancer on patients' partners and other key relatives: a review. Eur J Cancer. 2003; 39:1517-1524.

Playdon, M., Ferrucci, L. M., McCorkle, R., Stein, K. D., Cannady, R., Sanft, T., & Cartmel, B. (2016). Health information needs and preferences in relation to survivorship care plans of long-term cancer survivors in the American Cancer Society's Study of Cancer Survivors-I. Journal of cancer survivorship: research and practice, 10(4), 674–685. https://doi.org/10.1007/s11764-015-0513-4

Polkinghorne, D. E., (1989). Phenomenological research methods. In R. S. Valle & S. Halling (Eds.), Existential-phenomenological perspectives in psychology (pp.41-60). New York: Plenum.

Pollak, K. I., R. M. Arnold, A. S. Jeffreys, S. C. Alexander, M. K. Olsen, A. P. Abernethy, C. Sugg Skinner, K. L. Rodriguez, and J. A. Tulsky. (2007). Oncologist communication about emotion during visits with patients with advanced cancer. Journal of Clinical Oncology 25(36):5748-5752.

Polit, D., & Beck, C., (2008). Nursing research: Generating and asses-sing evidence for nursing practice (Vol. 8.). Philadelphia, PA: Lippincott William and Wilkins

Polit, D. F., & Beck, C. T., (2012). Nursing research: Generating and assessing evidence for nursing practice. Lippincott Williams & Wilkins.



Polit, D. F., & Beck, C. T., (2014). Essentials of Nursing Research: Appraising Evidence for Nursing Practice, 8th ed. Philadelphia: Wolters Kluwer Health/Lippincott Williams & Wilkins, 533p. First published 2001.

Portenoy, R. K., & Itri, L. M., (1999). Cancer-Related Fatigue: Guidelines for Evaluation and Management. http://theoncologist.alphamedpress.org/content/4/1/1.full.pdf

Portney, L. G., & Watkins, M. P., (2009). Foundations of Clinical Research: Applications to Practice (3rd ed.). New Jersey: Pearson/Prentice Hall.

Post, D.M., Cegala, D.J., Marinelli, T.M. (2001). Teaching patients to communicate with physicians: The impact of race. Journal of National Medical Association, 93, 6-12.

Posma, E. R., van Weert, J. C., Jansen, J., & Bensing, J. M. (2009). Older cancer patients' information and support needs surrounding treatment: An evaluation through the eyes of patients, relatives and professionals. BMC nursing, 8, 1. doi:10.1186/1472-6955-8-1

Potter, J. & Wetherell, M., (1987). Discourse and social psychology: Beyond attitudes and behaviour. London: Sage

Prakash, B., (2010). Patient satisfaction. Journal of cutaneous and aesthetic surgery, 3(3), 151–155. https://doi.org/10.4103/0974-2077.74491

Prates, ACL., Freitas-Junior, R., Prates, M. F. O., Veloso, M. F., Barros, N. M., (2017). Influence of Body Image in Women Undergoing Treatment for Breast Cancer. Rev Bras Ginecol Obstet. 2017 Apr;39(4):175-183. English. doi: 10.1055/s-0037-1601453. Epub 2017 Mar 30. PMID: 28359110.

Preston, M. M., (2010). "An Exploration of Appearance-Related Issues of Breast Cancer Treatment on Sense of Self, Self-Esteem, and Social Functioning in Women with Breast Cancer" (2010). Doctorate in Social Work (DSW) Dissertations. 10. https://repository.upenn.edu/edissertations_sp2/10



Preat, F., Simon, P., Noel, J-C., (2014). Differences in breast carcinoma immunohistochemical subtypes between immigrant Arab and European women. Diagn. Pathol. 2014; 9:26. doi: 10.1186/1746-1596-9-26.

Prip, A., Møller, K. A., Nielsen, D. L., Jarden, M., Olsen, M. H., Danielsen, A. K., (2018). The Patient–Healthcare Professional Relationship and Communication in the Oncology Outpatient Setting. A Systematic Review. Cancer Nursing: September/October 2018 - Volume 41 - Issue 5 - p E11–E22. doi: 10.1097/NCC.00000000000533

Price, L. A., (2019). Breast cancer. Thriving through treatment to recovery. Chapter 2. Emotional aftermath and effects on the immune system. Springer publishing company.

Prue, G., Santin, O., Porter, S., (2015). Assessing the needs of informal caregivers to cancer survivors: a review of the instruments. Psycho-oncology. 2015; 24(2): 121–9. pmid:24930811. DOI: 10.1002/pon.3609

Puchalski, C, M., (2001). "The role of spirituality in health care." Proceedings (Baylor University. Medical Center) vol. 14,4 (2001): 352-7. doi:10.1080/08998280.2001.11927788

Qian-Xin, W., Yan, B., Guo-Fang, L., & Cai-Yun, Z., (2017). "Perceived Health-related Stigma among Patients with Breast Cancer." Chinese Nursing Research 4.4 (2017): 158-61. Web.

Rabey, N. G., Lie, K. H., Kumiponjera, D., Erel, E., Simcock, J. W., & Malata, C. M. (2013). Salvage of Failed Prosthetic Breast Reconstructions by Autologous Conversion with Free Tissue Transfers. Eplasty, 13, e32.

Radwan, H., Hasan, H., Ballout, R. A., Rizk, R., (2018). The epidemiology of cancer in the United Arab Emirates: A systematic review [published correction appears in Medicine (Baltimore). 2019 Jan;98(2):e14220]. Medicine (Baltimore). 2018;97(50):e13618. doi:10.1097/MD.00000000013618.

https://www.ncbi.nlm.nih.gov/pmc/articles/PMC6320101/

Radwin, L. E., & Alster, K., (2002). Individualized nursing care: an empirically generated definition. Int Nurs Rev. 49, 54–63.



Rainbird, K. J., Perkins, J. J., Sanson-Fisher, R. W., (2005). The Needs Assessment for Advanced Cancer Patients (NA-ACP): a measure of the perceived needs of patients with advanced, incurable cancer. a study of validity, reliability and acceptability. Psycho-Oncology 2005,14(4):297-306.

Rah, B., Ali, S., Dar, M. I., & Afroze, D., (2019). Breast Cancer: Management and Survivorship, Cancer Survivorship, Dil Afroze, IntechOpen, DOI: 10.5772/intechopen.82297. Available from: https://www.intechopen.com/books/cancer-survivorship/breast-cancer-management-and-survivorship

Rahnama, M., Khoshknab, M. F., Maddah, S. S., Ahmadi, F., (2012). Iranian cancer patients' perception of spirituality: a qualitative content analysis study. BMC Nursing. 2012 Oct 9; 11():19.

Ramirez- Perdomo, A. G., Gallion, K. J., Perez, A., Munoz, E., Long Parma, D., Moreno, P. I., & Penedo, F. J. (2019). Improving quality of life among latino cancer survivors: Design of a randomized trial of patient navigation. Contemporary clinical trials, 76, 41–48. https://doi.org/10.1016/j.cct.2018.11.002

Rao, A., & Cohen, H. J., (2004). Symptom Management in the Elderly Cancer Patient: Fatigue, Pain, and Depression. JNCI Monographs, Volume 2004, Issue 32, July 2004, Pages 150–157, https://doi.org/10.1093/jncimonographs/lgh031

Rasool, G. H. (2000). The crescent and Islam: healing, nursing and the spiritual dimension. Some considerations towards an understanding of the Islamic perspectives on caring. Journal of advanced nursing, 32, 1476-84.

Ratzan, S. C., (2009). Integrating Health Literacy into Primary and Secondary Prevention Strategies, A commissioned paper, the Institute of Medicine Roundtable on Health Literacy Washington, DC, Appendix C, in Promoting Health Literacy to Encourage Prevention and Wellness: Workshop Summary 2011. Retrieved on 17 July 2019, http://www.nap.edu/read/13186/chapter/13.

Ravichandran, K., Al-Hamdan, N. A., Mohamed, G., (2011). Knowledge, attitude and behaviour among Saudis towards cancer prevention practice. J Fam Commun Med 18:135-142



Ray, C., & Baum, M., (1985). Psychosocial aspects of early breast cancer.

Rechis, R., Beckjord, E. B., Nutt, S., (2014). Potential benefits of treatment summaries for survivors' health and information needs: results from a livestrong survey. J Oncol Pract. 2014; 10:75–8. doi: 10.1200/JOP.2013.000973.

Recio-Saucedo, A., Gerty S., Foster, C., Eccles, D., Cutress, R. I., (2016). Information requirements of young women with breast cancer treated with mastectomy or breast conserving surgery: A systematic review. The Breast. 2016 Feb; 25:1-13. doi: 10.1016/j.breast.2015.11.001.

Recklitis, C. J., & Syrjala, K. L. (2017). Provision of integrated psychosocial services for cancer survivors post-treatment. The Lancet. Oncology, 18(1), e39–e50. doi:10.1016/S1470-2045(16)30659-3

Reed, S., Kassis, K., Nagel, R., Verback, N., Mahan, J.D., & Shell, R. (2015). Breaking bad news is a teachable skill in pediatric residents: A feasibility study of an educational intervention. Patient Education and Counseling, 98, 748–752. doi: 10.1016/j.pec.2015.02.015

Regan, T. W., Lambert, S. D., Girgis, A., Kelly, B., Kayser, K., Turner, J., (2012). Do couple-based interventions make a difference for couples affected by cancer? A systematic review. BMC Cancer. 2012 Jul 6; 12:279. doi: 10.1186/1471-2407-12-279. PMID: 22769228; PMCID: PMC3464780.

Remennick, L., (2006). The challenge of early breast cancer detection among immigrant and minority women in multicultural societies. Breast Journal, January 2006, Vol.12, pp. S103-S110

Renneker, R., & Cutler, M., (1952). Psychological Problems of Adjustment to Cancer of the

Breast. JAMA, 148(10): 833-838.

Renz, M., Bueche, D., Reichmuth, O., Schuett, Mao. M., Renz, U., Siebenrock, R., Strasser, F., (2019). Forgiveness and Reconciliation Processes in Dying Patients with Cancer. Am J Hosp Palliat Care. 2020 Mar;37(3):222-234. doi: 10.1177/1049909119867675. Epub



2019 Aug 5. Erratum in: Am J Hosp Palliat Care. 2019 Oct 16;1049909119884407. PMID: 31382757.

Reicher, S., & Taylor, S. (2005). Similarities and differences between traditions. The Psychologist, 18(9), 547-549

Reyes-Gibby, C. C., Anderson, K. O., Morrow, P. K., Shete, S., & Hassan, S., (2012). Depressive symptoms and health-related quality of life in breast cancer survivors. Journal of Women's Health, 21(3), 311-318.

Rezaei, M., Elyasi, F., Janbabai, G., Moosazadeh, M., & Hamzehgardeshi, Z. (2016). Factors Influencing Body Image in Women with Breast Cancer: A Comprehensive Literature Review. Iranian Red Crescent Medical Journal, 18(10), e39465. http://doi.org/10.5812/ircmj.39465

Resnicow, K., Baranowski, T., Ahluwalia, J. S., & Braithwaite, R. L., (1999). Cultural sensitivity in public health: defined and demystified, Ethn Dis, 1999, vol. 9 (pg. 10-21).

Richardson, A., Sitzia, J., Brown, V., Medina, J., & Richardson, A., (2005). Patients' needs assessment tools in cancer care: Principles and practice. London, England: King's College London

Richardson, A., Medina, J., Brown, V. Sitzia, J., (2007). Patients' needs assessment in cancer care: a review of assessment tools. Support Care Cancer 15, 1125–1144 (2007). https://doi.org/10.1007/s00520-006-0205-8

Richardson, A., Addington-Hall, J., Amir, Z., Foster, C., Stark, D., Armes, J., Brearley, S. G., Hodges, L., Hook, J., Jarrett, N., Stamataki, Z., Scott, I., Walker, J., Ziegler, L., & Sharpe, M. (2011). Knowledge, ignorance and priorities for research in key areas of cancer survivorship: findings from a scoping review. British journal of cancer, 105 Suppl 1(Suppl 1), S82–S94. https://doi.org/10.1038/bjc.2011.425

Riker, A. I., (2015). Breast Disease Comprehensive Management. Pg 523 Springer



Ringdal, G., (1996). Religiosity, Quality of Life, and Survival in Cancer Patients. Social Indicators Research, 38(2), 193-211. Retrieved August 25, 2021, from http://www.jstor.org/stable/27522926

Ritchie, J., Lewis, J., & Elam, G. (2003). Designing and selecting samples. In Jane Ritchie & Jane Lewis (Eds.), Qualitative research practice: A guide for social science students and researchers (pp.77-108). Thousand Oaks, CA: Sage Publication

Robinson, J., & Elkan, R., (1996). Health needs Assessment: Theory and Practice. Churchill Livingstone.

Robson, C., (2002). Real world research: A resource for social scientists and practitioner-researchers(2ndedn.). Oxford, UK: Blackwell.

Rodgers, B. L., & Cowles, K. V., (1997). A conceptual foundation for human suffering in nursing care and research. J Adv Nurs. 1997; 25:1048-1053

Rodriguez, M. S., & Cohen, S. (1998). Social support. Encyclopedia of Mental Health, 3,535-544. Retrieved from http://www.psy.cmu.edu/~scohen/socsupchap98.pdf

Rodriguez, D. P., Fins, J. J., Helmy, I., El Chaki, R., El Shazly, T., Wafaradi, D., & Mahfoud, Z. (2012). Truth-Telling and Cancer Diagnoses: Physician Attitudes and Practices in Qatar. The Oncologist, 17(11), 1469–1474. http://doi.org/10.1634/theoncologist.2012-0128

Rolnick, S. J., Altschuler, A., Nekhlyudov, L., Elmore, J. G., Greene, S. M., Harris, E. L., Herrinton, L. J., Barton, M. B., Geiger, A. M., Fletcher, S. W., (2007). What women wish they knew before prophylactic mastectomy. Cancer Nurs. 2007 Jul-Aug;30(4):285-91; quiz 292-3. doi: 10.1097/01.NCC.0000281733. 40856.c4. PMID: 17666977.

Röing, M., Hirsch, J. M., Holmström, I., (2006). Ways of understanding the encounter with head and neck cancer patients in the hospital dental team--a phenomenographic study. Support Care Cancer. 2006 Oct;14(10):1046-54. doi: 10.1007/s00520-006-0043-8. Epub 2006 Mar 24. PMID: 16572314.



Rood, J., Lissenberg-Witte, B. I., Eeltink, C., Stam, F., van Zuuren, F. J., Zweegman, S., & Verdonck-de Leeuw, I. M. (2018). The need for information among patients with hematological malignancies: Psychometric analyses of the 62-item Hematology Information Needs Questionnaire (HINQ-62). PloS one, 13(8), e0201699. doi: 10.1371/journal.pone.0201699

Rose, K., (1994). Unstructured and semi-structured interviewing: the author explores the complexities of conducting these interviews, and their value to nursing research. Nurs Res. 1994;1(3):23Y32.

Rose, J. H., (1990). Social support and cancer: adult patients' desire for support from family, friends and health professionals. Am J Community Psychol. 1990; 18:439–64

Rosenberg, A. R., Starks, H., Unguru, Y., Feudtner, C., & Diekema, D. (2017). Truth Telling in the Setting of Cultural Differences and Incurable Pediatric Illness: A Review. JAMA pediatrics, 171(11), 1113–1119. doi:10.1001/jamapediatrics.2017.2568

Rosenberg, S., (2020). Overcome the Challenges of Cancer Care. How to avoid pitfalls on the path of healing. Chapter 7: Chemotherapy: Chemotherapy and side effects to consider. Page 62. Published by Rowman & Littlefield.

Rosedale, M., (2009). Survivor loneliness of women following breast cancer. Oncology Nursing Forum. 2009;36(2):175–183. DOI: 10.1188/09.ONF.175-183.

Rossman, G. B., & Wilson, B. L., (1985). Number and words: Combining quantitative and qualitative methods in a single large-scale evaluation study. Evaluation Review 9 (5): 627–43

Ross, C., Steward, C., Sinacore, J. M., (1993). The importance of patient preferences in the measurement of health care satisfaction. Medical Care. 1993; 31:1138–1149.

Roussi, P., Krikeli, V., Hatzidimitriou, C., Koutri, I., (2007). Patterns of coping, flexibility in coping and psychological distress in women diagnosed with breast cancer. Cogn Ther Res. 2007; 31:97–109.



Rovai, A. P., Baker, J. D., & Ponton, M. K. (2014). Social Science Research Design and Statistics. Chesapeake, VA: Watertree Press LLC.

Row, A., & Carey, L., (2004). The effect of population-based needs assessments on health visitor practice. Primary healthcare research and development, 5, 179-186.

Rowland, J. H., Hewitt, M., & Ganz, P. A., (2006). Cancer survivorship: A new challenge in delivering quality cancer care Journal of Clinical Oncology. 5101–5104

Rowland, H. H., Kent, E. E., Forsythe, L. P., Loge, J. H., Hjorth, L., Glaser, A., Mattioli, V., Fossa, S., D., (2013). Cancer survivorship research in Europe and the United States: Where have we been, where are we going, and what can we learn from each other? Cancer 119:2094–2108,2013 (suppl 11)

Rowland, J. H., & Bellizzi, K. M., (2008). Cancer survivors and survivorship research: a reflection on today's successes and tomorrow's challenges. Hematol Oncol Clin North Am. 2008 Apr;22(2):181-200, v. doi: 10.1016/j.hoc.2008.01.008. PMID: 18395144.

Rowland, J. H., & Bellizzi, K. M. (2014). Cancer survivorship issues: life after treatment and implications for an aging population. Journal of clinical oncology: official journal of the American Society of Clinical Oncology, 32(24), 2662–2668. https://doi.org/10.1200/JCO.2014.55.8361

Rubin, H., & Rubin, I., (2005). Qualitative Interviewing: The Art of Hearing Data, 2ndedition. London: Sage

Ruhnke, G. W., Tak, H. J., Meltzer, D. O., (2020). Association of Preferences for Participation in Decision-making with Care Satisfaction Among Hospitalized Patients. JAMA Netw Open. 2020;3(10): e2018766. doi:10.1001/jamanetworkopen.2020.18766

Rushton, L., (2014). What are the Barriers to Spiritual Care in a Hospital Setting? British Journal of Nursing 23: 370–74.

Rutten, L. J., Arora, N. K., Bakos, A. D., Aziz, N., Rowland, J., (2005). Information needs and sources of information among cancer patients: a systematic review of research (1980–



2003). Patient Education and Counseling, 2005, Vol.57(3), pp.250-261. DOI: 10.1016/j.pec.2004.06.006.

Russo, S., Jongerius, C., Faccio, F., Silvia F.M.Pizzoli, S. F. M., Pinto, C. A., Veldwijk, J., Janssens, R., Simons, G., Falahee, M., de Bekker-Grob, Huys, I., Postmus, D., Kihlbom, U., Pravettoni, G., (2019). Understanding Patients' Preferences: A Systematic Review of Psychological Instruments Used in Patients' Preference and Decision Studies. Value in Health. Volume 22, Issue 4, April 2019, Pages 491-501. https://doi.org/10.1016/j.jval.2018.12.007

Ryan, D. A., Gallagher, P., Wright, S., Cassidy, E. M., (2011). Sensitivity and specificity of the Distress Thermometer and a two-item depression screen (Patient Health Questionnaire-2) with a 'help' question for psychological distress and psychiatric morbidity in patients with advanced cancer [published online ahead of print September 15, 2011]. Psychooncology.

Ryan, F., Coughlan, M., & Cronin, P. (2007). Step-by-step guide to critiquing research. Part 2: Qualitative research. British Journal ofNursing,16, 738–744. doi:10.12968/bjon.2007.16.12.23726

Ryan, G., & Bernard, H. (2000). Data management and analysis methods. In Denzin, N., Lincoln, Y. (Eds.), Handbook of qualitative research (2nd ed., pp. 769–802). Thousand Oaks, CA: Sage.

Sabiston, C. M., Wrosch, C., Fong, A. J., Brunet, J.; Gaudreau, P.; O'Loughlin, J.; Meterissian, S., (2018). Life after breast cancer: moving on, sitting down or standing still? A prospective study of Canadian breast cancer survivors BMJ Open 2018;8: e021770. doi: 10.1136/bmjopen-2018-021770

Saca-Hazboun, H., (2018). Knowledge and Health Beliefs about Breast Cancer Screening among Rural Palestinian Women: Pilot Study. https://sigma.nursingrepository.org/bitstream/handle/10755/20805/SacaHazboun_Dissertatio n.pdf?sequence=1&isAllowed=y

Saeed, S., Asim, M. & Sohail, M. M., (2021). Fears and barriers: problems in breast cancer diagnosis and treatment in Pakistan. BMC Women's Health 21, 151 (2021). https://doi.org/10.1186/s12905-021-01293-6



Safman, R. M., & Sobal, J., (2004). Qualitative sample extensiveness in health education research, Health education and behaviour, 31 pp. 9-21

Saha, S., Beach, M. C., & Cooper, L. A. (2008). Patient centeredness, cultural competence and healthcare quality. *Journal of the National Medical Association*, *100*(11), 1275–1285. https://doi.org/10.1016/s0027-9684(15)31505-4

Salem, H, & Daher-Nashif, S., (2020). Psychosocial Aspects of Female Breast Cancer in the Middle East and North Africa. Int. J. Environ. Res. Public Health 2020, 17, 6802; doi:10.3390/ijerph17186802 www.mdpi.com/journal/ijerph

Salem, A., & Salem, A-F., (2013). Breaking Bad News: Current Prospective and Practical Guideline for Muslim Countries. J Canc Educ (2013) 28:790–794DOI 10.1007/s13187-013-0523-8. Springer Science+Business Media New York 2013

Salonen, M. K., Kajantie, E., Osmond, C., Forsén, T., Ylihärsilä, H., Paile-Hyvärinen, M., ... Eriksson, J. G. (2009). Role of Socioeconomic Indicators on Development of Obesity from a Life Course Perspective. Journal of Environmental and Public Health, 2009, 625168. http://doi.org/10.1155/2009/625168

Sale, J. E. M., Lohfeld, L. H., Brazil, K., (2002). Revisiting the quantitative-qualitative debate: implications for mixed methods research. Quality and Quantity 36, 43–53.

Saldaña, J. (2016). The coding manual for qualitative researchers (3rd ed.). London, UK: Sage Publications Ltd.

Samdahl, D. M., (1999). Differences between quantitative and qualitative research. In Jackson, E. J., & Burton (Eds), T. L. Lesire studies: Prospects for the 21st century (pp. 119-133). State College, PA: Venture Publ.

Samson, A., & Zerter, B., (2003). The experience of spirituality in the psycho-social adaptation of cancer survivors. Journal of Pastoral Care Counseling. 2003 Fall; 57(3):329-43.

Sandelowski, M. (1995). Sample size in qualitative research. Research in Nursing & Health, 18, 179-183



Sanders, S. L., Bantrum, E. O., Owen, J. E., Thornton, A. A., Stanton, A. L., (2010). Supportive care needs in patients with lung cancer. Psycho-Oncol 19: 480–489. Psycho oncology. 2010 May;19(5):480-9. doi: 10.1002/pon.1577.

Sanson-Fisher, R., Girgis, A., Boyes, A., Bonevski, B., Burton, L., & Cook, P., (2000). The unmet supportive care needs of patients with cancer. Supportive Care Review Group. Cancer. 2000 Jan 1;88(1):226-37. doi: 10.1002/(sici)1097-0142(20000101)88:1<226: aid-cncr30>3.3.co;2-g. PMID: 10618627

Sarafis, P., Tsounis, A., Malliarou, M., & Lahana, E. (2013). Disclosing the truth: a dilemma between instilling hope and respecting patient autonomy in everyday clinical practice. Global journal of health science, 6(2), 128–137. doi:10.5539/gjhs.v6n2p128

Sarwer, D. B., Spitzer, J. C., (2012). Body image dysmorphic disorder in persons who undergo aesthetic medical treatments. Aesthet Surg J. 2012 Nov;32(8):999-1009. doi: 10.1177/1090820X12462715. Epub 2012 Sep 26. PMID: 23015692.

Saslow, D., Solomon, D., Lawson, H. W., Killackey, M., Kulasingam, S. L., Cain, J., Garcia, F. A., Moriarty, A. T., Waxman, A. G., Wilbur, D. C., (2012). American cancer society, american society for colposcopy and cervical pathology, and american society for clinical pathology screening guidelines for the prevention and early detection of cervical cancer. CA Cancer J Clin. 2012;62(3):147–172. doi: 10.3322/caac.21139.

Sato, F., (2005). Information provision structures of physician regarding surgical option for breast cancer patients and attitudes of patients toward those structures, Jpn J Health Behav Sci, 2005, vol. 20 (pg. 132-45) (in Japanese with English abstract).

Savage, J., (2000). One voice, different tunes: Issues raised by dual analysis of a segment of qualitative data. Journal of AdvancedNursing,31, 1493–1500. doi:10.1046/j.1365-2648.2000. 01432.x

Savina, S., & Zaydiner, B. (2019). Cancer-Related Fatigue: Some Clinical Aspects. Asia-Pacific journal of oncology nursing, 6(1), 7–9. doi: 10.4103/apjon.apjon_45_18



Schlairet, M., Heddon, M. A., Griffis, M., (2010). Piloting a needs assessment to guide development of a survivorship program for a community cancer center. Oncol Nurs Forum. 2010 Jul;37(4):501-8. doi: 10.1188/10.ONF.501-508. PMID: 20591810.

Schulz, R., Beach, S., (1999). Caregiving as a risk factor for mortality. Journal of the American Medical Association. 1999; 282:2215–2219. doi: 10.1001/jama.282.23.2215.

Schnoll, R. A; Malstrom, Michael; James, Calvin; Rothman, Randi L; Miller, Suzanne
M; Ridge, John A; Movsas, Benjamin; Unger, Michael; Langer, Corey; Goldberg,
Melvyn., (2002). Ireland: Elsevier Ireland Ltd. Patient education and counseling, 2002, Vol.46
(2), p.137-145

Schiavon, G., & Smith, I. E. (2014). Status of adjuvant endocrine therapy for breast cancer. Breast Cancer Research: BCR, 16(2), 206. http://doi.org/10.1186/bcr3636

Schmid-Buchi, S., Halfens, R. J. G., Dassen, T., & van den Borne, B. (2008). A review of psychosocial needs of breast-cancer patients and their relatives. Journal of Clinical Nursing, 17, 2895–2909

Schmidt, N. A., Brown, J., (2015). Evidence-Based Practice for Nurses: Appraisal and Application of Research. 3rd ed. Burlington, MA: Jones and Bartlett; 2015.

Schmitz, K. H., Stout, N. L., Andrews, K., Binkley, J. M. and Smith, R. A. (2012). Prospective evaluation of physical rehabilitation needs in breast cancer survivors. Cancer, 118: 2187–2190. doi:10.1002/cncr.27471

Schmitz, K. H., Neuhouser, M. L., Agurs-Collins, T., Zanetti, K. A., Cadmus-Bertram, L., Dean, L. T., Drake, B. F., (2013). Impact of Obesity on Cancer Survivorship and the Potential Relevance of Race and Ethnicity, JNCI: Journal of the National Cancer Institute, Volume 105, Issue 18, 18 September 2013, Pages 1344–1354, https://doi.org/10.1093/jnci/djt223

Schnur, J. B., Ouellette, S. C., Dilorenzo, T. A., Green, S., Montgomery, G. H., (2011). A qualitative analysis of acute skin toxicity among breast cancer radiotherapy patients.



Psychooncology. 2011 Mar;20(3):260-8. doi: 10.1002/pon.1734. PMID: 20238306; PMCID: PMC3729017.

Schoemaker, M. J., Jones, M. E., Wright, L. B., Griffin, J., McFadden, E., Ashworth, A., & Swerdlow, A. J. (2016). Psychological stress, adverse life events and breast cancer incidence: a cohort investigation in 106,000 women in the United Kingdom. Breast cancer research: BCR, 18(1), 72. doi:10.1186/s13058-016-0733-1

Schouten, B., Avau, B., Bekkering, G. T. E., Vankrunkelsven. P., Mebis, J., Hellings, J., Van Hecke, A., (2019). Systematic screening and assessment of psychosocial well-being and care needs of people with cancer. Cochrane Database Syst Rev. 2019 Mar 26;3:CD012387. doi: 10.1002/14651858.CD012387.pub2.

Schover, L. R., van der, Kaaij, M., van Dorst, E., Creutzberg, C., Huyghe, E., & Kiserud, C. E. (2014). Sexual dysfunction and infertility as late effects of cancer treatment. EJC supplements: EJC: official journal of EORTC, European Organization for Research and Treatment of Cancer ... [et al.], 12(1), 41–53. doi: 10.1016/j.ejcsup.2014.03.004

Schover, L. R., (1999). Counseling Cancer Patients about Changes in Sexual Function. Oncology (Williston Park). 1999 Nov;13(11):1585-91; discussion 1591-2, 1595-6. https://www.ncbi.nlm.nih.gov/pubmed/10581604'

Scotland, J., (2012). Exploring the Philosophical Underpinnings of Research: Relating Ontology and Epistemology to the Methodology and Methods of the Scientific, Interpretive, and Critical Research Paradigms. English Language Teaching; Vol. 5, No. 9; 2012. ISSN 1916-4742 E-ISSN 1916-4750. Published by Canadian Center of Science and Education doi:10.5539/elt.v5n9p9

Scotté, F., Bossi, P., Carola, E., Cudennec, T., Dielenseger, P., Gomes, F., Knox, S., Strasser, F., (2018). Addressing the quality of life needs of older patients with cancer: a SIOG consensus paper and practical guide, Annals of Oncology, Volume 29, Issue 8, August 2018, Pages 1718–1726, https://doi.org/10.1093/annonc/mdy228



Shea-Budgell, M. A., Kostaras, X., Myhill, K. P., & Hagen, N. A., (2014). Information needs and sources of information for patients during cancer follow-up. Current oncology (Toronto, Ont.), 21(4), 165–173. doi:10.3747/co.21.1932

Shulman, L. N., Ferris, L., Takanishi, D. M., & McKellar, D., (2015). Treatment Summaries and Survivorship Care Plans: The Approach by the Commission on Cancer to Increase Use. Journal of Oncology Practice 2015 11:1, 40-41. DOI: 10.1200/JOP.2014.002329

Seale, C., (1999). Quality in qualitative research. Qual Inquiry. 1999;5(4):465<478.

Seale, C., (2000). Using computers to analyse qualitative data. In D. Silverman (Ed.), Doing qualitative research: A practical handbook (pp. 155-174). London: Sage.

Seedhom, E. A., & Kamal, N. N., (2011). Factors Affecting Survival of Women Diagnosed with Breast Cancer in El-Minia Governorate, Egypt. International Journal of Preventive Medicine V.2 (3): 131–138. PMCID: PMC3143525. Available from: http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3143525/

Seidman, I. E., (1991). Interviewing as qualitative research. New York: Teacher College Press.

Segrin, C., T. A. Badger, P. Meek, A. M. Lopez, E. Bonham, & A. Sieger. (2005). Dyadic interdependence on affect and quality-of-life trajectories among women with breast cancer and their partners. Journal of Social and Personal Relationships 22(5):673–689.

Semin, J. N., Palm, D., Smith, L. M. et al. (2020). Understanding breast cancer survivors' financial burden and distress after financial assistance. Support Care Cancer 28, 4241–4248 (2020). https://doi.org/10.1007/s00520-019-05271-5

Senkus, E., & Jassem, J., (2011). Cardiovascular effects of systemic cancer treatment. Cancer Treat Rev. 2011 Jun; 37(4):300-11.

Sertöz O, Elbi HM, Noyan A, Alper M, Kapkac M. (2004). Effects of surgery type on body image, sexuality, self-esteem, and marital adjustment in breast cancer: a controlled study. Türk Psikiyatri Derg. 2004;15(4):264–75.



Setyowibowo, H., Purba, F. D., Hunfeld, JAM., Iskandarsyah, A., Sadarjoen, S. S., Passchier, J., & Sijbrandij, M., (2018). Quality of life and health status of Indonesian women with breast cancer symptoms before the definitive diagnosis: A comparison with Indonesian women in general. PLoS ONE 13(7): e0200966. https://doi.org/10.1371/journal.pone.0200966

Seyfried, T. N., & Huysentruyt, L. C. (2013). On the origin of cancer metastasis. Critical reviews in oncogenesis, 18(1-2), 43–73. https://doi.org/10.1615/critrevoncog.v18.i1-2.40

Seyfried, T., (2012). Cancer As a Metabolic Disease: On the Origin, Management, and Prevention of Cancer. Chapter 1: Images of cancer. https://ebookcentral-proquest-com.salford.idm.oclc.org/lib/salford/reader.action?docID=861782

Shahidi, J., Taghizadeh-Kermani, A., Gohari, M. R., Ghavamnasiri, M. R., Khoshroo, F., Pourali, L., & Cohen, S. R. (2014). Changes in daily activities of cancer patients after diagnosis: how do canadian and Iranian patients perceive the change? Iranian journal of cancer prevention, 7(1), 28–34.NLM

Shahid, S., Finn, L., Bessarab, D., Thompson, S. C., (2011). 'Nowhere to room ... nobody told them': logistical and cultural impediments to Aboriginal peoples' participation in cancer treatment. Aust Health Rev. 2011 May;35(2):235-41. doi: 10.1071/AH09835. PMID: 21612740.

Shahid, S., Teng, T. H., Bessarab, D., Aoun, S., Baxi, S., & Thompson, S. C. (2016). Factors contributing to delayed diagnosis of cancer among Aboriginal people in Australia: a qualitative study. BMJ open, 6(6), e010909. https://doi.org/10.1136/bmjopen-2015-010909

Shahrokni, A., Kim, S. J., Bosl, G. J., & Korc-Grodzicki, B., (2017). How we care for an older patient with cancer. Journal of oncology practice vol. 13,2 (2017): 95-102. doi:10.1200/JOP.2016.017608.

Shane W Pascoe, S. W., Neal, R. D., Allgar, V. L., Selby, P. J., Wright, E. P., (2004). Psychosocial care for cancer patients in primary care? Recognition of opportunities for cancer care, Family Practice, Volume 21, Issue 4, August 2004, Pages 437–442, https://doi.org/10.1093/fampra/cmh415



Shannon-Baker, P., (2015) Making paradigms meaningful inmixed methods research. Journal of Mixed Methods Research, Epub ahead of print 20 March 2015. DOI:10.1177/1558689815575861.

Shankar, A., Prasad, N., Roy, S. h., Chakraborty, A., Biswas, A., Patil, J., & Rath, G. K., (2017). Sexual Dysfunction in Females after Cancer Treatment: An Unresolved Issue. Asian Pacific journal of cancer prevention: APJCP, 18(5), 1177–1182. doi:10.22034/APJCP.2017.18.5.1177

Sharif, A. A., & Blair, I., (2011). The role of the hospital in the changing landscape of UAE health care: a focus on Dubai. World Hospitals & Health Services, 47(3), 11-3.

Sharma, P., Goswami, M., & Sharma, S., (2019). Needs of Cancer Patients Receiving Chemotherapy and Radiation Therapy: A Narrative Review. International Journal of Oncological Nursing, 2019; 5(2): 6-11p.

Sharma, A. D., Tiwari, G. K., Rai, P. K., (2020). 9The Spiritual Beliefs of Cancer Survivors: A Thematic Analysis. Psychology, Community & Health, [S.1.], v. 8, n. 1, p. 139-157, apr. 2020. ISSN 2182-438X. Available at: https://pch.psychopen.eu/article/view/290/241. Date accessed: 25 aug. 2021. doi: http://dx.doi.org/10.5964/pch.v8i1.290.

Sharp, L., & Carsin, A. E., (2013). Timmons A. Association between cancer-related financial stress and strain and psychological well-being among individuals living with cancer. PsychoOncology. 2013; 22:745-55.

Shapiro, D., Lasker, R., Bindman, A., Lee, P., (1993). Containing costs while improving quality of care: the role of profiling and practice guidelines. Annu Rev Public Health. 1993;14(1):219–241. doi: 10.1146/annurev.pu.14.050193.001251.

Shapiro, 2018. Cancer Survivorship. N Engl J Med 2018; 379:2438-2450 DOI: 10.1056/NEJMra1712502

Shaw, R., (2001). Why use interpretative phenomenological analysis in health psychology? Health Psychology Update, 10, (4), 48-52



Shaw, J., Butow, P., Sze, M., Young, J., & Goldstein, D., (2013). Reducing disparity in outcomes for immigrants with cancer: a qualitative assessment of the feasibility and acceptability of a culturally targeted telephone-based supportive care intervention. Support Care Cancer. 2013 Aug;21(8):2297-301. doi: 10.1007/s00520-013-1786-7.

Shaw, J., Zou, X., & Butow, P., (2015). Treatment decision making experiences of migrant cancer patients and their families in Australia. Patient Education & Counseling. 2015 Jun; 98(6):742-7. DOI: 10.1016/j.pec.2015.01.012.

Sheldon, L.K., Ellington, L., Barrett, R, Dudley, W.N., Clayton, M.F.& Rinaldi, K., (2009). Nurse responsiveness to cancer patient expressions of emotion. Patient Education and Counseling 76(1): 63–70.

Shelton, J. D., (2018). At Last! Universal Health Coverage That Prioritizes Health Impact: The Latest Edition of Disease Control Priorities (DCP3). Global Health: Science and Practice June 2018, 6(2):232-236; https://doi.org/10.9745/GHSP-D-18-00193.

Shell, J.A., & Kirsch, S., (2001). Psychosocial issues, outcomes, and quality of life. In S.E. Otto (Ed.), Oncology nursing (4th ed., pp. 948–972). St. Louis, MO: Mosby

Shen, Y., Dong, W., Esteva, F. J., (2007). Are there racial differences in breast cancer treatments and clinical outcomes for women treated at M.D. Anderson Cancer Center? Breast Cancer Research & Treatment. 2007; 102(3):347–356. DOI: 10.1007/s10549-006-9337-2

Shea–Budgell, M. A., Kostaras, X., Myhill, K. P., Hagen, N. A., (2014). Information needs and sources of information for patients during cancer follow-up. Rehabilitation and Survivorship, Current oncology (Toronto, Ont.), August 2014, Vol.21(4), pp.165-73. DOI: http://dx.doi.org/10.3747/co.21.1932

Sheppard, V.B., Christopher, J., & Nwabukwu, I. (2010). Breaking the silence barrier: Opportunities to address breast cancer in African-born women. Journal of the National Medical Association, 102, 461–468.



Shilling, V., Jenkins, V., Fallowfield, L., (2003). Factors affecting patient and clinician satisfaction with the clinical consultation: can communication skills training for clinicians improve satisfaction? Psychooncology 12:599–611. doi:10.1002/pon.731

Shim, E. J., Shin, Y. W., Jeon, H. J., Hahm, B. J., (2008). Distress and its correlates in Korean cancer patients: pilot use of the distress thermometer and the problem list. Psychooncology. 2008 Jun;17(6):548-55. doi: 10.1002/pon.1275. PMID: 17957764.

Shim E-J, Mehnert A, Koyama A, Cho S-J, Inui H, Paik N-S, et al. (2006). Health-related quality of life in breast cancer: A cross-cultural survey of German, Japanese, and South Korean patients. Breast Cancer Res Treat. 2006;99(3):341–50.

Shirazi, M., Shirazi, A., Bloom, J., (2013). Developing a Culturally Competent Faith-Based Framework to Promote Breast Cancer Screening Among Afghan Immigrant Women. Journal of Religion & Health. 2013 DOI: 10.1007/s10943-013-9793-z

Shorten, A., & Smith, J., (2017). Mixed methods research: expanding the evidence base. Research made Simple. Evid Based Nurs July 2017, volume 20, number 3. DOI: 10.1136/eb-2017-102699. https://ebn.bmj.com/content/ebnurs/20/3/74.full.pdf

Silbermann, M., Pitsillides, B., Al-Alfi, N., Omran, S., Al-Jabri, K., Elshamy, K., Ghrayeb, I., Livneh, J., Daher, M., Charalambous, H., Jafferri, A., Fink, R., & El-Shamy, M., (2013). Multidisciplinary care team for cancer patients and its implementation in several Middle Eastern countries, Annals of Oncology, Volume 24, Supplement 7, 2013, Pages vii41-vii47, ISSN 0923-7534, https://doi.org/10.1093/annonc/mdt265.

Silbermann, M., Arnaout, M., Daher, M., Nestoros, S., Pitsillides, B., Charalambous, H., Gultekin, M., Fahmi, R., Mostafa, K. A., Khleif, A. D., Manasrah, N., & Oberman, A., (2012). Palliative cancer care in Middle Eastern countries: accomplishments and challenges. Ann Oncol. 2012 Apr;23 Suppl 3(Suppl 3):15-28. doi: 10.1093/annonc/mds084. PMID: 22628412; PMCID: PMC3493144.

Siemsen, I., (2011). Patientovergange - Et eksplorativt studie af faktorer der påvirker sikkerheden af patientovergange [In Danish]. PhD Thesis. 2011, Technical University of Denmark, Department of Management Engineering



Silistraru, I., (2018). Shame and Humiliation of Breast Cancer Patients –CommunicationPitfallswithOncologyCases.http://cis01.central.ucv.ro/revistadestiintepolitice/files/numarul59_2018/11.pdf. Published by:Editura Universitaria Craiova

Silveira, M., EiPiero, A., Gerrity, M., Geudtner, C., (2000). Patients' knowledge of options at the end of life. Journal of the American Medical Association. 2000; 284:2483–2488. doi: 10.1001/jama.284.19.2483.

Silverman, D., (2001). Interpreting Qualitative Data. London: Sage.

Sines, D., Saunders, M., & Forbes-Burford, J., (2009). Community Healthcare Nursing. Chapter 5: Health needs assessment and the community nurse

Singer, S., Blettner, M., Kreienberg, R., Janni W, Wöckel A, Kühn T, Felberbaum R, Flock F, Schwentner L (2015). Breast Cancer Patients' Fear of Treatment: Results from the Multicenter Longitudinal Study BRENDA II. Breast Care 2015; 10:95-100. doi: 10.1159/000381933

Singh–Carlson, S., Wong, F., Martin, L., Nguyen, S. K. A., (2013). Breast cancer survivorship and South Asian women: understanding about the follow-up care plan and perspectives and preferences for information post treatment. Current Oncology. 2013 Apr; 20(2): e63–e79. DOI: 10.3747/co.20.1066

Skalla, K. A., & Ferrell, B., (2015). Challenges in Assessing Spiritual Distress in Survivors of Cancer. Clinical Journal of Oncology Nursing 19: 99–104

Skarstein, J., Aass, N., Fosså, S. D., Skovlund, E., Dahl, A. A., (2000). Anxiety and depression in cancer patients: relation between the hospital anxiety and depression scale and the European organization for research and treatment of cancer core quality of life questionnaire. J. Psychosom. Res., 49 (2000), pp. 27-34

Skrzypczak, M., Czerniak, U., & Laski, P. (2012). Selected elements of socio-demographic status and lifestyle as factors determining subjective assessment of life in women after



mastectomy. Contemporary oncology (Poznan, Poland), 16(6), 569–575. doi:10.5114/wo.2012.32492

Slade, M., Leese, M., Ruggeri, M., Kuipers, E., Tansella, M., & Thornicroft, G., (2004). Does meeting needs improve quality of life? Psychotherapy and Psychosomatics, 73, 183-189. doi:10.1159/000076456

Smith, J. A., Flowers, P., & Larkin, M., (2009). Interpretative phenomenological analysis: Theory, method and research. Los Angeles, CA: SAGE

Smith, K. C., Klassen, A. C., Coa, K. I., Hannum, S. M., (2016). The salience of cancer and the "survivor" identity for people who have completed acute cancer treatment: a qualitative study. J Cancer Surviv. 2016 Jun; 10(3):457-66.

Smith, S. L., Wai, E. S., Alexander, C., & Singh-Carlson, S., (2011). Caring for survivors of breast cancer: perspective of the primary care physician. Current oncology (Toronto, Ont.), 18(5), e218–e226. https://doi.org/10.3747/co.v18i5.793

Snowden, A., Young, J., White, C., Murray, E., Richard, C., Lussier, M-T., Macarthur, E., Storey, D., Schipani, S., Wheatley, D., Mcmahon, J., Ross, E., (2015). Evaluating holistic needs assessment in outpatient cancer care—a randomised controlled trial: the study protocol. British Medical Journal Publishing Group BMJ Open, Vol.5(5). DOI:10.1136/bmjopen-2014-006840

Snyder, K. A., & Pearse, W. (2010). Crisis, social support, and the family response: exploring the narratives of young breast cancer survivors. Journal of psychosocial oncology, 28(4), 413–431. doi:10.1080/07347332.2010.484830

Sobczak, K., Leoniuk, K., & Janaszczyk, A. (2018). Delivering bad news: patient's perspective and opinions. Patient preference and adherence, 12, 2397–2404. doi:10.2147/PPA.S183106. https://www.dovepress.com/delivering-bad-news-patients-perspective-and-opinions-peer-reviewed-fulltext-article-PPA



Søndergaard, E. G., Grøne, B. H., Wulff, C. N., Larsen, P. V., & Søndergaard, J., (2013). A survey of cancer patients' unmet information and coordination needs in handovers--a cross-sectional study. MC Res Notes. 2013 Sep 25; 6:378. DOI: 10.1186/1756-0500-6-378.

So, Vivian H.T., Channon, Andrew A., Ali, Mohamed M., Merdad, Leenad., Al Sabahi, Sultanag., Al Suwaidi, Hudah., Al Ajeel, Abdullahi., Osman, Nabile., Khoja, Tawfik A.M., (2019). Uptake of breast and cervical cancer screening in four Gulf Cooperation Council countries. European Journal of Cancer Prevention: September 2019 - Volume 28 - Issue 5 - p 451-456. doi: 10.1097/CEJ.000000000000466

Soothill, K., Morris, S. M., Harman, J., Francis, B., Thomas, C., & McIllmurray, M., (2001). The significant unmet needs of cancer patients: probing psychosocial concerns. Support Care Cancer 9: 597–605.

Soothill, K., Morris, S. M., Thomas, C., Harman, J. C, Francis, B. & McIllmurray, M. B., (2003). The universal, situational, and personal needs of cancer patients and their main carers. Eur J Oncol Nurs 7, 5–13

Soskolne, V., Marie, S; & Manor, O., (2007). Beliefs, recommendations and intentions are important explanatory factors of mammography screening behavior among Muslim Arab women in Israel. Health Education Research, Volume 22, Issue 5, 1 October 2007, Pages 665–676, https://doi.org/10.1093/her/cyl132

Soylar, P & Genc, M. F., (2016). The Treatment Approach and Social Support Needs for Patients with Breast Cancer. The journal of breast health, April 2016, Vol.12(2), pp.56-62. DOI: 10.5152/tjbh.2016.2753

Sprague, B. L., Trentham-Dietz, A., Gangnon, R. E., Ramchandani, R., Hampton, J. M., Robert, S. A., Remington, P., Newcomb, P. A., (2011). Socioeconomic status and survival after an invasive breast cancer diagnosis. Cancer, 117(7), 10.1002/cncr.25589. http://doi.org/10.1002/cncr.25589

Spiegel, D., (1994). Health caring: psychosocial support for patients with cancer. Cancer 1994; 74:1453–7.



Spiegel, D., & Giese-Davis, J., (2003). Depression and cancer: mechanisms and disease progression. Biological Psychiatry, 2003, Vol.54(3), pp.269-282.

Spittler, C. A., Pallikathayil, L., & Bott, M., (2012). Exploration of how women make treatment decisions after a breast cancer diagnosis [Online exclusive]. Oncology Nursing Forum, 39, E425–E433. doi: 10.1188/12.ONF. E425-E433

Spronk, I., Korevaar, J. C., Schellevis, F. G., Albreht, T., & Burgers, J. S. (2017). Evidence-based recommendations on care for breast cancer survivors for primary care providers: a review of evidence-based breast cancer guidelines. BMJ open, 7(12), e015118. https://doi.org/10.1136/bmjopen-2016-015118

Squiers, L., Finney, R. L. J., Treiman, K., Bright, M. A., & Hesse, B., (2005). Cancer patients' information needs across the cancer care continuum: evidence from the cancer information service. Journal of health communication,2005;10 (suppl 1):15–34. doi: 10.1080/10810730500263620

Sriwatanakul, K., Kelvie, W., Lasagna, L., Calimlim, J. F., Weis, O. F., Mehta, G., (1983). Studies with different types of visual analog scales for measurement of pain. Clin Pharmacol Ther. 1983 Aug;34(2):234-9. doi: 10.1038/clpt.1983.159. PMID: 6872418.

Stanton, A. L., Estes, M. A., Estes, N. C., Cameron, C. L., (1998). Danoff-Burg S, Irving LM. Treatment decision making and adjustment to breast cancer: a longitudinal study, J Consult Clin Psychol, 1998, vol. 66 (pg. 313-22)

Stanton, A. L., (2005). How and for whom? Asking questions about the utility of psychosocial interventions for individuals diagnosed with cancer. Journal of Clinical Oncology 23(22):4818–4820.

Stanton, A. L., Rowland, J. J., & Ganz, P. A., (2015). Life After Diagnosis and Treatment of Cancer in Adulthood Contributions from Psychosocial Oncology Research. February– March 2015 • American Psychologist © 2015 American Psychological Association 0003-066X/15/\$12.00 Vol. 70, No. 2, 159–174 http://dx.doi.org/10.1037/a0037875


Stebbins, R. A., (2001). Exploratory research in the social sciences [Kindle edition]. Thousand Oaks, CA: Sage.

Steckler, A., McLeroy, K. R., Goodman, R. M., Bird, S. T., McCormick, L., (1992). Toward integrating qualitative and quantitative methods: an introduction. Health Education Quarterly, 19 (1) (1992), pp. 1-8

Stephenson, P., Draucker, C., & Martsolf, D., (2003). The experience of spirituality in the lives of hospice patients. Journal of Hospice and Palliative Care Nursing, 5 (1) (2003), pp. 51-58

Stiefel, F., (2006). Communication in cancer care. Published by Springer

Stoll, B. A., (1986). Coping with Cancer Stress: With an Introduction by Avery D. Weissman. Chapter 7: Denial in cancer patients pg 66. Clinical Journal of Oncology Nursing. Vol.18 Suppl:15-22. DOI: 10.1188/14.CJON.S1.15-22.

Straker, N., (2013). Facing Cancer and the fear of death: A psychoanalytic perspective on treatment. Published by Jason Aronson

Strauss, A., & Corbin. J., (1990). Basics of Qualitative Research: Grounded Theory Principles Procedures and Techniques. Sage: USA.

Stein, K. D., Syrjala, K. L., & Andrykowski, M. A., (2008), Physical and psychological longterm and late effects of cancer. Cancer, 112: 2577–2592. doi:10.1002/cncr.23448

Stricker, C. T., & O'Brien, M., (2014). Implementing the commission on cancer standards for survivorship care plans.

Strømmen-Bakhtiar, A. (2020). Introduction to Digital Transformation and its Impact on Society. Informing Science press

Stuckey, H. L. (2014). The first step in data analysis: Transcribing and managing qualitative research data. Journal of Social Health and Diabetes, 2(1), 6–6. doi:10.4103/2321-0656.120254



Stump, T. K, Eghan, N., Egleston, B. L, Hamilton, O., Pirollo, M., Schwartz, J. S., Armstrong, K., Beck, J. R., Meropol, N. J., Wong, Y., (2013). Journal of Oncology Practice. 2013. Cost concerns of patients with cancer.

Suchman, A. L., Markakis, K., Beckman, H. B., Frankel, R., (1997). A model of empathic communication in the medical interview. Journal of the American Medical Association. 1997; 277:678–682. doi: 10.1001/jama.277.8.678.

Sullivan, R., Peppercorn, J., Sikora, K., Zalcberg, J., Meropol N. J., Amir, E., Khayat, D., Boyle, P., Autier, P., Tannock, I. F., Fojo, T., Siderov, J., Williamson, S., Camporesi, S., McVie, J. G., Purushotham, A. D., Naredi, P., Eggermont, A., Brennan, M. F., Steinberg, M. L., De Ridder, M., McCloskey, S. A., Verellen, D., Roberts, T., Storme, G., Hicks, R. J., Ell, P. J., Hirsch, B. R., Carbone, D. P., Schulman, K. A., Catchpole, P., Taylor, D., Geissler, J., Brinker, N. G., Meltzer, D., Kerr, D., Aapro, M., (2011). Delivering affordable cancer care in high-income countries. Lancet Oncology. 2011;12(10):933–980.

Sun, H., Yang, Y., Zhang, J., Liu, T., Wang, H., Garg, S., & Zhang, B. (2019). Fear of cancer recurrence, anxiety and depressive symptoms in adolescent and young adult cancer patients. Neuropsychiatric disease and treatment, 15, 857–865. https://doi.org/10.2147/NDT.S202432

Sun, W., Chen, K., Terhaar, A., Wiegmann, D. A., Heidrich, S. M., Tevaarwerk, A. J., & Sesto, M. E. (2016). Work-related barriers, facilitators, and strategies of breast cancer survivors working during curative treatment. Work (Reading, Mass.), 55(4), 783–795. https://doi.org/10.3233/WOR-162449

Surbone, A., (2008). Cultural aspects of communication in cancer care. Supportive Care in Cancer, 16(3), 235-240.

Surbone, A., & Halpern, M. T., (2016). Unequal cancer survivorship care: addressing cultural and sociodemographic disparities in the clinic. Support Care Cancer. 2016 Dec;24(12):4831-4833. DOI: 10.1007/s00520-016-3435-4



Surbone, A., Zwitter, M., Rajer, M., & Stiefel, R., (2013). New challenges in communication with cancer patients. Part IV: Multicultural aspects of are for cancer patients in Israel, pg327

Swenson, W. T., (2016). Survey of spiritual quality of life among survivorship and distress guidelines. Journal of Clinical Oncology 34, no. 15_suppl - published online before print. DOI: 10.1200/JCO.2016.34.15_suppl.e21566

Tadmouri, G. O., & Al-Sharhan, M., (1985). Cancers in the United Arab Emirates. Genetic Disorders in the Arab World: United Arab Emirates. https://www.cags.org.ae/cbc08ca.pdf

Taghian, N. R., Miller, C. L., Jammallo, L. S., O'Toole, J., Skolny, M. N., (2014). Lymphedema following breast cancer treatment and impact on quality of life: a review. Crit Rev Oncol Hematol. 2014;92(3):227–34

Taha, H., Al-Qutob, R., Nyström, L., Wahlström, R., & Berggren, V. (2012). "Voices of Fear and Safety" Women's ambivalence towards breast cancer and breast health: a qualitative study from Jordan. BMC Women's Health, 12(1), 21.

Takeuchi, N., Yoshida, S., Rokuhara, T., Ohno, Z., Koike, K., Hotta, k., (2016). What are the patient expectations from the palliative chemotherapy? VOLUME 27, SUPPLEMENT 7, VII102, NOVEMBER 01, 2016. DOI: https://doi.org/10.1093/annonc/mdw523.037

Tamburini, M., Gangeri, L., Brunelli, C., Boeri, P., Borreani, C., Bosisio, M., Karmann, F. C., Greco, M., Miccinesi, G., Murru, L., Trimigno, P., (2003). Cancer patients' needs during hospitalisation: a quantitative and qualitative study. BMC Cancer 2003, 3:12. Available from: http://www.biomedcentral.com/1471-2407/3/12

Tariman, J. D., Doorenbos, A., Schepp, K. G., Singhal, S., & Berry, D. L. (2014). Information Needs Priorities in Patients Diagnosed with Cancer: A Systematic Review. Journal of the Advanced Practitioner in Oncology, 5(2), 115–122. Available from: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4042668/

Tanyi, R. A., (2002). Towards clarification of the meaning of spirituality. Journal of Advanced Nursing, 2002 Sep;39(5):500-9. DOI: 10.1046/j.1365-2648.2002.02315.



Tashakkori, A., & Teddlie, C., (1998). Mixed methodology: Combining qualitative and quantita-tive approaches. Applied Social Research Methods Series, vol. 46. Thousand Oaks, CA:Sage

Tashakkori, A., & Teddlie, C., (2003). Handbook of Mixed Methods in Social & Behavioral Research. Chapter 11: Data collection strategies in mixed methods research. Pg 298. Sage publications

Tashakkori, A., & Teddlie, C. (Eds.). (2010). SAGE handbook of mixed methods in social and behavioral research (2nd ed.). Thousand Oaks, CA: Sage.

Takahashi, M., (2016). Cancer survivorship: current status of research, care, and policy in Japan, Japanese Journal of Clinical Oncology, Volume 46, Issue 7, July 2016, Pages 599–604, https://doi.org/10.1093/jjco/hyw057

Tatsuo, A., Megumi, U., Tomohiro, N., Toru, O., Nobuhiro, S., Tatsuya, T., Hiroko Y., (2015). Difference of patient's perceived need in breast cancer patients after diagnosis, Japanese Journal of Clinical Oncology, Volume 45, Issue 1, January 2015, Pages 75–80, https://doi.org/10.1093/jjco/hyu165

Tay, L. H., Hegney, D., Ang, E., (2011). Factors affecting effective communication between registered nurses and adult cancer patients in an inpatient setting: a systematic review. Int J Evid Based Healthc. 2011; 9:151-164.

Taylor, B., & Francis, K., (2013). Qualitative research in health sciences: Methodologies Methods and Process. Chapter 1: Introduction. Routledge, New York

Taylor, S. E., (1983). Adjustment to threatening events: A theory of cognitive adaptation. American Psychologist, 38, 1161–73.

Taylor, S. E., (2011). Social Support: A Review. The Oxford Handbook of Health Psychology.

Teddlie, C., & Tashakkori, A. (2003). Major issues and controversies in the use of mixed methods in the social and behavioral sciences. In A. Tashakkori & C. Teddlie (Eds.), Handbook



of mixed methods in social and behavioral research (1st ed.) (pp. 3-50). Thousand Oaks, CA: Sage.

Teddlie, C. & Tashakkori, A. (2009). Foundations of mixed methods research: Integrating quantitative and qualitative approaches in the social and behavioral science. Thousand Oaks CA: Sage

Tellj, M. L., Chang, E. T., Kurian, A. W., Keegan, T. H., McClure, L. A., Lichtensztajn, D., Ford, J., & Gomez, S., (2011). Asian ethnicity and breast cancer subtypes: a study from the California Cancer Registry. Breast Cancer Res Treat 127(2): 471–478.

Terwee, C. B., Bot, S. D., de Boer, M. R., van der Windt, D. A., Knol, D. L., Dekker, J., et al., (2007). Quality criteria were proposed for measurement properties of health status questionnaires. J Clin Epidemiol. 2007; 60(1): 34–42. pmid:17161752

Tetteh, D. A., & Faulkner, S. L. (2016). Sociocultural factors and breast cancer in sub-Saharan Africa: implications for diagnosis and management. Women's health (London, England), 12(1), 147–156. https://doi.org/10.2217/whe.15.76

Thacker, P. H., Lutgendorf, S. K., & Sood. A. K., (2007). The neuroendocrine impact of chronic stress on cancer. Cell Cycle 6(4):430–433.

The Cancer Atlas., (2021). CANCER SURVIVAL. American Cancer Society, Inc. https://canceratlas.cancer.org/the-burden/cancer-survival/(Accessed on 02nd Feb 2021).

The World Bank., (2021). Literacy rate, youth total (% of people ages 15-24) - United Arab Emirates. UNESCO Institute for Statistics (uis.unesco.org). Data as of September 2021.https://data.worldbank.org/indicator/SE.ADT.1524.LT.ZS?locations=AE

Thewes, B., Butow, P., Girgis, A., & Pendlebury., (2004). The Psychosocial needs of Breast cancer survivors; A qualitative study of the shared and unique needs of younger versus older survivors. DOI:10.1002/pon.710. Psychooncology. 2004 Mar;13(3):177-89. doi: 10.1002/pon.710. PMID: 15022153.



Thorne, S., (2000). Data analysis in qualitative research. Evidence Based Nursing, 3, 68–70. doi:10.1136/ebn.3.3.68

Thorsen, L., Gjerset, G. M., Loge, J. H., Kiserud, C. E., Skovlund, E., Flotten, T., Fossa, S. D., (2011). Cancer patients' needs for rehabilitation services. Acta Oncol. 2011, 50 (2): 212-222. 10.3109/0284186X.2010.531050

Threlfall, K. D., (1999). "Using focus groups as a consumer research tool." Journal of Marketing Practice: Applied Marketing Science 5, (4) 102-105

Tian, L., Cao, X., Feng, X., (2019). Evaluation of psychometric properties of needs assessment tools in cancer patients: A systematic literature review. PLoS ONE 14(1): e0210242. https://doi.org/10.1371/journal.pone.0210242

Tiew, L. H., Kwee, J. H., Creedy, D. K., & Chan, M. F., (2013). Hospice Nurses' Perspectives of Spirituality. Journal of Clinical Nursing 22: 2923–33.

Till, J. E., (2003). Evaluation of support groups for women with breast cancer: importance of the navigator role. Health and quality of life outcomes, 1, 16. doi:10.1186/1477-7525-1-16

Tix, A. P., & Frazier, P. A., (1998). The use of religious coping during stressful life events: main effects, moderation, and mediation, Journal of Consulting and Clinical Psychology, 1998, Vol.66(2), pp.411-422. DOI: 10.1037/0022-006X.66.2.411

Tobin, G. A., Begley, C. M., (2004). Methodological rigour within a qualitative framework. Journal of Advanced Nursing, 48, 388–396. doi:10.1111/j.1365-2648.2004.03207.x

Tompkins, C., Scanlon, K., Scott, E., Ream, E., Harding, S., & Armes, J. (2016). Survivorship care and support following treatment for breast cancer: a multi-ethnic comparative qualitative study of women's experiences. BMC health services research, 16(1), 401. doi:10.1186/s12913-016-1625-x

Torres, E., Dixon, C., & Richman, A. R., (2016). Understanding the breast cancer experience of survivors: a qualitative study of African American women in rural eastern North Carolina. J Cancer Educ, 31 (2016), pp. 198-206



Towers, R., (2016). Psychological Issues in Breast Cancer Survivorship. Chapter 19. Breast Cancer Survivorship, DOI 10.1007/978-3-319-41858-2_19. © Springer International Publishing Switzerland 2016

Trenchard, L., McGrath-Lone, L., & Ward, H., (2016). Ethnic variation in cancer patients' ratings of information provision, communication and overall care. Ethnicity and Health 2016; 21 (5): 515-533 (October 2016).

Truong, M., Paradies, Y., & Priest, N. (2014). Interventions to improve cultural competency in healthcare: a systematic review of reviews. BMC health services research, 14, 99. doi:10.1186/1472-6963-14-99

Tsai, T-I., Morisky, E. D., Kagawa-Singer, M., & Ashing-Giw, T. K., (2011). Acculturation in the adaptation of Chinese American women to breast cancer: a mixed-method approach. DOI: 10.1111/j.1365-2702.2011.03872.x, Journal of Clinical Nursing, 20, 3383–3393

Tsaras, K., Papathanasiou, I., Mitsi, D., Veneti, A., Kelesi, M., Zyga, S., Fradelos, E. C., (2018). Assessment of Depression and Anxiety in Breast Cancer Patients: Prevalence and Associated Factors. Asian Pac J Cancer Prev. 2018 Jun 25;19(6):1661-1669. DOI: 10.22034/APJCP.2018.19.6.1661

Tsuchiya, M., & Horn, S. A., (2009). European Journal of Cancer Care. An exploration of unmet information needs among breast cancer patients in Japan: a qualitative study. European Journal of Cancer Care, 2009, 18, 149–155. DOI: 10.1111/j.1365-2354.2008.00936.x

Tuckett, A. G., (2005). Applying thematic analysis theory to practice: A researcher's experience. Contemporary Nurse, 19(1-2), 75-87.

Tu, J., Kang, G., Zhong, J., & Cheng, Y. (2019). Outpatient communication patterns in a cancer hospital in China: A qualitative study of doctor-patient encounters. Health expectations: an international journal of public participation in health care and health policy, 22(3), 594–603. doi:10.1111/hex.12890

Tuli, F., (2010). The Basis of Distinction Between Qualitative and Quantitative Research in Social Science: Reflection on Ontological, Epistemological and Methodological Perspectives.



Ethiop. J. Educ. & Sc. Vol. 6 No 1 September 2010. https://www.ajol.info/index.php/ejesc/article/viewFile/65384/53078

Turnbull Macdonald, G. C., Baldassarre, F., Brown, P., Hatton-Bauer, J., Li, M., Green, E., & Lebel, S. (2012). Psychosocial care for cancer: a framework to guide practice, and actionable recommendations for Ontario. Current oncology (Toronto, Ont.), 19(4), 209–216. https://doi.org/10.3747/co.19.981

Turner, J. H. (1991). The emergence of functionalism. In J.H. Turner (Ed.), The structure of sociological theory (5th ed. Pp. 33-50). Belmont, CA: Wadsworth.

Turning Research into Practice, (2005). Brassey, J., (2005) TRIP database Identifying High Quality Medical Literature from a range of sources, New Review of Information Networking, 11:2, 229-234, DOI: 10.1080/13614570600573441

Turning Research into Practice, (2007). Fyfe, T., 2007. Turning Research into Practice (TRIP). Journal of the Medical Library Association. 2007 Apr; 95(2): 215-216

Turning Research into Practice, (2018). Fox, L. M., 2018. Journal of the medical library association. Vol 106, No. 2 (2018). doi: http://dx.doi.org/10.5195/jmla.2018.399

Türk, K. E., & Yılmaz, M., (2018). The Effect on Quality of Life and Body Image of Mastectomy Among Breast Cancer Survivors. European journal of breast health, 14(4), 205–210. https://doi.org/10.5152/ejbh.2018.3875

UAE Government Portal, 2019. Government.ae; The United Arab Emirates' Government portal. (2019).https://government.ae/en/information-and-services/public-holidays-andreligious-affairs/official-religion-and-religious-affairs-entities Accessed 19th May 2019

UAE Government Portal, 2021. Government.ae; The United Arab Emirates' Government portal. (2019). Fact sheet - The Official Portal of the UAE Government

Uchida, M., Akechi, T., Okuyama, T., Sagawa, R., Nakaguchi, T., Endo, C., Yamashita, H., Toyama, T., & Furukawa, T. A., (2011). Patients' Supportive Care Needs and Psychological Distress in Advanced Breast Cancer Patients in Japan, Japanese Journal of



Clinical Oncology, Volume 41, Issue 4, 1 April 2011, Pages 530–536, https://doi.org/10.1093/jjco/hyq230

Ullrich, A., Marx, G., Bergelt, C., Benze, G., Zhang, Y., Wowretzko, F., Heine, J., Dickel, L. M., Nauck, F., Bokemeyer, C., Oechsle, K., (2021). Supportive care needs and service use during palliative care in family caregivers of patients with advanced cancer: a prospective longitudinal study. Support Care Cancer. 2021 Mar;29(3):1303-1315. doi: 10.1007/s00520-020-05565-z. Epub 2020 Jul 6. PMID: 32632761; PMCID: PMC7843549.

United Nations (2019). World Population Prospects 2019. Department of Economic and Social Affairs Population Dynamics. World Population Prospects - Population Division - United Nations

Urquhart, R.; Murnaghan, S.; Kendell, C.; Sussman, J.; Porter, G.A.; Howell, D.; Grunfeld, E., (2021). What Matters in Cancer Survivorship Research? A Suite of Stakeholder-Relevant Outcomes. Curr. Oncol. 2021, 28, 3188-3200. https://doi.org/10.3390/curroncol28040277

Usta, Y. Y., (2012). Importance of social support in cancer patients. Asian Pac J Cancer Prev 2012; 13:3569-72.

http://journal.waocp.org/article_26722_8bf0472560a509b6154c38e36fa4b649.pdf

Vachon, M., (2006). Psychosocial distress and coping after cancer treatment. American Journal of Nursing, 106(3), 26–31.

van Dalen, I., Groothoff, J., Stewart R, & Spreeuwenberg, P., (2001). Motives for seeking a second opinion in orthopaedic surgery. J Health Serv Res Policy 2001; 6:195–201.

van de Wal, M., Servaes, P., Berry, R., Thewes, B., & Prins, J. (2018). Cognitive Behavior Therapy for Fear of Cancer Recurrence: A Case Study. Journal of clinical psychology in medical settings, 25(4), 390–407. doi:10.1007/s10880-018-9545-z

Van Londen, G., Beckjord, E., Dew, M., Cuijpers, P., Tadic, S., & Brufsky, A. (2013). Breast cancer survivorship symptom management: current perspective and future development. Breast cancer management, 2(1), 71–81. doi:10.2217/bmt.12.63



Van Mossel, C., Leitz, L., Scott, S., Daudt, H., Dennis, D., Watson, H., Alford, M., Mitchell, A., Payeur, N., Cosby, C., Levi-Milne, R., & Purkis, M. E., (2012). Information needs across the colorectal cancer care continuum: scoping the literature. European Journal of Cancer Care, 21, 296-320.

Van Mossel, C., Leitz, L., Watson, H., Daudt, H., Alford, M., Mitchell, A., Coady, N., & Payeur, N., (2014). Learning from the collective story: information needs of people with colorectal cancer', Journal of Nursing Education and Practice, 4(8), 125.

Vaziri, S. H., & Lotfi K, F., (2012). Sexuality after breast cancer: need for guideline. Iranian journal of cancer prevention, 5(1), 10–15.

Velengtas, P., Daling, J. R., (1994). Risk factors for breast cancer in young women. Monographs National Cancer Institute 16: 15–23.

Vickberg, S. M., (2001). Fears about breast cancer recurrence. Cancer Practice, 9, 237–243.

Villagomeza, R. L., (2005). Spiritual distress in adult cancer patients: toward conceptual clarity. Holist Nursing Practice. 2005 Nov-Dec; 19(6):285-94.

Vitaliano, P., (1997). Psychological and physical concomitants of caregiving; introduction to special issue. Annals of behavioral medicine. 1997; 19:75–77.

Vivar, G. C., & McQueen, A., (2005). Informational and emotional needs of long-term survivors of breast cancer. Journal of Advanced Nursing, 51(5), 520–528.

Vivar, C. G., Whyte, D. A., & McQueen, A., (2010). Again: The impact of recurrence on survivors of cancer and family members. Journal of Clinical Nursing, 19, 2048–2056. doi:10.1111/j.1365-2702.2009.03145.x

Vogel, B. A., Bengel, J., Helmes, A. W., (2008). Information and decision making: patients' needs and experiences in the course of breast cancer treatment. Patient Educ. Couns., 71 (2008), pp. 79-85



Von Essen. L., Enskär. K., Kreuger, A., (2000). Self-esteem, depression and anxiety among Swedish children and adolescents on and off cancer treatment. Acta Paediatrica 89(2): 229–236.

Vos, P. J., Garssen, B., Visser, A. P., Duivenvoorden, H. J., deHaes, H. C., (2004). Earlystage breast cancer: explaining level of psychosocial adjustment using structural equation modeling. J Behav Med. 2004 Dec;27(6):557-80. doi: 10.1007/s10865-004-0003-z. PMID: 15669444.

Vrontaras, N. (2018). Cancer Patients' Views on the Family Changes and the Family Social Support. Journal of European Psychology Students, 9(1), 16–27, DOI: https://doi.org/10.5334/jeps.403

Walaa, K. S., Taher, A. J., Jabari, El. C., Hajar, C., Adib, M. S., Harrision, O., (2012). Barriers of Breast Cancer Screening and Treatment among Women in Emirate of Abu Dhabi. Ethnicity & Disease, Volume 22, Spring 2012. Available from: http://www.researchgate.net/publication/228329242_Barriers_to_breast_cancer_screening_an d_treatment_among_women_in_Emirate_of_Abu_Dhabi

Waljee, j. F., Ubel, P. A., Atisha, D. M., Hu, E. S., Alderman, A. (2011). The choice for breast cancer surgery: can women accurately predict postoperative quality of life and disease-related stigma? Ann Surg Oncol, 18 (2011), pp. 2477-2482

Walijee, J. F., Hu, E. S., Ubel, P. A., Smith, D. M., Newman, L. A., Alderman, A. K., (2008). Effect of esthetic outcome after breast-conserving surgery on psychosocioal functioning and quality of life. J Clini Oncology 2008; 26:3331-3337

Wallberg, B., Michelson, H., Nystedt, M., Bolund, C., Degner, L.F. & Wilking, N. (2000). Information needs and preferences for participation in treatment decisions among Swedish Breast Cancer patients. Acta Oncologica, 39(4), 467-76. Taylor & Francis. Acta Oncologica, Vol.39(4), p.467-476. http://dx.doi.org/10.1080/028418600750013375

Wallace, T. A., Martin, S. N., Ambs, S., (2011). Interactions among genes, tumor biology and the environment in cancer health disparities: examining the evidence on a national and global scale. Carcinogenesis. 2011 Aug; 32(8): 1107–1121. DOI: 10.1093/carcin/bgr066.



Walsh, J., Harrison, J. D., Young, J. M., Butow, P. N., Solomon, M. J., & Masya, L. (2010). What are the current barriers to effective cancer care coordination? A qualitative study. BMC health services research, 10, 132. doi:10.1186/1472-6963-10-132

Walsh, J., Young, J. M., Harrison, J. D., Butow, P. N., Solomon, M. J., Masya, L., White, K., (2011). What is important in cancer care coordination? A qualitative investigation. Eur J Cancer Care. 2011, 20 (2): 220-227. 10.1111/j.1365-2354.2010.01187. x.

Wang, T., Molassiotis, A., Chung, B. P. M., Tan, J-Y., (2018). Unmet care needs of
advanced cancer patients and their informal caregivers: a systematic review. BMC Palliative
Care201817:96.Care201817:96.https://doi.org/10.1186/s12904-018-0346-9.https://bmcpalliatcare.biomedcentral.com/articles/10.1186/s12904-018-0346-9

Wang, J. H., Adams, I. F., Pasick, R. J., Gomez, S. L., Allen, L., Ma, G. X., Lee, M. X., Huang, E. (2013). Perceptions, Expectations, and Attitudes about Communication with Physicians among Chinese American and non-Hispanic White Women with Early-Stage Breast Cancer. Supportive Care in Cancer: Official Journal of the Multinational Association of Supportive Care in Cancer, 21(12), 3315–3325. http://doi.org/10.1007/s00520-013-1902-8

Wan-Tzu, L., Patsy, Y., & Raymond, J. C., (2018). Unmet supportive care needs and symptom burden in Taiwanese cancer survivors who have completed primary treatment. European Journal of Oncology Nursing. Vol. 35, Aug. 2018 Pg 79-84. https://doi.org/10.1016/j.ejon.2018.05.010

Ward, E., Jemal, A., Cokkinides, V., Singh, G. K., Cardinez, C; Ghafoor, A; Thun, M., (2004). Cancer Disparities by Race/Ethnicity and Socioeconomic Status. CA A Cancer Journal for Clinicians. Volume 54 Y Number 2 Y March/April 2004

Ware, J. E., Davies-Avery, A., Stewart, A. L., (1978). The measurement and meaning of patient satisfaction: a review of the literature. Health Med Care Service Rev 1978; 1:1–15

Waring, A. N., (2000). Breast cancer: reactions, choices, decisions. The Ochsner journal, 2(1), 40–46.



Warga, C. L., (2000). Menopause and the Mind: The Complete Guide to Coping with the Cognitive Effects of Perimenopause and Menopause, Including Memory Loss, Foggy Thinking, and Verbal Slips. New York: Simon and Schuster.

Warren, M.P., Shu, A. R., (2000). Dominguez JE. Menopause and Hormone Replacement. [Updated 2015 Feb 25]. In: De Groot LJ, Chrousos G, Dungan K, et al., editors. Endotext [Internet]. South Dartmouth (MA): MDText.com, Inc.; 2000-. Available from: https://www.ncbi.nlm.nih.gov/books/NBK279050/

Weber, R., (2004). The rhetoric of positivism versus interpretivism: A personal view. MIS Quarterly, 28(1), 3-12.

Weber, D., & O'Brien, K., (2017). Cancer and Cancer-Related Fatigue and the Interrelationships with Depression, Stress, and Inflammation. Journal of evidence-based complementary & alternative medicine, 22(3), 502–512. https://doi.org/10.1177/2156587216676122

Wei, D., Liu, X.-Y., Chen, Y.-Y., Zhou, X., & Hu, H.-P. (2016). Effectiveness of Physical, Psychological, Social, and Spiritual Intervention in Breast Cancer Survivors: An Integrative Review. Asia-Pacific Journal of Oncology Nursing, 3(3), 226–232. http://doi.org/10.4103/2347-5625.189813

Wells-Di, Gregorio., S., Porensky, E. K., Minotti, M., Brown, S., Snapp, J., Taylor, R. M., Adoliph, M. D., Everett, S., Lowther, K., Callahan, K., Streva, D., Heinke, V., Leno, D., Flowe, C., McVey, A., Andersen B. L., (2013). The James supportive care screening: integrating science into practice to meet the NCCN guidelines for distress management at a comprehensive cancer centre. Psycho Oncology, 2013;22(9): 2001-2008

Welman, J. C., & Kruger, S. J., (1999). Research methodology for the business and administrative sciences. Johannesburg, South Africa: International Thompson.

Weinstein, L. C., Stefancic, A., Cunningham, A. T., Hurley, K. E., Cabassa, L. J., & Wender, R. C. (2016). Cancer screening, prevention, and treatment in people with mental illness. CA: a cancer journal for clinicians, 66(2), 134–151. doi:10.3322/caac.21334



Wen, K-Y., & Gustafson, D, H., (2004). Needs Assessment for cancer patients and their families. Health and Quality of Life Outcomes20042:11. https://doi.org/10.1186/1477-7525-2-11. Available from: https://hqlo.biomedcentral.com/articles/10.1186/1477-7525-2-11

Wenzel, J., Jones, R. A., Klimmek, R., Krumm, S., Darrell, L. P., Song, D., Stearns, V., Ford, J. G. (2012). Cancer support and resource needs among African American older adults. Clinical journal of oncology nursing, 16(4), 372–377. doi:10.1188/12. CJON.372-377

Wertz. F. J. (2005). Phenomenological research methods in counseling psychology. Journal of Counseling Psychology, 52, 167–177.

Wessels, H., de Graeff, A., Wynia, K., de Heus, M., Kruitwagen, C. L., Woltjer, G. T., Teunissen, S. C., Voest, E. E., (2010). Gender-related needs and preferences in cancer care indicate the need for an individualized approach to cancer patients. Oncologist. 2010, 15 (6): 648-655. 10.1634/theoncologist.2009-0337

Wewers, M. E., & Lowe, N. K., (1990). A critical review of visual analogue scales in the measurement of clinical phenomena. Res Nurs Health. 1990 Aug;13(4):227-36. doi: 10.1002/nur.4770130405. PMID: 2197679.

Whelan, T. J., Mohide, E. A., Willan, A. R., Arnold, A., Tew, M., Sellick, S., et al. (1997). The supportive care needs of newly diagnosed cancer patients attending a regional cancer center. Cancer, 80(8), 1518-1524.

Whitman & Aldinger., (2009). Case studies in global school health promotion. From research to practice. Chapter 20: United Arab Emirates: Health promoting schools: strategies for policy change. Springer science and business media, LLC 2009.

White-Means, S. I., & Osmani, A. R. (2017). Racial and ethnic disparities in patient-provider communication with breast cancer patients: Evidence from 2011 MEPS and experiences with cancer supplement. INQUIRY: The Journal of Health Care Organization, Provision, and Financing. https://doi.org/10.1177/0046958017727104

Whitford, H. S., Olver, I. N., Peterson, M. J., (2008). Spirituality as a core domain in the assessment of quality of life in oncology. Psycho-oncology. Vol.17(11):1121-8.



Widdicombe, S. & Wooffitt, R., (1995), The language of youth subcultures: Social identity in action. Hemel Hempstead: Harvester Wheatsheaf.

Wiggers, J. H., Donovan, K. O., Redman, S., Sanson-Fisher, R. W., (1990)., Cancer patient satisfaction with care. Cancer 66:610 - 616, 1990 Crossref, Medline, Google Scholar. https://acsjournals.onlinelibrary.wiley.com/doi/pdf/10.1002/1097-0142(19900801)66:3% 3C610::AID-CNCR2820660335% 3E3.0.CO;2-T

Willig, C., (1999). Beyond appearances: A critical realist approach to social constructionism. In D. J. Nightingale & J. Cromby (Eds.), Social constructionist psychology: A critical analysis of theory and practice (pp. 37-51). Buckingham, UK: Open University Press.

Willig, C., (2008). Introducing qualitative research in psychology. 2nd edition. Open university press.

Wilkinson, S., (2000). Women with breast cancer talking causes: comparing content, biographical and discursive analyses. Feminism & Psychology, 10(4), 431-460

Wilt, J. A., Grubbs, J. B., Kenneth I., Exline, P & J., (2016). Religious and Spiritual Struggles, Past and Present: Relations to the Big Five and Well-Being. The International Journal for the Psychology of Religion 27: 51–64

Wisdom, J. P., Cavaleri, M. A., Onwuegbuzie, A. J., & Green, C. A. (2012), April). Methodological reporting in qualitative, quantitative, and mixed methods health services research articles. Health Services Research, 47(2), 721+. Retrieved from https://link-gale-com.salford.idm.oclc.org/apps/doc/A284552051/AONE?u=salcal2&sid=AONE&xid=35121 234

Wisdom, J., &U Creswell, J. W., (2013). Mixed Methods: Integrating Quantitative and Qualitative Data Collection and Analysis While Studying Patient-Centered Medical Home Models. Rockville, MD: Agency for Healthcare Research and Quality. February 2013. AHRQ Publication No. 13-0028-EF.

WislØff, F., Hjorth, M., (1997). Health-related quality of life assessed before and during chemotherapy predicts forsurvival in multiple myeloma.Br. J. Haematol.1997,97,29–37



Wong, M. C., Yee, K. C., Turner, P., (2008). A Structured Evidence-based Literature Review regarding the Effectiveness of Improvement Interventions in Clinical Handover. 2008, Australia: The eHealth Services Research Group, University of Tasmania

Wright, S., (1999). How to be happy at work. Nursing Time, 95: 26-8

Wright, J., Williams, R., & Wilkinson, J. R. (1998). Development and importance of health needs assessment. BMJ (Clinical research ed.), 316(7140), 1310–1313. https://doi.org/10.1136/bmj.316.7140.1310

Wright, P., (2019). Certified Hospice and Palliative Nurse (CHPN) Exam Review. A study guide with review questions. Chapter 18: 18: Psychosocial, Spiritual, and Cultural Needs. DOI: 10.1891/9780826119742.0018. Springer Publishing company

World Health Organisation's (1948). Preamble to the Constitution of the World Health Organization as adopted by the International Health Conference, New York: World Health Organization, 19-22 June 1946. Available from: http://www.who.int/suggestions/faq/en/..

World Health Organization. (2007). Strengthening health systems to improve health outcomes: WHO's framework for action. Geneva, Switzerland: Author.

World Health Organisation, (2012). WHOQOL: Measuring Quality of Life. Available online: https://www.who.int/healthinfo/survey/whoqol-qualityoflife/en/(accessed on 26 June 2020)

World Health Organization (WHO) (2014). World Cancer Report 2014 warns of a "tidal wave" of cancer. Available at: http://www.rttnews.com/2262790/world-cancer-report-2014-warns-of-a-tidal-wave-of-cancer.aspx (accessed 22 October 2019).

World Health Organization (WHO), (2015). Breast cancer: prevention and control. Available at: http://www.who.int/cancer/detection/breastcancer/en/

World Health Organization, (2017a). Projections of mortality and causes of death, 2015 and 2030. 2015. http://www.who.int/healthinfo/global_burden_disease/projections/en/ [Accessed 23 October 2017.]



World Health Organization (2017b). Guide to Cancer. Early Diagnosis. Chapter 1:IntroductionandScope.https://apps.who.int/iris/bitstream/handle/10665/254500/9789241511940-eng.pdf[Accessed2nd May 2018.]2018.]

World Health Organization (2017c). Guide to Cancer. Early Diagnosis. Chapter 2:UnderstandingEarlyDiagnosis.https://apps.who.int/iris/bitstream/handle/10665/254500/9789241511940-eng.pdf[Accessed2nd May 2018.]

World Health Organization, (2018). World Health Organization. International Agency for Research on Cancer. Estimated Number of Deaths in 2018, United Arab Emirates, All Cancers, Females, All Ages. 2018. Available online: http://gco.iarc.fr/today/home (accessed on 14 December 2019).

World Health Organisation, (2018a). https://www.who.int/cancer/PRGlobocanFinal.pdf

World Health Organisation, (2018b). https://www.who.int/news-room/fact-sheets/detail/cancer#:~:text=Around%20one%20third%20of%20deaths,tobacco%20use%2C%20and%20alcohol%20use.

World Health Organisation (WHO), (2018c). Integrating palliative care and symptom reliefinto primary health care. A WHO guide for planner, implementers and managers. Chapter 2.Accesstopalliativecare.https://apps.who.int/iris/bitstream/handle/10665/274559/9789241514477-eng.pdf

World Health Organization (2021). https://www.who.int/tools/whoqol/whoqol-bref

World Life Expectancy, (2017). http://www.worldlifeexpectancy.com/cause-of-death/breast-cancer/by-country/

World economic forum (2017). https://www.weforum.org/agenda/2017/03/these-are-the-wealthiest-countries-in-the-world



Wortman, C., (1984). Social support and the cancer patient. Cancer 1984; 53: (Suppl.) 2339 2360.

Woźniak, K., & Iżycki, D. (2014). Cancer: a family at risk. Przeglad menopauzalny = Menopause review, 13(4), 253–261. https://doi.org/10.5114/pm.2014.45002

Wronski, S. L., (2015). Defining cancer survivor and cancer survivorship: THE WHO, WHAT, AND WHEN. PSICOONCOLOGÍA. Vol. 12, Núm. 1, 2015, pp. 7-18 ISSN: 1696-7240 – DOI: 10.5209/rev_PSIC. 2015.v12.n1.48900

Wulff, C. N. (2012). The Effect of Hospital-Based Case Management in Cancer care Pathways. PhD Thesis. 2012, Aarhus University, Research Unit for General Practice, Unpublished

Xu, R. H., & Wong, E. L. (2017). Involvement in shared decision-making for patients in public specialist outpatient clinics in Hong Kong. Patient preference and adherence, 11, 505–512. https://doi.org/10.2147/PPA.S126316

Yanez, B., Edmondson, D., Stanton, A. L., Park, C. L., Kwan, L., Ganz, P. A., & Blank,
T. O. (2009). Facets of spirituality as predictors of adjustment to cancer: Relative contributions of having faith and finding meaning. Journal of Consulting and Clinical Psychology, 77(4), 730-741. http://dx.doi.org/10.1037/a0015820

Yauch, C. A., & Steudel, H. J. (2003). Complementary use of qualitative and quantitative cultural assessment methods. Organizational Research Methods, 6(4), 465

Yefimov, Vladimir., (2004). On Pragmatist Institutional Economics. IDEAS Working Paper Series from RePEc; Munich: Munich Personal RePEc Archive.

Yeung, S-C, J., (2009). Medical care of cancer patients. Published by BC Becker. Chap 72. Sexual Function Issues

Yi, J. C., & Syrjala, K. L., (2017). Anxiety and Depression in Cancer Survivors. The Medical clinics of North America, 101(6), 1099–1113. https://doi.org/10.1016/j.mcna.2017.06.005

Yin, R. K., (1993). Applications of case study research. Newbury Park, CA: Sage



Yin, R. K., (1994). Discovering the future of the case study method inevaluation research.Eval Pract. 1994; 15:283-290.

Young, J. S., (2014). Resources for Middle Eastern patients: online resources for culturally and linguistically appropriate services in home healthcare and hospice, Part 3. Home Healthcare Nurse. DOI: 10.1097/NHH.000000000000085.

Younis, M., Al-Rubaye, D., Haddad, H., Hammad, A., & Hijazi, M., (2016). Knowledge and Awareness of Breast Cancer among Young Women in the United Arab Emirates. Advances in Breast Cancer Research, 2016, 5, 163-176. http://www.scirp.org/journal/abcr. ISSN Online: 2168-1597. ISSN Print: 2168-1589. https://file.scirp.org/pdf/ABCR_2016103113304426.pdf

Yousefi, H., Abedi, H. A., (2011). Spiritual care in hospitalized patients. Iran J Nurs Midwifery Res 2011; 16:125.

Ypinazar, V. A., Margolis, S. A., (2006). Delivering Culturally Sensitive Care: The Perceptions of Older Arabian Gulf Arabs Concerning Religion, Health, and Disease. Qualitative Health Research. 2006;16(6):773-787.

Zabora, J.R., Blanchard, C., & Smith, E.D. (1997). The prevalence of psychological distress across the disease continuum. Journal of Psychosocial Oncology, 15, 75–87.

Zabora, J., BrintzenhofeSzoc, K., Curbow, B., Hooker, C., Piantadosi, S., (2001). The prevalence of psychological distress by cancer site. Psycho-oncology. Vol.10(1):19-28.

Zabora, J. R., (2015). Distress screening guidelines for oncology social workers. In G. Christ, C. Messner & L. Behar (Eds.), Handbook of oncology social work: Psychosocial care for people with cancer (pp. 115–119). New York, NY: Oxford University Press.

Zafar, S. Y., & Abernethy, A. P. (2013). Financial toxicity, Part I: a new name for a growing problem. Oncology (Williston Park, N.Y.), 27(2), 80–149.

Zaider, T.I., & Kissane, D.W. (2015). Psychosocial interventions for couples and families coping with cancer. In J.C. Holland, W.S. Breitbart, P.N. Butow, P.B. Jacobsen, M.J. Loscalzo,



& R. McCorkle (Eds.), Psycho-oncology (3rd ed., pp. 526–531). New York, NY: Oxford University Press

Zakri, J. & Karim, S. M., (2015). Breaking Cancer Bad News to Patients with Cancer: A Comprehensive Perspective of Patients, Their Relatives, and the Public—Example from a Middle Eastern Country. http://ascopubs.org/doi/full/10.1200/jgo.2015.001925

Zamanzadeh, V., Rassouli, M., Abbaszadeh, A., Nikanfar, A., Alavi-Majd, H., Ghahramanian, A., (2014). Factors Influencing communication between the patients with cancer and their nurses in oncology wards. DOI: 10.4103/0973-1075.125549. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3931236/; 2014 Jan-Apr; 20(1): 12–20.

Zapka, J. G., Taplin, S. H., Solberg, L. I., Manos, M. M., (2003). A framework for improving the quality of cancer care: the case of breast and cervical cancer screening. Cancer Epidemiol Biomarkers Prev. 2003 Jan;12(1):4-13. PMID: 12540497.

Zebrack, B. J., & Zeltzer, L. K., (2003). Quality of life issues and cancer survivorship. Current Problems in Cancer 27(4):198–211.

Zebrack, E. S., Penedo, F. J., Bustillo, N. E., Benedict, C., Rasheed, M., Lechner, S., Antoni, M. H. (2010). Longitudinal Effects of Social Support and Adaptive Coping on the Emotional Well-Being of Survivors of Localized Prostate Cancer. The Journal of Supportive Oncology, 8(5), 196–201.

Ziadeh, C., Ziogas, A., Jiang, L., Anton-Culver, H., (2018). Breast Cancer Characteristics in Middle Eastern Women Immigrants Compared with Non-Hispanic White Women in California, JNCI Cancer Spectrum, Volume 2, Issue 2, April 2018, pky014, https://doi.org/10.1093/jncics/pky014.

https://academic.oup.com/jncics/article/2/2/pky014/5025853

Zidan, J., Sikorsky, N., Basher, W., Sharabi, A., Friedman, E., & Steiner, M., (2012). Differences in pathological and clinical features of breast cancer in Arab as compared to Jewish women in Northern Israel. International Journal of Cancer. Volume 131, Issue 4, 924–929 (2012) VC 2011 UICC.



Ziebland, S., & McPherson, A., (2006). Making sense of qualitative data analysis: An introduction with illustrations from DIPEx (personal experiences of health and illness). Medical Education, 40 (5): 405 - 414.

Zimmermann, T., (2015). Intimate Relationships Affected by Breast Cancer: Interventions for Couples. Breast Care (Basel). 2015 Apr;10(2):102-8. doi: 10.1159/000381966. PMID: 26195938; PMCID: PMC4464013.

Ziner, K. W., Sledge, G. W., Bell, C. J., Johns, S., Miller, K. D., & Champion, V. L. (2012). Predicting fear of breast cancer recurrence and self-efficacy in survivors by age at diagnosis. Oncology nursing forum, 39(3), 287–295. https://doi.org/10.1188/12.ONF.287-295

Zhou, E. S., Penedo, F. J., Bustillo, N. E., Benedict, C., Rasheed, M., Lechner, S., Soloway, M., Kava, B. R., Schneiderman, N., & Antoni, M. H. (2010). Longitudinal effects of social support and adaptive coping on the emotional well-being of survivors of localized prostate cancer. The journal of supportive oncology, 8(5), 196–201. https://doi.org/10.1016/j.suponc.2010.09.004

Zucca, A., Sanson-Fisher, R., Waller, A., Carey, M., (2014). Patient-centred care: making cancer treatment centres accountable. Support Care Cancer. 2014 Jul;22(7):1989-97. doi: 10.1007/s00520-014-2221-4. Epub 2014 Apr 3. PMID: 24696084.



Chapter 8: APPENDICES

Following are the list of documents provided in this section:

Appendix 1: General Consent Form Tawam Hospital

Appendix 1a: General Hospital Consent Form - Original

Appendix 1b: General Hospital Consent Form - Highlighted

Appendix 1c: Email confirmation

Appendix 2: IRB Approvals

Appendix 2a: IRB approval Tawam Hospital

Appendix 2b: IRB approval Tawam Hospital - Extension

Appendix 2c: IRB approval Salford University

Appendix 3: Written Consent Form

Appendix 3a: Written Consent Form - English

Appendix 3b: Written Consent Form - Arabic

Appendix 4: Patient Information Sheet

Appendix 4a: Patient Information Sheet - English

Appendix 4b: Patient Information Sheet - Arabic

Appendix 5: Cancer Patients' Need Survey

Appendix 5a: Cancer Patients' Need Survey - English

Appendix 5b: Cancer Patients' Need Survey - Arabic



- Appendix 6: Data Collection Form Cancer Registry
- Appendix 7: Risk Assessment Form
- Appendix 8: Data Confidentiality Form
- **Appendix 9:** Survivorship Care Plan (SCP)
- Appendix 10: Quantitative Data Extraction File
- Appendix 11: Qualitative Interviews
- Appendix 12: Glossary of terms used (Abbreviations)
- Appendix 13: Approved Permission Mayo Clinic Survey



APPENDIX 1a: GENERAL CONSENT FOR HEALTHCARE SERVICES - ORIGINAL

	M Pa Di Er Lc Fl	IRN: atient Name: OB: nirates ID #: oc/Spec: N:	Age: Admit/Re	Sex g Date &	Nationality: Time:	
General	Consent for	Healthcare Se	rvices			
اية الصحية	ي خدمات الرع	وافقة عامة على تلق	إقرار م			
Patient Instructions					لمري <u>ض:</u>	تعليمات ال
The purpose of this form is to ensure that and its staff have permission to diagnose treatment to you (as well as obtain of relating to other matters such as use of y information and payment for services we p We will ask you to sign this form and ony	the hospital and provide confirmation your medical rovide)	تأكيد من أن كل من القيام بتشخيص حالتك مول على موافقة تتعلق الطبية والمبلغ المالي مها).	رار هذا هو ال يهم التصريح ب ك (كذلك الحص يدام معلوماتك بة التي نقوم بتقدير	وذج الإقر رظفيها لد العلاج لا مثل استخ قابل الخدم	الغرض من نم المستشفى و مو المرضية وتوفير بأمور أخرى م المستحق دفعه ما	*
will be effective for all future visits by hospital	you to the	ج وبمجرد الانتهاء من ل لكل الزيارات المقبلة	على هذا النموذ وذج نافذ المفعو	ك التوقيع ر هذا النم لمستشفى.	سوف نطلب منا ذلك، سيتم اعتبار التي ستقوم بها لا	*
I hereby acknowledge and authorize (a the legal representative of the patient on behalf of the patient and, in relati medical treatment matters, myself) the f	and if I am I authorize on to non- following:	لي للمريض وبالنيابة نخص العلاج الطبي)،	الممثّل القانون الأمور التي لا ن	(إذا كنت ا يتعلق با	وجبه وأفوض والمريض فيم	أقرّ أنا بم عن نفسي بما يلي:
1 Diagnosis and Treatment:					يص والعلاج:	1- التشخ
 the hospital and its physicians, professionals and employees wh involved in my care to perform s diagnostic and treatment proced considered by the attending necessary and appropriate for my m (including but not limited electrocardiograms and admini intravenous fluids, injections medications). 	healthcare no may be uuch medical ures as are physician(s) nedical needs to x-rays, stration of and other	للاجية التي قد ير اها سبة لحالتي الصحية سر، الأشعة السينية بالمحاليل الوريدية،	بية وإجراءات ع ن ضرورية ومنا بائها ومتخصص بالمثال لا الحص لهربائي والإمداد ردوية الأخرى).	نيصات ط ء المعالجير نشفى وأط على سبيل القلب الك الحقن و الا	إجراء أي تشذ الطبيب/الأطبا. من قبل المست والموظفين، (ومخطط صدي بالإضافة إلى ا	•
 As the practice of medicine is n science and outcomes of treatmen my medical condition, no guarante outcome of any procedures, tr examinations can be made to me. 	ot an exact t depend on es as to the reatment or	ما دقيقا وأن نتائج لك لا يوجد نتائج علاج أو الفحوصات	لطبية ليست علم تي الصحية، لذ اءات أو طرق ال	ارسات ال على حال من الإجرا لها.	بسبب أن المم العلاج تعتمد مضمونة لأي التي قد أخضع	•
 if I require an invasive surgical p agree to participate in human sub research, I will be asked to give informed consent 	rocedure or ject medical e additional	جراحي تداخلي أو ية التي تجرى على أقوم بإعطاء موافقة	وع إلى إجراء في البحوث الطب مني حيننذ أن	جة للخضر لمشاركة ف يستلزم لك.	في حالة الحا. مو افقتي على الإنسان، سوف إضافية على ذا	•
HA-AF-001-01-DT General Cons	ent for Healthca	re Services- DT Form			Page 1 of 5	



atient Name:		MRN:	FIN:
2 Trainee Staff: the participation in treatment of healthcare profession under the supervision of appropr personnel	من قبل n my care and nals in training, راف طاقم iate healthcare	ئة في تقديم الرعاية والعلاج ل إض التدريب وذلك تحت إشر	2- الموظفون المتدريون : المشارك متخصصي الرعاية الصحية لأغر الرعاية الصحية المناسب.
3 Insurance: the hospital to bill the m for my care and treatment to company or other payer, if I am insu	ی بمطالبة edical expenses الیف. an insurance red	تغطية تأمينية، تقوم المستشف تأمين صحي آخر بتغطية التك	3- التأمين: إذا كنت مشمول تحت شركة التأمين أو أي مزود خدمات :
4 Responsibility for Payment: the parall charges and expenses of the from my care and treatment if I are to the extent that such expenses a by insurance	yment by me of للمستشفى كن مؤمنا not insured or منحي تلك re not covered	د كافة المصاريف والنفقات لاج المقدم لى وذلك إذا لم أ لا تغطي شركة التأمين الم	4- مسؤولية الدفع: قيامي بتسدير التي تتعلق بالرعاية الصحية والعا تأمينا صحيا أو إلى الحد الذي التكاليف.
5 Personal Property: the hospital and responsible for any loss or damage to my belongings I have brought unless declared by me, provided to by hospital staff	its staff are not that may occur to the hospital and signed for	لا تتحمل المستشفى أو د يلحق بممتلكاتي الشخصية له قيامي بالإقرار أنني قمت مها بواسطتهم.	5- الممتلكات الشخصية: المسؤولية عن أي فقد أو ضرر ق بإحضارها للمستشفى إلا في حال لموظفي المستشفى والتوقيع باستلا،
 Disclosure of Medical Information and use of my medical information copies of my medical record and/o electronic medical record through device or the internet) to hospital staff involved in my care healthcare professionals outsid who may become involved in m those providing another opinion my care and treatment my insurance company, empl person paying for my medical example, to determine whether continued care or other treatmen necessary or which is require payment for my bill) hospital staff and others within the United Arab Emirates (includ registries) to assess and impro- quality and cost of care provice other patients, assist in the operation of the hospital or co- and scientific research and review approval, patients could be conta- terest to take part in the research 	the disclosure thicin (including praccess to my than electronic e the hospital ty care such as the or continuing over or other expenses (for ty admission, ent is medically ed to process and outside of ing clinical data ty care safety, ted to me and administrative onduct medical tys. Further to IRB cted for their in- studies.	الطبية: القيام بالإفصاح سخ من ملفي الطبي الإلكتر خاص بي من خلال أجيزة إلك في تقديم الرعاية لي سحية من خارج نطاق المستئد محية من خارج نطاق المستئد أو يستكملون تقديم الرعاية وا يستكملون تقديم الرعاية وا مسيل المثل، التحديد ما إذا ك ستمرة والعلاج ضروريا م بر هم داخل أو خارج دولة في ذلك سجلات البيانات في ذلك سجلات البيانات داخل والمراجعات الطبية والم م بتمويل أو تحصيل المبالغ المقدمة لي.	 6- الإفصاح عن المعلومات المعلومات الطبية (بما في ذلك ند المحلومات الطبية (بما في ذلك ند الانترنت) بواسطة: طاقم المستشفى المشاركة طاقم المستشفى المشاركة قد يشاركون في تقديم وتديم المركة التأمين أو صاحد بإعطاء رأي طبي آخر المحلود اللمستشفى والرعاية الم مصاريفي الطبية، أو ما هو مطلود المستشفى أو إجراء الأبري قد تقوم المستشفى أو إجراء الأبرعاية المؤسسات التي قد تقوم المؤسسات التقييم وتحسير العربية المتحدة (بما المستشفى أو إجراء الأبري قد تقوم المستشفى أو إجراء الأبريانية المستشفى أو إجراء الأبريانية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى نظير الرعاية المستشفى المستشى المستشفى المستشفى المستشفى المستشفى المستشفى المستشفى المستشفى المستشفى المستشفى المستشامى المستشامى المستشفى المستشفى المستشفى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشمى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشامى المستشمى المستشامى المستشامى المستشامى المستشامى المستشا



Patient Name:	MRN: FIN:
 Payments owed to the hospital for my care Regulatory, judicial and other governmental authorities where authorized or required by laws and regulations (including in relation to suspected illegal activities). 	 الهيئات الحكومية التنظيمية والقضائية و غير ها كما هو مُصرح لها، أو الذي تستلزمه القوانين واللوانح (بما في ذلك يما يتعلق بانشطة غير مشروعة مشتبه بها).
7 Request from Prior Healthcare Providers : the hospital and its staff to contact other healthcare providers whose names I have provided in order to obtain information related to the care and treatment provided to me by such providers.	7- تقديم طلب لمقدمي الرعاية الصحية السابقين: أن يقوم المستشفى وموظفيه بمخاطبة مقدمي الرعاية الصحية الأخرين الذين قمت بذكر أسماءهم من أجل الحصول على معلومات متعلقة بالرعاية الصحية والعلاج الذي تم تقديمه لي من قبلهم.
8 Blood withdrawal and Receipt : the drawing from my body of blood for diagnosis and testing and, in emergencies where I am unable to provide a separate consent, administration or transfusion of blood, blood plasma or other blood products deemed necessary by the healthcare professional staff.	8- سحب الدم واستلامه: إجراء سحب الدم بهدف التشخيص والفحص، وفي الحالات الطارئة التي لا أتمكن فيها من إعطاء موافقة منفصلة، القيام بتزويدي بالدم وبلازما الدم أو أي من مشتقاته الأخرى التي يعتبر ها طاقم الرعاية الصحية المختص ضرورية .
9 Tissues and Specimens: the hospital to examine, use, store and dispose of any tissue, fluids or specimens obtained from my body during my care and treatment.	9- الأنسجة والعينات: أن تقوم المستشفى بفحص واستخدام وتخزين والتخلص من أي نسيج أو السوانل أو العينات التي تم سحبها من جسمي خلال فترة الرعاية الصحية والعلاج المقدم لي.
10 Hospital Rules : the hospital to take appropriate actions for the safety and welfare of myself and other patients if I do not follow the rules of the hospital explained to me by hospital staff.	1 0- قوانين المستشفى : أن تتخذ المستشفى الإجراءات المناسبة لتحقيق السلامة والصحة لى وللمرضى الأخرين وذلك في حالة لم أقم باتباع قوانين المستشفى التي سبق وأن شرحت لي من قبل طاقم المستشفى.
11 Obstetrics/Gynecology Patients: if I am an obstetric patient, this General Consent for Healthcare Services also applies to any children born to me during my or their hospitalization.	11- مرضى أمراض النساء والولادة: : إذا أنا كنت من مرضى النساء الحوامل، سينطبق أيضا إقرار موافقة عامة على تلقي خدمات الرعاية الصحية هذا على أي أطفال يولدون أثناء مدة إقامتي أو إقامتهم في المستشفى.
12 Duration of Consent : the hospital and its staff to rely upon the above mentioned General Consent until I revoke them in writing (which I may do at any time by indicating my revocation on a copy of this or a new form and providing it to the registration staff of the hospital).	12- مدة الموافقة: إعتماد المستشفى وموظفيها إقرار الموافقة العامة الموضح أعلاه حتى أقوم أنا بالغانه كتابيًا في أي وقتموضحاً قيامي بالإلغاء على هذه النسخة أو على نموذج جديد وتقديمه لموظف التسجيل بالمستشفى).

SEHA-AF-001-01-DT

General Consent for Healthcare Services- DT Form

Page 3 of 5



Patient Name:	MRN: FIN:
 Additional Disclosure: release of my medical information (including copies of my medical record) to the following persons (in addition to any others entitled under applicable law): 1. First Name :	 13- إقرار إضافي: يجوز إعطاء المعلومات الطبية الخاصة بي (بما في ذلك شخ من ملفي الطبي للأشخاص التالية أسماؤهم (بالإضافة إلى أي أشخاص آخرين مخولين بموجب القانون المعمول به): 1. الاسم الأول:
*ID (Type):	الصلة: ** الهوية الشخصية (الذوع):
*ID (No.):	** الهوية الشخصية (الرقم):
* if known	** إذا ما كان معلوماً
□ Release only to me	□إعطاء المعلومات الطبية لي فقط
 14 Patient Rights Materials: I have received the following printed materials: Patient Rights and Responsibilities 	 14 وثائق حقوق المريض: لقد استامت الوثائق التالية: حقوق ومسؤوليات المريض
PATIENT SIGNATURE:	توقيع المريض
I have read and understand the information provided to me on this form	لقد قمت بقراءة وفهم المعلومات التي يحتوي عليه هذا النموذج
Patient Name:	اسم المريض:
Patient Signature :	توقيع المريض:
Date and Time:	التاريخ والوقت:
LEGAL REPRESENTATIVE SIGNATURE: (if patient is unable to or does not have legal capacity to sign form e.g. patient is under 18)	توقيع الممثل القانوني: (في حالة كان المريض غير قادر أو ليس لديه الأهلية القانونية تخوله لتوقيع هذا النموذج ، على سبيل المثال: تحت 18 من العمر).
I am the legal representative of the patient and I have read and understand the information provided to me on this form.	أنا، بصفتي الممثل القانوني للمريض، لقد قمت بقراءة وفهم المعلومات المقدمة لي في هذا النموذج.
Reason Patient Unable to Sign:	سبب عدم قدرة المريض على التوقيع:
Legal Representative Name (If applicable):	اسم الممثل القانوني (إن وجد):
Relationship to Patient:	العلاقة مع المريض:
Legal Representative Signature:	توقيع الممثل القانوني:

SEHA-AF-001-01-DT

General Consent for Healthcare Services- DT Form

Page 4 of 5



Patient Name:	MRN: FIN:
Date and Time:	التاريخ والوقت:
SEHA STAFF SIGNATURE:	توقيع موظف شركة صحة
I have explained to the patient (or legal representative) the purpose of this form, provided sufficient opportunity for questions and have answered any questions to his/her satisfaction.	لقد قمت أنا، بتقديم شرح للمريض (أو الممثّل القانوني) عن الغرض من هذا النموذج ، بالإضافة إلى توفير مساحة كافية لتلقي أي استفسارات وعليه فقد قمت بالإجابة عن أي أسئلة أو استفسارات على النحو الذي يرضيه.
Staff Name:	اسم الموظف:
Electronically Signed By:	التوقيع الإلكتروني:
Date and Time:	التاريخ والوقت:
AFFIRMATION BY STAFF (IF THE PATIENT IS UNCONSCIOUS OR UNABLE TO SIGN)	اقرار من قبل الموظف (في حالة كان المريض غير واعي أو غير قادر على التوقيع)
I confirm that medical staff has informed me that the patient is unable to sign this form on his/her own behalf due to a medical emergency (including unconsciousness)	أقر أنا، بأن الطاقم الطبي قد قام بإبلاغي بأن المريض غير قادر على القيام بالتوقيع على هذا النموذج بصفته الشخصية بسبب حالته الطبية الطارئة (بما في ذلك فقدان الوعي).
Reason Patient Unable to Sign:	سبب عدم قدرة المريض على التوقيع:
Staff Name:	اسم الموظف:
Electronically Signed By:	التوقيع الإلكتروني:
Name of Authorizing Medical Staff:	اسم موظف الرعاية الطبية المخول:
Date and Time:	التاريخ والوقت:
TRANSLATOR SIGNATURE (if any):	توقيع المترجم (إن وجد):
Translator Name:	اسم المترجم :
Translator Signature:	توقيع المترجم:
Date and Time:	التاريخ والوقت:

SEHA-AF-001-01-DT

General Consent for Healthcare Services- DT Form

Page 5 of 5



APPENDIX 1b: GENERAL CONSENT FOR HEALTHCARE SERVICES - HIGHLIGHTED

*(including consent to be contacted for research purposes in box 6)

Version 1, dated 03.10.18

Please see separate attachment for the complete document.

Box 6 (our highlighting) states:

Disclosure of Medical Information: the disclosure and use of my medical information (including copies of my medical record and/or access to my electronic medical record through an electronic device or the internet) to ...

hospital staff and others within and outside of the United Arab Emirates (including clinical data registries) to assess and improve the safety, quality and cost of care provided to me and other patients, assist in the administrative operation of the hospital or conduct medical and scientific research and reviews. Further to IRB approval, patients could be contacted for their interest to take part in the research studies.



APPENDIX1c:EMAILCONFIRMINGAUTHORIZATIONTOCONTACTPATIENTSTHROUGH THE REGISTRY BY TELEPHONE

Version 1, dated 03.10.18

Dear x,

Further to the IRB approval for the study No: 546, the central office would like to confirm that you have been authorized to contact patients through the registry identification process for the recruitment. We further confirm that the practice to call patients for their interest in participation of your study is legitimate and in line of ethics protocol for which you also have been granted the ethics approval.

Hope the email satisfies the ethics committee at your university.

Best wishes,

Central office of Cancer Registry	
NCR	
-	-
شفي توام Tawam Hospital	📖 💙 SEHA y ش
	شركة أبوظبي للخدمات الصحية 🚓
C ⁺⁹⁷¹ CCC ^{PO} Box.15258, Al C ^{ncr@seh}	Abu Dhabi Health Services Co. _{PJ.S.C}
5/6//444 Ain, UAE <u>a.ae</u> <u>a.ae</u>	الله المدر المفاق 📷 التفعرانين المفاق
نا: تطوير نظام صحي متميز للإرتقاء بالخدمات الصحية المقدمة إلى أعلى	مسالت المعادية المعادية المعادية المعادية المعادية المعادية المعادية المعادية المعادية المعادية المعادية المعاد
يات العالمية في الجودة الطبية وخدمة المتعاملين	المستو 🖉
Our Mission: Transform our healthcare system services t	o the
highest medical quality and customer care internat	ional
standards.	



APPENDIX 2a: IRB APPROVAL – TAWAM HOSPITAL



Tawam Human Research Ethics Committee (T-HREC) 22nd January 2018 Date: Ref. No.: SA/AJ/546 To: **Principal Investigator:** Dr. Maria Aamir Oncology Center (Dep) Department: Tawam Hospital Institute: ☑ New Research Study □Amendment □ Extension Subject: □ Revision An exploration of the needs of breast cancer survivors utilizing socio-**Research Title:** economic and clinical indicators-A mixed method study in a hospital setting. Retrospective Chart Review Study Type: Questionnaire/Survey (THREC-546) Ethics Committee Approval #: ☑ Favorable □ Favorable with Conditions Decision: □ Unfavorable **Progress Report Submission** Annual □ 6 Months **Requirement:**

Dear Dr. Aamir,

Study Expiry Date:

The Research Proposal was reviewed by the Ethics Committee members and voted towards the ethics approval.

31 March, 2019

Any ethical concern arising from the study in due course, should be informed. Annual report plus a terminal report are necessary and the Committee would appreciate receiving copies of abstracts and publications.

Studies approved can't be continued beyond the expiry date mentioned above. In case continuation of study is anticipated, extension request in the prescribed form should be submitted to the committee prior to 60 days of expiry date.

The Research Committee has been organized and operates according to the Good Clinical Practice (GCP) guidelines and the Department of Health, Abu Dhabi (DHAD).

It's mandatory to be compliant with the regulatory requirements of the Office of Research Governance ORG, Tawam, whenever required.

Yours sincerely, Sur

Dr. Sultan Al Karam Chair, Tawam Human Research Ethics Committee (T-HREC)



Email: TawamHREC@seha.ae | Tel: +971 3 7072240 | PO Box: 15258



APPENDIX 2b: IRB APPROVAL – TAWAM HOSPITAL



Tawam Human Research Ethics Committee (T-HREC)

02nd October 2019 Date: Ref. No.: AA/AJ/546 To: Dr. Maria Aamir **Principal Investigator:** Department: Oncology Center Institute: Tawam Hospital □ New Research Study □ Amendment ☑ Extension □ Revision An exploration of the needs of breast cancer survivors utilizing Subject: **Research Title:** socioeconomic and clinical indicators-A mixed method study in a hospital setting. Retrospective Chart Review & Questionnaires/Surveys Study Type: Ethics Committee Approval #: THREC-(546) Decision: Favorable Favorable with Conditions □ Unfavorable G 6 Months Annual **Progress Report Submission Requirement:** 30/09/2020 Study Expiry Date:

Dear Dr. Aamir,

Extension

The following documents were reviewed for extension by the Ethics Committee members and voted towards the ethics approval.

Extension.				
Extension Expiry Date:	30 September 2020			

Any ethical concern arising from the study in due course, should be informed. Annual report plus a terminal report are necessary and the Committee would appreciate receiving copies of abstracts and publications.

Studies approved can't be continued beyond the expiry date mentioned above. In case continuation of study is anticipated, extension request in the prescribed form should be submitted to the committee prior to 60 days of expiry date.

The Research Committee has been organized and operates according to the Good Clinical Practice (GCP) guidelines and the Department of Health, Abu Dhabi (DHAD).

It's mandatory to be compliant with the regulatory requirements of the Office of Research Governance ORG, Tawam, whenever required.

	TAWAM HOSPITAL
Yours sincerely,	JOINS HOPKINS SEILA - THIREC
Dr. Aysha Al Kaabi	لجنة أخلاقيات البحوث الإنسانية - مستشفى توام
Chair, Tawam Human Research Ethics Committee (T-HREC)	Tawam Human Research Ethics Committee
P.S. Primary investigator using data collectors from outside the hospital are required to	obtain HR department approval as visito

ــَشــفى توام - العين. أبوظــبى ص.ب: 15258 - الإمـارات العربية المـتحدة - هاتف : 7677444 3 00971 6 فاكس : 7677634 00971 3 Tawam Hospital - Al Ain - Abu Dhabi. P.O. Box 15258 - United Arab Emirates - Tel : 00971 3 7677444 - Fax: 00971 3 7677634

مىحة 🍤 seha



APPENDIX 2c: IRB APPROVAL – UNIVERSITY OF SALFORD



Research, Enterprise and Engagement Ethical Approval Panel

Doctoral & Research Support Research and Knowledge Exchange, Room 827, Maxwell Building, University of Salford, Manchester M5 4WT

T +44(0)161 295 2280

www.salford.ac.uk

22 February 2019

Dear Maria,

<u>RE: ETHICS APPLICATION–HSR1819-016 – 'An exploration of the psychosocial needs of breast</u> cancer survivors and its association with cultural differences and clinical indicators – A mixed method study in a hospital setting.'

Based on the information that you have provided, I am pleased to inform you that ethics application HSR1819-016 has been approved.

If there are any changes to the project and/or its methodology, then please inform the Panel as soon as possible by contacting <u>Health-ResearchEthics@salford.ac.uk</u>

Yours sincerely,

dhy An.

Professor Sue McAndrew Chair of the Research Ethics Panel



APPENDIX 3a & 3b: CONSENT FORM (ENGLISH)

Version 2b, dated 04/02/19

Title of study: Identifying the psychological needs of Emirati and expatriate breast cancer survivors in the U.A. E.: A mixed method investigation in a hospital setting.

Name of Researcher: Maria Aamir Please complete and sign this form after you have read and understood the study information sheet. Read the following statements, and select 'Yes' or 'No' in the box on the right hand side.				
1.	I confirm that I have read and understand the study information sheet	Yes/No		
	Version 2b, dated 22.01.19, for the above study.			
	I have had the opportunity to consider the information and to ask questions			
	which have been answered satisfactorily.	TT AT		
2.	I understand that my participation is voluntary and that I am free to	Yes/No		
	withdraw at any time, without giving any reason, and without my rights being affected.			
3.	If I do decide to withdraw I understand for my data to be removed	Yes/No		
	from the study, I must inform the researcher of my decision within			
	30 days of my participation.	Ves/No		
4.	I agree to participate by being interviewed and to having my interview	105/100		
	audio-recorded.			
5.	I understand that my personal details will be kept confidential and will not be			
	revealed to people outside the research team.	Yes/No		
6.	I understand that my anonymised data will be used in the researcher's thesis, resea	rch reports,		
	other academic publications, conferences/presentations, reports to Tawam Hopsita	al, Ministry		
	of Health, and further research.			



7.	If I choose to participate in	my native language	rather than English, I agree to	Yes/No
	having a translator present.			
				Yes/No
8.	I agree to take part in the st	udy.		
Name o	of participant	Date	Signature	
Name o	of person taking consent	Date	Signature	



APPENDIX 3b: CONSENT FORM - ARABIC

نموذج الموافقة

المشارك يرجى ملء النموذج بأكمله

نحن ندعوك للمشاركة في الدراسة البحثية التي أعدّها مجموعة من المهنين العاملين في مجال الصحة داخل مستشفى توام بغرض البحث والكشف عن احتياجات الناجيات من سرطان الثدي ومن أجل تقديم أفضل وأعلى مستوى للرعاية لمرضى السرطان في المستشفى .

قبل موافقتك على المشاركة في الدراسة ، نطلب منك قراءة ورقة معلومات المشترك ودائرة نعم/لا حسب الاقتضاء. إذا كانت لديك أي أسئلة أو استفسارات قبل التوقيع على نموذج الموافقة ، فيرجى التحدث إلى المحقق ماريا عامر.

أؤكد أنني قد قر أت ورقة المعلومات وأدركت تمامًا ما هو متوقع مني في هذه الدراسة. [نعم / لا] أؤكد أنه أتيحت لي الفرصة لطرح أي أسئلة وللرد عليها. [نعم / لا] أدرك أن إجاباتي والمعلومات التي سأقدمها سيتم الاحتفاظ بها حتى يتم فحص مشر وع البحث. [نعم / لا] أدرك أن إجاباتي والمعلومات التي سأقدمها سيتم الاحتفاظ بها حتى يتم فحص مشر وع البحث. [نعم / لا] أدرك أن مشاركتي طوعية وأنني حر في الانسحاب في أي وقت دون إبداء أي سبب ، دون أن تتأثر ر عايتي الطبية أو حقوقي القانونية. [نعم / لا] أدرك أنه بمجرد أن يتم الكشف عن بياناتي مجهولة الهوية والنتائج التي توصلت إليها ، قد لا يكون من الممكن سحبها، على الرغم من أنه سيتم بذل كل محاولة لاستخراج بياناتي ، حتى نقطة النشر. [نعم / لا] أدرك أن المعلومات الواردة في الاستطلاع سيتم تجميعها مع ردود المشاركين الأخرين ، وسيتم الكشف عن هويتهم أدرك أن المعلومات الواردة في الاستطلاع سيتم تجميعها مع ردود المشاركين الأخرين ، وسيتم الكشف عن هويتهم أدرك أن المعلومات الواردة في الاستطلاع سيتم تجميعها مع ردود المشاركين الأخرين ، وسيتم الكشف عن هويتهم أوقاق على المعلومات الواردة من ردود المسح التي تستخدمها في التقارير و المؤتمرات و الفعاليات التدريبية. أوافق على المعلومات الواردة من ردود المسح التي تستخدمها في التقارير و المؤتمرات و الفعاليات التدريبية.


يكن هناك اعتقاد بوجود خطر على نفسي أو ذه المعلومات مع مشرف أبحاثها. [نعم / لا]	، سرية ومجهولة تمامًا ما لم احث الرئيسي إلى مشاركة ها	معلومات سأقدمها سنبقر ي هذه الحالة سيحتاج الب	أدرك أن أي للأخرين ، وف
مد انتهاء الدر اسة. [نعم / لا]	معلوماتي لمدة سنة بع	ستشفى توام الذي يحتفظ ب	أوافق على مد
	لذه الدراسة	به على المشاركة في ه	أوافق بموجب
التاريخ	التوقيع	اسم المشارك	
التاريخ	التوقيع		اسم الباحث_



APPENDIX 4a: PATIENT INFORMATION SHEET - ENGLISH

Version 2, dated 04.02.19

Study Title

Identifying the psychological needs of Emirati and expatriate breast cancer survivors in the U.A. E.: A mixed method investigation in a hospital setting.

Invitation paragraph

I would like to invite you to take part in a research study. Before you decide, you need to understand why the research is being done and what it would involve for you. Please take time to read the following information carefully. Ask questions if anything you read is not clear or you would like more information. Take time to decide whether or not to take part.

Please take time to read the following information carefully and contact us for any questions if some part of the information is not clear to you or if you would like more information. You may decide to be in the study or not take part at all. If you do decide to take part in this study, you may stop at any time. However, before you decide, it is important that you understand why this research is being done and what it will involve. Your decision will not lead to any penalty or affect your regular medical care or any benefit to which you are otherwise entitled.

What is the purpose of the study?

The purpose of this study is to find out from patients what your supportive care needs are and have been during your illness. We want to know what kind of support might make your situation better and help you cope with the illness.

Why have I been invited?

You have been invited as you have been registered and treated at Tawam Hospital for the breast cancer and during the registration you signed to give consent to participate in health research which could benefit the community. The study is conducted by a PhD student Maria Aamir who is undertaking the research at the Salford University of Salford, UK to understand and explore the psychosocial needs of breast cancer survivors who have been treated at Tawam Hospital.

Do I have to take part?



'It is up to you to decide. We will describe the study and go through this information sheet which we will give to you. We will then ask you to sign a consent form to show that you have agreed to take part. You are free to withdraw at any time, without giving a reason and this will not affect the standard of care you receive.

If you decide to withdraw from the study after the completion of survey, and then simply inform the researcher within thirty days, your responses will be deleted and all related information will be destroyed; none of your information will be used in the study.

What will happen to me if I take part?

If you decide to participate in this study, you will be given a consent form to sign to indicate that you are agreeing to take part in this study voluntarily. Along with this, you will need to fill out a questionnaire which will take about 30 minutes to complete. Your responses will help us understand what has been important for you that could ease your life. The information to be asked about will include:

- Your needs including information, financial, medical, relationship, emotional and any other need that you think important for you.

- Personal details such as your age, gender, type of cancer, type of cancer treatment and when you commenced treatment will have been collected as well.

Some of you will be invited to a further interview after the questionnaire results are analysed. The choice of interview participants will be made to reflect a range of responses on the questionnaires and the purpose of the interview will be to explore in more depth and detail aspects of your personal experiences. The length of the interview will be 45-90 minutes depending on how detailed your responses are, and the interviews will be audio recorded and then transcribed. The interviews will take place in a private room in the hospital. Whether or not you are asked to participate in an interview and whether or not you choose to do so, you will continue with your cancer care follow up plan as per normal department follow up.

For both the questionnaires and the interviews, a hospital translator trained in confidentiality and data protection issues will be available if you have difficulty communicating in English and would prefer to use your native language. These translators have been trained to consider the sensitive nature of the survey and interview questions.



Expenses and payments?

As compensation, there are no expenses or payments to take part in the study however refreshments will be available free during the course of study which is during the survey and interview process.

What are the possible disadvantages and risks of taking part?

We will try to be very cautious with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if you need to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.

What are the possible benefits of taking part?

There may be no direct benefit to participation in this study but you may find it a positive experience to participate in this research because your participation can benefit you and others who have similar symptoms and experiencing the same challenges through their illness. Also, the study will offer an opportunity to improve cancer services and patient care by understanding what are the different type of needs you have in order to improve the way care is delivered and to prepare individuals more appropriately in the future.

What if there is a problem?

If you have a concern about any aspect of this study, you should ask to speak to the researcher (Maria Aamir, contact: 037072686) who will do their best to answer your questions. If you remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor (Dr. Luai Ahmed (luai.ahmed@uaeu.ac.ae) or Dr Linda Durbow Marshall (l.dubrow-marshall@salford.ac.uk). If the matter is still not resolved, please forward your concerns to Professor Susan McAndrew, Chair of the Health Research Ethical Approval Panel, Room MS1.91, Mary Seacole Building, Frederick Road Campus, University of Salford, Salford, M6 6PU. Tel: 0161 295 2278. E: s.mcandrew@salford.ac.uk.

Will my taking part in the study be kept confidential?

The researchers will need to collect personal data about you, which may be sensitive (for instance, date of birth and relevant health information). The researchers may also need to get some of your health information from your medical records. Any personal or health information will be kept private and confidential. It will be stored securely and accessible to authorised persons to be kept confidential. Your study details will be given a number so that your identity will not be apparent. The study records will be kept in a locked filing cabinet in the researcher's office at Tawam Hospital during the study and



in a locked archive for at least 10 years from the time the study is closed and will be destroyed at any time thereafter.

Authorised representatives of the study and the Research Ethics Committees may require access to your study records to verify study procedures and/or data. In all cases when dealing with your information, these people are required to comply with privacy laws that protect you. We will ensure that all the information you give us will be held in strictest confidence and will not be passed on to a third party in a form which can lead to the identification of individuals. All information which is collected about you during the course of the research will be kept strictly confidential, and any information about you which leaves the hospital will have your name and address removed so that you cannot be recognised.

Any data from the questionnaires and interviews will be kept on a secure university drive for three years, unless you ask for the information to be destroyed within 30 days of your participation. Your confidentiality will be further maintained by assigning a code number instead of your name to the information which you have provided either in the surveys or in the interviews.

What will happen if I don't carry on with the study?

If you withdraw from the study all the information and data collected from you, to date, will be destroyed provided you withdraw 30 days after your participation and your name will be removed from all the study files. There will be no negative consequences to you for withdrawing, and you will continue with your cancer care follow up plan as per normal department follow up.

What will happen to the results of the research study?

You will be sent a summary of the findings from this study. The results of the research will be made available to health professionals through medical journals or meetings, but you will not be identifiable in these communications. By taking part in this study you agree not to restrict the use of any data but only if you withdraw within thirty days.

Who is organising or sponsoring the research?

The Al Ain medical district Institution Board Review (IRB) committee and Salford University's School of Health and Society Research Ethics Committee have reviewed this study and given approval for this research. If you have any ethical concerns regarding the study, you can contact the secretary at the Tawam Hospital Institution Board Review (IRB) committee at Research Office on telephone number (+971 03 707 4739).



Further information and contact details

The study is conducted by the student Maria Aamir who is undertaking the research at the University of Salford, UK.

Contact persons:

Should you have questions about the study you may contact: Maria Aamir Phone No 037072686 email: <u>maamir@seha.ae</u>



APPENDIX 4b: PATIENT INFORMATION SHEET - **ARABIC**

نموذج معلومات المريض

نحن ندعوك للمشاركة في الدراسة البحثية التي أعدّها مجموعة من المهنين العاملين في مجال الصحة داخل مستشفى توام بغرض البحث والكشف عن احتياجات الناجيات من سرطان الثدي ومن أجل تقديم أفضل وأعلى مستوى للرعاية لمرضى السرطان في المستشفى.

ماء الباحثين:

ماريا عامر

يرجى أخذ وقتك لقراءة المعلومات التالية بعناية والاتصال بنا للحصول على أي سؤال إذا كان جزء من المعلومات غير واضح لك أو إذا كنت تر غب في مزيد من المعلومات. يرجى قراءة هذا قبل التوقيع على نموذج الموافقة هذا.

من يقوم بهذه الدراسة؟ هذه در اسة لمستشفى توام لفهم احتياجات مرضى السرطان ولتطوير تدخل يلبي تلك الاحتياجات غير الملباة. أجريت الدر اسة في الأساس من قبل الدكتورة ماريا عامر التي تجري البحث في مستشفى توام.

> للتنسيق والاتصال: إذا كانت لديك أسئلة حول الدر اسة ، فيمكنك الاتصال بـ: د. ماريا عامر هاتف رقم 03-707 2686

سيتم تزويد جميع المشاركين في الدر اسة بنسخة من ورقة المعلومات والموافقة نموذج لسجلاتهم الشخصية.

قد تقرر أن تكون في الدراسة أو لا تشارك على الإطلاق. إذا قررت المشاركة في هذه الدراسة ، فيمكنك التوقف في أي وقت. ومع ذلك ، قبل أن تقرر ، من المهم أن تفهم سبب إجراء هذا البحث وما الذي سيتضمنه.

لن يؤدي قرارك إلى أي عقوبة أو التأثير على رعايتك الطبية المنتظمة أو أي فائدة يحق لك الحصول عليها.

تشرح ورقة المعلومات التالية الدراسة وستنضمن تفاصيل مثل: - لماذا هذه الدراسة قد تكون مناسبة لك - طبيعة مشار كتك



- حقوقك ومسؤولياتك

ما هو الغرض من الدراسة؟ تهدف الدر اسة إلى استكشاف الاستخدام المحتمل لمنظمة العفو الدولية للتنبؤ بنتائج سرطان الثدي وتطوير خطط الر عاية الألية بناءً على الاحتياجات الفريدة للناجين.

ما هي مشاركتي في هذه الدراسة ، فسيتم إرسال نموذج الموافقة الذي ستوقعه بالموافقة على المشاركة في هذه إذا قررت المشاركة في هذه الدراسة ، فسيتم إرسال نموذج الموافقة الذي ستوقعه بالموافقة على المشاركة في هذه الدراسة طواعية. مع هذا ، ستحتاج إلى ملء استبيان سيستغرق استكماله حوالي 45 دقيقة إلى ساعة. ستساعدنا ردودك على فهم ما هو مهم بالنسبة لك والذي قد يخفف من حياتك. سيتم طرح المعلومات حول: - احتياجاتك بما في ذلك المعلومات ، والمالية ، والطبية ، والعلاقات ، والعاطغية وأي حاجة أخرى تعتقد أنها مهمة بالنسبة لك.

- التفاصيل الشخصية مثل عمرك وجنسك ونوع السرطان ونوع علاج السرطان ومتى بدأت العلاج سيتم جمعها أيضًا.

سوف تتوقف مشاركتك في الدر اسة بعد الانتهاء من الاستبيان.

هل هناك أي أسباب لا ينبغي أن أكون في هذه الدراسة ؟ لا ما هي التكاليف بالنسبة لي ؟ لن يتم تكبد أي تكاليف شخصية.

ما هى فوائد ممكنة من المشاركة؟

قد لا تكون هناك فائدة مباشرة للمشاركة في هذه الدراسة ولكن قد تجد أنها تجربة إيجابية للمشاركة في هذا البحث لأن مشاركتك يمكن أن تفيدك أنت والآخرين الذين لديهم أعراض مماثلة وتعاني من نفس التحديات من خلال مرضهم. كما ستوفر الدراسة فرصة لتحسين خدمات السرطان ورعاية المرضى من خلال فهم الأنواع المختلفة للاحتياجات التي تحتاج إليها من أجل تحسين طريقة تقديم الرعاية وإعداد الأفر اد بشكل أكثر ملاءمة في المستقبل.

كيف سيتم ضمان سلامتي؟



لن تؤثر المشاركة في هذه الدراسة على سلامتك الشخصية لأنك سترد على الاستبيان عن بُعد. ما هي البدائل التي يجب علي الذهاب إليها في هذه الدراسة؟ نود منك المشاركة ، ولكن إذا كنت لا تر غب في ذلك ، فلديك مطلق الحرية في الرفض. وبالمثل ، إذا كنت قد وافقت في البداية ، ثم ر غبت في الانسحاب ، فيمكنك القيام بذلك دون أي مشكلة. في كلتا الحالتين لن نطلب أي تفسير ، ولن تكون هناك أي آثار سلبية على الإطلاق بالنسبة لك. لن يؤثر هذا على الرعاية الحالية أو العلاج الطبي الذي نتلقاه بأي شكل من الأشكال.

ما هي الآثار الجانبية المحتملة والمخاطر ومضايقات المشاركة؟ سنحاول أن نكون حذرين للغاية فيما يتعلق بالأسئلة المطروحة ، لكن هذا موضوع حساس للكثيرين وفي بعض الأحيان ، قد يجده البعض كتجربة عاطفية أو مرهقة. في هذه الظروف ، إذا احتجت إلى إحالتك إلى أخصائي صحي للحصول على الدعم العاطفي ، فسوف يضمن الباحث حدوث ذلك في أقرب وقت ممكن. سيتم تقديم هذا الدعم دون أي تكلفة.

ماذا يحدث إذا غيرت رأيي بعد أن أرسلت إجاباتي وأرغب في الانسحاب من الدراسة؟ إذا قررت الانسحاب من الدراسة بعد الانتهاء من الدراسة ، ثم أخبر الباحث ببساطة وسيتم حذف إجاباتك وسيتم إتلاف جميع المعلومات ذات الصلة ، ولن يتم استخدام أي من معلوماتك في الدراسة.

ماذا يحدث في نهاية الدراسة؟ في نهاية الدراسة ، سوف تستمر في متابعة خطة متابعة العناية بالسرطان حسب متابعة القسم المعتادة.

هل تبقى مشاركتي في هذه الدراسة سرية؟

سيحتاج الباحثون إلى جمع بيانات شخصية عنك ، والتي قد تكون حساسة (على سبيل المثال ، تاريخ الميلاد والمعلومات الصحية ذات الصلة). قد يحتاج الباحثون أيضًا إلى الحصول على بعض المعلومات الصحية الخاصة بك من سجلاتك الطبية. سيتم الاحتفاظ بأي معلومات شخصية أو صحية خاصة وسرية. سيتم تخزينه بشكل آمن ومتاح للأشخاص المخولين للحفاظ على سريته. سيتم إعطاء تفاصيل در استك رقماً حتى لا تكون هويتك واضحة. سيتم الاحتفاظ بسجلات الدر اسة في خزانة ملفات مغلقة في مكتب الباحث في مستشفى توام أثناء الدر اسة وفي أر شيف مغلق لمدة 10 سنوات على الأقل من وقت إغلاق الدر اسة وسيتم إي وقت بعد ذلك.



قد يطلب الممثلون المعتمدون للدر اسة ولجان أخلاقيات البحث الوصول إلى سجلات الدر اسة للتحقق من إجراءات الدر اسة و/ أو البيانات. في جميع الحالات عند التعامل مع معلوماتك ، يتعين على هؤ لاء الأشخاص الامتثال لقوانين الخصوصية التي تحميك. سنضمن أن جميع المعلومات التي تقدمها لنا سيتم الاحتفاظ بها بسرية تامة ولن يتم نقلها إلى جهة خار جية في شكل يمكن أن يؤدي إلى تحديد هوية الأفر اد.

هل سأكتشف نتائج الدراسة؟

سيتم إرسال ملخص للنتائج التي توصلت إليها هذه الدراسة. ستتاح نتيجة البحث للمهنيين الصحيين من خلال المجلات الطبية أو الاجتماعات ، لكنك لن تكون محددًا في هذه الاتصالات. بالمشاركة في هذه الدراسة ، فإنك توافق على عدم تقييد استخدام أي بيانات ولكن فقط في حالة الانسحاب.

من سيراجع الدراسة؟

ستقوم لجنة مراجعة مجلس إدارة مستشفى توام بمراجعة هذه الدراسة وإعطاء الموافقة على إجراء هذا البحث. إذا كانت لديك أية مخاوف أخلاقية بشأن الدراسة ، فيمكنك الاتصال بلجنة مراجعة مجلس أمين السر (IRB) على رقم الهاتف (7072686 03 971+)

بمن أتصل وكيف إذا كان لدي أي أسئلة؟ إذا كان لديك أي أسئلة حول الدراسة ، يمكنك الاتصال بـ: Maria Aamir Tel: +971 03 707 2686 (من 7:00 ص إلى 4:00 م) ؛ البريد الإلكتروني: maamir@tawamhospital.ae



APPENDIX 5a: CANCER SURVIVORS' NEED SURVEY – ENGLISH

Version 1, dated 03.10.18

We invite you to take part in a research study at Tawam Hospital. This study is conducted by the group of health professionals to investigate the needs of breast cancer survivors to help provide the optimal and high-quality cancer care at the hospital.

First, we want you to know that taking part in this study is entirely voluntary and you may receive no benefit from taking part. Nevertheless, your contribution to this research is very important as this research will provide us knowledge that may help people and their health in the future.

There are ten major sections to complete in this questionnaire. It requires you to simply selecting the most appropriate answer from the choices given. The questionnaire will not take more than 30 minutes to answer. Please **complete this survey** carefully by giving your answers accurately and as fully as possible. Please be assured that all replies will be anonymised and you will not be identified at any stage to any third party. Your answers to this questionnaire will be strictly **CONFIDENTIAL** to the Tawam Hospital Research team only and will not be given to anyone else without your written permission. We would like to **THANK YOU** for taking part in the study.

Please complete this survey and return to the researcher. If you have difficulty in interpreting any question, please ask the researcher or the coordinator.

Thank you again. If you would like to have further information regarding any aspect of this study, please contact our research team and we would be delighted to provide you further information.

WHAT ARE YOUR NEEDS?

This survey of needs will provide awareness and insight into the needs of cancer survivors and provide leaders with valuable information needed to provide appropriate resources and plan supportive programs. Whether you completed treatment ten days or ten years ago, or continue to receive treatment, it's important we hear from you. Please take a few moments to complete this survey and share your thoughts with us.

Thank you.



SECTION 1: DEMOGRAPHICS (7 Items)

- Age: ____(Yrs)
- Nationality:
 - Emirati (1)
 - Expatriate (2)
- Year of cancer diagnosis _____
- Type of cancer _____
- Stage of cancer _____ (I, II, III IV, Unknown)
- Type of treatment received (check all that apply)
 - Surgery (1)
 - o Radiation (2)
 - \circ Chemotherapy (3)
 - \circ Hormonal (4)
 - \circ Other (8)
- Currently receiving treatment?
 - Yes (1)
 - No (0)

SECTION 2: QUALITY OF LIFE (1 Item)

Please circle the number (0-10) that describes your overall Quality of Life during the past week, including today:

0	1	2	3	4	5	6	7	8	9	10
As bao	l as									As good as
it can	be									it can be

As a cancer survivor you may experience some lasting side effects from your treatment. Please rate each topic according to how much concern it is causing you **today**.

Please circle the number on the scale from 0 (no concerns) to 5 (extreme concerns).

SECTION 3: PHYSICAL CONCERNS (20 Items)

	No Concerns					Extreme Concerns
	0	1	2	3	4	5
Pain						
Fatigue						
Sleep disturbance						
Memory and concentration						
Nausea / Vomiting						



Poor appetite			
Trouble swallowing			
Dental or mouth problems			
Weight changes			
Balance / Walking / Mobility			
Loss of strength			
Tingling or numbness in feet & hands (neuropathy)			
Swelling in legs or arms (lymphedema)			
Osteoporosis / Bone health			
Hair and skin care concerns			
Body changes			
Bowel or bladder changes			
Hot flashes			
Sexual concerns			
Fertility concerns			

SECTION 4: INFORMATION CONCERNS (7 Items)

	No Concerns					Extreme Concerns
	0	1	2	3	4	5
Being fully informed about my cancer and treatment						
Being fully informed about potential side effects and long-term adverse effects						
Being fully informed about any familial risk						
Being fully informed about any symptom appear (recurrence/new primary)						
Being fully informed about who to contact if any symptom appears (recurrence/new primary)						
Being fully informed about next follow- up date, place and physician						
Being fully informed about the Psychosocial support and resources to other support services						

SECTION 5: SOCIAL & FINANCIAL CONCERNS (6 Items)

No	Extrem	ie
Cor	cerns Concer	ms



	0	1	2	3	4	5
Managing household activities						
Caring for family members						
Talking about cancer with family and friends						
Returning to work						
Health insurance						
Financial concerns						

SECTION 6: EMOTIONAL CONCERNS (14 Items)

	No					Extreme
Defining a new sense of normal	0	1	2	3	4	5
Managing difficult emotions: anger, fear, sadness,						
depression, guilt, anxiety)						
Coping with grief and loss						
Living with uncertainty						
Fear of recurrence						
Managing stress						
Isolation / Feeling alone						
Intimacy issues						
Looking for the bright side: gratitude, forgiveness, love,						
happiness, contentment						
Having a sense of well being						
Changing relationships with spouse, family, and others						
Finding support resources						
Connecting to counseling services						
Genetic counseling (worry about your children getting cancer)						

SECTION 7: SPIRITUAL/RELIGIOUS CONCERNS (4 Items)

	No Concerns					Extreme Concerns
Religious or spiritual support	0	1	2	3	4	5
Religious or spiritual distress						
Loss of faith						



End of life concerns

SECTION 8: OTHER CONCERNS (1 Items)

Any other concerns that you may have, please specify

SECTION 9: Please indicate how you prefer to learn about the following topics. Check all that apply (4 Items)

	Written Materials	Class	Video	Healthcare Specialist	Cancer Survivor	Internet
Physical issues						
Social issues						
Emotional						
issues						
Spiritual issues						

SECTION 10: What specific topics are you interested in learning about? (Circle topics of interest) (4 Items)

HEALTHY LIVING CHOICES:	Nutrition	Safe exercise	Smoking cessation
FINANCIAL CONCERNS:		Living wills	Disability
ENHANCING COMMUNICATION:	With your doctors	With your spouse or significant other	
HEALTH SCREENINGS:	Cancer	Heart	

SECTION 11: What has been your primary source of strength during your cancer experience? (1 Items)



Additional comments:

Thank you for completing this survey. Your comments will be beneficial as we plan for the future needs of cancer survivors.

For more information regarding this survey, please contact Maria Aamir at 037072686 or maamir@seha.ae



APPENDIX 5B: CANCER SURVIVORS' NEED SURVEY - ARABIC

نحن ندعوك للمشاركة في الدر اسة البحثية التي أعدّها مجموعة من المهنين العاملين في مجال الصحة داخل مستشفى توام بغرض البحث والكشف عن احتياجات الناجيات من سرطان الثدى ومن أجل تقديم أفضل لمرضى السرطان في المستشفى. للر عابة وأعلى مستوى في البداية نودُ أن نشير بأن المشاركة في هذه الدراسة ستكون طوعية تمامًا ولن يتحصَّل المشارك على أي استحقاق أو مكافأة مقابل الإجابة أو المشاركة في هذا الاستبيان. ومع ذلك فتعتبر مشاركتك في هذه الدر اسة مهمّة جدًا لأنها سوف توفر لنا المعلومات والمعرفة التي ستسهم المستقبل. فى الناس من الكثير مساعدة وصحتهم فى يحتوى هذا الاستبيان على 10 أقسام رئيسة، تتطلب الإجابة واستكمال البيانات. وسيتطلب منك الاستبيان المقدمة. مبسط أن تقوم باختيار الإجابة الأكثر ملائمة من الخيارات و بشکل منك الإجابة على الاستبيان أكثر من 30 تستغر ق دقبقة لن بقدر المستطاع. على الاستبيان بشكل دقيق وكامل الإجابة منك نر جو نوّد أن نؤكد لكم بأنه لن يتم التعريف أو الكشف عن هويتك أو شخصك عند أي مرحلة و لأي فرد من أفراد الثالث الطرف وسيحتفظ فريق قسم الأبحاث في مستشفى توام بجميع إجابتك بشكل سرّي، ولن يتم تداولها مع أي شخص أخر من دون الحصول مسبقًا على إذن كتابي منك. ونحن نتقدم لك بجزيل الشكر على المشاركة في هذه الدر اسة. فضلًا بعد استكمال الإجابة على الاستبيان يرجى تسليمه للباحث، وفي حال واجهتك أي صعوبات في فهم ترجمتها يرجى الرجوع للباحث أو المنسّق لطلب المساعدة. الأسئلة أو شكرًا لك مجددًا وللمزيد من الأسئلة أو المعلومات المتعلِّقة بهذه الدراسة يرجى الاتصال بفريق البحث وسنكون سعداء عند تقديم المزيد من المعلومات والمساعدة. ماهي احتياجاتك؟ إن هذه الدراسة والاستبيان المتعلق باحتياجاتك سوف يوفر التوعية وسيسهم في إلقاء نظرة فاحصة على ا احتياجات الناجين من السرطان كما ستقدم للمسؤولين المعلومات القيّمة والمفيدة لتوفير الموارد المناسبة ا خطط الدعم. وبرامج رأيك وإجاباتك تهمنا، سواء أكملت العلاج منذ 10 أيام أو 10 سنوات أو لاتزال تتلقّى العلاج. فضلًا خصص بعضًا من وقتك للإجابة عن الاستبيان وشاركنا أرائك ومقترحاتك. شکر ً ا القسم الأول: المعلومات الديمغر افية (7 بنود)

القسم 1:

العمر: سنة



الجنسية:

سنة تشخيص إصابتك بالسرطان :....

نوع السرطان:..... نوع العلاج الذي تتلقاه (يرجى وضع إشارة عند كل ماينطبق)

الجراحة (1)

العلاج الإشعاعي (2)

العلاج الكيمائي (3)

العلاج الهرموني (4)

أنواع أخرى من العلاج (8)

هل تتلقى العلاج حاليًا؟

نعم
 لا
 القسم (2): نوعية الحياة (بند 1):

فضلًا ضع دائرة على الأرقام من 0-10 والتي تصف نوعية حياتك بشكل عام أثناء الأسبوع السابق بمافي ذلك اليوم :

السوء مايمكن



قد تصاب بأعراض دائمة ناتجة عن العلاج كمريض / مريضة ناجي/ة من السرطان. فضلًا يرجى تقييم كل موضوع بناءًا على مدى مقدار المخاوف والقلق الذي سببته لك هذه الأعراض اليوم

يرجى وضع دائرة حول الرقم على المقياس من 0 (ليس هناك مخاوف أو قلق) إلى رقم 5 (مخاوف شديدة)

القسم 3 المخاوف البدنية (20 بندًا)

المخاوف والقلق	4	3	2	1	لاتوجد	المخاوف و القلق
الشديد					مخاوف ()	الشديد من
5						
5						
						الألم
						b
						الاعياء والتعب
						اضطراب النوم
						الذاكرة وقلة التركيز
						الغثيان / القيئ (
						الاستفراغ)
						ضعف الشرورة
						صعوبة ومشاكل في
						البلع
						مشاكل في الفم أو
						الأسنان
						تغييرات في وزن
						الجسم
						\ ·



التوازن / المشي /			
الحركة			
فقدان القوة			
الشعور بالخدران			
والوخز في الأقدام			
واليدين (أعتلال			
عصبی)			
~ ~			
تورّم الساقيين أو			
الذراعين (استسقاء			
لمفاوي)			
هشاشة العظام / صحة			
العظم			
مخاوف وقلق بشأن			
صحة الشعر والجلد			
تغييرات في الجسم			
تغيير ات في المثانة			
والأحشاء (الأمعاء)			
هَبَّةُ الْحَرارَة (هوت			
فلاش)			
			-
شواغل ومخاوف			
جنسية			



القسم الرابع : المخاوف والقلق المعلوماتي 7بنود

المخاوف والقلق	4	3	2	1	لاتوجد مخاوف	
الشديد					0	
5						
3						
						على علم تام بشأن
						إصابتي بمرض
						السرطان والعلاج
						المقرر لي
						علی علم تام
						بالأضرار
						الجانبية المحتملة
						aliale le
						على علم تام
						بېخىمە «ت
						خطورة الإصابه
						الوراثية
						على علم تام
						بإمكانية ظهور
						أعراض (معاودة
						/ دردة و أوارة)
						(
						على علم تام
						بالشخص الذي
						يجب التواصل
						معه عند ظهور



			أي أعراض
			(معاودة أو جديدة
			وأولية)
			علی علم تام
			بتفاصيل الموعد
			التالي للمتابعة :
			تاريخه ومكانه
			واسم الطبيب
			in diale le
			على علم لام عل
			الدعم الإجتماعي
			والنفسي وموارد
			خدمات الدعم
			الأخرى

القسم الخامس : المخاوف والمشاغل الإجتماعية والمالية: (6 بنود)

المخاوف والقلق الشديد 5	4	3	2	1	لاتوجد مخاوف (
						المقدرة على القيام بالأعمال المنزلية
						رعاية أفراد الأسرة
						التحدث مع الأسرة والأصدقاء عن السرطان
						العودة للعمل
						التأمين الصحي
						المخاوف والقلق الخاص بالأمور المالية



القسم السادس : المخاوف والمشاغل العاطفية 14 بند

المخاوف والقلق	4	3	2	1	لاتوجد	المخاوف والقلق الشديد من
الشديد					مخاوف	
					0	
5						
						وجود شعور جديد طبيعي
						التغلب على المشاعر الصعبة :
						الغضب والخوف
						والحزن والاكتئاب والشعور بالذنب
						والقلق)
						التغلب على مشاعر الحزن والخسارة
						5 (<i>m</i> (
						عدم وجود استفرار معيشي
						الخوف من تكرار الإصابة أو معاودة
						المرض)
						التغلب على التو تر
						العزلة / الشعور بالوحدة
						مشاكل في العلاقة الحميمية
						البحث عن الجانب الإيجابي : الشكر
						مالحفه / التسامح والحب والفرح
						والمعق (، شمالت و ، شب و ، شرع
						والفناعة
						الشعور بالارتياح والرفاهية



حدوث تغيير في العلاقة مع الشريك				
في الحياة أو الأسرة أو الأخرين				
وجود مصادر للدعم				
الانضمام إلى خدمات المواساة والدعم				
الاستشارة الجينية (القلق من إصابة				
الأطفال بالسرطان)				
	1			

القسم السابع : المخاوف والمشاغل الدينية والروحانية (اربع بنود)

و القلق الشديد 5	المخاوف	4	3	2	1	لاتوجد مخاوف 0	المخاوف والقلق الشديد من
							وجود دعم روحي و ديني
							الضائقة الروحية والدينية
							فقدان الإيمان
							المخاوف المتعلقة بنهاية الحياة

القسم الثامن :اهتمامات ومخاوف أخرى بند واحد

هل لديك أي مخاوف أو اهتمامات أخرى، فضلًا أذكرها :

القسم التاسع : : فضلًا أذكر كيف تفضل أن نخبرك عن المواضيع التالية التي تهمك، ضع علامة أمام الموضوع الذي يهمك (4 بنود):

عن طريق شبكة	أحد الناجين	اختصاصي الحمادة	الفددية	محاضر ة/	المواد التثقيفية	وجود الدعم
المعلومات ۱۷۱کتر نیبة	من السببية ال	الرعاية المحية	الفيديو	الصف	التلقيفية المكتوبة	الروح الدين
الإلكترونية	السرطان					والديبي



			المشاكل الجسدية والبدنية
			المشاكل الإجتماعية
			المشاكل العاطفية
			المشاكل الروحانية

القسم العاشر : فضلًا حدد المواضيع التي تشغلك وترغب بمعرفة المزيد عنها، ضع علامة أمام الموضوع الذي يهمك (4 بنود):

الامتناع عن التدخين	التمارين الأمنة	التغذية	خيارات الحياة الصحية
العجز	وصايا البقاء		المشاكل المادية
	مع شريكك في الحياة أو أشخاص مهمين آخرين	مع أطبائك	تعزيز التواصل
	القلب	السرطان	الفحص الكشفي الصحي

القسم 11 أذكر مصدر قوتك الأساسي أثناء فترة معانتك من مرض السرطان؟ بند واحد

ملاحظات إضافية :



نشكر لك مشاركتك في هذا الاستبيان، إجاباتك وملاحظاتك سوف تفيدنا في خطتنا للإحتياجات المستقبلية الخاصة بالناجيين من السرطان:

للمزيد من الأسئلة والإستفسارات الخاصة بهذا الاستبيان يمكنكم الاتصال بالدكتورة ماريا عامر:

maamir@seha.ae 037072686



APPENDIX 6: DATA COLLECTION FORM CANCER REGISTRY

DEMOGRAPHICS	
AGE	
MARITAL STATUS	
FORMAL EDUCATION	
OCCUPATION	
NATIONALITY	
EMIRATE AT DIAGNOSIS	
EMIRATE CURRENT	
INSURANCE	
MOBILE/TELEPHONE	
CANCER INFORMATION	
DIAGNOSIS	
HISTOLOGY	
STAGE	
LATERALITY	
GRADE	
DIAGNOSTIC CONFIRMATION	



DATE OF INITIAL DIAGNOSIS	
DATE 1 st CONTACT WITH THE FACILITY	
DATE OF REFERRAL	
REFERRED HOSPITAL	
REFERRED PHYSICIAN	
MANAGING PHYSICIAN	
FOLLOWING PHYSICIAN	
TREATMENT INFORMATION	
TREATMENT STATUS	
SURGERY (Type & Date)	
CHEMOTHERAPY (Type & Date)	
RADIOTHERAPY (Type & Date)	
HORMONAL THERAPY (Type & Date)	
IMMUNOTHERAPY (Type & Date)	
PALLIATIVE CARE (Type & Date)	
ANY OTHER TREATMENT	Specify:
Reviewer Information:	·
Reviewer Name	
Tracking Number	



APPENDIX 7: RISK ASSESSMENT FORM

Version 1, dated 03.10.18

<u>ALL projects MUST include a risk assessment.</u> If this summary assessment of the risk proves insignificant, i.e., you answer 'no' to all of the questions, then no further action is necessary. However, if you identify any risks then you must identify the precautions you will put in place to control these.

<u>1. What is the title of the project?</u>

An exploration of psychosocial needs of breast cancer survivors and its association with cultural differences and clinical indicators – A mixed method study in a hospital setting

2. Is the project purely literature based? YES/NO

NO. It is supported by the literature.

Hazards	Risks	If yes, consider what precautions will be taken to minimise risk and discuss with your Supervisor	
Use of ionising or non- ionising radiation	Exposure to radiation	Obtain copy of existing risk assessment from place of research and attach a copy to this risk assessment summary.	
	NO		
Use of hazardous substances	Exposure to harmful substances	Obtain copy of existing risk assessment from place of research and attach a copy to this risk assessment summary.	
	NO		
Use of face-to-face interviews	Interviewing	<i>NB:</i> Greater precautions are required for medium & high- risk activities	
Interviewees could be upset by interview and become aggressive or		Consider:	

3. Please highlight the risk(s) which applies to your study:



violent toward Own researcher classmates=Low risk NO Other University students=Medium risk NO Use of face-to-face interviews Participants or interviewes could become upset by interview and suffer psychological effects Participants or interview and suffer paychological effects No Version of the planned interviews Participants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital. Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any and wisk to complain formally you can do this by contacting the Research Supervisor d, contact: 5078 or e f (ef@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Suson MeAndrew. Chiro the the the Research Ethical			
researcher classmates=Low risk NO Other University students=Medium risk NO Other University personnel=High risk NO Use of face-to-face interviews Non-University personnel=High risk NO Participants or A separate application will be made for face-to-face interviews. Participants or Participants will be contacted through the hospital source and an official number will be given for communication. The interview and suffer psychological effects Participants will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a healt professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any and wish to complain formally you can do this by contacting the Research Supervisor d, contact: 1234) two will do their bes to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor	violent toward	Own	
risk NOOther University students=Medium risk NOUse of face-to-face interviewsNon-University personnel=High risk NOUse of face-to-face interviewsYESConsider: A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interviewees could become upset by prychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews of the service paychological effectsInterviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a spectred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any and wisk to complain formally you can do this by contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wisk to complain formally you can do this by contacting the Research Supervisor d, contact: 5678 or e f (ef@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susen Meantow: Choir of the Health Procestor busices. Susen Meantow: Choir of the Health Procestored, participant scan forward their concerns to Professor Susen Meantow: Choir of the Health Procestored, participant scan forward their concerns to Professor Susen Meantow: Choir of the Health Procestored, particip	researcher	classmates=Low	
OtherUniversity students=Medium risk NONon-University personnel-High risk NoNon-University personnel-High risk NoUse of face-to-face interviewsYESConsider: A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interviewe and suffer psychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital. Interview and suffer psychological effectsInterview and suffer psychological effectsInterviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participants as a concern about any a spect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants result unappy and wish to complain formally you can do this by orntacting the Research Supervisor d, contact: 5678 or e f (ef@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professon Susen Meantrew. Chir of the Healtb Research Ethicid		risk NO	
Other University students=Medium risk NO Non-University personnel=High risk NO Use of face-to-face interviews YES Consider: A separate application will be made for face-to-face interviews YES Consider: Participants or interviewes could become upset by psychological effects Participants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital. Interviewes could become upset by interview and suffer psychological effects Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant heads to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor d, contact: 5678 or ef (c.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concern between Support Nethele the Path Pesserb Ethical			
Other University students=Medium risk NO Non-University personnel=High risk No Use of face-to-face interviews YES Consider: Participants or interviewees could become upset by interview and suffer psychological effects YES Consider: A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews. Participants or interviewees could perchological effects Participants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital. Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor cd, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concern border betried sond contacting the Research Supervisor cd, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward the Health Percent E			
OtherUniversity students=Medium risk NONon-University personnel=High riskNon-University personnel=High riskUse of face-to-face interviewsYESConsider: A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interviewe and suffer psychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviewe and suffer psychological effectsInterviews will be contacted through the hospital source and an official number will be given for communication. The interviews will be contacted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any ang spect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor cd, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan MeAndrew Chair of the Health Research Ethical			
students=Medium risk NOstudents=Medium risk NONon-University personnel=High risk NoNon-University personnel=High risk NoUse of face-to-face interviewsYESConsider: A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interviewe and suffer psychological effectsYESConsider: A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interviewe and suffer psychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews sould be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor cd, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan MeAndrew Chair of the Health Breserch Ethical		Other University	
risk NONon-University personnel-High risk NoUse of face-to-face interviewsYESConsider:Participants or interviewes could become upset by interviews and suffer psychological effectsYESConsider:Participants or interviews, but have completed this at this time in regard to the planned interviews.Participants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Sucan McAndreev Chair of the Health Research Ethical		students=Medium	
Non-University personnel=High risk No Use of face-to-face interviews Participants or interviewes could become upset by interview and suffer psychological effects Participants or Interview and suffer psychological effects Participants will be contacted through the hospital source and an official number will be given for communication. The interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or eff (eff@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Suce McAndreev Chair of the Health Preserve Fibrical Suce Suce Suce Suce Suce Suce Suce Suce		risk NO	
Non-University personnel=High riskNon-University personnel=High riskConsider:Use of face-to-face interviewsYESConsider:Participants or interviewesA separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interviewesParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.InterviewesInterviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Sucan McAndrew. Chair of the Health Breeserch Ethical			
Non-University personnel=High riskConsider:Use of face-to-face interviewsYESConsider:A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interview and suffer psychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contactingthe Research Supervisor cd, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor ueson the Andrew. Chair of the Health Research Ethical Susen McAndrew. Chair of the Health			
Non-University personnel=High riskNonUse of face-to-face interviewsYESConsider:A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interviewesParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.InterviewesInterviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor cd, contact: 5678 or e f (c.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Suean McAndewy. Chair of the Health Research Ethical Suean McAndewy. Chair of the Health Research Ethical			
personnel=High riskVESConsider:Use of face-to-face interviewsYESConsider:Participants or interviewesA separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interview and suffer psychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you cand ot his by contacting the Research Supervisor cd, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor ues of the lealth Professor Ender Suean Meahdew. Chair of the Health Besearch Ethical		Non-University	
ViskNoUse of face-to-face interviewsYESConsider: A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interviewescould become upset by interview and suffer psychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Suean McAndrew. Chair of the Health Besearch Ethical Suean McAndrew. Chair of the Health Besearch Ethical		personnel=High	
Use of face-to-face interviews YES Consider: A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews. Participants or Participants will be contacted through the hospital source and an official number will be given for communication. psychological effects Participants will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (c.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor		risk No	
Use of face-to-face interviewsYESConsider: A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants or interviewe and suffer psychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviewe scould uestion as a control of the transmission of the transmission of the transmission of the transmission of the transmission of the second of the transmission of the transmission of the second of the transmission of the second of the transmission		110	
interviewsA separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants orinterviewesinterviewescould become upset bypsychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor	Use of face-to-face	YES	Consider:
A separate application will be made for face-to-face interviews, but I have completed this at this time in regard to the planned interviews.Participants orintervieweescould become upset bypsychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor	interviews		
Participants orinterviews, but I have completed this at this time in regard to the planned interviews.Participants orinterviewesinterviewe could become upset by interview and suffer psychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor			A separate application will be made for face-to-face
Participants orto the planned interviews.Intervieweescouldbecomeupsetbyparticipants will be contacted through the hospital sourceand an official number will be given for communication.psychological effectsThe interviews will be held in a safe environment at Breastcare centre at Tawam Hospital.Interviews will be conducted very cautiously with thequestions asked but this is a sensitive topic for manyand at occasion, some may find it as an emotional orstressful experience. In those circumstances, if aparticipant needs to be referred to a health professionalfor emotional support, the researcher will ensure thisoccurs as soon as possible. This support will be offeredat no cost to you.Additionally, if any participant has a concern about anyaspect of this study, she should ask to speak to theresearcher (x y, contact: 1234) who will do their best toanswer your questions. If participants remain unhappyand wish to complain formally you can do this bycontacting the Research Supervisor c d, contact: 5678 ore f (e.f@salford.ac.uk). If the matter is still not resolved,participants can forward their concerns to ProfessorSuean McAndrew, Chair of the Health Research Ethical			interviews, but I have completed this at this time in regard
Participants orImage: Could become upset by interview and suffer psychological effectspsychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor cd, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor			to the planned interviews.
intervieweescould by interview and suffer psychological effectsParticipants will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor	Participants or		I
interviewees could become upset by interview and suffer psychological effects The interviews will be contacted through the hospital source and an official number will be given for communication. The interviews will be held in a safe environment at Breast care centre at Tawam Hospital. Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor	-		
becomeupsetby interviewParticipants will be contacted through the hospital source and an official number will be given for communication.psychological effectsThe interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor cd, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical	interviewees could		
interview and suffer psychological effectsand an official number will be given for communication.The interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, 	become upset by		Participants will be contacted through the hospital source
psychological effectsThe interviews will be held in a safe environment at Breast care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Research Ethical	interview and suffer		and an official number will be given for communication.
The interviews will be held in a safe environment at Breast care centre at Tawam Hospital. Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Sucan McAndrew. Chair of the Health Research Ethical	psychological effects		
care centre at Tawam Hospital.Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Besearch Ethical	1, 0, 0,		The interviews will be held in a safe environment at Breast
Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor			care centre at Tawam Hospital.
Interviews will be conducted very cautiously with the questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Research Ethical			
questions asked but this is a sensitive topic for many and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Research Ethical			Interviews will be conducted very cautiously with the
and at occasion, some may find it as an emotional or stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Research Ethical			questions asked but this is a sensitive topic for many
 and at occasion, some may find it as an emotional of stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Research Ethical 			and at occasion some may find it as an emotional or
 stressful experience. In those circumstances, if a participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Research Ethical 			and at occasion, some may find it as an emotional of
participant needs to be referred to a health professional for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Research Ethical			stressiui experience. In those circumstances, ii a
for emotional support, the researcher will ensure this occurs as soon as possible. This support will be offered at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Research Ethical			participant needs to be referred to a health professional
occurs as soon as possible. This support will be offered at no cost to you.Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew. Chair of the Health Research Ethical			for emotional support, the researcher will ensure this
at no cost to you. Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical			occurs as soon as possible. This support will be offered
Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical			at no cost to you
Additionally, if any participant has a concern about any aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical			
aspect of this study, she should ask to speak to the researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical			Additionally, if any participant has a concern about any
researcher (x y, contact: 1234) who will do their best to answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical			aspect of this study, she should ask to speak to the
answer your questions. If participants remain unhappy and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical			researcher (x y contact: 1234) who will do their best to
and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical			answer your questions. If nortiginants remain unbonny
and wish to complain formally you can do this by contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew Chair of the Health Research Ethical			answer your questions. It participants remain unnappy
contacting the Research Supervisor c d, contact: 5678 or e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical			and wish to complain formally you can do this by
e f (e.f@salford.ac.uk). If the matter is still not resolved, participants can forward their concerns to Professor Susan McAndrew, Chair of the Health Research Ethical			contacting the Research Supervisor c d, contact: 5678 or
participants can forward their concerns to Professor Susan McAndrew Chair of the Health Research Ethical			e f (e.f@salford.ac.uk). If the matter is still not resolved,
Susan McAndrew Chair of the Health Research Ethical			participants can forward their concerns to Professor
Susan Mor man or the realth Research rannan			Susan McAndrew, Chair of the Health Research Ethical



		Approval Panel, Room MS1.91, Mary Seacole Building, Frederick Road Campus, University of Salford, Salford, M6 6PU. Tel: 0161 295 2278. E: s.mcandrew@salford.ac.uk'.
Sensitive data	Exposure to data or information which may cause	Consider:
	upset or distress to the researcher YES	The researchers will collect personal data about the participants, which may be sensitive (for instance, date of birth and relevant health information). The researchers may also need to get some of patient's health information from participant's medical records. Any personal or health information will be kept private and confidential. It will be stored securely and accessible to authorised persons to be kept confidential. Participants will be given a number so that their identity will not be apparent. The study records will be kept in a locked filing cabinet in the researcher's office at Tawam Hospital during the study and in a locked archive for at least 10 years from the time the study is closed and will be destroyed at any time thereafter.
Physical activity	Exposure to levels of exertion unsuitable for an individual's level of fitness NO	Consider:
Equipment	Exposure to faulty or unfamiliar equipment. NO	Consider:
Sensitive issues i.e., Gender/Cultural e.g., when observing or dealing with undressed members of the opposite sex	Exposure to vulnerable situations/ sensitive issues that may cause distress to interviewer or interviewee YES	<i>Consider:</i> Translators will be placed for native speakers mainly for Emiratis who need clarification for any questions either with regards to questions in the in the study or any aspect of the study.
Children	NO	



Manual	handling	Exposure to an
activities		activity that could
		result in injury
		NO
		NO

If you have answered 'YES' to any of the hazards in section 3, then please list the proposed precautions below:

Please see notes above.

Signature of student: Maria Aamir

Date: 03/10/2018.....

Signature of Supervisor: Linda Dubrow-Marshall Date: 03/10/2018.....



APPENDIX 8: RISK ASSESSMENT FORM



Tawam Cancer Registry Data Confidentiality Agreement

You have inquired about obtaining access to the Cancer Data. Maintaining confidentiality of the cancer data in our Registry is of paramount importance and is upheld with great respect. For this reason, it is very important that you as the requestor specify your need for requested cancer data and the intent of their usage to avoid ever-present potential of its misuse.

By signature below TAWAM CANCER REGISTRY make such binding commitments, the requestor firm assures compliance with the below conditions.

The data will be treated as strictly confidential.

The data will not be used for any purpose other than that specifically set forth above.

The data will not be made available to any other individual, agency, institution, or firm and controls will be maintained to prevent unauthorized access.

No follow back of any type will be made to any individual, institution, or firm without prior approval.

All results will be restricted to study only and will not contact any individual, institution, or firm.

Confidential data will be destroyed or returned to the TAWAM CANCER REGISTRY after serving the purpose set forth above unless specific IRB is granted for their retention.

The Tawam Cancer Registry will be credited as the source of the data.

Authorship may be required when TCR makes substantial contribution to the data.



A copy of the results of the study will be furnished to the Tawam Cancer Registry within 60 days of the completion of the project.

Principal Investigator: <u>Maria Aamir</u>

Date: __03/10/2018_



APPENDIX 9: SURVIVORSHIP CARE PLAN



MRN:

Patient Name:

Date of Birth:

Tawam Hospital Cancer Treatment Summary and Survivorship Care Plan

- This Survivorship Care Plan is a cancer treatment summary and follow-up plan is provided to you to keep with your health care records and to share with your primary care provider. This summary is a brief record of major aspects of your cancer treatment.
- You can share your copy with any of your doctors or nurses. However, this is not a detailed or comprehensive record of your care.

General Information				
Patient Name:	Patient DOB:			
Patient phone:	Email:			
Health Care Providers (Inc	cluding Names, Institutio	n)		
Primary Care Provider:				
Surgeon:				
Radiation Oncologist:				
Medical Oncologist:				
Other Providers:				
Treatment	t Summary			
Diag	nosis			
Cancer Type/Location/Histology Subtype:		Diagnosis Date (year):		
Stage: II III III Not applicable				
Treat				
	Surgery Date(c) (vear):			
Surgery Li Yes Li No Surgery Date(s) (year):				
Surgical procedure/location/findings:				
Radiation 🗆 Yes 🔍 No Body area treated:	End	Date (year):		
Systemic Therapy (chemotherapy, hormonal therapy, othe	er) 🗆 Yes 🗌 No			
Names of Agents Used End Dates (year)				
Persistent symptoms or side effects at completion of treatment: No Yes (enter type(s)):				
Familial Cancer Risk Assessment				
Genetic/hereditary risk factor(s) or predisposing conditions:				

HIM-CTSSCP-11052016



Genetic counseling: Yes	No Genet Folic	tic testing results: ow-up Care Plan	
Need for ongoing (adjuvant)	treatment for cancer	Yes 🗆 No	
Additional treatment	name Pla	anned duration	Possible Side effects
	I		
	Sched	lule of next visits	
Physician/Clinic N	ame	When/How often	Related Tests
Please continue to see your p your age, including cancer set 1. Anything that represe 2. Anything that represe 3. Anything you are wor	rimary care provider for reening tests. Any sympto ents a brand new sympto ents a persistent sympto rried about that might b	r all general health ca toms should be broug om om e related to the cance	are recommended for a (man) (woman) ght to the attention of your provider: er coming back.
Possible late- and long-term e	effects that someone wi	th this type of cancer	r and treatment may experience:
Cancer survivors may experie please speak with your docto Emotional and mental her Physical Functioning assistance	nce issues with the area rs or nurses to find out l alth	as listed below. If you how you can get help □ Weight cha □ School/Wo	a have any concerns in these or other areas, b with them. anges Stopping smoking ork Financial advice or
 Memory or concentration Other 	loss 🗆 Parenting	Fertility	□ Sexual functioning
A number of lifestyle/behavio developing another cancer. D	ors can affect your ongo Discuss these recommen	ing health, including t dations with your doo	the risk for the cancer coming back or ctor or nurse:
□Tobacco use/cessation		Diet	:
□Alcohol use	cohol use 🗅 🗅 Sun screen use		
Delight management (loss/gain) Delight management (loss/gain)			
Resources you may be intere	sted in:		
Other comments:			
Prepared by: Name			Date:
Signature & S	.tamp	_	
Approved by: Name Signature & S	 itamp		Date:


APPENDIX 10: QUANTITATIVE DATA EXTRACTION FILES

Descriptive Analysis				
			Statistic	Std.
	F			Error
Physical Concerns	Mean	1	6.14	.857
	95% Confidence Interval for	Lower	4.45	
	Mean	Bound		
		Upper	7.83	
	5% Trimmed Mean	Boulld	3.97	
	Median		1.00	
	Variance		150.619	
	Std. Deviation		12.273	
	Minimum		0	
	Maximum		61	
	Range		61	
	Interguartile Range		6	
	Skewness		2.843	.170
	Kurtosis	7.801	.338	
Information Concerns	Mean	27.99	.770	
	95% Confidence Interval for	Lower	26.47	
	Mean	Bound		
		Upper	29.51	
		Bound		
	5% Trimmed Mean	29.13		
	Median	35.00		
	Variance	121.441		
	Std. Deviation	11.020		
	Minimum	0		
	Maximum	35		
	Range	35		
	Interquartile Range		12	
	Skewness		-1.420	.170
	Kurtosis		.593	.338
Social_Financial_Concerns	Mean	1	7.13	.442
	95% Confidence Interval for	Lower	6.26	
	Mean	Bound	8.00	
		Upper Bound	8.00	
	5% Trimmed Mean		6.71	
	Median		6.00	
	Variance		39.974	
	Std. Deviation		6.323	
	Minimum		0	
	Maximum		28	
	Range	28		



	Interquartile Range		12	
	Skewness		.726	.170
	Kurtosis		065	.338
Emotional Concerns	Mean		15.05	.761
	95% Confidence Interval for Mean		13.55	
	Mean	Bound	16.55	
		Upper Bound	16.55	
	5% Trimmed Mean	Dound	14.49	
	Median		13.00	
	Variance		118.733	
	Std. Deviation		10.896	
	Minimum		0	
	Maximum		54	
	Range Interquartile Range		54	
			17	
	Skewness		.672	.170
	Kurtosis		006	.338
Spiritual/Religious	Mean	•	.20	.064
Concerns	95% Confidence Interval for	Lower	.07	
	Mean	Bound		
		Upper	.33	
	50/ Trimmed Mean	Bound	02	
	3% Iffinited Mean		.02	
	Median		.00	
	Variance		.837	
	Std. Deviation		.915	
	Minimum		0	
	Maximum	8		
	Range		8	
	Interquartile Range		0	
	Skewness		5.718	.170
	Kurtosis	36.577	.338	



Tests of Normality								
	Kolmogorov-Smirnov ^a				Shapiro-Wilk			
	Statistic	df	Sig.	Statistic	df	Sig.		
Physical_Concerns	.309	205	.000	.551	205	.000		
Information_Concerns	.313	205	.000	.679	205	.000		
Social_Financial_Concerns	.130	205	.000	.915	205	.000		
Emotional_Concerns	.091	205	.000	.951	205	.000		
Spiritual_Relegious_Concerns	.518	205	.000	.232	205	.000		

a. Lilliefors Significance Correction

Physical Concerns









Information Concerns





Normal Q-Q Plot of Information_Concerns 1 0-Expected Normal -1 o -2" -3 ő 10 30 -10 20 40 Observed Value





Social or Financial Concerns











Emotional Concerns







Detrended Normal Q-Q Plot of Emotional_Concerns





Histogram Mean = .2 Std. Dev. = .915 N = 205

Spiritual Or Religious Concerns









DESCRIPTIVE STATISTICS - SEVERITY OF BREAST CANCER SURVIVORS' PSYCHOSOCIAL CONCERNS

	Ν	Mean	Std. Deviation
РНҮС1	205	.64	1.274
РНҮС2	204	.69	1.286
РНҮС3	205	.38	1.044
РНҮС4	203	.26	.760
РНҮС5	203	.26	.760
РНҮС6	205	.22	.803
РНҮС7	203	.26	.760
РНҮС8	203	.14	.625
РНҮС9	205	.35	.992
РНҮС10	203	.29	.912
РНҮС11	200	.29	.899
РНҮС12	204	.19	.631
РНҮС13	204	.15	.588
РНҮС14	203	.28	.903
РНҮС15	203	.47	1.166
РНҮС16	205	.36	.983
РНҮС17	202	.33	.969
РНҮС18	203	.33	1.073
РНҮС19	205	.09	.404
РНҮС20	205	.21	.786
INF_CONCERN1	205	3.89	1.829



INF_CONCERN2	205	4.35	1.479
INF_CONCERN3	201	4.04	1.691
INF_CONCERN4	202	4.06	1.692
INF_CONCERN5	203	3.97	1.778
INF_CONCERN6	204	3.96	1.776
INF_CONCERN7	204	3.95	1.791
SFC1	205	.85	1.204
SFC2	201	.86	1.168
SFC3	197	.61	1.052
SFC4	202	.94	1.420
SFC5	205	1.96	1.860
SFC6	205	1.96	1.860
EC1	205	1.04	1.380
EC2	204	.78	1.202
EC3	203	.40	.908
EC4	199	1.45	1.839
EC5	205	3.44	1.966
EC6	202	.98	1.250
EC7	203	.41	.915
EC8	205	.26	.718
EC9	200	.61	1.089
EC10	202	1.05	1.543
EC11	201	.29	.804
EC12	201	.46	1.104



EC13	200	.55	1.150
EC14	203	3.48	1.968
SRC1	205	.02	.279
SRC2	205	.02	.279
SRC3	205	.00	.000
SRC4	202	.16	.738
Valid N (listwise)	139		



Breast cancer survivors' most prevalent concerns by year of diagnosis

0–1 year since diagnosis, >=2018 (n =122)

	Ν	Mean	Std. Deviation
Physical_Concerns	122	7.00	13.368
Information_Concerns	122	26.60	11.582
Social_Financial_Concerns	122	6.30	6.160
Emotional_Concerns	122	13.89	10.570
Spiritual_Relegious_Concerns	122	.23	1.019
Valid N (listwise)	122		

2–5 years since diagnosis, 2012-2017 (n = 80)

	Ν	Mean	Std. Deviation
Physical_Concerns	80	4.31	8.710
Information_Concerns	80	29.96	9.995
Social_Financial_Concerns	80	8.27	6.410
Emotional_Concerns	80	17.04	11.302
Spiritual_Relegious_Concerns	80	.16	.754
Valid N (listwise)	80		

6 and earlier since diagnosis (n = 3)

	Ν	Mean	Std. Deviation
Physical_Concerns	3	19.67	33.201



Information_Concerns	3	32.00	5.196
Social_Financial_Concerns	3	10.00	7.211
Emotional_Concerns	3	9.33	4.041
Spiritual_Relegious_Concerns	3	.00	.000
Valid N (listwise)	3		



ASSOCIATION BETWEEN BREAST CANCER SURVIVORS' NATIONALITY & AGE

Case Processing Summary

	Valid		Missing		Total	
	Ν	Percent	Ν	Percent	Ν	Percent
Age group * Nation	205	100.0%	0	0.0%	205	100.0%

Cases

Age group * Nation Crosstabulation

			Na	tion	
			Emirati	Expatriate	Total
Age group	22-32	Count	1	11	12
		Expected Count	3.2	8.8	12.0
		% within Age group	8.3%	91.7%	100.0%
33-43	Count	20	44	64	
	Expected Count	16.9	47.1	64.0	
	44-54 Count Expec	% within Age group	31.3%	68.8%	100.0%
		Count	14	50	64
		Expected Count	16.9	47.1	64.0
55-65		% within Age group	21.9%	78.1%	100.0%
	55-65	Count	15	30	45
		Expected Count	11.9	33.1	45.0
		% within Age group	33.3%	66.7%	100.0%



>65		Count	4	16	20
		Expected Count	5.3	14.7	20.0
		% Within Age group	20.0%	80.0%	100.0%
Total		Count	54	151	205
		Expected Count	54.0	151.0	205.0
		% Within Age group	26.3%	73.7%	100.0%

Chi-Square Tests

	Value	df	Asymptotic Significance (2- sided)
Pearson Chi-Square	5.007ª	4	.287
Likelihood Ratio	5.479	4	.242
Linear-by-Linear Association	.069	1	.792
N of Valid Cases	205		

a. 1 cells (10.0%) have expected count less than 5. The minimum expected count is 3.16.

Symmetric Measures

		Value	Approximate Significance
Nominal by Nominal	Phi	.156	.287
	Cramer's V	.156	.287
N of Valid Cases		205	







ASSOCIATION BETWEEN BREAST CANCER SURVIVORS' STAGE OF CANCER & NATIONALITY

Case Processing Summary

	Cases						
	Valid		Missing		Total		
	Ν	Percent	Ν	Percent	Ν	Percent	
Stage * Nation	205	100.0%	0	0.0%	205	100.0%	

Stage * Nationality Crosstabulation

			Na		
			Emirati	Expatriate	Total
Stage	In-Situ	Count	5	8	13
		Expected Count	3.4	9.6	13.0
		% Within Stage	38.5%	61.5%	100.0%
	Stage I	Count	12	27	39
		Expected Count	10.3	28.7	39.0
		% Within Stage	30.8%	69.2%	100.0%
	Stage II	Count	31	74	105
		Expected Count	27.7	77.3	105.0
		% Within Stage	29.5%	70.5%	100.0%
	Stage III	Count	6	42	48
		Expected Count	12.6	35.4	48.0
		% Within Stage	12.5%	87.5%	100.0%
Total		Count	54	151	205



Expected Count	54.0	151.0	205.0
% Within Stage	26.3%	73.7%	100.0%

Stage * Nationality Crosstabulation

			Na		
			Emirati	Expatriate	Total
Stage	In-Situ	Count	5	8	13
		Expected Count	3.4	9.6	13.0
		% Within Stage	38.5%	61.5%	100.0%
	Stage I	Count	12	27	39
		Expected Count	10.3	28.7	39.0
		% Within Stage	30.8%	69.2%	100.0%
	Stage II	Count	31	74	105
		Expected Count	27.7	77.3	105.0
		% Within Modified Stage	29.5%	70.5%	100.0%
	Stage III	Count	6	42	48
		Expected Count	12.6	35.4	48.0
		% Within Modified Stage	12.5%	87.5%	100.0%
Total		Count	54	151	205
		Expected Count	54.0	151.0	205.0
		% Within Modified Stage	26.3%	73.7%	100.0%

Chi-Square Tests

		Asymptotic Significance (2-
Value	df	sided)



Pearson Chi-Square	6.666ª	3	.083
Likelihood Ratio	7.344	3	.062
Linear-by-Linear Association	4.949	1	.026
N of Valid Cases	205		

a. 1 cells (12.5%) have expected count less than 5. The minimum expected count is 3.42.

Symmetric Measures

		Value	Approximate Significance
Nominal by Nominal	Phi	.180	.083
	Cramer's V	.180	.083
	N of Valid Cases	205	





Group Statistics					
	Nation	Ν	Mean	Std. Deviation	Std. Error Mean
	Emirati	54	3.57	8.878	1.208
Physical_Concerns	Expatriate	151	7.05	13.183	1.073
	Emirati	54	26.41	14.395	1.959
Information_Concerns	Expatriate	151	28.56	9.524	.775
	Emirati	54	3.59	3.839	.522
Social_Financial_Concerns	Expatriate	151	8.39	6.561	.534
	Emirati	54	17.17	10.056	1.368
Emotional_Concerns	Expatriate	151	14.29	11.116	.905
	Emirati	54	.28	.960	.131
Spiritual_Relegious_Concerns	Expatriate	151	.17	.900	.073

T-Test – Psychosocial Concerns

Independent Sam	oles Test									
		Levene's Test for Equality of Variances			t-test for Equality of Means					
		F	Si g.	t	df	Sig. (2- tail ed)	Mean Differe nce	Std. Error Differe nce	95 Confi Interv th Diffe Lo wer	% dence /al of e rence Up per
Physical_Concerns	Equal varian ces assum ed	7.16 0	.0 08	- 1.7 98	203	.07 4	-3.479	1.935	- 7.2 95	.33 7
	Equal varian ces not			2.1 53	139. 000	.03 3	-3.479	1.616	6.6 73	- .28 4



	assum ed									
Information Concern	Equal varian ces assum ed	23.7 72	.0 00	1.2 31	203	.22 0	-2.149	1.745	- 5.5 90	1.2 92
S	Equal varian ces not assum ed			1.0 20	70.2 84	.31 1	-2.149	2.107	6.3 50	2.0 53
Social Financial Co	Equal varian ces assum ed	17.3 79	.0 00	- 5.0 67	203	.00 0	-4.798	.947	- 6.6 65	2.9 31
Social_Financial_Co ncerns	Equal varian ces not assum ed			- 6.4 23	159. 920	.00 0	-4.798	.747	6.2 73	3.3 23
	Equal varian ces assum ed	2.10 2	.1 49	1.6 71	203	.09 6	2.875	1.720	- .51 6	6.2 67
Emotional_Concerns	Equal varian ces not assum ed			1.7 53	102. 519	.08 3	2.875	1.640	.37 8	6.1 29
Spiritual_Relegious_ Concerns	Equal varian ces assum ed	1.86 4	.1 74	.72 7	203	.46 8	.106	.145	- .18 1	.39 2



	Equal varian ces not assum ed			.70 5	88.4 80	.48 3	.106	.150	.19 2	.40 3
--	-------------------------------	--	--	----------	------------	----------	------	------	----------	----------

Group Statistics – Physical Concerns												
	Nation	Ν	Mean	Std. Deviation	Std. Error Mean							
	Emirati	54	.33	.911	.124							
PHYC1	Expatriate	151	.75	1.366	.111							
	Emirati	54	.41	.981	.134							
PHYC2	Expatriate	150	.79	1.367	.112							
DUNCO	Emirati	54	.07	.428	.058							
PHYC3	Expatriate	151	.49	1.171	.095							
DINCA	Emirati	54	.09	.351	.048							
PHYC4	Expatriate	149	.32	.855	.070							
DUNCE	Emirati	54	.09	.351	.048							
PHYCS	Expatriate	149	.32	.855	.070							
DUNCK	Emirati	54	.19	.803	.109							
РНҮС6	Expatriate	151	.24	.806	.066							
DUNCZ	Emirati	54	.09	.351	.048							
PHYC/	Expatriate	149	.32	.855	.070							
DUVC9	Emirati	54	.02	.136	.019							
PHYC8	Expatriate	149	.19	.720	.059							
DUVCO	Emirati	54	.26	.915	.125							
PHIC9	Expatriate	151	.38	1.019	.083							
DUVC10	Emirati	54	.17	.795	.108							
PHYCIU	Expatriate	149	.34	.949	.078							
DUVC11	Emirati	53	.19	.810	.111							
PHICII	Expatriate	147	.32	.929	.077							
DUVC12	Emirati	54	.04	.191	.026							
rnici2	Expatriate	150	.24	.721	.059							
DUVC12	Emirati	54	.02	.136	.019							
PHYC13	Expatriate	150	.20	.676	.055							



	Emirati	53	.15	.533	.073
PHYC14	Expatriate	150	.32	.999	.082
DUNCIE	Emirati	53	.49	1.325	.182
PHYCIS	Expatriate	150	.46	1.109	.091
DUVC16	Emirati	54	.37	1.104	.150
FHICIO	Expatriate	151	.35	.939	.076
DHVC17	Emirati	54	.17	.505	.069
IIIICI/	Expatriate	148	.39	1.086	.089
DUVC19	Emirati	54	.24	.950	.129
PHICIO	Expatriate	149	.36	1.115	.091
DUVC10	Emirati	54	.06	.408	.056
PHICI9	Expatriate	151	.11	.403	.033
DUVCOO	Emirati	54	.15	.627	.085
PHIC20	Expatriate	151	.23	.836	.068

Independent Samples Test – Physical Concerns

	Levene's Test for Equality of Variances				t-test for Equality of Means							
		F	Sig.	t	df	Sig. (2- tailed)	Mean Difference	Std. Error Difference	95% Co Interva Diffe Lower	nfidence I of the prence Upper		
PHYC1	Equal variances assumed	14.017	.000	2.105	203	.037	422	.200	817	027		
	Equal variances not assumed			- 2.532	140.405	.012	422	.167	751	092		
PHYC2	Equal variances assumed	8.278	.004	- 1.904	202	.058	386	.203	786	.014		
	Equal variances not assumed			- 2.217	130.338	.028	386	.174	730	042		



РНҮС3	Equal variances assumed	27.524	.000	- 2.547	203	.012	416	.163	738	094
	Equal variances not assumed			3.725	202.879	.000	416	.112	636	196
PHYC4	Equal variances assumed	14.765	.000	- 1.857	201	.065	223	.120	459	.014
	Equal variances not assumed			- 2.628	198.024	.009	223	.085	390	056
РНҮС5	Equal variances assumed	14.765	.000	- 1.857	201	.065	223	.120	459	.014
	Equal variances not assumed			- 2.628	198.024	.009	223	.085	390	056
PHYC6	Equal variances assumed	.574	.450	417	203	.677	053	.128	305	.198
	Equal variances not assumed			418	93.791	.677	053	.127	306	.200
РНҮС7	Equal variances assumed	14.765	.000	- 1.857	201	.065	223	.120	459	.014
	Equal variances not assumed			- 2.628	198.024	.009	223	.085	390	056
РНҮС8	Equal variances assumed	12.466	.001	- 1.715	201	.088	169	.099	364	.025
	Equal variances not assumed			- 2.740	173.892	.007	169	.062	291	047
РНҮС9	Equal variances assumed	2.124	.147	793	203	.429	125	.157	435	.186
	Equal variances not assumed			835	103.255	.406	125	.150	422	.172
PHYC10	Equal variances assumed	4.880	.028	- 1.168	201	.244	169	.145	454	.116



	Equal variances not assumed			- 1.268	111.218	.207	169	.133	433	.095
PHYC11	Equal variances assumed	2.939	.088	910	198	.364	131	.144	415	.153
	Equal variances not assumed			970	104.630	.334	131	.135	399	.137
PHYC12	Equal variances assumed	17.960	.000	- 2.041	202	.043	203	.099	399	007
	Equal variances not assumed			- 3.157	192.150	.002	203	.064	330	076
PHYC13	Equal variances assumed	16.240	.000	- 1.957	202	.052	181	.093	364	.001
	Equal variances not assumed			- 3.119	178.120	.002	181	.058	296	067
PHYC14	Equal variances assumed	5.699	.018	- 1.173	201	.242	169	.144	453	.115
	Equal variances not assumed			- 1.542	169.747	.125	169	.110	385	.047
PHYC15	Equal variances assumed	.614	.434	.164	201	.870	.031	.187	338	.399
	Equal variances not assumed			.150	79.250	.881	.031	.203	374	.435
PHYC16	Equal variances assumed	.195	.659	.124	203	.901	.019	.156	289	.327
	Equal variances not assumed			.115	82.064	.909	.019	.169	316	.355
PHYC17	Equal variances assumed	9.031	.003	- 1.466	200	.144	225	.154	528	.078
	Equal variances not assumed			2.000	188.903	.047	225	.113	447	003



PHYC18	Equal variances assumed	1.577	.211	674	201	.501	115	.171	451	.222
	Equal variances not assumed			726	109.369	.469	115	.158	429	.199
PHYC19	Equal variances assumed	2.315	.130	787	203	.432	050	.064	177	.076
	Equal variances not assumed			782	92.316	.436	050	.064	178	.078
PHYC20	Equal variances assumed	1.688	.195	670	203	.503	084	.125	330	.162
	Equal variances not assumed			767	124.190	.445	084	.109	300	.132

Group Statistics – Social & Financial Concerns

	Nation	Ν	Mean	Std. Deviation	Std. Error Mean
SFC1	Emirati	54	.63	.917	.125
	Expatriate	151	.93	1.284	.104
SFC2	Emirati	54	.69	.886	.121
	Expatriate	147	.92	1.252	.103
SFC3	Emirati	50	.34	.519	.073
	Expatriate	147	.70	1.167	.096
SFC4	Emirati	53	.64	1.178	.162
	Expatriate	149	1.05	1.486	.122
SFC5	Emirati	54	.67	1.046	.142
	Expatriate	151	2.42	1.871	.152
SFC6	Emirati	54	.67	1.046	.142
	Expatriate	151	2.42	1.871	.152



Independent Samples Test – Social & Financial Concerns

		Levene's	Test for	st for						
		Equal	ity of							
		Varia	inces			t-tes	st for Equality	y of Means		
		F	Sig	t	df	Sig. (2- tailed)	Mean	Std. Error	95% Co Interva Diffe	nfidence l of the rence
		-	515.				Difference	Difference	Lower	opper
SFC1	Equal variances assumed	6.631	.011	- 1.600	203	.111	304	.190	679	.071
	Equal variances not assumed			- 1.868	130.649	.064	304	.163	626	.018
SFC2	Equal variances assumed	10.268	.002	- 1.256	199	.210	233	.186	599	.133
	Equal variances not assumed			- 1.468	133.223	.144	233	.159	547	.081
SFC3	Equal variances assumed	16.425	.000	2.113	195	.036	361	.171	697	024
	Equal variances not assumed			- 2.979	181.820	.003	361	.121	600	122
SFC4	Equal variances assumed	7.888	.005	- 1.795	200	.074	405	.226	851	.040
	Equal variances not assumed			- 2.002	114.574	.048	405	.203	807	004
SFC5	Equal variances assumed	44.090	.000	- 6.540	203	.000	-1.757	.269	-2.287	-1.227
	Equal variances not assumed			8.431	166.559	.000	-1.757	.208	-2.169	-1.346
SFC6	Equal variances assumed	44.090	.000	6.540	203	.000	-1.757	.269	-2.287	-1.227



Equal variances	-	166.559	.000	-1.757	.208	-2.169	-1.346
not assumed	8.431						



RELATIONSHIP BETWEEN (a) DEMOGRAPHIC FACTORS & PSYCHOSOCIAL CONCERNS

Relationship		Correlation (R-value)			Significance level (P-				
Age and physica	lconcern	-0.11	1		0 104				
Stage of ca	ncer and	-0.11	11		0.104				
physical concern		0.0-	F1		0.502				
Receiving treat	tment and	0.11	7		0.095				
physical concern	1								
Relationship		Cori	relation (R-value))	Significance level (P- value)				
Age and socia	l financial	-0.06	50		0.394				
concern									
Stage of cancer	and social	0.15	7		0.025*				
financial concern	1								
Receiving treat	tment and	0.24	б		0.000*				
social financial c	concern		Γ	1		T			
Model	Predictor		Standardised	T-va	lue	P(ignificanc	e		
	variable		regression			level)			
N. 1.11	D · ·		coefficient beta	2.62		0.000			
Model1	Receiving		0.246	3.62	2	0.000			
M. 1.10	treatment		0.220	2 5 1 5		0.000			
Model 2	Receiving		0.239	3.54	5	0.000			
	Stage of or		oor 0.145		1	0.022			
Delationshin	Stage of Ca	Com	0.143	2.13	Significan		(D		
Kelationship		Correlation (R-value)			value)	ice level	(r-		
Age and emotion	nal concern	0.13	3		0.058				
Stage of ca	ncer and	0.03	7		0.602				
emotional conce	rn								
Receiving treat	tment and	0.24	0		0.001*				
emotional conce	rn		~	I					
Model	Predictor		Standardised	T-va	lue	P(significan	ice		
	variable		regression			level)			
N 11	D · ·		coefficient beta	2.50	4	0.001			
Model	Receiving		0.240	3.52	4	0.001			
	treatment								
Relationship		Cori	relation (R-value))	Significan	ce level	(P -		
Kelationship			(· · · · · · · · · · · · · · · · · · ·		value)				
Age and spiritua	l concern	0.45			0.523				



Stage of canc	er and	-0.069	0.325
spiritual concern			
Receiving treatm	ent and	-0.024	0.734
spiritual concern			



ORRELATION BETWEEN QUALITY OF LIFE & PSYCHOSOCIAL CONCERNS OF BREAST CANCER SURVIVORS

		Overall	Physical_C	Information_	Social_Financia	Emotional_	Spiritual_Relegio
		_QOL	oncerns	Concerns	l_Concerns	Concerns	us_Concerns
Overall_QOL	Pears on Correl ation	1	483**	.221**	282**	360**	166*
	Sig. (2- tailed)		.000	.001	.000	.000	.017
	Ν	205	205	205	205	205	205
Physical_Concem s	Pears on Correl ation	483**	1	308**	.306**	.373**	.324**
	Sig. (2- tailed)	.000		.000	.000	.000	.000
	N	205	205	205	205	205	205
Information_Con cerns	Pears on Correl ation	.221**	308**	1	.107	.163*	192**
	Sig. (2- tailed)	.001	.000		.127	.019	.006
	N	205	205	205	205	205	205
Social_Financial_ Concerns	Pears on Correl ation	282**	.306**	.107	1	.524**	.104



	Sig. (2- tailed)	.000	.000	.127		.000	.138
	N	205	205	205	205	205	205
Emotional_Conce rns	Pears on Correl ation	360**	.373**	.163*	.524**	1	.246**
	Sig. (2- tailed)	.000	.000	.019	.000		.000
	N	205	205	205	205	205	205
Spiritual_Relegio us_Concerns	Pears on Correl ation	166*	.324**	192**	.104	.246**	1
	Sig. (2- tailed)	.017	.000	.006	.138	.000	
	N	205	205	205	205	205	205

**. Correlation is significant at the 0.01 level (2-tailed).

*. Correlation is significant at the 0.05 level (2-tailed).


MULTIPLE REGRESSION

Descriptive Statistics

	Mean	Std. Deviation	Ν
Overall_QOL	9.40	1.262	205
Physical_Concerns	6.14	12.273	205
Information_Concerns	27.99	11.020	205
Social_Financial_Concerns	7.13	6.323	205
Emotional_Concerns	15.05	10.896	205
Spiritual_Relegious_Concerns	.20	.915	205

Variables Entered/Removed^a

Model	Variables Entered	Variables Removed	Method		
1	Physical_Concerns		Stepwise (Criteria: Probability- of-F-to-enter <= .050, Probability-of-F-to-remove >= .100).		
2	Emotional_Concerns		Stepwise (Criteria: Probability- of-F-to-enter <= .050, Probability-of-F-to-remove >= .100).		
3	Information_Concerns		Stepwise (Criteria: Probability- of-F-to-enter <= .050, Probability-of-F-to-remove >= .100).		

a. Dependent Variable: Overall_QOL

Model Summary

Model R R Square Adjusted R Square Std. Error of the Es	stimate
---	---------



1	.483ª	.234	.230	1.108
2	.521 ^b	.271	.264	1.083
3	.540°	.292	.281	1.070

a. Predictors: (Constant), Physical_Concerns

b. Predictors: (Constant), Physical_Concerns, Emotional_Concerns

c. Predictors: (Constant), Physical_Concerns, Emotional_Concerns, Information_Concerns

Coefficients^a

		Unstandardized C	Coefficients	Standardized Coefficients		
Model		В	Std. Error	Beta	t	Sig.
1	(Constant)	9.700	.087		112.084	.000
	Physical_Concerns	050	.006	483	-7.864	.000
2	(Constant)	10.015	.129		77.538	.000
	Physical_Concerns	042	.007	405	-6.259	.000
	Emotional_Concerns	024	.007	209	-3.231	.001
3	(Constant)	9.551	.230		41.495	.000
	Physical_Concerns	035	.007	337	-4.817	.000
	Emotional_Concerns	030	.008	261	-3.868	.000
	Information_Concerns	.018	.008	.160	2.427	.016

a. Dependent Variable: Overall_QOL

Excluded Variables^a

					Collinearity
Model	Beta In	t	Sig.	Partial Correlation	Statistics



_						Tolerance
1	Information_Concerns	.079 ^b	1.231	.220	.086	.905
	Social_Financial_Concerns	147 ^b	-2.308	.022	160	.906
	Emotional_Concerns	209 ^b	-3.231	.001	222	.861
	Spiritual_Relegious_Concerns	011 ^b	168	.867	012	.895
2	Information_Concerns	.160°	2.427	.016	.169	.815
	Social_Financial_Concerns	067°	946	.345	067	.712
	Spiritual_Relegious_Concerns	.019°	.290	.772	.020	.877
3	Social_Financial_Concerns	084 ^d	-1.184	.238	083	.706
	Spiritual_Relegious_Concerns	.044 ^d	.683	.495	.048	.855

a. Dependent Variable: Overall_QOL

b. Predictors in the Model: (Constant), Physical_Concerns

c. Predictors in the Model: (Constant), Physical_Concerns, Emotional_Concerns

d. Predictors in the Model: (Constant), Physical_Concerns, Emotional_Concerns, Information_Concerns



RELATIONSHIP BETWEEN PHYSICAL CONCERNS AND

Relationship	Correlation (R-value)	Significance level (P-value)
Age and physical concerns	-0.114	0.104
Stage of cancer and physical	-0.041	0.562
concerns		
Receiving treatment and	0.117	0.095
physical concerns		



ELATIONSHIP BETWEEN INFORMATION CONCERNS AND

Relationship			Correl	Correlation (R-value)				Significance level (P-value)			
Age and information concerns			0.182	0.182				0.009			
Stage of cancer and information 0. concerns							0.52	28			
Receivin information	ng treatment tion concern	and s	0.179				0.01	10			
Variables Entered/Removed ^a											
Model	· · ·	Variables E	ntered	Va	ariables	Remove	ed		Method		
1	Age							Stepwise	(Criteria:		
								Probabil	ity-of-F-to-	enter <=	
								.050, Pro	bability-of-	F-to-	
								remove	>=.100).		
2	Curr	ent receiving	treatment					Stepwise	(Criteria:		
								Probabil	ity-of-F-to-	enter <=	
								.050, Pro	bability-of-	F-to-	
								remove	>=.100).		
a. Depend	dent Variable:	Information_	Concerns		1		1 Q				
Model S	Summary						nange Statis	tics			
	R	R Square	Adjusted R Square	Std. Error of the Estimate	R Squ Change	are F	Change	df1	df2	Sig. F Change	
1	.182 ^a	.033	.028	10.863	.033	6.9	956	1	203	.009	
2	.252 ^b	.063	.054	10.718	.030	6.	549	1	202	.011	
a. Predict	ors: (Constar	t), Age				I					
b. Predict	ors: (Constar	nt), Age, Curi	rent receivi	ng treatmen	nt						
ANOVA	A a										
Model			S	um of Squar	res	df	Me	ean Square	F	Sig.	
1	Regression			820).813	1	1	820.813	6.956	.009 ^b	
	Residual			23953	3.168	203	3	117.996	;		
	Total			24773	3.980	204	4				
2	Regression			1569	9.709	2	2	784.854	6.832	.001c	
	Residual			23204	4.272	202	2	114.873			
	Total			24773	3.980	204	4				
a. Depen	dent Variable:	Information_	Concerns								



b. Predictors: (Constant), Age												
c. Predictors: (Constant), Age, Current receiving treatment												
Coefficients ^a												
Model		Unstandard	lized Coefficients	Standardized Coefficients	t	Sig.						
		В	Std. Error	Beta								
1	(Constant)	19.732	3.222		6.125	.000						
	Age	.169	.064	.182	2.637	.009						
2	(Constant)	13.767	3.945		3.490	.001						
	Age	.165	.063	.177	2.602	.010						
	Current	3.888	1.523	.174	2.553	.011						
	receiving											
	treatment											
a. Dep	endent Variable: In	formation_Con	ncerns									



RELATIONSHIP BETWEEN SOCIAL & FINANCIAL CONCERNS AND

Relations	ship		Correla	tion (R- va	lue)	Sig	nificance	level (P-v	value)	
Age and concerns	Social finan	cial	-0.060			0.3	94			
Stage of financial	cancer and S concerns	ocial	0.157			0.0	25			
Receivin financial	g treatment concerns	and Social	0.246			<0.	001			
Variables Entered/Removed ^a										
Model		Variables E	ntered	V	ariables Ren	noved		Method	1	
1	Curre	ent receiving	treatment				Stepwise	e (Criteria:	:	
							Probabil	ity-of-F-te	o-enter <=	
							.050, Pro	obability-o	of-F-to-	
							remove	>=.100).		
2	Modi	fied Stage					Stepwise	e (Criteria:		
							Probabil	ity-of-F-te	o-enter <=	
							.050, Pro	obability-o	of-F-to-	
remove >= .100).										
a. Depe	a. Dependent Variable: Social_Financial_Concerns									
Model S	Summary					Change Stati	sucs			
	R	R Square	Adjusted R	Std. Error	R Square	F Change	df1	df2	Sig. F	
			Square	of the Estimate	Change				Change	
1	.246 ^a	.061	.056	6.143	.061	13.118	1	203	.000	
2	.286 b	.082	.073	6.089	.021	4.625	1	202	.033	
a. Predicto	ors: (Constan	t), Current r	eceiving tre	atment					·	
b. Predicte	ors: (Constan	t), Current 1	eceiving tre	eatment, Sta	ige					
ANOVA	a									
Model		Sum	of Squares	df	N	Aean Square]	F	Sig.	
1	Regression		494.960)	1	494.96	50 1	3.118	.000 ^b	
	Residual		7659.742	2	203	37.73	33			
	Total		8154.702	2	204					
2	Regression		666.418	3	2	333.20)9	8.988	.000°	
_	Residual		7488.285	5	202	37.07	/1			
	Total		8154.702	2	204					
a. Depend	lent Variable:	Social_Finar	icial_Concer	ns						
b. Predict	ors: (Constant)), Current rec	eiving treatm	nent						
c. Predicte	ors: (Constant)), Current rec	eiving treatn	nent, Modifi	ed Stage					
Coeffici	ents ^a									



Model		Unstandardized Coefficients		Standardized	t	Sig.
				Coefficients		
		В	Std. Error	Beta		
1	(Constant)	5.262	.670		7.851	.000
	Current receiving treatment	3.160	.872	.246	3.622	.000
2	(Constant)	2.055	1.632		1.259	.209
	Current receiving treatment	3.069	.866	.239	3.545	.000
	Modified Stage	1.118	.520	.145	2.151	.033
a. D	ependent Variable: Social_Finan	cial_Concerns				



RELATIONSHIP BETWEEN EMOTIONAL CONCERNS AND

Relationship		Correlati	on (R-val	ue)		Signifi	cance l	evel (P-valu	e)
Age and emotional concerns 0.		0.133				0.058			
Stage of cancer and concerns	emotional	0.037				0.602			
Receiving treatment emotional concerns	and	0.240				0.001			
		Varial	bles Enter	ed/Re	move	eda			
Model	Variables Fr	tono d	Ve		Dom	oved		Mathad	
			Vč	anabies	Kein	oveu	6 4		
I Curi	ent receiving	treatment				•	Stepv	vise (Criteria:	
							Proba	ibility-of-F-to	-enter <=
							.050,	Probability-o	f-F-to-
							remo	ve >= .100).	
a. Dependent Variable:	Emotional_C	oncerns				Change Statis	tics		
Model Summary						Change Statis	lies		
R ^a	R Square	Adjusted R	Std. Error	R Sq	uare	F Change	df1	df2	Sig. F
		Square	of the Estimate	Chang	e				Change
Model .240	.058	.053	10.604	0.58		12.417	1	203	.001
a. Predictors: (Constant), Current receiving treatment									
ANOVA ^a									
Model	Sum	of Squares	df		М	lean Square		F	Sig.
1 Regression		1396.120		1		1396.12	0	12.417	.001 ^b
Residual		22825.392		203		112.44	0		
Total		24221.512		204					
a. Dependent Variable:	Emotional_C	oncerns							
b. Predictors: (Constan	t), Current rec	eiving treatm	ent						
Coefficientsa									
Model		Unstan	dardized C	oefficie	ents	Standar	dized	t	Sig
1110 dol		Olistan		oomen	1100	Coeffic	rients	·	515.
		D		Std Em	or	Rot	9		
1 (Constant)		11	.917	<u>1</u>	157	Det	a	10.300	.000
Current recei	ving treatment	5	5.306	1.	506		.240	3.524	.001
· · · · ·									



RELATIONSHIP BETWEEN RELIGIOUS OR SPIRITUAL CONCERNS

AND DEMOGRAPHIC FACTORS

Relationship	Correlation (R-value)	Significance level (P-value)
Age and religious or spiritual	0.045	0.523
religious concerns		
Stage of cancer and religious or	-0.069	0.325
spiritual religious concerns		
Receiving treatment and	-0.024	0.734
religious or spiritual concerns		



RELATIONSHIP BETWEEN PSYCHOSOCIAL CONCERNS AND OVERALL QUALITY OF LIFE

Relation	ship	Cor	relation (R-va	lue)	Sign	ificance level	(P-value)
Quality of concern	of life & physical		-0.481			< 0.0001	
Quality of concern	of life & information		0.208			0.003	
Quality of concern	of life & social financial		-0.271			< 0.0001	
Quality of concern	of life & emotional		-0.366			< 0.0001	
Quality of concern	of life & spiritual		-0.166			< 0.0001	
Variable	es Entered/Removedª						
Model	Variables Entered	Variab	les Removed			Method	
1	Physical_Concerns				Stepwise (Crit	eria: Probabili	ty-of-F-to-
	-				enter <= .050,	Probability-of	-F-to-
					remove >= .10	00).	
2	Emotional_Concerns				Stepwise (Crit	eria: Probabili	ty-of-F-to-
					enter <= .050,	Probability-of	-F-to-
				_	remove >= .10)0).	
3	Information_Concerns				Stepwise (Crit	eria: Probabili	ty-of-F-to-
					enter <= .050,	Probability-of	F-F-to-
				_	remove >= .10	00).	
a. Deper	ident Variable: Overall Q	OL					
Model S	ummary						
Model	R	R Square	e Ad	juste	d R Square	Std. Error of	the Estimate
1	.483ª		.234	_	.230		1.108
2	.521 ^b		.271		.264		1.083
3	.540°		.292		.281		1.070
a. Predict	ors: (Constant), Physical_C	oncerns					
b. Predict	ors: (Constant), Physical_C	oncerns, Emoti	onal_Concerns				
c. Predict	ors: (Constant), Physical_C	oncerns, Emoti	onal_Concerns,]	Infor	mation_Concerns	5	
ANOVA	a						
Model	Sum	of Squares	df	Me	an Square	F	Sig.
1	Regression	75.894	1		75.894	61.848	.000 ^b
-	Residual	249.101	203		1.227		
	Total	324.995	204				
2	Regression	88.135	2		44.068	37.582	.000°
	Residual	236.860	202		1.173		
	Total	324.995	204				
3	Regression	94.880	3		31.627	27.625	.000 ^d



	Residual	230.115	201	1.145		
	Total	324.995	204			
a. Deper	ndent Variable: Overall QOI					
b. Predic	tors: (Constant). Physical	Concerns				
c Predic	tors: (Constant) Physical	Concerns Emotional Co	oncerns			
d Predic	etors: (Constant), Physical	Concerns Emotional C	oncerns Information	Concerns		
Coeffi	cients ^a	eoneems, Emotional_e	oncerns, monnation			
Model		Unstandard	ized Coefficients	Standardized	t f	Sig
Widdel		Clistundulu		Coefficients		515.
		B	Std Error	Beta		
1	(Constant)	9 700	087		112.084	000
1	Physical Concerns	- 050	006	5 - 48	3 -7 864	000
2	(Constant)	10.015	120)	77 538	000
2	Physical Concerns	042	.122	7 40	5 6 2 5 0	.000
	Emotional Concerns	042	.007	+0.	-0.239	.000
2		0.551		20	41.405	.001
3	(Constant)	9.551	.230		41.495	.000
	Physical_Concerns	035	.00.	33	/ -4.81/	.000
	Emotional_Concer	ns030	.008	26	1 -3.868	.000
	Information_Conce	erns .018	.008	.16	0 2.427	.016
a. Dep	a. Dependent Variable: Overall QOL					



APPENDIX 11a: QUALITATIVE INTERVIEWS – COMPILED TRANSCRIPTS			
1- PARTICIPANT - EMIRATI			
Interview	Exploratory comments		
I: Can you tell me what has happened since the moment you were diagnosed with breast cancer?	Cancer as a shock and unanticipated experience		
 P: Yeah, I discovered this disease before one year and half. Yeah, and how? it was by sudden. I put my hand on my breast and felt something is not normal, something hard like a bone in my breastso, I talked to my husband and he insisted me to go to the hospital and check. I went to Al-Zahra hospital in Sharja before coming to Tawam. Doctor at Al-Zahra hospital checked with her hand and asked me to do the biopsy. I felt that she was not happy and was worried. She asked me to take the biopsy because she wasn't sure and wasn't quite happy with the condition of my breast. And I got the biopsy near here and they told me that you have this breast cancer. And after that I decided to go to 	Openrelationshipwithhusband/Family supportSeeking care in different facility,lack of trust in one facility; healthdecision-making Willingness toget treatment		
I: What were your thoughts and feelings when you	Feeling unhappy		
P: What they told me? Of course, I was not happy because there were some worries and things to be	Feeling worried Feeling strong		
worried about but you Muslim always has faith so I said alhamdullila; it worked for me so I should say alhamdullila as I have faith. So, that was from my	Expressing gratitude Feeling strong		
husband, <mark>I was strong hamdullila</mark> .			



	Having faith, religion as source of faith and positivity Praising Allah often Support from husband Source of support
I: What are your experiences with having breast cancer, its treatment, side effects and care that you received at the hospital?	Problems due to treatment Health decision-making
P: of course, there were some difficulties especially throughout the mechanical treatment. All the side effects that I faced was in the chemo. The side effects were on my skin, darkness on my face, on my arms, on my neckand the foodmy appetite was very bad, couldn't eat. I lost weight, lost weight around five kilos. My eye brows started falling during the treatment, there was all the time movement but now I stopped	Skin changes, body changes outlook change Health went down Weight loss
<mark>chemo</mark> , things are better now alhamdullila.	Feeling better
I: What physical changes or difficulties you faced? P: About physical problems, the main problem I faced was the outlook and changes in my face and my skir; there were no eyebrows, no hair, like this	Changes in physical appearance
I: Did you face any difficulties in obtaining or getting information about your disease or care?	Access to information Sources of information



P: For information, of course now a days, there is no difficulty if you want to know anything; just google and YouTube and you will know everything.	Ease of getting information
 I: Did you have any social or emotional concerns with the either family, friend or relatives? P: I have found so much support from my family, from my relatives that's why they gave us strength hamdulilla. 	Supportivefamilyandfriends/relationshipsSource of strength
 I: Did you have any financial concerns during the disease management? P: No, I had no financial problems as it is covered hamdullila. 	No financial issues, Health coverage by the government. Praising God
I: Did you have any spiritual or religious concerns while having disease? P: When we have faith, it helps us pass through all the difficulties hamdullila. I feel sorry for Non-Muslims. I think how non-Muslims face the problems in their lives hamdullila, Islam is a gift, it is really a gift. It helps us to pass everything and I feel happy because you know the coming thing is better.	Describing value of faith/reliance on God Describing how faith helps in facing difficulties Praising Allah Describing the importance of importance of religion Praising religion/Islam



	Faith as source of positivity
	Feel happy and positive/hopeful
1: Tell me about the treatment choices and life	Multimodality treatment
changes you made after the treatment; how breast cancer impacted your life.	Willingness of treatment
P: I finished chemotherapy, I finished surgery, I finished	Long-term treatment
the radiation and now I am on the hormonal treatment.	Physical issues
I face with Chemo treatment mainly physical and	Outlook issues
outlook change. It was difficult but I found a way to deal with it through my faith and my family support .	Expressing difficulty
	Faith helped in dealing with the disease
	Supportive family
I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other)	
P: My major concern was physical appearance like, no hair, skin change	
I: How those influenced your breast cancer experience? (prompt: health, family, work, relationships).	Expressing difficulty.
P: Of course, as I said, it was challenging to deal the problems but hamdullila I managed all with my family	Expressing an ability to manage/feeling strong
and my doctor and most importantly with my faith in	Faithassourceofstrength/reliance on God



Allah, there were difficulties but I managed all	Praising Allah
	Family support
	Support from Doctor
	Support Holli Doctor
I: How did you manage to deal with them or from	Praising Allah
where you received the support?	Family support
P: My family and my faith in Allah made me pass	
through all difficulties.	
I: What do you think and what elements of the	
service delivery and care at the hospital you were	
satisfied or not satisfied with?	
P: I was satisfied and I am still satisfied especially with	Feeling confined
my Doctor Mouza Al Ameri. She is very nice doctor.	Support from the doctor
We don't feel like her as a doctor, she is like a friend or	
sister. She even calls us at our home to make sure and to	
do the check. So, I am very satisfied with the whole	
thing.	
I: What change or improvement you would expect if	
possible or what kind of support you need? (prompt:	
what comes to your mind, what do you want in terms	
of support that can make you feel better or to	Expectations
improve your experience).	Informationseeking
P: I really needed a Nutritionist. You just need someone	
to advise you how can one deal with appetite. What	Source of information
should we eat and what we shouldn't eat? That is what I	



	I
needed most. I am very far from the hospital (Al Ain). I am from Kalba it is two and half hour. So sometime I cannot reach to the doctor and in that case, internet helps me.	Location very far to access healthcare
2- PARTICIPANT - EMIRATI	
Interview	Exploratory comments
I: What were your thoughts and feelings when you were told you had breast cancer?	
P: Before three or four years, I had something under my arm. like small pimple and also my breast is not equal like my other one (yaani/means) something different I	Breast as abnormal in feel and look
feel it. Then I said, it happened but it will go. But it took time and there was wedding of my sister and then I said not now till my sister's wedding is over. You know then	Considering light, delayed seeking medical advice
after that I went to see the doctor here in Sharjah. They do the screening campaign to check the breast and when I went there to see the doctor without an appointment	Not a planned visit to the doctor Not thinking or expecting a health issue/disease
and I will return hamdullila. The doctor wrote me to do the mammogram after which he will see me. Then he did the mammogram and asked for how many days you	<mark>Feeling light/not taking it</mark> <mark>serious</mark>
are suffering from this and he said it's better to take the biopsy. He said directly that it's no good but we should take a biopsy and I will tell you exactly what you have.	Suspicion of disease by doctor Diagnosed after visiting the doctor
They took the biopsy and then they discovered that I have a cancer type (stage) II.	Fear of having bad news
Before they did the biopsy, I was all the time thinking if they will tell me something bad, what I will do But when he told me, I feel that the God gave me something	Feeling fearless



like power . I don't know, I wasn't afraid when I heard. I thought maybe I will cry. I said khalas (done), this is the moment, I will let. If this is what, God want it. Directly I asked what I should do and he said I will start do an operation, it's better directly.	Feeling strong and fearless by Allah Power given by Allah Allah's decision
I: What physical changes or difficulties you faced?	
P: See like, it was more pain on the left side, they took too much flesh from my left side but hamdullila I have	Pain and feeling lack of flesh after operation
my two son and daughter, they are with me. They told me, do like this, don't eat like this, put me in a good way	Family support,
you know for my food, my exercise. Every night I stayed in the hotel in Al Ain and I did exercise like walking	Looking after herself
hamdullila they were with me and I managed. I had no	Family helped in managing the disease
carethis is the last thing I care about. Also, my all family, all my sisters, they were all with me.	Physical look is the least priority
I had an operation and radiation but I have been alright during and after the treatments hamdullila.	Supportive family
You don't know what happened to me after operation,	Unsatisfactory care Incomplete operation quoted as
remove the lymph node under my arm. They took like	wrong by patient
half from my breast and the doctor said, khalas, you are ok and like this. When they took the biopsy, it written in the result, there are no lymph nodes involved and only	Seeking advice from another
the fat and my doctor, Dr. Riyad was so much angry	doctor showing lack of trust
removed everything. When I saw my doctor, directly I	



went to Hurrya kaabi, she is a very good doctor. She did the mammogram again for me and scanning and everything and she said its better to remove all and the second one; even if you don't have a cancer in that, remove it. Directly I went to Tawam to see Dr. Mouza and she made the surgery for me and I said remove both. She said you don't have a cancer in other one and I said, it's better for me to do it both.	RevisingtheinvestigationshowinglackoftrustintheshowinglackoftrustinthereceivedhealthcarefromhealthcareprovidersfromSoughtmultipleopinionsshowinglackoftrustinprovidersself-decisionmakingtoremovebothbreastself-decisionself-decisionself-decision
 I: Did you face any difficulties in obtaining or getting information about your disease or care? P: You know Dr. Hurrya, she explained to me everything and also Dr. Mouza. I had no problems with obtaining information at all. 	No problems in accessing information Satisfied with the given information
I: Did you have any social or emotional concerns with the either family, friend or relatives?	
P: No, alhamdullila, they are all big now and they are the one taking care of me. I don't have a husband, he passed away. Also, my husband, he had a cancer when he passed away. I was with him and I know everything.	Lost her husband due to cancer Known to cancer experience
Like this, I don't want to suffer when I think about him, I don't want to be like him. He was too much depressed. He couldn't control himself. That's is why he didn't even take the medicine. There are patients sometimes	Want to avoid depression since she had this experience with the husband
they give up and say khalasif I will die, I will just die. For me, No. I said myself, I will fight. Because I love	Self-motivation to fight cancer Want to live for children



my children, I want to see their children and become a grandmother. How you think, its very important. Think in a good way.	Thinking positive
 I: Did you have any financial concerns during the disease management? P: No, I had no financial or social issues. My healthcare is covered by insurance alhamdullilah. 	No financial or social issues Have full health coverage with insurance
 I: Did you have any spiritual or religious concerns while having disease? P: When I think about this, I said, may be Allah loves me. I discovered so many good thingspeoplelike this 	Feeling close to Allah Faith brings opportunities/good things
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: The difficult part was after operation. I had pain and sometime I was about to vomit. I suffered from this part so mainly physical. 	Difficulty in living with cancer Most challenging was the physical concern
 I: How did you manage to deal with them or from where you received the support? P:: Mainly from my children, my family and my sisters. I had lots of support from my family. I managed well with their support. 	Supportive family Managed the disease well because of family support
I: What do you think and what elements of the service delivery and care at the hospital you were satisfied or not satisfied with?	Confined with the care received at Tawam



P: Yes, I am satisfied with the care I received, how they treated me a good way. All the people are good even	
when I had radiation, they were helpful people hamdullilah.	
 I: What change or improvement you would expect if possible or what kind of support you need? (prompt: what comes to your mind, what do you want in terms of support that can make you feel better or to improve your experience). P: [didn't take chemotherapy, only radiation. The radiation department is nice but for chemotherapy when I had some appointment, there is too much rush. Sometimes, [have an appointment at one O'clock, I see the doctor at three O' clock, I don't know why. So fiming to see the doctor must be reduced. 	Self-selection of the treatment or shared decision-making Long waiting time, Timing should be reduced
3- PARTICIPANT - EMIRATI	
Interview	Exploratory comments
 I: What were your thoughts and feelings when you were told you had breast cancer? P: When I was told I had cancer; I was sort of shocked. Like I paused First I was little depressed of course it's 	Diagnosis was shocking Bad news



unexpected that you never thought of it. I just accepted	
that it happened, what else I could do.	
I: What physical changes or difficulties you faced?	Pain was the most challenging
P: Pain was the main problem. Also, I was having sleep	Sleep difficulties
disturbance but it was also because I lost my son. I had	
a good feed so I didn't lose weight. I would say I	Son death made her very
	emotional
managed with the support of my family and my doctors.	
It was sure not easy and sometimes pain and other	Family and doctor support
disturbances were too much for me to handle but I	helped a lot to manage the
1 1	helpeu a lot to manage the
managed somehow.	disease
As I said, I did not have much difficulty but of course	Suffered from pain
after Surgery I suffered pain. But that didn't impact my	
life as much when I lost my son. His loss made me	Son death made her emotional
emotional and low. I dealt with the cancer but his loss	Son dooth was more challonging
was the most challenging in my life because I was the	Son death was not e chanenging
	than cancer
one with him and taking care while he was ill.	
I: Did you face any difficulties in obtaining or getting	
information about your disease or care?	
	No information concerns
P: No, I was fully informed and multiple times about	
the side effects of medications I had. I was fully	Was fully informed
informed of the symptoms I had. Healthcare department	Health facility was approaching
was calling me regularly and asking me how I am doing.	her regularly
how I am progressing.	nor regularly
I: Did you have any social or emotional concerns	Not very difficult
with the either family friend or relatives?	



P: There was no as such difficulty. My sister in law had a cancer so I was comfortable talking to her and get the support I needed. I do not have much friend but I spoke to my family and they were all with me to support me. Well, I wasn't going out much and I was bit down so I was always with my family. But I had a fear of recurrence because I was diagnosed and then later, I was diagnosed for another cancer on other side so I was always having this fear that I might get another cancer. My husband was very supportive thought and I had no issue talking to him about it.	Sister in law had a cancer so got all the support within family Supportive family Feeling isolated/Not many friends Always with family/relying on family Fear of having cancer comeback again
I: Did you have any financial concerns during the disease management? P: I no financial problem; my healthcare was covered by insurance.	Husband was supportive and open to talk about issues No financial issue Health coverage by isurance
 I: Did you have any spiritual or religious concerns while having disease? P: I managed it all with my faith in Allah and Allah made it easy for me. I found more peace and courage with my strong faith in Allah and in my religion. 	Reliance and faith in Allah gave peace and strength
1: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical,	Son's death was the biggest challenge than cancer



information, social, emotional, spiritual or any other)	Son's death impacted her emotionally more than cancer
P: The biggest challenge was, my son passed away and that was more concerning to me than my own health. He had a comorbidity and was bed-ridden for a long time and I was taking care of him and he passed away just two months before I was diagnosed with cancer. When I was taking care of him, I knew I had a lump but when he passed away after that I decided to go and see the doctor. Before that I was ignoring my condition. This was the part that impacted me most and even now I think of my son that more than my condition.	Intentionally delayed in seeking medical care because of other life priorities (son)
 I: How did you manage to deal with them or from where you received the support? P: I did not have many health issues and my family was very supportive. I do not work and I have few friends but my relationship and my life were not altered due to cancer. I had other life challenges. Mainly family, my family was very supportive 	Not very social and have few friends Cancer was not the priority but other life challenges mainly son's health and death effected deeply
 I: What do you think and what elements of the service delivery and care at the hospital you were satisfied or not satisfied with? P: I was fully satisfied with the care I received. Everything was good and doctor kept on checking about everything. 	Very supportive family Fully confined with the care and health providers
I: What change or improvement you would expect if possible or what kind of support you need? (prompt: what comes to your mind, what do you want in terms	Meet all expectation, fully satisfied



of support that can make you feel better or to improve your experience). P: I don't think there is anything missing from the healthcare department. I received all the updates I needed from my doctors and nurses and the doctor was calling me to check on me how I am doing so I got the sufficient support from the healthcare department.	Doctors were asking and updating regularly Sufficient support by health providers
4- PARTICIPANT - EMIRATI	
Interview	Exploratory comments
 I: What were your thoughts and feelings when you were told you had breast cancer? P: I was surprised but it was my destiny. What I was going to do. I was scared though because I have small kids, I was worried and of course unhappy but it's from Allah and it took it easy. 	Feeling scared because of kids who are young Feeling worried Feeling unhappy Was not expecting cancer From Allah's believing he is the owner of our destinies Accepted it
I: What physical changes or difficulties you faced? P: I had chemotherapy, my eye-sight dropped and I couldn't see particularly at night. Mainly the left eye. At night I had all blurry vision. I got pain in my bones and body. When I was taking her chemo, my hair fell. And	Had multimodality treatments, surgery, chemo radiation Had pain, blurry vision, hair fall



then I took radiation for a month, I lost weight. When I had radiation on breast, I had pain on the area where I had surgery but it wasn't that much. So that was all. Sure, I had difficulties but I was able to manage my daily life and tasks as normal.	Lived with difficulties but managed daily life
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: They were clear and I had no difficulty in obtaining information. I was told everything.	Was fully informed, had no problems obtaining information
I: Did you have any social or emotional concerns with the either family, friend or relatives? P: Everything and everyone was good and supportive. I had no emotional or social concerns hamdullila. In our family, there are lots of people who have cancer. My sister also has breast cancer so it wasn't that much. Also, my work was OK and flexible and they knew my condition and allowed me to leave at twelve because I couldn't stay more as I got tired. They told me; you can leave early. My family treated me very normally. Of course, it was hard for them to know that I have cancer but they didn't reat me as something is abnormal. Hamdullila, I didn't experience any psychological or emotional issues with family or with my friend or colleagues.	Good supportive family No emotional or social issues Family has cancer history so it was ok to talk about it and family treated her normally not something abnormal At work, people and employer were supportive and considerate towards her condition



I: Did you have any financial concerns during the disease management?P: No, I had no issues with finance, it is covered and they did everything.	No financial difficult, had heath insurance
 I: Did you have any spiritual or religious concerns while having disease? P: when I was diagnosed with cancer, since then I got much closer to Allah. During my disease and while I was receiving my treatment, I found myself closer to Allah. 	Got closer to Allah due to disease
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: I think the most difficult for me was, I was away from my kids because I had my surgery done in the Germany and this was the biggest challenge for me to be apart from kids. It wasn't easy. 	The biggest challenge was not the cancer but being away from family because of cancer treatment
I: How did you manage to deal with them or from where you received the support? P: Mainly I got all support from Allah and my family.	Reliance on Allah Supportive family
I: What do you think and what elements of the service delivery and care at the hospital you were satisfied or not satisfied with?	Fully satisfied with the received care



P: I was satisfied with the care I received. I didn't have any issue really.				
I: What change or improvement you would expect if possible or what kind of support you need? (prompt: what comes to your mind, what do you want in terms of support that can make you feel better or to improve your experience). P: Tawam pharmacy services are very slow, I think that should be improved for faster services and better delivery.	Pharmacy improved for	services • efficiency	to	be

5- PARTICIPANI - EMIRATI	
Interview	Exploratory comments
I: What were your thoughts and feelings when you were told you had breast cancer?	
P: I don't know, I was expecting that. I don't know why. I felt that I was having a breast cancer soon. Doctor was	It was not a shock but expected
hesitating to tell me and I told her, I am having breast cancer and she said yes. Nothing, not because of	Her sister had it but she didn't have a feeling that she was
anything, just like this, I just felt I was having a breast cancer. My sister had a breast cancer and she passed away but it doesn't relate to the fact that because my	having because of her. It was a feeling she will have it
sister had it.	
I: What physical changes or difficulties you faced?	Had multimodality treatment,
P: Pain was the most of difficult. Since I did the radiation, my skin turned darker at that place. I had the	radiation, surgery, hormone therapy.

5- PARTICIPANT - EMIRATI



surgery. Plus, the look of the surgery is very bad. It is	
affecting me psychologically. They told me that pain	Pain was the most challenging
will go away after one year. It has been more than one	Surgery affected her
year and I am having that pain, seriously unbearable	physiologically
pain but they just gave me sick leave for just six months	
and then they let me go and said I am fit but I was not	Sleep disturbance
fit. And since I was taking this hormone therapy, I was	
having my periods more than twenty days, so my health	
became very bad. I am having flue frequently. Most days	
I have this flue. There is no day when I don't have	
congestive nose. My immune system became down and	
I told them and they said everything is fine. How? I	
wasn't like this! Whole night I couldn't sleep because of	
my congestive nose, how could I do daily work. Also, I	
am having asthma so it effects my chest so until now I	
am suffering from this.	
I: Did you face any difficulties in obtaining or getting	
I: Did you face any difficulties in obtaining or getting information about your disease or care?	
I: Did you face any difficulties in obtaining or getting information about your disease or care?	
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the	Lack of information provided
 I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the university and they are coming to say to you. They don't becaute a set in a set	Lack of information provided by doctors
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the university and they are coming to say to you. They don't know the result; they think I don't have anything so their	Lack of information provided by doctors
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the university and they are coming to say to you. They don't know the result; they think I don't have anything so their answers were not convenient to me at all. Some words	Lack of information provided by doctors Not being normal or do normal
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the university and they are coming to say to you. They don't know the result; they think I don't have anything so their answers were not convenient to me at all. Some words were related to me but others not especially my health	Lack of information provided by doctors Not being normal or do normal day work
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the university and they are coming to say to you. They don't know the result; they think I don't have anything so their answers were not convenient to me at all. Some words were related to me but others not especially my health was going down and they didn't answer me. They said,	Lack of information provided by doctors Not being normal or do normal day work
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the university and they are coming to say to you. They don't know the result; they think I don't have anything so their answers were not convenient to me at all. Some words were related to me but others not especially my health was going down and they didn't answer me. They said, you are fine and this is normal. What's normal? Normal	Lack of information provided by doctors Not being normal or do normal day work
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the university and they are coming to say to you. They don't know the result; they think I don't have anything so their answers were not convenient to me at all. Some words were related to me but others not especially my health was going down and they didn't answer me. They said, you are fine and this is normal. What's normal? Normal is when people can do and practice daily life. They can	Lack of information provided by doctors Not being normal or do normal day work
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the university and they are coming to say to you. They don't know the result; they think I don't have anything so their answers were not convenient to me at all. Some words were related to me but others not especially my health was going down and they didn't answer me. They said, you are fine and this is normal. What's normal? Normal is when people can do and practice daily life. They can go and come as they wish. This is not normal!	Lack of information provided by doctors Not being normal or do normal day work
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: Well, all doctors are like they remember in the university and they are coming to say to you. They don't know the result; they think I don't have anything so their answers were not convenient to me at all. Some words were related to me but others not especially my health was going down and they didn't answer me. They said, you are fine and this is normal. What's normal? Normal is when people can do and practice daily life. They can go and come as they wish. This is not normal! I don't think they told me my health would be like this.	Lack of information provided by doctors Not being normal or do normal day work



need to do the operation, radiation. We need to do the hormone therapy which I didn't complete because had this period for long time and I was outside the country and when I came back, I told my doctor this and she said, you are just sayingof course I am saying, I am not doing something from my mind that I am sick or	Lack of understanding by doctors Lack of trust in doctors Lack of support from doctors
something. Even if I am another country and I told you that I have this, how could you know that that I am having this or I don't have. You won't see inside me. What is this. Doctors have to believe what patients is saying. I am not creating these things.	
 I: Did you have any social or emotional concerns with the either family, friend or relatives? P: I couldn't tell my mother; I am having this disease. I used to tell her that I am going outside for my studies and I used to hide in my sister's house. When I used to be back home, I still used to have this pain but I had to hide it. My mother passed away and I couldn't even tell her, you know. They saw it affected me too much and after two months, my sister who had a cancer, she also died. That is like destroyed me really. 	Had to be silent, couldn't share Feeling regret Family death effected a lot (mother and sister both died)
 I: Did you have any financial concerns during the disease management? P: Alhamdullila no, I didn't have any financial problem because this is all from the government. 	No financial prob Health coverage by the government
I: Did you have any spiritual or religious concerns while having disease?	Faith became stronger Praising God Reliance on God



P: this disease made my faith into God stronger. Because the way God stands with me, nobody can be with me. In this situation, you will feel how God is closest to you.	
I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other)	Pain was the biggest concern Sleep disturbance
P: Pain, physical concerns for sure. This was the biggest challenge because the pain, pain on the chest cannot let you sleep and you have to be on drugs and drugs also	Difficulty in managing work life balance due to drugs
not good for you. Its like, you have to manage your pain plus you have to go to the work. How could you go to	Difficult living <mark>Multimodality treatment</mark>
the work like this? Until now, you know I cannot wear the bra. It's very hard for me to wear the bra because when I wear the bra, my pain increases. For example,	Decided to stop hormone <mark>therapy</mark>
until now when my nieces and nephews come and hug me, my breast is like somebody hit me or something like this slightly and I feel pain. I had a radiation: surgery	Lack of trust in doctors
and I started this hormone therapy then I stopped it as I told you and all doctors are like take ittake ithow	Unsatisfied delivery from doctors
can I take it? Even I asked them, did you have a patient like me, this happened and they said let's try. What's let's try? I am asking them give me information. Did it	
happen before, is it something naturally happening? Do you have other cases or not? So, in that situation, they just want to do what they learned.	
I: How did you manage to deal with them or from where you received the support?	Lack of support from people
	Colleagues at work were supportive



 P. I didn't have the support except people at my work helped me. Still I am suffering that I have to go to work and I still don't have a reason that could stop me to go the work, even the doctors didn't support me in this. However, I had all the support from my sisters but I couldn't tell them what is in my heart because you know they were also going through this tragedy. And they also lost one sister because of cancer. I have to be careful what I am saying to them and all this thing. 	Problems in managing work and life Supportive family Hide emotions Family had several deaths	
 I: What do you think and what elements of the service delivery and care at the hospital you were satisfied or not satisfied with? P: Cancer speciality compare to others; people are very caring. You have very good treatment so you cannot say that you didn't have a good care but lack of information from the doctors, this affects. 	Satisfied with care received Information from doctors lacking	
1 – EXPAT PARTICIPANT		
Interview	Exploratory comments	
 I: What were your thoughts and feelings when you were told you had breast cancer? P: I was shocked and devastated to hear that I have cancer. It was sense of denial in the beginning as I never thought myself being in that situation. I had no idea, it is gonna be that tough. 	Cancer was a shocking news Couldn't believe Unexpected news	
I: What physical changes or difficulties you faced?		



P: My life has changed including me, my appearance,	Life changed after cancer in all
my feelings, my priorities. Everything has changed to	aspects
the extent that its all just new. It has changed my life	-
completely. <mark>I put myself into forum mainly because of</mark>	Joined the forum
the effects chemo has had on me. It was after the first	Feeling devastated
chemo session. It was like I was devastated. As I	
explained to you, I associated myself with the forum and	Joined the forum
doctors explained to me. I did start my chemo on	Vomiting and diuarrhoea were
Wednesday and it was only on Sunday when I woke up,	intense
I started vomiting and diarrhoea and my stomach was so	
tortured. Nothing prepared me for that. Yes, I prepared	No one prepared her for those
myself for that but not to that level that will float me. I	intensed physical symptoms
had read, I had researched but I did not know that it was	Unable to go to the bathroom.
such a fine balance that you had to run between	was not able to do anything
diarrhoea and normal bathroom needs that needs to be	
there that was not there. That was so like today I am	Had intense pain just like
going to bathroom and after that I couldn't go back to it	childbirth
and it was escalated so much so first time, I was just	Sleen problems due to fotique
float. My whole face showed it. I was just in my bed; I	sicep problems due to langue
couldn't do anything just couldn't do anything.	Had hair fall that her pillow
Pain in terms of I do not take the tablet, the yellow tablet	was covered with hair that she
I do not take one. The intensity of pain and cramps:	had to remove all and used veet
having all your periods and childbirth is just nothing like	cream
that swear to God. And other was fatigue, I couldn't	Recouse of chamatherapy had
leave my pillowthat was the level of fatigue. And	bed teset in mouth
Inability of sleep, I was just awake. I would sleep one	bau taset in mouth
hour and then awake. My hair went off. I was just	Also because of chemo, she lost
showing to my nephew the picture of my pillow after my	7 kilos
second chemo. And my pillow, it was just full of hair,	Chow howelf strong and
that was the amount of hair I lost. I just went to the bath	buow nersen strong and
room, I didn't have the shaver, I just put the Veet all over	positive



my head And taste in mouth, anything I eat, it tastes off. I couldn't eat anything and whole six months of chemo because of that and because of side effects of chemo I lost 7 kilos I have a ten-year-old daughter so I had to be brave and strong for her. You know what I mean so I tried to get myself positive as I could. Having said that I had lots of friend assisting me. My principle was assisting me. I am a teacher so she took away some of my classes so it won't drain me out. She put another teacher with me in the class so if I can't do anymore, she	Supportive friends Supportive employer Difficult living Effected psychology
life was just, I just lived for weekend so I could just stay in bed. I just did the most essential groceries in my house and put like noodles, easy things for my daughter to make for herself. It had also psychological effect, I became very depressed	
I: Did you face any difficulties in obtaining or getting information about your disease or care?P: Yes, I was fully informed. My doctor, my oncologist	
she explained to me what's going to happen but as I said I didn't own it. I didn't realise that's gonna be you that's gonna happen to you like other people. Everybody experience is different for example sense of taste was	Was fully informed but didn't own it Sense of taste and memory loss



will be more patients waiting outsider her room, you	Concern notions has loss of
know what I mean. You know all my questions were	cancer patient has lots of
answered, I have to say that but like I have had time to	questions
think about it and I have more questions so whose gonna	
follow that. It cannot be the doctor because she has other	
patients to see. So, example, having my double	
mastectomy, when do I stop it? Yes, I was told and I	Usually healthcare providers
remember everything what the doctor said at that time	miss psychological and
but there was time when I was like what do I do now?	emotional aspect
How to empty it and everything. They go very efficient	
in everything but the psychological and emotional side,	You are unaware and not
nobody said you are gonna be depressed as afterwards.	prepared for lots of questions
If you are gonna see the mirror for the first time, you are	Live with only daughter and
gonna break down in tears no body prepared you for	have to be strong for her
that. Side effects, yes gonna be there but to what	have to be strong for her
levelI cannot say questions answered but addressed.	<mark>Cry alone</mark>
I am here alone with my daughter, it's difficult because	
I have to be strong for her. The only time, I can be	Had bad side effects of chemo,
myself is when she is gone to bed and I can lie and I can	diarrhoea
<mark>cry into my pillow</mark> . The side effects of the chemo, to use	Severe side effects must be
a bad wordjust terrible it is. Side effects are long, its	informed earlier
like, one time I found myself its like crawl to the	
bathroom like I have diarrhoea. Tell me these things, I	Life will not be the same
don't want to hear that you are gonna be OK cause you	side effects of chemo are severe
are not going to be OK. Your life just will not be the	and that should be informed by
same again. For example, when I recovered, fifty	doctors
percent of the hair has grown back. One person said,	
after one year of chemo, she cannot still remember how	cannot go and feel like telling
she coped till today. She had to rely on people to tell her	her situation to friends and
oh, this is what happened and so. So, be clear like my	<mark>colleagues</mark>
doctor said chemo will not but the side effects throw	cancer survivors chould feel
you. I didn't feel like I want to go my friends and	safe to talk about their issues
	sale to talk about their issues










Lot of the time, initially first few weeks of my chemo, I	
didn't do it. After that it had to be postponed to next	
Wednesday because it wasn't approved. I have had a lot	
of issues in the beginning. I can show you my phone	
calls; every time I come for chemo, I have learned that	
the day before I had to call them to ask them if it is	
approved. The ladies here, they are putting the request	
and then tell me that I put it in ten days ago and say they	
are still working on it. It is ridiculous! But to be honest,	
I would be at home and it would have been coming out	
of my pocket, I am extremely grateful that I am here in	
the UAE. That burden has been taken of me because I	
work for the ministry of education. My chemo is	
basically paid for, alhamdullila totally alhamdullila.	
And I am so lucky that people who are coming to my	
life especially Dr Mouza has been brilliant.	
nie ospeciality 21 moulu nus ocen omnand	
I: Did you have any spiritual or religious concerns	
I: Did you have any spiritual or religious concerns while having disease?	Religion to hold to during
I: Did you have any spiritual or religious concerns while having disease?	Religion to hold to during
 I: Did you have any spiritual or religious concerns while having disease? P: We need religion like something to hold onto. Since the birth of my daughter, before that I was kind of wild 	Religion to hold to during difficult time
 I: Did you have any spiritual or religious concerns while having disease? P: We need religion like something to hold onto. Since the birth of my daughter, before that I was kind of wild. Conception was an issue with me and I kind of made a 	Religion to hold to during difficult time
 I: Did you have any spiritual or religious concerns while having disease? P: We need religion like something to hold onto. Since the birth of my daughter, before that I was kind of wild. Conception was an issue with me and I kind of made a bet if I had a baby and I could carry it to the term then I 	Religion to hold to during difficult time
 I: Did you have any spiritual or religious concerns while having disease? P: We need religion like something to hold onto. Since the birth of my daughter, before that I was kind of wild. Conception was an issue with me and I kind of made a bet if I had a baby and I could carry it to the term then I would change my life. I went into hijab after that and I 	Religion to hold to during difficult time Being survivor made her
 I: Did you have any spiritual or religious concerns while having disease? P: We need religion like something to hold onto. Since the birth of my daughter, before that I was kind of wild. Conception was an issue with me and I kind of made a bet if I had a baby and I could carry it to the term then I would change my life. I went into hijab after that and I kind of decided I have always been a Muslim but now 	Religion to hold to during difficult time Being survivor made her practice even more
 I: Did you have any spiritual or religious concerns while having disease? P: We need religion like something to hold onto. Since the birth of my daughter, before that I was kind of wild. Conception was an issue with me and I kind of made a bet if I had a baby and I could carry it to the term then I would change my life. I went into hijab after that and I kind of decided I have always been a Muslim but now I am gonna be a stonged Muslim. I made a lot of 	Religion to hold to during difficult time Being survivor made her practice even more Being survivor made her cry
 I: Did you have any spiritual or religious concerns while having disease? P: We need religion like something to hold onto. Since the birth of my daughter, before that I was kind of wild. Conception was an issue with me and I kind of made a bet if I had a baby and I could carry it to the term then I would change my life. I went into hijab after that and I kind of decided I have always been a Muslim but now I am gonna be a stonged Muslim. I made a lot of Tahajjud (night prayer) and salah in the Ramadan. I just 	Religion to hold to during difficult time Being survivor made her practice even more Being survivor made her cry while praying
 I: Did you have any spiritual or religious concerns while having disease? P: We need religion like something to hold onto. Since the birth of my daughter, before that I was kind of wild. Conception was an issue with me and I kind of made a bet if I had a baby and I could carry it to the term then I would change my life. I went into hijab after that and I kind of decided I have always been a Muslim but now I am gonna be a stonged Muslim. I made a lot of Tahajjud (night prayer) and salah in the Ramadan. I just sit on my muslaah (prayer mat) and cry cry in the 	Religion to hold to during difficult time Being survivor made her practice even more Being survivor made her cry while praying Beligion guides you
 I: Did you have any spiritual or religious concerns while having disease? P: We need religion like something to hold onto. Since the birth of my daughter, before that I was kind of wild. Conception was an issue with me and I kind of made a bet if I had a baby and I could carry it to the term then I would change my life. I went into hijab after that and I kind of decided I have always been a Muslim but now I am gonna be a stonged Muslim. I made a lot of Tahajjud (night prayer) and salah in the Ramadan. I just sit on my muslaah (prayer mat) and cry cry in the middle of the night. I have done more sunna prayers. I 	Religion to hold to during difficult time Being survivor made her practice even more Being survivor made her cry while praying Religion guides you



but in an effort to sooth myself. Because if you don't have religion, you are lost. You know what I mean, so yes, I have gone closer to my God	Being survivor make you close to God
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: I had double mastectomy and my breast were removed. After that I had chemotherapy and hormonal. At this point of my life, as you can see, even 	Had both breast removed and chemo and hormonal therapy after that Anything afftect you after treatment even little thing
the small thing will affect your life. It effects your sexuality. You know what I mean. It effects how you dress. Like inside my apartment, I would go without my Sheila (scarf) but even in the balcony, I am gonna cover myself because I just don't want everybody to see. I	Her sexuality was effected Covered herself so people can't see Choice of dress became difficult
it's difficult. It affects me what I choose to wear. It is long sleeve, it is not tight fitted, I don't wear my swim wear, you know, its changed. Every aspect of your life changes drastically.	Couldn't have conversation with people, became very difficult Fear of losing job
It was my face. Because it's like I would have conversation with my gardener; I can't do this anymore. This is too difficult.	The highest concerns were physical and <mark>emotional</mark>
That is always at the back of my mind, tomorrow I can lose my job and then what do I do? Physical concerns and emotional concerns were the most important issues I had to deal with the cancer.	Lot of psychological and emotional effect on her, thought it will be easy
The psychological and emotional impact it had on me, I didn't know how this will affect my life like the first	Didn't take hormone tablets thinking it wont happen to me



time I had lump on my side so I thought, they will take it out and I will be good. I didn't realise taking the hormonal tablets after. I did not think its gonna happen to me.	When it came on the other side then wanted to talk to someone
You're always like it is gonna be the next person. So, when it recurred on the other side, I was like, no, this is really serious and that point, I needed to speak to someone.	<mark>No network available to help</mark> her
I did not have the financial ability, there is no network that tells you that someone you can speak to. I didn't have it.	Decision to remove both breasts was intense, wanted to speak to someone
The decision to remove both of my breast, it was such an immense step; I needed someone to speak to.	Self-perception changed Will affect me finding a partner <mark>It was difficult to show her</mark>
Because my view of myself changed and it was going to change immensely after and coming to terms to my new body, I can tell you at this point that I do not have partner in my life but this is gonna impact me having one again.	daughter Self-acceptance was imp at first Didn't want mirror to show full body
I don't wanna share as I said I have daughter, I didn't want her to let her see because first I had to own it to myself, I had to look at it and I didn't after the operation. I would just not look into the mirror. And I had a mirror the one full body mirror which I put it in my room and	It was devastating for daughter too Daughter wanted support too
now I have just small round one and I needed to speak to someone. And I feel my daughter needed to speak to	



had

information about places and

little

Non-locals

someone apart from me because we have been here alone for ten years and because my nephew came and he is with us here now and we didn't have anyone. And this was devastating for her. Because I had lost my best friend, she had colon cancer. Feb 22^{nd,} she passed away so she thought immediately that was gonna happen to her. So, I feel she needs someone, she needs just to speak to let it out.

Also, because I am an expat, if it was a local lady, they would know who to speak because they know the place and connections. Because I am not from here, I do not know, I can go there or go there. It would be an immense help.

I am blessed with a very good principle alhamdullila and she has been immensely supportive. Like I said, the accommodation they made for me, me to have absence, absence, absence so she said, I could come to the school. If I had to use my key, come in and sign in so like I am not absent. Not letting Ministry of education, they have basically covered me alhamdullila but this is because of the calibre of person she is. But to be honest the previous principle, she is the new one; the previous would had done the same. But I don't know the Ministry of education, what is their policy, you know what I mean. Of course, you are conflicting stories. Luckily, I would be put in the position where I was, I had to face that.

When I came for my first operation, Dr. Mouza said, almost like my entourage, there was always someone with me. A friend of mine Amina, we were together so I went for my operation, I was unable to cook anything

people who could support Supportive employer and co workers Supportive healthcare

providers. Colleagues

Received lots of support from friends

Fear of having cancer again



so we have like EMT in school, English medium	
teachers and they made like every night a different	Preparing her daughter
person will come and bring me lunch or dinner then my	emotionally so she can speak
friend Amina she is local, she said No, Noshe is my	about her fears
sister. I am not lying, I swear to God, every day I had	Satisfied with care
lunch, I had dinner. It would be left at my door. Every	
day she sent, somebody from cleaning company, every	Provision of information must
day my apartment got cleaned because I cannot, I must	<mark>be improved</mark>
not do anything. Alhamdullia she took me like her sister	
and she did it for the whole of my period when I was	
recovering till I do it and I said Amina, I can do it and	
only then she did stop. I am so blessed, absolutely	
blessed but I might be the exception, you know what I	
mean. This may not happen to others.	
The fear of recurrence that it will come back, it was	
always there in the back of my mind that I have to do	
revisions for my daughter if I am not there anymore and	
I have to prepare her for that situation and like we never	
know what will happen so we talked about it. As I told	
her she need to speaks about her fears or concerns like	
anytime she has a question, she needs to ask me.	
I was satisfied with the care but there should be an	
improvement in information delivery	
PARTICIPANT 2- EXPAT	
Interview	Exploratory comments
I. What were your thoughts and feelings when you	
were told you had breast cancer?	
were toru you hau di cast cancer :	Cancer was a shock, sad news
	It was scary



 P: First, I remember, I was sitting down and cry. It was a complete surprise for me that such a thing happened to me. Second, what I have to do. And of course, when you don't know what is chemo, how does it affect you. This scared you because you don't know anything. L cannot explain in few words. My friend she started 	
reading about it L cannot forget L couldn't read about it	
L couldn't hear about it. I don't know how to evplain	
this is something related to something you are going to	Life changed after treatment
do know about it. She told me Nadia, there are some	Mixed emotion
steps: First you are going to accept this and then will	
define everything When I had the treatment I	
understood it's my life now Of course it was angry like	Life shanged as it was in next
sadness fears it was like no emotions at all. So, all	Life changed as it was in past
emotions or mixed emotions. Lam not a kind of person	
who will sit angry why it happened to me or I am scared	
what I am scare of? Yes of course it can happen to	
anyone of course if you are standing next to the God	
and you are thinking I am ready yet actually I will	
believe It's difficult to explain emotions. If you	
compare me one and half year back and now it is	
different	
I: What physical changes or difficulties you faced?	Fatigue was the biggest issue
have any other problems from any treatment. I was very	
lucky.	No other major issues
I: Did you face any difficulties in obtaining or getting	
information about your disease or care?	



P: We are living in the world of information. We can	Information is easy to obtain
open the google and check on whatever we want. Also,	these days
we are not living out of social world. In my case I can	
say that, it was someone, I didn't know anything and	Information must be sought
even you offer me all the information which is enough	from doctors
for me but it all must be for my doctor. See if I have	Doctors should have correct
information what I am going to do with that. Of course,	information as it can impact
I understand what you mean for example, if I know	lives
something, it will change my life but no, it will not	
change my life. If my doctor doesn't know some	Was fully informed
information, it will affect me but if I don't know some	
information, I don't think it will change my life.	
I think yes, I was fully informed, even more I needed to	
know for example, Dr. Mouza she did not explain to me	
what was the treatment, she approximately explained to	
me but all what I needed to know deeply, Dr. Danijela	
explained to me and of course she is my oncologist. She	
actually explained to me more than I needed to know.	
I: Did you have any social or emotional concerns	
with the either family, friend or relatives?	
	People changed after cancer
P: You know through this sickness, almost all my	Do not have many friends, not
people changed, people whom I was with and thought	very supportive
they are close and I thought they are my relative, I am	
not talking about my husband, I am talking about my	
friends, close friends. Some of them they chill out with	Appreciating Cod's holp
me but I can say they are not so many and mostly I don't	Appreciating Gou sheip
see them. Because they show themselves that actually	Supportive employer
they are not my friends. They are scared more than me,	
they have decided to stay away from me.	



You know I think God is with me cause the treatment did not affect me that I had to stop the work. My director in the school, he knows what happened to me. Actually, he gave me one month off when I started my chemo so I can continue my job. Thanks God, may be because of my job because I knew that I needed to go, I needed to wake up, I had to go to my children to teach them; maybe it is some stamina that helps you in this I did 12 chemo then now 6 cycles of chemo tablet and tomorrow will be last day of tablet. Then I will go to Dr.	<mark>Had chemotherapy</mark>
Danijella to know what is next. That's why I did scan to see if all is alright and then next week Dr. Danijella will tell me, what will happen in my life.	
I. Did you have any financial concorne during the	
 1: Did you have any financial concerns during the disease management? P: Actually, my insurance doesn't work here, it was given from my job. Because I am from Abu Dhabi. have SEHA mandate. It is really hard for me for example in my case. I will tell you what Dr. Danijella and Dr. 	Treatment covered under government mandate Insurance didn't cover







P: You know when you pray, it releases your fears.	Prayers have positive effect on
Through prayers, the words you read in a particular way,	brain
it already gives your brain a different world and of	
course the feelings, when you are in this situation and	Now became much sensitive
your really are going to think about this. For me it's	unlike past, cries more
easier and I think I got more closer to the God. Actually,	Been more close to God
I never go out of his sight but may be before I prayed	
and I didn't cry but this time I pray and I cry so this I	
<mark>can say it's different</mark> . I think I am closed to the God. We	
are different people and with me it's like see when you	
are happy, you are closed to the God, if you are sad, you	
are closed to the God. I was waiting for something like	
it will be horrible and I wouldn't be able to wake up	
from the bed but thanks God. 90% is his job.	
I: Can you tell me what were your main concerns or	Emotions were the biggest
I: Can you tell me what were your main concerns or biggest challenges: (prompt: focusing physical.	Emotions were the biggest
I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information social emotional spiritual or any	Emotions were the biggest challenge
I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other)	Emotions were the biggest challenge Must have supportive people
I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other)	Emotions were the biggest challenge Must have supportive people
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest 	Emotions were the biggest challenge Must have supportive people
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest challenge for me because you need lots of strength and 	Emotions were the biggest challenge Must have supportive people
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest challenge for me because you need lots of strength and it is very important for you to have supportive people. 	Emotions were the biggest challenge Must have supportive people
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest challenge for me because you need lots of strength and it is very important for you to have supportive people. I: How did you manage to deal with them or from 	Emotions were the biggest challenge Must have supportive people
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest challenge for me because you need lots of strength and it is very important for you to have supportive people. I: How did you manage to deal with them or from where you received the support? 	Emotions were the biggest challenge Must have supportive people Not to feel anything is better sometimes/having no emotions
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest challenge for me because you need lots of strength and it is very important for you to have supportive people. I: How did you manage to deal with them or from where you received the support? 	Emotions were the biggest challenge Must have supportive people Not to feel anything is better sometimes/having no emotions is better
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest challenge for me because you need lots of strength and it is very important for you to have supportive people. I: How did you manage to deal with them or from where you received the support? P: The main support I got it from myself, though I am 	Emotions were the biggest challenge Must have supportive people Not to feel anything is better sometimes/having no emotions is better
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest challenge for me because you need lots of strength and it is very important for you to have supportive people. I: How did you manage to deal with them or from where you received the support? P: The main support I got it from myself, though I am very weak emotionally. I put myself through lots of 	Emotions were the biggest challenge Must have supportive people Not to feel anything is better sometimes/having no emotions is better Praising God for not having
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest challenge for me because you need lots of strength and it is very important for you to have supportive people. I: How did you manage to deal with them or from where you received the support? P: The main support I got it from myself, though I am very weak emotionally. I put myself through lots of thing. Sometimes its better not feel anything. May be 	Emotions were the biggest challenge Must have supportive people Not to feel anything is better sometimes/having no emotions is better Praising God for not having other side effects of treatment
 I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: Dealing with emotions, may be, was the biggest challenge for me because you need lots of strength and it is very important for you to have supportive people. I: How did you manage to deal with them or from where you received the support? P: The main support I got it from myself, though I am very weak emotionally. I put myself through lots of thing. Sometimes its better not feel anything. May be sometimes we don't know how strong we are and may 	Emotions were the biggest challenge Must have supportive people Not to feel anything is better sometimes/having no emotions is better Praising God for not having other side effects of treatment



will tell us. Really this is a miracle that I really didn't get any bad effects from the treatment and I think it is from God.	
I: What do you think and what elements of the service delivery and care at the hospital you were satisfied or not satisfied with?	
P: I went to the hospital and I was told don't worry, she is a nice doctor. Actually, when I started with Dr. Danijella, I didn't know that Dr. Mouza is my surgeon so it's really fun. I was like, she is really fun, I didn't know where I needed to go but already, I know where to go but that time she talked to me (both of them). Of course, she explained to me like when I came, she was very concerned, she was next to me when they standing next to my bed when they researched about whether it is	Friendly and supportive health providers Supportive doctor showed concerns
All the doctors, nurses, they have like become more concerned than my friends as I told you. From the first step when I started here, I see faces that are smiling saying you will be ok. I do remember, the first chemo, there was Syrian nurse, she did my first chemo. I cried; I don't know how to explained. I felt there is something in body, I don't know, I felt I won't wake up. She was standing right next to me and said why are you crying,	Expressing gratitude to health providers/government
are you missing someone. Everything will be OK. I really appreciate SEHA or whoever created this program for expatriate to get this treatment because for example, what about me. You know I am living here for quite long and job is here like my life is here. For example, if I didn't get the treatment here. I really	Suggesting enhanced coverage Feeling happy about the care received







need. I think like it must be next to the doctor like psychologist	
PARTICIPANT 3- EXPAT	
Interview	Exploratory comments
I: What were your thoughts and feelings when you were told you had breast cancer?	Cancer was a shocking news
 P: When I heard the first time, I was shocked, sure it was a shock for me but later I said its ok Alhamdulillah, I will pray and I told my god I agree for everything I 	Considered as from God and accepted it
have so my sister brought my mother. This gave me strength. Also, my sister is with me, my husbandNO, he is X. he didn't stand up with me. For gave myself the	Mother and sister were supportive
strength, my mother, my sister is with me.	Didn't receive support from husband
 I: What physical changes or difficulties you faced? P: Everything changed for me hair, body, everything. had also lots of difficulty with my stomach, my hand, 	Everything changed after cancer
my mouth but after sessions I went to work. Because my manager she told me, you can sit no problem; your salary is OK but for me I said, No, I should go to work	Faced difficulty with stomach, mouth and hand Supportive co-workers
because I don't want to think too much and if I will sit at home and I will think too much and I will feel I am weak. I didn't want to feel this.	Didn't want to feel weak so tried to engaged herself with work
Also, I can see from my mother's eyes, she is very sad about me. For me, I will show her, I am good, I am strong.	Had to be strong for mother as she was sad for her Life changed after treatment



After the treatment, everything changed for me. Also,	Hand discolouration due to
my hand got too much black and my colleague told me Shaima why your hand is like this, you want to wash your hand and I told them No, its from the treatment.	reatment Showed positivism and hope
My another colleague told me why you told her about it and I said no problem, its time and InshAllah it will pass.	Tiredness after chemo
On the first session, I was tired for sure but you know, I am driving from Abu Dhabi to Al Ain	Hospital very far from
and its considerably very far for a patient	residence
when I have session and after that my mother sat with me for nine months. I come along with my mother and after that,	Support from mother
I took my treatment, it was concentrated chemo.	Had chemo treatment
After I had session, I drove back to Abu Dhabi to Al Ain. OK, I felt so much pain and tiredness but I did manage.	Pain and tiredness after treatment
On the first session, after three days, I was little bit OK but because I had so much problem with my stomach, I	Stomach pain, vomiting and hairfall after treatment
for me I would put make up at least. And also, my family would see me that I am OK at least.	wearing makeup to show people I am OK
I: Did you face any difficulties in obtaining or getting information about your disease or care?	
P: On the session, they gave me some medication and I slept. I was not informed I feel but after that yes, sometimes I went to emergency, three times because I	Was informed about the treatment after effects
had too much headache and vomiting. The nurse gave me a book and told me if you have fever you should go	Had often vomiting and headache after treatment







 I: Did you have any spiritual or religious concerns while having disease? P: Alhamdullila, I got more closer to Allah. From the beginning, I accepted, I prayed and I told my Allah, I agree with all I have. 	Accepted cancer as it is God's decision
I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other)	Main concerns were physical and emotional
P: I would say physical and emotional, two. I prefer if I receive more support from health providers about the physical issues I have. May be doctors can also provide emotional support.	Anticipate further support from doctors for physical and emotional domains Had lots of pain and physical issues
I have too much pain and you know I work in sales and I should stand for hours but hamdullila I try to manage not that I can manage but I try to manage every day. It's difficult to accept, also I told you about the medicine, the hormonal treatment, I didn't feel anything. Its difficult for me	It was difficult to accept Hormone treatment didn't help Tried to strengthen herself and start living
I didn't think too much about depression and I tell to myself, now I am taking treatment, lets live now. I will put tomorrow up to God. Because I accepted all from the beginning. I had surgery, I removed my breast and its difficult for	For future, leave it to God Had breast removal, it was difficult decision
I: How did you manage to deal with them or from where you received the support?	



P: I got the main support from my God Alhamdullila after that my mother.	Got the strength from God and <mark>mother</mark>
I: What do you think and what elements of the service delivery and care at the hospital you were satisfied or not satisfied with?	Support from God
 P: Hamdullila, it is from my God, I had too much support here. You know before I came here, maybe I took two months from Abu Dhabi to Sharjah to Dubai to get the right treatment. After I came here, hamdullila from my God, I feel I am comfortable and I am friends with everybody here. 	Satisfied with care
I: What change or improvement you would expect if possible or what kind of support you need? (prompt: what comes to your mind, what do you want in terms of support that can make you feel better or to improve your experience). P: May be if doctors can provide emotional support	Anticipate emotional support from doctors
PARTICIPANT 4 - EXPAT	
Interview	Exploratory comments
I: What were your thoughts and feelings when you were told you had breast cancer?P: In Oct 2018, I was diagnosed with breast cancer with stage II cancer. When I heard the news, I didn't feel bad, I was positive and I felt normal.	It was normal, didn't feel bad when she heard had cancer
I: What physical changes or difficulties you faced?	Received chemo



P: First I received the chemotherapy. Thad lots of side effects mainly vomit but I managed. After chemotherapy, I started maintenance for Herceptin for two months. On March 11 th , I did my surgery. After surgery, I had radiation on my right Breast. In 2019, I took Femara for hormonal treatment until now. I have had mild pain in my knee, not too much and I have become fatty. Before I was sixty-five kilos and now, I am seventy-two. I have no problems as such in daily tasks or daily living, its normal.	Lots of side effect mainly vomitingHad hormonal therapy, had surgery and radiation tooMild painWeight gainDaily life wasn't altered, normal daily living
I: Did you face any difficulties in obtaining or getting information about your disease or care? P: yes, I was given all sorts of information and I had no issues with that.	No problems in obtaining in formation
I: Did you have any social or emotional concerns with the either family, friend or relatives? P: I didn't find any emotional or social issues. My family has been very supportive and they supported me all.	No emotional or social concerns Supportive family
I: Did you have any financial concerns during the disease management? P: I had no problems financially because my treatment is covered by Daman (insurance). I don't work, I manage my four children.	Had insurance coverage No financial problems



I: Did you have any spiritual or religious concerns while having disease?P: no, I rather got closer to God and my religion.	Got closer to God and religion
I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical, information, social, emotional, spiritual or any other) P: I would say none was challenging for me, my life didn't change much. I am happy with my family and I managed all.	Life didn't change much <mark>Supportive family</mark>
I: How did you manage to deal with them or from where you received the support?P: My God and my family really supported me.	Support from God and family
 I: What do you think and what elements of the service delivery and care at the hospital you were satisfied or not satisfied with? P: I was fully satisfied with care I received. For me its all good, the service Tawam hospital provided. Today I am a survivor because of all the doctors and staff at Tawam Hospital Oncology department. 	Satisfied with care received at hospital
I: What change or improvement you would expect if possible or what kind of support you need? (prompt: what comes to your mind, what do you want in terms of support that can make you feel better or to improve your experience).	Suggested enhanced coverage



P: If hospital can provide enhanced coverage for those who cannot afford the treatment, this would be very helpful. But it is all good and I think they are doing all fine.	
PARTICIPANT 5- EXPAT	
Interview	Exploratory comments
I: What were your thoughts and feelings when you were told you had breast cancer?	
P: I was scared, I was shocked, I was shocked really.The doctor told me that I need to remove my breast; I had no other choice.	Was scared and cancer was a shock
There was a shock in the beginning when doctor told me to remove my breast completely but you know my family, they were there for me.	Supportive family
Look, there were some women/patients out there, they were waiting out there; they also had an appointment with the doctor so I told her I need five minutes I need to stay alone. I went there and there were three of them; they also did the same surgery; they removed the breast. So, they told me that, its fine, its so easy. It won't take like more than three day.	
I: What physical changes or difficulties you faced? P: There were no side effects. I didn't take any chemo; I just had my surgery done. There was pain but after surgery two to three weeks, everything was fine. It was so easy, no difficulty like I didn't do anything.	Had no adverse effects from drug Had chemo and surgery Had a little pain but after surgery but manageable



I would say it didn't change for me even one percent, I went to my work as same after the surgery. I am doing my life activity as before, nothing changed for me at all. Even like two months, there was little difficulty in activities say for walking, sitting, you know but after that it was ok, nothing changed. Everything was like the past. But there was difficulty like when I saw myself in the mirror, it was a bit psychological or say emotional.	Life didn't change, it is as same as before There were some difficulties after the treatment but later if was all ok Everything is as it was in the past Couldn't see herself in the mirror. It was psychologically disturbing
Treatment that I am taking now; the doctor told me that I have to take the tablets for five years at least. I: Did you face any difficulties in obtaining or getting information about your disease or care? P: The doctor told me everything, every single detail about the treatment, everything to do. All the information I got and the details also.	Have to take tablets for 5 years (currently on hormone therapy) Was fully informed about treatment and everything.
 I: Did you have any social or emotional concerns with the either family, friend or relatives? P: See, it didn't impact me at all. I didn't tell anything about it. I only told my daughter, my son and that's it. 	Cancer didn't impact her life Didn't tell anyone about cancer only daughter and son Didn't tell anyone, kept it to herself



I didn't tell anyone about it because you know, they would be worried about me, I didn't want that so I kept it to myself. And you know when people about it they will be shocked; oh, how would you do it, it is dangerous. I didn't want that negativity. My husband passed away seven year ago but I got all the support from my family. I can't believe it has been one year and half since I did it, it was so easy alhamdullila. The best thing that I did it alhamdulliah because if I didn't do it, I would be suffering from it for. If I didn't remove it, it may have gone to other place and I didn't know what	People could be shocked and it could be negative Husband died but has full supportive family Had no problems due to cancer even after a year Made a timely decision to be treated Pleased with herself that she did
I don't regret anything. I: Did you have any financial concerns during the disease management? P: I had my insurance and it was totally fine. I had no problems anything alhamdulilla.	the surgery, has no regrets Had no insurance problems
I: Did you have any spiritual or religious concerns while having disease?P: I believe if anything happens to anyone, it is for a reason so Allah he knows what is better for you so it was totally fine. It made me closer to God.	Believe in Allah that everything has reasons Became close to God
I: Can you tell me what were your main concerns or biggest challenges; (prompt: focusing physical,	Main concern was physical



information, social, emotional, spiritual or any	
other)	
P: May be the physical concerns were more challenging	
in term of like surgery made something happened to me,	
it was dangerous to my life, I thought like this. This is	
the only thing. Because I am old so I thought surgery is	
dangerous for me, may be anaesthesia is not good for	
meonly that.	
I: How did you manage to deal with them or from	Supportive kids
where you received the support?	
P: My daughter and my son, really. Maybe they	
encouraged me so much to do. They were more scared	
about it; I was like I will do it anyways but they were	
scared about it.	
I. What do you think and what alamants of the	Sotiefied with core received
1. What do you think and what elements of the	Satisfied with care received
service derivery and care at the hospital you were	
satisfied or not satisfied with?	
P: The care I received was perfect. I was satisfied with	
everything. I got the care that I need and even more	
really	
I: What change or improvement you would expect if	
possible or what kind of support you need? (prompt:	
what comes to your mind, what do you want in terms	
of support that can make you feel better or to	Decided not to take
improve your experience).	chemotherapy or radiation
P: you know, if I had to choose the chemotherapy, I	Made a choice to get hormonal
P: you know, if I had to choose the chemotherapy, I would take the tablet not chemotherapy so it was good	<mark>Made a choice to get hormonal</mark> treatment



fine with me. It was just the best that I didn't do these								nese
two	treatments.	Ι	am	satisfied	that	Ι	am	on
table	tablets/hormonal treatment, it is easier.							



APPENDIX 11b: SELECTED FIELD NOTES FROM PARTICIPANTS INTERVIEWS

SITE: Breast	EXPATRIATES									
Centre	02 M	4.3.4 2020								
Interview date	02 Mar 2020	4 Mar 2020	^{3¹⁴} Mar 2020	29 Jul 2020	4 Sep 2020					
Participant ID	Participant 1	Participant 2	Participant 3	Participant 4	Participant 5					
Interview Questions	What were your thoughts and feelings when you were told you had breast cancer?	Can you tell me what your main concerns or biggest challenges were; (prompt: focusing physical, information, social, emotional, spiritual or any other?	What physical changes or difficulties you faced?	Did you have any financial concerns during the disease management?	How did you manage to deal with them or from where you received the support?					
Field Notes	Sounds recalling with lots of pauses yet very decisive about answers. She provides about as much detail on her experience on her own, her family, work and family is a	Very positive women, lots of clarity in her thoughts but upon asking about the biggest challenge seems like she has been upset with friend who left her during crisis.	Looks very occupied and struggling due to cancer and trying to explain how tired she is due to cancer and long- distance drive to visit her doctor. Sound relief while talking about her employment and	Sounds positive and hopeful. Talks lots of positive, using Gods name a lot. Lots of appreciation toward family and God.	Raised hands for dua (prayer) as gesture of gratitude toward God. Very calm and collective. Looks quite satisfied and answering positively to most questions					



	the main worry for her being a		supportive workplace		
	mother who had relied on her.				
SITE: Breast Cancer		L	EMIRATIS		
Centre					
Interview	29 Jul 2020	15 Sep 2020	27 Sep 2020	9 Nov 2020	4 th Oct 2020
date					
Participant	Participant	Participant	Participant	Participant 4	Participant
ID	1	2	3		5
Field Notes	Using Allahs name with lots of gratitude. I sensed that 'religion' is a big part of her life. She is firm in beliefs that God's been with her through the difficult times	Answered expressing painful memories. Raising hands closing eyes, nodding head, all are signs.	She lost her son, and it seems that this is the heaviest than her disease. Looks low and tired but normal. She paused several times and I sense she is still in grief of her son's death. I should not explore the question where she is quite and getting sensitive.	N/A (praising God)	Happy laughing and very engaged. Smiling and positive to most answers with lots of gratitude and using Allah's name. Most of her support came from her family.



APPENDIX 11c: CODING & THEMES

CODING OF INTERVIEWS (COLOUR CODING)

Theme 1: Living experience with Breast Cancer

Sub-theme 1: Diagnosis of Cancer

Sub-theme 2: Living with cancer

Sub-theme 3: Difficulty in access to health care

Sub-theme 4: Health decision-making

Theme 2: Breast Cancer Survivors' Concerns

Sub-theme 1: Physical concerns

Sub-theme 2: Information concerns

Sub-theme 3: Social & Financial concerns

Sub-theme 4: Emotional concerns

Sub-theme 5: Spiritual/Religious concerns

Theme 3: Expectations or support needed

Theme 1: Living experience with Breast Cancer										
Sub-theme 1: Diagnosis of Cancer										
(Emir ati) Partic ipant 1	(Emir ati) Partici pant 2	(Emi rati) Parti cipan t 3	(Emir ati) Partic ipant 4	(Emir ati) Partici pant 5	(Expat) Partici pant 6	(Expat) Partic ipant 7	(Expat) Partic ipant 8	(Expat) Partic ipant 9	(Expat) Partici pant 10	
It was by sudde n	somet hing differe	When I was told I had	I was surpri sed but it	I don't know, I was	I was shocke d and devasta	It was a compl ete	When I heard the first	In Oct 2018, I was diagno	I was shocke d, I was shocke	



University of Salford MANCHESTER 642

nt I	cance	was	expecti	ted to	surpris	time. I	sed	d
feel it	r: I	mv	ng that.	hear	e for	was	with	really.
100110	was	destin	ing tillati	that I	me	shocke	breast	The
	sort	V	I felt	have	that	d sure	cancer	doctor
	of	y	that I		unat such o	it was	with	told ma
	ohooly	it's	was	cancer	such a	n was		that I
	SHOCK	from	having		tning	a	stage	that I
	ed.	Allah	a		happe	shock		need to
	Like I	and it	breast		ned to	for me	cancer	remove
	pause	took	cancer		me.	but		my
	d	it	soon			later I	When	breast;
		10 096W	50011.			said its	I heard	I had no
		casy.	Doctor			ok	the	other
			was			Alham	news,	choice.
			hesitati			dulilla	1	
			ng to			h, I	didn't	
			tell me			will	feel	
			and I			pray	bad, I	
			told				was	
			her I				positiv	
			am				e and I	
			having				felt	
			having				norma	
			breast				1	
			cancer					
			and she					
			said					
			yes.					
			I inst					
			felt_I					
			having					
			navnig					
			a					



			breast cancer. My sister had a breast cancer and she passed away but it doesn't relate to the fact that becaus e my sister had it.			
somet	Не	It was	It was			There
hing is	(Docto	not	sense			was a
not	r) said	what	of			shock
norma	directl	I was	denial			in the
1	y that	expec				beginni
	it's no	ting				ng
	good	or I				when
		thoug				doctor
		ht 1				told me
		woul				to
		d				remove



ever			my
face			breast
but of			comple
cours			tely
e I			
later			
accep			
ted as			
part			
of the			
realit			
y. I			
was			
not			
angry			
or sad			
but			
shock			
ed			
you			
can			
say as			
it hit			
me			
from			
nowh			
ere,			
somet			
hing			
unex			
pecte			
d that			



[
		you				
		never				
		thoug				
		ht of				
		it.				
somet	they	I was	I never			
hing	discov	diaon	though			
hard	ered	osed	tilougii			
liko o	that I	and	t musalf			
like a	tilat I		haing			
bone	nave a	then	being			
in my	cancer	later,	in that			
breast	type	I was	situatio			
	(stage)	diagn	n			
	П	osed				
		for				
		anoth				
		er				
		cance				
		r on				
		other				
		side				
She						
(Docto						
r) was						
not						
hanny						
and						
was						
worrie						
d						



She					
(Docto					
r)					
wasn't					
sure					
and					
wasn't					
quite					
happy					
with					
the					
conditi					
on of					
my					
breast					
They					
told					
me					
that					
you					
have					
this					
breast					
cancer					



Sub-theme 2: Living with difficulties										
Sub-the	eme 2: Li I have been alright during and after the treatm ents hamdu llila.	ving wit I felt my life chang ed to slow and I beca me thoug htful and consi derat e towar derat i derat y than mysel	h difficu I had difficu Ities but I was able to manag e my daily life and tasks as norma I.	Ities	I had no idea, it is gonna be that tough	. If you compa re me one and half year back and now it is differe nt	After the treatm ent, everyt hing change d for me	: I would say none was challe nging for me, my life didn't chang e much	I would say it didn't change for me even one percent , I went to my work as same after the surgery . I am doing my life activity as before, nothing	
		famil y than mysel f.							activity as before, nothing change d for me at all	
It was challe nging		I just accep ted		They just gave	My life has change	When I had the		I have no proble	Everyth ing was	

Г



to deal	that it	me	d	treatm		ms as	like the
with	happe	sick	includi	ent, I		such	past.
the	ned,	leave	ng me,	unders		in	
proble	what	for just	my	tood		daily	
ms	else I	six	appeara	it's my		tasks	
	could	months	nce, my	life		or	
	do.	and	feeling	now		daily	
		then	s, my			living,	
		they let	prioriti			its	
		me go	es.			norma	
		and	Everyt			1.	
		said I	hing				
		am fit	has				
		but I	change				
		was	d to the				
		not fit	extent				
			that its				
			all just				
			new. It				
			has				
			change				
			d my				
			life				
			comple				
			tely.				
The	T/		M		F (τ
Inere	it was		My		Everyt		i can't
were	sure				ning		beneve
some	not		or life		cnange		it has
aifficu	easy		was		a for		been
Ities .	and		just, I		me		one
especi	somet		Just		hair,		year


ally	imes		lived	body,	and
throug	pain		for	everyt	half
hout	and		weeken	hing	since I
the	other		d so I		did it, it
mecha	distur		could		was so
nical	bance		just		easy
treatm	s		stay in		alhamd
ent.	were		bed. I		ullila
	too		just did		
	much		the		
	for		most		
	me to		essenti		
	handl		al		
	e but		groceri		
	Ι		es in		
	mana		my		
	ged		house		
	some		and put		
	how		like		
			noodles		
			, easy		
			things		
			for my		
			daught		
			er to		
			make		
			for		
			herself		
<u> </u>					
Now I	There		Your		
am on	was		life just		



the	no as	will not		
hormo	such	be the		
nal	diffic	same		
treatm	ulty	again.		
ent.		For		
You		exampl		
know		e, when		
that it		I		
lasts		recover		
for		ed, fifty		
five		percent		
years		of the		
		hair has		
		grown		
		back.		
		One		
		person		
		said,		
		after		
		one		
		year of		
		chemo,		
		she		
		cannot		
		still		
		remem		
		ber		
		how		
		she		
		coped		
		till		
		today.		



			She had		
			to rely		
			on		
			people		
			to tell		
			her oh,		
			this is		
			what		
			happen		
			ed and		
			so.		
There		They	Its such		
were		said,	a big		
difficu		you are	thing,		
lties		fine	its like		
but I		and	a		
manag		this is	paradig		
ed all		normal	m shift,		
hamdu			its		
llila		What's	never		
		normal	the		
		?	same.		
		Norma	Like		
		l is	before,		
		when	I would		
		people	like, oh		
		can do	look at		
		and	this		
		practic	line,		
		e daily	you		
		life.	know		



		They	that		
		can go	was the		
		and	main		
		come	concer		
		as they	n,		
		wish.	health		
		This is	wise. It		
		not	was so		
		normal	shallow		
		1	; you		
			got		
			other		
			things		
			now. I		
			couldn'		
			t be		
			bothere		
			d		
			anymor		
			e		
		These	At this		
		They	At this		
		said,			
		you are	my life,		
		fine	as you		
		and	can see,		
		this is	even		
		normal	the		
		·	small		
		What's	thing		
		normal	will		
		?	affect		



		Norma	your		
		l is	life.		
		when	You		
		people	know		
		can do	what I		
		and	mean.		
		practic	It		
		e daily	effects		
		life.	how		
		They	you		
		can go	dress.		
		and	Like		
		come	inside		
		as they	my		
		wish.	apartm		
		This is	ent, I		
		not	would		
		normal	go		
		!	without		
			my		
			Sheila		
			(scarf)		
			but		
			even in		
			the		
			balcon		
			y, I am		
			gonna		
			cover		
			myself		
			becaus		
			e I just		



			don't		
			want		
			everyb		
			ody to		
			see. I		
			have		
			become		
			more		
			comfor		
			table		
			not		
			having		
			breast		
			but it's		
			difficul		
			t		
			Every		
			aspect		
			of your		
			life		
			change		
			s		
			drastica		
			lly		
		Until	It was		
		now	n was		
		now,	faco		
		you know I	Pacer		
		KHOW I	becaus		
		cannot			
		wear	like l		
		the bra.	would		



		It's	have		
		very	convers		
		hard	ation		
		for me	with		
		to wear	my		
		the bra	garden		
		becaus	er; I		
		e when	can't		
		I wear	do this		
		the bra,	anymor		
		my	e. This		
		pain	is too		
		increas	difficul		
		es. For	t.		
		exampl			
		e, until			
		now			
		when			
		my			
		nieces			
		and			
		nephe			
		ws			
		come			
		and			
		hug			
		me, my			
		breast			
		is like			
		someb			
		ody hit			
		me or			



		someth			
		ing like			
		this			
		slightl			
		y and I			
		feel			
		pain			
		C+:11 I	Decours		
		Sull I	Decaus		
		am	e my		
		sufferi	view of		
		ng that	myself		
		I have	change		
		to go to	d and it		
		work	was		
		and I	going		
		still	to		
		don't	change		
		have a	immen		
		reason	sely		
		that	after		
		could	and		
		stop	coming		
		me to	to		
		go the	terms		
		work	to my		
			new		
			body, I		
			can tell		
			you at		
			this		
			point		



			that I		
			tilat I		
			do not		
			have		
			partner		
			in mv		
			life but		
			this is		
			gonna		
			impact		
			me		
			having		
			naving		
			one		
			again.		
		I told			
		my			
		doctor			
		this			
		and			
		she			
		anid			
		saiu,			
		you			
		are			
		just			
		saying			
		of			
		course			
		t			
		1 am			
		saying,			
		I am			
		not			
		doing			
		8			



		somet			
		hing			
		from			
		my			
		mind			
		that I			
		am			
		sick or			
		somet			
		hing.			
		8			
		Doctor			
		s have			
		to			
		believe			
		what			
		patien			
		ts is			
		saying.			
		I am			
		not			
		creati			
		ng			
		these			
		things.			
		How			
		could			
		you			
		know			
		that			
		that I			
		am			



				having			
				this or			
				I don't			
				have.			
				You			
				won't			
				see			
				inside			
				me.			
Sub-the	eme 3: Di	fficulty	in access	to health	icare		
I am	they			<mark>.11</mark>	When	VOU	
i ann	did			doctors		know	
for	wrong			are like	first	L om	
from	ononoti			toko	diagnas	1 an	
	operati				ulagnos		
					ea, n		
nospit	me,			e it les	you	Abu	
	wallan			1tno	ѕреак		
A1n)	•			w can I	to a	to Al	
				take it?	doctor	Aın	
				Even I	in her	and its	
				asked	office,	consid	
				them,	her	erably	
				did you	time is	very	
				have a	gonna	far for	
				patient	be	a	
				like	limited	patient	
				me,	becaus		
				this (e there		



			happen ed and they said let's	will be more patients waiting outside		
			ury. What's let's try	r ner room,		
Somet ime I cannot reach to the doctor	They didn't remov e the lymph node under my arm					
	my doctor, Dr. Riyad was so much angry becaus e its wrong operati on and they					



	should								
	have								
	remov								
	ed								
	evervt								
	hing								
Sub-the	me <u>4.</u> Ha	ealth dea	rision-m	aking					
Sub-th				aning					
I went	Then I	When	<mark>I had</mark>	<mark>I did</mark>	<mark>I put</mark>	I did	I took	First I	<mark>I didn't</mark>
to Al-	<mark>said, it</mark>	<mark>I was</mark>	<mark>chemo</mark>	the	myself	12	my	receiv	<mark>take</mark>
Zahra	happe	<mark>takin</mark>	therap	<mark>radiati</mark>	<mark>into</mark>	chemo	treatm	ed the	<mark>any</mark>
hospit	ned	g care	y	on	<mark>forum</mark>	then	ent, it	chemo	<mark>chemo;</mark>
al in	<mark>but it</mark>	<mark>of</mark>			mainly	now 6	was	therap	<mark>I just</mark>
Sharja	<mark>will go</mark>	<mark>him, I</mark>			becaus	cycles	concen	У	<mark>had my</mark>
before		<mark>knew</mark>			e of the	of	trated		surgery
comin		<mark>I had</mark>			effects	chemo	chemo		<mark>done</mark> .
g to		a			<mark>chemo</mark>	tablet			
Tawa		<mark>lump</mark>			<mark>has had</mark>	and			
		<mark>but</mark>			<mark>on me.</mark>	tomorr			
		when				ow			
		he				will be			
		passe				last			
		d				day of			
		away				tablet.			
		after							
		that I							
		decid							
		<mark>ed to</mark>							
		go							
		and							



		see						
		<mark>the</mark>						
		docto						
		<mark>r</mark> .						
Ι	<mark>But it</mark>	Befor	<mark>I was</mark>	<mark>I had</mark>	I	I	After	Treatm
decide	took	e that	taking	the	associa	change	chemo	ent that
d to go	<mark>time</mark>	<mark>I was</mark>	<mark>her</mark>	surger	ted	<mark>d my</mark>	therap	<mark>I am</mark>
to	and	ignori	<mark>chemo</mark>	y	myself	treatm	y, I	<mark>taking</mark>
Tawa	there	ng			with	ent	started	now;
m	was	<mark>my</mark>			<mark>the</mark>	and	mainte	<mark>the</mark>
hospit	weddi	<mark>condi</mark>			<mark>forum</mark>	when I	nance	doctor
al to	<mark>ng of</mark>	tion				<mark>change</mark>	for	told me
compl	<mark>my</mark>					<mark>d my</mark>	Herce	<mark>that I</mark>
ete my	sister					treatm	ptin	have to
treatm	and and					ent, I	for	take the
ent.	then I					<mark>took</mark>	two	tablets
	<mark>said</mark>					<mark>radiati</mark>	month	for five
	not					<mark>on and</mark>	s. On	<mark>years at</mark>
	now					<mark>my</mark>	March	<mark>least</mark> .
	till my					treatm	11 th , I	
	sister's					ent	did my	
	weddi					togeth	surger	
	ng is					er.	у.	
	over.						After	
	<mark>You</mark>						surger	
	<mark>know</mark>						у, І	
	then (had	
	after						radiati	
	<mark>that I</mark>						on on	
	went						my	



	to see the doctor here in Sharja h							right Breast	
but now I	when I went		I had surger	I was taking	I had do <u>uble</u>	I did	I had surger	In 2019, I	<mark>The</mark> best
stoppe	there		v	this	mastect	chemo	v. I	took	thing
d	to see			hormo	omy	then	remov	Femar	that I
chemo	the			ne	and my	now 6	ed my	a for	did it
	doctor			therap	breast	cycles	<mark>breast</mark>	hormo	<mark>alhamd</mark>
	withou			y	were	of	and its	nal	ulliah
	<mark>t an</mark>				remove	chemo	<mark>difficu</mark>	treatm	because
	<mark>appoin</mark>				d. After	tablet	<mark>lt for</mark>	ent	if I
	tment				that I	and	<mark>me but</mark>	until	didn't
	and				had	tomorr	I	now	do it, I
	though				chemot	ow	accept		would
	t they				herapy	will be	ed so I		be
	will				and	last	agree		sufferin
	see me				hormon	day of	tor		g from
	they				a1	lablet	hing		<mark>IT 101</mark> .
	will						mig.		
	sav								
	nothin								
	g and I								
		1							
	will								
	will return								
I	will return			We					: vou



chemoon andi.e. <th< th=""></th<>
IndiaitIndia indiaIndia indiaIndia
y, IononininininfinisheIIIIIIIIIdIIIIIIIIIIIIsurgerII
finisheImage: Some stat
dimageimageimageimageimageimagesurgerimageimageimageimageimageimagey, Iimageimageimageimageimageimagefinisheimageimageimageimageimageimagefinisheimageimageimageimageimageimagedimageimageimageimageimageimagedimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimageimageimageimageimageimagenadaimag
surgerididn'tafterininI wouldy, Icomplecompleinitititfinisheininititinititd theinininininitititradiatiininininininitinon andininininininininnow Iininininininininam onininininininininhormoininininininininnalin <td< td=""></td<>
y, Iiicompleiake thefinisheiieiiake thed theiieiiiabletradiatiiiiiiion andiiiiiiinow Iiiiiiiiam oniiiiiiiherapyiiiiiiinow Iiiiiiiiam oniiiiiiiiherapyiiiiiiiiinaliiiiiiiiiinaliiiiiiiiiiinaliiiiiiiiiiiiinaliii
finisheImage: set of the set o
d theII
radiatiImage: selection of the s
on andImage: selection of the se
now III
am onimage: state
theImage: second se
hormoImage: state
nal Image: state sta
treatm ent find find find find find find find find
ent chemo or
radiatio
n so it
was
just
fine fine
with
me. It
was
just the
best
that I
didn't



						<mark>do</mark>
						<mark>these</mark>
						<mark>two</mark>
						treatme
						nts. I
						am
						satisfie
						d that I
						am on
						tablets/
						ai
						treatme
						nt, it is
						easier.
And I	I went		L had a	So. if		
got the	to		radiati	there is		
bionsy	Hurrya		on	like a		
near	kaabi		surger	forum		
here	$\frac{Kaa01}{2}$		v and I	if there		
mene						
	very					
	good					
	doctor.		normo	wno		
	Sne		ne	nave		
	did the		therap	gone		
	mamm		y then I	through		
	ogram		stoppe	the		
	again		<mark>d it</mark>	same		
	for me			things,		
	and			they		
	scanni			<mark>can</mark>		



ng and relate	
everyt There	
hing should	
be a	
safe	
place to	
speak	
about	
she When	
(Docto	
r) said	
its	
better	
to	
remov if they	
e all can	
and the	
even if	
you tell, tell,	
don't listen	
have a this is	
cancer what	
in that, you	
remov need to	
e it look for	
and this	
is how	



	gonna		
	feel		
l went	There		
to	was a		
Tawa	lady		
<mark>m to</mark>	and she		
see Dr.	was on		
Mouza	her first		
and	session		
she	when I		
made	was on		
the	my 3rd		
surger	or 4th		
<mark>y for</mark>	session		
me	, she		
	came		
	from		
	Fujaira		
	h to		
	Tawam		
	. She		
	asked		
	mehow		
	does it		
	feel		
	now		
	where		
	you are		
	and		
	what		
	you		



		need to			
		look for			
		and <mark>I</mark>			
		told her			
		listen			
		look for			
		this and			
		this and			
		this and			
		then (
		after			
		<mark>couple</mark>			
		of			
		session			
		<mark>s down</mark>			
		<mark>the</mark>			
		line, we			
		landed			
		up			
		togethe			
		<mark>r in the</mark>			
		<mark>same</mark>			
		room			
		and she			
		said the			
		<mark>same</mark>			
		thing			
		with			
		her.			
		She			
		said she			
		<mark>did on</mark>			
			1		



			1	1	1
		Wedne			
		<mark>sday.</mark>			
		<mark>She</mark>			
		<mark>said</mark>			
		<mark>from</mark>			
		<mark>the</mark>			
		<mark>Sunday</mark>			
		to			
		Thursd			
		ay she			
		couldn'			
		<mark>t do</mark>			
		anythin			
		<mark>g so</mark>			
		what I			
		<mark>had</mark>			
		told her			
		<mark>so</mark>			
		basicall			
		y			
		prepare			
		d her			
		for that			
		so we			
		need			
		that.			
I said					
remov					
e both					
(breast					
). <mark>She</mark>					



	said								
	<mark>you</mark>								
	<mark>don't</mark>								
	<mark>have a</mark>								
	cancer								
	in								
	other								
	one								
	<mark>and I</mark>								
	<mark>said,</mark>								
	<mark>it's</mark>								
	better								
	for me								
	<mark>to do it</mark>								
	<mark>both</mark>								
	T								
	didn't								
	take								
	chemo								
	therap								
	v. only								
	radiati								
	on								
Theme 2: Breast Cancer Survivors' Concerns									
Sub-the	Sub-theme 1: Physical concerns								
The	I had	Pain	my	Pain	That	Mainl	Everyt	I had	There
side	somet	was	eye-	was the	was so	У	hing	lots of	were no
effects	hing	the	sight	most of	like	fatigue	change	side	side
were	under	main	dropp	difficul	today I	was	d for	effects	effects.
	my		ed and	t	am	the	me	mainly	



on my	arm,	probl	Ι	going	bigges	hair,	vomit	
skin	like	em	could	to	t	body,	but I	
	small		n't see	bathroo	challe	everyt	manag	
	pimple		partic	m and	nge. I	hing. I	ed.	
	and		ularly	after	didn't	had		
	also		at	that I	have	also		
	my		night.	couldn'	any	lots of		
	breast		Mainl	t go	other	difficu		
	is not		y the	back to	proble	lty		
	equal		left	it and it	ms	with		
	like		eye.	was	from	my		
	my		At	escalat	any	stomac		
	other		night I	ed so	treatm	h, my		
	one		had all	much	ent. I	hand,		
			blurry	so first	was	my		
			vision	time, I	very	mouth		
				was	lucky	but		
				just		after		
				float.		sessio		
				My		ns I		
				whole		went		
				face		to		
				showed		work.		
				it. I was				
				just in				
				my				
				bed; I				
				couldn'				
				t do				
				anythin				
				g just				
				couldn'				



University of Salford	
6	77
U	12

					t do			
					anythin			
					g.			
1 1	T 1 1	T	T /	XX 71 1	T 1.1		т 1	701
darkne	I had	1 was	1 got	whole	1 did	my	I have	Inere
ss on	no	havin	pain	night I	start	hand	had	was
my	issues	g	in my	couldn	my	got too	mild	pain
face,	with	sleep	bones	't sleep	chemo	much	pain in	but
on my	my	distur	and		on	black	my	after
arms,	physic	bance	body		Wedne	and	knee,	surgery
on my	al				sday	my	not too	two to
neck	appear				and it	collea	much	three
	ance				was	gue	and I	weeks,
					only on	told	have	everyth
					Sunday	me	becom	ing was
					when I	Shaim	e fatty.	fine. It
					woke	a why	Before	was so
					up, I	your	I was	easy,
					started	hand is	sixty-	no
					vomitin	like	five	difficul
					g and	this,	kilos	ty like I
					diarrho	you	and	didn't
					ea and	want	now, I	do
					my	to	am	anythin
					stomac	wash	sevent	g.
					h was	your	y-two.	
					so	hand		
					torture	and I		
					d	told		
						them		
						No, its		
						from		



DIC	
STER	
	673

					the	
					treatm	
					ent.	
) I		T had	Dein in	A ft and T	1:1
my) 1	my	I had	Pain in	After I	like
appetit	don't	hair	this	terms	had .	two
e was	care	fell	period	of I do	sessio	months
very	this is		for	not take	n, I	, there
bad, I	the last		long	the	drove	was
couldn	thing I		time	tablet,	back	little
't eat	care			the	to Abu	difficul
	about.			yellow	Dhabi	ty in
				tablet, I	to Al	activiti
				do not	Ain.	es say
				take	OK, I	for
				one.	felt so	walkin
				The	much	g,
				intensit	pain	sitting,
				y of	and	you
				pain	tiredne	know
				and	ss but I	but
				cramps	did	after
				;	manag	that it
				having	e. On	was ok,
				all your	the	nothing
				periods	first	change
				and	sessio	d
				childbir	n, after	
				th is	three	
				just	days, I	
				nothing	was	
				like	little	
1				-		



				that	bit OK	
				swear	but	
				to God	becaus	
					e I had	
					so	
					much	
					proble	
					m with	
					my	
					stomac	
					h, I	
					was	
					vomiti	
					ng,	
					also	
					my	
					mouth,	
					for	
					sure	
					my	
					hair	
					fell	
I lost	They	Llost	Pain	other	someti	May be
weight	took	weigh	physic	was	mes I	the
Llost	like	t.	al	fatigue	went	physica
weight	half		concer	I	to	1
around	from		ns for	couldn'	emerg	concern
five	mv		sure.	t leave	ency.	s were
kilos.	breast			mv	three	more
	and the			pillow	times	challen
	doctor			that	becaus	ging in



					wee the	o I had	torm of
	said,				was the	e i nad	
	khalas,				level of	too	like
	you				fatigue.	much	surgery
	are ok					headac	made
						he and	somethi
						vomiti	ng
						ng.	happen
							ed to
							me, it
							was
							danger
							ous to
							my life,
							Ι
							thought
							like this
The main proble m I faced was the outloo k and chang es in my face and		I had a good feed so I didn't lose weig ht	I had pain on the area where I had surger y but it wasn't that much	This was the biggest challen ge becaus e the pain, pain on the chest cannot let you sleep	Inabilit y of sleep, I was just awake. I would sleep one hour and then awake	On the first sessio n, I was tired for sure	



my						
skin		As I				
SKIII		said, I				
		did				
		not				
		have				
		much				
		diffic				
		ulty				
		but of				
		cours				
		e				
		after				
		Surge				
		ry I				
		suffer				
		ed				
		pain.				
	Pain		They	It		
	and		told me	effects		
	feeling		that	vour		
	lack of		pain	sexualit		
	flesh		will go	v.		
	after		away	5		
	operati		after			
	on		one			
			vear. It			
			has			
			been			
			more			
			than			
			one			



			year and I am having that pain, serious ly unbear able pain I was having my periods more than twenty days, so my health becam e very			
			bad			
Mv	The		I am	My hair	someti	
eye	difficu		having	went	mes I	
brows	lt part		flue	off. I	went	
started	was		freque	was	to	
falling	after		ntly.	just	emerg	
during	operati		Most	showin	encv.	
the	on. I		days I	g to my	three	



treatm	had	have	nephew	times	
ent	pain	this	the	becaus	
	and	flue.	picture	e I had	
	someti	There	of my	too	
	me I	is no	pillow	much	
	was	day	after	headac	
	about	when I	my	he and	
	to	don't	second	vomiti	
	vomit.	have	chemo.	ng	
	Ι	conges	And		
	suffere	tive	my		
	d from	nose.	pillow,		
	this	Му	it was		
	part so	immun	just full		
	mainly	e	of hair,		
	physic	system	that		
	al	becam	was the		
		e down	amount		
		and I	of hair I		
		told	lost. I		
		them	just		
		and	went to		
		they	the bath		
		said	room, I		
		everyt	didn't		
		hing is	have		
		fine.	the		
		How? I	shaver,		
		wasn't	I just		
		like	put the		
		this!	Veet all		
			over		



			my		
			head		
There		Whole	And	I had	
were		night I	taste in	some	
no		couldn	mouth,	days, I	
eyebro		't sleep	anythin	couldn	
ws, no		becaus	g I eat,	't	
hair		e of my	it tastes	stand	
		conges	off, I	becaus	
		tive	couldn'	e, I had	
		nose,	t eat	too	
		how	anythin	much	
		could I	g and	vomiti	
		do	whole	ng, I	
		daily	six	took	
		work.	months	water,	
		Also, I	of	Ι	
		am	chemo	vomite	
		having	becaus	d, I	
		asthma	e of	had	
		so it	that	too	
		effects		much	
		my		headac	
		chest		he and	
		so until		I went	
		now I		to	
		am		emerg	
		sufferi		ency	
		ng		may	
		from		be	
		this.			



				three	
				times.	
The			becaus	Physic	
main			e of	al	
proble			side	concer	
ms I			effects	ns	
face			of	were	
with			chemo	the	
Chem			I lost 7	highes	
0			kilos	t.	
treatm					
ent					
mainly					
physic					
al and					
outloo					
k					
chang					
e.					
All the			Physica		
side			1		
effects			concer		
that I			ns and		
faced			emotio		
was in			nal		
the			concer		
chemo			ns were		
			the		
			most		
			most		



University of Salford	
MANCHESTER	
	681
	001

		importa		
		nt		
		issues I		
		had to		
		deal		
		with		
		the		
		cancer.		
		The	I told	
		side	you	
		effects	about	
		of the	the	
		chemo,	medici	
		to use a	ne, the	
		bad	hormo	
		word	nal	
		just	treatm	
		terrible	ent, I	
		it is.	didn't	
		Side	feel	
		effects	anythi	
		are	ng. Its	
		long,	difficu	
		its like,	lt for	
		one	me	
		time I		
		found		
		myself		
		its like		
		crawl		
		to the		



			bathroo		
			m like I		
			have		
			diarrho		
			ea		
				T 1	
My				I have	
major				too	
concer				much	
n was				pain	
physic				and	
al				you	
appear				know I	
ance				work	
like,				in	
no				sales	
hair,				and I	
skin				should	
chang				stand	
e				for	
				hours	
				but	
				hamdu	
				llila I	
				try to	
				manag	
				e not	
				that I	
				can	
				manag	
				e hut I	
				try to	
				uy 10	



bro	
IESTER	683

							manag		
							e		
							every		
							day.		
							-1 T		
							also I		
							tola		
							you		
							about		
							the		
							medici		
							ne, the		
							hormo		
							nal		
							treatm		
							ent, I		
							didn't		
							feel		
							anythi		
							ng. Its		
							difficu		
							lt for		
							me		
							•••		
Sub-theme 2: Information concerns									
There	Dr.	I was	They	they	Nothin	We are	On the	I was	The
is no	Hurrya	fully	were	think I	g	living	sessio	given	doctor
difficu	, she	infor	clear	don't	prepare	in the	n, they	all	told me
lty if	explai	med	and I	have	d me	world	gave	sorts	everyth
you	ned to	and	had no	anythi	for that.	of	me	of	ing,
want	me	multi	difficu	ng so	Yes, I	inform	some	inform	every



to	everyt	ple	lty in	their	prepare	ation.	medic	ation	single
know	hing	times	obtain	answer	d	We	ation	and I	detail
anythi	and	about	ing	s were	myself	can	and I	had no	about
ng	also	the	infor	not	for that	open	slept. I	issues	the
	Dr.	side	matio	conven	but not	the	was	with	treatme
	Mouza	effect	n. I	ient to	to that	google	not	that	nt,
		s of	was	me at	level	and	inform		everyth
	T 1 1	medi	told	all	that	check	ed		ing to
	I had	catio	everyt		will	on			do. All
	no	ns.	hing		float	whate			the
	proble	T			me.	ver we			informa
	ms	I was				want.			tion I
	with	fully							got and
	obtaini	infor							the
	ng	med							details
	inform	of the							also
	ation	symp							
	at all	toms							
		I had.							
		Healt							
		hcare							
		depar							
		tment							
		was							
		callin							
		g me							
		regul							
		arly							
		and							
		askin							
		g me							
		how I							


	am					
	doing					
	, how					
	I am					
	progr					
	essin					
	g					
Inet		Somo	I had	T		
Jusi		Some	I llau	I dida't		
googie		words	read, 1			
and		were	nad	KNOW		
YouT		related	researc	anythi		
ube		to me	hed but	ng and		
and		but	I did	even		
you		others	not	you		
will		not	know	offer		
know		especia	that it	me all		
everyt		lly my	was	the		
hing		health	such a	inform		
		was	fine	ation		
		going	balance	which		
		down	that	is		
		and	you had	enoug		
		they	to run	h for		
		didn't	betwee	me but		
		answer	n	it all		
		me	diarrho	must		
			ea and	be for		
			normal	my		
			bathroo	doctor		
			m			
			needs			



			that			
			needs			
			to be			
			there			
			. that			
			was not			
			there			
Intern		I am	Yes, I	if I		
et		asking	was	know		
helps		them	fully	somet		
me		give	inform	hing, it		
		me	ed. My	will		
		inform	doctor,	change		
		ation.	my	my life		
		Did it	oncolo	but no,		
		happen	gist she	it will		
		before,	explain	not		
		is it	ed to	change		
		someth	me	my		
		ing	what's	life. If		
		natural	going	my		
		ly	to	doctor		
		happen	happen	doesn'		
		ing?	but as I	t know		
		Do you	said I	some		
		have	didn't	inform		
		other	own it.	ation,		
		cases	I didn't	it will		
		or not?	realise	affect		
		So, in	that's	me but		
		that	gonna	if I		



		situatio	be you	don't		
		n, they	that's	know		
		just	gonna	some		
		want to	happen	inform		
		do	to you	ation, I		
		what	like	don't		
		they	other	think it		
		learned	people	will		
				change		
				my		
				life.		
What		look of	I was	Lwoo		
should		inform	1 was	, 1 was		
Should		ation	inform	inform		
and		from	ad	ad		
what		the	about	eu,		
what		destore	the side	even I		
we		thic	offooto	monded		
silouid		, uns	effects	heeded		
n t		arrects.	OI alta ana an	[[0]		
			cnemo;	KNOW		
I hat is			1			
what I			remem			
neede			ber my			
d most			doctor			
			told me			
			all. I			
			was			
			fully			
			inform			
			ed			
			about			



University of Salford	
MANCHESTER	100
	688

		all the			
		sympto			
		ms and			
		side			
		effects.			
		You	I don't		
		know	remem		
		all my	ber but		
		questio	I don't		
		ns were	think I		
		answer	was		
		ed, I	told if		
		have to	I face		
		say that	such		
		but like	and		
		I have	such		
		had	issues,		
		time to	I can		
		think	go to		
		about it	this		
		and I	person		
		have	or that		
		more	person		
		questio	but I		
		ns so	think		
		whose	this is		
		gonna	somet		
		follow	hing		
		that, It	you		
		cannot	need		
		be the			



		doctor		
		becaus		
		e she		
		has		
		other		
		patients		
		to see.		
		1		
		naving		
		my		
		double		
		mastect		
		omy,		
		when		
		do I		
		stop it?		
		Yes, I		
		was		
		told		
		and I		
		remem		
		ber		
		everyth		
		ing		
		what		
		the		
		doctor		
		said at		
		that		
		time		
		but		
		there		



		WOG		
		was		
		time		
		when I		
		was		
		like		
		what do		
		I do		
		now?		
		They		
		go very		
		efficien		
		t in		
		evervth		
		ing but		
		the		
		psychol		
		ogical		
		and		
		emotio		
		nal		
		side		
		siue,		
		nobody		
		said		
		you are		
		gonna		
		be		
		depress		
		ed as		
		afterwa		
		rds		



		If you		
		are		
		gonna		
		see the		
		mirror		
		for the		
		first		
		time,		
		vou are		
		gonna		
		break		
		down		
		in tears		
		no		
		body		
		prepare		
		d vou		
		for that		
		TOI tilat.	 	
		Side		
		effects,		
		yes		
		gonna		
		be there		
		but to		
		what		
		level		
		I		
		cannot		
		say		
		questio		
		ns		



		answer		
		ed but		
		address		
		ed		
		. Tell		
		me		
		these		
		things,		
		I don't		
		want to		
		hear		
		that		
		you are		
		gonna		
		be OK		
		cause		
		you are		
		not		
		going		
		to be		
		OK		
		So, be		
		clear		
		like my		
		doctor		
		said		
		chemo		
		will not		
		but the		
		side		
		effects		



				throw				
				you				
				I was				
				also				
				inform				
				ed of				
				all the				
				service				
				Service				
				s				
				le if				
				needed.				
				once I				
				had to				
				come to				
				ER on				
				Friday				
				evenin				
				g. After				
				9, I had				
				to call				
				my				
				friend,				
				I had				
				fever. I				
				just				
	1	1	1		1	1	1	1



				showed						
				that so I						
				went to						
				him.						
				But it						
				wasn't						
				covere						
				d where						
				to go or						
				whom						
				to						
				speak						
				to if I						
				have						
				any						
				social						
				needs						
				Becaus						
				e I am						
				not						
				from						
				here, I						
				do not						
				know, I						
				can go						
				there or						
				go						
				there						
Sub-the	Sub-theme 3: Social & Financial concerns									



I had no financi al proble ms	No, I had no financi al or social issues	no finan cial probl em	Everyt hing and every one was good and suppo rtive. I had no emoti onal or social or social concer ns hamd	no, I didn't have any financi al proble m becaus e this is all from the govern ment.	Dr. Mouza will tell you the issues I had with NEXtC are which is my provide r from Ministr y of educati on.	I have SEHA manda te. It is really hard for me for examp le in my case;	My insura nce bad, bad, now bad, reatm took treatm ent under manda te becaus e my insura nce doesn' anythi ng.	I had no proble ms financi ally becaus e my treatm ent is covere d by Dama n (insura nce).	I had my insuran ce and it was totally fine. I had no proble ms anythin g alhamd ulilla.
It is covere d	healthc are is covere d by insura nce alham dullila h	my healt hcare was cover ed by insur ance	my work was OK and flexibl e and they knew my condit	:Ididn'thavethesupportexceptpeopleatmywork	Someti mes I would come and the approv al will not go through entirely so	Becau se I do not have a manda te approv ed from Abu Dhabi	, for my surger y I had money proble ms and I couldn 't manag	I don't work, I manag e my four childre n	



ion and	helped me.	couple of	for the gynaec	e the money	
allowe d me		session s I had	ologic al	for surger	
to		<mark>to miss</mark>	cancer	y.	
leave at			. They said if		
twelve			you		
becau			have a		
se I			cancer of		
n't			Breast,		
stay			we can		
as I			help you		
got			with		
tired.			that		
told			cannot		
<mark>me;</mark>			cover		
you can			this		
leave			ologic		
early.			al		
			cancer care		
		when I			
		had			
		started			
		my chemot			



					1
		herapy,			
		the first			
		session			
		had to			
		ha			
		UC			
		postpo			
		ned			
		becaus			
		e they			
		didn't			
		approv			
		en			
		They	it is		
		said go	most		
		salu go	most		
		to the	comfo		
		social	rtable		
		worker	that if I		
		and	get the		
		speak	gynaec		
		to the	ologist		
			horo I		
		social	nere, 1		
		worker	cannot		
		and	pay		
		they (two		
		asked if	thousa (
		I had	nd		
		insuran	dirham		
		oo ond I	c for		
			<u>s 101</u>		
		said	gynaec		
		yes, I	ologist		
		have	. I did		



		insuran	th1s		
		ce and	once		
		they	here		
		said	but		
		Saru			
		then	th1s 1s,		
		<mark>you</mark>	in my		
		<mark>cannot</mark>	job, it		
		have it,	is quite		
		ro to	expens		
			·		
		tne	ive.		
		principl	<mark>So,</mark>		
		e	this is		
			the		
			challe		
			nge I		
			faced		
			here,		
			becaus		
			e I		
			would		
			like to		
			be here		
			but I		
			cannot		
			afford		
			it this		
			amoun		
			t		
			-		
		they	$\mathbf{M}\mathbf{y}$		
		told	gynaec		
		that I	ology		
			ology		



	-	 -				-	
			have to	is not			
			non ton	2011240			
			pay ten	covere			
			thousan	d here.			
			d and I	lts not			
			said I	going			
			salu I	going			
			<mark>can't</mark>	to			
			1	<u> </u>			
			do	affect			
			that. I	me as			
				iiio us			
			<mark>can't</mark>	such. I			
			offord	o o tu o 11			
			arroru	actuan			
			it.	V			
				really			
				annrec			
				appree			
				iate all			
				1			
				what			
				has			
				happe			
				nod to			
				neu io			
				me the			
				chemo			
				the			
				, the			
				doctor			
				s, all			
				covere			
				d, its			
				11			
				really			
				big			
				help			
				and it			
				saved			
				my life			
1		1			1		







		<mark>hospita</mark>		
		l was		
		very		
		accom		
		<mark>modati</mark>		
		<mark>ng.</mark>		
		Lot of		
		the		
		<mark>time,</mark>		
		initially		
		first		
		few		
		weeks		
		of my		
		<mark>chemo,</mark>		
		I didn't		
		do it.		
		After		
		<mark>that it</mark>		
		<mark>had to</mark>		
		be		
		postpo		
		ned to		
		next		
		Wedne		
		sday		
		becaus		
		e it		
		wasn't		
		approv		
		ed.		



		la anna		
		nave		
		had a		
		lot of		
		icenae		
		188008		
		in the		
		beg1nn1		
		ng		
		11 8		
		every		
		· • •		
		time I		
		come		
		for		
		ahama		
		chemo,		
		I have		
		learned		
		that the		
		day		
		before I		
		had to		
		<mark>call</mark>		
		thom to		
		them to		
		ask		
		them if		
		it ie		
		n 15		
		approv		
		. 1		
		ea. The		
		ladies		
		here,		
		they are		
		they are		
		putting		
		the		
		request		



	1			[]
		and		
		then		
		tall ma		
		ten me		
		that I		
		put it in		
		ten		
		dava		
		uays		
		ago and		
		say		
		they are		
		otill		
		<u>Still</u>		
		workin		
		g on it.		
		lt 1S		
		ridioulo		
		Hulculo		
		us!		
		But to		
		be		
		honost		
		nonest,		
		I would		
		be at		
		be at		
		be at home		
		be at home		
		be at home and it		
		be at home and it would		
		be at home and it would		
		be at home and it would have		
		be at home and it would have		
		be at home and it would have been		
		be at home and it would have been		
		be at home and it would have been coming		
		be at home and it would have been coming		
		be at home and it would have been coming out of		
		be at home and it would have been coming out of		
		be at home and it would have been coming out of my		
		be at home and it would have been coming out of my		
		be at home and it would have been coming out of my pocket,		







to my	<mark>son</mark>	in	issues	didn't	friend	throug	sister	with	<mark>my</mark>
husba	and and	<mark>law</mark>	with	suppor	assistin	h this	brough	<mark>my</mark>	family,
nd	daught	had a	<mark>financ</mark>	t me in	<mark>g me</mark>	sickne	t my	family	they
	er,	cance	e, it is	<mark>this</mark>		ss,	mother	and I	were
	they	r so I	covere			almost	. This	manag	there
	are	was	<mark>d and</mark>			all my	gave	ed all.	for me
	with	comf	they (people	me		
	<mark>me.</mark>	ortabl	<mark>did</mark>			change	strengt		
	They	e	everyt			<mark>d,</mark>	<mark>h.</mark>		
	told	<u>talkin</u>	hing			people			
	me, do	g to				whom			
	<mark>like</mark>	her				I was			
	this,	and				with			
	<mark>don't</mark>	get				and			
	eat like	the				though			
	this,	suppo				t they			
	put me	rt I				are			
	in a	neede				close			
	good	<mark>d</mark> .				and I			
	way					though			
	you					t they			
	know (are my			
	for my					relativ			
	food,					e			
	my								
	exercis								
	e								
That	they	T	Mx	L had	My	Some	Also	My	My
was	were	spoke	family	all the	principl	of		family	daughte
from	with	to my	treate			them	sistor	hae	r and
my	meand	famil	d mo	t from	e was	they	is with	heen	
шу	me and		u me		assistill	ulley	15 with	Deell	my son,



husba	I	y and	very	<mark>my</mark>	g me. I	chill	<mark>me</mark> ,	very	really.
nd, I	manag	they	norma	sisters	am a	out	my	suppor	Maybe
was	ed	were	lly. Of		teacher	with	husban	tive	they (
strong		all	course		so she	me but	d	and	encoura
		with	, it		took	I can	NO,	they	ged me
		me to	was		away	say	he is	suppor	so
		<mark>suppo</mark>	hard		some of	they	X. he	ted me	much
		rt me.	for		my	are not	<mark>didn't</mark>	<mark>all.</mark>	to do.
			them		classes	so	stand		They
			to		so it	many	<mark>up</mark>		were
			know		won't	and	with		more
			that I		<mark>drain</mark>	mostly	me.		scared
			have		me out.	I don't			about
			cancer		She put	see			it; <mark>I was</mark>
			but		another	them.			like I
			they		teacher	Becau			will do
			didn't		with	se they			it
			treat		me in	show			anyway
			me as		the	themse			s but
			somet		class so	lves			they (
			hing is		if I	that			were
			abnor		<mark>can't</mark>	actuall			scared
			mal		do	y they			about
					anymor	are not			<mark>it.</mark>
					e, she	my			
					takes (friends			
					over.	. They			
						are			
						scared			
						more			
						than			
						me,			



					they		
					have		
					decide		
					d to		
					stay		
					away		
					from		
					me.		
							_
I have	my all	My	Mainl	Everyb	My	Becau	M <mark>y</mark>
found	family,	husba	y I got	ody	directo	se my	husban
SO	all my	nd	all	was	r in the	manag	d
much	sisters,	was	suppo	most	school	er she	passed
suppor	they	very	rt	helpful	, he	told	away
t from	were	<mark>suppo</mark>	from		<u>knows</u>	<mark>me,</mark>	seven
my	all	rtive	Allah		what	you	year
family	with	thoug	and		happe	can sit	ago but
, from	me	<mark>ht</mark>	my		ned to	no	I got all
my		and I	family		me.	proble	the
relativ		had			Actual	<mark>m;</mark>	support
es		no			ly, he	your	from
that's		issue			gave	salary	my
why		<mark>talkin</mark>			me one	is OK	family
they		g to			month	but for	
gave		<mark>him</mark>			off	me I	
us		about			when I	<mark>said,</mark>	
strengt		<mark>it.</mark>			started	No, I	
h					<mark>my</mark>	should	
hamdu					<u>chemo</u>	go to	
llila					so I	work	
					can		
					contin		



					ue my		
					<mark>job</mark>		
т	Moinl	mu		I om	I do	when I	т
found	v from	famil		I alli blessed	remem	have	. I didn't
				with	bor		toll
a way	abildro	y was		with a	the	sessio	onythin
to deal		very		very	first	n and	anytiin
throug	fomily	suppo		good	ahama	that	
h my	and	IIIve			thorp		
fo.i4h					, there		
Taith	my				was	mother	
and	sisters.				Syrian	sat	daughte
my				and sne	nurse,	With	r, my
family	lots of			has	she did	me for	son and
suppor	suppor			been	my	nine	that's it
t	t from			immen	first	month	
	my			sely	chemo	<mark>S.</mark>	
	family.			support	. I		
	I			ive	cried; I		
	manag				don't		
	ed well				know		
	with				how to		
	their				explai		
	suppor				ned. I		
	t				felt		
					there is		
					somet		
					hing in		
					body, I		
					don't		
					<mark>know,</mark>		
					I felt I		



			won't		
			wake		
			<mark>up.</mark>		
			She		
			was		
			standi		
			ng		
			rigni		
			next to		
			me and		
			<mark>said</mark>		
			why		
			are		
			you		
			crying,		
			are		
			VOII		
			missin		
			g		
			someo		
			ne.		
			Everyt		
			hing		
			will be		
			<mark>OK</mark>		
Ι	Mainl		All the	From	
manag	y		doctor	<mark>my</mark>	
ed all	<mark>famil</mark>		<mark>s,</mark>	<mark>husban</mark>	
with	y, my		nurses,	d, No,	
my	famil		they	there	
family	y was		have	was no	
5					



and	very			like	suppor	
my	<mark>suppo</mark>			becom	t. <mark>I felt</mark>	
doctor	rtive			e more	suppor	
				concer	t from	
				ned	my	
				<mark>than</mark>	mother	
				my	, from	
				friends	<mark>my</mark>	
					sister.	
					Actual	
					<mark>ly,</mark>	
					from (
					my	
					family	
My	I		When I		So, my	
family	have		came		family	
and	few		for my		gave	
my	friend		first		me	
faith	<mark>S</mark>		operati		this (
in			on, Dr.		not my	
Allah			Mouza		husban	
made			said,		d. I	
me			almost		have	
pass			like my		<mark>two</mark>	
throug			entoura		kids	
h all			ge,		and	
difficu			there		my	
lties			was		husban	
			always		d but	
			someon		he	



		e with	didn't	
		me. A	suppor	
		friend	t me	
		of mine	emotio	
		Amina,	nally,	
		we	he	
		were	didn't	
		togethe	suppor	
			suppor	
		r so i	t me	
		went	rinanci	
		for my	ally. It	
		operati	is	
		on,	difficu	
			lt for	
			me,	
			<mark>my</mark>	
			husban	
			d has	
			money	
			he	
			could	
			have	
			given	
			me	
			money	
			tor my	
			surger	
			y but	
			he	
			didn't	
			give	
			me. He	



				let me	
				go to	
				Hilal	
				Ahmar	
				(Charit	
				($)$ $)$ $	
				this to	
				manag	
				e my	
				surger	
				y. It	
				was	
				difficu	
				lt for	
				me	
				inc.	
Doctor			I am	Alham	
Mouza			not	dullila	
Al			lving. I	after	
Ameri			swear	that	
sho is			to God	my	
, SHE 15			to 000,	iiiy (1	
a very			every	mother	
nice			day I		
doctor			had		
. We			lunch, I		
don't			had		
feel			dinner.		
like			It		
her_as			would		
a			be left		
doctor			at my		
shalie			door		
, sne is			door.		



Lixe Every Triend ady she Sol sent Sol somebo dy from cleanin Cleanin cleanin g compa ny, cvery day my apartm cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned becaus cleaned <th></th> <th></th> <th></th> <th></th> <th></th> <th></th>						
friene: day she sent, Siste: somebo dy from cleanin g compa g compa nu ny, every day my apartin ent got cleaned becaus ent got cleaned g apartin ent got someto g apartin g apartin g apartin ent got someto g apartin g apartin g	like a			Every		
Image: Sintering in the sent, Sintering in the source of the so	friend			day she		
Sister Sister Sister Sister Sister Sister Sister Sister Sister Sister Sister Sister Sister Sister Sister Sister Si	or			sent,		
dy from cleanin g compa ny. every day my apartn ent got cleaned becaus e e g Cannot, I must not do anythin g, Alham dullia she took me like her sister and she did it for the whole	sister			somebo		
cleanin g compa ny, every day my apartm ent got cleaned becaus e I				dy from		
g compa ny, every day my apartm ent got cleaned becaus e becaus e i				cleanin		
Image: Image:				g		
ny,everyday myapartment_gotcleanedbecauseIcannot,I				compa		
everyday myapartment gotcleanedbecauseegAlhamdulliashetookme likehersisterand shedid itfor thewhole				<mark>ny,</mark>		
day myapartment gotcleanedbecausegcannot,Imustnot doanything.Alhamdulliashetookme likehersisterand shedid _ itfor thewhole				every		
apartm ent_got cleaned becaus e l becaus e l cannot, l l mot_do anythin g. Alham dullia she look me like her sister and she did_it for the whole				day my		
Image: Image:				apartm		
cleaned becaus e i i <tr< th=""><th></th><th></th><th></th><th>ent got</th><th></th><th></th></tr<>				ent got		
Image: second				cleaned		
e I cannot, I must not not anythin g. Alham dullia she took me like her sister and she did it for whole				becaus		
I I I I <th></th> <th></th> <th></th> <th>e I</th> <th></th> <th></th>				e I		
IImustnotdoanything.g.Alhamdulliadulliabisbiscookbisme likebishersisterand shediddiditfor thebiswholebis				<mark>cannot,</mark>		
not do anythin g. g. Alham dullia dullia b. b. <th></th> <th></th> <th></th> <th>I must</th> <th></th> <th></th>				I must		
anything.Alhamdulliadulliashetookme likehersisterand shedid itfor thewhole				not do		
g.Alhamdulliadulliashetooktookme likehersisterand shedid itfor thewhole				anythin		
Alhamdulliadulliasheshetookme likehersisterand shedid itfor thewhole				<mark>g.</mark>		
dulliadulliasheshetookme likeliiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiii				Alham		
she took ne like her sister and she did it for the whole				dullia		
took me like her sister and she did it for the whole				<mark>she</mark>		
me like her sister and she did it for the whole				took		
her sister and she did it for the whole				me like		
sister and she did it for the whole				her		
and she did for the whole				sister		
did it for the whole				and she		
for the whole				did it		
whole				for the		
				whole		



			of my period when I was recover ing till I do it and I			
			said Amina, I can do it and only then she did stop.			
He insiste d me to go to the hospit al and check				it is very import ant for you to have suppor tive people		
				She (docto r) is really fun		



						she (docto r) was very concer ned, she was next to me when they standi ng they standi ng when they bed when they bed when they bed when they standi ing researc hed about wheth er it is			
						not.			
Sub-the	e <mark>me 4: En</mark>	notional	concern	I <mark>S</mark>					
I was	<mark>I was</mark>	, I	I was	the	I <mark>have a</mark>	<mark>First, I</mark>	becaus	<mark>I was</mark>	But
not	all the	was	scared	<mark>look of</mark>	ten-	remem	e I	scared	there a construction of the second se
happy	<mark>time</mark>	<mark>little</mark>	thoug	<mark>the</mark>	year-	ber, I	<mark>don't</mark>		was



becaus	<mark>thinkin</mark>	depre	<mark>h</mark>	surger	<mark>old</mark>	was	want	difficul
e there	<mark>g if</mark>	ssed	<mark>becau</mark>	<mark>y is</mark>	<mark>daught</mark>	<mark>sitting</mark>	to	ty like
were	they (<mark>of</mark>	se I	very	er so I	<mark>down</mark>	<mark>think</mark>	when I
some	will	<mark>cours</mark>	have a state of the second sec	<mark>bad. It</mark>	<mark>had to</mark>	and	too	<mark>saw</mark>
worrie	tell me	e it's	<mark>small</mark>	is and the second second second second second second second second second second second second second second se	be	cry	much	myself
s and	somet	not	<mark>kids</mark> ,	affecti	<mark>brave</mark>		<mark>and if I</mark>	<mark>in the</mark>
things	<mark>hing</mark>	good		<mark>ng me</mark>	and		will sit	<mark>mirror,</mark>
to be	<mark>bad,</mark>	news		<mark>psycho</mark>	strong		<mark>at</mark>	<mark>it was a</mark>
worrie	what I	<mark>but</mark>		logical	for her.		home	<mark>bit</mark>
d	<mark>will do</mark>	<mark>later I</mark>		<mark>ly</mark>			<mark>and I</mark>	<mark>psychol</mark>
about		accep					will	<mark>ogical</mark>
but		ted					<mark>think</mark>	<mark>or say</mark>
you		the and a constant					too	emotio
		realit					much	<mark>nal</mark>
		y					<mark>and I</mark>	
							will	
							feel I	
							<mark>am</mark>	
							weak.	
							I	
							didn't	
							want	
							to feel	
							<mark>this</mark> .	
							<mark>Also, I</mark>	
							<mark>can</mark>	
							see	
							<mark>from</mark>	
							<mark>my</mark>	
							mother	
							's	
							eyes,	



							<mark>she is</mark>		
							very		
							sad		
							<mark>about</mark>		
							<mark>me.</mark>		
							<mark>For</mark>		
							me, I		
							will		
							<mark>show</mark>		
							<mark>her, I</mark>		
							<mark>am</mark>		
							<mark>good, I</mark>		
							am		
							strong		
Things	T	it was	I was	T	L tried	Someti	After	T	L didn't
are	• wasn't	also	worrie	• couldn	to get	mes its	that. I	• didn't	tell
better	afraid	becau	d and	't tell	my self	hetter	will be	find	anvone
now	when I	se I	of	mv	positiv	not	in	anv	about it
alham	heard	lost	course	mother	e as I	feel	front	emotio	because
dullila		my	unhap	; I am	could	anythi	of him	nal or	vou
		son	py	having		ng.	even	social	- know,
			1.	this c		-	more	issues	they
							III OI C	100000	UTIC y
				disease			strong	100400	would
				<mark>disease</mark> . <mark>I used</mark>			strong becaus	100400	would be
				disease . <mark>I used</mark> to tell			strong becaus e he	100400	would be worried
				disease . I used to tell her that			strong becaus e he wante		would be worried about
				disease . I used to tell her that I am			strong becaus e he wante d to		would be worried about me, I
				disease . I used to tell her that I am going			strong becaus e he wante d to see I		would be worried about me, I didn't
				disease . I used to tell her that I am going outside			strong becaus e he wante d to see I am		would be worried about me, I didn't want
				disease . I used to tell her that I am going outside for my			strong becaus e he wante d to see I am weak.		would be worried about me, I didn't want that so I



				and I used to hide in my sister's house. When I used to be back home, I still used to have this pain but I had to hide it			eve n I am, I will not show him I am weak.	to myself. And you know when people about it they will be shocke d; oh, how would you do it, it is danger ous. I didn't want that negativ ity.
I was	<mark>my</mark>	<mark>I after</mark>	I	<mark>My</mark>	It had	what I	but <mark>for</mark>	lf I
satisfi	husban	Surge	didn't	mother	also	have to	me I	didn't
ed and	d, he	ry I	experi	passed	psychol	$\frac{do.}{\Delta r d - c}$	would	remove
1 am	nad a	suffer	ence	away	ogical	And of	put malea	It, It
suill	when	ed nain	any psych	and I	became	when		may have
satisti	ho	P 111.		couldin	voru	, when	up at	nave
	passed	that	al or	tell	very	don't	And	other



away. I	didn't	emoti	her.	depress	know	also.	place
was	impa	onal	vou	ed	what is	mv	and I
with	ct mv	issues	know		chemo	family	didn't
him	lifeas	with			how	would	know
and I	much	family			does it	see me	what
	when	or			affect	that I	would
overvt		with				am	hannen
bing					you.		
mng		fri en d					
	SON.				scared	least.	really
					you		
	loss	collea			becaus		that it
	made	gues.			e you		was
	me				don't		remove
	emoti				know		d. I
	onal				anythi		don't
	and				<mark>ng</mark> .		regret
	low						anythin
							<mark>g.</mark>
L don't	T	. L had	They	It was	Of	T	
want	dealt	no	saw it	like I	course	- would	
to	with	emoti	affecte	was	it was	sav	
suffer	the	onal	d me	devasta	anory	physic	
when I	cance	or	too	ted	like	al and	
think	r but	social	much		sadnes	emotio	
about	his	concer	and		suarros	nal	
him I		ne	after		e,	<mark>1101</mark>	
don't	wae	hamd			it was		
wont	the		monthe		like no		
to be	most	uma	montuis				
			, my				
			sister		ns at		
him.	engin		<mark>who</mark>		all. So,		



	He	<mark>g in</mark>		<mark>had a</mark>		<mark>all,</mark>			
	was	<mark>my</mark>		cancer,		<mark>emotio</mark>			
	too	<mark>life</mark>		<mark>she</mark>		<mark>ns or</mark>			
	much	becau		<mark>also</mark>		mixed			
	depres	se I		died.		emotio			
	sed	was		That is		<mark>ns</mark>			
		the and a constant		<mark>like</mark>					
		one		<mark>destroy</mark>					
		with		ed me					
		<mark>him</mark>		really.					
		and							
		<mark>takin</mark>							
		g care							
		while							
		<mark>he</mark>							
		was							
		<mark>ill</mark> .							
	. <mark>I said</mark>	I do	In our	I	I am	before	lt's		. If I
	myself	not	family	couldn	here	I	difficu		didn't
	, I will	have	, there	't tell	alone	prayed	lt to		remove
	<mark>fight.</mark>	much	are	them	with	and I	accept		it, it
		friend	lots of	what is	my	didn't			may
			people	in my	daught	cry but			have
			who	heart	er, it's	<u>this</u>			gone to
			have	becaus	difficul	time I			other
			cancer	e you	t	pray			place
			. My	know	becaus	and I			and I
			sister	they	e I have	<mark>cry so</mark>			didn't
			also	were	to be	this I			know
			has	also	strong	can			what
1	1	1	1 .		<u>C</u> 1		1	l	1 1


		cancer	throug		differe		happen
		<mark>so it</mark>	<mark>h this</mark>		<mark>nt</mark>		<mark>so I am</mark>
		<mark>wasn't</mark>	traged				really
		<mark>that</mark>	<mark>y</mark> .				<mark>happy</mark>
		much.					<mark>that it</mark>
							was
							remove
							d. I
							don't
							regret
							<mark>anythin</mark>
							<mark>g.</mark>
I love	I	I think	A <mark>nd</mark>	The	Dealin	My	
<mark>my</mark>	wasn'	the a	they	only	g with	anothe	
<u>childre</u>	t	most	also	time, I	emotio	r	
<mark>n</mark> , <mark>I</mark>	going	<mark>difficu</mark>	<mark>lost</mark>	<mark>can be</mark>	<mark>ns,</mark>	<mark>collea</mark>	
want	out	<mark>lt for</mark>	one	myself	<mark>may</mark>	gue	
<mark>to see</mark>	much	<mark>me</mark>	sister	<mark>is when</mark>	<mark>be,</mark>	told	
their (and I	was, I	becaus	<mark>she is</mark>	was	me	
<mark>childre</mark>	was	was	<mark>e of</mark>	gone to	<mark>the</mark>	why	
<mark>n and</mark>	<mark>bit</mark>	<mark>away</mark>	cancer.	bed and	bigges	you	
becom	<mark>down</mark>	<mark>from</mark>	<mark>I have</mark>	<mark>I can lie</mark>	t	<mark>told</mark>	
<mark>e a</mark>	so I	<mark>my</mark>	to be	and I	<mark>challe</mark>	her	
<mark>grand</mark>	was	<mark>kids</mark>	<mark>careful</mark>	<mark>can cry</mark>	<mark>nge for</mark>	<mark>about</mark>	
mother	<mark>alway</mark>	<mark>becau</mark>	what I	<mark>into my</mark>	<mark>me</mark>	<mark>it and I</mark>	
	<mark>s with</mark>	se I	<mark>am</mark>	<mark>pillow</mark> .	becaus	<mark>said no</mark>	
	<mark>my</mark>	<mark>had</mark>	<mark>saying</mark>		<mark>e you</mark>	<mark>proble</mark>	
	<mark>famil</mark>	<mark>my</mark>	to them		need	<mark>m, its</mark>	
	y	surger	<mark>and all</mark>		lots of	<mark>time</mark>	
		y done	<mark>this</mark>		strengt	and	
		<mark>in the</mark>	<mark>thing.</mark>		<mark>h and</mark>	<mark>InshAl</mark>	



	Germ		<mark>it is</mark>	<mark>lah it</mark>	
	any any		very	will	
	and		<mark>import</mark>	pass.	
	this (ant for		
	was		you to		
	the		have		
	bigges		suppor		
	t		tive		
	- challe		people		
	nge				
	for me		-		
	to be				
	apart				
	from				
	kids.				
	It				
	wasn't				
	easy				
		That is		I	
		<mark>always</mark>		<mark>didn't</mark>	
		<mark>at the</mark>		<mark>think</mark>	
		back of		too	
		<mark>my</mark>		much	
		mind,		<mark>about</mark>	
		tomorr		depres	
		<mark>ow I</mark>		sion	
		<mark>can</mark>		<mark>and I</mark>	
		lose my		<mark>tell to</mark>	
		job and		<mark>myself</mark>	
		then (<mark>, now I</mark>	
				<mark>am</mark>	



				what do		<mark>taking</mark>	
				<mark>I do?</mark>		treatm	
						ent,	
						lets	
						live	
						now	
How []	<mark>I had</mark>			I didn't			
you	a fear			feel			
<mark>think,</mark>	<mark>of</mark>			like I			
<mark>its</mark>	recurr			want to			
very	ence			<mark>go my</mark>			
<mark>import</mark>				friends			
<mark>ant</mark> .				and			
Think				<mark>colleag</mark>			
<mark>in a</mark>				<mark>ues to</mark>			
good				tell			
way				them			
				like I			
				feel			
				this.			
				like			
				they			
				have			
				enough			
				of them			
	I was			I feel			
	<mark>alway</mark>			like I			
	<mark>s</mark>			<mark>cannot</mark>			
	<mark>havin</mark>			<mark>go to</mark>			
	<mark>g this</mark>			them (
	fear			all the			
	1	1	1		1		1



	tha tha	t I		<mark>time, I</mark>			
	mi	<mark>ght</mark>		feel			
	get			<mark>like I</mark>			
	anc	<mark>th</mark>		<mark>cannot</mark>			
	er			vent			
	can	ce		<mark>out my</mark>			
	r			<mark>story</mark>			
	The			. I just			
	big	ge		<mark>sit on</mark>			
	<mark>st</mark>			<mark>my</mark>			
	<mark>cha</mark>	lle		<mark>muslaa</mark>			
	nge			<mark>h</mark>			
	was	<mark>,</mark>		<mark>(prayer</mark>			
	<mark>my</mark>			<mark>mat)</mark>			
	son			and			
	pas	se		cry			
	d			<mark>cry in</mark>			
	awa	<mark>iy</mark>		<mark>the</mark>			
	anc			middle			
	<mark>tha</mark>	t		of the			
	was	5		night			
	mo	re					
	cor	ce					
	rni.	ng					
	to	ne					
	<mark>tha</mark>	n					
	my						
	ow	1					
	hea	lt					
	<mark>h</mark> .						
1			1	1	1	1	



	<mark>I was</mark>	. The		
	<mark>takin</mark>	decisio		
	g care	<mark>n to</mark>		
	of	remove		
	him	<mark>both of</mark>		
	and	<mark>my</mark>		
	he	<mark>breast,</mark>		
	<mark>passe</mark>	<mark>it was</mark>		
	d	such an		
	<mark>away</mark>	<mark>immen</mark>		
	<mark>just</mark>	se step;		
	<mark>two</mark>	needed		
	mont	someon		
	<mark>hs</mark>	<mark>e to</mark>		
	<mark>befor</mark>	speak		
	e I	to		
	was			
	<mark>diagn</mark>			
	osed			
	with			
	cance			
	r			
	This	Δς Ι		
	was	told her		
	the	she		
	nart	need to		
	that	speaks		
	impa	about		
	cted	her		
	me	fears or		
	most	concer		



_					
	and		ns like		
	<mark>even</mark>		<mark>anytim</mark>		
	now I		<mark>e she</mark>		
	think		has a		
	of my		questio		
	son		n she		
	that		n, she		
	that		neeus		
	more		to ask		
	than		me.		
	<mark>my</mark>				
	<mark>condi</mark>				
	tion				
			The		
			fear of		
			recurre		
			nce that		
			<mark>it will</mark>		
			come		
			back, it		
			was		
			always		
			there in		
			the		
			back of		
			<mark>my</mark>		
			mind		
			<mark>that I</mark>		
			have to		
			do		
			revisio		
			ns tor		



ford	
IESTER	727

		<mark>my</mark>				
		<mark>daught</mark>				
		er if I				
		<mark>am not</mark>				
		there				
		<mark>anymor</mark>				
		e and I				
		have to				
		prepare				
		<mark>her for</mark>				
		<mark>that</mark>				
		<mark>situatio</mark>				
		<mark>n and</mark>				
		like we				
		never				
		<mark>know</mark>				
		what				
		will				
		happen				
		so we				
		talked				
		about				
		it.				
		And I				
		<mark>feel my</mark>				
		<mark>daught</mark>				
		er				
		needed				
		to				
		speak				
		to				
			1	1	1	1



		someon		
		e apart		
		from		
		me		
		becaus		
		e we		
		have		
		have		
		nere		
		alone		
		for ten		
		years		
		and		
		becaus		
		e my		
		nephew		
		<mark>came</mark>		
		and he		
		<mark>is with</mark>		
		us here		
		now		
		and we		
		dida't		
		nave		
		anyone		
		. And		
		<mark>this</mark>		
		<mark>was</mark>		
		<mark>devasta</mark>		
		ting for		
		her. I		
		feel she		



			needs		
			someon		
			e, she		
			needs		
			<mark>just to</mark>		
			speak		
			to let it		
			out.		
			L don't		
			shoro os		
			nave		
			daught		
			er, I		
			didn't		
			want		
			her to		
			let her		
			see		
			becaus		
			<mark>e first I</mark>		
			<mark>had to</mark>		
			<mark>own it</mark>		
			to		
			<mark>myself,</mark>		
			<mark>I had to</mark>		
			<mark>look at</mark>		
			it and I		
			<mark>didn't</mark>		
	1				



		<mark>after</mark>		
		<mark>the</mark>		
		<mark>operati</mark>		
		<mark>on. I</mark>		
		would		
		<mark>just not</mark>		
		<mark>look</mark>		
		<mark>into the</mark>		
		mirror.		
		And I		
		<mark>had a</mark>		
		mirror		
		the one		
		full		
		body		
		mirror		
		which I		
		<mark>put it in</mark>		
		<mark>my</mark>		
		room		
		and		
		now I		
		have		
		just		
		- small		
		round		
		one and		
		I		
		needed		
		to		
		speak		
		to		
		_		



		someon		
		<mark>e.</mark>		
		amotio		
		concer		
		ns were		
		the		
		most		
		importa		
		nt		
		issues I		
		had to		
		deal		
		with		
		the		
		cancer.		
		The		
		<mark>psychol</mark>		
		ogical		
		and and		
		<mark>emotio</mark>		
		nal		
		impact		
		it had		
		on me.		
		I didn't		
		know		
		how		
		this		



					will				
					will affect				
					my life				
					like the				
					first				
					time I				
					had				
					lump				
					on my				
					side so				
					I				
					- thought				
					, they				
					will				
					take it				
					out and				
					I will				
					<mark>be good</mark>				
							<u> </u>		
Sub-the	eme 5: Sp	iritual/l	Religious	s concerns	5		Γ		
Musli	when	I	when	this	We	You	I told	My	I
m	he told	mana	I was	disease	need	know I	my	God	believe
always	me, I	ged it	diagn	made	religion	think	god I	and	if
has	feel	all	osed	my	like	God is	agree	my	anythin
faith	that	with	with	faith	someth	with	for	family	g
so I	the	my	cancer	into	ing to	me	everyt	really	happen
said	God	faith	, since	God	hold	cause	hing	suppor	s to
alham	gave	in	then I	stronge	onto	the		ted	anyone,
dullila	me	Allah	got	r.		treatm		me.	it is for
	somet		much			ent did			a
	hing		closer			not			reason
						affect			so



	like		to			me			Allah
	power		Allah.			that I			he
						had to			knows
						stop			what is
						the			better
						work.			for you
									so it
									was
									totally
									fine. It
									made
									me
									closer
									to God.
It	If this	Allah	Durin	Becaus	I have	You	Alham	Mv	
worke	is	made	g my	e the	always	know	dullila,	religio	
d for	what,	it	diseas	way	been a	when	I got	us	
me so	God	easy	e and	God	Muslim	you	more	belief	
Ι	want	for	while	stands	but	pray, it	closer	really	
should	it.	me	I was	with	now I	release	to	helped	
say			receiv	me,	am	s your	Allah.	me	
alham			ing	nobod	gonna	fears.	From	throug	
dullila			my	y can	be a		the	h the	
as I			treatm	be with	stonged		beginn	diseas	
have			ent, I	me.	Muslim		ing, I	e and	
faith			found		. I made		accept	to	
			mysel		a lot of		ed, I	cope	
			f		Tahajju		prayed	up	
			closer		d (night		and I		
			to		prayer)		told		
			Allah		and		my		



					salah in		Allah,	
					the		I agree	
					Ramad		with	
					an		all I	
							have.	
When	Allah	I	Mainl	In this	I have	Throu	I	
we	loves	found	y I got	situatio	done	gh		
have	me.	more	all	n, you	more	prayer		
faith,		peace	suppo	will	sunna	s, the		
it		and	rt	feel	prayers	words		
helps		coura	from	how		you		
us		ge	Allah	God is		read in		
pass		with	and	closest		a		
throug		my	my	to you.		particu		
h all		stron	family			lar		
the		g				way, it		
difficu		faith				alread		
lties		in				y gives		
hamdu		Allah				your		
llila		and				brain a		
		in my				differe		
		religi				nt		
		on				world		
						and of		
						course		
						the		
						feeling		
						s,		
						when		
						you		
						are in		



				this situati on and your really are going to think about this.		
I think how non- Musli ms face the proble ms in their lives hamdu llila	I discov ered so many good things peo ple		Becaus e if you don't have religion , you are lost.	I think I got more closer to the God	I will put tomorr ow up to God. Becau se I accept ed all from the beginn ing.	
Islam is a gift; it is really a gift.			I have gone closer to my God	when you are happy, you are	I got the main suppor t from	



It			closed	my	
helps			to the	God	
us to			God, if		
pass			you		
everyt			are		
hing			sad,		
			you		
			are		
			closed		
			to the		
			God.		
Ι			Really	Hamd	
found			this is	ullila,	
a way			a	it is	
to deal			miracl	from	
with it			e that I	my	
throug			really	God, I	
h my			didn't	had	
faith			get any	too	
			bad	much	
			effects	suppor	
			from	t here	
			the		
			treatm		
			ent and		
			I think		
			it is		
			from		
			God.		
Most					
import					
mport					



antly									
with									
my									
faith									
in									
Allah,									
there									
were									
difficu									
lties									
but I									
manag									
ed all									
hamdu									
llila									
my									
foith									
in									
111 Δllah									
made									
me									
nass									
throug									
h all									
difficu									
lties									
Theme	3: Expect	tations o	o <mark>r supp</mark> ol	rt needed					
Musli	Yes, I	I was	I was	You	I was	I am so	May	I was	The
m	am	fully	satisfi	have	satisfie	happy	be if	fully	care I
always	satisfie	satisfi	ed	very	d with	about	doctor	satisfi	receive



has	d with	ed	with	good	the	the	s can	ed	d was
faith	the	with	the	treatm	care	nurser	provid	with	perfect.
so I	care I	the	care I	ent so	but	s and	e	care I	I was
said	receive	care I	receiv	you	there	researc	emoti	receiv	satisfie
alham	d	recei	ed. I	cannot	should	hers. I	onal	ed. For	d with
dullila		ved	didn't	say	be an	can	suppo	me its	everyth
			have	that	improv	say	rt	all	ing. I
			any	you	ement	that		good,	got the
			issue	didn't	in	they		the	care
			really	have a	inform	are		servic	that I
				good	ation	with us		e	need
				care	deliver	in this.		Tawa	and
					y	It's		m	even
						their		hospit	more
						job		al	really
						and		provid	
						people		ed.	
						can do		Today	
						their		I am a	
						job in		surviv	
						differe		or	
						nt		becaus	
						way,		e of all	
						see		the	
						here		doctor	
						it's		s and	
						really		staff at	
						helpful		Tawa	
								m	
								Hospit	
								al	
								Oncol	



							ogy	
							depart	
							ment.	
It	they	Every	Tawa	Lack of	I have	Ι	If	
worke	treated	thing	m	inform ation	two	prefer	hospit	
d for	me a	was	pharm	from	ways	if I	al can	
me so	good	good	acy	the doctor	of my	receiv	provi	
Ι	way	and	servic	s, this	surger	e	de	
should		docto	es are	affects	y so	more	enhan	
say		r kept	very		plastic	suppo	ced	
alham		on	slow, I		surger	rt	cover	
dullila		check	think		у	from	age	
as <mark>I</mark>		ing	that		doesn'	health	for	
have		about	should		t come	provid	those	
faith		every	be		under	ers	who	
		thing.	impro		manda	about	cannot	
			ved		te	the	afford	
			for		approv	physic	the	
			faster		ed	al	treatm	
			servic		treatm	issues	ent,	
			es and		ent but	I have.	this	
			better		in my	May	would	
			delive		case,	be	be	
			ry.		better	doctor	very	
					to go	s can	helpfu	
					for this	also	l. But	
					operati	provid	it is all	
					on. So,	e	good	
					for	emoti	and I	
					some	onal	think	
					patient		they	



				s, like	suppo	are	
				in my	rt.	doing	
				case		all	
				like		fine.	
				somet			
				hing to			
				do			
				with			
				the			
				increa			
				se or			
				enhan			
				ced			
				covera			
				ge			
I was	All the	I		Expan	After I		
satisfi	people	don't		d the	came		
ed,	are	think		mand	here,		
and I	good	there		ate,	hamdu		
am	even	is		this	llila		
still	when I	anyth		will	from		
satisfi	had	ing		help	<mark>my</mark>		
ed	radiati	missi		me of	God, I		
especi	on,	ng		course	feel I		
ally	they	from		but	am		
with	were	the		will	comfo		
ту	helpful	healt		help	rtable		
Docto	people	hcare		my			
r	hamdu	depar		doctor			
	llilah.	tment		as			
				well.			



				-	
			I have		
			told		
			you if		
			possibl		
			e, in		
			cases		
			of		
			researc		
			h if		
			you		
			add		
			such		
			manda		
			tes to		
			cover		
			other		
			co-		
			existin		
			g		
			proble		
			ms, it		
			will		
			help.		
			It's not		
			just		
			helpful		
			, it's		
			very		
			helpful		
I have	T		I no alle		
1 nave	l				
an	recei		apprec		



appoin	ved		iate		
tment	all		SEHA		
at one	the		or		
O'cloc	updat		whoev		
k, I see	es I		er		
the	neede		create		
doctor	d		d this		
at	from		progra		
three	my		m for		
0'	docto		expatri		
clock	rs and		ate to		
So	nurse		get this		
timing	s and		treatm		
to see	the		ent		
the	docto				
doctor	r was				
must	callin				
be	g me				
reduc	to				
ed.	check				
	on				
	me				
	how I				
	am				
	doing				
	so I				
	got				
	the				
	suffic				
	ient				
	suppo				
	rt				



from				
the				
healt				
hcare				
depar				
tment				



APPENDIX 12: GLOSSARY OF TERMS USED (ABBREVIATIONS)

ACS: American Cancer Society

Adjuvant therapy: Additional treatment(s) that are offered after a surgical procedure, designed to target microscopic traces of tumour which may have been left behind.

Advanced Stage: Mainly Stage IV, advanced breast cancer means that a cancer that began in the breast has spread to another part of the body.

ALND (Axillary Lymph Node Dissection): An axillary lymph node dissection (ALND) is surgery to remove lymph nodes from the armpit (underarm or axilla).

Arabs: Those who speak Arabic as their native tongue. Arabs are not a distinct ethnic group, since there are both white Arabs and black Arabs. Majority of Arabs are Muslims, but some are Jews and Christians too. In this research Arabs are considered as Muslims.

Aromatase Inhibitors: A type of hormonal therapy that reduces the levels of oestrogen in the body by blocking aromatase, an enzyme which helps to form oestrogen from other steroids. Examples include Anastrozole (Arimidex), Letrozole and Exemestane.

ASCO: American Society of Clinical Oncology

Brachytherapy: It is a form of radiotherapy where a sealed radiation source is placed inside or next to the area requiring treatment.

Breast reconstruction: The formation of a breast shape after a total mastectomy (see mastectomy), using a synthetic implant or tissue from the woman's body.

Chemotherapy: The use of anti-cancer drugs to destroy cancer cells. This treatment can be given alone or as part of adjuvant therapy.

Clinical factors: This research will focus on clinical attributes including primary site, cancer stage and treatment modalities.

DoH: Department of Health (Abu Dhabi)

UAE: United Arab Emirates

HAAD: Health Authority Abu Dhabi

Early Stage: Breast cancer that has not spread beyond the breast or the axillary lymph nodes. This includes ductal carcinoma in situ and stage I, stage IIA, stage IIB, and stage IIIA breast cancers.



Emirati Nationals: UAE citizens/nationals

Expatriate: Any non-UAE National who has entered the Emirate of Abu Dhabi for the purpose of work or residence under a permanent or temporary work or residence permit issued by the State. The term "Expatriate Resident" shall also include the resident families and dependants of the Expatriate Resident who enter the Emirate for the purpose of residence.

FOCP: Friends of Cancer Patients

Herceptin: Also known as Trastuzumab, it is a type of biological therapy used to reduce the risk of breast cancer coming back in women with HER2 positive breast cancer (type of breast cancer).

Hormonal Therapy: Type of treatment used to inhibit the growth of hormones in the body which are responsive to cancer cells (also see: Arimidex, Aromatase inhibitors and Tamoxifen). **Invasive Breast Cancer:** Type of breast cancer where the cancer cells have spread outside the lining of the ducts or lobules into the surrounding breast tissue.

IARC: International Agency for Research on Cancer-GLOBOCAN

IOM: Institute of Medicine

Lumpectomy: It is a surgery to remove cancer (the lump) and some other abnormal tissue from your breast. Lumpectomy is a form of "breast-conserving" or "breast preservation" surgery.

Lymphoedema: Swelling of the arm or breast due to a build-up of lymphatic fluid.

Mastectomy: It is the removal of the whole breast. **Metastasis:** The spread of cancer to other organs in the body.

MoH: Ministry of Health

MOHAP: Ministry of Health and Prevention

Multiple Primary (MP): Multiple primary cancers are usually defined as primary malignant tumours of different histological origins in one person. In Breast, multiple tumours with the same histologic type appearing in the breast and in a different site are regarded as multiple primary tumours unless stated as metastatic tumours.

Muslims: Those who follow or practice Islam (religion).

Nationality: Representation of the passport held by an individual. However, in this study the term nationality is used in cultural context classified into two groups: Emirati Nationals and Expatriates

NCCN Guidelines: National Comprehensive Cancer Network Guidelines



Neoadjuvant therapy: Treatment given as a first step to shrink a tumour before the main treatment, which is usually surgery, is given.

Primary Breast Cancer: It is a cancer that hasn't spread beyond the breast or the lymph nodes under the arm.

Prophylactic contralateral mastectomy: It is a risk-reducing mastectomy performed for the patient diagnosed with an invasive or a non-invasive breast cancer.

Psychosocial Concerns: This research will cover psychosocial issues including physical, informational or educational, social and financial, emotional and spiritual or religious concerns which are translated into patients' needs issues of cancer, beyond the diagnosis and treatment phases. Survivorship includes issues related to the ability to get health care and follow-up treatment, late effects of treatment, second cancers, and quality of life.

Quran: Muslims Holy book

Radiotherapy: Type of treatment which uses high energy x-rays to destroy cancer cells.

Recurrence: Cancer that has recurred (come back), usually after a period of time during which the cancer could not be detected.

SEHA: Abu Dhabi Health Services

Survivor: An individual diagnosed with cancer from the time of initial diagnosis until his or her death.

Survivorship: In cancer, survivorship focuses on the health and life of a person with cancer post treatment until the end of life. However, in this research the term survivorship encapsulates the survivors' experience/needs throughout cancer trajectory starting from the diagnosis.

Tamoxifen: Type of hormonal therapy which uses an anti-oestrogen drug to prevent oestrogen in the body from attaching to breast cancer cells

WHO: World Health Organisation.



APPENDIX 13: MAYO CLINIC APPROVAL FOR THE USAGE OF SURVEY

From: Cancer Education Program Sent: Thursday, June 25 6:29 PM To: 'Maria Aamir' Subject: RE: Permission to use cancer needs survey for the PhD dissertation

Dear Dr. Aamir,

Thank you for your note and interest in the Cancer Survivorship Survey Tool that was developed at Mayo Clinic. We are happy to share the tool. I am attaching the survey in English and Spanish. You have permission to use as-is or adapt the survey for your research. Our only request is to credit Mayo Clinic for the tool, please reference the survey development from our program. Original survey developed by the Cancer Education Program, Mayo Clinic Cancer Center – Cancer Survivor Survey of Needs. I also attached a copy of our survey results/study manuscript from 2013.

I would love to hear back from you if you decide to publish, the survey results would be interesting also to our group.

Kind Regards,

Wendy

From: Maria Aamir [mailto:drmaria75@hotmail.com] Sent: Thursday, June 25 7:03 AM To: Cancer Education Program Subject: [EXTERNAL] Permission to use cancer needs survey for the PhD dissertation

Hi,

I am writing you to ask permission if I can use the cancer needs survey developed by PeterMac Mayo clinic to be used to assess cancer patients needs in the UAE (in a hospital setting). I am not intended to change any of the items apart from nationality classification into local emiratis and non-local expatriates.



Kindly if I can have some information about the internal validity of the survey how it was validated and what is the validity of it while developing the instrument (attached). Kindly if you could get me in touch with the authors or who could provide me the above information along with the permission to use it.

I look forward to obtain the approval to use it for my PhD studies.

Dr. Maria Aamir

+971508550292

My professional profile can be accessed through the below link:

https://www.linkedin.com/in/dr-maria-aamir-89b70415